

ON THE BEILINSON-BLOCH-KATO CONJECTURE FOR RANKIN-SELBERG MOTIVES

Yifeng Liu, Yichao Tian, Xiao Liang, Wei Zhang, Xinwen Zhu

▶ To cite this version:

Yifeng Liu, Yichao Tian, Xiao Liang, Wei Zhang, Xinwen Zhu. ON THE BEILINSON-BLOCH-KATO CONJECTURE FOR RANKIN-SELBERG MOTIVES. 2019. hal-02425406

HAL Id: hal-02425406 https://hal.science/hal-02425406

Preprint submitted on 30 Dec 2019

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.

ON THE BEILINSON–BLOCH–KATO CONJECTURE FOR RANKIN–SELBERG MOTIVES

YIFENG LIU, YICHAO TIAN, LIANG XIAO, WEI ZHANG, AND XINWEN ZHU

ABSTRACT. In this article, we study the Beilinson–Bloch–Kato conjecture for motives corresponding to the Rankin–Selberg product of conjugate self-dual automorphic representations, within the framework of the Gan–Gross–Prasad conjecture. We show that if the central critical value of the Rankin–Selberg *L*-function does not vanish, then the Bloch–Kato Selmer group with coefficients in a favorable field of the corresponding motive vanishes. We also show that if the class in the Bloch–Kato Selmer group constructed from certain diagonal cycle does not vanish, which is conjecturally equivalent to the nonvanishing of the central critical first derivative of the Rankin–Selberg *L*-function, then the Bloch–Kato Selmer group is of rank one.

Contents

1. Introduction	3
1.1. Main results	3
1.2. Road map for the article	5
1.3. Notations and conventions	8
Acknowledgements	10
2. Galois cohomology and Selmer groups	11
2.1. Preliminaries on adic modules	11
2.2. Local Galois cohomology	12
2.3. Some Galois-theoretical lemmas	14
2.4. Reduction of Selmer groups	16
2.5. Localization of Selmer groups	18
2.6. A Rankin–Selberg example	20
3. Preliminaries on hermitian structures	22
3.1. Hermitian spaces and unitary Hecke algebras	22
3.2. Unitary Shimura varieties and sets	25
3.3. Generalized CM type and reflexive closure	28
3.4. Unitary abelian schemes	29
3.5. A CM moduli scheme	34
4. Unitary moduli schemes: smooth case	36
4.1. Construction of moduli schemes	36
4.2. Basic correspondence on special fiber	39
4.3. Source of basic correspondence and Tate cycles	43
4.4. Functoriality under special morphisms	47
4.5. Second geometric reciprocity law	53
5. Unitary moduli schemes: semistable case	55
5.1. Construction of moduli schemes	56
5.2. Basic correspondence on balloon stratum	59

2010 Mathematics Subject Classification. 11G05, 11G18, 11G40, 11R34.

Date: December 23, 2019.

5.3. Basic correspondence on ground stratum	62
5.4. Basic correspondence on link stratum	68
5.5. Cohomology of link stratum	70
5.6. Intersection on ground stratum	72
5.7. Incidence maps on ground stratum	78
5.8. Weight spectral sequence	83
5.9. Special results in the rank 3 case	87
5.10. Functoriality under special morphisms	90
5.11. First geometric reciprocity law	99
6. Tate classes and arithmetic level raising	107
6.1. Preliminaries on automorphic representations	107
6.2. Tate classes in the odd rank case	110
6.3. Arithmetic level raising in the even rank case	112
6.4. Proof of Theorem 6.3.4	116
7. Explicit reciprocity laws for Rankin–Selberg motives	121
7.1. Setup for automorphic Rankin–Selberg motives	121
7.2. First explicit reciprocity law	121
7.3. Second explicit reciprocity law	128
8. Proof of main theorems	131
8.1. Admissible primes for coefficient field	131
8.2. Main theorem in the Selmer rank 0 case	134
8.3. Main theorem in the Selmer rank 1 case	139
Appendix A. Unitary Deligne–Lusztig varieties	143
A.1. Unitary Deligne–Lusztig varieties in the smooth case	143
A.2. Unitary Deligne–Lusztig varieties in the semistable case	145
Appendix B. Computation in Hecke algebras	149
B.1. Characters of the dual group	149
B.2. Two Hecke algebras	151
B.3. Enumeration of Hecke operators in the even rank case	154
B.4. Enumeration of Hecke operators in the odd rank case	158
Appendix C. Some representation theory for unitary groups	162
C.1. Local base change for unitary groups	162
C.2. Tate–Thompson representations	163
C.3. Results from endoscopic classification	170
Appendix D. Some trace formulae argument	171
D.1. Vanishing of cohomology off middle degree	171
D.2. Dimension of old forms	173
Appendix E. Deformation of Galois representations	176
E.1. Extension of essentially conjugate self-dual representations	177
E.2. Deformation problems	178
E.3. Fontaine–Laffaille deformations	180
E.4. Representations of the tame group	184
E.5. Minimally ramified deformations	188
E.6. Level-raising deformations	196
E.7. An almost minimal $R=T$ theorem	198
E.8. Rigidity of automorphic Galois representations	205
References	209

1. INTRODUCTION

In this article, we study the Beilinson–Bloch–Kato conjecture for motives corresponding to the Rankin–Selberg product of conjugate self-dual automorphic representations of $\operatorname{GL}_n(\mathbb{A}_F) \times$ $\operatorname{GL}_{n+1}(\mathbb{A}_F)$ for a CM number field F, within the framework of the Gan–Gross–Prasad conjecture [GGP12] for the pair of unitary groups $U(n) \times U(n+1)$. For the background on the Beilinson– Bloch–Kato conjecture, which is a generalization of the famous Birch and Swinnerton-Dyer conjecture from elliptic curves to higher dimensional algebraic varieties, we refer to the introduction of [Liu16].

1.1. Main results. Let F/F^+ be a totally imaginary quadratic extension of a totally real number field. We first state one of our main results that is least technical to understand.

Theorem 1.1.1 (Corollary 8.2.5). Let $n \ge 2$ be an integer. Let A and A' be two modular elliptic curves over F^+ such that $\operatorname{End}(A_{\overline{F}}) = \operatorname{End}(A'_{\overline{F}}) = \mathbb{Z}$. Suppose that

- (a) $A_{\overline{F}}$ and $A'_{\overline{F}}$ are not isogenous to each other;
- (b) both $\operatorname{Sym}^{n-1} A$ and $\operatorname{Sym}^n A'$ are modular; and
- (c) $[F^+:\mathbb{Q}] > 1$ if $n \ge 3$.

If the (central critical) L-value $L(n, \operatorname{Sym}^{n-1} A_F \times \operatorname{Sym}^n A'_F)$ does not vanish, then the Bloch-Kato Selmer group

$$\mathrm{H}^{1}_{f}(F, \mathrm{Sym}^{n-1}\mathrm{H}^{1}_{\mathrm{\acute{e}t}}(A_{\overline{F}}, \mathbb{Q}_{\ell}) \otimes_{\mathbb{Q}_{\ell}} \mathrm{Sym}^{n}\mathrm{H}^{1}_{\mathrm{\acute{e}t}}(A'_{\overline{F}}, \mathbb{Q}_{\ell})(n))$$

vanishes for all but finitely many rational primes ℓ .

Remark 1.1.2. The finite set of rational primes ℓ that are excluded in Theorem 1.1.1 can be effectively bounded. We now explain the three conditions in Theorem 1.1.1.

- (a) is necessary for ℓ to satisfy (L3) and (L5) in Definition 8.1.1. Otherwise, there might be no rational primes ℓ satisfying (L3) and (L5).
- (b) is necessary since our approach only applies to Galois representations arising from automorphic representations. We summarise the current knowledge on the modularity of symmetric powers of elliptic curves in Remark 8.2.6.
- (c) is necessary only for technical reasons. First, we do not know Hypothesis 3.2.9, which concerns cohomology of unitary Shimura varieties, yet for $N \ge 4$ if $F^+ = \mathbb{Q}$. Second, we do not have (an appropriate replacement for) Theorem D.1.3, a result generalizing [CS17], when $F^+ = \mathbb{Q}$ since the corresponding Shimura variety is not proper. Indeed, as long as we have these results as expected, (c) can be lifted.

Theorem 1.1.1 is a special case of a more general result concerning the Bloch–Kato Selmer groups of Galois representations associated to conjugate self-dual automorphic representations. To reduce the burden of long and technical terminology in the future, we first introduce the following definition, which will serve for the entire article.

Definition 1.1.3. We say that a complex representation Π of $GL_N(\mathbb{A}_F)$ with $N \geq 1$ is relevant if

- (1) Π is an irreducible cuspidal automorphic representation;
- (2) $\Pi \circ \mathbf{c} \simeq \Pi^{\vee}$, where $\mathbf{c} \in \operatorname{Gal}(F/F^+)$ is the complex conjugation;
- (3) for every archimedean place τ of F, Π_{τ} is isomorphic to the (irreducible) principal series representation induced by the characters ($\arg^{N-1}, \arg^{N-3}, \ldots, \arg^{3-N}, \arg^{1-N}$), where $\arg: \mathbb{C}^{\times} \to \mathbb{C}^{\times}$ is the *argument character* defined by the formula $\arg(z) \coloneqq z/\sqrt{z\overline{z}}$.

Now we can state our main result in the context of automorphic representations, of which Theorem 1.1.1 is a special case. Till the end of the next subsection, we will take an integer $n \ge 2$, and denote by n_0 and n_1 the unique even and odd numbers in $\{n, n+1\}$, respectively. **Theorem 1.1.4** (Theorem 8.2.1). Let Π_0 and Π_1 be relevant representations of $\operatorname{GL}_{n_0}(\mathbb{A}_F)$ and $\operatorname{GL}_{n_1}(\mathbb{A}_F)$, respectively. Let $E \subseteq \mathbb{C}$ be a strong coefficient field of both Π_0 and Π_1 (Definition 3.2.5). Suppose $[F^+:\mathbb{Q}] > 1$ if $n \geq 3$. If $L(\frac{1}{2}, \Pi_0 \times \Pi_1) \neq 0$, then for all admissible primes λ of E with respect to (Π_0, Π_1) , the Bloch–Kato Selmer group $\operatorname{H}^1_f(F, \rho_{\Pi_0,\lambda} \otimes_{E_\lambda} \rho_{\Pi_1,\lambda}(n))$ vanishes. Here, $\rho_{\Pi_\alpha,\lambda}$ is the Galois representation of F with coefficients in E_λ associated to Π_α for $\alpha = 0, 1$, as described in Proposition 3.2.4 and Definition 3.2.5.

In fact, Theorem 8.2.1 is slightly stronger than the one stated here.

Remark 1.1.5. The notion of admissible primes appeared in Theorem 1.1.4 is introduced in Definition 8.1.1, which consists of a long list of assumptions, some of which are rather technical. Here, we would like to comment on the essence of these assumptions.

- (L1) is elementary and excludes only finitely many primes λ .
- (L2) is elementary and excludes only finitely many primes λ .
- (L3) is expected to hold for every prime λ if and only if the (conjectural) automorphic product $\Pi_0 \boxtimes \Pi_1$, which is an irreducible admissible representation of $\operatorname{GL}_{n(n+1)}(\mathbb{A}_F)$, remains cuspidal.
- (L4) is expected to hold for all but finitely many primes λ .
- (L5) is basically saying that, under (L4), the image of the pair of residual Galois representations $(\bar{\rho}_{\Pi_0,\lambda}, \bar{\rho}_{\Pi_1,\lambda})$ contains an element of a particular form. It is expected to hold for all but finitely many primes λ if the two automorphic representations Π_0 and Π_1 are not correlated in some manner. For example, when n = 2, we expect that as long as Π_1 is not an automorphic twist of Sym² Π_0 after any base change, then (L5) holds for all but finitely many primes λ .
- (L6) is a technical assumption that is only used in the argument of an R=T theorem concerning Galois deformations in Appendix E. It is expected to hold for all but finitely many primes λ (see Conjecture E.8.1 and Theorem E.8.4).
- (L7) is a technical assumption for the vanishing of certain Hecke localized cohomology of unitary Shimura varieties off middle degree. In fact, when $[F^+ : \mathbb{Q}] > 1$, (L7) holds for all but finitely many primes λ by Corollary D.1.4.

In fact, we have dedicated ourselves to obtaining the following family of abstract examples in which all but finitely many primes are admissible. Note that neither the following theorem nor Theorem 1.1.1 implies the other.

Theorem 1.1.6 (Corollary 8.2.3). Let Π_0 , Π_1 , and E be as in Theorem 1.1.4. Suppose that

- (a) there exists a very special inert prime \mathfrak{p} of F^+ (Definition 3.3.4) such that $\Pi_{0,\mathfrak{p}}$ is Steinberg, and $\Pi_{1,\mathfrak{p}}$ is unramified whose Satake parameter contains 1 exactly once¹;
- (b) for $\alpha = 0, 1$, there exists a nonarchimedean place w_{α} of F such that $\Pi_{\alpha, w_{\alpha}}$ is supercuspidal; (c) $[F^+ : \mathbb{Q}] > 1$ if $n \ge 3$.

If $L(\frac{1}{2}, \Pi_0 \times \Pi_1) \neq 0$, then for all but finitely many primes λ of E, the Bloch-Kato Selmer group $\mathrm{H}^1_f(F, \rho_{\Pi_0,\lambda} \otimes_{E_\lambda} \rho_{\Pi_1,\lambda}(n))$ vanishes.

Remark 1.1.7. In (a) of Theorem 1.1.6, if the CM field F is Galois or contains an imaginary quadratic field, then a very special inert prime of F^+ is simply a prime of F^+ that is inert in F, of degree 1 over \mathbb{Q} , whose underlying rational prime is odd and unramified in F.

Now we state our result in the (Selmer) rank 1 case. Let Π_0 and Π_1 be relevant representations of $\operatorname{GL}_{n_0}(\mathbb{A}_F)$ and $\operatorname{GL}_{n_1}(\mathbb{A}_F)$, respectively. Let $E \subseteq \mathbb{C}$ be a strong coefficient field of both Π_0 and Π_1

¹Note that the Satake parameter of $\Pi_{1,\mathfrak{p}}$ has to contain 1 at least once by Definition 1.1.3(2).

(Definition 3.2.5). Suppose that the global epsilon factor of $\Pi_0 \times \Pi_1$ is -1. Then the Beilinson– Bloch–Kato conjecture predicts that if $L'(\frac{1}{2}, \Pi_0 \times \Pi_1) \neq 0$, then the Bloch–Kato Selmer group $\mathrm{H}^1_f(F, \rho_{\Pi_0,\lambda} \otimes_{E_\lambda} \rho_{\Pi_1,\lambda}(n))$ has rank 1. However, what we can prove now is half of this implication. Namely, for every prime λ of E, we will construct explicitly an element Δ_{λ} in (the direct sum of finitely many copies of) $\mathrm{H}^1_f(F, \rho_{\Pi_0,\lambda} \otimes_{E_\lambda} \rho_{\Pi_1,\lambda}(n))$ in Subsection 8.3 (more precisely, Δ_{λ} is the class in (8.9)). In fact, by Conjecture 8.3.1 and Beilinson's conjecture on the injectivity of the ℓ -adic Abel–Jacobi map, the nonvanishing of Δ_{λ} is equivalent to the nonvanishing of $L'(\frac{1}{2}, \Pi_0 \times \Pi_1)$. Then our theorem in the rank 1 case reads as follows.

Theorem 1.1.8 (Theorem 8.3.2). Let Π_0 and Π_1 be relevant representations of $\operatorname{GL}_{n_0}(\mathbb{A}_F)$ and $\operatorname{GL}_{n_1}(\mathbb{A}_F)$, respectively. Let $E \subseteq \mathbb{C}$ be a strong coefficient field of both Π_0 and Π_1 (Definition 3.2.5). Suppose $[F^+ : \mathbb{Q}] > 1$ if $n \geq 3$. For all admissible primes λ of E with respect to (Π_0, Π_1) , if $\Delta_{\lambda} \neq 0$, then the Bloch-Kato Selmer group $\operatorname{H}^1_f(F, \rho_{\Pi_0,\lambda} \otimes_{E_{\lambda}} \rho_{\Pi_1,\lambda}(n))$ is of dimension 1 over E_{λ} .

In fact, Theorem 8.3.2 is slightly stronger than the one stated here. We also have an analogue of Theorem 1.1.6 in the rank 1 case, whose statement we omit.

Remark 1.1.9. In both Theorem 1.1.4 and Theorem 1.1.8, the assumption that $[F^+ : \mathbb{Q}] > 1$ if $n \geq 3$ can be lifted once Hypothesis 3.2.9 is known for $N \geq 4$ when $F^+ = \mathbb{Q}$.

1.2. Road map for the article. The very basic idea of bounding Selmer groups as in our main theorems follows from Kolyvagin [Kol90], namely, we construct a system of torsion Galois cohomology classes serving as annihilators of (reduction of) Selmer groups. However, our system is not a generalization of the Euler–Kolyvagin system originally constructed by Kolyvagin. Instead, our system is constructed via level-raising congruences, which was first introduced by Bertolini and Darmon in the case of Heegner points in the study of certain Iwasawa main conjecture of elliptic curves [BD05]. The first example where such level-raising system was used to bound Selmer groups beyond the Heegner point case was performed by one the us in [Liu16], for the so-called twisted triple product automorphic motives. In the sequels [Liu19] and [LT], the case of the so-called cubic triple product automorphic motives was also studied. From this point of view, our current article is a vast generalization of the previous results mentioned above. We have to point out that, although the fundamental ideas do not vary too much, the level of difficulty of realizing all the steps in our current work is tremendously higher than all of the past ones. In fact, in order to study the arithmetic level-raising for unitary groups of even ranks at least 4, we have to use the theory of Galois deformations, which seems to be a new application of the latter.

The following is a road map for reading the main part of the article, where we indicate the need from the five appendices in the parentheses.



The proof of Theorem 1.1.8 is based on the proof of Theorem 1.1.4. We may regard the transition from the rank 0 case to the rank 1 case as an induction step. As seen from the road map, for the rank 0 case alone, Section 4, Subsection A.1, Subsection 7.3, and, of course, Subsection 8.3 are not needed. However, we strongly recommend the readers to go through Section 4 even if they are only interested in the rank 0 case, as Section 4 is an appropriate warm-up for reading Section 5, which is parallel but much more complicated.

In what follows, we explain the main steps in the proof of Theorem 1.1.4. Some of the notations in the rest of this subsection are *ad hoc*, only for the purpose of explaining ideas, hence will be obsolete or differ from the main text.

The initial step (which although will not appear until Subsection 8.2) is to translate the condition that $L(\frac{1}{2}, \Pi_0 \times \Pi_1) \neq 0$ into a more straightforward statement. This is exactly the content of the global Gan–Gross–Prasad conjecture [GGP12]. In fact, as stated in Lemma 8.2.2, we may construct a pair of hermitian spaces $(V_n^{\circ}, V_{n+1}^{\circ})$ over F (with respect to F/F^+) in which V_n° is totally positive definite of rank n, and $V_{n+1}^{\circ} = V_n^{\circ} \oplus F \cdot 1$ where 1 has norm 1. For $\alpha = 0, 1$, put $\mathrm{Sh}(V_{n_{\alpha}}^{\circ}) \coloneqq \mathrm{U}(V_{n_{\alpha}}^{\circ})(F^+) \setminus \mathrm{U}(V_{n_{\alpha}}^{\circ})(\mathbb{A}_{F^+}^{\infty})$ as a Shimura (pro-)set. We may further find cuspidal automorphic representations π_0 and π_1 contained in the space of locally constant functions on $\mathrm{Sh}(V_{n_0}^{\circ})$ and $\mathrm{Sh}(V_{n_1}^{\circ})$ satisfying $\mathrm{BC}(\pi_0) \simeq \Pi_0$ and $\mathrm{BC}(\pi_1) \simeq \Pi_1$, respectively, such that

(1.1)
$$\mathcal{P}(f_0, f_1) \coloneqq \int_{\operatorname{Sh}(\mathbf{V}_n^\circ)} f_0(h) f_1(h) \mathrm{d}h \neq 0$$

for some $f_0 \in \pi_0$ and $f_1 \in \pi_1$ valued in O_E . Such result was first obtained by one of us [Zha14] under some local restrictions. Those restrictions are all lifted till very recently through some new techniques in the study of trace formulae [BPLZZ]. In what follows, we will fix open compact subgroups of $U(V_{n_0}^{\circ})(\mathbb{A}_{F^+}^{\infty})$ and $U(V_{n_1}^{\circ})(\mathbb{A}_{F^+}^{\infty})$ that fix f_0 and f_1 , respectively, and will carry them implicitly in the notation.

The next step is to bring the set $Sh(V_{n_{\alpha}}^{\circ})$ into arithmetic geometry so that the period (1.1) can be related to certain Galois cohomology classes. Now we choose a special inert prime **p** of F^+ (see Definition 3.3.4) with sufficiently large underlying rational prime p, so that all data appeared so far are unramified above p. For $\alpha = 0, 1$, we attach to $V_{n_{\alpha}}^{\circ}$ canonically a strictly semistable scheme $\mathbf{M}_{\mathfrak{p}}(\mathbf{V}_{n_{\alpha}}^{\circ})$ over Spec \mathbb{Z}_{p^2} of relative dimension $n_{\alpha} - 1$, whose complex generic fiber is non-canonically isomorphic to the disjoint union of finitely many Shimura varieties attached to the *nearby* hermitian space of $V_{n_{\alpha}}^{\circ}$ by changing local components at \mathfrak{p} and one archimedean place. Moreover, we can write its special fiber $M_{\mathfrak{p}}(V_{n_{\alpha}}^{\circ})$ over $\operatorname{Spec} \mathbb{F}_{p^2}$ as a union of $M_{\mathfrak{p}}^{\circ}(V_{n_{\alpha}}^{\circ})$ and $M_{\mathfrak{p}}^{\bullet}(V_{n_{\alpha}}^{\circ})$, in which $M_{\mathfrak{p}}^{\circ}(V_{n_{\alpha}}^{\circ})$ is geometrically a $\mathbb{P}^{n_{\alpha}-1}$ -fibration over the Shimura set $\operatorname{Sh}(V_{n_{\alpha}}^{\circ})$. However, the reality is much more intricate, as the geometry of the other stratum $M^{\bullet}_{\mathfrak{p}}(V^{\circ}_{n_{\alpha}})$, which is rather mysterious, will also involve in the later computation. In fact, one key effort we pay is to show that only the basic locus of the stratum $M^{\bullet}_{\mathfrak{p}}(V^{\circ}_{n_{\alpha}})$ will play a role in the computation. For the basic locus, we show that its normalization is geometrically a fibration over the Shimura set $\operatorname{Sh}(\operatorname{V}_{n_{\alpha}}^{\circ})$ (but with a slightly different level structure at \mathfrak{p}) by certain Deligne-Lusztig varieties of dimension $r_{\alpha} \coloneqq \lfloor \frac{n_{\alpha}}{2} \rfloor$, introduced in Subsection A.2. The study of various geometric aspects of the scheme $\mathbf{M}_{\mathfrak{p}}(\mathbf{V}_{n_{\alpha}}^{\circ})$, including its associated Rapoport-Zink spectral sequence and its functorial behavior from n to n + 1, will be carried out in Section 5.

The automorphic input will be thrown into the scheme $\mathbf{M}_{\mathfrak{p}}(\mathbf{V}_{n_{\alpha}}^{\circ})$ from the third step, in Section 6, where we study the local Galois cohomology of certain cohomology of $\mathbf{M}_{\mathfrak{p}}(\mathbf{V}_{n_{\alpha}}^{\circ})$ localized at some Hecke ideals. More precisely, we fix an admissible prime λ of E with respect to (Π_0, Π_1) , and denote by O_{λ} and k_{λ} the ring of integers and the residue field of E_{λ} , respectively. For $\alpha = 0, 1$, the Satake parameters of Π_{α} induce a homomorphism $\phi_{\alpha} \colon \mathbb{T}_{n_{\alpha}} \to k_{\lambda}$ with kernel \mathfrak{m}_{α} , where $\mathbb{T}_{n_{\alpha}}$ is certain abstract spherical Hecke algebra for unitary groups of rank n_{α} . When $\alpha = 0$ (resp. $\alpha = 1$),

we need to study the singular (resp. unramified) part of the local Galois cohomology

(1.2)
$$\mathrm{H}^{1}(\mathbb{Q}_{p^{2}},\mathrm{H}^{n_{\alpha}-1}_{\mathfrak{T}}(\overline{\mathrm{M}}_{\mathfrak{p}}(\mathrm{V}^{\circ}_{n_{\alpha}}),\mathrm{R}\Psi O_{\lambda}(r_{\alpha}))_{\mathfrak{m}_{\alpha}}),$$

where $\overline{\mathrm{M}}_{\mathfrak{p}}(\mathrm{V}_{n_{\alpha}}^{\circ}) \coloneqq \mathrm{M}_{\mathfrak{p}}(\mathrm{V}_{n_{\alpha}}^{\circ}) \otimes_{\mathbb{F}_{p^2}} \overline{\mathbb{F}}_p$, and $\mathrm{H}_{\mathfrak{T}}$ denotes certain invariant part of the étale cohomology (a subtlety that can be ignored at this moment). The question boils down to the arithmetic levelraising phenomenon (resp. existence of Tate cycles) when $\alpha = 0$ (resp. $\alpha = 1$). However, in both cases, we have to rely on the recent progress on the Tate conjecture for Shimura varieties achieved by some of us [XZ]. Now we would like to continue the discussion on the case where $\alpha = 0$, since it is more interesting and more involved, and omit the case where $\alpha = 1$. The first key point is to figure out the correct condition so that the level-raising phenomenon (namely, from unramified to mildly ramified at the place \mathfrak{p}) happens on the cohomology (1.2) in a way that can be understood: we say that \mathfrak{p} is a *level-raising prime* with respect to λ if $\ell \nmid p(p^2 - 1)$, and the mod λ Satake parameter of $\Pi_{0,\mathfrak{p}}$ contains the pair $\{p, p^{-1}\}$ exactly once and does not contain the pair $\{-1, -1\}$.

(1.3)
$$\mathrm{H}^{1}_{\mathrm{sing}}(\mathbb{Q}_{p^{2}},\mathrm{H}^{n_{0}-1}_{\mathfrak{T}}(\overline{\mathrm{M}}_{\mathfrak{p}}(\mathrm{V}^{\circ}_{n_{0}}),\mathrm{R}\Psi O_{\lambda}(r_{0}))/\mathfrak{m}_{0})\simeq O_{\lambda}[\mathrm{Sh}(\mathrm{V}^{\circ}_{n_{0}})]/\mathfrak{m}_{0}$$

of O_{λ} -modules of finite length. Note that by our condition on \mathfrak{p} , the right-hand side of (1.3) is nonvanishing, which implies that the left-hand side is also nonvanishing; in other words, we see the level-raising phenomenon in $\mathrm{H}^{n_0-1}_{\mathfrak{T}}(\overline{\mathrm{M}}_{\mathfrak{p}}(\mathrm{V}^{\circ}_{n_0}), \mathrm{R}\Psi O_{\lambda}(r_0))$. The proof of (1.3) is the technical heart of this article (for example, it uses materials from all of the five appendices). Through studying the geometry and intersection theory on the special fiber $\mathrm{M}_{\mathfrak{p}}(\mathrm{V}^{\circ}_{n_0})$ in Section 5 and some of the appendices, we can conclude that $O_{\lambda}[\mathrm{Sh}(\mathrm{V}^{\circ}_{n_0})]/\mathfrak{m}_0$ is canonically a subquotient of $\mathrm{H}^1_{\mathrm{sing}}(\mathbb{Q}_{p^2},\mathrm{H}^{n_0-1}_{\mathfrak{T}}(\overline{\mathrm{M}}_{\mathfrak{p}}(\mathrm{V}^{\circ}_{n_0}),\mathrm{R}\Psi O_{\lambda}(r_0))/\mathfrak{m}_0)$. Thus, it remains to show that the two sides of (1.3) have the same cardinality. For this, we use the theory of Galois deformations. We construct a global Galois deformation O_{λ} -algebra $\mathrm{R}^{\mathrm{mix}}$ with two quotient algebras $\mathrm{R}^{\mathrm{unr}}$ and $\mathrm{R}^{\mathrm{ram}}$, together with a natural $\mathrm{R}^{\mathrm{unr}}$ -module $\mathrm{H}^{\mathrm{unr}}$ and a natural $\mathrm{R}^{\mathrm{ram}}$ -module $\mathrm{H}^{\mathrm{ram}}$. They satisfy the following relation: if we put $\mathrm{R}^{\mathrm{cong}} := \mathrm{R}^{\mathrm{unr}} \otimes_{\mathrm{R}^{\mathrm{mix}}} \mathrm{R}^{\mathrm{ram}}$, which is an Artinian O_{λ} -algebra, then we have natural isomorphisms

$$\begin{aligned} & \mathsf{H}^{\mathrm{unr}} \otimes_{\mathsf{R}^{\mathrm{unr}}} \mathsf{R}^{\mathrm{cong}} \simeq O_{\lambda}[\mathrm{Sh}(\mathrm{V}_{n_{0}}^{\circ})]/\mathfrak{m}_{0}, \\ & \mathsf{H}^{\mathrm{ram}} \otimes_{\mathsf{R}^{\mathrm{ram}}} \mathsf{R}^{\mathrm{cong}} \simeq \mathrm{H}^{1}_{\mathrm{sing}}(\mathbb{Q}_{p^{2}}, \mathrm{H}^{n_{0}-1}_{\mathfrak{T}}(\overline{\mathrm{M}}_{\mathfrak{p}}(\mathrm{V}_{n_{0}}^{\circ}), \mathrm{R}\Psi O_{\lambda}(r_{0}))/\mathfrak{m}_{0}). \end{aligned}$$

Thus, we only need to show that H^{unr} and H^{ram} are both finite free over R^{unr} and R^{ram} , respectively, of the same rank. The finite-freeness follows from an $\mathsf{R}=\mathsf{T}$ theorem, proved in Appendix E. It is worth pointing out that our $\mathsf{R}=\mathsf{T}$ theorem is over the initial base field F, that is, we do not take a favourable CM extension of F as people usually do like in [CHT08, Tho12], for example; in particular, we have to deal with certain ramification at nonsplit places of F^+ . The comparison of ranks can be performed over E_{λ} , which turns out to be an automorphic problem and is solved in Subsection 6.4 based on Subsection D.2. Therefore, we obtain (1.3). In practice, we also need a mod λ^m version of (1.3).

The fourth step is to merge the study of (1.2) for n_0 and n_1 together, to obtain the so-called *first explicit reciprocity law* for the Rankin–Selberg product of Galois representations. As an application, we construct a system of torsion Galois cohomology classes whose image in the singular part of the local Galois cohomology at \mathbf{p} of the product Galois representation is controlled by the period integral (1.1). This step is sort of routine, once we have enough knowledge on (1.2); it is completed in Subsection 7.2.

The final step of the proof of Theorem 1.1.4 will be performed in Subsection 8.2, where we use the system of torsion Galois cohomology classes constructed in the previous step, together with some Galois theoretical facts from Section 2, to bound the Selmer group, which is possible due to the nonvanishing of (1.1).

1.3. Notations and conventions. In this subsection, we setup some common notations and conventions for the entire article, including appendices, unless otherwise specified. The notations in the previous two subsections will not be relied on from this moment, and should not be kept for further reading.

Generalities:

- O Denote by $\mathbb{N} = \{0, 1, 2, 3, ...\}$ the monoid of nonnegative integers.
- O We only apply the operation $\sqrt{-}$ to positive real numbers, which takes values in positive real numbers as well.
- **O** For a set S, we denote by $\mathbb{1}_S$ the characteristic function of S.
- **O** The eigenvalues or generalized eigenvalues of a matrix over a field k are counted with multiplicity (namely, dimension of the corresponding eigenspace or generalized eigenspace); in other words, they form a multi-subset of an algebraic extension of k.
- For every rational prime p, we fix an algebraic closure $\overline{\mathbb{Q}}_p$ of \mathbb{Q}_p with the residue field $\overline{\mathbb{F}}_p$. For every integer $r \geq 1$, we denote by \mathbb{Q}_{p^r} the subfield of $\overline{\mathbb{Q}}_p$ that is an unramified extension of \mathbb{Q}_p of degree r, by \mathbb{Z}_{p^r} its ring of integers, and by \mathbb{F}_{p^r} its residue field.
- O For a nonarchimedean place v of a number field K, we write ||v|| for the cardinality of the residue field of K_v .
- We use standard notations from the category theory. The category of sets is denoted by Set. For a category \mathfrak{C} , we denote by $\mathfrak{C}^{\mathrm{op}}$ its opposite category, and denote by $\mathfrak{C}_{/A}$ the category of morphisms to A for an object A of \mathfrak{C} . For another category \mathfrak{D} , we denote by $\mathsf{Fun}(\mathfrak{C}, \mathfrak{D})$ the category of functors from \mathfrak{C} to \mathfrak{D} . In particular, we denote by $\mathsf{PC} := \mathsf{Fun}(\mathfrak{C}^{\mathrm{op}}, \mathsf{Set})$ the category of presheaves on \mathfrak{C} , which contains \mathfrak{C} as a full subcategory by the Yoneda embedding. Isomorphisms in a category will be indicated by \simeq .
- O All rings (but not algebras) are commutative and unital. For a ring L and an L-algebra L', we denote by Mod(L') the category of left L'-modules.
- **O** If a base ring is not specified in the tensor operation \otimes , then it is \mathbb{Z} .
- O For a ring L and a set S, denote by L[S] the L-module of L-valued functions on S of finite support.

Definition 1.3.1. Let K be a subfield of \mathbb{C} . We say that an intermediate extension $K \subseteq K' \subseteq \mathbb{C}$ is K-normal if K'/K is finite, and every automorphism in $\operatorname{Aut}(\mathbb{C}/K)$ stabilizes K'.

Algebraic geometry:

- O We denote by the category of schemes by Sch and its full subcategory of locally Noetherian schemes by Sch'. For a scheme S (resp. Noetherian scheme S), we denote by Sch_{/S} (resp. Sch_{/S}) the category of S-schemes (resp. locally Noetherian S-schemes). If S = Spec R is affine, we also write Sch_{/R} (resp. Sch_{/R}) for Sch_{/S} (resp. Sch_{/S}).
- O For a scheme X over an affine scheme Spec R and a commutative R-algebra S, we write $X \otimes_R S$ or even X_S for $X \times_{\text{Spec } R} \text{Spec } S$.
- O For a scheme S in characteristic p for some rational prime p, we denote by $\sigma: S \to S$ the absolute p-power Frobenius morphism. For a perfect field κ of characteristic p, we denote by $W(\kappa)$ its Witt ring, and by abuse of notation, $\sigma: W(\kappa) \to W(\kappa)$ the canonical lifting of the p-power Frobenius map.
- O For a scheme S and a locally free \mathcal{O}_S -module \mathcal{V} of finite rank, we denote by $\mathbb{P}(\mathcal{V}) \to S$ the moduli scheme of quotient line bundles of \mathcal{V} over S, known as the *projective fibration* associated to \mathcal{V} .

- O For a scheme S and (sheaves of) \mathcal{O}_S -modules \mathcal{F} and \mathcal{G} , we denote by $\mathcal{H}om(\mathcal{F},\mathcal{G})$ the quasi-coherent sheaf of \mathcal{O}_S -linear homomorphisms from \mathcal{F} to \mathcal{G} .
- O For two positive integers r, s, we denote by $M_{r,s}$ the scheme of r-by-s matrices, and put $M_r := M_{r,r}$ for short; we also denote by $\operatorname{GL}_r \subseteq M_r$ the subscheme of invertible r-by-r matrices. Then GL_1 is simply the multiplicative group $\mathbf{G}_m := \mathbb{Z}[T, T^{-1}]$; but we will distinguish between GL_1 and \mathbf{G}_m according to the context.
- For a number field K, a commutative group scheme $G \to S$ equipped with an action by O_K over some base scheme S, and an ideal $\mathfrak{a} \subset O_K$, we denote by $G[\mathfrak{a}]$ the maximal closed subgroup scheme of G annihilated by all elements in \mathfrak{a} .
- By a *coefficient ring* for étale cohomology, we mean either a finite ring, or a finite extension of \mathbb{Q}_{ℓ} , or the ring of integers of a finite extension of \mathbb{Q}_{ℓ} . In the latter two cases, we regard the étale cohomology via suitable ℓ -adic formalism. We say that a coefficient ring L is *n*-coprime for a positive integer n if n is invertible in L in the first case, and $\ell \nmid n$ in the latter two cases.

Group theory: Let G and $\tilde{\Gamma}$ be groups, and Γ a subgroup of $\tilde{\Gamma}$. Let L be a ring.

- O Denote by Γ^{ab} the maximal abelian quotient of Γ .
- O For a homomorphism $\rho: \Gamma \to \operatorname{GL}_r(L)$ for some $r \ge 1$, we denote by $\rho^{\vee}: \Gamma \to \operatorname{GL}_r(L)$ the contragredient homomorphism, which is defined by the formula $\rho^{\vee}(x) = {}^{\mathrm{t}}\rho(x)^{-1}$ for every $x \in \Gamma$.
- O For a homomorphism $\rho: \Gamma \to G$ and an element $\gamma \in \Gamma$ that normalizes Γ , we let $\rho^{\gamma}: \Gamma \to G$ be the homomorphism defined by $\rho^{\gamma}(x) = \rho(\gamma x \gamma^{-1})$ for every $x \in \Gamma$.
- O We say that two homomorphisms $\rho_1, \rho_2 \colon \Gamma \to G$ are conjugate if there exists an element $g \in G$ such that $\rho_1 = g \circ \rho_2 \circ g^{-1}$, that is, $\rho_1(x) = g\rho_2(x)g^{-1}$ for every $x \in \Gamma$.
- O The L-module L[G] is naturally an L-algebra, namely, the group algebra of G with coefficients in L.
- O Suppose that G is a locally compact and totally disconnected topological group. For an open compact subgroup K of G, the L-module $L[K \setminus G/K]$ (of bi-K-invariant compactly supported L-valued functions on G) is naturally an L-algebra, where the algebra structure is given by the composition of cosets. In particular, the unit element of $L[K \setminus G/K]$ is always $\mathbb{1}_K$.
- O For every integer $r \ge 1$, we denote by J_r the standard upper triangular nilpotent Jordan block

$$\begin{pmatrix} 0 & 1 & 0 & \cdots & 0 \\ 0 & 1 & \cdots & 0 \\ & \ddots & \ddots & \vdots \\ & & 0 & 1 \\ & & & & 0 \end{pmatrix}$$

or size r-by-r.

Combinatorics:

Notation 1.3.2. We recall the λ -analogue of binomial coefficients:

$$[0]_{\lambda} = 1, \quad [n]_{\lambda} = \frac{\lambda^n - 1}{\lambda - 1}, \quad [n]_{\lambda}! = [n]_{\lambda} \cdot [n - 1]_{\lambda} \cdots [1]_{\lambda}, \quad \begin{bmatrix} n \\ m \end{bmatrix}_{\lambda} = \frac{[n]_{\lambda}!}{[n - m]_{\lambda}! \cdot [m]_{\lambda}!}$$

for integers $0 \le m \le n$. For $r \ge 0$ and $q \in \mathbb{N}$, we put

$$\begin{cases} \mathbf{d}_{r,q} \coloneqq \sum_{\delta=0}^{r} (-1)^{\delta} (2\delta+1) q^{\delta(\delta+1)} \begin{bmatrix} 2r+1\\ r-\delta \end{bmatrix}_{-q}, \\ \mathbf{d}_{r,q}^{\bullet} \coloneqq \frac{1}{q+1} \left(\mathbf{d}_{r,q} + \frac{(-q)^{r+1}-1}{q+1} (q+1) (q^3+1) \cdots (q^{2r-1}+1) \right). \end{cases}$$

Ground fields:

- O Let $c \in Aut(\mathbb{C}/\mathbb{Q})$ be the complex conjugation.
- O Throughout the article, we fix a *subfield* $F \subseteq \mathbb{C}$ that is a number field and is stable under c; it is assumed to be a CM field except in Section 2.
- O Let $F^+ \subseteq F$ be the maximal subfield on which c acts by the identity.
- O Let \overline{F} be the Galois closure of F in \mathbb{C} . Put $\Gamma_F \coloneqq \operatorname{Gal}(\overline{F}/F)$ and $\Gamma_{F^+} \coloneqq \operatorname{Gal}(\overline{F}/F^+)$.
- O Denote by Σ_{∞} (resp. Σ_{∞}^+) the set of complex embeddings of F (resp. F^+) with $\tau_{\infty} \in \Sigma_{\infty}$ (resp. $\underline{\tau}_{\infty} \in \Sigma_{\infty}^+$) the default one. For $\tau \in \Sigma_{\infty}$, we denote by τ^{c} the its complex conjugation.
- O For every rational prime p, denote by Σ_p^+ the set of all p-adic places of F^+ .
- O Denote by Σ_{ram}^+ the union of Σ_p^+ for all p ramified in F.
- O Denote by η_{F/F^+} : $\Gamma_{F^+} \to \{\pm 1\}$ the character associated to the extension F/F^+ .
- O For every prime ℓ , denote by $\epsilon_{\ell} \colon \Gamma_{F^+} \to \mathbb{Z}_{\ell}^{\times}$ the ℓ -adic cyclotomic character.

For every place v of F^+ , we

- O put $F_v := F \otimes_{F^+} F_v^+$; and define $\delta(v)$ to be 1 (resp. 2) if v splits (resp. does not split) in F;
- O fix an algebraic closure \overline{F}_v^+ of F_v^+ containing \overline{F} ; and put $\Gamma_{F_v^+} \coloneqq \operatorname{Gal}(\overline{F}_v^+/F_v^+)$ as a subgroup of Γ_{F^+} ;
- O for a homomorphism r from Γ_{F^+} to another group, denote by r_v the restriction of r to the subgroup $\Gamma_{F^+_v}$.

For every nonarchimedean place w of F, we

- O identify the Galois group Γ_{F_w} with $\Gamma_{F_v^+} \cap \Gamma_F$ (resp. $c(\Gamma_{F_v^+} \cap \Gamma_F)c)$, where v is the underlying place of F^+ , if the embedding $F \hookrightarrow \overline{F}_v^+$ induces (resp. does not induce) the place w;
- O let $I_{F_w} \subseteq \Gamma_{F_w}$ be the inertia subgroup;
- O let κ_w be the residue field of F_w , and identify its Galois group Γ_{κ_w} with Γ_{F_w}/I_{F_w} ;
- O denote by $\phi_w \in \Gamma_{F_w}$ a lifting of the *arithmetic* Frobenius element in Γ_{κ_w} .

Definition 1.3.3. We say that two subsets Σ_1^+ and Σ_2^+ of nonarchimedean places of F^+ are strongly disjoint if there is no rational prime underlying places from both sets.

Acknowledgements. This article is the main outcome of an AIM SQuaREs project *Geometry* of Shimura varieties and arithmetic application to L-functions by the five authors from 2017 to 2019. We would like to express our sincere gratitude and appreciation to the American Institute of Mathematics for their constant and generous support of the project, and to the staff members at the AIM facility in San Jose, California for their excellent coordination and hospitality.

We would like to thank Sug Woo Shin and Yihang Zhu for the discussion concerning Hypothesis 3.2.9 and the endoscopic classification for unitary groups, Zipei Nie and Jun Su for suggesting a proof of a combinatorial lemma (Lemma B.3.3), Ana Caraiani and Peter Scholze for the discussion concerning Subsection D.1, and Kai-Wen Lan for the discussion concerning the reference [LS18].

The research of Y. L. is partially supported by the NSF grant DMS-1702019 and a Sloan Research Fellowship. The research of L. X. is partially supported by the NSF grant DMS-1502147 and DMS-1752703. The research of W. Z. is partially supported by the NSF grant DMS-1838118 and DMS-1901642. The research of X. Z. is partially supported by the NSF grant DMS-1902239.

2. Galois cohomology and Selmer groups

In this section, we make the Galois theoretical preparation for the proof of the main theorems. Most discussions in this section are generalizations from [Liu16, Liu19]. The material of this section will not be used until Section 6. In Subsection 2.1, we collect some lemmas on ℓ -adic modules with certain group actions. In Subsection 2.2, we study local Galois cohomology. In Subsection 2.3, we perform the discussion that is typical for Kolyvagin's type of argument. The Selmer group and its variant will be introduced in Subsection 2.4. In Subsection 2.5, we study localization of Selmer groups. In Subsection 2.6, we study an example related to the Rankin–Selberg product.

We will start from a more general setup in order to make the discussion applicable to the orthogonal case as well, which may be studied in the future. Thus, we fix a *subfield* $F \subseteq \mathbb{C}$ that is a number field, *not* necessarily CM.

We fix an odd rational prime ℓ that is unramified in F, and consider a finite extension $E_{\lambda}/\mathbb{Q}_{\ell}$, with the ring of integers O_{λ} and the maximal ideal λ of O_{λ} . Recall that $\epsilon_{\ell} \colon \Gamma_{F^+} \to \mathbb{Z}_{\ell}^{\times}$ is the ℓ -adic cyclotomic character.

2.1. Preliminaries on adic modules. Let Γ be a profinite group and L a commutative topological \mathbb{Z}_{ℓ} -algebra.

Notation 2.1.1. We denote by $\mathsf{Mod}(\Gamma, L)$ the category of finitely generated *L*-modules equipped with a continuous action of Γ , and by $\mathsf{Mod}(\Gamma, L)_{tor}$ (resp. $\mathsf{Mod}(\Gamma, L)_{fr}$) the full subcategory of $\mathsf{Mod}(\Gamma, L)$ consisting of those whose underlying *L*-module is torsion (resp. free).

For the rest of this subsection, we further assume that Γ is a topologically finitely generated abelian group and L is Noetherian.

Definition 2.1.2. We say that an $L[\Gamma]$ -module M is weakly semisimple if

- (1) M is an object of $\mathsf{Mod}(\Gamma, L)$; and
- (2) the natural map $M^{\Gamma} \to M_{\Gamma}$ is an isomorphism.

Lemma 2.1.3. Let M be an $L[\Gamma]$ -module that is an object of $Mod(\Gamma, L)$. Then

- (1) $M_{\Gamma} = 0$ implies $M^{\Gamma} = 0$;
- (2) M is weakly semisimple if and only if $(M/M^{\Gamma})_{\Gamma} = 0$.

Proof. For (1), for every maximal ideal \mathfrak{m} of L, let \mathfrak{m}' be the maximal ideal of $L[[\Gamma]]$ topologically generated by \mathfrak{m} and Γ . As $M_{\Gamma} = 0$, we have $M \otimes_{L[\Gamma]} (L[[\Gamma]]/\mathfrak{m}') = 0$ hence $M_{\mathfrak{m}'} = 0$ by Nakayama's lemma since $L[[\Gamma]]$ is Noetherian. Thus, we have $(M^{\Gamma})_{\mathfrak{m}} = (M^{\Gamma})_{\mathfrak{m}'} = 0$, which implies $M^{\Gamma} = 0$. For (2), consider the short exact sequence $0 \to M^{\Gamma} \to M \to M/M^{\Gamma} \to 0$. Suppose that

For (2), consider the short exact sequence $0 \to M^{\Gamma} \to M \to M/M^{\Gamma} \to 0$. Suppose that M is weakly semisimple. Then the natural map $(M^{\Gamma})_{\Gamma} \to M_{\Gamma}$ is an isomorphism. Thus, we have $(M/M^{\Gamma})_{\Gamma} = 0$. Conversely, suppose $(M/M^{\Gamma})_{\Gamma} = 0$. Then we have $\mathrm{H}^{1}(\Gamma, M/M^{\Gamma}) = 0$, and $\mathrm{H}^{0}(\Gamma, M/M^{\Gamma}) = 0$ by (1). From the exact sequence

$$\mathrm{H}^{0}(\Gamma, M/M^{\Gamma}) \to \mathrm{H}^{1}(\Gamma, M^{\Gamma}) \to \mathrm{H}^{1}(\Gamma, M) \to \mathrm{H}^{1}(\Gamma, M/M^{\Gamma})$$

we know that natural map $M^{\Gamma} \to M_{\Gamma}$, which coincides with $H^{1}(\Gamma, M^{\Gamma}) \to H^{1}(\Gamma, M)$ is an isomorphism.

The lemma is proved.

Lemma 2.1.4. We have

- (1) A finite direct sum of weakly semisimple $L[\Gamma]$ -modules is weakly semisimple.
- (2) A subquotient $L[\Gamma]$ -module of a weakly semisimple $L[\Gamma]$ -module is weakly semisimple.

Proof. Part (1) is obvious.

For (2), we take a weakly semisimple $L[\Gamma]$ -module M. By Lemma 2.1.3(2), we have $(M/M^{\Gamma})_{\Gamma} = 0$.

Let N be an $L[\Gamma]$ -submodule of M. Since L is Noetherian, N is an object of $\mathsf{Mod}(\Gamma, L)$. As $N^{\Gamma} = N \cap M^{\Gamma}$, we have an inclusion $N/N^{\Gamma} \hookrightarrow M/M^{\Gamma}$. Since $(M/NM^{\Gamma})_{\Gamma} = 0$, we have $\mathrm{H}^{0}(\Gamma, M/NM^{\Gamma}) = 0$ by Lemma 2.1.3(1), which implies $(N/N^{\Gamma})_{\Gamma} = 0$. By Lemma 2.1.3(2) again, we know that N is weakly semisimple.

Let N be a quotient $L[\Gamma]$ -module of M. Then we have a quotient map $M/M^{\Gamma} \to N/N^{\Gamma}$. Thus, we have $(N/N^{\Gamma})_{\Gamma} = 0$. By Lemma 2.1.3(2) again, we know that N is weakly semisimple. Part (2) is proved.

Lemma 2.1.5. Let M be an $O_{\lambda}[\Gamma]$ -module that is an object of $\mathsf{Mod}(\Gamma, O_{\lambda})_{\mathrm{fr}}$. Suppose that $M \otimes_{O_{\lambda}} O_{\lambda}/\lambda$ is weakly semisimple, and $\dim_{E_{\lambda}}(M \otimes_{O_{\lambda}} E_{\lambda})^{\Gamma} \geq \dim_{O_{\lambda}/\lambda}(M \otimes_{O_{\lambda}} O_{\lambda}/\lambda)^{\Gamma}$. Then M is weakly semisimple as well, and $\dim_{E_{\lambda}}(M \otimes_{O_{\lambda}} E_{\lambda})^{\Gamma} = \dim_{O_{\lambda}/\lambda}(M \otimes_{O_{\lambda}} O_{\lambda}/\lambda)^{\Gamma}$.

Proof. Since M is a finitely generated free O_{λ} -module, both M^{Γ} and M/M^{Γ} are finitely generated free O_{λ} -modules. In particular, the map $M^{\Gamma} \otimes_{O_{\lambda}} O_{\lambda}/\lambda \to (M \otimes_{O_{\lambda}} O_{\lambda}/\lambda)^{\Gamma}$ is injective. As we have

$$\dim_{O_{\lambda}/\lambda} M^{\Gamma} \otimes_{O_{\lambda}} O_{\lambda}/\lambda = \operatorname{rank}_{O_{\lambda}} M^{\Gamma} = \dim_{E_{\lambda}} (M \otimes_{O_{\lambda}} E_{\lambda})^{\Gamma},$$

the map $M^{\Gamma} \otimes_{O_{\lambda}} O_{\lambda}/\lambda \to (M \otimes_{O_{\lambda}} O_{\lambda}/\lambda)^{\Gamma}$ is an isomorphism. Thus, we have

$$\dim_{E_{\lambda}}(M \otimes_{O_{\lambda}} E_{\lambda})^{\Gamma} = \dim_{O_{\lambda}/\lambda}(M \otimes_{O_{\lambda}} O_{\lambda}/\lambda)^{\Gamma}$$

and

$$(M/M^{\Gamma}) \otimes_{O_{\lambda}} O_{\lambda}/\lambda \simeq (M \otimes_{O_{\lambda}} O_{\lambda}/\lambda)/(M \otimes_{O_{\lambda}} O_{\lambda}/\lambda)^{\Gamma}$$

As $M \otimes_{O_{\lambda}} O_{\lambda}/\lambda$ is weakly semisimple, we have $((M/M^{\Gamma}) \otimes_{O_{\lambda}} O_{\lambda}/\lambda)_{\Gamma} = 0$ by Lemma 2.1.3(2). By Nakayama's lemma, we have $(M/M^{\Gamma})_{\Gamma} = 0$, which implies that M is weakly semisimple by Lemma 2.1.3(2). The lemma is proved.

To end this subsection, we record the following definition which slightly generalizes [Liu16, Definition 5.1], which will be used in later sections.

Definition 2.1.6. Consider an O_{λ} -module M and an element $x \in M$. We define the *exponent* and the *order* of x to be

$$\exp_{\lambda}(x, M) \coloneqq \min\{d \in \mathbb{Z}_{\geq 0} \cup \{\infty\} \mid \lambda^{d} x = 0\},\\ \operatorname{ord}_{\lambda}(x, M) \coloneqq \sup\{d \in \mathbb{Z}_{\geq 0} \mid x \in \lambda^{d} M\} \in \mathbb{Z}_{\geq 0} \cup \{\infty\},$$

respectively.

2.2. Local Galois cohomology. In this subsection, we study Galois cohomology locally at nonarchimedean places of F. Let w be a nonarchimedean place of F.

Notation 2.2.1. For a commutative topological \mathbb{Z}_{ℓ} -algebra L and ? $\in \{$, tor, fr}, we

(1) put $\mathsf{Mod}(F_w, L)_? \coloneqq \mathsf{Mod}(\Gamma_{F_w}, L)_?$;

(2) denote by -(j): $\mathsf{Mod}(F_w, L)_? \to \mathsf{Mod}(F_w, L)_?$ the functor of j-th Tate twist for $j \in \mathbb{Z}$; and

(3) denote by $-^{\vee}$: $\mathsf{Mod}(F_w, L)^{\mathrm{op}}_? \to \mathsf{Mod}(F_w, L)_?$ the functor sending M to $\operatorname{Hom}_L(M, L)$.

We also denote

 $-_{\mathbb{Q}}: \operatorname{\mathsf{Mod}}(F_w, O_\lambda) \to \operatorname{\mathsf{Mod}}(F_w, E_\lambda)$

the base change functor sending M to $M \otimes_{O_{\lambda}} E_{\lambda}$, and

$$-^*: \operatorname{\mathsf{Mod}}(F_w, O_\lambda)^{\operatorname{op}} \to \operatorname{\mathsf{Mod}}(F_w, O_\lambda)$$

the E_{λ} -Pontryagin duality functor sending M to $\operatorname{Hom}_{O_{\lambda}}(M, E_{\lambda}/O_{\lambda})$. For every pair $m, m' \in \{1, 2, \ldots, \infty\}$ with $m' \geq m$, we have a "reduction modulo λ^{m} " functor

$$\mathbf{L}^{(m)} \coloneqq - \otimes_{O_{\lambda}} O_{\lambda}/\lambda^{m} \colon \mathsf{Mod}(F_{w}, O_{\lambda}/\lambda^{m'}) \to \mathsf{Mod}(F_{w}, O_{\lambda}/\lambda^{m}).$$

We usually write - for $-^{(1)}$.

Recall that for every object $\mathbf{R} \in \mathsf{Mod}(F_w, O_\lambda)$, we have a local Tate pairing

(2.1)
$$\langle , \rangle_w \colon \mathrm{H}^1(F_w, \mathrm{R}) \times \mathrm{H}^1(F_w, \mathrm{R}^*(1)) \xrightarrow{\cup} \mathrm{H}^2(F_w, E_\lambda/O_\lambda) \simeq E_\lambda/O_\lambda,$$

which we will study in the following.

First, we study the case where ℓ is invertible in κ_w .

Definition 2.2.2. For every object R in either $Mod(F_w, E_\lambda)$ or $Mod(F_w, O_\lambda)$, we put

$$\mathrm{H}^{1}_{\mathrm{sing}}(F_{w}, \mathrm{R}) \coloneqq \mathrm{H}^{1}(\mathrm{I}_{F_{w}}, \mathrm{R})^{\Gamma_{\kappa_{w}}};$$

and denote by $\mathrm{H}^{1}_{f}(F_{w}, \mathbf{R})$ the kernel of the canonical map

$$\partial_w \colon \mathrm{H}^1(F_w, \mathrm{R}) \to \mathrm{H}^1_{\mathrm{sing}}(F_w, \mathrm{R}),$$

called the *finite part* of $H^1(F_w, \mathbb{R})$.

By the inflation-restriction exact sequence (see, for example, [Liu19, Lemma 2.6]), we know that ∂_w is surjective, and $\mathrm{H}^1_f(F_w, \mathbb{R})$ is canonically isomorphic to $\mathrm{H}^1(\kappa_w, \mathbb{R}^{\mathrm{I}_{F_w}})$.

Lemma 2.2.3. For $\mathbb{R} \in \mathsf{Mod}(F_w, O_\lambda)_{tor}$, the finite parts $\mathrm{H}^1_f(F_w, \mathbb{R})$ and $\mathrm{H}^1_f(F_w, \mathbb{R}^*(1))$ are the exact annihilators of each other under the local Tate pairing \langle , \rangle_w (2.1).

Proof. This is well-known. In fact, the cup product of $H^1_f(F_w, \mathbb{R})$ and $H^1_f(F_w, \mathbb{R}^*(1))$ factors through $H^2(\kappa_w, \mathbb{R}^{I_{F_w}} \otimes \mathbb{R}^*(1)^{I_{F_w}})$, which is the zero group. The lemma then follows from an easy computation of length and the fact that the pairing \langle , \rangle_w is perfect.

Second, we study the case that κ_w has characteristic ℓ . In particular, F_w is a finite unramified extension of \mathbb{Q}_{ℓ} . Denote by $-_0$: $\mathsf{Mod}(F_w, O_\lambda) \to \mathsf{Mod}(F_w, \mathbb{Z}_{\ell})$ the obvious forgetful functor.

Definition 2.2.4. Let $a \leq b$ be two integers.

- (1) For an object $\mathbf{R} \in \mathsf{Mod}(F_w, \mathbb{Z}_\ell)_{tor}$, we say that \mathbf{R} is crystalline with Hodge–Tate weights in [a, b] if $\mathbf{R} = \mathbf{R}''/\mathbf{R}'$ where $\mathbf{R}' \subseteq \mathbf{R}''$ are two Γ_{F_w} -stable \mathbb{Z}_ℓ -lattices in a crystalline \mathbb{Q}_ℓ -representation of Γ_{F_w} with Hodge–Tate weights in [a, b].
- (2) For an object $R \in \mathsf{Mod}(F_w, \mathbb{Z}_\ell)$, we say that R is crystalline with Hodge-Tate weights in [a, b] if $R/\ell^m R$ is a torsion crystalline module with Hodge-Tate weights in [a, b] for every integer $m \ge 1$.
- (3) For an object $R \in \mathsf{Mod}(F_w, O_\lambda)$, we say that R is crystalline with Hodge-Tate weights in [a, b] if R_0 is.

Definition 2.2.5. For an object $\mathbf{R} \in \mathsf{Mod}(F_w, O_\lambda)$ that is crystalline with Hodge–Tate weights in [a, b], we define $\mathrm{H}^1_f(F_w, \mathbf{R})$ to be the subset of $\mathrm{H}^1(F_w, \mathbf{R}) = \mathrm{H}^1(F_w, \mathbf{R}_0)$ consisting of elements s represented by an extension

$$0 \to \mathbf{R}_0 \to \mathbf{R}_s \to \mathbb{Z}_\ell \to 0$$

in the category $\mathsf{Mod}(F_w, \mathbb{Z}_\ell)$ such that \mathbb{R}_s is crystalline with Hodge–Tate weights in [a, b].

It follows that $\mathrm{H}^{1}_{f}(F_{w}, \mathbb{R})$ is an O_{λ} -submodule of $\mathrm{H}^{1}(F_{w}, \mathbb{R})$.

Proof. We have a canonical map $\operatorname{Tr}: (\mathbb{R}^*)_0 \to (\mathbb{R}_0)^*$ in the category $\operatorname{\mathsf{Mod}}(F_w, \mathbb{Z}_\ell)$ induced by the trace map $\operatorname{Tr}_{E_\lambda/\mathbb{Q}_\ell}$, which induces a map $\operatorname{H}^1(F_w, \mathbb{R}^*(1)) \to \operatorname{H}^1(F_w, (\mathbb{R}_0)^*(1))$ under which the image of $\operatorname{H}^1_f(F_w, \mathbb{R}^*(1))$ is contained in $\operatorname{H}^1_f(F_w, (\mathbb{R}_0)^*(1))$. Take elements $x \in \operatorname{H}^1_f(F_w, \mathbb{R})$ and $y \in \operatorname{H}^1_f(F_w, \mathbb{R}^*(1))$. Then we have for every $a \in O_\lambda$,

$$\operatorname{Tr}_{E_{\lambda}/\mathbb{Q}_{\ell}}(a\langle x, y\rangle_{w}) = \operatorname{Tr}_{E_{\lambda}/\mathbb{Q}_{\ell}}\langle ax, y\rangle_{w} = \langle ax, \operatorname{Tr}(y)\rangle_{w} \in \mathbb{Q}_{\ell}/\mathbb{Z}_{\ell}.$$

However, $\langle ax, \text{Tr}(y) \rangle_w = 0$ by [Niz93, Proposition 6.2]. The lemma follows.

2.3. Some Galois-theoretical lemmas. In this subsection, we generalize some lemmas from [Liu16]. For a finite set Σ of places of F, we denote by $\Gamma_{F,\Sigma}$ the Galois group of the maximal subextension of \overline{F}/F that is unramified outside Σ .

Notation 2.3.1. For a commutative topological \mathbb{Z}_{ℓ} -algebra L and ? $\in \{$, tor, fr}, we put

$$\mathsf{Mod}(F,L)_? \coloneqq \varinjlim_{\Sigma} \mathsf{Mod}(\Gamma_{F,\Sigma},L)_?$$

where the colimit is taken over all finite sets Σ of places of F with inflation as transition functors. We have functors -(j), $-^{\vee}$, $-_{\mathbb{Q}}$, $-^*$, and $\bar{-}^{(m)}$ similar to those in Notation 2.2.1. For an object $\mathbf{R} \in \mathsf{Mod}(F, L)$ and $i \in \mathbb{Z}$, we put

$$\mathrm{H}^{i}(F, \mathrm{R}) \coloneqq \varinjlim_{\Sigma'} \mathrm{H}^{i}(\Gamma_{F,\Sigma}, \mathrm{R}).$$

Moreover, for every place w of F, we have the restriction functor $Mod(F, L) \to Mod(F_w, L)$; and denote

$$\operatorname{loc}_w \colon \operatorname{H}^i(F, \mathbb{R}) \to \operatorname{H}^i(F_w, \mathbb{R})$$

the localization map.

Definition 2.3.2 ([Liu16, Definition 5.1]). Let G be a profinite group. For an object $R \in Mod(G, O_{\lambda})_{tor}$, we define its *reducibility depth* to be the smallest integer $\mathfrak{r}_{R} \geq 0$ such that

- (1) if R' is a G-stable O_{λ} -submodule that is not contained in λR , then R' contains $\lambda^{\mathfrak{r}_{R}} R$;
- (2) for every positive integer m, the group $\operatorname{End}_{O_{\lambda}[G]}(\bar{\mathbb{R}}^{(m)})/O_{\lambda} \cdot \operatorname{id}$ is annihilated by $\lambda^{\mathfrak{r}_{\mathbb{R}}}$.

Note that if $R/\lambda R$ is absolutely irreducible, then $r_R = 0$.

Lemma 2.3.3. Let $R \in Mod(F, O_{\lambda})$ be an object such that $R_{\mathbb{Q}}$ is absolutely irreducible. Then there exists an integer \mathfrak{r}_R depending on R only, such that $\overline{R}^{(m)}$ has reducibility depth at most \mathfrak{r}_R for every positive integer m.

Proof. The same argument in [Liu16, Lemma 5.2] apply to our case as well, with \mathbb{Z}/p^n replaced by O_{λ}/λ^m .

Now we fix a positive integer m. Consider an object $\mathbb{R} \in \mathsf{Mod}(F, O_{\lambda}/\lambda^m)_{\mathrm{fr}}$. We denote by $\rho: \Gamma_F \to \mathrm{GL}(\mathbb{R})$ the associated homomorphism. Let F_{ρ}/F be the Galois extension fixed by the kernel of ρ , and $G \coloneqq \mathrm{Gal}(F_{\rho}/F)$ the image of ρ . we have the restriction map

(2.2)
$$\operatorname{Res}_{\rho} \colon \operatorname{H}^{1}(F, \mathbb{R}) \to \operatorname{H}^{1}(F_{\rho}, \mathbb{R})^{G} = \operatorname{Hom}_{G}(\Gamma_{F_{\rho}}^{\mathrm{ab}}, \mathbb{R}),$$

where $\Gamma_{F_{\rho}}^{ab} \coloneqq \operatorname{Gal}(F_{\rho}^{ab}/F_{\rho})$ with $F_{\rho}^{ab} \subseteq \overline{F}$ the maximal abelian extension of F_{ρ} , which is equipped with the natural conjugation action by $G = \operatorname{Gal}(F_{\rho}/F)$.

Lemma 2.3.4. Suppose that either one of the following two assumptions holds:

- (a) the image of Γ_F in GL(R) contains a nontrivial scalar element;
- (b) $\dim_{O_{\lambda}/\lambda} \bar{\mathbf{R}} \leq \min\{\frac{\ell+1}{2}, \ell-3\}, \ \bar{\mathbf{R}} \ is \ a \ semisimple \ O_{\lambda}/\lambda[\Gamma_F]$ -module, and moreover $\operatorname{Hom}_{O_{\lambda}/\lambda[\Gamma_F]}(\operatorname{End}(\bar{\mathbf{R}}), \bar{\mathbf{R}}) = 0.$

Then the map $\operatorname{Res}_{\rho}(2.2)$ is injective.

Proof. By the inflation-restriction exact sequence, it suffices to show that $H^1(G, \mathbb{R}) = 0$.

In the situation (a), it follows that G contains a nontrivial scalar element of order coprime to ℓ . Then by the same argument in [Gro91, Proposition 9.1], we have $H^1(G, \mathbb{R}) = 0$.

Now we consider the situation (b). We prove by induction that $H^1(G, \mathbb{R}^{(i)}) = 0$ for $1 \le i \le m$. Suppose $H^1(G, \mathbb{R}^{(j)}) = 0$ for $1 \le j \le i < m$. By the short exact sequence

$$0 \to \bar{\mathbf{R}}^{(i+1)} \otimes_{O_{\lambda}/\lambda^{i+1}} \lambda^{i}/\lambda^{i+1} \to \bar{\mathbf{R}}^{(i+1)} \to \bar{\mathbf{R}}^{(i)} \to 0$$

of $O_{\lambda}[G]$ -modules, in which $\bar{\mathbf{R}}^{(i+1)} \otimes_{O_{\lambda}/\lambda^{i+1}} \lambda^i / \lambda^{i+1}$ is isomorphic to $\bar{\mathbf{R}}$, we know that $\mathrm{H}^1(G, \bar{\mathbf{R}}^{(i+1)}) = 0$. 0. Therefore, it remains to check the initial step that $\mathrm{H}^1(G, \bar{\mathbf{R}}) = 0$.

Let $G^i \subseteq G$ be the kernel of the composite homomorphism $G \to \operatorname{GL}(\mathbb{R}) \to \operatorname{GL}(\overline{\mathbb{R}}^{(i)})$ for $1 \leq i \leq m$, so we obtain a filtration $0 = G^m \subseteq G^{m-1} \subseteq G^1 \subseteq G$ of normal subgroups of G. We prove by induction that $\operatorname{H}^1(G/G^i, \overline{\mathbb{R}}) = 0$. For i = 1, since $\overline{\mathbb{R}}$ is a faithful semisimple $O_{\lambda}/\lambda[G/G^1]$ -module, G/G^1 has no nontrivial normal ℓ -subgroup. As $\dim_{O_{\lambda}/\lambda} \overline{\mathbb{R}} \leq \ell - 3$, we have $\operatorname{H}^1(G/G^1, \overline{\mathbb{R}}) = 0$ by [Gur99, Theorem A]. Suppose $\operatorname{H}^1(G/G^j, \overline{\mathbb{R}}) = 0$ for $1 \leq j \leq i < m$. By the inflation-restriction exact sequence

$$0 \to \mathrm{H}^{1}(G/G^{i}, \bar{\mathrm{R}}) \to \mathrm{H}^{1}(G/G^{i+1}, \bar{\mathrm{R}}) \to \mathrm{Hom}_{G}(G^{i}/G^{i+1}, \bar{\mathrm{R}}),$$

it suffices to show that $\operatorname{Hom}_{G}(G^{i}/G^{i+1}, \overline{\mathbb{R}}) = 0$, or equivalently, $\operatorname{Hom}_{O_{\lambda}/\lambda[G]}(G^{i}/G^{i+1} \otimes O_{\lambda}/\lambda, \overline{\mathbb{R}}) = 0$. Note that G^{i}/G^{i+1} is an $\mathbb{F}_{\ell}[G]$ -submodule of $\operatorname{End}(\overline{\mathbb{R}})$, hence $(G^{i}/G^{i+1}) \otimes O_{\lambda}/\lambda$ is an $O_{\lambda}/\lambda[G]$ -submodule of $\operatorname{End}(\overline{\mathbb{R}}) \otimes (O_{\lambda}/\lambda) \simeq \operatorname{End}(\overline{\mathbb{R}})^{d}$, where $d := [O_{\lambda}/\lambda : \mathbb{F}_{\ell}]$ is the degree. Since $\overline{\mathbb{R}}$ is a semisimple $O_{\lambda}/\lambda[G]$ -module and $2 \dim_{O_{\lambda}/\lambda} \overline{\mathbb{R}} < \ell+2$, by [Ser94, Corollaire 1], we know that $\operatorname{End}(\overline{\mathbb{R}})$ is a semisimple $O_{\lambda}/\lambda[G]$ -module. In particular, we have $\operatorname{Hom}_{O_{\lambda}/\lambda[G]}(G^{i}/G^{i+1} \otimes O_{\lambda}/\lambda, \overline{\mathbb{R}}) = 0$ as $\operatorname{Hom}_{G}(\operatorname{End}(\overline{\mathbb{R}}), \overline{\mathbb{R}}) = 0$.

The lemma is proved.

Remark 2.3.5. In Lemma 2.3.4, assumption (a) is well-known to deduce the injectivity, and in fact the surjectivity as well, for the map $\operatorname{Res}_{\rho}$; this is the assumption in all previous works concerning Selmer groups of elliptic curves or their products and symmetric powers. However, for reduction of general automorphic Galois representations, assumption (a) is very hard to verify. Thus, we find an alternative, namely, assumption (b) for the injectivity of $\operatorname{Res}_{\rho}$; it looks much more complicated than (a), nevertheless can be achieved under certain mild conditions; see Corollary 8.2.3.

The map $\operatorname{Res}_{\rho}(2.2)$ induces an O_{λ} -linear pairing

$$[,]: \mathrm{H}^{1}(F, \mathrm{R}) \times \Gamma^{\mathrm{ab}}_{F_{\rho}} \to \mathrm{R},$$

such that the action of G on $\Gamma_{F_{\rho}}^{ab}$ is compatible with that on R. Let S be a finitely generated O_{λ}/λ^{m} submodule of $\mathrm{H}^{1}(F, \mathrm{R})$, and let F_{S}/F_{ρ} be the finite abelian extension such that $\mathrm{Gal}(F_{\rho}^{ab}/F_{S})$ is
the subgroup of $\Gamma_{F_{\rho}}^{ab}$ consisting of γ satisfying $[s, \gamma] = 0$ for every $s \in S$. As in [Liu16, Section 5.1],
we introduce a sequence \mathfrak{f} that is given by $\mathfrak{f}(0) = 1$, $\mathfrak{f}(1) = 1$, $\mathfrak{f}(2) = 4$, $\mathfrak{f}(r+1) = 2(\mathfrak{f}(r)+1)$ for $r \geq 2$.

Lemma 2.3.6. Let the notation be as above. Suppose that either one of the two assumptions in Lemma 2.3.4 is satisfied. Then the induced pairing

$$[,]: S \times \operatorname{Gal}(F_S/F_{\rho}) \to \mathbb{R}$$

induces an injective map θ_S : $\operatorname{Gal}(F_S/F_{\rho}) \to \operatorname{Hom}_{O_{\lambda}}(S, \mathbb{R})$ of abelian groups that is compatible with G-actions. Moreover, if S is a free $O_{\lambda}/\lambda^{m-m'}$ -module of rank r_S for some integer $0 \leq m' \leq m$, then the O_{λ} -submodule of $\operatorname{Hom}_{O_{\lambda}}(S, \mathbb{R})$ generated by the image of θ_S contains $\lambda^{\mathfrak{f}(r_S)\mathfrak{r}_{\mathbb{R}}} \operatorname{Hom}_{O_{\lambda}}(S, \mathbb{R})$, where $\mathfrak{r}_{\mathbb{R}} \leq m$ is the reducibility depth of \mathbb{R} .

Proof. The same argument in [Liu16, Lemma 5.4] apply to our case as well, with \mathbb{Z}/p^n replaced by $O_{\lambda}/\lambda^{m-m'}$. Note that the proof only uses the injectivity, not the surjectivity, of the map $\operatorname{Res}_{\rho}$ (2.2).

2.4. Reduction of Selmer groups. We recall the following definition of the Bloch–Kato Selmer group from [BK90].

Definition 2.4.1 (Bloch–Kato Selmer group). For an object $\mathbb{R} \in \mathsf{Mod}(F, E_{\lambda})$, we define the Bloch–Kato Selmer group $\mathrm{H}^{1}_{f}(F, \mathbb{R})$ of \mathbb{R} to be the E_{λ} -subspace of $\mathrm{H}^{1}(F, \mathbb{R})$ consisting of elements s such that

- (1) $\operatorname{loc}_w(s) \in \operatorname{H}^1_f(F_w, \mathbb{R})$ (Definition 2.2.2) for every nonarchimedean place w of F; and
- (2) $\operatorname{loc}_w(s) \in \operatorname{H}^1_f(F_w, \mathbb{R}) := \ker (\operatorname{H}^1(F_w, \mathbb{R}) \to \operatorname{H}^1(F_w, \mathbb{R} \otimes_{\mathbb{Q}_\ell} \mathbb{B}_{\operatorname{cris}}))$ for every place w above ℓ , where $\mathbb{B}_{\operatorname{cris}}$ is Fontaine's crystalline period ring for \mathbb{Q}_ℓ .

Definition 2.4.2. Consider an object $R \in Mod(F, O_{\lambda})_{fr}$.

- (1) We define the *(integral) Bloch-Kato Selmer group* $H^1_f(F, \mathbb{R})$ of \mathbb{R} to be inverse image of $H^1_f(F, \mathbb{R}_{\mathbb{Q}})$ under the obvious map $H^1(F, \mathbb{R}) \to H^1(F, \mathbb{R}_{\mathbb{Q}})$.
- (2) For $m \in \{1, 2, ..., \infty\}$, we define $\mathrm{H}^{1}_{f, \mathrm{R}}(F, \overline{\mathrm{R}}^{(m)})$ to be the image of $\mathrm{H}^{1}_{f}(F, \mathrm{R})$ under the obvious map $\mathrm{H}^{1}(F, \mathrm{R}) \to \mathrm{H}^{1}(F, \overline{\mathrm{R}}^{(m)})$.

Lemma 2.4.3. Consider an object $R \in Mod(F, O_{\lambda})_{fr}$. Suppose that we are in one of the two following cases

- (1) w is a nonarchimedean place of F not above ℓ at which R is unramified.
- (2) w is a place of F above ℓ at which $\mathbb{R}_{\mathbb{Q}}$ is crystalline with Hodge–Tate weights in [a, b] with $a \leq 0 \leq b$ and $b a \leq \ell 2$.

Then for every positive integer m, we have

(2.3)
$$\operatorname{loc}_{w}(\operatorname{H}^{1}_{f,\operatorname{R}}(F, \overline{\operatorname{R}}^{(m)})) \subseteq \operatorname{H}^{1}_{f}(F_{w}, \overline{\operatorname{R}}^{(m)}).$$

Proof. For (1), as R is unramified at v, the natural map

$$\mathrm{H}^{1}(\mathrm{I}_{F_{w}},\mathrm{R}) = \mathrm{Hom}(\mathrm{I}_{F_{w}},\mathrm{R}) \to \mathrm{H}^{1}(\mathrm{I}_{F_{w}},\mathrm{R}_{\mathbb{Q}}) = \mathrm{Hom}(\mathrm{I}_{F_{w}},\mathrm{R}_{\mathbb{Q}})$$

is injective. We deduce from the following map of exact sequences

that $\mathrm{H}^1_f(F_w, \mathbb{R})$ consists of exactly the elements of $\mathrm{H}^1(F_w, \mathbb{R})$ whose image in $\mathrm{H}^1(F_w, \mathbb{R}_{\mathbb{Q}})$ belongs to $\mathrm{H}^1_f(F_w, \mathbb{R}_{\mathbb{Q}})$. From this, we conclude that loc_w sends $\mathrm{H}^1_f(F, \mathbb{R})$ into $\mathrm{H}^1_f(F_w, \mathbb{R})$. Moreover, it is clear that the image of $\mathrm{H}^1_f(F_w, \mathbb{R})$ under the reduction map $\mathrm{H}^1(F_w, \mathbb{R}) \to \mathrm{H}^1(F_w, \overline{\mathbb{R}}^{(m)})$ is contained in $\mathrm{H}^{1}_{f}(F_{w}, \bar{\mathrm{R}}^{(m)})$. (In fact, the image is exactly equal to $\mathrm{H}^{1}_{f}(F_{w}, \bar{\mathrm{R}}^{(m)})$.) The desired inclusion (2.3) follows from this by the definition of $\mathrm{H}^{1}_{f,\mathrm{R}}(F, \bar{\mathrm{R}}^{(m)})$.

For (2), by the main result of [Liu07], $\mathrm{H}^1_f(F_w, \mathrm{R})$ is exactly the preimage of $\mathrm{H}^1_f(F_w, \mathrm{R}_{\mathbb{Q}})$ via the natural map $\mathrm{H}^1(F_w, \mathrm{R}) \to \mathrm{H}^1(F_w, \mathrm{R}_{\mathbb{Q}})$. It is clear then from the definition that the image of $\mathrm{H}^1_f(F_w, \mathrm{R})$ in $\mathrm{H}^1(F_w, \overline{\mathrm{R}}^{(m)})$ lies in $\mathrm{H}^1_f(F_w, \overline{\mathrm{R}}^{(m)})$.

We recall the notion of purity for a local Galois representation.

Definition 2.4.4. Let w be a nonarchimedean place of F not above ℓ . Consider an object $\mathbb{R} \in \mathsf{Mod}(F_w, E_\lambda)$. Let $WD(\mathbb{R})$ be the attached Weil–Deligne representation, and $\operatorname{gr}_n WD(\mathbb{R})$ be the *n*-th graded piece of the monodromy filtration on $WD(\mathbb{R})$. For $\mu \in \mathbb{Z}$, we say that \mathbb{R} is *pure of weight* μ if $\operatorname{gr}_n WD(\mathbb{R})$ is pure of weight $\mu + n$ for each n, that is, the eigenvalues of ϕ_w on $\operatorname{gr}_n WD(\mathbb{R})$ are Weil $||w||^{-(\mu+n)}$ -numbers.

From now to the end of this section, we suppose that the complex conjugation c restricts to an automorphism of F (of order at most two). We adopt the notation concerning ground fields in Subsection 1.3; in particular, we put $F^+ := F^{c=1}$. We also have a functor

$$-^{c}$$
: Mod $(F, L) \rightarrow Mod(F, L)$

induced by the conjugation by c.

Lemma 2.4.5. For every object $\mathbb{R} \in \mathsf{Mod}(F, E_{\lambda})$, the functor $-^{\mathsf{c}}$ induces an isomorphism $\mathrm{H}^{1}_{f}(F, \mathbb{R}) \simeq \mathrm{H}^{1}_{f}(F, \mathbb{R}^{\mathsf{c}})$

of Selmer groups.

Proof. Regard elements in $H^1(F, -)$ as extensions. Then applying $-^c$ to extensions induces maps $H^1(F, \mathbb{R}) \to H^1(F, \mathbb{R}^c), \quad H^1(F, \mathbb{R}^c) \to H^1(F, \mathbb{R})$

which are inverse to each other. It is clear that conditions (1) and (2) in Definition 2.4.1 are preserved under such maps. The lemma follows.

Proposition 2.4.6. Let R be an object in $Mod(F, O_{\lambda})_{fr}$ satisfying $\mathbb{R}^{c}_{\mathbb{Q}} \simeq \mathbb{R}^{\vee}_{\mathbb{Q}}(1)$ and such that $\mathbb{R}_{\mathbb{Q}}$ is pure of weight -1 at every nonarchimedean place w of F not above ℓ . Take a finite set Σ of places of F. Then there exists a positive integer m_{Σ} , depending on R and Σ , such that for every free O_{λ} -submodule S of $\mathrm{H}^{1}_{f}(F, \mathbb{R})$ that is saturated in $\mathrm{H}^{1}_{f}(F, \mathbb{R})/\mathrm{H}^{1}_{f}(F, \mathbb{R})_{\mathrm{tor}}$ and every integer $m > m_{\Sigma}$, we have:

- (1) $S^{(m)}$, the image of S in $H^1_{f,R}(F, \overline{R}^{(m)})$, is a free O_{λ}/λ^m -module of the same rank as S;
- (2) $\operatorname{loc}_w(\lambda^{m_{\Sigma}}S^{(m)}) = 0$ for every nonarchimedean place $w \in \Sigma$ not above ℓ .

Proof. Part (1) follows from the same argument for [Liu16, Lemma 5.9].

For (2), we look at the map

$$\operatorname{loc}_{\Sigma}^{\infty\ell} \colon \operatorname{H}^{1}_{f, \mathbb{R}}(F, \bar{\mathbb{R}}^{(m)}) \to \bigoplus_{w \in \Sigma, w \nmid \infty\ell} \operatorname{H}^{1}(F_{w}, \bar{\mathbb{R}}^{(m)}).$$

For every $w \nmid \infty \ell$, since $\mathbb{R}_{\mathbb{Q}}$ is of pure weight of -1 at w, $\mathbb{R}_{\mathbb{Q}}^{c}$ and $\mathbb{R}_{\mathbb{Q}}^{\vee}(1)$ are of pure weight of -1 at w as well. Thus, we have $\mathrm{H}^{0}(F_{w}, \mathbb{R}_{\mathbb{Q}}) = 0$ and $\mathrm{H}^{2}(F_{w}, \mathbb{R}_{\mathbb{Q}}) \simeq \mathrm{H}^{0}(F_{w}, \mathbb{R}_{\mathbb{Q}}^{\vee}(1))^{\vee} = 0$, hence $\mathrm{H}^{1}(F_{w}, \mathbb{R}_{\mathbb{Q}}) = 0$ by the Euler characteristic formula (see also the proof of [Nek07, Proposition 4.2.2(1)]). Thus, $\mathrm{H}^{1}(F_{w}, \mathbb{R})$ is annihilated by $\lambda^{m_{w}}$ for some integer $m_{w} \geq 0$. We may enlarge m_{w} so that $\lambda^{m_{w}}$ also annihilates $\mathrm{H}^{2}(F_{w}, \mathbb{R})_{\mathrm{tor}}$. Then it follows that $\mathrm{H}^{1}(F_{w}, \mathbb{R}^{(m)})$ is annihilated by $\lambda^{2m_{w}}$. Now if we put $m_{\Sigma} := \max\{2m_{w} \mid w \in \Sigma, w \nmid \infty \ell\}$, then (2) follows. This completes the proof of the proposition. To end this subsection, we recall the extension along *j*-polarization. This has been introduced in [CHT08, Section 1] when $[F:F^+] = 2$. We introduce the group scheme \mathscr{G}_N . When $[F:F^+] = 1$, it is just $\operatorname{GL}_N \times \operatorname{GL}_1$. When $[F:F^+] = 2$, it is the one in Notation E.1.1.

Definition 2.4.7. For a commutative topological \mathbb{Z}_{ℓ} -algebra L, an integer j, and an object R in Mod(F, L), a *j*-polarization of R is an isomorphism

$$\Xi \colon \mathrm{R}^{\mathsf{c}} \xrightarrow{\sim} \mathrm{R}^{\vee}(j)$$

in Mod(F, L), such that $\Xi^{c,\vee}(j) = \mu_{\Xi} \cdot (-1)^{j+1} \cdot \Xi$ for some $\mu_{\Xi} \in \{\pm 1\}$. We say that R is *j*-polarizable if there exists a *j*-polarization.

Construction 2.4.8. Let R be a nonzero object in $\mathsf{Mod}(F, L)_{\mathrm{fr}}$ with the associated continuous homomorphism $\rho \colon \Gamma_F \to \mathrm{GL}(\mathbb{R})$, equipped with a *j*-polarization $\Xi \colon \mathbb{R}^{\mathfrak{c}} \xrightarrow{\sim} \mathbb{R}^{\vee}(j)$. Choose an isomorphism $\mathbb{R} \simeq L^{\oplus N}$ of the underlying *L*-modules for a unique integer $N \ge 1$.

(1) When $[F:F^+] = 1$, we let

$$\rho_+ \colon \Gamma_{F^+} \to \mathscr{G}_N(L)$$

be the continuous homomorphism sending $g \in \Gamma_{F^+} = \Gamma_F$ to $(\rho(g), \epsilon_{\ell}^j(g))$.

(2) When $[F:F^+] = 2$, the *j*-polarization Ξ gives rise to an element $B \in \operatorname{GL}_N(L)$ as in Lemma E.1.3(2) for the pair $(\rho, \eta_{F/F}^{\mu_{\Xi}} \epsilon_{\ell}^j)$. We let

$$\rho_+\colon \Gamma_F \to \mathscr{G}_N(L)$$

be the continuous homomorphism given in Lemma E.1.3(2).

In both cases, we call ρ_+ an *extension* of ρ , which depends on the choice of a basis of R.

2.5. Localization of Selmer groups. In this subsection, we study the behavior of Selmer groups under localization maps.

Notation 2.5.1. We take a nonzero object $\mathbb{R} \in \mathsf{Mod}(F, O_{\lambda})_{\mathrm{fr}}$ with the associated homomorphism $\rho \colon \Gamma_F \to \mathrm{GL}(\mathbb{R})$, together with a *j*-polarization $\Xi \colon \mathbb{R}^{\mathfrak{c}} \xrightarrow{\sim} \mathbb{R}^{\vee}(j)$. We fix an isomorphism $\mathbb{R} \simeq O_{\lambda}^{\oplus N}$. Let

$$\rho_+ \colon \Gamma_{F^+} \to \mathscr{G}_N(O_\lambda)$$

be the extension of ρ from Construction 2.4.8. For every integer $m \ge 1$, we have the induced homomorphisms

$$\bar{\rho}^{(m)} \colon \Gamma_F \to \operatorname{GL}(\bar{\mathbf{R}}^{(m)}) \simeq \operatorname{GL}_N(O_\lambda/\lambda^m),$$
$$\bar{\rho}^{(m)}_+ \colon \Gamma_{F^+} \to \mathscr{G}_N(O_\lambda/\lambda^m),$$

and we omit the superscript (m) when m = 1.

We denote by $F^{(m)} := F_{\bar{\rho}^{(m)}}$ and $F_{+}^{(m)}$ the subfields of \overline{F} fixed by ker $\bar{\rho}^{(m)}$ and ker $\bar{\rho}_{+}^{(m)}$, respectively. In particular, we have $F \subseteq F^{(m)} \subseteq F_{+}^{(m)} \subseteq F^{(m)}(\zeta_{\ell^m})$.

Notation 2.5.2. For a positive integer m and an element $\gamma \in (\operatorname{GL}_N(O_{\lambda}/\lambda^m) \times (O_{\lambda}/\lambda^m)^{\times})\mathfrak{c} \subseteq \mathscr{G}_N(O_{\lambda}/\lambda^m)$, we denote by $h_{\gamma} \in \operatorname{GL}_N(O_{\lambda}/\lambda^m)$ the first component of $\gamma^{[F:F^+]} \in \operatorname{GL}_N(O_{\lambda}/\lambda^m) \times (O_{\lambda}/\lambda^m)^{\times}$.

Now we fix a positive integer m and an element $\gamma \in (\operatorname{GL}_N(O_\lambda/\lambda^m) \times (O_\lambda/\lambda^m)^{\times})\mathfrak{c} \subseteq \mathscr{G}_N(O_\lambda/\lambda^m)$. The following definition is essentially [Liu16, Definition 5.6].

Definition 2.5.3. We say that a place $w_+^{(m)}$ of $F_+^{(m)}$ is γ -associated if it is coprime to $\infty \ell$, unramified over F^+ , unramified in F_S , and such that its Frobenius substitution in $\operatorname{Gal}(F_+^{(m)}/F^+) \simeq \operatorname{im} \bar{\rho}_+^{(m)}$ coincides with γ .

Consider a finitely generated O_{λ} -submodule S of $\mathrm{H}^{1}_{f,\mathrm{R}}(F, \bar{\mathrm{R}}^{(m)})$. We have the finite abelian extension $F_{S}/F^{(m)}$ from Subsection 2.3. Suppose that either one of the two assumptions in Lemma 2.3.4 is satisfied. Then by Lemma 2.3.6, we have an injective map

$$\theta_S \colon \operatorname{Gal}(F_S/F^{(m)}) \to \operatorname{Hom}_{O_\lambda}(S, \overline{\mathbb{R}}^{(m)})$$

of abelian groups, equivariant under the action of $\operatorname{Gal}(F^{(m)}/F)$. Take a γ -associated place $w_+^{(m)}$ of $F_+^{(m)}$, and denote by its underlying places of $F^{(m)}$ and F by $w^{(m)}$ and w, respectively. Since $F_S/F^{(m)}$ is abelian, $w^{(m)}$ has a well-defined Frobenius substitution $\Psi_{w^{(m)}} \in \operatorname{Gal}(F_S/F^{(m)})$. Denote by $G_{S,\gamma}$ the subset of $\operatorname{Gal}(F_S/F^{(m)})$ of elements $\Psi_{w^{(m)}}$ for all γ -associated places $w_+^{(m)}$.

On the other hand, as ϕ_w acts on $\overline{\mathbb{R}}^{(m)}$ by h_{γ} , we have an isomorphism

(2.4)
$$\mathrm{H}_{f}^{1}(F_{w},\bar{\mathrm{R}}^{(m)}) = \mathrm{H}^{1}(\kappa_{w},\bar{\mathrm{R}}^{(m)}) \simeq \bar{\mathrm{R}}^{(m)}/(h_{\gamma}-1)\bar{\mathrm{R}}^{(m)}$$

that sends a 1-cocycle on Γ_{κ_w} to its image of ϕ_w . By Lemma 2.4.3 and the fact that the underlying place of F^+ of $w^{(m)}_+$ is inert in F, we have the localization map

$$\operatorname{loc}_w \colon \mathrm{H}^1_{f,\mathrm{R}}(F,\bar{\mathrm{R}}^{(m)}) \to \mathrm{H}^1_f(F_w,\bar{\mathrm{R}}^{(m)})$$

From (2.4), we know that $loc_w(s)(\Psi_{w^{(m)}})$ is a well-defined element in $(\bar{\mathbf{R}}^{(m)})^{h_{\gamma}}$ for every $s \in S$.

Lemma 2.5.4. Suppose that either one of the two assumptions in Lemma 2.3.4 is satisfied.

- (1) If $w_{+}^{(m)}$ is a γ -associated place of $F_{+}^{(m)}$, then we have $\theta_{S}(\Psi_{w^{(m)}})(s) = \operatorname{loc}_{w}(s)(\Psi_{w^{(m)}})$ for every $s \in S$.
- (2) Suppose $\gamma \in \operatorname{im} \bar{\rho}^{(m)}_+$, and that the order of h_{γ} is coprime to ℓ . Then we have

$$G_{S,\gamma} = \theta_S^{-1} \operatorname{Hom}(S, (\bar{\mathbf{R}}^{(m)})^{h_{\gamma}})$$

Proof. The same arguments in [Liu16, Lemma 5.7 and Lemma 5.8] apply to the current case as well with \mathbb{Q} replaced by F.

By Lemma 2.5.4, for every $r \in \mathbb{N}$, we have a map

$$\theta_{S,\gamma}^r \colon G_{S,\gamma}^r \to \operatorname{Hom}_{O_{\lambda}}(S, ((\bar{\mathbb{R}}^{(m)})^{h_{\gamma}})^{\oplus r}))$$

of abelian groups induced by θ_S .

Proposition 2.5.5. We make the following assumptions:

- (1) $R_{\mathbb{O}}$ is absolutely irreducible;
- (2) either one of the two assumptions in Lemma 2.3.4 is satisfied;
- (3) the order of h_{γ} is coprime to ℓ ; γ belongs to im $\bar{\rho}_{+}^{(m)}$;
- (4) $(\bar{\mathbf{R}}^{(m)})^{h_{\gamma}}$ is free over O_{λ}/λ^{m} of rank r_{γ} for some $r_{\gamma} \in \mathbb{N}$; and
- (5) S is a free $O_{\lambda}/\lambda^{m-m_0}$ -module of rank r_S for some $m_0 \in \mathbb{N}$ and $r_S \in \mathbb{N}$.

Then there exists an element $(\Psi_1, \ldots, \Psi_{r_{\gamma}r_S}) \in G_{S,\gamma}^{r_{\gamma}r_S}$ such that the image of the homomorphism $\theta_{S,\gamma}^{r_{\gamma}r_S}(\Psi_1, \ldots, \Psi_{r_{\gamma}r_S})$ contains $\lambda^{m_0+\mathfrak{f}(r_S)\mathfrak{r}_{R}}((\overline{\mathbf{R}}^{(m)})^{h_{\gamma}})^{\oplus r_S}$, where \mathfrak{r}_{R} and $\mathfrak{f}(r_S)$ are the integers appearing in Lemma 2.3.3 and Lemma 2.3.6, respectively.

Proof. By Lemma 2.3.3 and Lemma 2.3.6, the O_{λ} -submodule generated by the image of θ_S contains $\lambda^{\mathfrak{f}(r_S)\mathfrak{r}_{\mathbb{R}}} \operatorname{Hom}_{O_{\lambda}}(S, \overline{\mathbb{R}}^{(m)})$. Since h_{γ} has order coprime to ℓ , $\operatorname{Hom}_{O_{\lambda}}(S, (\overline{\mathbb{R}}^{(m)})^{h_{\gamma}})$ is a direct summand of $\operatorname{Hom}_{O_{\lambda}}(S, \overline{\mathbb{R}}^{(m)})$. It follows from Lemma 2.5.4(2) that the O_{λ} -submodule generated by $\theta_S(G_{S,\gamma})$ contains $\lambda^{\mathfrak{f}(r_S)\mathfrak{r}_{\mathbb{R}}} \operatorname{Hom}_{O_{\lambda}}(S, (\overline{\mathbb{R}}^{(m)})^{h_{\gamma}})$. As $(\overline{\mathbb{R}}^{(m)})^{h_{\gamma}}$ is free O_{λ}/λ^m -module of rank r_{γ} and S is a free $O_{\lambda}/\lambda^{m-m_0}$ -module of rank r_S , the proposition follows immediately.

Definition 2.5.6. Let the notation be as in Proposition 2.5.5. We say that an $r_{\gamma}r_{S}$ -tuple $(\Psi_{1}, \ldots, \Psi_{r_{\gamma}r_{S}}) \in G_{S,\gamma}^{r_{\gamma}r_{S}}$ is (S, γ) -abundant if the image of the homomorphism $\theta_{S,\gamma}^{r_{\gamma}r_{S}}(\Psi_{1}, \ldots, \Psi_{r_{\gamma}r_{S}})$ contains $\lambda^{m_{0}+\mathfrak{f}(r_{S})\mathfrak{r}_{R}}((\mathbb{R}^{(m)})^{h_{\gamma}})^{\oplus r_{S}}$.

Remark 2.5.7. In the applications later, we will use Proposition 2.5.5 with $r_{\gamma} = 1$ and $r_S \in \{1, 2\}$.

2.6. A Rankin–Selberg example. In this subsection, we discuss an example that is related to the Rankin–Selberg motives, which will be considered later. We take objects $R_{\alpha} \in Mod(F, O_{\lambda})_{fr}$ for $\alpha = 0, 1$ of rank $n_{\alpha} > 0$ with the associated homomorphism $\rho_{\alpha} \colon \Gamma_{F} \to GL(R_{\alpha})$, together with a $(1 - \alpha)$ -polarization $\Xi_{\alpha} \colon R_{\alpha}^{c} \xrightarrow{\sim} R_{\alpha}^{\vee}(1 - \alpha)$. We fix isomorphisms $R_{\alpha} \simeq O_{\lambda}^{\oplus n_{\alpha}}$ for $\alpha = 0, 1$.

We assume that $n_0 = 2r_0$ is even and $n_1 = 2r_1 + 1$ is odd. Put $\mathbf{R} := \mathbf{R}_0 \otimes_{O_\lambda} \mathbf{R}_1$; $\rho := \rho_0 \otimes \rho_1 \colon \Gamma_F \to \mathbf{GL}(\mathbf{R})$; and $\Xi := \Xi_0 \otimes \Xi_1 \colon \mathbf{R}^{\mathfrak{c}} \xrightarrow{\sim} \mathbf{R}^{\vee}(1)$ which is a 1-polarization of \mathbf{R} .

For a homomorphism ρ from Γ_F and a place w of F, we write ρ_w for the restriction of ρ to the subgroup Γ_{F_w} . Moreover, for clarity, we denote by $\bar{\epsilon}_{\ell}^{(m)} \colon \Gamma_{F^+} \to (O_{\lambda}/\lambda^m)^{\times}$ the reduction of ϵ_{ℓ} modulo λ^m for a positive integer m, and put $\bar{\epsilon}_{\ell} \coloneqq \bar{\epsilon}_{\ell}^{(1)}$ for simplicity.

Lemma 2.6.1. Let the notation be as above. Take a totally real F^+ -normal intermediate extension $F^+ \subseteq F' \subseteq \mathbb{C}$ (Definition 1.3.1) and a polynomial $\mathscr{P}(T) \in \mathbb{Z}[T]$. The following three statements are equivalent:

- (1) There exists a nonarchimedean place v of F^+ inert in F (under the unique place w of F) and splits completely in F' such that
 - (a) ℓ does not divide $||v|| \mathscr{P}(||v||)$;
 - (b) both $\bar{\rho}_{0,w}$ and $\bar{\rho}_{1,w}$ are unramified and semisimple;
 - (c) the trivial character appears in each of $\bar{\rho}_{0,w}$, $\bar{\rho}_{1,w}$, and $\bar{\rho}_w$ with multiplicity one;
 - (d) if $[F:F^+] = 2$, then the unramified character sending ϕ_w to -1 does not appear in $\overline{\rho}_{0,w}$;
 - (e) if $[F:F^+] = 2$, then the unramified character sending ϕ_w to $-\|v\|$ does not appear in $\bar{\rho}_{1,w}$.
- (2) For every positive integer m, the image of the restriction of the homomorphism

$$(\bar{\rho}_{0+}^{(m)}, \bar{\rho}_{1+}^{(m)}, \bar{\epsilon}_{\ell}^{(m)}) \colon \Gamma_{F^+} \to \mathscr{G}_{n_0}(O_{\lambda}/\lambda^m) \times \mathscr{G}_{n_1}(O_{\lambda}/\lambda^m) \times (O_{\lambda}/\lambda^m)^{\gamma}$$

(see Notation 2.5.1 for the notation) to $\operatorname{Gal}(\overline{F}/F')$ contains an element $(\gamma_0, \gamma_1, \xi)$ satisfying (a) $\mathscr{P}(\xi)$ is invertible in O_{λ}/λ^m ;

- (b) for $\alpha = 0, 1, \gamma_{\alpha}$ belongs to $(\operatorname{GL}_{n_{\alpha}}(O_{\lambda}/\lambda^m) \times (O_{\lambda}/\lambda^m)^{\times})\mathfrak{c}$ with order coprime to ℓ ;
- (c) the kernels of $h_{\gamma_0} 1$, $h_{\gamma_1} 1$, and $h_{\gamma_0} \otimes h_{\gamma_1} 1$ (Notation 2.5.2) are all free over O_{λ}/λ^m of rank 1;
- (d) if $[F:F^+] = 2$, then h_{γ_0} does not have an eigenvalue that is equal to -1 in O_{λ}/λ ;
- (e) if $[F:F^+] = 2$, then h_{γ_1} does not have an eigenvalue that is equal to $-\xi$ in O_{λ}/λ .
- (3) Part (2) holds for m = 1.

Proof. For a nonarchimedean place v of F^+ , we choose an arithmetic Frobenius element ϕ_v in $\Gamma_{F^+_v}$.

First, we show that (1) implies (3). Let v be such a place in (1). Then (the conjugacy class of) ϕ_v belongs to (the normal subgroup) $\operatorname{Gal}(\overline{F}/F')$ as v splits completely in F'. Let $(\gamma_0, \gamma_1, \xi)$ be the image of ϕ_v under the homomorphism $(\bar{\rho}_{0+}, \bar{\rho}_{1+}, \bar{\epsilon}_{\ell})$, which is well-defined by (1b). Then (2a) follows from (1a); (2b) follows from the fact that v is inert in F and (1b); (2c) follows from (1c); (2d) follows from (1d); and (2e) follows from (1e).

Second, we show that (3) implies (2). Suppose that $(\gamma_0, \gamma_1, \xi)$ is an element in the image of $(\bar{\rho}_{0+}, \bar{\rho}_{1+}, \bar{\epsilon}_{\ell})|_{\mathrm{Gal}(\overline{F}/F')}$ satisfying (2a)–(2e). For every given positive integer m, take an element $(\gamma'_0, \gamma'_1, \xi')$ in the image of $(\bar{\rho}_{0+}^{(m)}, \bar{\rho}_{1+}^{(m)}, \bar{\epsilon}_{\ell}^{(m)})|_{\mathrm{Gal}(\overline{F}/F')}$ whose reduction is $(\gamma_0, \gamma_1, \xi)$. Then after raising sufficiently large ℓ -power, $(\gamma'_0, \gamma'_1, \xi')$ will satisfy (2a)–(2e).

Last, we show that (2) implies (1). Take an element $(\gamma_0, \gamma_1, \xi)$ is an element in the image of $(\bar{\rho}_{0+}, \bar{\rho}_{1+}, \bar{\epsilon}_{\ell})|_{\text{Gal}(\overline{F}/F')}$ satisfying (2a)–(2d). By the Chebotarev density theorem, we can find a nonarchimedean place v of F^+ coprime to ℓ , unramified in F, and splits completely in F', such that $(\bar{\rho}_{0+}, \bar{\rho}_{1+}, \bar{\epsilon}_{\ell})$ is unramified at v and sends ϕ_v to $(\gamma_0, \gamma_1, \xi)$. Then v is inert in F by (2b). Moreover, (1a) follows from (2a); (1b) follows from (2b); (1c) follows from (2c); (1d) follows from (2d); and (1e) follows from (2e).

The lemma is proved.

Lemma 2.6.2. Take a positive integer m. Let v be a nonarchimedean place of F^+ coprime to ℓ and inert in F, such that the homomorphism $(\bar{\rho}_{0+}^{(m)}, \bar{\rho}_{1+}^{(m)}, \bar{\epsilon}_{\ell}^{(m)})$ is unramified at v and sends an arithmetic Frobenius element in $\Gamma_{F_v^+}$ to an element $(\gamma_0, \gamma_1, \xi)$ satisfying (2a)-(2c) in Lemma 2.6.1. Then all of $\mathrm{H}^1_f(F_w, \bar{\rho}_0^{(m)})$, $\mathrm{H}^1_f(F_w, \bar{\rho}_1^{(m)})$, and $\mathrm{H}^1_f(F_w, \bar{\rho}^{(m)})$ are free O_{λ}/λ^m -modules of rank 1, where w is the place of F above v.

Proof. We only consider $\mathrm{H}_{f}^{1}(F_{w}, \bar{\rho}^{(m)})$; and the other two cases are similar. By (2b), we may write $\bar{\rho}_{w}^{(m)} \simeq \bigoplus_{i=1}^{N} \rho'_{i}$ where each ρ'_{i} is residually irreducible. By (2c), we know that the trivial representation appears in $\{\rho'_{i}\}_{i=1}^{N}$ exactly once, say ρ'_{1} . Moreover, for i > 1, the residual representation of ρ'_{i} is not trivial, hence we have $\mathrm{H}_{f}^{1}(F_{w}, \rho'_{i}) = 0$. Therefore, we have

$$\mathrm{H}_{f}^{1}(F_{w},\bar{\rho}^{(m)})\simeq\mathrm{H}_{f}^{1}(F_{w},\rho_{1}')=\mathrm{H}_{f}^{1}(F_{w},O_{\lambda}/\lambda^{m})$$

which is a free O_{λ}/λ^m -module of rank 1.

In the remaining part, we discuss an example in the Rankin–Selberg case using elliptic curves. Let A_0 and A_1 be two elliptic curves over F^+ . For a rational prime ℓ (that is odd and unramified in F), we put

$$\mathbf{R}_{\alpha} \coloneqq (\operatorname{Sym}_{\mathbb{Z}_{\ell}}^{n_{\alpha}-1} \operatorname{H}^{1}_{\operatorname{\acute{e}t}}(A_{\alpha \overline{F}}, \mathbb{Z}_{\ell}))(r_{\alpha})$$

as a $\mathbb{Z}_{\ell}[\Gamma_F]$ -module for $\alpha = 0, 1$. Then \mathbb{R}_{α} is an object in $\mathsf{Mod}(F, \mathbb{Z}_{\ell})_{\mathrm{fr}}$ of rank n_{α} with a canonical $(1 - \alpha)$ -polarization $\Xi_{\alpha} \colon \mathbb{R}_{\alpha}^{\mathsf{c}} \xrightarrow{\sim} \mathbb{R}_{\alpha}^{\vee}(1 - \alpha)$. Put $\mathbb{R} \coloneqq \mathbb{R}_0 \otimes_{\mathbb{Z}_{\ell}} \mathbb{R}_1$ and $\Xi \coloneqq \Xi_0 \otimes \Xi_1$ as above.

Proposition 2.6.3. Suppose that $A_{0\overline{F}}$ and $A_{1\overline{F}}$ are not isogenous to each other and $\operatorname{End}(A_{0\overline{F}}) = \operatorname{End}(A_{1\overline{F}}) = \mathbb{Z}$. Take a totally real F^+ -normal intermediate extension $F^+ \subseteq F' \subseteq \mathbb{C}$ and a polynomial $\mathscr{P}(T) \in \mathbb{Z}[T]$. Then for sufficiently large ℓ , we have that

- (1) the image of $\bar{\rho} \colon \Gamma_F \to \operatorname{GL}(\mathbb{R} \otimes \mathbb{F}_\ell)$ contains a nontrivial scalar element;
- (2) all of $\bar{\rho}_0$, $\bar{\rho}_1$, and $\bar{\rho}_0 \otimes \bar{\rho}_1$ are absolutely irreducible; and
- (3) Lemma 2.6.1(3) holds for F', $\mathscr{P}(T)$, and the coefficient field \mathbb{F}_{ℓ} .

Proof. For $\alpha = 0, 1$ and every ℓ , we have the homomorphism

$$\bar{\rho}_{A_{\alpha,\ell}} \colon \Gamma_F \to \mathrm{GL}(\mathrm{H}^1_{\mathrm{\acute{e}t}}(A_{\alpha\overline{F}}, \mathbb{F}_\ell)) \simeq \mathrm{GL}_2(\mathbb{F}_\ell).$$

Then we have $\bar{\rho}_{\alpha} = (\text{Sym}^{n_0-1}\bar{\rho}_{A_{\alpha},\ell})(r_{\alpha})$ for $\alpha = 0, 1$. By our assumption on $A_{0\overline{F}}$ and $A_{1\overline{F}}$, and [Ser72, Théorème 6], for sufficiently large ℓ , the image of the homomorphism

$$(\bar{\rho}_{A_0,\ell},\bar{\rho}_{A_1,\ell},\bar{\epsilon}_\ell)\colon\Gamma_F\to\mathrm{GL}_2(\mathbb{F}_\ell)\times\mathrm{GL}_2(\mathbb{F}_\ell)\times\mathbb{F}_\ell^{\times}$$

consists exactly of the elements (g_0, g_1, ξ) satisfying det $g_0 = \det g_1 = \xi^{-1}$. Then both (1) and (2) follow immediately.

For (3), take an element $g \in \Gamma_F$ such that its image under $(\bar{\rho}_{A_0,\ell}, \bar{\rho}_{A_1,\ell}, \bar{\epsilon}_\ell)$ is in the conjugate class of

$$\left(\begin{pmatrix} a & 0 \\ 0 & 1 \end{pmatrix}, \begin{pmatrix} ab & 0 \\ 0 & b^{-1} \end{pmatrix}, a^{-1} \right)$$

for $a, b \in \mathbb{F}_{\ell}^{\times}$ satisfying

- O $\mathscr{P}(a^{-1}) \neq 0$,
- $O((a^{2i}(ab^2)^{2j})^{[F':F^+]} \neq 1 \text{ for } (i,j) \in \{r_0, r_0 1, \dots, 1 r_0\} \times \{r_1, r_1 1, \dots, -r_1\} \text{ except for } (0,0),$
- O $(a^{2i-1})^{[F':F^+]} \neq -1$ for $i \in \{r_0, r_0 1, \dots, 1 r_0\}$, and
- $O(a(ab^2)^{2j})^{[F':F^+]} \neq -1 \text{ for } j \in \{r_1, r_1 1, \dots, -r_1\}.$

Such pair (a, b) always exists for sufficiently large ℓ . Then it is straightforward to check that the image $g^{[F':F^+]}$ c under $(\bar{\rho}_{0+}, \bar{\rho}_{1+}, \bar{\epsilon}_{\ell})$ (under the notation of Lemma 2.6.1) satisfies (2a)–(2e) of Lemma 2.6.1. In particular, (3) follows.

3. Preliminaries on hermitian structures

In this section, we collect some constructions and results concerning objects with certain hermitian structure. In Subsection 3.1, we introduce hermitian spaces, their associated unitary groups and unitary Hecke algebras. In Subsection 3.2, we introduce unitary Shimura varieties and unitary Shimura sets. In Subsection 3.3, we review the notion of (generalized) CM types. In Subsection 3.4, we collect some facts about abelian schemes with hermitian structure, which will be parameterized by our unitary Shimura varieties. In Subsection 3.5, we introduce a moduli scheme parameterizing CM abelian varieties, which is an auxiliary moduli space in order to equip our unitary Shimura variety a moduli interpretation.

Let $N \geq 1$ be an integer.

3.1. Hermitian spaces and unitary Hecke algebras. We start by recalling the notion of the coefficient field for an automorphic representation of $\operatorname{GL}_N(\mathbb{A}_F)$. Let Π be an irreducible cuspidal automorphic (complex) representation of $\operatorname{GL}_N(\mathbb{A}_F)$.

Definition 3.1.1. The *coefficient field* of Π is defined to be the smallest subfield of \mathbb{C} , denoted by $\mathbb{Q}(\Pi)$, such that for every $\rho \in \operatorname{Aut}(\mathbb{C}/\mathbb{Q}(\Pi))$, Π and $\Pi \otimes_{\mathbb{C},\rho} \mathbb{C}$ are isomorphic.

For a nonarchimedean place w of F such that Π_w is unramified, let

$$\boldsymbol{\alpha}(\Pi_w) \coloneqq \{\alpha(\Pi_w)_1, \dots, \alpha(\Pi_w)_N\} \subseteq \mathbb{C}$$

be the Satake parameter of Π_w and $\mathbb{Q}(\Pi_w) \subseteq \mathbb{C}$ be the subfield generated by the coefficients of the polynomial

$$\prod_{i=1}^{N} \left(T - \alpha(\Pi_w)_i \cdot \sqrt{\|w\|}^{N-1} \right) \in \mathbb{C}[T].$$

Lemma 3.1.2. The coefficient field $\mathbb{Q}(\Pi)$ is the composition of $\mathbb{Q}(\Pi_w)$ for all nonarchimedean places w of F such that Π_w is unramified.

Proof. Let $\mathbb{Q}(\Pi)'$ be the composition of $\mathbb{Q}(\Pi_w)$ for such w. By the construction of unramified principal series, it is clear that for every $\gamma \in \operatorname{Aut}(\mathbb{C}/\mathbb{Q}(\Pi)')$ and every w, Π_w and $\Pi_w \otimes_{\mathbb{C},\gamma} \mathbb{C}$ have the same Satake parameter hence are isomorphic. By the strong multiplicity one property, we know that for $\gamma \in \operatorname{Aut}(\mathbb{C}/\mathbb{Q}(\Pi)')$, Π and $\Pi \otimes_{\mathbb{C},\gamma} \mathbb{C}$ are isomorphic. Thus, $\mathbb{Q}(\Pi)$ is contained in $\mathbb{Q}(\Pi)'$. Conversely, for $\gamma \in \operatorname{Aut}(\mathbb{C}/\mathbb{Q}(\Pi))$, Π_w and $\Pi_w \otimes_{\mathbb{C},\gamma} \mathbb{C}$ must have the same Satake parameter for every w, which implies that γ fixes $\mathbb{Q}(\Pi_w)$ for every w. Thus, $\mathbb{Q}(\Pi_w)$ is contained in $\mathbb{Q}(\Pi)$ for every w, which implies $\mathbb{Q}(\Pi)' \subseteq \mathbb{Q}(\Pi)$. The lemma follows. \square

Definition 3.1.3 (Abstract Satake parameter). Let *L* be a ring. For a multi-subset $\boldsymbol{\alpha} := \{\alpha_1, \ldots, \alpha_N\} \subseteq L$, we put

$$P_{\alpha}(T) \coloneqq \prod_{i=1}^{N} (T - \alpha_i) \in L[T].$$

Consider a nonarchimedean place v of F^+ not in $\Sigma^+_{\rm ram}$.

- (1) Suppose that v is inert in F. We define an *(abstract) Satake parameter* in L at v of rank N to be a multi-subset $\boldsymbol{\alpha} \subseteq L$ of cardinality N. We say that $\boldsymbol{\alpha}$ is unitary if $P_{\boldsymbol{\alpha}}(T) = (-T)^N \cdot P_{\boldsymbol{\alpha}}(T^{-1})$.
- (2) Suppose that v splits in F. We define an *(abstract) Satake parameter* in L at v of rank N to be a pair $\boldsymbol{\alpha} \coloneqq (\boldsymbol{\alpha}_1; \boldsymbol{\alpha}_2)$ of multi-subsets $\boldsymbol{\alpha}_1, \boldsymbol{\alpha}_2 \subseteq L$ of cardinality N, indexed by the two places w_1, w_2 of F above v. We say that $\boldsymbol{\alpha}$ is unitary if $P_{\boldsymbol{\alpha}_1}(T) = c \cdot T^N \cdot P_{\boldsymbol{\alpha}_2}(T^{-1})$ for some constant $c \in L^{\times}$.

For two Satake parameters α_0 and α_1 in L at v of rank n_0 and n_1 , respectively, we may form their tensor product $\alpha_0 \otimes \alpha_1$ which is of rank $n_0 n_1$ in the obvious way. If α_0 and α_1 are both unitary, then so is $\alpha_0 \otimes \alpha_1$.

Notation 3.1.4. We denote by Σ_{Π}^+ the smallest (finite) set of nonarchimedean places of F^+ containing Σ_{ram}^+ such that Π_w is unramified for every nonarchimedean place w of F not above Σ_{Π}^+ . Take a nonarchimedean place v of F^+ not in Σ_{Π}^+ .

- (1) If v is inert in F, then we put $\alpha(\Pi_v) \coloneqq \alpha(\Pi_w)$ for the unique place w of F above v.
- (2) If v splits in F into two places w_1 and w_2 , then we put $\boldsymbol{\alpha}(\Pi_v) \coloneqq (\boldsymbol{\alpha}(\Pi_{w_1}); \boldsymbol{\alpha}(\Pi_{w_2}))$.

Thus, $\boldsymbol{\alpha}(\Pi_v)$ is a Satake parameter in \mathbb{C} at v of rank N.

Definition 3.1.5. Let v be a nonarchimedean place of F^+ inert in F, and L a ring in which ||v|| is invertible. Let $P \in L[T]$ be a monic polynomial of degree N satisfying $P(T) = (-T)^N \cdot P(T^{-1})$.

- (1) When N is odd, we say that P is Tate generic at v if the constant term of $T^{-1}P(T+1)$ is invertible in L.
- (2) When N is odd, we say that P is intertwining generic at v if P(-||v||) is invertible in L.
- (3) When N is even, we say that P is *level-raising special at* v if P(||v||) = 0 and the constant term of $T^{-1}P(T + ||v||)$ is invertible in L.
- (4) When N is even, we say that P is intertwining generic at v if P(-1) is invertible in L.

Remark 3.1.6. Suppose that L is a field in the above definition. Note that when N is odd, 1 appears in the Satake parameter and all other elements appear in pairs of the form $\{\alpha, \alpha^{-1}\}$; when N is even, elements in the Satake parameter appear in pairs of the form $\{\alpha, \alpha^{-1}\}$. Then

- (1) means that 1 appears exact once in the Satake parameter;
- (2) means that the pair $\{-\|v\|, -\|v\|^{-1}\}$ does not appear in the Satake parameter;
- (3) means that the pair $\{||v||, ||v||^{-1}\}$ appears exactly once in the Satake parameter;
- (4) means that the pair $\{-1, -1\}$ does not appear in the Satake parameter.

We now introduce hermitian spaces.

Definition 3.1.7 (Hermitian space). Let R be a commutative $O_{F^+}[(\Sigma_{\text{ram}}^+)^{-1}]$ -algebra. A hermitian space over $O_F \otimes_{O_{F^+}} R$ of rank N is a free $O_F \otimes_{O_{F^+}} R$ -module V of rank N together with a perfect pairing

$$(,)_{\mathrm{V}}: \mathrm{V} \times \mathrm{V} \to O_F \otimes_{O_{F^+}} R$$

that is $O_F \otimes_{O_{F^+}} R$ -linear in the first variable and $(O_F \otimes_{O_{F^+}} R, \mathsf{c} \otimes \mathrm{id}_R)$ -linear in the second variable, and satisfies $(x, y)_V = (y, x)_V^{\mathsf{c}}$ for $x, y \in V$. We denote by U(V) the group of $O_F \otimes_{O_{F^+}} R$ -linear isometries of V, which is a reductive group over R.

Moreover, we denote by V_{\sharp} the hermitian space $V \oplus O_F \otimes_{O_{F^+}} R \cdot 1$ where 1 has norm 1. For an $O_F \otimes_{O_{F^+}} R$ -linear isometry $f \colon V \to V'$, we have the induced isometry $f_{\sharp} \colon V_{\sharp} \to V'_{\sharp}$.

Let v be a nonarchimedean place of F^+ not in Σ^+_{ram} . Let $\Lambda_{N,v}$ be the unique up to isomorphism hermitian space over $O_{F_v} = O_F \otimes_{O_{F^+}} O_{F^+_v}$ of rank N, and $U_{N,v}$ its unitary group over $O_{F^+_v}$. Under a suitable basis, the associated hermitian from of $\Lambda_{N,v}$ is given by the matrix

$$\begin{pmatrix} 0 & \cdots & 0 & 1 \\ 0 & \cdots & 1 & 0 \\ \vdots & \ddots & \vdots & \vdots \\ 1 & \cdots & 0 & 0 \end{pmatrix}$$

Consider the local spherical Hecke algebra

$$\mathbb{T}_{N,v} \coloneqq \mathbb{Z}[\mathbb{U}_{N,v}(O_{F_v^+}) \setminus \mathbb{U}_{N,v}(F_v^+) / \mathbb{U}_{N,v}(O_{F_v^+})].$$

Note that according to our convention, the unit element is $\mathbb{1}_{U_{N,v}(O_{F_v^+})}$. Let $A_{N,v}$ be the maximal split diagonal subtorus of $U_{N,v}$, and $X_*(A_{N,v})$ be its cocharacter group. Then there is a well-known Satake transform

(3.1)
$$\mathbb{T}_{N,v} \to \mathbb{Z}[\|v\|^{\pm \delta(v)/2}][A_{N,v}(F_v^+)/A_{N,v}(O_{F_v^+})] \simeq \mathbb{Z}[\|v\|^{\pm \delta(v)/2}][X_*(A_{N,v})]$$

as a homomorphism of algebras. Choose a uniformizer ϖ_v of F_v^+ .

Construction 3.1.8. Let *L* be a commutative $\mathbb{Z}[||v||^{\pm \delta(v)/2}]$ -algebra. Let α be a unitary Satake parameter in *L* at *v* of rank *N*. There are two cases.

(1) Suppose that v is inert in F. Then a set of representatives of $A_{N,v}(F_v^+)/A_{N,v}(O_{F_v^+})$ can be taken as

 $\{(\varpi_v^{t_1},\ldots,\varpi_v^{t_N}) \mid t_1,\ldots,t_N \in \mathbb{Z} \text{ satisfying } t_i + t_{N+1-i} = 0 \text{ for all } 1 \le i \le N\}.$

Choose an order in $\boldsymbol{\alpha}$ as $(\alpha_1, \ldots, \alpha_N)$ satisfying $\alpha_i \alpha_{N+1-i} = 1$; we have a unique homomorphism

$$\mathbb{Z}[\|v\|^{\pm \delta(v)/2}][\mathcal{A}_{N,v}(F_v^+)/\mathcal{A}_{N,v}(O_{F_v^+})] \to L$$

of $\mathbb{Z}[\|v\|^{\pm\delta(v)/2}]$ -algebras sending the class of $(\varpi_v^{t_1}, \ldots, \varpi_v^{t_N})$ to $\prod_{i=1}^{\lfloor \frac{N}{2} \rfloor} \alpha_i^{t_i}$. Composing with the Satake transform (3.1), we obtain a ring homomorphism

$$\phi_{\alpha} \colon \mathbb{T}_{N,v} \to L.$$

It is independent of the choices of the uniformizer ϖ_v and the order in α .

(2) Suppose that v splits in F into two places w_1 and w_2 . Then a set of representatives of $A_{N,v}(F_v^+)/A_{N,v}(O_{F_v^+})$ can be taken as

$$\left\{ \begin{pmatrix} \varpi_v^{t_1} & & \\ & \ddots & \\ & & \varpi_v^{t_N} \end{pmatrix}, \begin{pmatrix} \varpi_v^{-t_N} & & \\ & \ddots & \\ & & & \varpi_v^{-t_1} \end{pmatrix} \middle| t_1, \dots, t_N \in \mathbb{Z} \right\},$$

where the first diagonal matrix (resp. the second diagonal matrix) is regarded as an element in $A_{N,v}(F_{w_1})$ (resp. $A_{N,v}(F_{w_2})$). Choose orders in α_1 and α_2 as $(\alpha_{1,1}, \ldots, \alpha_{1,N})$ and $(\alpha_{2,1}, \ldots, \alpha_{2,N})$ satisfying $\alpha_{1,i}\alpha_{2,N+1-i} = 1$; we have a unique homomorphism

$$\mathbb{Z}[\|v\|^{\pm\delta(v)/2}][\mathcal{A}_{N,v}(F_v^+)/\mathcal{A}_{N,v}(O_{F_v^+})] \to L$$

of $\mathbb{Z}[\|v\|^{\pm\delta(v)/2}]$ -algebras sending the class of $(\varpi_v^{t_1}, \ldots, \varpi_v^{t_N}; \varpi_v^{-t_N}, \ldots, \varpi_v^{-t_1})$ to $\prod_{i=1}^N \alpha_{1,i}^{t_i}$. Composing with the Satake transform (3.1), we obtain a ring homomorphism

$$\phi_{\alpha} \colon \mathbb{T}_{N,v} \to L.$$

It is independent of the choices of the uniformizer ϖ_v , the order of the two places of F above v, and the orders in α_1 and α_2 .

Definition 3.1.9 (Abstract unitary Hecke algebra). For a finite set Σ^+ of nonarchimedean places of F^+ containing Σ^+_{ram} , we define the *abstract unitary Hecke algebra away from* Σ^+ to be the restricted tensor product

$$\mathbb{T}_N^{\Sigma^+} \coloneqq \bigotimes_v' \mathbb{T}_{N, \iota}$$

over all $v \notin \Sigma_{\infty}^+ \cup \Sigma^+$ with respect to unit elements. It is a commutative \mathbb{Z} -algebra (that is, a ring under our convention).

Construction 3.1.10. Suppose that Π satisfies $\Pi \circ \mathbf{c} \simeq \Pi^{\vee}$. For $v \notin \Sigma_{\Pi}^+$, the Satake parameter $\boldsymbol{\alpha}(\Pi_v)$ is unitary. Thus by Construction 3.1.8, we have a homomorphism

$$\phi_{\Pi} \coloneqq \bigotimes_{v \notin \Sigma_{\infty}^{+} \cup \Sigma_{\Pi}^{+}} \phi_{\alpha(\Pi_{v})} \colon \mathbb{T}_{N}^{\Sigma_{\Pi}^{+}} \to \mathbb{C}$$

which takes value in $\mathbb{Q}(\Pi)$ by Lemma 3.1.2. Here, we regard \mathbb{C} as a $\mathbb{Z}[\|v\|^{\pm \delta(v)/2}]$ -algebra by sending $\|v\|^{\pm \delta(v)/2}$ to $\sqrt{\|v\|}^{\pm \delta(v)}$. If moreover Π is regular algebraic, then it is cohomological [Clo90, Lemme 3.14]; hence $\mathbb{Q}(\Pi)$ is a number field and ϕ_{Π} takes value in $O_{\mathbb{Q}(\Pi)}$.

At last, we introduce some categories of open compact subgroups, which will be used later.

Definition 3.1.11. Let V be a hermitian space over F of rank N. Let \Box be a finite set of nonarchimedean places of F^+ .

- (1) (Neat subgroups [Lan13]) We consider $U(V)(\mathbb{A}_{F^+}^{\infty,\Box})$ as a subgroup of $\operatorname{GL}_F(V)(\mathbb{A}_F^{\infty,\Box})$ (by choosing an arbitrary basis). For a nonarchimedean place v of F^+ not in \Box and an element $g_v \in U(V)(F_v^+)$, it makes sense to talk about the eigenvalues of g_v in \overline{F}_v^+ , which contains $\overline{\mathbb{Q}}$. Let $\Gamma(g_v)$ be the subgroup of $(\overline{F}_v^+)^{\times}$ generated by the eigenvalues of g_v . Note that the torsion subgroup $\Gamma(g_v)_{\text{tors}}$ lies in $\overline{\mathbb{Q}}^{\times}$. We say an element $g = (g_v) \in U(V)(\mathbb{A}_{F^+}^{\infty,\Box})$ is *neat* if $\bigcap_{v\notin \Box} \Gamma(g_v)_{\text{tors}} = \{1\}$, and a subgroup $K \subseteq U(V)(\mathbb{A}_{F^+}^{\infty,\Box})$ is *neat* if all its elements are neat.
- (2) We define a category $\mathfrak{K}(V)^{\square}$ whose objects are neat open compact subgroups K of $U(V)(\mathbb{A}_{F^+}^{\infty,\square})$, and a morphism from K to K' is an element $g \in K \setminus U(V)(\mathbb{A}_{F^+}^{\infty,\square})/K'$ satisfying $g^{-1}Kg \subseteq K'$. Denote by $\mathfrak{K}'(V)^{\square}$ the subcategory of $\mathfrak{K}(V)^{\square}$ that allows only identity double cosets as morphisms.
- (3) We define a category $\mathfrak{K}(V)_{sp}^{\Box}$ whose objects are pairs $K = (K_{\flat}, K_{\sharp})$ where K_{\flat} is an object of $\mathfrak{K}(V)^{\Box}$ and K_{\sharp} is an object of $\mathfrak{K}(V_{\sharp})^{\Box}$ such that $K_{\flat} \subseteq K_{\sharp} \cap U(V)(\mathbb{A}_{F^{+}}^{\infty,\Box})$, and a morphism from $K = (K_{\flat}, K_{\sharp})$ to $K' = (K'_{\flat}, K'_{\sharp})$ is an element $g \in K_{\flat} \setminus U(V)(\mathbb{A}_{F^{+}}^{\infty,\Box})/K'_{\flat}$ such that $g^{-1}K_{\flat}g \subseteq K'_{\flat}$ and $g^{-1}K_{\sharp}g \subseteq K'_{\sharp}$. We have the obvious functors

$$-_{\flat} \colon \mathfrak{K}(V)^{\square}_{\operatorname{sp}} \to \mathfrak{K}(V)^{\square}, \quad -_{\sharp} \colon \mathfrak{K}(V)^{\square}_{\operatorname{sp}} \to \mathfrak{K}(V_{\sharp})^{\square}$$

sending $K = (K_{\flat}, K_{\sharp})$ to K_{\flat} and K_{\sharp} , respectively. Note that $\mathfrak{K}(V)_{sp}^{\Box}$ is a non-full subcategory of $\mathfrak{K}(V)^{\Box} \times \mathfrak{K}(V_{\sharp})^{\Box}$.

When \Box is the empty set, we suppress it from all the notations above.

3.2. Unitary Shimura varieties and sets. We introduce hermitian spaces over F that will be used in this article.

Definition 3.2.1. Let V be a hermitian space over F of rank N.

- (1) We say that V is standard definite if it has signature (N, 0) at every place in Σ_{∞}^+ .
- (2) We say that V is standard indefinite if it has signature (N-1, 1) at $\underline{\tau}_{\infty}$ and (N, 0) at other places in Σ_{∞}^+ .

First, we introduce unitary Shimura varieties. Take a standard indefinite hermitian space V over F of rank N. We have a functor

$$\operatorname{Sh}(\mathrm{V}, -) \colon \mathfrak{K}(\mathrm{V}) \to \operatorname{Sch}_{/F}$$

 $\mathrm{K} \mapsto \operatorname{Sh}(\mathrm{V}, \mathrm{K})$

of Shimura varieties associated to the reductive group $\operatorname{Res}_{F^+/\mathbb{Q}} U(V)$ and the Deligne homomorphism

h:
$$\operatorname{Res}_{\mathbb{C}/\mathbb{R}} \mathbf{G}_m \to (\operatorname{Res}_{F^+/\mathbb{Q}} \mathrm{U}(\mathrm{V})) \otimes_{\mathbb{Q}} \mathbb{R}$$

$$z \mapsto \left(\begin{pmatrix} \overline{z}/z \\ 1_{N-1} \end{pmatrix}, 1_N, \dots, 1_N \right) \in \mathrm{U}(\mathrm{V})(F_{\underline{\tau}_{\infty}}^+) \times \prod_{\underline{\tau} \in \Sigma_{\infty}^+, \underline{\tau} \neq \underline{\tau}_{\infty}} \mathrm{U}(\mathrm{V})(F_{\underline{\tau}}^+),$$

where we have identified $U(V)(F_{\mathcal{I}_{\infty}}^+)$ with a subgroup of $GL_N(\mathbb{C})$ via the complex embedding τ_{∞} of F.

Second, we introduce unitary Shimura sets. Take a standard definite hermitian space V over F of rank N. We have a functor

$$\begin{aligned} \operatorname{Sh}(\mathrm{V},-) \colon \mathfrak{K}(\mathrm{V}) &\to \mathsf{Set} \\ \mathrm{K} &\mapsto \operatorname{Sh}(\mathrm{V},\mathrm{K}) \coloneqq \mathrm{U}(\mathrm{V})(F^+) \backslash \mathrm{U}(\mathrm{V})(\mathbb{A}_{F^+}^\infty) / \mathrm{K}. \end{aligned}$$

Remark 3.2.2. Whether the notion Sh(V, -) stands for a scheme or a set depends on whether V is standard indefinite or standard definite; so there will be no confusion about notation. Of course, one can equip with Sh(V, -) a natural scheme structure when V is standard definite; but we will not take this point of view in this article.

We now recall the notion of global base change.

Definition 3.2.3 (Global base change). Let V be a hermitian space over F of rank N, and π a discrete automorphic representation of U(V)(\mathbb{A}_{F^+}).

- (1) We define a global base change of π is an automorphic representation $BC(\pi)$ of $GL_N(\mathbb{A}_F)$ that is a finite isobaric sum of discrete automorphic representations such that $BC(\pi)_v \simeq$ $BC(\pi_v)$ holds for all but finitely many nonarchimedean places v of F^+ such that π_v is unramified. By the strong multiplicity one property for GL_N [PS79], if $BC(\pi)$ exists, then it is unique up to isomorphism.
- (2) We say that π is *stable* if BC(π) exists and is cuspidal.²

Proposition 3.2.4. Let Π be a relevant representation of $GL_N(\mathbb{A}_F)$.

- (1) For every nonarchimedean place w of F, Π_w is tempered.
- (2) For every isomorphism $\iota_{\ell} \colon \mathbb{C} \xrightarrow{\sim} \overline{\mathbb{Q}}_{\ell}$, there is a semisimple continuous homomorphism

$$\rho_{\Pi,\iota_{\ell}}\colon \Gamma_F \to \mathrm{GL}_N(\overline{\mathbb{Q}}_\ell),$$

unique up to conjugation, satisfying that for every nonarchimedean place w of F, the Frobenius semisimplification of the associated Weil–Deligne representation of $\rho_{\Pi,\iota_{\ell}}|_{\Gamma_{F_w}}$ corresponds to the irreducible admissible representation $\iota_{\ell}\Pi_w |\det|_w^{\frac{1-N}{2}}$ of $\operatorname{GL}_N(F_w)$ under the local Langlands correspondence. Moreover, $\rho_{\Pi,\iota_{\ell}}^{\mathsf{c}}$ and $\rho_{\Pi,\iota_{\ell}}^{\vee}(1-N)$ are conjugate.

Proof. Part (1) is [Car12, Theorem 1.2]. For (2), the Galois representation $\rho_{\Pi,\iota_{\ell}}$ is constructed in [CH13, Theorem 3.2.3], and the local-global compatibility is obtained in [Car12, Theorem 1.1] and [Car14, Theorem 1.1]. The last property in (2) follows from the previous one and the Chebotarev density theorem.

²This is slightly more restrictive than the usual definition of stable representations by requiring $BC(\pi)$ discrete.

Definition 3.2.5. We say that a subfield $E \subseteq \mathbb{C}$ is a *strong coefficient field* of Π if E is a number field containing $\mathbb{Q}(\Pi)$ (Definition 3.1.1); and for every prime λ of E, there exists a continuous homomorphism

$$\rho_{\Pi,\lambda} \colon \Gamma_F \to \mathrm{GL}_N(E_\lambda),$$

necessarily unique up to conjugation, such that for every isomorphism $\iota_{\ell} \colon \mathbb{C} \xrightarrow{\sim} \overline{\mathbb{Q}}_{\ell}$ inducing the prime λ , $\rho_{\Pi,\lambda} \otimes_{E_{\lambda}} \overline{\mathbb{Q}}_{\ell}$ and $\rho_{\Pi,\iota_{\ell}}$ are conjugate, where $\rho_{\Pi,\iota_{\ell}}$ is the homomorphism from Proposition 3.2.4(2).

Remark 3.2.6. By [CH13, Proposition 3.2.5], strong coefficient field of Π exists. Moreover, under Hypothesis 3.2.9 below, $\mathbb{Q}(\Pi)$ is already a strong coefficient field of Π if Π is isomorphic to BC(π) for a relevant pair (V, π) (see Definition 3.2.7 below) in which V is standard *indefinite*.

Definition 3.2.7. Consider a pair (V, π) where V is a hermitian space over F, and π a discrete automorphic representation of $U(V)(\mathbb{A}_{F^+})$. We say that (V, π) is *relevant* if either one of the following two situations happens:

(1) V is standard definite, and π^{∞} appears in

$$\lim_{K \in \mathfrak{K}'(V)} \mathbb{C}[Sh(V,K)];$$

(2) V is standard indefinite, and π^{∞} appears in

$$\lim_{\mathbf{K}\in\widehat{\mathfrak{K}'}(\mathbf{V})} \iota_{\ell}^{-1} \mathbf{H}^{i}_{\mathrm{\acute{e}t}}(\mathrm{Sh}(\mathbf{V},\mathbf{K})_{\overline{F}},\overline{\mathbb{Q}}_{\ell})$$

for some isomorphism $\iota_{\ell} \colon \mathbb{C} \xrightarrow{\sim} \overline{\mathbb{Q}}_{\ell}$ and some $i \in \mathbb{Z}$.

Proposition 3.2.8. Let (V, π) be a relevant pair. Then $BC(\pi)$ exists.

k

Proof. When V is standard definite, this is proved in [Lab, Corollaire 5.3]. When V is standard indefinite, this is proved in [Shi, Theorem 1.1].³

Hypothesis 3.2.9. Consider an integer $N \geq 1$. For every standard indefinite hermitian space V over F of rank N, every discrete automorphic representation π of $U(V)(\mathbb{A}_{F^+})$ such that $BC(\pi)$ exists and is a relevant representation of $GL_N(\mathbb{A}_F)$, and every isomorphism $\iota_{\ell} \colon \mathbb{C} \xrightarrow{\sim} \overline{\mathbb{Q}}_{\ell}$, the semisimplification of the $\overline{\mathbb{Q}}_{\ell}[\Gamma_F]$ -module

$$W^{N-1}(\pi) \coloneqq \operatorname{Hom}_{\overline{\mathbb{Q}}_{\ell}[\mathrm{U}(\mathrm{V})(\mathbb{A}_{F^{+}}^{\infty})]} \left(\iota_{\ell} \pi^{\infty}, \varinjlim_{\widehat{\mathfrak{K}'}(\widetilde{\mathrm{V}})} \mathrm{H}^{N-1}_{\mathrm{\acute{e}t}}(\mathrm{Sh}(\mathrm{V}, \mathrm{K})_{\overline{F}}, \overline{\mathbb{Q}}_{\ell}) \right)$$

is isomorphic to the underlying $\overline{\mathbb{Q}}_{\ell}[\Gamma_F]$ -module of $\rho_{\mathrm{BC}(\pi),\iota_{\ell}}^{\mathsf{c}}$.

Proposition 3.2.10. Hypothesis 3.2.9 holds for $N \leq 3$, and for N > 3 if $[F^+ : \mathbb{Q}] > 1$.

Proof. By Proposition C.3.1, we know that $W^{N-1}(\pi)$ is of dimension N.

The case for N = 1 follows directly from the definition of the canonical model of Shimura varieties over reflex fields. The case for N = 2 is proved in [Liu, Theorem D.6(2)]. The case for N = 3 when $F^+ = \mathbb{Q}$ follows from the main result of [Rog92]. The case for $N \ge 3$ when $[F^+:\mathbb{Q}] > 1$ will be proved in [KSZ].

³In fact, in [Shi], the author considers the case for unitary similitude group and assumes that F contains an imaginary quadratic field. However, we can obtain the result in our setup by modifying the argument as in the proof of Theorem D.1.3.

3.3. Generalized CM type and reflexive closure. We denote by $\mathbb{N}[\Sigma_{\infty}]$ the commutative monoid freely generated by the set Σ_{∞} , which admits an action of $\operatorname{Aut}(\mathbb{C}/\mathbb{Q})$ via the set Σ_{∞} .

Definition 3.3.1. A generalized CM type of rank N is an element

$$\Psi = \sum_{\tau \in \Sigma_{\infty}} r_{\tau} \tau \in \mathbb{N}[\Sigma_{\infty}]$$

satisfying $r_{\tau} + r_{\tau^{c}} = N$ for every $\tau \in \Sigma_{\infty}$. For such Ψ , we define its *reflex field* $F_{\Psi} \subseteq \mathbb{C}$ to be the fixed subfield of the stabilizer of Ψ in Aut(\mathbb{C}/\mathbb{Q}). A *CM type* is simply a generalized CM type of rank 1. For a CM type Φ , we say that Φ contains τ if its coefficient r_{τ} equals 1.

Definition 3.3.2. We define the *reflexive closure* of F, denoted by F_{rflx} , to be the subfield of \mathbb{C} generated by F and F_{Φ} for every CM type Φ of F. Put $F_{\text{rflx}}^+ := (F_{\text{rflx}})^{c=1}$.

Remark 3.3.3. It is clear that F_{rflx} is a CM field, and is *F*-normal (Definition 1.3.1); F_{rflx}^+ is the maximal totally real subfield of F_{rflx} , and is F^+ -normal. In many cases, we have $F_{\text{rflx}} = F$ hence $F_{\text{rflx}}^+ = F^+$; for example, when *F* is Galois or contains an imaginary quadratic field.

Definition 3.3.4. We say that a prime \mathfrak{p} of F^+ is special inert if the following are satisfied:

- (1) \mathfrak{p} is inert in F;
- (2) the underlying rational prime p of \mathfrak{p} is odd and is unramified in F;
- (3) \mathfrak{p} is of degree one over \mathbb{Q} , that is, $F_{\mathfrak{p}}^+ = \mathbb{Q}_p$.

By abuse of notation, we also denote by \mathfrak{p} for its induced prime of F.

We say that a special inert prime \mathfrak{p} of F^+ is very special inert if there exists a prime \mathfrak{p}' of F^+_{rflx} above \mathfrak{p} satisfying $(F^+_{rflx})_{\mathfrak{p}'} = F^+_{\mathfrak{p}} (= \mathbb{Q}_p)$.

Remark 3.3.5. In Definition 3.3.4, (3) is proposed only for the purpose of simplifying computations on Dieudonné modules in Sections 4 and 5; it is not really necessary as results in these two sections should remain valid without (3). However, dropping (3) will vastly increase the burden of notations and computations in those two sections, where the technicality is already heavy.

In what follows in this article, we will often take a rational prime p that is unramified in F, and an isomorphism $\iota_p \colon \mathbb{C} \xrightarrow{\sim} \overline{\mathbb{Q}}_p$. By composing with ι_p , we regard Σ_{∞} also as the set of p-adic embeddings of F. We also regard \mathbb{Q}_p as a subfield of \mathbb{C} via ι_p^{-1} .

Notation 3.3.6. We introduce the following important notations.

- (1) In what follows, whenever we introduce some finite unramified extension $\mathbb{Q}_{?}^{?}$ of \mathbb{Q}_{p} , we denote by $\mathbb{Z}_{?}^{?}$ its ring of integers and put $\mathbb{F}_{?}^{?} := \mathbb{Z}_{?}^{?}/p\mathbb{Z}_{?}^{?}$.
- (2) For every $\tau \in \Sigma_{\infty}$, we denote by $\mathbb{Q}_{p}^{\tau} \subseteq \mathbb{C}$ the composition of $\tau(F)$ and \mathbb{Q}_{p} , which is unramified over \mathbb{Q}_{p} . For a scheme $S \in \operatorname{Sch}_{\mathbb{Z}_{p}^{\tau}}$ and an \mathcal{O}_{S} -module \mathcal{F} with an action $O_{F} \to \operatorname{End}_{\mathcal{O}_{S}}(\mathcal{F})$, we denote by \mathcal{F}_{τ} the maximal \mathcal{O}_{S} -submodule of \mathcal{F} on which O_{F} acts via the homomorphism $\tau: O_{F} \to \mathbb{Z}_{p}^{\tau} \to \mathcal{O}_{S}$.
- (3) We denote by $\mathbb{Q}_p^{\diamond} \subseteq \mathbb{C}$ the composition of \mathbb{Q}_p^{τ} for all $\tau \in \Phi$, which is unramified over \mathbb{Q}_p . We can identify Σ_{∞} with $\operatorname{Hom}(O_F, \mathbb{Z}_p^{\diamond}) = \operatorname{Hom}(O_F, \mathbb{F}_p^{\diamond})$. In particular, the *p*-power Frobenius map σ acts on Σ_{∞} .
- (4) For a generalized CM type Ψ of rank N, we denote by $\mathbb{Q}_p^{\Psi} \subseteq \mathbb{C}$ the composition of \mathbb{Q}_p , F, and F_{Ψ} , which is contained in \mathbb{Q}_p^{\diamond} .
- (5) For a (functor in) scheme over $\mathbb{Z}_{?}^{?}$ written like $\mathbf{X}_{?}(\cdots)$, we put $X_{?}(\cdots) \coloneqq \mathbf{X}_{?}(\cdots) \otimes_{\mathbb{Z}_{?}^{?}} \mathbb{F}_{?}^{?}$ and $\mathbf{X}_{?}^{\eta}(\cdots) \coloneqq \mathbf{X}_{?}(\cdots) \otimes_{\mathbb{Z}_{?}^{?}} \mathbb{Q}_{?}^{?}$. For a (functor in) scheme over $\mathbb{F}_{?}^{?}$ written like $X_{?}^{?}(\cdots)$, we put $\overline{X}_{?}^{?}(\cdots) \coloneqq X_{?}^{?}(\cdots) \otimes_{\mathbb{F}_{?}^{?}} \overline{\mathbb{F}}_{p}$. Similar conventions are applied to morphisms as well.

3.4. Unitary abelian schemes. We first introduce some general notations about abelian schemes.

Notation 3.4.1. Let A be an abelian scheme over a scheme S. We denote by A^{\vee} the dual abelian variety of A over S. We denote by $\mathrm{H}_{1}^{\mathrm{dR}}(A/S)$ (resp. Lie_{A/S}, and $\omega_{A/S}$) for the relative de Rham homology (resp. Lie algebra, and dual Lie algebra) of A/S, all regarded as locally free \mathcal{O}_{S} -modules. We have the following Hodge exact sequence

(3.2)
$$0 \to \omega_{A^{\vee}/S} \to \mathrm{H}_{1}^{\mathrm{dR}}(A/S) \to \mathrm{Lie}_{A/S} \to 0$$

of sheaves on S. When the base S is clear from the context, we sometimes suppress it from the notation.

Definition 3.4.2 (Unitary abelian scheme). We prescribe a subring $\mathbb{P} \subseteq \mathbb{Q}$. Let S be a scheme in $\mathsf{Sch}_{/\mathbb{P}}$.

- (1) An O_F -abelian scheme over S is a pair (A, i) in which A is an abelian scheme over S and $i: O_F \to \operatorname{End}_S(A) \otimes \mathbb{P}$ is a homomorphism of algebras.
- (2) A unitary O_F -abelian scheme over S is a triple (A, i, λ) in which (A, i) is an O_F -abelian scheme over S, and $\lambda \colon A \to A^{\vee}$ is a quasi-polarization such that $i(a^c)^{\vee} \circ \lambda = \lambda \circ i(a)$ for every $a \in O_F$, and there exists $c \in \mathbb{P}^{\times}$ making $c\lambda$ a polarization.
- (3) For two O_F -abelian schemes (A, i) and (A', i') over S, a (quasi-)homomorphism from (A, i) to (A', i') is a (quasi-)homomorphism $\varphi \colon A \to A'$ such that $\varphi \circ i(a) = i'(a) \circ \varphi$ for every $a \in O_F$. We will usually refer to such φ as an O_F -linear (quasi-)homomorphism.

Moreover, we will usually suppress the notion i if it is insensitive.

Definition 3.4.3 (Signature type). Let Ψ be a generalized CM type of rank N (Definition 3.3.1). Consider a scheme $S \in \mathsf{Sch}_{O_{F_{\Psi}} \otimes \mathbb{P}}$. We say that an O_F -abelian scheme (A, i) over S has signature type Ψ if for every $a \in O_F$, the characteristic polynomial of i(a) on $\operatorname{Lie}_{A/S}$ is given by

$$\prod_{\tau \in \Sigma_{\infty}} (T - \tau(a))^{r_{\tau}} \in \mathcal{O}_S[T].$$

Construction 3.4.4. Let K be an $O_{F_{\Psi}} \otimes \mathbb{P}$ -algebra that is an algebraically closed field. Suppose that we are given a unitary O_F -abelian scheme (A_0, i_0, λ_0) over K of signature type Φ that is a CM type, and a unitary O_F -abelian scheme (A, i, λ) over K of signature type Ψ . For every set \Box of places of \mathbb{Q} containing ∞ and the characteristic of K, if not zero, we construct a hermitian space

$$\operatorname{Hom}_{F\otimes_{\mathbb{A}}\mathbb{A}^{\square}}^{\lambda_{0},\lambda}(\operatorname{H}_{1}^{\operatorname{\acute{e}t}}(A_{0},\mathbb{A}^{\square}),\operatorname{H}_{1}^{\operatorname{\acute{e}t}}(A,\mathbb{A}^{\square}))$$

over $F \otimes_{\mathbb{Q}} \mathbb{A}^{\square} = F \otimes_{F^+} (F^+ \otimes_{\mathbb{Q}} \mathbb{A}^{\square})$, with the underlying $F \otimes_{\mathbb{Q}} \mathbb{A}^{\square}$ -module

$$\operatorname{Hom}_{F\otimes_{\mathbb{Q}}\mathbb{A}^{\square}}(\operatorname{H}_{1}^{\operatorname{\acute{e}t}}(A_{0},\mathbb{A}^{\square}),\operatorname{H}_{1}^{\operatorname{\acute{e}t}}(A,\mathbb{A}^{\square}))$$

equipped with the pairing

$$(x,y) \coloneqq i_0^{-1} \left((\lambda_{0*})^{-1} \circ y^{\vee} \circ \lambda_* \circ x \right) \in i_0^{-1} \operatorname{End}_{F \otimes_{\mathbb{Q}} \mathbb{A}^{\square}} (\operatorname{H}_1^{\operatorname{\acute{e}t}}(A_0, \mathbb{A}^{\square})) = F \otimes_{\mathbb{Q}} \mathbb{A}^{\square}.$$

Now we take a rational prime p that is unramified in F, and take the prescribed subring \mathbb{P} in Definition 3.4.2 to be $\mathbb{Z}_{(p)}$. We also choose an isomorphism $\iota_p \colon \mathbb{C} \simeq \overline{\mathbb{Q}}_p$, and adopt Notation 3.3.6.

Definition 3.4.5. Let A and B be two abelian schemes over a scheme $S \in \mathsf{Sch}_{\mathbb{Z}_{(p)}}$. We say that a quasi-homomorphism (resp. quasi-isogeny) $\varphi \colon A \to B$ is a quasi-p-homomorphism (resp. quasi-p-isogeny) if there exists some $c \in \mathbb{Z}_{(p)}^{\times}$ such that $c\varphi$ is a homomorphism (resp. isogeny). A quasi-isogeny φ is prime-to-p if both φ and φ^{-1} are quasi-p-isogenies. We say that a quasipolarization λ of A is p-principal if λ is a prime-to-p quasi-isogeny. Note that for a unitary O_F -abelian scheme (A, i, λ) , the quasi-polarization λ is a quasi-*p*-isogeny. To continue, take a generalized CM type $\Psi = \sum_{\tau \in \Sigma_{\infty}} r_{\tau} \tau$ of rank N.

Remark 3.4.6. Let A be an O_F -abelian scheme of signature type Ψ over a scheme $S \in \mathsf{Sch}_{\mathbb{Z}_p^{\tau}}$ for some $\tau \in \Sigma_{\infty}$. Then (3.2) induces a short exact sequence

$$0 \to \omega_{A^{\vee}/S,\tau} \to \mathrm{H}_{1}^{\mathrm{dR}}(A/S)_{\tau} \to \mathrm{Lie}_{A/S,\tau} \to 0$$

of locally free \mathcal{O}_S -modules of ranks $N - r_\tau$, N, and r_τ , respectively. If S belongs to $\mathsf{Sch}_{/\mathbb{Z}_p^{\Diamond}}$, then we have decompositions

$$H_1^{dR}(A/S) = \bigoplus_{\tau \in \Sigma_{\infty}} H_1^{dR}(A/S)_{\tau}$$

$$Lie_{A/S} = \bigoplus_{\tau \in \Sigma_{\infty}} Lie_{A/S,\tau},$$

$$\omega_{A/S} = \bigoplus_{\tau \in \Sigma_{\infty}} \omega_{A/S,\tau}$$

of locally free \mathcal{O}_S -modules.

Notation 3.4.7. Take $\tau \in \Sigma_{\infty}$. Let (A, λ) be a unitary O_F -abelian scheme of signature type Ψ over a scheme $S \in \mathsf{Sch}_{/\mathbb{Z}_p^r}$. We denote

$$\langle , \rangle_{\lambda,\tau} \colon \mathrm{H}_{1}^{\mathrm{dR}}(A/S)_{\tau} \times \mathrm{H}_{1}^{\mathrm{dR}}(A/S)_{\tau^{c}} \to \mathcal{O}_{S}$$

the \mathcal{O}_S -bilinear pairing induced by the polarization λ , which might be degenerate. Moreover, for an \mathcal{O}_S -submodule $\mathcal{F} \subseteq \mathrm{H}_1^{\mathrm{dR}}(A/S)_{\tau}$, we denote by $\mathcal{F}^{\perp} \subseteq \mathrm{H}_1^{\mathrm{dR}}(A/S)_{\tau^c}$ its (right) orthogonal complement under the above pairing, if λ is clear from the context.

Next we review some facts from the Serre-Tate theory [Kat81] and the Grothendieck-Messing theory [Mes72], tailored to our application. Let Ψ be a generalized CM type of rank N such that $r_{\tau}r_{\tau^c} = 0$ for every τ not above $\underline{\tau}_{\infty}$. Consider a closed immersion $S \hookrightarrow \hat{S}$ in $\operatorname{Sch}_{/\mathbb{Z}_p^{\Psi}}$ with an ideal equipped with a locally nilpotent PD-structure, and a unitary O_F -abelian scheme (A, λ) of signature type Ψ over S. We let $\operatorname{H}_1^{\operatorname{cris}}(A/\hat{S})$ be the evaluation of the first relative crystalline homology of A/S at the PD-thickening $S \hookrightarrow \hat{S}$, which is a locally free $\mathcal{O}_{\hat{S}} \otimes O_F$ -module. The polarization λ induces a pairing

(3.3)
$$\langle , \rangle_{\lambda,\tau_{\infty}}^{\operatorname{cris}} \colon \mathrm{H}_{1}^{\operatorname{cris}}(A/\hat{S})_{\tau_{\infty}} \times \mathrm{H}_{1}^{\operatorname{cris}}(A/\hat{S})_{\tau_{\infty}^{\mathsf{c}}} \to \mathcal{O}_{\hat{S}}$$

We define two groupoids

- O Def $(S, \hat{S}; A, \lambda)$, whose objects are unitary O_F -abelian schemes $(\hat{A}, \hat{\lambda})$ of signature type Ψ over \hat{S} that lift (A, λ) ;
- O Def' $(S, \hat{S}; A, \lambda)$, whose objects are pairs $(\hat{\omega}_{\tau_{\infty}}, \hat{\omega}_{\tau_{\infty}^{c}})$ where for each $\tau = \tau_{\infty}, \tau_{\infty}^{c}, \hat{\omega}_{\tau} \subseteq H_{1}^{cris}(A/\hat{S})_{\tau}$ is a subbundle that lifts $\omega_{A^{\vee}/S,\tau} \subseteq H_{1}^{dR}(A/S)_{\tau}$, such that $\langle \hat{\omega}_{\tau_{\infty}}, \hat{\omega}_{\tau_{\infty}^{c}} \rangle_{\lambda,\tau_{\infty}}^{cris} = 0$.

Proposition 3.4.8. The functor from $Def(S, \hat{S}; A, \lambda)$ to $Def'(S, \hat{S}; A, \lambda)$ sending $(\hat{A}, \hat{\lambda})$ to $(\omega_{\hat{A}^{\vee}/\hat{S}, \tau_{\infty}}, \omega_{\hat{A}^{\vee}/\hat{S}, \tau_{\infty}^{c}})$ is a natural equivalence.

Proof. By étale descent, we may replace $S \hookrightarrow \hat{S}$ by $S \otimes_{\mathbb{Z}_p^{\Psi}} \mathbb{Z}_p^{\Diamond} \hookrightarrow \hat{S} \otimes_{\mathbb{Z}_p^{\Psi}} \mathbb{Z}_p^{\Diamond}$. Then we have a decomposition

$$\mathbf{H}_{1}^{\mathrm{cris}}(A/\hat{S}) = \bigoplus_{\tau \in \Sigma_{\infty}} \mathbf{H}_{1}^{\mathrm{cris}}(A/\hat{S})_{\tau}$$

similar to the one in Notation 3.3.6. Note that for $\tau \notin \{\tau_{\infty}, \tau_{\infty}^{c}\}$, the subbundle $\omega_{A^{\vee}/S,\tau}$ has a unique lifting to either zero or the entire $H_{1}^{cris}(A/\hat{S})_{\tau}$. Thus, the proposition follows from the Serre–Tate and Grothendieck–Messing theories.

To end this subsection, we review some notions for abelian schemes in characteristic p.

Notation 3.4.9. Let A be an abelian scheme over a scheme $S \in \mathsf{Sch}_{/\mathbb{F}_p}$. Put

$$A^{(p)} \coloneqq A \times_{S,\sigma} S,$$

where σ is the absolute Frobenius morphism of S. Then we have

- (1) a canonical isomorphism $\mathrm{H}_{1}^{\mathrm{dR}}(A^{(p)}/S) \simeq \sigma^{*}\mathrm{H}_{1}^{\mathrm{dR}}(A/S)$ of \mathcal{O}_{S} -modules; (2) the Frobenius homomorphism $\mathrm{Fr}_{A} \colon A \to A^{(p)}$ which induces the Verschiebung map

$$\mathbb{V}_A \coloneqq (\mathrm{Fr}_A)_* \colon \mathrm{H}_1^{\mathrm{dR}}(A/S) \to \mathrm{H}_1^{\mathrm{dR}}(A^{(p)}/S)$$

of \mathcal{O}_S -modules;

(3) the Verschiebung homomorphism $\operatorname{Ver}_A \colon A^{(p)} \to A$ which induces the Frobenius map

$$\mathbf{F}_A \coloneqq (\operatorname{Ver}_A)_* \colon \operatorname{H}_1^{\operatorname{dR}}(A^{(p)}/S) \to \operatorname{H}_1^{\operatorname{dR}}(A/S)$$

of \mathcal{O}_S -modules.

For a subbundle H of $\mathrm{H}_{1}^{\mathrm{dR}}(A/S)$, we denote by $H^{(p)}$ the subbundle of $\mathrm{H}_{1}^{\mathrm{dR}}(A^{(p)}/S)$ that corresponds to $\sigma^* H$ under the isomorphism in (1). In what follows, we will suppress A in the notations F_A and V_A if the reference to A is clear.

Remark 3.4.10. In Notation 3.4.9, we have ker $\mathbf{F} = \operatorname{im} \mathbf{V} = \omega_{A(p)/S}$ and ker $\mathbf{V} = \operatorname{im} \mathbf{F}$.

If $S = \operatorname{Spec} \kappa$ for a field κ of characteristic p, then we have a canonical isomorphism $\mathrm{H}_{1}^{\mathrm{dR}}(A^{(p)}/\kappa) \simeq \mathrm{H}_{1}^{\mathrm{dR}}(A/\kappa) \otimes_{\kappa,\sigma} \kappa$. Thus, by abuse of notation, we have

O the (κ, σ) -linear Frobenius map $F: H_1^{dR}(A/\kappa) \to H_1^{dR}(A/\kappa)$ and O if κ is perfect, the (κ, σ^{-1}) -linear Verschiebung map $V: H_1^{dR}(A/\kappa) \to H_1^{dR}(A/\kappa)$.

Suppose that κ is perfect. Recall that we have the *covariant* Dieudonné module $\mathcal{D}(A)$ associated to the p-divisible group $A[p^{\infty}]$, which is a free $W(\kappa)$ -module, such that $\mathcal{D}(A)/p\mathcal{D}(A)$ is canonically isomorphic to $H_1^{dR}(A/\kappa)$. Moreover, again by abuse of notation, we have

O the $(W(\kappa), \sigma)$ -linear Frobenius map $F: \mathcal{D}(A) \to \mathcal{D}(A)$ lifting the one above, and

O the $(W(\kappa), \sigma^{-1})$ -linear Verschiebung map $V: \mathcal{D}(A) \to \mathcal{D}(A)$ lifting the one above,

respectively, satisfying $F \circ V = V \circ F = p$.

Remark 3.4.11. Take $\tau \in \Sigma_{\infty}$. For a scheme $S \in \mathsf{Sch}_{/\mathbb{F}_p^{\tau}}$ and an O_F -abelian scheme A over S, we have $(H_1^{dR}(A/S)_{\tau})^{(p)} = H_1^{dR}(A^{(p)}/S)_{\sigma\tau}$ under Notations 3.3.6 and 3.4.9.

If $S = \operatorname{Spec} \kappa$ for a perfect field κ (containing \mathbb{F}_p^{τ}), then applying Notation 3.3.6 to the $W(\kappa)$ module $\mathcal{D}(A)$, we obtain $W(\kappa)$ -submodules $\mathcal{D}(A)_{\sigma^i\tau} \subseteq \mathcal{D}(A)$ for every $i \in \mathbb{Z}$. From Remark 3.4.10, we obtain

O the $(W(\kappa), \sigma)$ -linear Frobenius map $F: \mathcal{D}(A)_{\tau} \to \mathcal{D}(A)_{\sigma\tau}$ and

O the $(W(\kappa), \sigma^{-1})$ -linear Verschiebung map $\mathbb{V} \colon \mathcal{D}(A)_{\tau} \to \mathcal{D}(A)_{\sigma^{-1}\tau}$

by restriction. We have canonical isomorphisms and inclusions:

$$\mathbb{V}\mathcal{D}(A)_{\sigma\tau}/p\mathcal{D}(A)_{\tau} \simeq \omega_{A^{\vee},\tau} \subseteq \mathcal{D}(A)_{\tau}/p\mathcal{D}(A)_{\tau} \simeq \mathrm{H}_{1}^{\mathrm{dR}}(A)_{\tau}$$

Notation 3.4.12. Take $\tau \in \Sigma_{\infty}$. Let (A, λ) be a unitary O_F -abelian scheme of signature type Ψ over Spec κ for a perfect field κ containing \mathbb{F}_{p}^{τ} . We have a pairing

$$\langle , \rangle_{\lambda,\tau} \colon \mathcal{D}(A)_{\tau} \times \mathcal{D}(A)_{\tau^{c}} \to W(\kappa)$$

lifting the one in Notation 3.4.7. We denote by $\mathcal{D}(A)^{\vee}_{\tau}$ the $W(\kappa)$ -dual of $\mathcal{D}(A)_{\tau}$, as a submodule of $\mathcal{D}(A)_{\tau^c} \otimes \mathbb{Q}$. In what follows, unless we specify, the dual is always with respect to the default quasi-polarization.

The following lemma will be repeatedly used in later discussion.

Lemma 3.4.13. Suppose that F^+ is contained in \mathbb{Q}_p (via the embedding $\tau: F^+ \hookrightarrow \mathbb{C} \simeq \overline{\mathbb{Q}}_p$) with \mathfrak{p} the induced p-adic prime. Let $\varpi \in O_{F^+}$ be an element such that $\operatorname{val}_{\mathfrak{p}}(\varpi) = 1$. Consider two O_F -abelian schemes A and B over a scheme $S \in \operatorname{Sch}_{/\mathbb{F}_{p^2}}$. Let $\alpha: A \to B$ and $\beta: B \to A$ be two O_F -linear quasi-p-isogenies (Definition 3.4.5) such that $\beta \circ \alpha = \varpi \cdot \operatorname{id}_A$ (hence $\alpha \circ \beta = \varpi \cdot \operatorname{id}_B$). Then

(1) For $\tau \in \{\tau_{\infty}, \tau_{\infty}^{c}\}$, the induced maps

$$\alpha_{*,\tau} \colon \mathrm{H}_{1}^{\mathrm{dR}}(A/S)_{\tau} \to \mathrm{H}_{1}^{\mathrm{dR}}(B/S)_{\tau},$$
$$\beta_{*,\tau} \colon \mathrm{H}_{1}^{\mathrm{dR}}(B/S)_{\tau} \to \mathrm{H}_{1}^{\mathrm{dR}}(A/S)_{\tau}$$

satisfy the relations ker $\alpha_{*,\tau} = \operatorname{im} \beta_{*,\tau}$ and ker $\beta_{*,\tau} = \operatorname{im} \alpha_{*,\tau}$; and these kernels and images are locally free \mathcal{O}_S -modules.

(2) We have

 $\operatorname{rank}_{\mathcal{O}_S}\operatorname{Lie}_{B/S,\tau_{\infty}}-\operatorname{rank}_{\mathcal{O}_S}\operatorname{Lie}_{A/S,\tau_{\infty}}=\operatorname{rank}_{\mathcal{O}_S}(\ker \alpha_{*,\tau_{\infty}})-\operatorname{rank}_{\mathcal{O}_S}(\ker \alpha_{*,\tau_{\infty}^{c}}).$

- (3) Let λ_A and λ_B be two quasi-polarizations on A and B, respectively, so that (A, λ_A) and (B, λ_B) become unitary O_F -abelian schemes of dimension $n[F^+ : \mathbb{Q}]$. Suppose that $\alpha^{\vee} \circ \lambda_B \circ \alpha = \varpi \lambda_A$.
 - (a) If both λ_A and λ_B are p-principal, then we have

 $\operatorname{rank}_{\mathcal{O}_S}(\ker \alpha_{*,\tau_{\infty}}) + \operatorname{rank}_{\mathcal{O}_S}(\ker \alpha_{*,\tau_{\infty}^{\mathsf{c}}}) = n.$

(b) If λ_A is p-principal and ker $\lambda_B[\mathfrak{p}^{\infty}]$ is of rank p^2 , then we have

 $\operatorname{rank}_{\mathcal{O}_{S}}(\ker \alpha_{*,\tau_{\infty}}) + \operatorname{rank}_{\mathcal{O}_{S}}(\ker \alpha_{*,\tau_{\infty}^{c}}) = n - 1.$

(c) If ker $\lambda_A[\mathfrak{p}^{\infty}]$ is of rank p^2 and λ_B is p-principal, then we have

 $\operatorname{rank}_{\mathcal{O}_S}(\ker \alpha_{*,\tau_{\infty}}) + \operatorname{rank}_{\mathcal{O}_S}(\ker \alpha_{*,\tau_{\infty}^{c}}) = n+1.$

(d) If both ker $\lambda_A[\mathfrak{p}^{\infty}]$ and ker $\lambda_B[\mathfrak{p}^{\infty}]$ are of rank p^2 , respectively, then we have

 $\operatorname{rank}_{\mathcal{O}_S}(\ker \alpha_{*,\tau_{\infty}}) + \operatorname{rank}_{\mathcal{O}_S}(\ker \alpha_{*,\tau_{\infty}^{\mathsf{c}}}) = n.$

(4) Let λ_A and λ_B be two quasi-polarizations on A and B, respectively, so that (A, λ_A) and (B, λ_B) become unitary O_F -abelian schemes of dimension $n[F^+ : \mathbb{Q}]$. Suppose that $\alpha^{\vee} \circ \lambda_B \circ \alpha = \lambda_A$. If ker $\lambda_A[\mathfrak{p}^{\infty}]$ is of rank p^2 and λ_B is p-principal, then we have

 $\operatorname{rank}_{\mathcal{O}_S}(\ker \alpha_{*,\tau_{\infty}}) + \operatorname{rank}_{\mathcal{O}_S}(\ker \alpha_{*,\tau_{\infty}^{\mathsf{c}}}) = 1.$

Proof. We may assume S connected; and up to replacing α , β and ϖ by a common $\mathbb{Z}_{(p)}^{\times}$ -multiple, we may also that α and β are genuine isogenies.

For (1), it suffices to show that the induced maps

$$\alpha_* \colon \mathrm{H}_1^{\mathrm{dR}}(A/S) \otimes_{O_{F^+}} \mathbb{Z}_p \to \mathrm{H}_1^{\mathrm{dR}}(B/S) \otimes_{O_{F^+}} \mathbb{Z}_p,$$

$$\beta_* \colon \mathrm{H}_1^{\mathrm{dR}}(B/S) \otimes_{O_{F^+}} \mathbb{Z}_p \to \mathrm{H}_1^{\mathrm{dR}}(A/S) \otimes_{O_{F^+}} \mathbb{Z}_p$$

satisfy the relations ker $\alpha_* = \operatorname{im} \beta_*$ and ker $\beta_* = \operatorname{im} \alpha_*$; and these kernels and images are locally free \mathcal{O}_S -modules.

Note that $A[\mathfrak{p}]$, $B[\mathfrak{p}]$, ker $\alpha[\mathfrak{p}]$, and ker $\beta[\mathfrak{p}]$ are all locally free finite group schemes over S with an action by $O_F/\mathfrak{p}O_F$. By the relation among α, β, ϖ , we may assume that $A[\mathfrak{p}]$ and $B[\mathfrak{p}]$ have degree p^{2d} ; ker $\alpha[\mathfrak{p}]$ has degree p^r ; and ker $\beta[\mathfrak{p}]$ has degree p^{2d-r} . As $\beta_* \circ \alpha_* = 0$ and $\alpha_* \circ \beta_* = 0$, it suffices to show that both ker α_* and im β_* (resp. both ker β_* and im α_*) are locally direct factors of $\mathrm{H}_1^{\mathrm{dR}}(A/S) \otimes_{O_{F^+}} \mathbb{Z}_p$ (resp. $\mathrm{H}_1^{\mathrm{dR}}(B/S) \otimes_{O_{F^+}} \mathbb{Z}_p$) of rank r (resp. 2d - r), which will follow if we can show that coker α_* and coker β_* are locally free \mathcal{O}_S -modules of rank r and 2d - r, respectively. We now prove that coker α_* is a locally free \mathcal{O}_S -modules of rank r; and the other case is similar. We follow the argument in [dJ93, Lemma 2.3]. Consider the big crystalline site $(S/\mathbb{Z}_p)_{cris}$ with the structural sheaf \mathcal{O}_S^{cris} . Let $\mathcal{D}(A[\mathfrak{p}^{\infty}])$ and $\mathcal{D}(B[\mathfrak{p}^{\infty}])$ denote by the covariant Dieudonné crystals on $(S/\mathbb{Z}_p)_{cris}$ of p-divisible groups $A[\mathfrak{p}^{\infty}]$ and $B[\mathfrak{p}^{\infty}]$, respectively. They are locally free \mathcal{O}_S^{cris} -modules. We have a short exact sequence

$$(3.4) \quad 0 \to \alpha_* \mathcal{D}(A[\mathfrak{p}^{\infty}]) / \varpi \mathcal{D}(B[\mathfrak{p}^{\infty}]) \to \mathcal{D}(B[\mathfrak{p}^{\infty}]) / \varpi \mathcal{D}(B[\mathfrak{p}^{\infty}]) \to \mathcal{D}(B[\mathfrak{p}^{\infty}]) / \alpha_* \mathcal{D}(A[\mathfrak{p}^{\infty}]) \to 0$$

and a surjective map

(3.5)
$$\alpha_* \colon \mathcal{D}(A[\mathfrak{p}^\infty]) / \beta_* \mathcal{D}(B[\mathfrak{p}^\infty]) \to \alpha_* \mathcal{D}(A[\mathfrak{p}^\infty]) / \varpi \mathcal{D}(B[\mathfrak{p}^\infty])$$

of $\mathcal{O}_S^{\text{cris}}$ -modules. To show that $\operatorname{coker} \alpha_*$ is a locally free \mathcal{O}_S -module of rank r, it suffices to show that $\mathcal{D}(B[\mathfrak{p}^{\infty}])/\alpha_*\mathcal{D}(A[\mathfrak{p}^{\infty}])$ is a locally free $\mathcal{O}_S^{\text{cris}}/p\mathcal{O}_S^{\text{cris}}$ -module of rank r. By [BBM82, Proposition 4.3.1], $\mathcal{D}(B[\mathfrak{p}^{\infty}])/\varpi \mathcal{D}(B[\mathfrak{p}^{\infty}])$ is a locally free $\mathcal{O}_S^{\text{cris}}/p\mathcal{O}_S^{\text{cris}}$ -module of rank 2d. Thus, by (3.4) and (3.5), it suffices to show that the $\mathcal{O}_S^{\text{cris}}/p\mathcal{O}_S^{\text{cris}}$ -modules $\alpha_*\mathcal{D}(A[\mathfrak{p}^{\infty}])/\varpi \mathcal{D}(B[\mathfrak{p}^{\infty}])$ and $\mathcal{D}(B[\mathfrak{p}^{\infty}])/\alpha_*\mathcal{D}(A[\mathfrak{p}^{\infty}])$ are locally generated by r and 2d - r sections, respectively. However, this can be easily checked using classical Dieudonné modules after base change to geometric points of S. Thus, (1) is proved.

For (2), we know from (1) that both ker $\alpha_{*,\tau_{\infty}}$ and ker $\alpha_{*,\tau_{\infty}^{c}}$ are locally free \mathcal{O}_{S} -modules. We may assume that $S = \operatorname{Spec} \kappa$ for a perfect field κ containing $\mathbb{F}_{p^{2}}$. Put $r \coloneqq \dim_{\kappa} \operatorname{Lie}_{A/\kappa,\tau_{\infty}}$ and $s \coloneqq \dim_{\kappa} \operatorname{Lie}_{B/\kappa,\tau_{\infty}}$. Then we have

$$s = \dim_{\kappa}(\omega_{B^{\vee}/\kappa,\tau_{\infty}^{\mathsf{c}}}) = \dim_{\kappa}\frac{\mathsf{V}\mathcal{D}(B)_{\tau_{\infty}}}{p\mathcal{D}(B)_{\tau_{\infty}^{\mathsf{c}}}}, \quad r = \dim_{\kappa}(\omega_{A^{\vee}/\kappa,\tau_{\infty}^{\mathsf{c}}}) = \dim_{\kappa}\frac{\mathsf{V}\mathcal{D}(A)_{\tau_{\infty}}}{p\mathcal{D}(A)_{\tau_{\infty}^{\mathsf{c}}}}$$

Thus, we obtain

(3.6)
$$s - r = \dim_{\kappa} \frac{\mathsf{V}\mathcal{D}(B)_{\tau_{\infty}}}{p\mathcal{D}(B)_{\tau_{\infty}^{c}}} - \dim_{\kappa} \frac{\mathsf{V}\mathcal{D}(A)_{\tau_{\infty}}}{p\mathcal{D}(A)_{\tau_{\infty}^{c}}}$$

Regarding $\mathcal{D}(A)$ as a submodule of $\mathcal{D}(B)$ via α_* , it follows that

$$(3.6) = \dim_{\kappa} \frac{\mathsf{V}\mathcal{D}(B)_{\tau_{\infty}}}{\mathsf{V}\mathcal{D}(A)_{\tau_{\infty}}} - \dim_{\kappa} \frac{p\mathcal{D}(B)_{\tau_{\infty}^{\mathsf{c}}}}{p\mathcal{D}(A)_{\tau_{\infty}^{\mathsf{c}}}} = \dim_{\kappa} \frac{\mathcal{D}(B)_{\tau_{\infty}}}{\mathcal{D}(A)_{\tau_{\infty}}} - \dim_{\kappa} \frac{\mathcal{D}(B)_{\tau_{\infty}^{\mathsf{c}}}}{\mathcal{D}(A)_{\tau_{\infty}^{\mathsf{c}}}} = \dim_{\kappa} (\ker \alpha_{*,\tau_{\infty}}) - \dim_{\kappa} (\ker \alpha_{*,\tau_{\infty}^{\mathsf{c}}}).$$

Thus, (2) is proved.

For (3), by assumption on λ_A , the alternating paring

$$\langle , \rangle^{\mathrm{dR}}_{\tau_{\infty}} \colon \mathrm{H}^{\mathrm{dR}}_{1}(A/S)_{\tau_{\infty}} \times \mathrm{H}^{\mathrm{dR}}_{1}(A/S)_{\tau_{\infty}^{\mathrm{c}}} \to \mathcal{O}_{S}$$

induced by λ_A is perfect.

In case (a), we have ker $\alpha_{*,\tau_{\infty}} = (\ker \alpha_{*,\tau_{\infty}^{c}})^{\perp}$, the orthogonal complement of ker $\alpha_{*,\tau_{\infty}^{c}} = \operatorname{im} \beta_{*,\tau_{\infty}^{c}}$ under $\langle , \rangle_{\tau}^{\mathrm{dR}}$, since λ_{B} is also *p*-principal.

In case (b), ker $\alpha_{*,\tau_{\infty}}$ is a subbundle of $(\ker \alpha_{*,\tau_{\infty}^{c}})^{\perp}$ of corank 1. The identity follows immediately from the identity $\operatorname{rank}_{\mathcal{O}_{S}}((\ker \alpha_{*,\tau_{\infty}^{c}})^{\perp}) + \operatorname{rank}_{\mathcal{O}_{S}}(\ker \alpha_{*,\tau_{\infty}^{c}}) = n.$

In case (c), $(\ker \alpha_{*,\tau_{\infty}})^{\perp}$ is a subbundle of $\ker \alpha_{*,\tau_{\infty}^{c}}$ of corank 1. The identity follows immediately from the identity $\operatorname{rank}_{\mathcal{O}_{S}}((\ker \alpha_{*,\tau_{\infty}})^{\perp}) + \operatorname{rank}_{\mathcal{O}_{S}}(\ker \alpha_{*,\tau_{\infty}}) = n$.

In case (d), we have both situations in (b) and (c), and the identity follows by a similar reason. The proof for (4) is similar to (3). We leave the detail to readers. \Box

3.5. A CM moduli scheme. In this subsection, we introduce an auxiliary moduli scheme parameterizing certain CM abelian varieties, which will be used in Sections 4 and 5.

Definition 3.5.1. Let R be a commutative $\mathbb{Z}[(\operatorname{disc} F)^{-1}]$ -algebra.

(1) A rational skew-hermitian space over $O_F \otimes R$ of rank N is a free $O_F \otimes R$ -module W of rank N together with an R-bilinear skew-symmetric perfect pairing

$$\langle , \rangle_{\mathrm{W}} \colon \mathrm{W} \times \mathrm{W} \to R$$

satisfying $\langle ax, y \rangle_{W} = \langle x, a^{c}y \rangle_{W}$ for every $a \in O_{F} \otimes R$ and $x, y \in W$.

- (2) Let W and W' be two rational skew-hermitian spaces over $O_F \otimes R$, a map $f: W \to W'$ is a *similitude* if f is an $O_F \otimes R$ -linear isomorphism such that there exists some $c(f) \in R^{\times}$ satisfying $\langle f(x), f(y) \rangle_{W'} = c(f) \langle x, y \rangle_W$ for every $x, y \in W$.
- (3) Two rational skew-hermitian spaces over $O_F \otimes R$ are *similar* if there exists a similitude between them.
- (4) For a rational skew-hermitian space W over $O_F \otimes R$, we denote by GU(W) its group of similitude as a reductive group over R; it satisfies that for every ring R' over R, GU(W)(R') is the set of self-similitude of the rational skew-hermitian space W $\otimes_R R'$ over $O_F \otimes R'$.

We define a subtorus $T_0 \subseteq (\operatorname{Res}_{O_F/\mathbb{Z}} \mathbf{G}_m) \otimes \mathbb{Z}[(\operatorname{disc} F)^{-1}]$ such that for every commutative $\mathbb{Z}[(\operatorname{disc} F)^{-1}]$ -algebra R,

$$T_0(R) = \{ a \in O_F \otimes R \mid \operatorname{Nm}_{F/F^+} a \in R^{\times} \}.$$

Now we take a rational prime p that is unramified in F. We take the prescribed subring \mathbb{P} in Definition 3.4.2 to be $\mathbb{Z}_{(p)}$.

Remark 3.5.2. Let W_0 be a rational skew-hermitian space over $O_F \otimes \mathbb{Z}_{(p)}$ of rank 1. Then $\mathrm{GU}(W_0)$ is canonically isomorphic to $T_0 \otimes_{\mathbb{Z}[(\operatorname{disc} F)^{-1}]} \mathbb{Z}_{(p)}$. Moreover, the set of similarity classes of rational skew-hermitian spaces W'_0 over $O_F \otimes \mathbb{Z}_{(p)}$ of rank 1 such that $W'_0 \otimes_{\mathbb{Z}_{(p)}} \mathbb{A}$ is similar to $W_0 \otimes_{\mathbb{Z}_{(p)}} \mathbb{A}$ is canonically isomorphic to

$$\ker^{1}(\mathbf{T}_{0}) \coloneqq \ker \left(\mathrm{H}^{1}(\mathbb{Q}, \mathbf{T}_{0}) \to \prod_{v \leq \infty} \mathrm{H}^{1}(\mathbb{Q}_{v}, \mathbf{T}_{0}) \right),$$

which is a finite abelian group.

Definition 3.5.3. Let Φ be a CM type. We say that a rational skew-hermitian space W_0 over $O_F \otimes \mathbb{Z}_{(p)}$ of rank 1 has type Φ if for every $x \in W_0$ and every totally imaginary element $a \in F^{\times}$ satisfying Im $\tau(a) > 0$ for all $\tau \in \Phi$, we have $\langle ax, x \rangle_{W_0} \geq 0$.

Definition 3.5.4. For a rational skew-hermitian space W_0 over $O_F \otimes \mathbb{Z}_{(p)}$ of rank 1 and type Φ and an open compact subgroup $K_0^p \subseteq T_0(\mathbb{A}^{\infty,p})$, we define a presheaf $\mathbf{T}_p^1(W_0, K_0^p)$ on $\mathsf{Sch}'_{O_{F_\Phi} \otimes \mathbb{Z}_{(p)}}$ as follows: for every $S \in \mathsf{Sch}'_{O_{F_\Phi} \otimes \mathbb{Z}_{(p)}}$, we let $\mathbf{T}_p^1(W_0, K_0^p)(S)$ be the set of equivalence classes of triples $(A_0, \lambda_0, \eta_0^p)$ where

- $O(A_0, \lambda_0)$ is a unitary O_F -abelian scheme of signature type Φ over S such that λ_0 is p-principal;
- O η_0^p is a K_0^p -level structure, that is, for a chosen geometric point *s* on every connected component of *S*, a $\pi_1(S, s)$ -invariant K_0^p -orbit of similitude

$$\eta_0^p \colon \mathrm{W}_0 \otimes_{\mathbb{Q}} \mathbb{A}^{\infty, p} \to \mathrm{H}_1^{\mathrm{\acute{e}t}}(A_{0s}, \mathbb{A}^{\infty, p})$$

of rational skew-hermitian spaces over $F \otimes_{\mathbb{Q}} \mathbb{A}^{\infty,p}$, where $\mathrm{H}_{1}^{\mathrm{\acute{e}t}}(A_{0s}, \mathbb{A}^{\infty,p})$ is equipped with the rational skew-hermitian form induced by λ_{0} .

Two triples $(A_0, \lambda_0, \eta_0^p)$ and $(A'_0, \lambda'_0, \eta_0^{p'})$ are equivalent if there exists a prime-to- $p O_F$ -linear quasiisogeny $\varphi_0: A_0 \to A'_0$ carrying (λ_0, η_0^p) to $(c\lambda'_0, \eta_0^{p'})$ for some $c \in \mathbb{Z}^{\times}_{(p)}$.

For an object $(A_0, \lambda_0, \eta_0^p) \in \mathbf{T}_p^1(W_0, K_0^p)(\mathbb{C})$, its first homology $H_1(A_0(\mathbb{C}), \mathbb{Z}_{(p)})$ is a rational skewhermitian space over $O_F \otimes \mathbb{Z}_{(p)}$ induced by λ_0 , which is of rank 1 and type Φ , and is everywhere locally similar to W_0 . Thus, by Remark 3.5.2, we obtain a map

$$\mathbf{w} \colon \mathbf{T}_p^1(\mathbf{W}_0, \mathbf{K}_0^p)(\mathbb{C}) \to \ker^1(\mathbf{T}_0)$$

sending $(A_0, \lambda_0, \eta_0^p) \in \mathbf{T}_p^1(W_0, K_0^p)(\mathbb{C})$ to the similarity class of $H_1(A_0(\mathbb{C}), \mathbb{Q})$.

It is known that when K_0^p is neat, $\mathbf{T}_p^1(W_0, K_0^p)$ is a scheme finite and étale over $O_{F_{\Phi}} \otimes \mathbb{Z}_{(p)}$. We define $\mathbf{T}_p(W_0, K_0^p)$ to be the minimal open and closed subscheme of $\mathbf{T}_p^1(W_0, K_0^p)$ containing $\mathbf{w}^{-1}(W_0)$. The group $\mathbf{T}_0(\mathbb{A}^{\infty,p})$ acts on $\mathbf{T}_p(W_0, K_0^p)$ via the formula

$$a.(A_0, \lambda_0, \eta_0^p) = (A_0, \lambda_0, \eta_0^p \circ a)$$

whose stabilizer is $T_0(\mathbb{Z}_{(p)})K_0^p$. In fact, $T_0(\mathbb{A}^{\infty,p})/T_0(\mathbb{Z}_{(p)})K_0^p$ is the Galois group of the Galois morphism

$$\mathbf{T}_p(\mathbf{W}_0, \mathbf{K}_0^p) \to \operatorname{Spec}(O_{F_\Phi} \otimes \mathbb{Z}_{(p)}).$$

р

Definition 3.5.5. We denote by \mathfrak{T} the groupoid of $T_0(\mathbb{A}^{\infty,p})/T_0(\mathbb{Z}_{(p)})K_0^p$, that is, a category with a single object * with $\operatorname{Hom}(*,*) = T_0(\mathbb{A}^{\infty,p})/T_0(\mathbb{Z}_{(p)})K_0^p$.

Remark 3.5.6. As $\mathbf{T}_p(\mathbf{W}_0, \mathbf{K}_0^p)$ is object in $\mathsf{Sch}_{/O_{F_\Phi} \otimes \mathbb{Z}_{(p)}}$ with an action by $\mathbf{T}_0(\mathbb{A}^{\infty,p})/\mathbf{T}_0(\mathbb{Z}_{(p)})\mathbf{K}_0^p$, it induces a functor from \mathfrak{T} to $\mathsf{Sch}_{/O_{F_\Phi} \otimes \mathbb{Z}_{(p)}}$, which we still denote by $\mathbf{T}_p(\mathbf{W}_0, \mathbf{K}_0^p)$. In what follows, we may often have another category \mathfrak{C} and will regard $\mathbf{T}_p(\mathbf{W}_0, \mathbf{K}_0^p)$ as a functor from $\mathfrak{C} \times \mathfrak{T}$ to $\mathsf{Sch}_{/O_{F_\Phi} \otimes \mathbb{Z}_{(p)}}$ as the composition of the projection functor $\mathfrak{C} \times \mathfrak{T} \to \mathfrak{T}$ and the functor $\mathbf{T}_p(\mathbf{W}_0, \mathbf{K}_0^p) \colon \mathfrak{T} \to \mathsf{Sch}_{/O_{F_\Phi} \otimes \mathbb{Z}_{(p)}}$.

Notation 3.5.7. For a functor $X: \mathfrak{T} \to \mathsf{Sch}$ and a coefficient ring L, we denote

$$\mathrm{H}^{i}_{\mathfrak{T}}(X, L(j)) \subseteq \mathrm{H}^{i}_{\mathrm{\acute{e}t}}(X(\ast), L(j)), \quad \mathrm{H}^{i}_{\mathfrak{T},c}(X, L(j)) \subseteq \mathrm{H}^{i}_{\mathrm{\acute{e}t},c}(X(\ast), L(j))$$

the maximal L-submodules, respectively, on which $T_0(\mathbb{A}^{\infty,p})/T_0(\mathbb{Z}_{(p)})K_0^p$ acts trivially.

Definition 3.5.8. Let κ be an algebraically closed field of characteristic p, and L a p-coprime coefficient ring. For a functor $X: \mathfrak{T} \to \mathsf{Sch}_{/\kappa}$ such that X(*) is smooth of finite type of dimension d, we define the \mathfrak{T} -trace map

$$\int_X^{\mathfrak{T}} \colon \mathrm{H}^{2d}_{\mathfrak{T},c}(X(*), L(d)) \to L$$

to be the composite map

$$\mathrm{H}^{2d}_{\mathfrak{T},c}(X(\ast),L(d)) \hookrightarrow \mathrm{H}^{2d}_{c}(X(\ast),L(d)) \to \bigoplus_{Y} \mathrm{H}^{2d}_{c}(Y,L(d)) \xrightarrow{\sum \mathrm{tr}_{Y}} L_{s}^{2d}(Y,L(d)) \xrightarrow{\sum \mathrm{tr}_{Y}} L_{s$$

where $\{Y\}$ is a set of representatives of \mathfrak{T} -orbits on the connected components of X(*), and the second map is the natural projection. It is clear that the above composite map does not depend on the choice of $\{Y\}$.
4. Unitary moduli schemes: smooth case

In this section, we define and study certain smooth integral moduli scheme whose generic fiber is the product of a unitary Shimura variety and an auxiliary CM moduli. Since the materials in this section are strictly in the linear order, we will leave the summary of contents to each subsection.

We fix a special inert prime (Definition 3.3.4) \mathfrak{p} of F^+ (with the underlying rational prime p). We take the prescribed subring \mathbb{P} in Definition 3.4.2 to be $\mathbb{Z}_{(p)}$. We choose the following data

- O a CM type Φ containing τ_{∞} ;
- O a rational skew-hermitian space W_0 over $O_F \otimes \mathbb{Z}_{(p)}$ of rank 1 and type Φ (Definition 3.5.3);
- **O** a neat open compact subgroup $K_0^p \subseteq T_0(\mathbb{A}^{\infty,p})$;
- O an isomorphism $\iota_p \colon \mathbb{C} \simeq \overline{\mathbb{Q}}_p$ such that $\iota_p \circ \underline{\tau}_\infty \colon F^+ \hookrightarrow \overline{\mathbb{Q}}_p$ induces the place \mathfrak{p} of F^+ ; O an element $\varpi \in O_{F^+}$ that is totally positive and satisfies $\operatorname{val}_{\mathfrak{p}}(\varpi) = 1$, and $\operatorname{val}_{\mathfrak{q}}(\varpi) = 0$ for every prime $\mathfrak{q} \neq \mathfrak{p}$ of F^+ above p.

We adopt Notation 3.3.6. In particular, \mathbb{F}_p^{Φ} contains \mathbb{F}_{p^2} . Since W_0 and K_0^p are insensitive and will never be changed in the remaining part of this section, we will not include them in all notations. However, we will keep the prime \mathfrak{p} in notations as, in later application, we need to choose different primes in a crucial step. Put $\mathbf{T}_{\mathfrak{p}} \coloneqq \mathbf{T}_{p}(\mathbf{W}_{0}, \mathbf{K}_{0}^{p}) \otimes_{O_{F_{\Phi}} \otimes \mathbb{Z}_{(p)}} \mathbb{Z}_{p}^{\Phi}$.

4.1. Construction of moduli schemes. In this subsection, we construct our initial moduli schemes. We start from the datum $(V, \{\Lambda_{\mathfrak{a}}\}_{\mathfrak{a}|p})$ where

O V is a standard *indefinite* hermitian space (Definition 3.1.7) over F of rank $N \geq 1$, and

O $\Lambda_{\mathfrak{q}}$ is a self-dual $O_{F_{\mathfrak{q}}}$ -lattice in $V \otimes_F F_{\mathfrak{q}}$ for every prime \mathfrak{q} of F^+ above p.

Before defining the moduli functor, we need the following lemma to make sense of the later definition.

Lemma 4.1.1. The field \mathbb{Q}_p^{Φ} contains F_{Ψ} with $\Psi = N\Phi - \tau_{\infty} + \tau_{\infty}^{\mathsf{c}}$, which is a generalized CM type of rank N, for every $N \geq 1$.

Proof. Take $\rho \in \operatorname{Aut}(\mathbb{C}/\mathbb{Q}_p^{\Phi}) \subseteq \operatorname{Aut}(\mathbb{C}/F)$. Then we have $\rho \Phi = \Phi$ and $\rho \tau_{\infty} = \tau_{\infty}$. Thus, we have $\rho(N\Phi - \tau_{\infty} + \tau_{\infty}^{\mathsf{c}}) = N\Phi - \tau_{\infty} + \tau_{\infty}^{\mathsf{c}}$ for every $N \geq 1$. The lemma follows. \Box

Recall that we have the category $\mathsf{Sch}'_{\mathbb{Z}^{\Phi}_p}$ of locally Noetherian schemes over \mathbb{Z}^{Φ}_p , and $\mathsf{PSch}'_{\mathbb{Z}^{\Phi}_p}$ the category of presheaves on $\mathsf{Sch}'_{\mathbb{Z}^{\Phi}}$.

Definition 4.1.2. We define a functor

$$\begin{split} \mathbf{M}_{\mathfrak{p}}(\mathbf{V},-) \colon \mathfrak{K}(\mathbf{V})^{p} \times \mathfrak{T} &\to \mathsf{PSch}'_{\mathbb{Z}_{p}^{\Phi}}\\ \mathbf{K}^{p} &\mapsto \mathbf{M}_{\mathfrak{p}}(\mathbf{V},\mathbf{K}^{p}) \end{split}$$

such that for every $S \in \mathsf{Sch}'_{\mathbb{Z}_p^{\Phi}}, \mathbf{M}_{\mathfrak{p}}(\mathbf{V}, \mathbf{K}^p)(S)$ is the set of equivalence classes of sextuples $(A_0, \lambda_0, \eta_0^p; A, \lambda, \eta^p)$ where

- $O(A_0, \lambda_0, \eta_0^p)$ is an element in $\mathbf{T}_{\mathfrak{p}}(S)$;
- O (A, λ) is a unitary O_F -abelian scheme of signature type $N\Phi \tau_{\infty} + \tau_{\infty}^{c}$ over S (Definitions 3.4.2 and 3.4.3) such that λ is *p*-principal;
- $O \eta^p$ is a K^p-level structure, that is, for a chosen geometric point s on every connected component of S, a $\pi_1(S, s)$ -invariant K^p-orbit of isomorphisms

$$\eta^p \colon \mathcal{V} \otimes_{\mathbb{Q}} \mathbb{A}^{\infty, p} \to \operatorname{Hom}_{F \otimes_{\mathbb{Q}} \mathbb{A}^{\infty, p}}^{\lambda_0, \lambda}(\mathcal{H}_1^{\operatorname{\acute{e}t}}(A_{0s}, \mathbb{A}^{\infty, p}), \mathcal{H}_1^{\operatorname{\acute{e}t}}(A_s, \mathbb{A}^{\infty, p}))$$

of hermitian spaces over $F \otimes_{\mathbb{Q}} \mathbb{A}^{\infty,p} = F \otimes_{F^+} \mathbb{A}^{\infty,p}_{F^+}$. See Construction 3.4.4 (with $\square =$ $\{\infty, p\}$) for the right-hand side.

Two sextuples $(A_0, \lambda_0, \eta_0^p; A, \lambda, \eta^p)$ and $(A'_0, \lambda'_0, \eta_0^{p'}; A', \lambda', \eta^{p'})$ are equivalent if there are prime-to-p O_F -linear quasi-isogenies $\varphi_0: A_0 \to A'_0$ and $\varphi: A \to A'$ such that

- $\bigcirc \varphi_0 \text{ carries } \eta_0^p \text{ to } \eta_0^{p'};$
- O there exists $c \in \mathbb{Z}_{(p)}^{\times}$ such that $\varphi_0^{\vee} \circ \lambda_0' \circ \varphi_0 = c\lambda_0$ and $\varphi^{\vee} \circ \lambda' \circ \varphi = c\lambda$;
- O the K^p-orbit of maps $v \mapsto \varphi_* \circ \eta^p(v) \circ (\varphi_{0*})^{-1}$ for $v \in V \otimes_{\mathbb{Q}} \mathbb{A}^{\infty,p}$ coincides with $\eta^{p'}$.

On the level of morphisms,

- O a morphism $g \in \mathrm{K}^p \setminus \mathrm{U}(\mathrm{V})(\mathbb{A}_F^{\infty,p})/\mathrm{K}^{p'}$ of $\mathfrak{K}(\mathrm{V})^p$ maps $\mathbf{M}_{\mathfrak{p}}(\mathrm{V},\mathrm{K}^p)(S)$ to $\mathbf{M}_{\mathfrak{p}}(\mathrm{V},\mathrm{K}^{p'})(S)$ by changing η^p to $\eta^p \circ g$; and
- **O** a morphism a of \mathfrak{T} acts on $\mathbf{M}_{\mathfrak{p}}(\mathbf{V}, \mathbf{K}^p)(S)$ by changing η_0^p to $\eta_0^p \circ a$.

We have apparently the forgetful morphism

$$(4.1) \mathbf{M}_{\mathfrak{p}}(\mathbf{V}, -) \to \mathbf{T}_{\mathfrak{p}}$$

in $\operatorname{Fun}(\mathfrak{K}(V)^p \times \mathfrak{T}, \operatorname{PSch}'_{\mathbb{Z}_p^{\Phi}})$, the category of functors from $\mathfrak{K}(V)^p \times \mathfrak{T}$ to $\operatorname{PSch}'_{\mathbb{Z}_p^{\Phi}}$. Here, we regard $\mathbf{T}_{\mathfrak{p}}$ as an object in $\operatorname{Fun}(\mathfrak{K}(V)^p \times \mathfrak{T}, \operatorname{Sch}'_{\mathbb{Z}_p^{\Phi}})$ as in Remark 3.5.6. According to Notation 3.3.6, we shall denote by the base change of (4.1) to \mathbb{F}_p^{Φ} by $\operatorname{M}_{\mathfrak{p}}(V, -) \to \operatorname{T}_{\mathfrak{p}}$, which is a morphism in $\operatorname{Fun}(\mathfrak{K}(V)^p \times \mathfrak{T}, \operatorname{PSch}'_{\mathbb{F}_p^{\Phi}})$.

Theorem 4.1.3. The morphism (4.1) is represented by a quasi-projective smooth scheme over $\mathbf{T}_{\mathfrak{p}}$ of relative dimension N-1. Moreover, for every $\mathbf{K}^p \in \mathfrak{K}(\mathbf{V})^p$, we have a canonical isomorphism for the relative tangent sheaf

$$\mathcal{T}_{\mathbf{M}_{\mathfrak{p}}(\mathbf{V},\mathbf{K}^{p})/\mathbf{T}_{\mathfrak{p}}} \simeq \mathcal{H}om\left(\omega_{\mathcal{A}^{\vee},\tau_{\infty}},\mathbf{H}_{1}^{dR}(\mathcal{A})_{\tau_{\infty}}/\omega_{\mathcal{A}^{\vee},\tau_{\infty}}\right)$$

where $(\mathcal{A}_0, \lambda_0, \eta_0^p; \mathcal{A}, \lambda, \eta^p)$ is the universal object over $\mathbf{M}_{\mathfrak{p}}(\mathbf{V}, \mathbf{K}^p)$. Moreover, (4.1) is projective if and only if its base change to \mathbb{Q}_p^{Φ} is.

Proof. This is well-known. We sketch the computation on the tangent sheaf hence the proof of the smoothness for readers' convenience. Take an object $K^p \in \mathfrak{K}(V)^p$. Since both K_0^p and K^p are neat, $\mathbf{M}_{\mathfrak{p}}(V, K^p)$ is an algebraic space. Thus, we have the universal object $(\mathcal{A}_0, \lambda_0, \eta_0^p; \mathcal{A}, \lambda, \eta^p)$ over $\mathbf{M}_{\mathfrak{p}}(V, K^p)$. By a standard argument in deformation theory, using Proposition 3.4.8, we know that the morphism $\mathbf{M}_{\mathfrak{p}}(V, K^p) \to \mathbf{T}_{\mathfrak{p}}$ is separated and smooth; and we have a canonical isomorphism for the tangent sheaf

$$\mathcal{T}_{\mathbf{M}_{\mathfrak{p}}(\mathbf{V},\mathbf{K}^{p})/\mathbf{T}_{\mathfrak{p}}} \simeq \mathcal{H}om\left(\omega_{\mathcal{A}^{\vee},\tau_{\infty}},\mathbf{H}_{1}^{\mathrm{dR}}(\mathcal{A})_{\tau_{\infty}}/\omega_{\mathcal{A}^{\vee},\tau_{\infty}}\right)$$

which is locally free of rank N - 1. Moreover, the canonical sheaf of $\mathbf{M}_{\mathfrak{p}}(\mathbf{V}, \mathbf{K}^p)$ is ample; hence $\mathbf{M}_{\mathfrak{p}}(\mathbf{V}, \mathbf{K}^p)$ is a quasi-projective scheme. The theorem is proved.

Let $K_{\mathfrak{q}}$ be the stabilizer of $\Lambda_{\mathfrak{q}}$ for every $\mathfrak{q} \mid p$; and put $K_p \coloneqq \prod_{\mathfrak{q}\mid p} K_{\mathfrak{q}}$. As show in [RSZ, Section 3.2], there is a *canonical* "moduli interpretation" isomorphism of varieties over \mathbb{Q}_p^{Φ}

(4.2)
$$\mathbf{M}_{\mathfrak{p}}^{\eta}(\mathbf{V},-) \xrightarrow{\sim} \operatorname{Sh}(\mathbf{V},-\mathbf{K}_{p}) \times_{\operatorname{Spec} F} \mathbf{T}_{\mathfrak{p}}^{\eta}$$

in $\operatorname{Fun}(\mathfrak{K}(V)^p \times \mathfrak{T}, \operatorname{Sch}_{\mathbb{Q}_p^{\Phi}})_{/\mathbf{T}_p^{\eta}}$, where \mathfrak{T} acts on $\operatorname{Sh}(V, -K_p) \times_{\operatorname{Spec} F} \mathbf{T}_p^{\eta}$ through the second factor. See also Remark 4.1.5 below.

Lemma 4.1.4. Let L be a p-coprime coefficient ring. The two specialization maps

$$\begin{aligned} & \mathrm{H}^{i}_{\mathfrak{T},c}(\mathbf{M}_{\mathfrak{p}}(\mathrm{V},-)\otimes_{\mathbb{Z}_{p}^{\Phi}}\overline{\mathbb{Q}}_{p},L) \to \mathrm{H}^{i}_{\mathfrak{T},c}(\overline{\mathrm{M}}_{\mathfrak{p}}(\mathrm{V},-),L), \\ & \mathrm{H}^{i}_{\mathfrak{T}}(\mathbf{M}_{\mathfrak{p}}(\mathrm{V},-)\otimes_{\mathbb{Z}_{p}^{\Phi}}\overline{\mathbb{Q}}_{p},L) \to \mathrm{H}^{i}_{\mathfrak{T}}(\overline{\mathrm{M}}_{\mathfrak{p}}(\mathrm{V},-),L), \end{aligned}$$

are both isomorphisms. In particular, (4.2) induces isomorphisms

$$\begin{aligned} \mathrm{H}^{i}_{\mathrm{\acute{e}t},c}(\mathrm{Sh}(\mathrm{V},-\mathrm{K}_{p})_{\overline{F}},L) &\simeq \mathrm{H}^{i}_{\mathfrak{T},c}(\overline{\mathrm{M}}_{\mathfrak{p}}(\mathrm{V},-),L), \\ \mathrm{H}^{i}_{\mathrm{\acute{e}t}}(\mathrm{Sh}(\mathrm{V},-\mathrm{K}_{p})_{\overline{F}},L) &\simeq \mathrm{H}^{i}_{\mathfrak{T}}(\overline{\mathrm{M}}_{\mathfrak{p}}(\mathrm{V},-),L), \end{aligned}$$

in $\operatorname{Fun}(\mathfrak{K}(\mathcal{V})^p, \operatorname{\mathsf{Mod}}(L[\operatorname{Gal}(\overline{\mathbb{Q}}_p/\mathbb{Q}_p^{\Phi})]))$ for every $i \in \mathbb{Z}$. Here, $\operatorname{Gal}(\overline{\mathbb{Q}}_p/\mathbb{Q}_p^{\Phi})$ is regarded as a subgroup of $\operatorname{Gal}(\overline{F}/F)$ under our fixed isomorphism $\iota_p \colon \mathbb{C} \simeq \overline{\mathbb{Q}}_p$.

Proof. Since $\mathbf{M}_{\mathfrak{p}}(\mathbf{V}, -)$ is smooth over \mathbb{Z}_p^{Φ} , we have a canonical isomorphism $L \simeq \mathbb{R}\Psi L$. When $\mathbf{M}_{\mathfrak{p}}(\mathbf{V}, -)$ is proper, this is simply the proper base change. When $\mathbf{M}_{\mathfrak{p}}(\mathbf{V}, -)$ is not proper, this follows from [LS18, Corollary 5.20].

Remark 4.1.5. For readers' convenience, we describe (4.2) on complex points, which determines the isomorphism uniquely. It suffices to assign to every point

$$x = (A_0, \lambda_0, \eta_0^p; A, \lambda, \eta^p) \in \mathbf{M}_{\mathfrak{p}}(\mathbf{V}, \mathbf{K}^p)(\mathbb{C}).$$

a point in

$$\mathrm{Sh}(\mathrm{V},\mathrm{K}^{p}\mathrm{K}_{p})(\mathbb{C}) = \mathrm{U}(\mathrm{V})(F^{+}) \setminus \left(\mathrm{V}(\mathbb{C})_{-}/\mathbb{C}^{\times} \times \mathrm{U}(\mathrm{V})(\mathbb{A}_{F^{+}}^{\infty})/\mathrm{K}^{p}\mathrm{K}_{p}\right)$$

where $V(\mathbb{C})_{-}/\mathbb{C}^{\times}$ is the set of negative definite complex lines in $V \otimes_F \mathbb{C}$. Put

$$\mathbf{V}_x \coloneqq \mathrm{Hom}_F(\mathrm{H}_1(A_0(\mathbb{C}), \mathbb{Q}), \mathrm{H}_1(A(\mathbb{C}), \mathbb{Q}))$$

equipped with a pairing in the way similar to Construction 3.4.4, which becomes a hermitian space over F of rank N. Moreover, it is standard indefinite. By the comparison between singular homology and étale homology, we have a canonical isometry of hermitian spaces

$$\rho\colon \mathcal{V}_x\otimes_{\mathbb{Q}}\mathbb{A}^{\infty,p}\xrightarrow{\sim} \mathrm{Hom}_{F\otimes_{\mathbb{Q}}\mathbb{A}^{\infty,p}}^{\lambda_0,\lambda}(\mathcal{H}_1^{\mathrm{\acute{e}t}}(A_0,\mathbb{A}^{\infty,p}),\mathcal{H}_1^{\mathrm{\acute{e}t}}(A,\mathbb{A}^{\infty,p}))$$

which implies that $V_x \otimes_{\mathbb{Q}} \mathbb{A}^{\infty,p} \simeq V \otimes_{\mathbb{Q}} \mathbb{A}^{\infty,p}$ by the existence of the level structure η^p . On the other hand, we have a canonical decomposition

$$\operatorname{Hom}_{O_F \otimes \mathbb{Z}_p}(\operatorname{H}_1^{\operatorname{\acute{e}t}}(A_0, \mathbb{Z}_p), \operatorname{H}_1^{\operatorname{\acute{e}t}}(A, \mathbb{Z}_p)) = \bigoplus_{\mathfrak{q} \mid p} \Lambda_{x, \mathfrak{q}}$$

of $O_F \otimes \mathbb{Z}_p$ -modules in which $\Lambda_{x,\mathfrak{q}}$ is a self-dual lattice in $V \otimes_F F_{\mathfrak{q}}$ for every prime \mathfrak{q} of F^+ above p. Thus, by the Hasse principle for hermitian spaces, this implies that hermitian spaces V_x and V are isomorphic. Choose an isometry $\eta_{rat} \colon V_x \to V$. Thus, we obtain an isometry

$$g^p \coloneqq \eta_{\mathrm{rat}} \circ \rho^{-1} \circ \eta^p \colon \mathcal{V} \otimes_{\mathbb{Q}} \mathbb{A}^{\infty, p} \to \mathcal{V} \otimes_{\mathbb{Q}} \mathbb{A}^{\infty, p}$$

as an element in $U(V)(\mathbb{A}_{F^+}^{\infty,p})$. For every \mathfrak{q} above p, there exists an element $g_{\mathfrak{q}} \in U(V)(F_{\mathfrak{q}}^+)$ such that $g_{\mathfrak{q}}\Lambda_{\mathfrak{q}} = \eta_{\mathrm{rat}}\Lambda_{x,\mathfrak{q}}$. Together, we obtain an element $g_x \coloneqq (g^p, (g_{\mathfrak{q}})_{\mathfrak{q}|p}) \in U(V)(\mathbb{A}_{F^+}^\infty)$. Finally,

$$l_x \coloneqq \{ \alpha \in \operatorname{Hom}_F(\operatorname{H}_1^{\operatorname{dR}}(A_0/\mathbb{C}), \operatorname{H}_1^{\operatorname{dR}}(A/\mathbb{C})) \mid \alpha(\omega_{A_0^{\vee}, \tau_{\infty}}) \subseteq \omega_{A^{\vee}, \tau_{\infty}} \}$$

is a line in $V_x(\mathbb{C})$ such that $\eta_{rat}(l_x)$ is an element in $V(\mathbb{C})_-/\mathbb{C}^{\times}$. It is easy to check that the coset

$$\mathrm{U}(\mathrm{V})(F^+)(\eta_{\mathrm{rat}}(l_x), g_x\mathrm{K}^p\mathrm{K}_p)$$

does not depend on the choice of η_{rat} , hence gives rise an element in $\text{Sh}(V, K^p K_p)(\mathbb{C})$. It is clear that the action of a morphism a of \mathfrak{T} on x does not change the above coset. 4.2. Basic correspondence on special fiber. In this subsection, we construct and study the basic correspondence on the special fiber $M_{\mathfrak{p}}(V, -)$. Recall that we have chosen an element $\varpi \in O_{F^+}$ that is totally positive and satisfies $\operatorname{val}_{\mathfrak{p}}(\varpi) = 1$, and $\operatorname{val}_{\mathfrak{q}}(\varpi) = 0$ for every prime $\mathfrak{q} \neq \mathfrak{p}$ of F^+ above p.

Definition 4.2.1. We define a functor

$$\begin{split} \mathrm{S}_{\mathfrak{p}}(\mathrm{V},-) \colon \mathfrak{K}(\mathrm{V})^{p} \times \mathfrak{T} &\to \mathsf{PSch}'_{/\mathbb{F}_{p}^{\Phi}} \\ \mathrm{K}^{p} &\mapsto \mathrm{S}_{\mathfrak{p}}(\mathrm{V},\mathrm{K}^{p}) \end{split}$$

such that for every $S \in \mathsf{Sch}'_{/\mathbb{F}_p^{\Phi}}$, $S_{\mathfrak{p}}(V, K^p)(S)$ is the set of equivalence classes of sextuples $(A_0, \lambda_0, \eta_0^p; A^\star, \lambda^\star, \eta^{p\star})$ where

- O $(A_0, \lambda_0, \eta_0^p)$ is an element in $T_{\mathfrak{p}}(S)$;
- O (A^*, λ^*) is a unitary O_F -abelian scheme of signature type $N\Phi$ over S such that ker $\lambda^*[p^{\infty}]$ is trivial (resp. contained in $A^*[\mathfrak{p}]$ of rank p^2) if N is odd (resp. even);
- $\bigcirc \eta^{p\star}$ is, for a chosen geometric point s on every connected component of S, a $\pi_1(S, s)$ invariant K^p-orbit of isomorphisms

$$\eta^{p\star} \colon \mathcal{V} \otimes_{\mathbb{Q}} \mathbb{A}^{\infty,p} \to \operatorname{Hom}_{F \otimes_{\mathbb{Q}} \mathbb{A}^{\infty,p}}^{\varpi\lambda_0,\lambda^{\star}}(\mathcal{H}_1^{\operatorname{\acute{e}t}}(A_{0s},\mathbb{A}^{\infty,p}),\mathcal{H}_1^{\operatorname{\acute{e}t}}(A_s^{\star},\mathbb{A}^{\infty,p}))$$

of hermitian spaces over $F \otimes_{\mathbb{Q}} \mathbb{A}^{\infty,p} = F \otimes_{F^+} \mathbb{A}_{F^+}^{\infty,p}$. Note that here we are using $\varpi \lambda_0$ rather that λ_0 .

The equivalence relation and the action of morphisms in $\mathfrak{K}(\mathbf{V})^p \times \mathfrak{T}$ are defined similarly as in Definition 4.1.2.

We have apparently the forgetful morphism

$$S_{\mathfrak{p}}(V, -) \to T_{\mathfrak{p}}$$

in $\mathsf{Fun}(\mathfrak{K}(V)^p \times \mathfrak{T}, \mathsf{PSch}'_{/\mathbb{F}_n^{\Phi}})$ which is represented by finite and étale schemes.

Now we take a point $s^{\star} = (A_0, \lambda_0, \eta_0^p; A^{\star}, \lambda^{\star}, \eta^{p\star}) \in S_{\mathfrak{p}}(V, K^p)(\kappa)$ where κ is a field containing \mathbb{F}_p^{Φ} . By Remark 3.4.10, we have the (κ, σ) -linear Frobenius map

$$\mathbf{F} \colon \mathrm{H}_{1}^{\mathrm{dR}}(A^{\star}/\kappa)_{\tau_{\infty}} \to \mathrm{H}_{1}^{\mathrm{dR}}(A^{\star}/\kappa)_{\sigma\tau_{\infty}} = \mathrm{H}_{1}^{\mathrm{dR}}(A^{\star}/\kappa)_{\tau_{\infty}^{\mathsf{c}}}.$$

We define a pairing

$$\{ , \}_{s^{\star}} \colon \mathrm{H}_{1}^{\mathrm{dR}}(A^{\star}/\kappa)_{\tau_{\infty}} \times \mathrm{H}_{1}^{\mathrm{dR}}(A^{\star}/\kappa)_{\tau_{\infty}} \to \mathrm{H}_{2}^{\mathrm{dR}}(A^{\star}/\kappa)_{\tau_{\infty}} \to \mathrm{H}_{2}^{\mathrm{$$

by the formula $\{x, y\}_{s^{\star}} \coloneqq \langle \mathsf{F}x, y \rangle_{\lambda^{\star}, \tau^{c}_{\infty}}$ (Notation 3.4.7). To ease notation, we put

 $\mathscr{V}_{s^{\star}} \coloneqq \mathrm{H}_{1}^{\mathrm{dR}}(A^{\star}/\kappa)_{\tau_{\infty}}.$

Lemma 4.2.2. The pair $(\mathscr{V}_{s^*}, \{,\}_{s^*})$ is admissible of rank N (Definition A.1.1). In particular, the Deligne-Lusztig variety $\mathrm{DL}_{s^*} := \mathrm{DL}(\mathscr{V}_{s^*}, \{,\}_{s^*}, \lceil \frac{N+1}{2} \rceil)$ (Definition A.1.2) is a geometrically irreducible projective smooth scheme in $\mathrm{Sch}_{/\kappa}$ of dimension $\lfloor \frac{N-1}{2} \rfloor$ with a canonical isomorphism for its tangent sheaf

 $\mathcal{T}_{\mathrm{DL}_{s^{\star}}/\kappa} \simeq \mathcal{H}om\left(\mathcal{H}/\mathcal{H}^{\dashv}, (\mathscr{V}_{s^{\star}})_{\mathrm{DL}_{s^{\star}}}/\mathcal{H}\right)$

where $\mathcal{H} \subseteq (\mathscr{V}_{s^*})_{\mathrm{DL}_{s^*}}$ is the universal subbundle.

Proof. It follows from the construction that $\{, \}_{s^*}$ is (κ, σ) -linear in the first variable and κ -linear in the second variable. By the signature condition Definition 4.2.1(2), the map $\mathbf{F} \colon \mathrm{H}_1^{\mathrm{dR}}(A^*/\kappa)_{\tau_{\infty}} \to \mathrm{H}_1^{\mathrm{dR}}(A^*/\kappa)_{\tau_{\infty}^{\mathsf{c}}}$ is an isomorphism, and the pairing $\langle \mathbf{F}, \rangle_{\lambda^*,\tau_{\infty}^{\mathsf{c}}}$ has kernel of rank 0 (resp. 1) if N is odd (resp. even). Thus, by Proposition A.1.3, it suffices to show that $(\mathscr{V}_{s^*}, \{, \}_{s^*})$ is admissible.

Note that we have a canonical isomorphism $(\mathscr{V}_{s^{\star}})_{\overline{\kappa}} = \mathrm{H}_{1}^{\mathrm{dR}}(A^{\star}/\kappa)_{\tau_{\infty}} \otimes_{\kappa} \overline{\kappa} \simeq \mathrm{H}_{1}^{\mathrm{dR}}(A^{\star}_{\overline{\kappa}}/\overline{\kappa})_{\tau_{\infty}}$, and that the $(\overline{\kappa}, \sigma)$ -linear Frobenius map $F: \mathrm{H}_{1}^{\mathrm{dR}}(A^{\star}_{\overline{\kappa}}/\overline{\kappa})_{\tau_{\infty}} \to \mathrm{H}_{1}^{\mathrm{dR}}(A^{\star}_{\overline{\kappa}}/\overline{\kappa})_{\tau_{\infty}^{c}}$ and the $(\overline{\kappa}, \sigma^{-1})$ -linear

Verschiebung map $V: \operatorname{H}_{1}^{\mathrm{dR}}(A_{\overline{\kappa}}^{\star}/\overline{\kappa})_{\tau_{\infty}} \to \operatorname{H}_{1}^{\mathrm{dR}}(A_{\overline{\kappa}}^{\star}/\overline{\kappa})_{\tau_{\infty}^{c}}$ are both isomorphisms. Thus, we obtain a $(\overline{\kappa}, \sigma^{2})$ -linear isomorphism $V^{-1}F: \operatorname{H}_{1}^{\mathrm{dR}}(A_{\overline{\kappa}}^{\star}/\overline{\kappa})_{\tau_{\infty}} \to \operatorname{H}_{1}^{\mathrm{dR}}(A_{\overline{\kappa}}^{\star}/\overline{\kappa})_{\tau_{\infty}}$. Denote by \mathscr{V}_{0} the subset of $\operatorname{H}_{1}^{\mathrm{dR}}(A_{\overline{\kappa}}^{\star}/\overline{\kappa})_{\tau_{\infty}}$ on which $V^{-1}F = \operatorname{id}$, which is an $\mathbb{F}_{p^{2}}$ -linear subspace. Since the *p*-divisible group $A[\mathfrak{p}^{\infty}]$ is supersingular, by Dieudonné's classification of crystals, the canonical map $\mathscr{V}_{0} \otimes_{\mathbb{F}_{p^{2}}} \overline{\kappa} \to \operatorname{H}_{1}^{\mathrm{dR}}(A^{\star}/\overline{\kappa})_{\tau_{\infty}} = (\mathscr{V}_{s^{\star}})_{\overline{\kappa}}$ is an isomorphism. For $x, y \in \mathscr{V}_{0}$, we have

$$\{x,y\}_{s^{\star}} = \langle \mathsf{F}x,y \rangle_{\lambda^{\star},\tau_{\infty}^{\mathsf{c}}} = \langle x, \mathsf{V}y \rangle_{\lambda^{\star},\tau_{\infty}}^{\sigma} = \langle x, \mathsf{F}y \rangle_{\lambda^{\star},\tau_{\infty}}^{\sigma} = -\langle \mathsf{F}y,x \rangle_{\lambda^{\star},\tau_{\infty}}^{\sigma} = -\{y,x\}_{s^{\star}}^{\sigma}.$$

Thus, $(\mathscr{V}_{s^*}, \{,\}_{s^*})$ is admissible. The lemma follows.

Definition 4.2.3. We define a functor

$$\begin{split} \mathbf{B}_{\mathfrak{p}}(\mathbf{V},-) \colon \mathfrak{K}(\mathbf{V})^{p} \times \mathfrak{T} &\to \mathsf{PSch}_{/\mathbb{F}_{p}^{\Phi}}^{/\mathbb{F}_{p}^{\Phi}} \\ \mathbf{K}^{p} &\mapsto \mathbf{B}_{\mathfrak{p}}(\mathbf{V},\mathbf{K}^{p}) \end{split}$$

such that for every $S \in \mathsf{Sch}'_{/\mathbb{F}_p^{\Phi}}$, $B_{\mathfrak{p}}(V, K^p)(S)$ is the set of equivalence classes of decuples $(A_0, \lambda_0, \eta_0^p; A, \lambda, \eta^p; A^*, \lambda^*, \eta^{p*}; \alpha)$ where

- O $(A_0, \lambda_0, \eta_0^p; A, \lambda, \eta^p)$ is an element of $M_{\mathfrak{p}}(V, K^p)(S)$;
- $O(A_0, \lambda_0, \eta_0^p; A^\star, \lambda^\star, \eta^{p\star})$ is an element of $S_{\mathfrak{p}}(V, K^p)(S)$;
- $O \alpha : A \to A^*$ is an O_F -linear quasi-*p*-isogeny (Definition 3.4.5) such that
 - (a) ker $\alpha[p^{\infty}]$ is contained in $A[\mathfrak{p}]$;
 - (b) we have $\varpi \cdot \lambda = \alpha^{\vee} \circ \lambda^{\star} \circ \alpha$; and
 - (c) the K^{*p*}-orbit of maps $v \mapsto \alpha_* \circ \eta^p(v)$ for $v \in V \otimes_{\mathbb{Q}} \mathbb{A}^{\infty,p}$ coincides with η^{p*} .

Two decuples $(A_0, \lambda_0, \eta_0^p; A, \lambda, \eta^p; A^\star, \lambda^\star, \eta^{p\star}; \alpha)$ and $(A'_0, \lambda'_0, \eta_0^{p\prime}; A', \lambda', \eta^{p\prime}; A^{\star\prime}, \lambda^{\star\prime}, \eta^{p\star\prime}; \alpha')$ are equivalent if there are prime-to-p O_F -linear quasi-isogenies $\varphi_0: A_0 \to A'_0, \varphi: A \to A'$, and $\varphi^\star: A^\star \to A^{\star\prime}$ such that

- O φ_0 carries η_0^p to $\eta_0^{p'}$;
- O there exists $c \in \mathbb{Z}_{(p)}^{\times}$ such that $\varphi_0^{\vee} \circ \lambda_0' \circ \varphi_0 = c\lambda_0$, $\varphi^{\vee} \circ \lambda' \circ \varphi = c\lambda$, and $\varphi^{\star \vee} \circ \lambda^{\star \prime} \circ \varphi^{\star} = c\lambda^{\star}$;
- O the K^p-orbit of maps $v \mapsto \varphi_* \circ \eta^p(v) \circ (\varphi_{0*})^{-1}$ for $v \in V \otimes_{\mathbb{Q}} \mathbb{A}^{\infty,p}$ coincides with $\eta^{p'}$;
- O the K^p-orbit of maps $v \mapsto \varphi_*^{\star} \circ \eta^{p\star}(v) \circ (\varphi_{0*})^{-1}$ for $v \in V \otimes_{\mathbb{Q}} \mathbb{A}^{\infty,p}$ coincides with $\eta^{p\star'}$;
- **O** $\varphi^* \circ \alpha = \alpha' \circ \varphi$ holds.

On the level of morphisms,

- O a morphism $g \in \mathrm{K}^p \setminus \mathrm{U}(\mathrm{V})(\mathbb{A}_F^{\infty,p})/\mathrm{K}^{p'}$ of $\mathfrak{K}(\mathrm{V})^p$ maps $\mathrm{B}_{\mathfrak{p}}(\mathrm{V},\mathrm{K}^p)(S)$ to $\mathrm{B}_{\mathfrak{p}}(\mathrm{V},\mathrm{K}^{p'})(S)$ by changing $\eta^p, \eta^{p\star}$ to $\eta^p \circ g, \eta^{p\star} \circ g$, respectively; and
- O a morphism a of \mathfrak{T} acts on $M_{\mathfrak{p}}(V, K^p)(S)$ by changing η_0^p to $\eta_0^p \circ a$.

We obtain in the obvious way a correspondence

(4.3)
$$S_{\mathfrak{p}}(V, -) \stackrel{\pi}{\longrightarrow} B_{\mathfrak{p}}(V, -) \stackrel{\iota}{\longrightarrow} M_{\mathfrak{p}}(V, -)$$

in $\operatorname{Fun}(\mathfrak{K}(V)^p \times \mathfrak{T}, \operatorname{PSch}'_{/\mathbb{F}_n^\Phi})_{/\mathrm{T}_p}$.

Definition 4.2.4 (Basic correspondence). We refer to (4.3) as the *basic correspondence*⁴ on $M_{\mathfrak{p}}(V, -)$, with $S_{\mathfrak{p}}(V, -)$ being the *source* of the basic correspondence.

Theorem 4.2.5. In the diagram (4.3), take a point

 $s^{\star} = (A_0, \lambda_0, \eta_0^p; A^{\star}, \lambda^{\star}, \eta^{p\star}) \in \mathcal{S}_{\mathfrak{p}}(\mathcal{V}, \mathcal{K}^p)(\kappa)$

where κ is a field containing \mathbb{F}_p^{Φ} . Put $B_{s^*} \coloneqq \pi^{-1}(s^*)$, and denote by $(\mathcal{A}, \lambda, \eta^p; \alpha)$ the universal object over the fiber B_{s^*} .

⁴We adopt this terminology since the image of ι is in fact the basic locus of $M_{\mathfrak{p}}(V, -)$.

(1) The fiber B_{s^*} is a smooth scheme over κ , with a canonical isomorphism for its tangent bundle

$$\mathcal{T}_{\mathrm{B}_{s^{\star}/\kappa}} \simeq \mathcal{H}om\left(\omega_{\mathcal{A}^{\vee},\tau_{\infty}}, \ker \alpha_{*,\tau_{\infty}}/\omega_{\mathcal{A}^{\vee},\tau_{\infty}}\right).$$

(2) The restriction of ι to B_{s^*} is locally on B_{s^*} a closed immersion, with a canonical isomorphism for its normal bundle

$$\mathcal{N}_{\iota|B_{s^{\star}}} \simeq \mathcal{H}om\left(\omega_{\mathcal{A}^{\vee},\tau_{\infty}}, \operatorname{im} \alpha_{*,\tau_{\infty}}\right).$$

(3) The assignment sending a point $(A_0, \lambda_0, \eta_0^p; A, \lambda, \eta^p; A^\star, \lambda^\star, \eta^{p\star}; \alpha) \in B_{s^\star}(S)$ for every $S \in Sch'_{/\kappa}$ to the subbundle

$$H \coloneqq (\check{\alpha}_{*,\tau_{\infty}})^{-1} \omega_{A^{\vee}/S,\tau_{\infty}} \subseteq \mathrm{H}_{1}^{\mathrm{dR}}(A^{\star}/S)_{\tau_{\infty}} = \mathrm{H}_{1}^{\mathrm{dR}}(A^{\star}/\kappa)_{\tau_{\infty}} \otimes_{\kappa} \mathcal{O}_{S} = (\mathscr{V}_{s^{\star}})_{S},$$

where $\check{\alpha}: A^* \to A$ is the (unique) O_F -linear quasi-p-isogeny such that $\check{\alpha} \circ \alpha = \varpi \cdot \mathrm{id}_A$, induces an isomorphism

$$\zeta_{s^{\star}} \colon \mathcal{B}_{s^{\star}} \xrightarrow{\sim} \mathcal{DL}_{s^{\star}} = \mathcal{DL}(\mathscr{V}_{s^{\star}}, \{,\}_{s^{\star}}, \lceil \frac{N+1}{2} \rceil).$$

In particular, B_{s^*} is a geometrically irreducible projective smooth scheme in Sch_{κ} of dimension $\lfloor \frac{N-1}{2} \rfloor$ by Lemma 4.2.2. In particular, ι is of pure codimension $\lfloor \frac{N}{2} \rfloor$.

Proof. For an object $(A_0, \lambda_0, \eta_0^p; A, \lambda, \eta^p; A^*, \lambda^*, \eta^{p*}; \alpha) \in B_p(V, K^p)(S)$, Definition 4.2.3(a) implies that there is a (unique) O_F -linear quasi-*p*-isogeny $\check{\alpha}: A^* \to A$ such that $\check{\alpha} \circ \alpha = \varpi \cdot id_A$ hence $\alpha \circ \check{\alpha} = \varpi \cdot id_{A^*}$. Moreover, we have the following properties from Definition 4.2.3:

- (a') ker $\check{\alpha}[p^{\infty}]$ is contained in $A^{\star}[\mathfrak{p}]$;
- (b') we have $\varpi \cdot \lambda^* = \breve{\alpha}^{\vee} \circ \lambda \circ \breve{\alpha}$; and

(c') the K^{*p*}-orbit of maps $v \mapsto \overline{\omega}^{-1} \check{\alpha}_* \circ \eta^{\star p}(v)$ for $v \in V \otimes_{\mathbb{Q}} \mathbb{A}^{\infty, p}$ coincides with η^p .

First, we show (1). It is clear that B_{s^*} is a scheme of finite type over κ . Consider a closed immersion $S \hookrightarrow \hat{S}$ in $\operatorname{Sch}'_{\kappa}$ defined by an ideal sheaf \mathcal{I} satisfying $\mathcal{I}^2 = 0$. Take a point $x = (A_0, \lambda_0, \eta_0^p; A, \lambda, \eta^p; A^*, \lambda^*, \eta^{p^*}; \alpha) \in B_{s^*}(S)$. To compute lifting of x to \hat{S} , we use the Serre–Tate and Grothendieck–Messing theories. Note that lifting α is equivalent to lifting both α and $\check{\alpha}$, satisfying (b,c) in Definition 4.2.3 and (b',c') above, respectively. Thus, by Proposition 3.4.8, to lift x to an \hat{S} -point is equivalent to lifting

O $\omega_{A^{\vee}/S,\tau_{\infty}}$ to a subbundle $\hat{\omega}_{A^{\vee},\tau_{\infty}}$ of $\mathrm{H}_{1}^{\mathrm{cris}}(A/\hat{S})_{\tau_{\infty}}$ (of rank 1),

O $\omega_{A^{\vee}/S,\tau_{\infty}^{c}}$ to a subbundle $\hat{\omega}_{A^{\vee},\tau_{\infty}^{c}}$ of $\mathrm{H}_{1}^{\mathrm{cris}}(A/\hat{S})_{\tau_{\infty}^{c}}$ (of rank N-1),

subject to the following requirements

- (a") $\hat{\omega}_{A^{\vee},\tau_{\infty}}$ and $\hat{\omega}_{A^{\vee},\tau_{\infty}}$ are orthogonal under $\langle , \rangle_{\lambda,\tau_{\infty}}^{\text{cris}}$ (3.3); and
- (b") $\check{\alpha}_{*,\tau_{\mathbf{c}}^{\mathbf{c}}}\mathbf{H}_{1}^{\mathrm{cris}}(A^{\star}/\hat{S})_{\tau_{\mathbf{c}}^{\mathbf{c}}}$ is contained in $\hat{\omega}_{A^{\vee},\tau_{\mathbf{c}}^{\mathbf{c}}}$.

Since $\langle , \rangle_{\lambda,\tau_{\infty}}^{\text{cris}}$ is a perfect pairing, $\hat{\omega}_{A^{\vee},\tau_{\infty}}$ uniquely determines $\hat{\omega}_{A^{\vee},\tau_{\infty}^{c}}$ by (a"). Moreover, by Property (b') above, we know that ker $\alpha_{*,\tau_{\infty}}$ and im $\check{\alpha}_{*,\tau_{\infty}^{c}}$ are orthogonal complement to each other under $\langle , \rangle_{\lambda,\tau_{\infty}}^{\text{cris}}$. Thus, (b") is equivalent to

(c") $\hat{\omega}_{A^{\vee},\tau_{\infty}}$ is contained in the kernel of $\alpha_{*,\tau_{\infty}} \colon \mathrm{H}_{1}^{\mathrm{cris}}(A/\hat{S})_{\tau_{\infty}} \to \mathrm{H}_{1}^{\mathrm{cris}}(A^{*}/\hat{S})_{\tau_{\infty}}$.

To summarize, lifting x to an \hat{S} -point is equivalent to lifting $\omega_{A^{\vee}/S,\tau_{\infty}}$ to a subbundle $\hat{\omega}_{A^{\vee},\tau_{\infty}}$ of ker $\alpha_{*,\tau_{\infty}}$. In other words, the subset of $B_{s^*}(\hat{S})$ above x is canonically a torsor over $\operatorname{Hom}_{\mathcal{O}_S}(\omega_{A^{\vee},\tau_{\infty}})$, (ker $\alpha_{*,\tau_{\infty}}/\omega_{A^{\vee},\tau_{\infty}}$) $\otimes_{\mathcal{O}_S} \mathcal{I}$). Thus, (1) follows.

Next, we show (2). By Theorem 4.1.3, we have a canonical isomorphism

$$\iota_{\kappa}^{*}\mathcal{T}_{\mathrm{M}_{\mathfrak{p}}(\mathrm{V},\mathrm{K}^{p})/\kappa}|_{\mathrm{B}_{s^{\star}}}\simeq\mathcal{H}om\left(\omega_{\mathcal{A}^{\vee},\tau_{\infty}},\mathrm{H}_{1}^{\mathrm{dR}}(\mathcal{A})_{\tau_{\infty}}/\omega_{\mathcal{A}^{\vee},\tau_{\infty}}\right),$$

and the induced map $\mathcal{T}_{B_{s^{\star}}/\kappa} \to \iota_{\kappa}^{*} \mathcal{T}_{M_{\mathfrak{p}}(V,K^{p})/\kappa}|_{B_{s^{\star}}}$ is identified with the canonical map

$$\mathcal{H}om\left(\omega_{\mathcal{A}^{\vee},\tau_{\infty}},\ker\alpha_{*,\tau_{\infty}}/\omega_{\mathcal{A}^{\vee},\tau_{\infty}}\right)\to\mathcal{H}om\left(\omega_{\mathcal{A}^{\vee},\tau_{\infty}},\mathrm{H}_{1}^{\mathrm{dR}}(\mathcal{A})_{\tau_{\infty}}/\omega_{\mathcal{A}^{\vee},\tau_{\infty}}\right).$$

It is clearly injective, with cokernel canonically isomorphic to

$$\mathcal{H}om\left(\omega_{\mathcal{A}^{\vee},\tau_{\infty}},\operatorname{im}\alpha_{*,\tau_{\infty}}\right)$$
.

Thus, (2) follows.

Finally, we show (3). We first show that ζ_{s^*} has the correct image, namely, H is a locally free \mathcal{O}_S -module of rank $\lceil \frac{N+1}{2} \rceil$, and satisfies $(FH^{(p)})^{\perp} \subseteq H$. Lemma 3.4.13(1,2,3) implies that H is locally free, and

$$\operatorname{rank}_{\mathcal{O}_S}(\ker \alpha_{*,\tau_{\infty}}) - \operatorname{rank}_{\mathcal{O}_S}(\ker \alpha_{*,\tau_{\infty}^{\mathsf{c}}}) = 1,$$

$$\operatorname{rank}_{\mathcal{O}_S}(\ker \alpha_{*,\tau_{\infty}}) + \operatorname{rank}_{\mathcal{O}_S}(\ker \alpha_{*,\tau_{\infty}^{\mathsf{c}}}) = 2\lceil \frac{N}{2} \rceil - 1.$$

Thus, we have $\operatorname{rank}_{\mathcal{O}_S}(\ker \alpha_{*,\tau_{\infty}}) = \lceil \frac{N}{2} \rceil$ and

$$\operatorname{rank}_{\mathcal{O}_S}(\ker \breve{\alpha}_{*,\tau_{\infty}}) = N - \operatorname{rank}_{\mathcal{O}_S}(\ker \alpha_{*,\tau_{\infty}}) = \lceil \frac{N-1}{2} \rceil.$$

On the other hand, as $\omega_{A^{\vee}/S,\tau_{\infty}}$ has rank 1 and $\omega_{A^{*\vee}/S,\tau_{\infty}}$ has rank 0, $\omega_{A^{\vee}/S,\tau_{\infty}}$ is contained in the kernel of $\alpha_{*,\tau_{\infty}}$ hence in the image of $\breve{\alpha}_{*,\tau_{\infty}}$. Together, we obtain rank_{\mathcal{O}_S} $H = \lceil \frac{N+1}{2} \rceil$. From the equalities

$$\begin{split} \breve{\alpha}_{*,\tau_{\infty}^{\mathsf{c}}}(\mathsf{F}H^{(p)}) &= \breve{\alpha}_{*,\tau_{\infty}^{\mathsf{c}}}\mathsf{F}_{A^{\star}}\left((\breve{\alpha}_{*,\tau_{\infty}})^{-1}\omega_{A^{\vee}/S,\tau_{\infty}}\right)^{(p)} = \breve{\alpha}_{*,\tau_{\infty}^{\mathsf{c}}}\mathsf{F}_{A^{\star}}(\breve{\alpha}_{*,\tau_{\infty}^{\mathsf{c}}}^{(p)})^{-1}\omega_{A^{(p)\vee}/S,\tau_{\infty}^{\mathsf{c}}} \\ &= \mathsf{F}_{A}\breve{\alpha}_{*,\tau_{\infty}^{\mathsf{c}}}^{(p)}(\breve{\alpha}_{*,\tau_{\infty}^{\mathsf{c}}}^{(p)})^{-1}\omega_{A^{(p)\vee}/S,\tau_{\infty}^{\mathsf{c}}} = \mathsf{F}_{A}\omega_{A^{(p)\vee}/S,\tau_{\infty}^{\mathsf{c}}} = 0 \end{split}$$

and the fact that $\mathbf{F}H^{(p)}$ and ker $\check{\alpha}_{*,\tau_{\infty}^{c}}$ are both subbundles of $\mathrm{H}_{1}^{\mathrm{dR}}(A^{*}/S)_{\tau_{\infty}^{c}}$ of rank $\lceil \frac{N+1}{2} \rceil$, we know $\mathbf{F}H^{(p)} = \ker \check{\alpha}_{*,\tau_{\infty}^{c}}$. By Definition 4.2.3(b) and the definition of $\check{\alpha}$, we have

$$\langle \ker \breve{\alpha}_{*,\tau_{\infty}^{\mathsf{c}}}, \operatorname{im} \alpha_{*,\tau_{\infty}} \rangle_{\lambda^{*},\tau_{\infty}^{\mathsf{c}}} = \langle \breve{\alpha}_{*,\tau_{\infty}^{\mathsf{c}}} \ker \breve{\alpha}_{*,\tau_{\infty}^{\mathsf{c}}}, \operatorname{H}_{1}^{\mathrm{dR}}(A/S)_{\tau_{\infty}} \rangle_{\lambda,\tau_{\infty}^{\mathsf{c}}} = 0.$$

which implies

$$\ker \breve{\alpha}_{*,\tau_{\infty}} = \operatorname{im} \alpha_{*,\tau_{\infty}} \subseteq (\ker \breve{\alpha}_{*,\tau_{\infty}^{\mathsf{c}}})^{\perp} = (\mathsf{F}H^{(p)})^{\perp}.$$

As both sides are subbundles of $\mathrm{H}_{1}^{\mathrm{dR}}(A^{*}/S)_{\tau_{\infty}}$ of rank $\lceil \frac{N-1}{2} \rceil$, we must have ker $\check{\alpha}_{*,\tau_{\infty}} = (\mathrm{F}H^{(p)})^{\perp}$. In particular, we have $(\mathrm{F}H^{(p)})^{\perp} \subseteq H$. Thus, $\zeta_{s^{*}}$ is defined as we claim.

Since the target of ζ_{s^*} is smooth over κ by Lemma 4.2.2, to see that ζ_{s^*} is an isomorphism, it suffices to check that for every algebraically closed field κ' containing κ

(3.1) ζ_{s^*} induces a bijection on κ' -points; and

(3.2) ζ_{s^*} induces an isomorphism on the tangent spaces at every κ' -point.

To ease notation, we may assume that $\kappa' = \kappa$ hence is perfect in particular.

For (3.1), we construct an inverse to the map $\zeta_{s^*}(\kappa)$. Take a point $y \in \mathrm{DL}_{s^*}(\kappa)$ represented by a κ -linear subspace $H \subseteq \mathscr{V}_{s^*} = \mathrm{H}_1^{\mathrm{dR}}(A^*/\kappa)_{\tau_{\infty}}$. We regard F and V as those sesquilinear maps in Remark 3.4.10. In particular, we have $(\mathrm{F}H)^{\perp} \subseteq H$. For every $\tau \in \Sigma_{\infty}$, we define a $W(\kappa)$ submodule $\mathcal{D}_{A,\tau} \subseteq \mathcal{D}(A^*)_{\tau}$ as follows.

- O If $\tau \notin \{\tau_{\infty}, \tau_{\infty}^{\mathsf{c}}\}$, then $\mathcal{D}_{A,\tau} = \mathcal{D}(A^{\star})_{\tau}$.
- O We set $\mathcal{D}_{A,\tau_{\infty}} := \mathbb{V}^{-1}\tilde{H}^{\mathfrak{c}}$, where $\tilde{H}^{\mathfrak{c}}$ is the preimage of H^{\perp} under the reduction map $\mathcal{D}(A^{\star})_{\tau_{\infty}^{\mathfrak{c}}} \to \mathcal{D}(A^{\star})_{\tau_{\infty}^{\mathfrak{c}}}/p\mathcal{D}(A^{\star})_{\tau_{\infty}^{\mathfrak{c}}} = \mathrm{H}_{1}^{\mathrm{dR}}(A^{\star})_{\tau_{\infty}^{\mathfrak{c}}}.$
- O We set $\mathcal{D}_{A,\tau_{\infty}^{c}} \coloneqq F\tilde{H}$, where \tilde{H} is the preimage of H under the reduction map $\mathcal{D}(A^{\star})_{\tau_{\infty}} \to \mathcal{D}(A^{\star})_{\tau_{\infty}}/p\mathcal{D}(A^{\star})_{\tau_{\infty}} = \mathrm{H}_{1}^{\mathrm{dR}}(A^{\star})_{\tau_{\infty}}.$

Finally, put $\mathcal{D}_A := \bigoplus_{\tau \in \Sigma_{\infty}} \mathcal{D}_{A,\tau}$ as a $W(\kappa)$ -submodule of $\mathcal{D}(A^*)$. We show that it is stable under F and V. It suffices to show that both F and V stabilize $\mathcal{D}_{A,\tau_{\infty}} \oplus \mathcal{D}_{A,\tau_{\infty}^{c}}$, which breaks into checking that

- O $\mathsf{F}\mathcal{D}_{A,\tau_{\infty}} \subseteq \mathcal{D}_{A,\tau_{\infty}^{c}}$, that is, $\mathsf{F}\mathsf{V}^{-1}\tilde{H}^{\mathsf{c}} \subseteq \mathsf{F}\tilde{H}$. It suffices to show that $\mathsf{V}^{-1}(H^{\perp})$ (as a subspace of $\mathrm{H}_{1}^{\mathrm{dR}}(A^{\star})_{\tau_{\infty}}$) is contained in H. However, $\mathsf{V}^{-1}(H^{\perp}) = (\mathsf{F}H)^{\perp}$, which is contained in H.
- O $\mathcal{FD}_{A,\tau_{\infty}^{c}} \subseteq \mathcal{D}_{A,\tau_{\infty}}$, that is, $\mathcal{FF}\tilde{H} \subseteq \mathbb{V}^{-1}\tilde{H}^{c}$. It suffices to show $p\mathcal{F}\tilde{H} \subseteq \tilde{H}^{c}$, which obviously holds.
- O $V\mathcal{D}_{A,\tau_{\infty}} \subseteq \mathcal{D}_{A,\tau_{\infty}^{c}}$, that is, $VV^{-1}\tilde{H}^{c} \subseteq F\tilde{H}$. it suffices to show $H^{\perp} \subseteq FH$ as subspaces of $H_{1}^{dR}(A^{\star})_{\tau_{\infty}^{c}}$, which follows from $(FH)^{\perp} \subseteq H$.
- O $V\mathcal{D}_{A,\tau_{\infty}^{c}} \subseteq \mathcal{D}_{A,\tau_{\infty}}$, that is, $VF\tilde{H} \subseteq V^{-1}\tilde{H}^{c}$. It is obvious as $V^{-1}\tilde{H}^{c}$ contains $p\mathcal{D}(A^{\star})_{\tau_{\infty}}$.

Thus, $(\mathcal{D}_A, \mathbf{F}, \mathbf{V})$ is a Dieudonné module over $W(\kappa)$. By the Dieudonné theory, there is an O_F abelian scheme A over κ with $\mathcal{D}(A)_{\tau} = \mathcal{D}_{A,\tau}$ for every $\tau \in \Sigma_{\infty}$, and an O_F -linear p-isogeny $\alpha \colon A \to A^*$ inducing the inclusion of Dieudonné modules $\mathcal{D}(A) = \mathcal{D}_A \subseteq \mathcal{D}(A^*)$. Moreover, since $\mathfrak{p}\mathcal{D}(A^*) \subseteq \mathcal{D}(A)$, we have ker $\alpha[p^{\infty}] \subseteq A[\mathfrak{p}]$.

Let $\lambda: A \to A^{\vee}$ be the unique quasi-polarization such that $\varpi \lambda = \alpha^{\vee} \circ \lambda^* \circ \alpha$. We claim that λ is *p*-principal. It is enough to show the induced pairing

$$p^{-1} \cdot \langle , \rangle_{\lambda, \tau_{\infty}} \colon \mathcal{D}(A)_{\tau_{\infty}} \times \mathcal{D}(A)_{\tau_{\infty}^{c}} \to W(\kappa)$$

(Notation 3.4.12) is non-degenerate. Since \tilde{H} is $W(\kappa)$ -dual to $p^{-1}\tilde{H}^{\mathfrak{c}}$, hence $\mathcal{D}(A)_{\tau_{\infty}^{\mathfrak{c}}} = \mathsf{F}\tilde{H}$ is dual to $\mathsf{V}^{-1}(p^{-1}\tilde{H}^{\mathfrak{c}}) = p^{-1}\mathsf{V}^{-1}\tilde{H}^{\mathfrak{c}} = p^{-1}\mathcal{D}(A)_{\tau_{\infty}}$, the above pairing is non-degenerate.

It is an easy consequence of Lemma 3.4.13(2,3) that the O_F -abelian scheme A has signature type $N\Phi - \tau_{\infty} + \tau_{\infty}^{c}$. Finally, let η^{p} be the unique K^{p} -level structure such that Definition 4.2.3(c) is satisfied. Putting together, we obtain a point $x = (A_0, \lambda_0, \eta_0^p; A, \lambda, \eta^p; A^{\star}, \lambda^{\star}, \eta^{p\star}; \alpha) \in B_{s^{\star}}(\kappa)$ such that $\zeta_{s^{\star}}(x) = y$. It is easy to see that such assignment gives rise to an inverse of $\zeta_{s^{\star}}(\kappa)$; hence (3.1) follows immediately.

For (3.2), let \mathcal{T}_x and \mathcal{T}_y be the tangent spaces at x and y as in (3.1), respectively. By (1) and Lemma 4.2.2, we have canonical isomorphisms

$$\mathcal{T}_x \simeq \operatorname{Hom}_{\kappa}(\omega_{A^{\vee},\tau_{\infty}}, \ker \alpha_{*,\tau_{\infty}}/\omega_{A^{\vee},\tau_{\infty}}), \quad \mathcal{T}_y \simeq \operatorname{Hom}_{\kappa}(H/(\mathbb{F}H)^{\perp}, \operatorname{H}_1^{\mathrm{dR}}(A^{\star})_{\tau_{\infty}}/H)$$

Moreover, by the definition of ζ_{s^*} , the map $(\zeta_{s^*})_* \colon \mathcal{T}_x \to \mathcal{T}_y$ is induced by the following two maps

$$H/(\mathbf{F}H)^{\perp} = (\breve{\alpha}_{*,\tau_{\infty}})^{-1} \omega_{A^{\vee},\tau_{\infty}} / \ker \breve{\alpha}_{*,\tau_{\infty}} \xrightarrow{\check{\alpha}_{*,\tau_{\infty}}} \omega_{A^{\vee},\tau_{\infty}},$$
$$\mathrm{H}_{1}^{\mathrm{dR}}(A^{\star})_{\tau_{\infty}}/H = \mathrm{H}_{1}^{\mathrm{dR}}(A^{\star})_{\tau_{\infty}}/(\breve{\alpha}_{*,\tau_{\infty}})^{-1} \omega_{A^{\vee},\tau_{\infty}} \xrightarrow{\check{\alpha}_{*,\tau_{\infty}}} \ker \alpha_{*,\tau_{\infty}}/\omega_{A^{\vee},\tau_{\infty}},$$

both being isomorphisms. Thus, (3.2) hence (3) follow.

Remark 4.2.6. In Theorem 4.2.5, when K^p is sufficiently small, the restriction of ι to B_{s^*} is a closed immersion for every point $s^* \in S_p(V, K^p)(\kappa)$ and every field κ containing \mathbb{F}_p^{Φ} .

4.3. Source of basic correspondence and Tate cycles. In this subsection, we study the source $S_{\mathfrak{p}}(V, -)$ of the basic correspondence. We will describe the set $S_{\mathfrak{p}}(V, -)(\overline{\mathbb{F}}_p)$ in terms of certain Shimura set and study its Galois action. Such a description is not canonical, which depends on the choice of a definite uniformization datum defined as follows.

Definition 4.3.1. We define a *definite uniformization datum for* V (at \mathfrak{p}) to be a collection of $(V^*, \mathfrak{i}, \{\Lambda^*_{\mathfrak{q}}\}_{\mathfrak{q}|p})$ where

- O V^{\star} is a standard definite hermitian space over F of rank N;
- $O \ {\tt i} \colon \mathcal{V} \otimes_{\mathbb{Q}} \mathbb{A}^{\infty,p} \to \mathcal{V}^{\star} \otimes_{\mathbb{Q}} \mathbb{A}^{\infty,p} \ {\rm is \ an \ isometry};$
- O for every prime \mathfrak{q} of F^+ above p other than \mathfrak{p} , $\Lambda^{\star}_{\mathfrak{q}}$ is a self-dual $O_{F_{\mathfrak{q}}}$ -lattice in $V^{\star} \otimes_F F_{\mathfrak{q}}$; and
- O $\Lambda_{\mathfrak{p}}^{\star}$ is an $O_{F_{\mathfrak{p}}}$ -lattice in $V^{\star} \otimes_{F} F_{\mathfrak{p}}$ satisfying $p\Lambda_{\mathfrak{p}}^{\star} \subseteq (\Lambda_{\mathfrak{p}}^{\star})^{\vee}$ such that $(\Lambda_{\mathfrak{p}}^{\star})^{\vee}/p\Lambda_{\mathfrak{p}}^{\star}$ has length 0 (resp. 1) if N is odd (resp. even).

By the Hasse principle for hermitian spaces, there exists a definite uniformization datum for which we fix one. Let $K^*_{\mathfrak{q}}$ be the stabilizer of $\Lambda^*_{\mathfrak{q}}$ for every \mathfrak{q} over p; and put $K^*_p \coloneqq \prod_{\mathfrak{q}|p} K^*_{\mathfrak{q}}$. The isometry \mathfrak{i} induces an equivalence of categories $\mathfrak{i} \colon \mathfrak{K}(V)^p \xrightarrow{\sim} \mathfrak{K}(V^*)^p$.

Construction 4.3.2. We now construct a *uniformization map*, denoted by the Greek letter *upsilon*

(4.4)
$$\upsilon \colon \mathrm{S}_{\mathfrak{p}}(\mathrm{V}, -)(\overline{\mathbb{F}}_p) \to \mathrm{Sh}(\mathrm{V}^{\star}, (\mathtt{i}-)\mathrm{K}_p^{\star}) \times \mathrm{T}_{\mathfrak{p}}(\overline{\mathbb{F}}_p)$$

in $\mathsf{Fun}(\mathfrak{K}(V)^p \times \mathfrak{T}, \mathsf{Set})_{/\mathrm{T}_n(\overline{\mathbb{F}}_n)}$, which turns out to be an isomorphism.

Take a point $s^{\star} = (A_0, \lambda_0, \eta_0^p; A^{\star}, \lambda^{\star}, \eta^{p\star}) \in S_{\mathfrak{p}}(V, K^p)(\overline{\mathbb{F}}_p)$. Let

$$\mathcal{V}_{s^{\star}} \coloneqq \operatorname{Hom}_{O_{F}}(A_{0}, A^{\star}) \otimes \mathbb{Q}$$

be the space of O_F -linear quasi-homomorphisms. We equip with V_{s^*} a pairing

$$(x,y) = \overline{\omega}^{-1} \cdot \lambda_0^{-1} \circ y^{\vee} \circ \lambda^* \circ x \in \operatorname{End}_{\overline{\mathbb{F}}_p}(A_0) \otimes \mathbb{Q} = F,$$

which becomes a hermitian space over F. Note that we have an extra factor ϖ^{-1} in the above pairing. Moreover, for every prime \mathfrak{q} of F^+ above p, put

$$\Lambda_{s^{\star},\mathfrak{q}} \coloneqq \operatorname{Hom}_{O_{F}}(A_{0}[\mathfrak{q}^{\infty}], A^{\star}[\mathfrak{q}^{\infty}])$$

which is an $O_{F_{\mathfrak{q}}}$ -lattice in $(V_{s^{\star}})_{\mathfrak{q}}$ since A^{\star} is isogenous to A_0^N .

Now we construct v, whose process is very similar to Remark 4.1.5. Note that we have an isometry

 $\rho \colon \mathcal{V}_{s^\star} \otimes_{\mathbb{Q}} \mathbb{A}^{\infty,p} \xrightarrow{\sim} \mathrm{Hom}_{F \otimes_{\mathbb{Q}} \mathbb{A}^{\infty,p}}^{\varpi \lambda_0,\lambda^\star}(\mathcal{H}_1^{\mathrm{\acute{e}t}}(A_0,\mathbb{A}^{\infty,p}),\mathcal{H}_1^{\mathrm{\acute{e}t}}(A^\star,\mathbb{A}^{\infty,p})).$

By Lemma 4.3.3 below, we can choose an isometry $\eta_{rat}: V_{s^*} \to V^*$. Thus, we obtain an isometry

$$g^p \coloneqq \eta_{\mathrm{rat}} \circ \rho^{-1} \circ \eta^{p\star} \circ \mathbf{i}^{-1} \colon \mathrm{V}^{\star} \otimes_{\mathbb{Q}} \mathbb{A}^{\infty, p} \to \mathrm{V}^{\star} \otimes_{\mathbb{Q}} \mathbb{A}^{\infty, p}$$

as an element in $U(V^*)(\mathbb{A}_{F^+}^{\infty,p})$. By Lemma 4.3.3(1,2), for every \mathfrak{q} above p, there exists an element $g_{\mathfrak{q}} \in U(V^*)(F_{\mathfrak{q}}^+)$ such that $g_{\mathfrak{q}}\Lambda_{\mathfrak{q}}^* = \eta_{\mathrm{rat}}\Lambda_{s^*,\mathfrak{q}}$. Together, we obtain an element $g_{s^*} \coloneqq (g^p, (g_{\mathfrak{q}})_{\mathfrak{q}|p}) \in U(V^*)(\mathbb{A}_{F^+}^\infty)$ such that the double coset $U(V^*)(F)g(\mathfrak{i}K^p)K_p^*$ depends only on the point s^* . Thus, it allows us to define

$$\upsilon(s^{\star}) \coloneqq \left(\mathrm{U}(\mathrm{V}^{\star})(F)g_{s^{\star}}(\mathtt{i}\mathrm{K}^{p})\mathrm{K}_{p}^{\star}, (A_{0}, \lambda_{0}, \eta_{0}^{p}) \right) \in \mathrm{Sh}(\mathrm{V}^{\star}, (\mathtt{i}\mathrm{K}^{p})\mathrm{K}_{p}^{\star}) \times \mathrm{T}_{\mathfrak{p}}(\overline{\mathbb{F}}_{p}).$$

Lemma 4.3.3. The hermitian spaces V_{s^*} and V^* are isomorphic. Moreover,

- (1) for every prime \mathfrak{q} of F^+ above p other than \mathfrak{p} , the lattice $\Lambda_{s^*,\mathfrak{q}}$ is self-dual;
- (2) the lattice $\Lambda_{s^*,\mathfrak{p}}$ satisfies $p\Lambda_{s^*,\mathfrak{p}} \subseteq (\Lambda_{s^*,\mathfrak{p}})^{\vee}$ such that $(\Lambda_{s^*,\mathfrak{p}})^{\vee}/p\Lambda_{s^*,\mathfrak{p}}$ has length 0 (resp. 1) if N is odd (resp. even).

Proof. We first prove (1) and (2).

For (1), note that $A^*[\mathfrak{q}^{\infty}]$ is isomorphic to $(A_0[\mathfrak{q}^{\infty}])^N$, equipped with the polarization $\lambda^*[\mathfrak{q}^{\infty}]$ that is principal. Thus, $\Lambda_{s^*,\mathfrak{q}}$ is self-dual as $\lambda_0[\mathfrak{q}^{\infty}]$ is principal and $\operatorname{val}_{\mathfrak{q}}(\varpi) = 0$.

For (2), note that $A^*[\mathfrak{p}^{\infty}]$ is isomorphic to $(A_0[\mathfrak{p}^{\infty}])^N$, equipped with the polarization $\lambda^*[\mathfrak{p}^{\infty}]$ satisfying such that ker $\lambda^*[\mathfrak{p}^{\infty}]$ is trivial (resp. contained in $A^*[\mathfrak{p}]$ of rank p^2) if N is odd (resp. even). Thus, the statement follows as $\lambda_0[\mathfrak{p}^{\infty}]$ is principal and $\operatorname{val}_{\mathfrak{p}}(\varpi) = 1$.

Now to prove the main statement, it suffices to show that

- (i) V_{s^*} is totally positive definite; and
- (ii) the hermitian spaces $V_{s^*} \otimes_{\mathbb{Q}} \mathbb{A}^{\infty,p}$ and $V \otimes_{\mathbb{Q}} \mathbb{A}^{\infty,p}$ are isomorphic.

For (i), it follows from the same argument in [KR14, Lemma 2.7].

For (ii), we have a map

$$V_{s^{\star}} \otimes_{\mathbb{Q}} \mathbb{A}^{\infty,p} \to \operatorname{Hom}_{F \otimes_{\mathbb{Q}} \mathbb{A}^{\infty,p}}^{\varpi \lambda_{0},\lambda^{\star}}(\operatorname{H}_{1}^{\operatorname{\acute{e}t}}(A_{0},\mathbb{A}^{\infty,p}),\operatorname{H}_{1}^{\operatorname{\acute{e}t}}(A^{\star},\mathbb{A}^{\infty,p}))$$

of hermitian spaces, which is injective. As both sides have rank N and the right-hand side is isomorphic to $V \otimes_{\mathbb{Q}} \mathbb{A}^{\infty,p}$, (ii) follows.

Proposition 4.3.4. The uniformization map v (4.4) is an isomorphism. Moreover, the induced action of $\operatorname{Gal}(\overline{\mathbb{F}}_p/\mathbb{F}_p^{\Phi})$ on the target of v factors through the projection map

$$\operatorname{Sh}(\mathrm{V}^{\star},(\mathtt{i}-)\mathrm{K}_p^{\star})\times \mathrm{T}_{\mathfrak{p}}(\overline{\mathbb{F}}_p) \to \mathrm{T}_{\mathfrak{p}}(\overline{\mathbb{F}}_p).$$

Proof. We first show that v is an isomorphism. Take a point $t = (A_0, \lambda_0, \eta_0^p) \in T_{\mathfrak{p}}(\overline{\mathbb{F}}_p)$. It suffices to show that, for every $K^p \in \mathfrak{K}(V)^p$, the restriction

$$\upsilon \colon \mathrm{S}_{\mathfrak{p}}(\mathrm{V},\mathrm{K}^p)(\overline{\mathbb{F}}_p)_{/t} \to \mathrm{Sh}(\mathrm{V}^\star,(\mathrm{i}\mathrm{K}^p)\mathrm{K}_p^\star)$$

to the fiber over t is an isomorphism. The injectivity follows directly from the definition. For the surjectivity, it suffices to show that for every $g \in U(V^*)(\mathbb{A}_{F^+}^{\infty,p})$, there is an object $s^* = (A_0, \lambda_0, \eta_0^p; A^*, \lambda^*, \eta^{p*}) \in S_p(V, K^p)(\overline{\mathbb{F}}_p)_{/t}$ whose image under v is the image of g in $Sh(V^*, (iK^p)K_p^*)$. To construct s^* , we take an O_F -lattice Λ^* in V^* satisfying $O_F \otimes_F F_p = \Lambda_p^*$. Put $A^* := A_0 \otimes_{O_F} \Lambda^*$, which is equipped with a unique quasi-polarization λ^* such that the canonical isomorphism

$$\mathbf{V}^{\star} \otimes_{\mathbb{Q}} \mathbb{A}^{\infty, p} \simeq \operatorname{Hom}_{F \otimes_{\mathbb{Q}} \mathbb{A}^{\infty, p}}(\mathrm{H}_{1}^{\operatorname{\acute{e}t}}(A_{0}, \mathbb{A}^{\infty, p}), \mathrm{H}_{1}^{\operatorname{\acute{e}t}}(A^{\star}, \mathbb{A}^{\infty, p}))$$

of $F \otimes_{\mathbb{Q}} \mathbb{A}^{\infty,p}$ -modules is an isometry of hermitian spaces. We let $\eta^{p\star}$ be the map

$$V \otimes_{\mathbb{Q}} \mathbb{A}^{\infty, p} \xrightarrow{g \circ i} V^{\star} \otimes_{\mathbb{Q}} \mathbb{A}^{\infty, p} = \operatorname{Hom}_{F \otimes_{\mathbb{Q}} \mathbb{A}^{\infty, p}}^{\varpi \lambda_{0}, \lambda^{\star}} (\operatorname{H}_{1}^{\operatorname{\acute{e}t}}(A_{0}, \mathbb{A}^{\infty, p}), \operatorname{H}_{1}^{\operatorname{\acute{e}t}}(A^{\star}, \mathbb{A}^{\infty, p})).$$

Then $v(s^{\star}) = g$ in Sh(V^{\star}, (iK^p)K^{\star}_p). Thus, v is an isomorphism.

Since v is an isomorphism, the Galois group $\operatorname{Gal}(\overline{\mathbb{F}}_p/\mathbb{F}_p^{\Phi})$ acts on the target of v. We show that it acts trivially on the first factor of the target of v. Take an element $\varsigma \in \operatorname{Gal}(\overline{\mathbb{F}}_p/\mathbb{F}_p^{\Phi})$ and a point $s^* = (A_0, \lambda_0, \eta_0^p; A^*, \lambda^*, \eta^{p*}) \in \operatorname{S}_{\mathfrak{p}}(V, \operatorname{K}^p)(\overline{\mathbb{F}}_p)$. Then ςs^* is simply represented by $(A_0^{\varsigma}, \lambda_0^{\varsigma}, \eta_0^{p\varsigma}; A^{*\varsigma}, \lambda^{*\varsigma}, \eta^{p*\varsigma})$, the ς -twist of the previous object. We then have a canonical isomorphism

$$\mathbf{V}_{\varsigma s^{\star}} = \mathrm{Hom}_{O_F}(A_0^{\varsigma}, A^{\star\varsigma}) \otimes \mathbb{Q} \simeq \mathrm{Hom}_{O_F}(A_0, A^{\star}) \otimes \mathbb{Q} = \mathbf{V}_{s^{\star}}$$

of hermitian spaces. Unraveling the definition, we see that $g_{s^*} = g_{\varsigma s^*}$. Thus, we have

$$\upsilon(\varsigma s^{\star}) \coloneqq \left(\mathrm{U}(\mathrm{V}^{\star})(F) g_{s^{\star}}(\mathrm{i}\mathrm{K}^{p}) \mathrm{K}_{p}^{\star}, (A_{0}^{\varsigma}, \lambda_{0}^{\varsigma}, \eta_{0}^{p\varsigma}) \right).$$

The proposition follows.

Next, we define an action of the Hecke algebra $\mathbb{Z}[\mathrm{K}_{\mathfrak{p}}^{\star} \setminus \mathrm{U}(\mathrm{V}^{\star})(F_{\mathfrak{p}}^{+})/\mathrm{K}_{\mathfrak{p}}^{\star}]$ on $\mathrm{S}_{\mathfrak{p}}(\mathrm{V}, -)$ via finite étale correspondences, that is compatible with the uniformization map (4.4).

Construction 4.3.5. For every element $g \in K^{\star}_{\mathfrak{p}} \setminus U(V^{\star})(F^{+}_{\mathfrak{p}})/K^{\star}_{\mathfrak{p}}$, we define a functor

$$\begin{split} \mathbf{S}_{\mathfrak{p}}(\mathbf{V},-)_g \colon \mathfrak{K}(\mathbf{V})^p \times \mathfrak{T} &\to \mathsf{PSch}'_{/\mathbb{F}_p^{\Phi}} \\ \mathbf{K}^p &\mapsto \mathbf{S}_{\mathfrak{p}}(\mathbf{V},\mathbf{K}^p)_g \end{split}$$

such that for every $S \in \mathsf{Sch}'_{/\mathbb{F}_p^{\Phi}}$, $S_{\mathfrak{p}}(V, K^p)_g(S)$ is the set of equivalence classes of decuples $(A_0, \lambda_0, \eta_0^p; A^\star, \lambda^\star, \eta^{p\star}; A_g^\star, \lambda_g^\star, \eta_g^{p\star}; \phi^\star)$ where

O $(A_0, \lambda_0, \eta_0^p; A^\star, \lambda^\star, \eta^{p\star})$ and $(A_0, \lambda_0, \eta_0^p; A_g^\star, \lambda_g^\star, \eta_g^{p\star})$ are both elements in $S_{\mathfrak{p}}(V, K^p)(S)$; and O $\phi^\star : A^\star \to A_g^\star$ is an O_F -linear quasi-isogeny such that (a) $\phi^{\star \vee} \circ \lambda_g^\star \circ \phi^\star = \lambda^\star$;

- (b) $\phi^{\star}[\mathfrak{p}^{\infty}]: A^{\star}[\mathfrak{p}^{\infty}] \to A_{g}^{\star}[\mathfrak{p}^{\infty}]$ is a quasi-isogeny of height zero under which the two lattices $\operatorname{Hom}_{O_{F}}(A_{0s}[\mathfrak{p}^{\infty}], A_{s}^{\star}[\mathfrak{p}^{\infty}])$ and $\operatorname{Hom}_{O_{F}}(A_{0s}[\mathfrak{p}^{\infty}], A_{gs}^{\star}[\mathfrak{p}^{\infty}])$ are at the relative position determined by g for every geometric point s of S;
- (c) $\phi^{\star}[\mathfrak{q}^{\infty}]$ is an isomorphism for every prime \mathfrak{q} of F^{+} above p that is not \mathfrak{p} ; and
- (d) the K^{*p*}-orbit of maps $v \mapsto \phi_*^* \circ \eta^{p*}(v)$ for $v \in V \otimes_{\mathbb{Q}} \mathbb{A}^{\infty,p}$ coincides with η_a^{p*} .

The equivalence relation and the action of morphisms in $\mathfrak{K}(\mathbf{V})^p \times \mathfrak{T}$ are defined similarly as in Definition 4.2.3. Then we construct the *Hecke correspondence* (of g) to be the morphism

(4.5)
$$\operatorname{Hk}_{g} \colon \operatorname{S}_{\mathfrak{p}}(\mathbf{V}, -)_{g} \to \operatorname{S}_{\mathfrak{p}}(\mathbf{V}, -) \times \operatorname{S}_{\mathfrak{p}}(\mathbf{V}, -)$$

in $\mathsf{Fun}(\mathfrak{K}(V)^p \times \mathfrak{T}, \mathsf{PSch}'_{/\mathbb{F}_p^\Phi})_{/\mathrm{T}_p}$ induced by the assignment

 $(A_0, \lambda_0, \eta_0^p; A^\star, \lambda^\star, \eta^{p\star}; A_g^\star, \lambda_g^\star, \eta_g^{p\star}; \phi^\star) \mapsto ((A_0, \lambda_0, \eta_0^p; A^\star, \lambda^\star, \eta^{p\star}), (A_0, \lambda_0, \eta_0^p; A_g^\star, \lambda_g^\star, \eta_g^{p\star})).$

Here, the product in (4.5) is also taken in the category $\operatorname{Fun}(\mathfrak{K}(V)^p \times \mathfrak{T}, \operatorname{\mathsf{PSch}}'_{/\mathbb{F}^{\Phi}_p})_{/\mathbb{T}_p}$, that is, $S_{\mathfrak{p}}(V, -) \times S_{\mathfrak{p}}(V, -)$ is a functor sending K^p to $S_{\mathfrak{p}}(V, K^p) \times_{\mathbb{T}_p} S_{\mathfrak{p}}(V, K^p)$ on which \mathfrak{T} acts diagonally.

Proposition 4.3.6. For every $g \in K^*_{\mathfrak{p}} \setminus U(V^*)(F^+_{\mathfrak{p}})/K^*_{\mathfrak{p}}$, we have

- (1) The morphism Hk_g (4.5) is finite étale; in particular, it is a morphism in $\operatorname{Fun}(\mathfrak{K}(V)^p \times \mathfrak{T}, \operatorname{Sch}_{/\mathbb{F}_p^\Phi})_{/\mathrm{T}_p}$.
- (2) The uniformization map v (4.4) lifts uniquely to an isomorphism making the diagram

$$\begin{array}{ccc} \mathrm{S}_{\mathfrak{p}}(\mathrm{V},-)_{g}(\overline{\mathbb{F}}_{p}) & \xrightarrow{\upsilon} & \mathrm{Sh}(\mathrm{V}^{\star},(\mathtt{i}-)(g\mathrm{K}_{p}^{\star}g^{-1}\cap\mathrm{K}_{p}^{\star})) \times \mathrm{T}_{\mathfrak{p}}(\overline{\mathbb{F}}_{p}) \\ & & \downarrow \\ \mathrm{S}_{\mathfrak{p}}(\mathrm{V},-)(\overline{\mathbb{F}}_{p}) \times_{\mathrm{T}_{\mathfrak{p}}(\overline{\mathbb{F}}_{p})} \mathrm{S}_{\mathfrak{p}}(\mathrm{V},-)(\overline{\mathbb{F}}_{p}) & \xrightarrow{\upsilon \times \upsilon} \left(\mathrm{Sh}(\mathrm{V}^{\star},(\mathtt{i}-)\mathrm{K}_{p}^{\star}) \times \mathrm{Sh}(\mathrm{V}^{\star},(\mathtt{i}-)\mathrm{K}_{p}^{\star}) \right) \times \mathrm{T}_{\mathfrak{p}}(\overline{\mathbb{F}}_{p}) \end{array}$$

in $\operatorname{Fun}(\mathfrak{K}(V)^p \times \mathfrak{T}, \operatorname{Set})_{/\operatorname{T}_{\mathfrak{p}}(\overline{\mathbb{F}}_p)}$ commutative, where the right vertical map is induced by the set-theoretical Hecke correspondence of g.

Proof. For (1), it suffices to consider those $K^p \in \mathfrak{K}(V)^p$ that are sufficiently small. Then the morphism $Hk_g \colon S_{\mathfrak{p}}(V, K^p)_g \to S_{\mathfrak{p}}(V, K^p) \times_{T_{\mathfrak{p}}} S_{\mathfrak{p}}(V, K^p)$ is closed, hence represented by a finite étale scheme. Part (2) follows directly from the definition.

Remark 4.3.7. In fact, the proof of Proposition 4.3.6(1) together with Proposition 4.3.4 imply that Hk_g is a local isomorphism.

Remark 4.3.8. Note that since $K_{\mathfrak{p}}^{\star}$ is a special maximal open compact subgroup of $U(V^{\star})(F_{\mathfrak{p}}^{+})$, the algebra $\mathbb{Z}[K_{\mathfrak{p}}^{\star} \setminus U(V^{\star})(F_{\mathfrak{p}}^{+})/K_{\mathfrak{p}}^{\star}]$ is commutative. Moreover, when N is odd, $\Lambda_{s^{\star},\mathfrak{p}}$ is a self-dual lattice under the pairing $\varpi \cdot (,)_{V^{\star}}$; hence $\mathbb{Z}[K_{\mathfrak{p}}^{\star} \setminus U(V^{\star})(F_{\mathfrak{p}}^{+})/K_{\mathfrak{p}}^{\star}]$ is canonically isomorphic to $\mathbb{T}_{N,\mathfrak{p}}$.

Let L be a p-coprime coefficient ring. The uniformization map (4.4) induces an isomorphism

$$L[\mathrm{Sh}(\mathrm{V}^{\star},(\mathtt{i}-)\mathrm{K}_{p}^{\star})] \simeq \mathrm{H}^{0}_{\mathfrak{T}}(\overline{\mathrm{S}}_{\mathfrak{p}}(\mathrm{V},-),L) = \mathrm{H}^{0}_{\mathfrak{T}}(\mathrm{S}_{\mathfrak{p}}(\mathrm{V},-),L)$$

in $\operatorname{Fun}(\mathfrak{K}(V)^p, \operatorname{\mathsf{Mod}}(L[\mathrm{K}^*_{\mathfrak{p}} \setminus \mathrm{U}(V^* \otimes_F F_{\mathfrak{p}})/\mathrm{K}^*_{\mathfrak{p}}]))$ by Proposition 4.3.6. Recall from Theorem 4.2.5(3) that the morphism ι in (4.3) is of pure codimension $\lfloor \frac{N}{2} \rfloor$.

Construction 4.3.9. Put $r \coloneqq \lfloor \frac{N}{2} \rfloor \ge 0$. We construct a pair of maps

$$\begin{cases} \operatorname{inc}_{!}^{\star} \colon L[\operatorname{Sh}(\mathbf{V}^{\star},(\mathbf{i}-)\mathbf{K}_{p}^{\star})] \xrightarrow{\sim} \mathrm{H}_{\mathfrak{T}}^{0}(\mathbf{S}_{\mathfrak{p}}(\mathbf{V},-),L) \\ \xrightarrow{\pi^{\star}} \mathrm{H}_{\mathfrak{T}}^{0}(\mathbf{B}_{\mathfrak{p}}(\mathbf{V},-),L) \xrightarrow{\iota_{!}} \mathrm{H}_{\mathfrak{T}}^{2r}(\mathbf{M}_{\mathfrak{p}}(\mathbf{V},-),L(r)), \\ \operatorname{inc}_{\star}^{\star} \colon \mathrm{H}_{\mathfrak{T}}^{2(N-r-1)}(\mathbf{M}_{\mathfrak{p}}(\mathbf{V},-),L(N-r-1)) \xrightarrow{\iota^{\star}} \mathrm{H}_{\mathfrak{T}}^{2(N-r-1)}(\mathbf{B}_{\mathfrak{p}}(\mathbf{V},-),L(N-r-1)) \\ \xrightarrow{\pi_{!}} \mathrm{H}_{\mathfrak{T}}^{0}(\mathbf{S}_{\mathfrak{p}}(\mathbf{V},-),L) \xrightarrow{\sim} L[\operatorname{Sh}(\mathbf{V}^{\star},(\mathbf{i}-)\mathbf{K}_{p}^{\star})], \end{cases}$$

in $\operatorname{Fun}(\mathfrak{K}(\mathbf{V})^p, \operatorname{\mathsf{Mod}}(L))$.

Theorem 4.3.10. Suppose N = 2r + 1 odd with $r \ge 0$. Then the composite map $\operatorname{inc}_{\star}^* \circ \operatorname{inc}_{!}^*$ is equal to the Hecke operator

$$\mathtt{T}^{\star}_{N,\mathfrak{p}}\coloneqq \sum_{\delta=0}^{r}\mathtt{d}_{r-\delta,p}\cdot \mathtt{T}_{N,\mathfrak{p};\delta}\in \mathbb{T}_{N,\mathfrak{p}}$$

in which the numbers $\mathbf{d}_{r-\delta,p}$ are introduced in Notation 1.3.2, and the Hecke operators $\mathbf{T}_{N,\mathfrak{p};\delta}$ are introduced in Notation B.2.1 (as $\mathbf{T}_{N;\delta}^{\circ}$).

Note that by Remark 4.3.8, $L[Sh(V^{\star}, (i-)K_p^{\star})]$ is a $\mathbb{T}_{N,p}$ -module when N is odd.

Proof. This is [XZ, Theorem 9.3.5].

4.4. Functoriality under special morphisms. In this subsection, we study the behavior of various moduli schemes under the special morphisms, which is closely related to the Rankin–Selberg motives for $GL_n \times GL_{n+1}$.

We start from the datum $(V_n, \{\Lambda_{n,\mathfrak{q}}\}_{\mathfrak{q}|p})$ as in the beginning of Subsection 4.1, but with V_n of rank $n \geq 1$. We then have the induced datum

$$(\mathcal{V}_{n+1}, \{\Lambda_{n+1,\mathfrak{q}}\}_{\mathfrak{q}|p}) \coloneqq ((\mathcal{V}_n)_{\sharp}, \{(\Lambda_{n,\mathfrak{q}})_{\sharp}\}_{\mathfrak{q}|p})$$

of rank n + 1 by Definition 3.1.7. For $N \in \{n, n + 1\}$, we let $K_{N,\mathfrak{q}}$ be the stabilizer of $\Lambda_{N,\mathfrak{q}}$, and put $K_{N,p} \coloneqq \prod_{\mathfrak{q}|p} K_{N,\mathfrak{q}}$. Recall the category $\mathfrak{K}(V_n)_{\mathrm{sp}}^p$ and functors $-\flat, -\sharp$ from Definition 3.1.11. To unify notation, we put $-n \coloneqq -\flat$ and $-n+1 \coloneqq -\sharp$. There are five stages of functoriality we will consider.

The first stage concerns Shimura varieties. The canonical inclusions

$$V_n \hookrightarrow V_{n+1}, \quad {\Lambda_{n,\mathfrak{q}} \hookrightarrow \Lambda_{n+1,\mathfrak{q}}}_{\mathfrak{q}|p}$$

induce a morphism

(4.6)
$$\operatorname{sh}_{\uparrow} \colon \operatorname{Sh}(\operatorname{V}_{n}, -_{n}\operatorname{K}_{n,p}) \to \operatorname{Sh}(\operatorname{V}_{n+1}, -_{n+1}\operatorname{K}_{n+1,p})$$

in $\operatorname{Fun}(\mathfrak{K}(V_n)_{sp}^p, \operatorname{Sch}_{/F})$, known as the special morphism.

For the second stage of functoriality, we have a morphism

(4.7)
$$\mathbf{m}_{\uparrow} \colon \mathbf{M}_{\mathfrak{p}}(\mathbf{V}_n, -_n) \to \mathbf{M}_{\mathfrak{p}}(\mathbf{V}_{n+1}, -_{n+1})$$

in $\operatorname{Fun}(\mathfrak{K}(\mathcal{V}_n)_{\operatorname{sp}}^p \times \mathfrak{T}, \operatorname{Sch}_{\mathbb{Z}_p^{\Phi}})_{/\mathbf{T}_p}$ sending an object $(A_0, \lambda_0, \eta_0^p; A, \lambda, \eta^p) \in \mathbf{M}_p(\mathcal{V}_n, \mathcal{K}_n^p)(S)$ to the object $(A_0, \lambda_0, \eta_0^p; A \times A_0, \lambda \times \lambda_0, \eta^p \oplus (\operatorname{id}_{A_0})_*) \in \mathbf{M}_p(\mathcal{V}_{n+1}, \mathcal{K}_{n+1}^p)(S)$. We then have the following commutative diagram

in $\operatorname{Fun}(\mathfrak{K}(\mathcal{V}_n)_{\operatorname{sp}}^p \times \mathfrak{T}, \operatorname{Sch}_{/\mathbb{Q}_p^\Phi})_{/\mathbf{T}_{\mathfrak{p}}^\eta}$.

At the third stage of functoriality, we study the basic correspondence (4.3) under the special morphisms. We will complete a commutative diagram in $\operatorname{Fun}(\mathfrak{K}(V_n)_{sp}^p \times \mathfrak{T}, \operatorname{Sch}_{/\mathbb{F}_p^\Phi})_{/T_p}$ as follows

$$(4.9) \qquad S_{\mathfrak{p}}(V_{n+1}, -n+1) \xleftarrow{\pi_{n+1}} B_{\mathfrak{p}}(V_{n+1}, -n+1) \xrightarrow{\iota_{n+1}} M_{\mathfrak{p}}(V_{n+1}, -n+1) \xrightarrow{\uparrow} M_{\mathfrak{p}}(V_{n+1}, -n+1) \xrightarrow{\downarrow} M_{\mathfrak{p}}(V_{n+1}, -n+1)$$

in which the lower-left square is Cartesian; and the lower (resp. upper) line is the basic correspondences on $M_{\mathfrak{p}}(V_n, -n)$ (resp. $M_{\mathfrak{p}}(V_{n+1}, -n+1)$) as introduced in Definition 4.2.4.

Definition 4.4.1. We define a functor

$$\begin{split} \mathbf{S}_{\mathfrak{p}}(\mathbf{V}_{n},-)_{\mathrm{sp}} \colon \mathfrak{K}(\mathbf{V}_{n})_{\mathrm{sp}}^{p} \times \mathfrak{T} &\to \mathsf{PSch}'_{/\mathbb{F}_{p}^{\Phi}} \\ \mathbf{K}^{p} &\mapsto \mathbf{S}_{\mathfrak{p}}(\mathbf{V}_{n},\mathbf{K}^{p})_{\mathrm{sp}} \end{split}$$

such that for every $S \in \mathsf{Sch}'_{/\mathbb{F}_p^{\Phi}}$, $S_{\mathfrak{p}}(V_n, K^p)_{\mathrm{sp}}(S)$ is the set of equivalence classes of decuples $(A_0, \lambda_0, \eta_0^p; A^\star, \lambda^\star, \eta^{p\star}; A_{\mathfrak{b}}^\star, \eta_{\mathfrak{b}}^{\mathfrak{b}}; \delta^\star)$ where

- O $(A_0, \lambda_0, \eta_0^p; A^\star, \lambda^\star, \eta^{p\star})$ is an element in $S_p(V_n, K_n^p)(S)$;
- $O(A_0, \lambda_0, \eta_0^p; A_{\natural}^{\star}, \lambda_{\natural}^{\star}, \eta_{\natural}^{p\star}) \text{ is an element in } S_{\mathfrak{p}}(V_{n+1}, K_{n+1}^p)(S);$
- $O \delta^* : A^* \times A_0 \to A^*_{\natural}$ is an O_F -linear quasi-*p*-isogeny (Definition 3.4.5) such that
 - (a) ker $\delta^{\star}[p^{\infty}]$ is contained in $(A^{\star} \times A_0)[\mathfrak{p}];$
 - (b) we have $\lambda^* \times \varpi \lambda_0 = \delta^{*\vee} \circ \lambda_{\natural}^* \circ \delta^*$; and
 - (c) the K_{n+1}^p -orbit of maps $v \mapsto \delta_*^\star \circ (\eta^{p\star} \oplus (\mathrm{id}_{A_0})_*)(v)$ for $v \in \mathrm{V}_{n+1} \otimes_{\mathbb{Q}} \mathbb{A}^{\infty,p}$ coincides with $\eta_{\mathfrak{h}}^{p\star}$.

The equivalence relation and the action of morphisms in $\mathfrak{K}(V_n)_{sp}^p \times \mathfrak{T}$ are defined similarly as in Definition 4.2.3.

We have apparently the forgetful morphism

$$S_{\mathfrak{p}}(V_n, -)_{sp} \to T_{\mathfrak{p}}$$

in $\operatorname{Fun}(\mathfrak{K}(V_n)_{\operatorname{sp}}^p \times \mathfrak{T}, \operatorname{PSch}'_{/\mathbb{F}_p^{\Phi}})$ which is represented by finite and étale schemes. By definition, we have the two forgetful morphisms

$$\begin{split} \mathbf{s}_{\downarrow} \colon \mathbf{S}_{\mathfrak{p}}(\mathbf{V}_{n}, -)_{\mathrm{sp}} &\to \mathbf{S}_{\mathfrak{p}}(\mathbf{V}_{n}, -_{n}), \\ \mathbf{s}_{\uparrow} \colon \mathbf{S}_{\mathfrak{p}}(\mathbf{V}_{n}, -)_{\mathrm{sp}} &\to \mathbf{S}_{\mathfrak{p}}(\mathbf{V}_{n+1}, -_{n+1}) \end{split}$$

in $\operatorname{Fun}(\mathfrak{K}(\mathcal{V}_n)_{\operatorname{sp}}^p \times \mathfrak{T}, \operatorname{Sch}_{/\mathbb{F}_p^{\Phi}})_{/\mathrm{T}_p}$.

Lemma 4.4.2. We have the following properties concerning s_{\perp} .

(1) When n is odd, s_{\perp} is an isomorphism, and the morphism

 $s_{\uparrow} \circ s_{\downarrow}^{-1} \colon S_{\mathfrak{p}}(V_n, -_n) \to S_{\mathfrak{p}}(V_{n+1}, -_{n+1})$

is given by the assignment

 $(A_0, \lambda_0, \eta_0^p; A^\star, \lambda^\star, \eta^{p\star}) \mapsto (A_0, \lambda_0, \eta_0^p; A^\star \times A_0, \lambda^\star \times \varpi \lambda_0, \eta^{p\star} \times (\mathrm{id}_{A_0})_*).$

(2) When n is even, s_{\downarrow} is finite étale of degree p + 1.

Proof. Take an object K^p of $\mathfrak{K}(V_n)_{sp}^p$, and a point $x = (A_0, \lambda_0, \eta_0^p; A^\star, \lambda^\star, \eta^{p\star}) \in S_{\mathfrak{p}}(V_n, K_n^p)(\kappa)$ for some perfect field κ containing \mathbb{F}_n^{Φ} .

For (1), it suffices to show that the fibre $s_{\downarrow}^{-1}(x)$ consists of the single point with the extra datum $(A_{\natural}^{\star}, \lambda_{\natural}^{\star}, \eta_{\natural}^{p\star}; \delta^{\star}) = (A^{\star} \times A_0, \lambda^{\star} \times \varpi \lambda_0, \eta^{p\star} \times \eta_0^p; id)$. This follows from the fact that δ^{\star} as in Definition 4.4.1 induces an equivalence between $(A_{\natural}^{\star}, \lambda_{\natural}^{\star}, \eta_{\natural}^{p\star})$ and $(A^{\star} \times A_0, \lambda^{\star} \times \varpi \lambda_0, \eta^{p\star} \times \eta_0^p)$.

For (2), we note first that a point in the fibre $s_{\downarrow}^{-1}(x)$ is determined by the quasi-*p*-isogeny δ^* , which is in turn determined, up to equivalence, by a totally isotropic (O_F/\mathfrak{p}) -subgroup of $\ker(\lambda^* \times \varpi \lambda_0)$ of order p^2 . We classify such subgroups by using Dieudonné theory. Let $\mathcal{D}(A^* \times A_0)_{\tau_{\infty}^{c}}^{\vee}$ be the dual lattice of $\mathcal{D}(A^* \times A_0)_{\tau_{\infty}^{c}}$ (Notation 3.4.12) but with respect to the quasi-polarization $\lambda^* \times \varpi \lambda_0$. The quotient $\mathscr{W}_x := \mathcal{D}(A^* \times A_0)_{\tau_{\infty}^{c}}^{\vee}/\mathcal{D}(A^* \times A_0)_{\tau_{\infty}}$ is κ -vector space of dimension 2 equipped with an induced *nondegenerate* hermitian pairing. Then the hermitian space \mathscr{W}_x is admissible in the sense of Definition A.1.1 with underlying hermitian space over \mathbb{F}_{p^2} given by $\mathscr{W}_{x,0} := \mathscr{W}_x^{\mathsf{V}^{-1}\mathsf{F}=1}$. Then $\mathscr{W}_{x,0}$ is an \mathbb{F}_{p^2} -vector space of dimension 2. By the classical Dieudonné theory for finite group schemes over κ , the set of totally isotropic (O_F/\mathfrak{p}) -subgroups of $\ker(\lambda^* \times \varpi \lambda_0)$ of order p^2 is in natural bijection with the set of isotropic \mathbb{F}_{p^2} -lines in $\mathscr{W}_{x,0}$, which has cardinality p+1.

Definition 4.4.3. We define $B_{\mathfrak{p}}(V_n, -)_{sp}$ to be the fiber product indicated in the following Cartesian diagram

$$\begin{array}{c|c} \mathbf{B}_{\mathfrak{p}}(\mathbf{V}_{n},-)_{\mathrm{sp}} & \xrightarrow{\pi_{\mathrm{sp}}} & \mathbf{S}_{\mathfrak{p}}(\mathbf{V}_{n},-)_{\mathrm{sp}} \\ & & \downarrow^{\mathrm{s}_{\downarrow}} \\ & & \downarrow^{\mathrm{s}_{\downarrow}} \\ & & \mathbf{B}_{\mathfrak{p}}(\mathbf{V}_{n},-_{n}) & \xrightarrow{\pi_{n}} & \mathbf{S}_{\mathfrak{p}}(\mathbf{V}_{n},-_{n}) \end{array}$$

in $\mathsf{Fun}(\mathfrak{K}(\mathcal{V}_n)^p_{\mathrm{sp}} \times \mathfrak{T}, \mathsf{Sch}_{/\mathbb{F}_p^{\Phi}})_{/\mathrm{T}_{\mathfrak{p}}}.$

Lemma 4.4.4. The assignment sending an object

 $((A_0, \lambda_0, \eta_0^p; A, \lambda, \eta^p; A^\star, \lambda^\star, \eta^{p\star}; \alpha), (A_0, \lambda_0, \eta_0^p; A^\star, \lambda^\star, \eta^{p\star}; A^\star_{\natural}, \lambda^\star_{\natural}, \eta^{p\star}_{\natural}; \delta^\star))$

of $B_{\mathfrak{p}}(V_n, K^p)_{sp}(S)$ to

(4.10)
$$(A_0, \lambda_0, \eta_0^p; A \times A_0, \lambda \times \lambda_0, \eta^p \oplus (\mathrm{id}_{A_0})_*; A_{\natural}^{\star}, \lambda_{\natural}^{\star}, \eta_{\natural}^{p\star}; \delta^{\star} \circ (\alpha \times \mathrm{id}_{A_0}))$$

defines a morphism

$$\mathbf{b}_{\uparrow} \colon \mathbf{B}_{\mathfrak{p}}(\mathbf{V}_n, -)_{\mathrm{sp}} \to \mathbf{B}_{\mathfrak{p}}(\mathbf{V}_{n+1}, -n+1)$$

in $\operatorname{Fun}(\mathfrak{K}(V_n)_{\operatorname{sp}}^p \times \mathfrak{T}, \operatorname{Sch}_{/\mathbb{F}_p^{\Phi}})_{/\mathrm{T}_{\mathfrak{p}}}.$

Proof. The lemma amounts to showing that (4.10) is an object of $B_{\mathfrak{p}}(V_{n+1}, K_{n+1}^p)(S)$. Put $\alpha_{\natural} := \delta^* \circ (\alpha \times id_{A_0}) : A \times A_0 \to A_{\natural}^*$. The only nontrivial condition in Definition 4.2.3 to check is that ker $\alpha_{\natural}[p^{\infty}]$ is contained in $(A \times A_0)[\mathfrak{p}]$. For this, we may assume $S = \operatorname{Spec} \kappa$ for a perfect field κ containing \mathbb{F}_n^{Φ} .

Consider the following injective maps of Dieudonné modules

$$\mathcal{D}(A)_{\tau} \oplus \mathcal{D}(A_0)_{\tau} \xrightarrow{\alpha_{*,\tau} \oplus \mathrm{id}} \mathcal{D}(A^*)_{\tau} \oplus \mathcal{D}(A_0)_{\tau} \xrightarrow{\delta_{*,\tau}^*} \mathcal{D}(A_{\natural}^*)_{\tau}$$

for every $\tau \in \Sigma_{\infty}$. We have the inclusion $\mathcal{D}(A^{\star}_{\natural})_{\tau} \subseteq \mathcal{D}(A^{\star})^{\vee}_{\tau^{c}} \oplus \varpi^{-1}\mathcal{D}(A_{0})_{\tau}$ (Notation 3.4.12). Thus, it suffices to show $p\mathcal{D}(A^{\star})^{\vee}_{\tau^{c}} \subseteq \mathcal{D}(A)_{\tau}$ for every $\tau \in \Sigma_{\infty}$. For $\tau \notin \{\tau_{\infty}, \tau^{c}_{\infty}\}$, we have $\mathcal{D}(A^{\star})^{\vee}_{\tau^{c}} = \mathcal{D}(A)_{\tau}$. It remains to show $p\mathcal{D}(A^{\star})^{\vee}_{\tau^{c}} \subseteq \mathcal{D}(A)_{\tau}$ for $\tau \in \{\tau_{\infty}, \tau^{c}_{\infty}\}$. Recall the subspace $H \coloneqq (\check{\alpha}_{*,\tau_{\infty}})^{-1} \omega_{A^{\vee}/\kappa,\tau_{\infty}} \subseteq \mathrm{H}_{1}^{\mathrm{dR}}(A^{\star}/\kappa)_{\tau_{\infty}}$ from Theorem 4.2.5. Under the notation in proof of Theorem 4.2.5, since $(\mathbf{F}H)^{\perp} \subseteq H$, we have $p\mathcal{D}(A^{\star})_{\tau_{\infty}}^{\vee} \subseteq \tilde{H}$ hence $p\tilde{\mathcal{D}}(A^{\star})_{\tau_{\infty}}^{\vee} \subseteq \tilde{H}^{\mathsf{c}}$. Thus, we have

$$p\mathcal{D}(A^{\star})_{\tau_{\infty}^{\mathsf{c}}}^{\vee} = p\mathbb{V}^{-1}(\mathcal{D}(A^{\star})_{\tau_{\infty}}^{\vee}) \subseteq \mathbb{V}^{-1}\tilde{H}^{\mathsf{c}} = \mathcal{D}(A)_{\tau_{\infty}},$$
$$p\mathcal{D}(A^{\star})_{\tau_{\infty}}^{\vee} = p\mathbb{F}(\mathcal{D}(A^{\star})_{\tau_{\infty}^{\mathsf{c}}}^{\vee}) \subseteq \mathbb{F}\tilde{H} = \mathcal{D}(A)_{\tau_{\infty}^{\mathsf{c}}}.$$

The lemma follows.

By the above lemma, we obtain our desired diagram (4.9). Moreover, we have the following result.

Proposition 4.4.5. When n is even, the square

$$\begin{array}{ccc}
\mathbf{B}_{\mathfrak{p}}(\mathbf{V}_{n+1}, -n+1) & \xrightarrow{\iota_{n+1}} & \mathbf{M}_{\mathfrak{p}}(\mathbf{V}_{n+1}, -n+1) \\
& & & \uparrow^{\mathbf{b}_{\uparrow}} & & \uparrow^{\mathbf{m}_{\uparrow}} \\
& & & \mathbf{B}_{\mathfrak{p}}(\mathbf{V}_{n}, -)_{\mathrm{sp}} & \xrightarrow{\iota_{n} \circ \mathbf{b}_{\downarrow}} & & \mathbf{M}_{\mathfrak{p}}(\mathbf{V}_{n}, -n)
\end{array}$$

extracted from the diagram (4.9) is Cartesian.

We remark that the above proposition is not correct on the nose when n is odd and at least 3.

Proof. The square in the proposition induces a morphism

$$\iota_{\rm sp} \colon {\rm B}_{\mathfrak{p}}({\rm V}_n, -)_{\rm sp} \to {\rm B}_{\mathfrak{p}}({\rm V}_{n+1}, -_{n+1}) \times_{{\rm M}_{\mathfrak{p}}({\rm V}_{n+1}, -_{n+1})} {\rm M}_{\mathfrak{p}}({\rm V}_n, -_n).$$

We need to prove that ι_{sp} is an isomorphism. By Theorem 4.2.5, we know that ι_{sp} is locally for the Zariski topology on the source a closed immersion, such that both the source and the target are smooth. Thus, it suffices to show that for a given algebraically closed field κ containing \mathbb{F}_p^{Φ} , we have that

(1) $\iota_{\rm sp}(\kappa)$ is an isomorphism in $\mathsf{Fun}(\mathfrak{K}(\mathbf{V}_n)^p_{\rm sp} \times \mathfrak{T}, \mathsf{Set})$; and

(2) for every $\mathbf{K}^p \in \mathfrak{K}(\mathbf{V}_n)_{sp}^p$ and every $x \in \mathbf{B}_{\mathfrak{p}}(\mathbf{V}_n, \mathbf{K}^p)_{sp}(\kappa)$, the induced diagram

of tangent spaces is a Cartesian square of κ -modules.

For (1), we take an object $K^p \in \mathfrak{K}(V_n)_{sp}^p$ and construct an inverse of $\iota_{sp}(\kappa)$. Take a point

$$(A_0, \lambda_0, \eta_0^p; A, \lambda, \eta^p; A_{\natural}^{\star}, \lambda_{\natural}^{\star}, \eta_{\natural}^{p\star}; \alpha_{\natural})$$

in the target of $\iota_{\rm sp}(\kappa)$. Then α_{\natural} induces an inclusion

$$\mathcal{D}(A)_{\tau} \oplus \mathcal{D}(A_0)_{\tau} \subseteq \mathcal{D}(A_{\natural}^{\star})_{\tau}$$

of Dieudonné modules, which is an equality if $\tau \notin \{\tau_{\infty}, \tau_{\infty}^{c}\}$. We put

$$\mathcal{D}_{A^{\star}} \coloneqq \bigoplus_{\tau \in \Sigma_{\infty}} \mathcal{D}_{A^{\star},\tau}$$

where $\mathcal{D}_{A^{\star},\tau} = \mathcal{D}(A)_{\tau}$ for $\tau \notin \{\tau_{\infty}, \tau_{\infty}^{\mathsf{c}}\}$ and $\mathcal{D}_{A^{\star},\tau} = \mathcal{D}(A_{\natural}^{\star})_{\tau} \cap p^{-1}\mathcal{D}(A)_{\tau}$ for $\tau \in \{\tau_{\infty}, \tau_{\infty}^{\mathsf{c}}\}$. Then $\mathcal{D}_{A^{\star}}$ is a Dieudonné module containing $\mathcal{D}(A)$. By the Dieudonné theory, there is an O_F -abelian

scheme A^* over κ with $\mathcal{D}(A^*)_{\tau} = \mathcal{D}_{A^*,\tau}$ for every $\tau \in \Sigma_{\infty}$, and an O_F -linear isogeny $\alpha \colon A \to A^*$ inducing the inclusion of Dieudonné modules $\mathcal{D}(A) \subseteq \mathcal{D}(A^*)$. We factors α_{\natural} as

$$A \times A_0 \xrightarrow{\alpha \times \mathrm{id}_{A_0}} A^\star \times A_0 \xrightarrow{\delta^\star} A_{\natural}^\star.$$

It is clear that there is a unique quasi-polarization λ^* of A^* such that $\lambda^* \times \varpi \lambda_0 = \delta^{*\vee} \circ \lambda_{\natural}^* \circ \delta^*$. Let η^{p*} be the K_n^p -level structure induced from η^p under α . We claim that the datum

$$((A_0, \lambda_0, \eta_0^p; A, \lambda, \eta^p; A^\star, \lambda^\star, \eta^{p\star}; \alpha), (A_0, \lambda_0, \eta_0^p; A^\star, \lambda^\star, \eta^{p\star}; A^\star_{\natural}, \lambda^\star_{\natural}, \eta^{p\star}_{\natural}; \delta^\star))$$

gives rise to an element in $B_{\mathfrak{p}}(V_n, K^p)_{sp}(\kappa)$. It suffices to show that $(A_0, \lambda_0, \eta_0^p; A^*, \lambda^*, \eta^{p*})$ is an element in $S_{\mathfrak{p}}(V_n, K_n^p)(\kappa)$. Moreover precisely, we need to show that

- (1.1) the O_F -abelian scheme A^* has signature type $n\Phi$; and
- (1.2) ker $\lambda^{\star}[p^{\infty}]$ is contained in $A^{\star}[\mathfrak{p}]$ of degree p^2 .

To prove these, we add two auxiliary properties

- (1.3) the composite map $\mathcal{D}(A_{\natural}^{\star})_{\tau} \subseteq p^{-1}\mathcal{D}(A)_{\tau} \oplus p^{-1}\mathcal{D}(A_{0})_{\tau} \to p^{-1}\mathcal{D}(A_{0})_{\tau}$ is surjective for $\tau \in \{\tau_{\infty}, \tau_{\infty}^{\mathsf{c}}\}$; and
- (1.4) the cokernel of the inclusion $\mathcal{D}(A^{\star})_{\tau} \oplus \mathcal{D}(A_0)_{\tau} \subseteq \mathcal{D}(A_{\natural}^{\star})_{\tau}$ is isomorphic to κ for $\tau \in \{\tau_{\infty}, \tau_{\infty}^{c}\}$.

For (1.3), if not surjective, then we have $\mathcal{D}(A_{\sharp}^{\star})_{\tau} \subseteq p^{-1}\mathcal{D}(A)_{\tau} \oplus \mathcal{D}(A_{0})_{\tau}$ for both $\tau \in \{\tau_{\infty}, \tau_{\infty}^{\mathsf{c}}\}$. As $\varpi \lambda \times \varpi \lambda_{0} = \alpha_{\sharp}^{\vee} \circ \lambda_{\sharp}^{\star} \circ \alpha_{\sharp}$, this contradicts with the fact that λ_{\sharp}^{\star} is *p*-principal.

For (1.4), it follows (1.3) and the fact that the kernel of $\mathcal{D}(A_{\natural}^{\star})_{\tau} \to p^{-1}\mathcal{D}(A_{0})_{\tau}$ is $\mathcal{D}(A^{\star})_{\tau}$ for $\tau \in \{\tau_{\infty}, \tau_{\infty}^{c}\}.$

For (1.1), it amounts to showing that $F: \mathcal{D}(A^*)_{\tau} \to \mathcal{D}(A^*)_{\tau^c}$ is an isomorphism for every $\tau \in \Phi$. This is obvious for $\tau \neq \tau_{\infty}$. When $\tau = \tau_{\infty}$, this follows from (iv) and the fact that both $F: \mathcal{D}(A^*_{\sharp})_{\tau} \to \mathcal{D}(A^*_{\sharp})_{\tau^c}$ and $F: \mathcal{D}(A_0)_{\tau} \to \mathcal{D}(A_0)_{\tau^c}$ are isomorphisms.

For (1.2), it follows from (1.4) and the fact that $\lambda_{\natural}^{\star}$ is *p*-principal.

Thus, (1) is proved.

For (2), the diagram (4.11) is identified with

by Theorem 4.1.3 and Theorem 4.2.5. However, it is an easy consequence of (1.3) that ker $\alpha_{\natural*,\tau_{\infty}} \cap$ $\mathrm{H}_{1}^{\mathrm{dR}}(A)_{\tau_{\infty}} = \ker \alpha_{*,\tau_{\infty}}$. Thus, the above diagram is Cartesian; and (2) follows.

At the fourth stage of functoriality, we compare the special morphisms for basic correspondences and for Deligne–Lusztig varieties. Take a point

$$s^{\star} = (A_0, \lambda_0, \eta_0^p; A^{\star}, \lambda^{\star}, \eta^{p\star}; A_{\natural}^{\star}, \lambda_{\natural}^{\star}, \eta_{\natural}^{p\star}; \delta^{\star}) \in \mathcal{S}_{\mathfrak{p}}(\mathcal{V}_n, \mathcal{K}^p)_{\mathrm{sp}}(\kappa)$$

for a field κ containing \mathbb{F}_p^{Φ} . Put

$$s_n^\star \coloneqq \mathbf{s}_{\downarrow}(s^\star), \quad s_{n+1}^\star \coloneqq \mathbf{s}_{\uparrow}(s^\star);$$

and denote by $B_{s^{\star}}$, $B_{s_n^{\star}}$, and $B_{s_{n+1}^{\star}}$ their preimages under π_{sp} , π_n , and π_{n+1} , respectively. By Lemma 4.2.2, we have admissible pairs $(\mathscr{V}_{s_n^{\star}}, \{,\}_{s_n^{\star}})$ and $(\mathscr{V}_{s_{n+1}^{\star}}, \{,\}_{s_{n+1}^{\star}})$. As in Construction A.1.5, we extend the pair $(\mathscr{V}_{s_n^{\star}}, \{,\}_{s_n^{\star}})$ to $(\mathscr{V}_{s_n^{\star}, \sharp}, \{,\}_{s_n^{\star}, \sharp})$. Then the homomorphism $\delta^{\star} \colon A^{\star} \times A_0 \to A_{\natural}^{\star}$ induces a κ -linear map

$$\delta_{s^\star} \colon \mathscr{V}_{s^\star_n, \sharp} \to \mathscr{V}_{s^\star_{n+1}}$$

satisfying $\{\delta_{s^{\star}}(x), \delta_{s^{\star}}(y)\}_{s_{n+1}^{\star}} = \{x, y\}_{s_{n}^{\star},\sharp}$ for every $x, y \in \mathscr{V}_{s_{n}^{\star},\sharp}$. By Construction A.1.5, we obtain a morphism

$$\delta_{s^{\star}\uparrow} \colon \mathrm{DL}_{s_n^{\star}} = \mathrm{DL}(\mathscr{V}_{s_n^{\star}}, \{ \ , \ \}_{s_n^{\star}}, \lceil \frac{n+1}{2} \rceil) \to \mathrm{DL}_{s_{n+1}^{\star}} = \mathrm{DL}(\mathscr{V}_{s_{n+1}^{\star}}, \{ \ , \ \}_{s_{n+1}^{\star}}, \lceil \frac{n+2}{2} \rceil)$$

of corresponding Deligne–Lusztig varieties.

Proposition 4.4.6. Let the notation be as above. The following diagram



in $\operatorname{Sch}_{\kappa}$ commutes, where $\zeta_{s_n^{\star}}$ and $\zeta_{s_{n+1}^{\star}}$ are the isomorphisms in Theorem 4.2.5(3). In particular, b₁: B_{s^{\star}} \to B_{s_{n+1}^{\star}} is an isomorphism if n is odd, and is a regular embedding of codimension one if n is even.

Proof. Note that by Lemma 4.4.2, the restricted morphism $b_{\downarrow} \colon B_{s^{\star}} \to B_{s^{\star}_n}$ is an isomorphism. Thus, the last claim follows from the commutativity and Proposition A.1.6.

When n is odd, the commutativity is obvious. When n is even, it suffices to show that for every point

$$(A_0, \lambda_0, \eta_0^p; A, \lambda, \eta^p; A^\star, \lambda^\star, \eta^{p\star}; \alpha) \in \mathcal{B}_{s^\star}(S),$$

we have

 $\delta_{*,\tau_{\infty}}^{\star}\left((\breve{\alpha}_{*,\tau_{\infty}})^{-1}\omega_{A^{\vee}/S,\tau_{\infty}} \oplus \mathrm{H}_{1}^{\mathrm{dR}}(A_{0}/S)_{\tau_{\infty}}\right) = (\breve{\alpha}_{\natural*,\tau_{\infty}})^{-1}\omega_{A^{\vee}\times A_{0}^{\vee}/S,\tau_{\infty}}$ (4.12)

in view of the diagram



in which $\check{\alpha} \circ \alpha = \varpi \cdot \mathrm{id}_A$ and $\check{\alpha}_{\natural} \circ \check{\alpha}_{\natural} = \varpi \cdot \mathrm{id}_{A \times A_0}$. Since both sides of (4.12) have the same rank, it suffices to show that

$$\check{\alpha}_{\natural*,\tau_{\infty}}\left(\delta_{*,\tau_{\infty}}^{\star}\left((\check{\alpha}_{*,\tau_{\infty}})^{-1}\omega_{A^{\vee}/S,\tau_{\infty}}\oplus\mathrm{H}_{1}^{\mathrm{dR}}(A_{0}/S)_{\tau_{\infty}}\right)\right)\subseteq\omega_{A^{\vee}\times A_{0}^{\vee}/S,\tau_{\infty}},$$

which is obvious as ϖ annihilates $\mathrm{H}_{1}^{\mathrm{dR}}(A_{0}/S)_{\tau_{\infty}}$. The proposition is proved.

At the final stage of functoriality, we relate the special morphisms for sources of basic correspondences to Shimura sets under the uniformization map v (4.4).

Notation 4.4.7. As in Definition 4.3.1, we choose a definite uniformization datum $(\mathcal{V}_{n}^{\star}, \mathbf{i}_{n}, \{\Lambda_{n,\mathfrak{q}}^{\star}\}_{\mathfrak{q}|p})$ for V. We also fix a definite uniformization datum $(\mathcal{V}_{n+1}^{\star}, \mathbf{i}_{n+1}, \{\Lambda_{n+1,\mathfrak{q}}^{\star}\}_{\mathfrak{q}|p})$ for V_{n+1} satisfying

- O $V_{n+1}^{\star} = (V_n^{\star})_{\sharp}$ and $\mathbf{i}_{n+1} = (\mathbf{i}_n)_{\sharp}$; O $\Lambda_{n+1,\mathfrak{q}}^{\star} = (\Lambda_{n,\mathfrak{q}}^{\star})_{\sharp}$ for $\mathfrak{q} \neq \mathfrak{p}$; and
- $O (\Lambda_{n,\mathfrak{p}}^{\star})_{\sharp} \subseteq \Lambda_{n+1,\mathfrak{p}}^{\star} \subseteq p^{-1}(\Lambda_{n,\mathfrak{p}}^{\star})_{\sharp}^{\vee}.$

Let $K_{n+1,\mathfrak{q}}^{\star}$ be the stabilizer of $\Lambda_{n+1,\mathfrak{q}}^{\star}$ for every \mathfrak{q} over p; and put $K_{n+1,\mathfrak{p}}^{\star} \coloneqq \prod_{\mathfrak{q}|p} K_{n+1,\mathfrak{q}}^{\star}$. Moreover, we put $K_{sp,\mathfrak{p}}^{\star} \coloneqq K_{n,\mathfrak{p}}^{\star} \cap K_{n+1,\mathfrak{p}}^{\star}$ (as a subgroup of $K_{n,\mathfrak{p}}^{\star}$) and $K_{sp,\mathfrak{p}}^{\star} \coloneqq K_{sp,\mathfrak{p}}^{\star} \times \prod_{\mathfrak{q}\neq\mathfrak{p}} K_{n,\mathfrak{q}}^{\star}$.

Remark 4.4.8. When n is odd, since $(\Lambda_{n,\mathfrak{p}}^{\star})^{\vee} = p\Lambda_{n,\mathfrak{p}}^{\star}$, we must have $\Lambda_{n+1,\mathfrak{p}}^{\star} = (\Lambda_{n,\mathfrak{p}}^{\star})_{\sharp}$ as well hence $\mathcal{K}_{sp,p}^{\star} = \mathcal{K}_{n,p}^{\star}$. When n is even, the number of choices of $\Lambda_{n+1,\mathfrak{p}}^{\star}$ is p+1.

Similar to Construction 4.3.2, we may construct a uniformization map

(4.13)
$$\upsilon_{\rm sp} \colon {\rm S}_{\mathfrak{p}}({\rm V}_n, -)_{\rm sp}(\overline{\mathbb{F}}_p) \to {\rm Sh}({\rm V}_n^{\star}, (\mathtt{i}_n - {}_n){\rm K}_{{\rm sp}, p}^{\star}) \times {\rm T}_{\mathfrak{p}}(\overline{\mathbb{F}}_p)$$

in $\operatorname{Fun}(\mathfrak{K}(V_n)_{\operatorname{sp}}^p \times \mathfrak{T}, \operatorname{Set})_{/\operatorname{T}_n(\overline{\mathbb{F}}_n)}$ which is an isomorphism, whose details we leave to readers.

Proposition 4.4.9. The following diagram

$$\begin{split} & \operatorname{S}_{\mathfrak{p}}(\operatorname{V}_{n+1}, -_{n+1})(\overline{\mathbb{F}}_{p}) \xrightarrow{v_{n+1}} \operatorname{Sh}(\operatorname{V}_{n+1}^{\star}, (\mathbf{i}_{n+1} -_{n+1})\operatorname{K}_{n+1,p}^{\star}) \times \operatorname{T}_{\mathfrak{p}}(\overline{\mathbb{F}}_{p}) \\ & \underset{s_{\uparrow}(\overline{\mathbb{F}}_{p})}{\overset{(4.4)}{}} \xrightarrow{\operatorname{Sh}(\operatorname{V}_{n+1}^{\star}, (\mathbf{i}_{n+1} -_{n+1})\operatorname{K}_{n+1,p}^{\star}) \times \operatorname{T}_{\mathfrak{p}}(\overline{\mathbb{F}}_{p})} \\ & \underset{s_{\downarrow}(\overline{\mathbb{F}}_{p})}{\overset{(1.13)}{}} \xrightarrow{v_{\operatorname{sp}}} \operatorname{Sh}(\operatorname{V}_{n}^{\star}, (\mathbf{i}_{n} -_{n})\operatorname{K}_{\operatorname{sp},p}^{\star}) \times \operatorname{T}_{\mathfrak{p}}(\overline{\mathbb{F}}_{p})} \\ & \underset{s_{\downarrow}(\overline{\mathbb{F}}_{p})}{\overset{(1.13)}{}} \xrightarrow{v_{n}} \operatorname{Sh}(\operatorname{V}_{n}^{\star}, (\mathbf{i}_{n} -_{n})\operatorname{K}_{n,p}^{\star}) \times \operatorname{T}_{\mathfrak{p}}(\overline{\mathbb{F}}_{p})} \end{split}$$

in $\operatorname{Fun}(\mathfrak{K}(V_n)_{\operatorname{sp}}^p \times \mathfrak{T}, \operatorname{Set})_{/\operatorname{T}_{\mathfrak{p}}(\overline{\mathbb{F}}_p)}$ commutes, where $\operatorname{sh}_{\downarrow}^{\star}$ and $\operatorname{sh}_{\uparrow}^{\star}$ are obvious maps on Shimura sets. Moreover, the induced actions of $\operatorname{Gal}(\overline{\mathbb{F}}_p/\mathbb{F}_p^{\Phi})$ on all terms on the right-hand side factor through the projection to the factor $\operatorname{T}_{\mathfrak{p}}(\overline{\mathbb{F}}_p)$.

Proof. The commutativity follows directly from definition. The proof of the last claim is same to Proposition 4.3.4.

4.5. Second geometric reciprocity law. In this subsection, we state and prove a theorem we call *second geometric reciprocity law*, which can be regarded a geometric template for the second explicit reciprocity law studied in Subsection 7.3 once throw the automorphic input.

We keep the setup in Subsection 4.4. However, we allow - = (-n, -n+1) to be an object of $\mathfrak{K}(\mathbf{V}_n)^p \times \mathfrak{K}(\mathbf{V}_{n+1})^p$, rather than $\mathfrak{K}(\mathbf{V}_n)_{sp}^p$. Denote by n_0 and n_1 the unique even and odd numbers in $\{n, n+1\}$, respectively. Write $n_0 = 2r_0$ and $n_1 = 2r_1 + 1$ for unique integers $r_0, r_1 \ge 1$. In particular, we have $n = r_0 + r_1$. Let L be a p-coprime coefficient ring.

To ease notation, we put $X_{n_{\alpha}}^{?} \coloneqq X_{\mathfrak{p}}^{?}(V_{n_{\alpha}}, -n_{\alpha})$ for meaningful triples $(X, ?, \alpha) \in \{\mathbf{M}, \mathbf{M}, \mathbf{B}, \mathbf{S}\} \times \{, \eta\} \times \{0, 1\}.$

Construction 4.5.1. We construct two maps and two graphs.

(1) For every integers i, j, we define

$$\operatorname{loc}_{\mathfrak{p}}^{\prime} \colon \operatorname{H}_{\operatorname{\acute{e}t}}^{\iota}(\operatorname{Sh}(\operatorname{V}_{n_{0}}, -_{n_{0}}\operatorname{K}_{n_{0}, p}) \times_{\operatorname{Spec} F} \operatorname{Sh}(\operatorname{V}_{n_{1}}, -_{n_{1}}\operatorname{K}_{n_{1}, p}), L(j)) \to \operatorname{H}_{\mathfrak{T}}^{\iota}(\operatorname{M}_{n_{0}} \times_{\operatorname{T}_{\mathfrak{p}}} \operatorname{M}_{n_{1}}, L(j))$$

to be the composition of the localization map

$$loc_{\mathfrak{p}} \colon \mathrm{H}^{i}_{\mathrm{\acute{e}t}}(\mathrm{Sh}(\mathrm{V}_{n_{0}},-_{n_{0}}\mathrm{K}_{n_{0},p}) \times_{\mathrm{Spec}\,F} \mathrm{Sh}(\mathrm{V}_{n_{1}},-_{n_{1}}\mathrm{K}_{n_{1},p}),L(j)) \rightarrow \mathrm{H}^{i}_{\mathrm{\acute{e}t}}((\mathrm{Sh}(\mathrm{V}_{n_{0}},-_{n_{0}}\mathrm{K}_{n_{0},p}) \times_{\mathrm{Spec}\,F} \mathrm{Sh}(\mathrm{V}_{n_{1}},-_{n_{1}}\mathrm{K}_{n_{1},p})) \otimes_{F} \mathbb{Q}_{p}^{\Phi},L(j))$$

the pullback map

$$\mathrm{H}^{i}_{\mathrm{\acute{e}t}}((\mathrm{Sh}(\mathrm{V}_{n_{0}},-_{n_{0}}\mathrm{K}_{n_{0},p})\times_{\mathrm{Spec}\,F}\mathrm{Sh}(\mathrm{V}_{n_{1}},-_{n_{1}}\mathrm{K}_{n_{1},p}))\otimes_{F}\mathbb{Q}_{p}^{\Phi},L(j))\to\mathrm{H}^{i}_{\mathfrak{T}}(\mathbf{M}^{\eta}_{n_{0}}\times_{\mathbf{T}^{\eta}_{\mathfrak{p}}}\mathbf{M}^{\eta}_{n_{1}},L(j))$$

induced from (4.2), and the isomorphism

 $\mathrm{H}^{i}_{\mathfrak{T}}(\mathrm{M}_{n_{0}}\times_{\mathrm{T}_{\mathfrak{p}}}\mathrm{M}_{n_{1}},\mathrm{R}\Psi L(j))\xrightarrow{\sim}\mathrm{H}^{i}_{\mathfrak{T}}(\mathrm{M}_{n_{0}}\times_{\mathrm{T}_{\mathfrak{p}}}\mathrm{M}_{n_{1}},L(j))$

due to the fact $L \simeq R\Psi L$ by Theorem 4.1.3.

(2) Analogous to Construction 4.3.9, we define the map

$$\operatorname{inc}_{!}^{\star,\star} \colon L[\operatorname{Sh}(\operatorname{V}_{n_{0}}^{\star},(\mathfrak{i}_{n_{0}}-_{n_{0}})\operatorname{K}_{n_{0},p}^{\star})] \otimes_{L} L[\operatorname{Sh}(\operatorname{V}_{n_{1}}^{\star},(\mathfrak{i}_{n_{1}}-_{n_{1}})\operatorname{K}_{n_{1},p}^{\star})] \\ \xrightarrow{\sim} \operatorname{H}_{\mathfrak{T}}^{0}(\operatorname{S}_{n_{0}},L) \otimes_{L} \operatorname{H}_{\mathfrak{T}}^{0}(\operatorname{S}_{n_{1}},L) = \operatorname{H}_{\mathfrak{T}}^{0}(\operatorname{S}_{n_{0}}\times_{\operatorname{T}_{\mathfrak{p}}}\operatorname{S}_{n_{1}},L) \\ \xrightarrow{(\pi_{n_{0}}\times\pi_{n_{1}})^{*}} \operatorname{H}_{\mathfrak{T}}^{0}(\operatorname{B}_{n_{0}}\times_{\operatorname{T}_{\mathfrak{p}}}\operatorname{B}_{n_{1}},L) \xrightarrow{(\iota_{n_{0}}\times\iota_{n_{1}})_{!}} \operatorname{H}_{\mathfrak{T}}^{2n}(\operatorname{M}_{n_{0}}\times_{\operatorname{T}_{\mathfrak{p}}}\operatorname{M}_{n_{1}},L(n))$$

in $\operatorname{Fun}(\mathfrak{K}(\mathbf{V}_n)^p \times \mathfrak{K}(\mathbf{V}_{n+1})^p, \operatorname{\mathsf{Mod}}(L)).$

Suppose that – is taken in the subcategory $\Re(V_n)_{sp}^p$.

(3) We define $\Delta \operatorname{Sh}(V_n, -_n K_{n,p})$ to be the graph of the morphism $\operatorname{sh}_{\uparrow}$ (4.6), as a closed subscheme of $\operatorname{Sh}(V_{n_0}, -_{n_0} K_{n_0,p}) \times_{\operatorname{Spec} F} \operatorname{Sh}(V_{n_1}, -_{n_1} K_{n_1,p})$, which gives rise to a class

 $[\triangle \operatorname{Sh}(\operatorname{V}_n, -_n \operatorname{K}_{n,p})] \in \operatorname{H}^{2n}_{\operatorname{\acute{e}t}}(\operatorname{Sh}(\operatorname{V}_{n_0}, -_{n_0} \operatorname{K}_{n_0,p}) \times_{\operatorname{Spec} F} \operatorname{Sh}(\operatorname{V}_{n_1}, -_{n_1} \operatorname{K}_{n_1,p}), L(n))$

by the absolute cycle class map.

(4) We define $\Delta \operatorname{Sh}(\operatorname{V}_{n}^{\star},(\mathbf{i}_{n}-\mathbf{n})\operatorname{K}_{\operatorname{sp},p}^{\star})$ to be the graph of the correspondence $(\operatorname{sh}_{\downarrow}^{\star},\operatorname{sh}_{\uparrow}^{\star})$, which is a subset of $\operatorname{Sh}(\operatorname{V}_{n_{0}}^{\star},(\mathbf{i}_{n_{0}}-\mathbf{n}_{0})\operatorname{K}_{n_{0},p}^{\star}) \times \operatorname{Sh}(\operatorname{V}_{n_{1}}^{\star},(\mathbf{i}_{n_{1}}-\mathbf{n}_{1})\operatorname{K}_{n_{1},p}^{\star})$.

The following theorem, which we call the second geometric reciprocity law, relates the class $[\Delta \operatorname{Sh}(V_n, -_n K_{n,p})]$ with an explicit class coming from the Shimura set.

Theorem 4.5.2 (Second geometric reciprocity law). Suppose that – is taken in the subcategory $\mathfrak{K}(V_n)_{sp}^p$. We have

$$\mathsf{T}^{\star}_{n_{1},\mathfrak{p}}\cdot(\mathrm{id}\times\pi_{n_{1}})_{!}(\mathrm{id}\times\iota_{n_{1}})^{*}\mathrm{loc}_{\mathfrak{p}}'\left([\bigtriangleup\operatorname{Sh}(\mathsf{V}_{n},-_{n}\mathsf{K}_{n,p})]\right)=(\mathrm{id}\times\pi_{n_{1}})_{!}(\mathrm{id}\times\iota_{n_{1}})^{*}\mathrm{inc}_{!}^{\star,\star}(\mathbb{1}_{\bigtriangleup\operatorname{Sh}(\mathsf{V}^{\star}_{n},(\mathtt{i}_{n}-_{n})\mathsf{K}^{\star}_{\mathrm{sp},p}))$$

in $\mathrm{H}^{2r_0}_{\mathfrak{T}}(\mathrm{M}_{n_0} \times_{\mathrm{T}_{\mathfrak{p}}} \mathrm{S}_{n_1}, L(r_0))$, where $\mathrm{T}^{\star}_{n_1,\mathfrak{p}} \in \mathbb{T}_{n_1,\mathfrak{p}}$ is the Hecke operator appeared in Theorem 4.3.10.

Note that by Proposition 4.3.6 and Remark 4.3.8, $H_{\mathfrak{T}}^{2r_0}(M_{n_0} \times_{T_{\mathfrak{p}}} S_{n_1}, L(r_0))$ is a $\mathbb{T}_{n_1,\mathfrak{p}}$ -module. For readers' convenience, we illustrate the identity in the above theorem through the following diagram



Proof. We denote

$$\mathbf{m}_{\bigtriangleup} \colon \mathbf{M}_n o \mathbf{M}_n imes_{\mathbf{T}_p} \mathbf{M}_{n+1} = \mathbf{M}_{n_0} imes_{\mathbf{T}_p} \mathbf{M}_{n_1}$$

the diagonal morphism of the correspondence (id, \mathbf{m}_{\uparrow}) (4.7) in $\mathsf{Fun}(\mathfrak{K}(\mathbf{V}_n)_{\mathrm{sp}}^p \times \mathfrak{T}, \mathsf{Sch}_{\mathbb{Z}_p^{\Phi}})_{/\mathbf{T}_p}$. Then we have the identity

$$\operatorname{loc}_{\mathfrak{p}}'([\bigtriangleup \operatorname{Sh}(\operatorname{V}_{n},-_{n}\operatorname{K}_{n,p})]) = \operatorname{m}_{\bigtriangleup !}[\operatorname{M}_{n}] \in \operatorname{H}_{\mathfrak{T}}^{2n}(\operatorname{M}_{n}\times_{\operatorname{T}_{\mathfrak{p}}}\operatorname{M}_{n+1},L(n))$$

by the commutative diagram (4.8).

Put $B_{sp} \coloneqq B_{\mathfrak{p}}(V_n, -)_{sp}$ for short, and denote

$$\mathbf{b}_{\Delta} \coloneqq (\mathbf{b}_{\downarrow}, \mathbf{b}_{\uparrow}) \colon \mathbf{B}_{\mathrm{sp}} \to \mathbf{B}_n \times_{\mathbf{T}_{\mathfrak{p}}} \mathbf{B}_{n+1} = \mathbf{B}_{n_0} \times_{\mathbf{T}_{\mathfrak{p}}} \mathbf{B}_{n_1}$$

the diagonal morphism of the correspondence $(b_{\downarrow}, b_{\uparrow})$. By Proposition 4.4.5 (resp. Lemma 4.4.2) when $n = n_0$ (resp. $n = n_1$), the following commutative diagram

$$\begin{array}{c|c} \mathbf{B}_{\mathrm{sp}} \xrightarrow{(\iota_{n_0} \times \mathrm{id}) \circ \mathbf{b}_{\bigtriangleup}} \mathbf{M}_{n_0} \times_{\mathbf{T}_{\mathfrak{p}}} \mathbf{B}_{n_1} \\ \downarrow & & \downarrow^{\mathrm{id} \times \iota_{n_1}} \\ \mathbf{M}_n \xrightarrow{\mathbf{m}_{\bigtriangleup}} \mathbf{M}_{n_0} \times_{\mathbf{T}_{\mathfrak{p}}} \mathbf{M}_{n_1} \end{array}$$

is Cartesian. Then by Proper Base Change, we have

$$\begin{aligned} \mathbf{T}_{n_{1},\mathfrak{p}}^{\star}.(\mathrm{id}\times\pi_{n_{1}})_{!}(\mathrm{id}\times\iota_{n_{1}})^{*}\mathbf{m}_{\bigtriangleup !}[\mathbf{M}_{n}] &= \mathbf{T}_{n_{1},\mathfrak{p}}^{\star}.(\mathrm{id}\times\pi_{n_{1}})_{!}((\iota_{n_{0}}\times\mathrm{id})\circ\mathbf{b}_{\bigtriangleup})_{!}(\iota_{n}\circ\mathbf{b}_{\downarrow})^{*}[\mathbf{M}_{n}] \\ &= \mathbf{T}_{n_{1},\mathfrak{p}}^{\star}.(\mathrm{id}\times\pi_{n_{1}})_{!}((\iota_{n_{0}}\times\mathrm{id})\circ\mathbf{b}_{\bigtriangleup})_{!}[\mathbf{B}_{\mathrm{sp}}]. \end{aligned}$$

The commutative diagram

$$\begin{array}{c|c} \mathbf{B}_{\mathrm{sp}} & \xrightarrow{(\iota_{n_0} \times \mathrm{id}) \circ \mathbf{b}_{\bigtriangleup}} & \mathbf{M}_{n_0} \times_{\mathrm{T}_{\mathfrak{p}}} \mathbf{B}_{n_1} \\ (\mathrm{id} \times \pi_{n_1}) \circ \mathbf{b}_{\bigtriangleup} & & & & & & \\ \mathbf{B}_{n_0} \times_{\mathrm{T}_{\mathfrak{p}}} \mathbf{S}_{n_1} & \xrightarrow{\iota_{n_0} \times \mathrm{id}} & \mathbf{M}_{n_0} \times_{\mathrm{T}_{\mathfrak{p}}} \mathbf{S}_{n_1} \end{array}$$

implies the identity

$$\mathbf{T}_{n_1,\mathfrak{p}}^{\star}.(\mathrm{id}\times\pi_{n_1})_!((\iota_{n_0}\times\mathrm{id})\circ\mathrm{b}_{\bigtriangleup})_![\mathrm{B}_{\mathrm{sp}}]=\mathbf{T}_{n_1,\mathfrak{p}}^{\star}.(\iota_{n_0}\times\mathrm{id})_!((\mathrm{id}\times\pi_{n_1})\circ\mathrm{b}_{\bigtriangleup})_![\mathrm{B}_{\mathrm{sp}}].$$

Now by the definition of B_{sp} (Definition 4.4.3), we have

$$((\mathrm{id} \times \pi_{n_1}) \circ \mathrm{b}_{\triangle})_![\mathrm{B}_{\mathrm{sp}}] = (\pi_{n_0} \times \mathrm{id})^* (\mathbb{1}_{\triangle \operatorname{Sh}(\mathrm{V}_n^\star, (\mathtt{i}_n - n)\mathrm{K}_{\mathrm{sp}, p}^\star)}).$$

In all, we have

$$\mathbf{T}_{n_1,\mathfrak{p}}^{\star}.(\mathrm{id}\times\pi_{n_1})!(\mathrm{id}\times\iota_{n_1})^*\mathbf{m}_{\triangle}![\mathbf{M}_n] = (\iota_{n_0}\times\mathrm{id})!(\pi_{n_0}\times\mathrm{id})^*(\mathbf{T}_{n_1,\mathfrak{p}}^{\star}.\mathbb{1}_{\triangle\operatorname{Sh}(\mathbf{V}_n^{\star},(\mathbf{i}_n-n)\mathbf{K}_{\mathrm{sp},p}^{\star})}),$$

which, by Theorem 4.3.10, equals

$$(\iota_{n_0} \times \mathrm{id})_! (\pi_{n_0} \times \mathrm{id})^* (\mathrm{id} \times \pi_{n_1})_! (\mathrm{id} \times \iota_{n_1})^* (\mathrm{id} \times \iota_{n_1})_! (\mathrm{id} \times \pi_{n_1})^* (\mathbb{1}_{\triangle \operatorname{Sh}(\mathrm{V}_n^\star, (\mathfrak{i}_n - n)\mathrm{K}_{\operatorname{sp}, p}^\star)}) = (\mathrm{id} \times \pi_{n_1})_! (\mathrm{id} \times \iota_{n_1})^* \mathrm{inc}_!^{\star,\star} (\mathbb{1}_{\triangle \operatorname{Sh}(\mathrm{V}_n^\star, (\mathfrak{i}_n - n)\mathrm{K}_{\operatorname{sp}, p}^\star)}).$$

The theorem follows.

5. Unitary moduli schemes: semistable case

In this section, we define and study certain semistable integral moduli scheme whose generic fiber is the product of a unitary Shimura variety and an auxiliary CM moduli. Since the materials in this section are strictly in the linear order, we will leave the summary of contents to each subsection.

We fix a special inert prime (Definition 3.3.4) \mathfrak{p} of F^+ (with the underlying rational prime p). We take the prescribed subring \mathbb{P} in Definition 3.4.2 to be $\mathbb{Z}_{(p)}$. We choose following data

O a CM type Φ containing τ_{∞} ;

- O a rational skew-hermitian space W_0 over $O_F \otimes \mathbb{Z}_{(p)}$ of rank 1 and type Φ (Definition 3.5.3);
- O a neat open compact subgroup $K_0^p \subseteq T_0(\mathbb{A}^{\infty,p})$;
- O an isomorphism $\mathbb{Q}_p \simeq \mathbb{C}$ that induces the place \mathfrak{p} of F^+ ;
- O an element $\varpi \in O_{F^+}$ that is totally positive and satisfies $\operatorname{val}_{\mathfrak{p}}(\varpi) = 1$, and $\operatorname{val}_{\mathfrak{q}}(\varpi) = 0$ for every prime $\mathfrak{q} \neq \mathfrak{p}$ of F^+ above p.

We adopt Notation 3.3.6. In particular, \mathbb{F}_p^{Φ} contains \mathbb{F}_{p^2} . Since W_0 and K_0^p are insensitive and will never be changed in the remaining part of this section, we will not include them in all notations. However, we will keep the prime \mathfrak{p} in notations as later in application, we need to choose different primes in a crucial step. Put $\mathbf{T}_{\mathfrak{p}} := \mathbf{T}_p(W_0, K_0^p) \otimes_{O_{F_{\Phi}} \otimes \mathbb{Z}_{(p)}} \mathbb{Z}_p^{\Phi}$.

5.1. Construction of moduli schemes. In this subsection, we construct our initial moduli schemes. We start from the datum $(V^{\circ}, \{\Lambda^{\circ}_{\mathfrak{q}}\}_{\mathfrak{q}|p})$ where

- O V° is a standard *definite* hermitian space (Definition 3.1.7) over F of rank $N \ge 1$, and
- O for every prime \mathfrak{q} of F^+ above p, a self-dual $O_{F_{\mathfrak{q}}}$ -lattice $\Lambda^{\circ}_{\mathfrak{q}}$ in $\mathcal{V}^{\circ} \otimes_F F_{\mathfrak{q}}$.

Definition 5.1.1. We define a functor

$$\begin{split} \mathbf{M}_{\mathfrak{p}}(\mathbf{V}^{\circ},-) \colon \mathfrak{K}(\mathbf{V}^{\circ})^{p} \times \mathfrak{T} &\to \mathsf{PSch}'_{/\mathbb{Z}_{p}^{\Phi}} \\ \mathbf{K}^{p \circ} &\mapsto \mathbf{M}_{\mathfrak{p}}(\mathbf{V}^{\circ},\mathbf{K}^{p \circ}) \end{split}$$

such that for every $S \in \mathsf{Sch}'_{\mathbb{Z}_p^{\Phi}}, \mathbf{M}_{\mathfrak{p}}(\mathcal{V}^\circ, \mathcal{K}^{p\circ})(S)$ is the set of equivalence classes of sextuples $(A_0, \lambda_0, \eta_0^p; A, \lambda, \eta^p)$ where

- $O(A_0, \lambda_0, \eta_0^p)$ is an element in $\mathbf{T}_{\mathfrak{p}}(S)$;
- O (A, λ) is a unitary O_F -abelian scheme of signature type $N\Phi \tau_{\infty} + \tau_{\infty}^{c}$ over S (Definitions 3.4.2 and 3.4.3) such that ker $\lambda[p^{\infty}]$ is contained in $A[\mathfrak{p}]$ of rank p^2 ;
- O η^p is a K^{*p*°}-level structure, that is, for a chosen geometric point *s* on every connected component of *S*, a $\pi_1(S, s)$ -invariant K^{*p*°}-orbit of isomorphisms

$$\eta^p \colon \mathcal{V}^{\circ} \otimes_{\mathbb{Q}} \mathbb{A}^{\infty, p} \to \operatorname{Hom}_{F \otimes_{\mathbb{Q}} \mathbb{A}^{\infty, p}}^{\lambda_{0, \lambda}}(\mathcal{H}_{1}^{\operatorname{\acute{e}t}}(A_{0s}, \mathbb{A}^{\infty, p}), \mathcal{H}_{1}^{\operatorname{\acute{e}t}}(A_{s}, \mathbb{A}^{\infty, p}))$$

of hermitian spaces over $F \otimes_{\mathbb{Q}} \mathbb{A}^{\infty,p} = F \otimes_{F^+} \mathbb{A}^{\infty,p}_{F^+}$. See Construction 3.4.4 (with $\square = \{\infty, p\}$) for the right-hand side.

The equivalence relation and the action of morphisms in $\mathfrak{K}(\mathcal{V}^{\circ})^{p} \times \mathfrak{T}$ are defined similarly as in Definition 4.1.2.

Remark 5.1.2. In the definition of the moduli functor $\mathbf{M}_{\mathfrak{p}}(\mathbf{V}^{\circ}, -)$, we use the *definite* hermitian space \mathbf{V}° to define the tame level structure – this is different from the usual treatment. The reason for doing this is to make the uniformization map (5.4) for certain stratum in the special fiber of $\mathbf{M}_{\mathfrak{p}}(\mathbf{V}^{\circ}, -)$ canonical, since our main interest is the Shimura set $\mathrm{Sh}(\mathbf{V}^{\circ}, -\mathbf{K}_{p}^{\circ})$, while the trade-off is that the relation between the generic fiber of $\mathbf{M}_{\mathfrak{p}}(\mathbf{V}^{\circ}, -)$ and unitary Shimura varieties cannot be made canonical (see Definition 5.1.6).

We have apparently the forgetful morphism

(5.1)
$$\mathbf{M}_{\mathfrak{p}}(\mathbf{V}^{\circ}, \mathbf{-}) \to \mathbf{T}_{\mathfrak{p}}$$

in $\operatorname{Fun}(\mathfrak{K}(V^{\circ})^{p} \times \mathfrak{T}, \operatorname{PSch}'_{\mathbb{Z}_{p}^{\Phi}})$, which is representable by quasi-projective schemes. According to Notation 3.3.6, we shall denote by the base change of (5.1) to \mathbb{F}_{p}^{Φ} by $\operatorname{M}_{\mathfrak{p}}(V^{\circ}, -) \to \operatorname{T}_{\mathfrak{p}}$, which is a morphism in $\operatorname{Fun}(\mathfrak{K}(V^{\circ})^{p} \times \mathfrak{T}, \operatorname{Sch}_{/\mathbb{F}_{p}^{\Phi}})$.

Definition 5.1.3. For every $K^{p\circ} \in \mathfrak{K}(V^{\circ})^{p}$, let $(\mathcal{A}_{0}, \lambda_{0}, \eta_{0}^{p}; \mathcal{A}, \lambda, \eta^{p})$ be the universal object over $\mathbf{M}_{\mathfrak{p}}(V^{\circ}, K^{p\circ})$. We define

(1) $M_{\mathfrak{p}}^{\circ}(V^{\circ}, K^{p^{\circ}})$ to be the locus of $M_{\mathfrak{p}}(V^{\circ}, K^{p^{\circ}})$ on which $\omega_{\mathcal{A}^{\vee}, \tau_{\infty}}$ coincides with $H_{1}^{dR}(\mathcal{A})_{\tau_{\infty}^{\circ}}^{\perp}$, which we call the *balloon stratum*,⁵

⁵This terminology is borrowed from an unpublished note by Kudla and Rapoport, where they study the corresponding Rapoport–Zink space. The intuition becomes clear after Theorem 5.2.4 where we show that this stratum is a projective space fibration over a zero-dimensional scheme.

(2) $M^{\bullet}_{\mathfrak{p}}(V^{\circ}, K^{p\circ})$ to be the locus of $M_{\mathfrak{p}}(V^{\circ}, K^{p\circ})$ on which $H^{dR}_{1}(\mathcal{A})^{\perp}_{\tau_{\infty}}$ is a line subbundle of $\omega_{\mathcal{A}^{\vee}, \tau^{\circ}_{\infty}}$, which we call the *ground stratum*;

(3) $\mathrm{M}^{\dagger}_{\mathfrak{p}}(\mathrm{V}^{\circ}, \mathrm{K}^{p\circ})$ to be $\mathrm{M}^{\circ}_{\mathfrak{p}}(\mathrm{V}^{\circ}, \mathrm{K}^{p\circ}) \cap \mathrm{M}^{\bullet}_{\mathfrak{p}}(\mathrm{V}^{\circ}, \mathrm{K}^{p\circ})$, which we call the *link stratum*.⁶ We denote

$$\begin{split} \mathbf{m}^{\dagger \circ} \colon \mathbf{M}^{\dagger}_{\mathfrak{p}}(\mathbf{V}^{\circ}, -) &\to \mathbf{M}^{\circ}_{\mathfrak{p}}(\mathbf{V}^{\circ}, -), \\ \mathbf{m}^{\dagger \bullet} \colon \mathbf{M}^{\dagger}_{\mathfrak{p}}(\mathbf{V}^{\circ}, -) &\to \mathbf{M}^{\bullet}_{\mathfrak{p}}(\mathbf{V}^{\circ}, -), \end{split}$$

the obvious inclusion morphisms.

Remark 5.1.4. When N = 1, the ground stratum and link stratum are both empty.

Theorem 5.1.5. For every $K^{p\circ} \in \mathfrak{K}(V^{\circ})^p$, we have

(1) The scheme $\mathbf{M}_{\mathfrak{p}}(\mathbf{V}^{\circ}, \mathbf{K}^{p\circ})$ is quasi-projective and strictly semi-stable over $\mathbf{T}_{\mathfrak{p}}$ of relative dimension N-1; and we have

$$M_{\mathfrak{p}}(V^{\circ}, K^{p \circ}) = M_{\mathfrak{p}}^{\circ}(V^{\circ}, K^{p \circ}) \bigcup M_{\mathfrak{p}}^{\bullet}(V^{\circ}, K^{p \circ}).$$

Moreover, (5.1) is projective if and only if its base change to \mathbb{Q}_p^{Φ} is.

- (2) The loci M^o_p(V^o, K^{po}) and M^o_p(V^o, K^{po}) are both closed subsets of M_p(V^o, K^{po}), (whose induced reduced schemes are) smooth over T_p.
- (3) We have a canonical isomorphism for the relative tangent sheaf

$$\mathcal{T}_{\mathrm{M}^{\circ}_{\mathfrak{p}}(\mathrm{V}^{\circ},\mathrm{K}^{p\circ})/\mathrm{T}_{\mathfrak{p}}}\simeq\mathcal{H}om\left(\omega_{\mathcal{A}^{\vee},\tau_{\infty}^{c}},\mathrm{Lie}_{\mathcal{A},\tau_{\infty}^{c}}\right)$$

(4) When $N \geq 2$, the relative tangent sheaf $\mathcal{T}_{M^{\bullet}_{\mathfrak{p}}(V^{\circ}, K^{p\circ})/T_{\mathfrak{p}}}$ fits canonically into a sequence

$$0 \longrightarrow \mathcal{H}om\left(\omega_{\mathcal{A}^{\vee},\tau_{\infty}}, \omega_{\mathcal{A}^{\vee},\tau_{\infty}}^{\perp}/\omega_{\mathcal{A}^{\vee},\tau_{\infty}}\right) \longrightarrow \mathcal{T}_{\mathcal{M}_{\mathfrak{p}}^{\bullet}(\mathcal{V}^{\circ},\mathcal{K}^{p\circ})/\mathcal{T}_{\mathfrak{p}}} \longrightarrow \mathcal{H}om\left(\omega_{\mathcal{A}^{\vee},\tau_{\infty}^{\mathsf{c}}}/\mathcal{H}_{1}^{\mathrm{dR}}(\mathcal{A})_{\tau_{\infty}}^{\perp}, \mathrm{Lie}_{\mathcal{A},\tau_{\infty}^{\mathsf{c}}}\right) \longrightarrow 0.$$

(5) When $N \geq 2$, we have a canonical isomorphism for the relative tangent sheaf

$$\mathcal{T}_{\mathrm{M}^{\dagger}_{\mathfrak{p}}(\mathrm{V}^{\circ},\mathrm{K}^{p\circ})/\mathrm{T}_{\mathfrak{p}}} \simeq \mathcal{H}om\left(\omega_{\mathcal{A}^{\vee},\tau_{\infty}^{c}}/\mathrm{H}_{1}^{\mathrm{dR}}(\mathcal{A})_{\tau_{\infty}}^{\perp},\mathrm{Lie}_{\mathcal{A},\tau_{\infty}^{c}}\right)$$

Proof. For (1), the (quasi-)projectiveness part is well-known. We consider the remaining assertions. Take a point $x = (A_0, \lambda_0, \eta_0^p; A, \lambda, \eta^p) \in \mathbf{M}_{\mathfrak{p}}(\mathcal{V}^\circ, \mathcal{K}^{p\circ})(\kappa)$ for a perfect field κ containing \mathbb{F}_p^{Φ} , and denote by the completed local ring of $\mathbf{M}_{\mathfrak{p}}(\mathcal{V}^\circ, \mathcal{K}^{p\circ})$ at x by \mathcal{O}_x . We have a $W(\kappa)$ -bilinear pairing $\langle , \rangle_{\lambda,\tau_{\infty}} : \mathcal{D}(A)_{\tau_{\infty}} \times \mathcal{D}(A)_{\tau_{\infty}^c} \to W(\kappa)$ as in Notation 3.4.12. By repeatedly applying Proposition 3.4.8, we have for every commutative Artinian $W(\kappa)$ -algebra R that $\operatorname{Hom}_{W(\kappa)}(\mathcal{O}_x, R)$ is the set of R-subbundle

$$M_{\tau_{\infty}} \subseteq \mathcal{D}(A)_{\tau_{\infty}} \otimes_{W(\kappa)} R, \quad M_{\tau_{\infty}^{c}} \subseteq \mathcal{D}(A)_{\tau_{\infty}^{c}} \otimes_{W(\kappa)} R$$

of ranks 1 and N-1 lifting $\omega_{A^{\vee}/\kappa,\tau_{\infty}}$ and $\omega_{A^{\vee}/\kappa,\tau_{\infty}^{c}}$, respectively, such that $\langle M_{\tau_{\infty}}, M_{\tau_{\infty}^{c}} \rangle_{\lambda,\tau_{\infty}} = 0$. We choose isomorphisms $\mathcal{D}(A)_{\tau_{\infty}} \simeq W(\kappa)^{\oplus N}$ and $\mathcal{D}(A)_{\tau_{\infty}^{c}} \simeq W(\kappa)^{\oplus N}$ under which the pairing $\langle , \rangle_{\lambda,\tau_{\infty}}$ is given by

$$\langle (x_1,\ldots,x_N), (y_1,\ldots,y_N) \rangle_{\lambda,\tau_{\infty}} = px_1y_1 + x_2y_2 + \cdots + x_Ny_N.$$

There are four possible cases.

- (i) If $\omega_{A^{\vee}/\kappa,\tau_{\infty}}$ is generated by $(1,0,\ldots,0)$ and $\omega_{A^{\vee}/\kappa,\tau_{\infty}^{c}}$ contains $(1,0,\ldots,0)$, then possibly after changing coordinates, we may assume that $\omega_{A^{\vee}/\kappa,\tau_{\infty}^{c}} = \{(y_{1},\ldots,y_{N-1},0)\}$. Then we have $\mathcal{O}_{x} \simeq W(\kappa)[[x_{1},\ldots,x_{N-1},x_{N}]]/(x_{1}x_{N}-p)$.
- (ii) If $\omega_{A^{\vee}/\kappa,\tau_{\infty}}$ is generated by $(1,0,\ldots,0)$ and $\omega_{A^{\vee}/\kappa,\tau_{\infty}^{c}}$ does not contain $(1,0,\ldots,0)$, then possibly after changing coordinates, we may assume that $\omega_{A^{\vee}/\kappa,\tau_{\infty}^{c}} = \{(0,y_{2},\ldots,y_{N})\}$. It is clear that $M_{\tau_{\infty}}$ is determined by $M_{\tau_{\infty}^{c}}$; and $\mathcal{O}_{x} \simeq W(\kappa)[[x_{2},\ldots,x_{N}]]$.

⁶This is the stratum linking balloons to the ground.

- (iii) If $\omega_{A^{\vee}/\kappa,\tau_{\infty}}$ is not generated by $(1,0,\ldots,0)$ and $\omega_{A^{\vee}/\kappa,\tau_{\infty}}$ contains $(1,0,\ldots,0)$, then possibly after changing coordinates, we may assume that $\omega_{A^{\vee}/\kappa,\tau_{\infty}}$ is generated by $(0,\ldots,0,1)$. It is clear that $M_{\tau_{\infty}^{c}}$ is determined by $M_{\tau_{\infty}}$; and $\mathcal{O}_{x} \simeq W(\kappa)[[x_{1},\ldots,x_{N-1}]]$.
- (iv) If $\omega_{A^{\vee}/\kappa,\tau_{\infty}}$ is not generated by $(1, 0, \ldots, 0)$ and $\omega_{A^{\vee}/\kappa,\tau_{\infty}^{c}}$ does not contain $(1, 0, \ldots, 0)$, then this would not happen.

Together with the fact that $\mathbf{M}_{\mathfrak{p}}(\mathbf{V}^{\circ}, \mathbf{K}^{p\circ}) \otimes \mathbb{Q}$ is smooth of dimension N-1, $\mathbf{M}_{\mathfrak{p}}(\mathbf{V}^{\circ}, \mathbf{K}^{p\circ})$ is strictly semi-stable over $\mathbf{T}_{\mathfrak{p}}$ of relative dimension N-1. Moreover, $\mathbf{M}_{\mathfrak{p}}^{\circ}(\mathbf{V}^{\circ}, \mathbf{K}^{p\circ})$ is the locus where (i) or (ii) happens; and $\mathbf{M}_{\mathfrak{p}}^{\bullet}(\mathbf{V}^{\circ}, \mathbf{K}^{p\circ})$ is the locus where (i) or (iii) happens. Thus, both (1) and (2) follow.

For (3–5), we will use deformation theory. For common use, we consider a closed immersion $S \hookrightarrow \hat{S}$ in $\operatorname{Sch}_{/\mathrm{T}_p}$ defined by an ideal sheaf \mathcal{I} with $\mathcal{I}^2 = 0$. Take an S-point $(A_0, \lambda_0, \eta_0^p; A, \lambda, \eta^p)$ in various schemes we will consider. By Proposition 3.4.8, we need to lift $\omega_{A^{\vee}, \tau_{\infty}}$ and $\omega_{A^{\vee}, \tau_{\infty}^c}$ to subbundles $\hat{\omega}_{A^{\vee}, \tau_{\infty}} \subseteq \operatorname{H}_1^{\operatorname{cris}}(A/\hat{S})_{\tau_{\infty}}$ and $\hat{\omega}_{A^{\vee}, \tau_{\infty}^c} \subseteq \operatorname{H}_1^{\operatorname{cris}}(A/\hat{S})_{\tau_{\infty}^c}$, respectively, that are orthogonal to each other under the pairing (3.3).

For (3), since we require $\langle \hat{\omega}_{A^{\vee},\tau_{\infty}}, \mathrm{H}_{1}^{\mathrm{cris}}(A/\hat{S})_{\tau_{\infty}^{c}} \rangle_{\lambda,\tau_{\infty}}^{\mathrm{cris}} = 0$, it remains to lift $\hat{\omega}_{A^{\vee},\tau_{\infty}^{c}}$ without restriction. Thus, (3) follows by Remark 3.4.6.

For (4), we need to first find lifting $\hat{\omega}_{A^{\vee},\tau_{\infty}^{c}}$ that contains $\mathrm{H}_{1}^{\mathrm{cris}}(A/\hat{S})_{\tau_{\infty}}^{\perp}$; and then find lifting $\hat{\omega}_{A^{\vee},\tau_{\infty}}$ satisfying $\langle \hat{\omega}_{A^{\vee},\tau_{\infty}}, \hat{\omega}_{A^{\vee},\tau_{\infty}^{c}} \rangle_{\lambda,\tau_{\infty}}^{\mathrm{cris}} = 0$. Thus, (4) follows by Remark 3.4.6.

For (5), we only need to find lifting $\hat{\omega}_{A^{\vee},\tau_{\infty}^{c}}$ that contains $\mathrm{H}_{1}^{\mathrm{cris}}(A/\hat{S})_{\tau_{\infty}}^{\perp}$, which implies (5). \Box

In the remaining part of this subsection, we discuss the relation between $\mathbf{M}_{\mathfrak{p}}(V^{\circ}, -)$ and unitary Shimura varieties. Since we use a standard definite hermitian space to parameterize the level structures, such relation is not canonical, which depends on the choice of an indefinite uniformization datum defined as follows.

Definition 5.1.6. We define an *indefinite uniformization datum for* V[°] (at \mathfrak{p}) to be a collection of $(V', \mathbf{j}, \{\Lambda'_{\mathfrak{q}}\}_{\mathfrak{q}|p})$ where

- O V' is a standard indefinite hermitian space over F of rank N;
- $O \ j \colon V^{\circ} \otimes_{\mathbb{Q}} \mathbb{A}^{\infty,p} \to V' \otimes_{\mathbb{Q}} \mathbb{A}^{\infty,p} \text{ is an isometry};$
- Ο for every prime **q** of F^+ above *p* other than **p**, $\Lambda'_{\mathfrak{q}}$ is a self-dual $O_{F_{\mathfrak{q}}}$ -lattice in V' ⊗_F $F_{\mathfrak{q}}$; and Ο $\Lambda'_{\mathfrak{p}}$ is an $O_{F_{\mathfrak{p}}}$ -lattice in V' ⊗_F $F_{\mathfrak{p}}$ satisfying $\Lambda'_{\mathfrak{p}} \subseteq (\Lambda'_{\mathfrak{p}})^{\vee}$ and $(\Lambda'_{\mathfrak{p}})^{\vee}/\Lambda'_{\mathfrak{p}}$ has length 1.

By the Hasse principle for hermitian spaces, there exists an indefinite uniformization datum for which we fix one. Let $K'_{\mathfrak{q}}$ be the stabilizer of $\Lambda'_{\mathfrak{q}}$ for every \mathfrak{q} over p; and put $K'_p := \prod_{\mathfrak{q}|p} K'_{\mathfrak{q}}$. The isometry \mathfrak{j} induces an equivalence of categories $\mathfrak{j}: \mathfrak{K}(V^\circ)^p \xrightarrow{\sim} \mathfrak{K}(V')^p$.

Then similar to Remark 4.1.5, we obtain a "moduli interpretation" isomorphism

(5.2)
$$\mathbf{M}_{\mathfrak{p}}^{\eta}(\mathbf{V}^{\circ},-) \xrightarrow{\sim} \operatorname{Sh}(\mathbf{V}',\mathfrak{j}-\mathbf{K}'_{p}) \times_{\operatorname{Spec} F} \mathbf{T}_{\mathfrak{p}}^{\eta}$$

in $\operatorname{\mathsf{Fun}}(\mathfrak{K}(\mathcal{V}^{\circ})^{p} \times \mathfrak{T}, \operatorname{\mathsf{Sch}}_{/\mathbb{Q}_{p}^{\Phi}})_{/\mathbb{T}_{\mathfrak{p}}^{\eta}}$, where \mathfrak{T} acts on $\operatorname{Sh}(\mathcal{V}', \mathfrak{j}-\mathcal{K}'_{p}) \times_{\operatorname{Spec} F} \mathbb{T}_{\mathfrak{p}}^{\eta}$ via the second factor.

Lemma 5.1.7. Let L be a p-coprime coefficient ring. The two specialization maps

$$\begin{aligned} & \mathrm{H}^{i}_{\mathfrak{T},c}(\mathbf{M}_{\mathfrak{p}}(\mathrm{V}^{\circ},-)\otimes_{\mathbb{Z}_{p}^{\Phi}}\overline{\mathbb{Q}}_{p},L)\to\mathrm{H}^{i}_{\mathfrak{T},c}(\overline{\mathrm{M}}_{\mathfrak{p}}(\mathrm{V}^{\circ},-),\mathrm{R}\Psi L), \\ & \mathrm{H}^{i}_{\mathfrak{T}}(\mathbf{M}_{\mathfrak{p}}(\mathrm{V}^{\circ},-)\otimes_{\mathbb{Z}_{p}^{\Phi}}\overline{\mathbb{Q}}_{p},L)\to\mathrm{H}^{i}_{\mathfrak{T}}(\overline{\mathrm{M}}_{\mathfrak{p}}(\mathrm{V}^{\circ},-),\mathrm{R}\Psi L), \end{aligned}$$

are both isomorphisms. In particular, (5.2) induces isomorphisms

$$\begin{split} \mathrm{H}^{i}_{\mathrm{\acute{e}t},c}(\mathrm{Sh}(\mathrm{V}', \mathbf{j}-\mathrm{K}'_{p})_{\overline{F}}, L) &\simeq \mathrm{H}^{i}_{\mathfrak{T},c}(\overline{\mathrm{M}}_{\mathfrak{p}}(\mathrm{V}^{\circ}, -), \mathrm{R}\Psi L), \\ \mathrm{H}^{i}_{\mathrm{\acute{e}t}}(\mathrm{Sh}(\mathrm{V}', \mathbf{j}-\mathrm{K}'_{p})_{\overline{F}}, L) &\simeq \mathrm{H}^{i}_{\mathfrak{T}}(\overline{\mathrm{M}}_{\mathfrak{p}}(\mathrm{V}^{\circ}, -), \mathrm{R}\Psi L), \end{split}$$

in $\operatorname{Fun}(\mathfrak{K}(\mathcal{V}^{\circ})^{p}, \operatorname{\mathsf{Mod}}(L[\operatorname{Gal}(\overline{\mathbb{Q}}_{p}/\mathbb{Q}_{p}^{\Phi})]))$ for every $i \in \mathbb{Z}$. Here, $\operatorname{Gal}(\overline{\mathbb{Q}}_{p}/\mathbb{Q}_{p}^{\Phi})$ is regarded as a subgroup of $\operatorname{Gal}(\overline{F}/F)$ under our fixed isomorphism $\iota_{p} \colon \mathbb{C} \simeq \overline{\mathbb{Q}}_{p}$.

Proof. When $\mathbf{M}_{\mathfrak{p}}(V, -)$ is proper, this is simply the proper base change. When $\mathbf{M}_{\mathfrak{p}}(V, -)$ is not proper, this follows from [LS18, Corollary 5.20].

Remark 5.1.8. When $[F^+ : \mathbb{Q}] > 1$, the Shimura variety $\operatorname{Sh}(V', \operatorname{K}^{p'}\operatorname{K}'_p)$ is proper over F for $\operatorname{K}^{p'} \in \mathfrak{K}(V')^p$. We explain that $\operatorname{Sh}(V', \operatorname{K}^{p'}\operatorname{K}'_p)$ has proper smooth reduction at every place w of F above $\Sigma_p^+ \setminus \{\mathfrak{p}\}$.

Take a place w of F above $\Sigma_p^+ \setminus \{\mathfrak{p}\}$. Choose a CM type Φ containing τ_{∞} and an isomorphism $\mathbb{C} \simeq \overline{\mathbb{Q}}_p$ that induces w (not the unique place above \mathfrak{p} !). Put $\mathbf{T}_w \coloneqq \mathbf{T}_p(\mathbf{W}_0, \mathbf{K}_0^p) \otimes_{O_{F_\Phi} \otimes \mathbb{Z}_{(p)}} \mathbb{Z}_p^{\Phi}$. We define a functor $\mathbf{M}_w(\mathbf{V}', \mathbf{K}^{p'})$ on $\mathsf{Sch}'_{\mathbb{Z}_p^{\Phi}}$ such that for every $S \in \mathsf{Sch}'_{\mathbb{Z}_p^{\Phi}}$, $\mathbf{M}_w(\mathbf{V}', \mathbf{K}^{p'})(S)$ is the set of equivalence classes of sextuples $(A_0, \lambda_0, \eta_0^p; A, \lambda, \eta^p)$ where

- $O(A_0, \lambda_0, \eta_0^p)$ is an element in $\mathbf{T}_w(S)$;
- O (A, λ) is a unitary O_F -abelian scheme of signature type $N\Phi \tau_{\infty} + \tau_{\infty}^{c}$ over S (Definitions 3.4.2 and 3.4.3) such that ker $\lambda[p^{\infty}]$ is contained in $A[\mathfrak{p}]$ of rank p^2 ;
- O η^p is a K^p-level structure, similarly defined as in Definition 5.1.1.

Then $\mathbf{M}_w(\mathbf{V}', \mathbf{K}^{p'})$ is represented by a projective scheme over \mathbb{Z}_p^{Φ} . An easy computation of the tangent sheaf as in Theorem 4.1.3 shows that $\mathbf{M}_w(\mathbf{V}', \mathbf{K}^{p'})$ is smooth of relative dimension N-1. Moreover, we have a canonical isomorphism

$$\mathbf{M}_{w}^{\eta}(\mathbf{V}',\mathbf{K}^{p\prime})\simeq \mathrm{Sh}(\mathbf{V}',\mathbf{K}^{p\prime}\mathbf{K}_{p}')\times_{\mathrm{Spec}\,F}\mathbf{T}_{w}^{\eta}$$

over \mathbf{T}_w^{η} . Thus, $\mathrm{Sh}(\mathbf{V}', \mathbf{K}_p''\mathbf{K}_p')$ has proper smooth reduction at w as \mathbf{T}_w is finite étale over O_{F_w} .

5.2. Basic correspondence on balloon stratum. In this subsection, we construct and study the basic correspondence on the balloon stratum $M_{\mathfrak{p}}^{\circ}(V^{\circ}, -)$.

Definition 5.2.1. We define a functor

$$\begin{split} \mathbf{S}^{\circ}_{\mathfrak{p}}(\mathbf{V}^{\circ},-) \colon \mathfrak{K}(\mathbf{V}^{\circ})^{p} \times \mathfrak{T} &\to \mathsf{PSch}'_{/\mathbb{F}^{\Phi}_{p}} \\ \mathbf{K}^{p\circ} &\mapsto \mathbf{S}^{\circ}_{\mathfrak{p}}(\mathbf{V}^{\circ},\mathbf{K}^{p\circ}) \end{split}$$

such that for every $S \in \mathsf{Sch}'_{/\mathbb{F}_p^{\Phi}}$, $S^{\circ}_{\mathfrak{p}}(V^{\circ}, K^{p\circ})(S)$ is the set of equivalence classes of sextuples $(A_0, \lambda_0, \eta_0^p; A^{\circ}, \lambda^{\circ}, \eta^{p\circ})$ where

- O $(A_0, \lambda_0, \eta_0^p)$ is an element in $T_{\mathfrak{p}}(S)$;
- $O(A^{\circ}, \lambda^{\circ})$ is a unitary O_F -abelian scheme of signature type $N\Phi$ over S such that λ° is p-principal;
- $\bigcirc \eta^{p^{\circ}}$ is, for a chosen geometric point s on every connected component of S, a $\pi_1(S, s)$ invariant $K^{p^{\circ}}$ -orbit of isomorphisms

$$\eta^{p\circ} \colon \mathcal{V}^{\circ} \otimes_{\mathbb{Q}} \mathbb{A}^{\infty,p} \to \operatorname{Hom}_{F \otimes_{\mathbb{Q}} \mathbb{A}^{\infty,p}}^{\lambda_{0},\lambda^{\circ}}(\mathcal{H}_{1}^{\operatorname{\acute{e}t}}(A_{0s},\mathbb{A}^{\infty,p}),\mathcal{H}_{1}^{\operatorname{\acute{e}t}}(A_{s}^{\circ},\mathbb{A}^{\infty,p}))$$

of hermitian spaces over $F \otimes_{\mathbb{Q}} \mathbb{A}^{\infty,p} = F \otimes_{F^+} \mathbb{A}_{F^+}^{\infty,p}$.

The equivalence relation and the action of morphisms in $\mathfrak{K}(\mathcal{V}^{\circ})^{p} \times \mathfrak{T}$ are defined similarly as in Definition 4.1.2.

We have apparently the forgetful morphism

$$S^{\circ}_{\mathfrak{p}}(V^{\circ}, -) \to T_{\mathfrak{p}}$$

in $\operatorname{Fun}(\mathfrak{K}(V^{\circ})^{p} \times \mathfrak{T}, \operatorname{\mathsf{PSch}}'_{\mathbb{F}_{p}^{\Phi}})$ which is represented by finite and étale schemes.

Now we take a point $s^{\circ} = (A_0, \lambda_0, \eta_0^p; A^{\circ}, \lambda^{\circ}, \eta^{p_{\circ}}) \in S^{\circ}_{\mathfrak{p}}(V^{\circ}, K^{p_{\circ}})(\kappa)$ where κ is a perfect field containing \mathbb{F}_p^{Φ} . By Remark 3.4.10, the (κ, σ^{-1}) -linear Verschiebung map

$$\mathbb{V} \colon \mathrm{H}_{1}^{\mathrm{dR}}(A^{\circ}/\kappa)_{\tau_{\infty}} \to \mathrm{H}_{1}^{\mathrm{dR}}(A^{\circ}/\kappa)_{\sigma^{-1}\tau_{\infty}} = \mathrm{H}_{1}^{\mathrm{dR}}(A^{\circ}/\kappa)_{\tau_{\infty}^{\circ}}$$

is an isomorphism. Thus, we obtain a (κ, σ) -linear isomorphism

$$\mathbf{V}^{-1} \colon \mathrm{H}^{\mathrm{dR}}_{1}(A^{\circ}/\kappa)_{\tau_{\infty}^{\mathsf{c}}} \to \mathrm{H}^{\mathrm{dR}}_{1}(A^{\circ}/\kappa)_{\tau_{\infty}}$$

We define a non-degenerate pairing

$$\{ , \}_{s^{\circ}} \colon \mathrm{H}_{1}^{\mathrm{dR}}(A^{\circ}/\kappa)_{\tau_{\infty}^{c}} \times \mathrm{H}_{1}^{\mathrm{dR}}(A^{\circ}/\kappa)_{\tau_{\infty}^{c}} \to \kappa$$

by the formula $\{x, y\}_{s^{\circ}} \coloneqq \langle \mathbb{V}^{-1}x, y \rangle_{\lambda^{\circ}, \tau_{\infty}}$ (Notation 3.4.7). To ease notation, we put

$$\mathscr{V}_{s^{\circ}} \coloneqq \mathrm{H}_{1}^{\mathrm{dR}}(A^{\circ}/\kappa)_{\tau_{\infty}^{\mathsf{c}}}.$$

By the same proof of Lemma 4.2.2, we know that $(\mathscr{V}_{s^{\circ}}, \{,\}_{s^{\circ}})$ is admissible. Thus, we have the Deligne-Lusztig variety $DL_{s^{\circ}} := DL(\mathscr{V}_{s^{\circ}}, \{,\}_{s^{\circ}}, N-1)$ (Definition A.1.2).

Definition 5.2.2. We define a functor

$$\begin{split} \mathbf{B}^{\circ}_{\mathfrak{p}}(\mathbf{V}^{\circ},-) \colon \mathfrak{K}(\mathbf{V}^{\circ})^{p} \times \mathfrak{T} &\to \mathsf{PSch}'_{/\mathbb{F}^{\Phi}_{p}} \\ \mathbf{K}^{p\circ} &\mapsto \mathbf{B}^{\circ}_{\mathfrak{p}}(\mathbf{V}^{\circ},\mathbf{K}^{p\circ}) \end{split}$$

such that for every $S \in \mathsf{Sch}'_{/\mathbb{F}_p^{\Phi}}$, $\mathrm{B}^{\circ}_{\mathfrak{p}}(\mathrm{V}^{\circ}, \mathrm{K}^{p\circ})(S)$ is the set of equivalence classes of decuples $(A_0, \lambda_0, \eta_0^p; A, \lambda, \eta^p; A^{\circ}, \lambda^{\circ}, \eta^{p\circ}; \beta)$ where

- O $(A_0, \lambda_0, \eta_0^p; A, \lambda, \eta^p)$ is an element of $\mathrm{M}^{\circ}_{\mathfrak{p}}(\mathrm{V}^{\circ}, \mathrm{K}^{p\circ})(S)$;
- O $(A_0, \lambda_0, \eta_0^p; A^\circ, \lambda^\circ, \eta^{p\circ})$ is an element of $S^{\circ}_{\mathfrak{p}}(V^\circ, K^{p\circ})(S)$;
- O $\beta: A \to A^{\circ}$ is an O_F -linear quasi-*p*-isogeny (Definition 3.4.5) such that
 - (a) ker $\beta[p^{\infty}]$ is contained in $A[\mathfrak{p}]$;
 - (b) we have $\lambda = \beta^{\vee} \circ \lambda^{\circ} \circ \beta$; and
 - (c) the K^{*p*°}-orbit of maps $v \mapsto \beta_* \circ \eta^p(v)$ for $v \in V^\circ \otimes_{\mathbb{Q}} \mathbb{A}^{\infty,p}$ coincides with η^{p° .

The equivalence relation and the action of morphisms in $\mathfrak{K}(\mathcal{V}^{\circ})^{p} \times \mathfrak{T}$ are defined similarly as in Definition 4.2.3.

We obtain in the obvious way a correspondence

(5.3)
$$S^{\circ}_{\mathfrak{p}}(V^{\circ},-) \stackrel{\pi^{\circ}}{\longleftarrow} B^{\circ}_{\mathfrak{p}}(V^{\circ},-) \stackrel{\iota^{\circ}}{\longrightarrow} M^{\circ}_{\mathfrak{p}}(V^{\circ},-)$$

in $\mathsf{Fun}(\mathfrak{K}(\mathcal{V}^{\circ})^{p} \times \mathfrak{T}, \mathsf{PSch}'_{/\mathbb{F}_{p}^{\Phi}})_{/\mathrm{T}_{\mathfrak{p}}}.$

Definition 5.2.3 (Basic correspondence). We refer to (5.3) as the *basic correspondence* on the balloon stratum $M^{\circ}_{\mathfrak{p}}(V^{\circ}, -)$, with $S^{\circ}_{\mathfrak{p}}(V^{\circ}, -)$ being the *source* of the basic correspondence.

Theorem 5.2.4. In the diagram (5.3), ι° is an isomorphism. Moreover, for every point $s^{\circ} = (A_0, \lambda_0, \eta_0^p; A^{\circ}, \lambda^{\circ}, \eta^{p \circ}) \in S_{\mathfrak{p}}^{\circ}(V^{\circ}, K^{p \circ})(\kappa)$ where κ is a perfect field containing \mathbb{F}_p^{Φ} , if we put $B_{s^{\circ}}^{\circ} := \pi^{\circ - 1}(s^{\circ})$, then the assignment sending $(A_0, \lambda_0, \eta_0^p; A, \lambda, \eta^p; A^{\circ}, \lambda^{\circ}, \eta^{p \circ}; \beta) \in B_{s^{\circ}}^{\circ}(S)$ to the subbundle

$$H \coloneqq \beta_{*,\tau_{\infty}^{\mathsf{c}}} \omega_{A^{\vee}/S,\tau_{\infty}^{\mathsf{c}}} \subseteq \mathrm{H}_{1}^{\mathrm{dR}}(A^{\circ}/S)_{\tau_{\infty}^{\mathsf{c}}} = \mathrm{H}_{1}^{\mathrm{dR}}(A^{\circ}/\kappa)_{\tau_{\infty}^{\mathsf{c}}} \otimes_{\kappa} \mathcal{O}_{S} = (\mathscr{V}_{s^{\circ}})_{S}$$

induces an isomorphism $\zeta_{s^{\circ}}^{\circ} \colon \mathbf{B}_{s^{\circ}}^{\circ} \xrightarrow{\sim} \mathbb{P}(\mathscr{V}_{s^{\circ}})$ satisfying that

(1) $\zeta_{s^{\circ}}^{\circ}$ restricts to an isomorphism

$$\zeta^{\circ}_{s^{\circ}} \colon \mathrm{B}^{\circ}_{s^{\circ}} \bigcap \iota^{\circ -1} \mathrm{M}^{\dagger}_{\mathfrak{p}}(\mathrm{V}^{\circ}, \mathrm{K}^{p^{\circ}}) \xrightarrow{\sim} \mathrm{DL}_{s^{\circ}} = \mathrm{DL}(\mathscr{V}_{s^{\circ}}, \{ , \}_{s^{\circ}}, N-1);$$

(2) we have an isomorphism

$$\mathcal{H}om\left(\omega_{\mathcal{A}^{\vee},\tau_{\infty}},\omega_{\mathcal{A}^{\vee},\tau_{\infty}^{c}}^{\perp}/\omega_{\mathcal{A}^{\vee},\tau_{\infty}}\right)\simeq(\zeta_{s^{\circ}}^{\circ})^{*}\mathcal{O}_{\mathbb{P}(\mathscr{V}_{s^{\circ}})}(-(p+1)).$$

In particular, $B_{s^{\circ}}^{\circ} \cap \iota^{\circ -1} M_{\mathfrak{p}}^{\dagger}(V^{\circ}, K^{p^{\circ}})$ is a Fermat hypersurface in $B_{s^{\circ}}^{\circ} \simeq \mathbb{P}(\mathscr{V}_{s^{\circ}})$.

Proof. Take an object $K^{p\circ} \in \mathfrak{K}(V^{\circ})^{p}$. It is clear that $B^{\circ}_{\mathfrak{p}}(V^{\circ}, -)$ is a scheme. We denote by $(\mathcal{A}_{0}, \lambda_{0}, \eta^{p}_{0}; \mathcal{A}, \lambda, \eta^{p}; \mathcal{A}^{\circ}, \lambda^{\circ}, \eta^{p\circ}; \beta)$ the universal object over $B^{\circ}_{\mathfrak{p}}(V^{\circ}, K^{p\circ})$.

First, we show that ι° is an isomorphism. It is an easy exercise from Grothendieck–Messing theory that the canonical map $\mathcal{T}_{B^{\circ}_{\mathfrak{p}}(V^{\circ},K^{p\circ})/T_{\mathfrak{p}}} \to \iota^{\circ*}\mathcal{T}_{M^{\circ}_{\mathfrak{p}}(V^{\circ},K^{p\circ})/T_{\mathfrak{p}}}$ is an isomorphism. Thus, it suffices to show that $\iota^{\circ}(\kappa')$ is a bijection for every algebraically closed field κ' containing κ . To ease notation, we may assume $\kappa' = \kappa$. We construct an inverse of $\iota^{\circ}(\kappa)$. Take a point $(A_0, \lambda_0, \eta_0^p; A, \lambda, \eta^p) \in \mathrm{M}^{\circ}_{\mathfrak{p}}(\mathrm{V}^{\circ}, \mathrm{K}^{p\circ})(\kappa).$ Write $\tilde{\omega}_{A^{\vee}, \tau_{\infty}}$ the preimage of $\omega_{A^{\vee}, \tau_{\infty}}$ under the reduction map $\mathcal{D}(A)_{\tau_{\infty}} \to \mathrm{H}_{1}^{\mathrm{dR}}(A/\kappa)_{\tau_{\infty}}$. As $\langle \omega_{A^{\vee},\tau_{\infty}}, \mathrm{H}_{1}^{\mathrm{dR}}(A/\kappa)_{\tau_{\infty}^{\mathsf{c}}} \rangle_{\lambda,\tau_{\infty}} = 0$, we have $\mathcal{D}(A)_{\tau_{\infty}^{\mathsf{c}}}^{\vee} = p^{-1}\tilde{\omega}_{A^{\vee},\tau_{\infty}}$. Now we put $\mathcal{D}_{A^{\circ},\tau} \coloneqq \mathcal{D}(A)_{\tau}$ for $\tau \neq \tau_{\infty}$, and $\mathcal{D}_{A^{\circ},\tau_{\infty}} \coloneqq p^{-1}\tilde{\omega}_{A^{\vee},\tau_{\infty}}$. We claim that $\mathcal{D}_{A^{\circ}} \coloneqq$ $\bigoplus_{\tau \in \Sigma_{\infty}} \mathcal{D}_{A^{\circ},\tau}$ is a Dieudonné module, which amounts to the inclusions $\mathcal{F}\mathcal{D}_{A^{\circ},\tau_{\infty}} \subseteq \mathcal{D}_{A^{\circ},\tau_{\infty}}$ and $\mathbb{VD}_{A^{\circ},\tau_{\infty}} \subseteq \mathcal{D}_{A^{\circ},\tau_{\infty}^{\circ}}$. The first one is obvious; and the second one is equivalent to the first one as $\mathcal{D}_{A^{\circ},\tau_{\infty}}$ and $\mathcal{D}_{A^{\circ},\tau_{\infty}^{\circ}}$ are integrally dual under $\langle , \rangle_{\lambda,\tau_{\infty}}^{cris}$. Then by the Dieudonné theory, there is an O_F -abelian scheme A° over κ with $\mathcal{D}(A^{\circ})_{\tau} = \mathcal{D}_{A^{\circ},\tau}$ for every $\tau \in \Sigma_{\infty}$, and an O_F -linear isogeny $\beta: A \to A^{\circ}$ inducing the inclusion of Dieudonné modules $\mathcal{D}(A) \subset \mathcal{D}(A^{\circ})$. By Lemma 3.4.13(2,4), the O_F -abelian scheme A° has signature type $N\Phi$. Let λ° be the unique quasipolarization of A° satisfying $\lambda = \beta^{\vee} \circ \lambda^{\circ} \circ \beta$, which is *p*-principal as $\mathcal{D}_{A^{\circ},\tau_{\infty}^{\circ}} = \mathcal{D}_{A^{\circ},\tau_{\infty}}^{\vee}$. Finally, we let $\eta^{p\circ}$ be the map sending $v \in V^{\circ} \otimes_{\mathbb{Q}} \mathbb{A}^{\infty,p}$ to $\beta_* \circ \eta^p(v)$. Thus, we obtain an object $(A_0, \lambda_0, \eta_0^p; A, \lambda, \eta^p; A^{\circ}, \lambda^{\circ}, \eta^{p^{\circ}}; \beta) \in S^{\circ}_{\mathfrak{p}}(V^{\circ}, K^{p^{\circ}})(\kappa)$. It is straightforward to check that such assignment gives rise to an inverse of $\iota^{\circ}(\kappa)$.

Second, we show that $\zeta_{s^{\circ}}^{\circ}$ is well-defined, namely, H is a subbundle of rank N-1. By Lemma 3.4.13(2,4) and Definition 5.2.2(b), we have rank_{\mathcal{O}_S}(ker $\beta_{*,\tau_{\infty}}$) – rank_{\mathcal{O}_S}(ker $\beta_{*,\tau_{\infty}^{\circ}}$) = 1 and rank_{\mathcal{O}_S}(ker $\beta_{*,\tau_{\infty}^{\circ}}$) + rank_{\mathcal{O}_S}(ker $\beta_{*,\tau_{\infty}^{\circ}}$) = 1. Thus, $\beta_{*,\tau_{\infty}^{\circ}}$ is an isomorphism, hence H is a subbundle of rank N-1.

Third, we show that $\zeta_{s^{\circ}}^{\circ}$ is an isomorphism. Denote by $\mathcal{H} \subseteq (\mathscr{V}_{s^{\circ}})_{\mathbb{P}(\mathscr{V}_{s^{\circ}})}$ the universal subbundle (of rank N-1). Then we have a canonical isomorphism

$$\mathcal{T}_{\mathbb{P}(\mathscr{V}_{s^{\circ}})/\kappa} \simeq \operatorname{Hom}_{\mathcal{O}_{\mathbb{P}(\mathscr{V}_{s^{\circ}})}} \left(\mathcal{H}, \operatorname{H}_{1}^{\mathrm{dR}}(A^{\circ}/\kappa)_{\tau_{\infty}^{\mathsf{c}}}/\mathcal{H}\right).$$

By Theorem 5.1.5(1) and the fact that $\beta_{*,\tau_{\infty}^{c}}$ is an isomorphism, we obtain an isomorphism

$$\left(\iota^{\circ*}\mathcal{T}_{\mathrm{M}^{\circ}_{\mathfrak{p}}(\mathrm{V}^{\circ},\mathrm{K}^{p\circ})/\mathrm{T}_{\mathfrak{p}}}\right)|_{\mathrm{B}^{\circ}_{s^{\circ}}}\xrightarrow{\sim}\zeta^{\circ*}_{s^{\circ}}\mathcal{T}_{\mathbb{P}}(\mathscr{V}_{s^{\circ}})/\kappa.$$

Thus, to show that $\zeta_{s^\circ}^\circ : \mathbb{B}_{s^\circ}^\circ \to \mathbb{P}(\mathscr{V}_{s^\circ})$ is an isomorphism, it suffices to construct an inverse of $\zeta_{s^\circ}^\circ(\kappa')$ for every algebraically closed field κ' containing κ . To ease notation, we may assume $\kappa' = \kappa$. Take a κ -linear subspace $H \subseteq \mathscr{V}_{s^\circ} = \mathrm{H}_1^{\mathrm{dR}}(A^\circ)_{\tau_\infty^\circ}$ of rank N-1. Let \tilde{H} denote by its preimage under the reduction map $\mathcal{D}(A^\circ)_{\tau_\infty^\circ} \to \mathrm{H}_1^{\mathrm{dR}}(A^\circ)_{\tau_\infty^\circ}$. We put $\mathcal{D}_{A,\tau} \coloneqq \mathcal{D}(A^\circ)_{\tau}$ for $\tau \neq \tau_\infty$, and $\mathcal{D}_{A,\tau_\infty} \coloneqq \mathbb{V}^{-1}\tilde{H} \subseteq \mathcal{D}(A^\circ)_{\tau_\infty}$. It is clear that $\mathcal{D}_A \coloneqq \bigoplus_{\tau \in \Sigma_\infty} \mathcal{D}_{A,\tau}$ is a Dieudonné module. By the Dieudonné theory, there is an O_F -abelian scheme A over κ with $\mathcal{D}(A)_{\tau} = \mathcal{D}_{A,\tau}$ for every $\tau \in \Sigma_\infty$, and an O_F -linear isogeny $\beta \colon A \to A^\circ$ inducing the inclusion of Dieudonné modules $\mathcal{D}(A) \subseteq \mathcal{D}(A^\circ)$. By a similar argument as for ι° , we obtain a point $(A, \lambda, \eta^p; \beta) \in \mathrm{B}_{s^\circ}^\circ(\kappa)$; and it follows that such assignment is an inverse of $\zeta_{s^\circ}^\circ(\kappa)$.

Finally, we check the two properties of $\zeta_{s^{\circ}}^{\circ}$.

For (1), we check that the closed subscheme $\zeta_{s^{\circ}}^{\circ}(\mathbf{B}_{s^{\circ}}^{\circ} \cap \iota^{\circ-1}\mathbf{M}_{\mathfrak{p}}^{\dagger}(\mathbf{V}^{\circ},\mathbf{K}^{p^{\circ}}))$ coincides with $\mathrm{DL}(\mathscr{V}_{s^{\circ}}, \{, \}_{s^{\circ}}, N-1)$. Recall that $\mathrm{M}_{\mathfrak{p}}^{\dagger}(\mathbf{V}^{\circ}, \mathbf{K}^{p^{\circ}})$ is define by the condition

$$\mathrm{H}^{1}_{\mathrm{dR}}(A/S)^{\perp}_{\tau_{\infty}} \subseteq \omega_{A^{\vee}/S,\tau_{\infty}^{\mathsf{c}}}.$$

Note that we have $H = \beta_{*,\tau_{\infty}^{c}} \omega_{A^{\vee}/S,\tau_{\infty}^{c}}$ and $\mathbb{V}^{-1}H^{(p)} = \beta_{*,\tau_{\infty}} \mathbb{H}_{1}^{\mathrm{dR}}(A/S)_{\tau_{\infty}}$, which implies $(\mathbb{V}^{-1}H^{(p)})^{\perp} = (\beta_{*,\tau_{\infty}} \mathbb{H}_{1}^{\mathrm{dR}}(A/S)_{\tau_{\infty}})^{\perp} = \beta_{*,\tau_{\infty}^{c}} (\mathbb{H}_{\mathrm{dR}}^{1}(A/S)_{\tau_{\infty}})^{\perp}$. Applying the isomorphism $\beta_{*,\tau_{\infty}^{c}}$, the above condition is equivalent to

$$(\mathbf{V}^{-1}H^{(p)})^{\perp} \subseteq H,$$

which is the condition defining $DL(\mathscr{V}_{s^{\circ}}, \{, \}_{s^{\circ}}, N-1)$.

For (2), we have

$$\omega_{A^{\vee},\tau_{\infty}} = \ker \beta_{*,\tau_{\infty}} \simeq \mathrm{H}_{1}^{\mathrm{dR}} (A^{\circ}/S)_{\tau_{\infty}} / \beta_{*,\tau_{\infty}} \mathrm{H}_{1}^{\mathrm{dR}} (A/S)_{\tau_{\infty}} = \mathrm{H}_{1}^{\mathrm{dR}} (A^{\circ}/S)_{\tau_{\infty}} / \mathrm{V}^{-1} H^{(p)}$$

and

$$\omega_{A^{\vee},\tau_{\infty}^{\mathsf{c}}}^{\perp}/\omega_{A^{\vee},\tau_{\infty}}\simeq\beta_{*,\tau_{\infty}}\omega_{A^{\vee},\tau_{\infty}^{\mathsf{c}}}^{\perp}=(\beta_{*,\tau_{\infty}^{\mathsf{c}}}\omega_{A^{\vee}/S,\tau_{\infty}^{\mathsf{c}}})^{\perp}=H^{\perp}.$$

Thus, we have

$$\omega_{\mathcal{A}^{\vee},\tau_{\infty}} \simeq \zeta_{s^{\circ}}^{\circ*} \mathcal{O}_{\mathbb{P}(\mathscr{V}_{s^{\circ}})}(p), \quad \omega_{\mathcal{A}^{\vee},\tau_{\infty}^{c}}^{\perp} / \omega_{\mathcal{A}^{\vee},\tau_{\infty}} \simeq \zeta_{s^{\circ}}^{\circ*} \mathcal{O}_{\mathbb{P}(\mathscr{V}_{s^{\circ}})}(-1)$$

from which (2) follows.

The theorem is all proved.

Corollary 5.2.5. When $N \ge 2$, the normal bundle of the closed immersion

1

$$\mathrm{m}^{\dagger \bullet} \colon \mathrm{M}^{\dagger}_{\mathfrak{p}}(\mathrm{V}^{\circ}, \mathrm{K}^{p\circ}) \to \mathrm{M}^{\bullet}_{\mathfrak{p}}(\mathrm{V}^{\circ}, \mathrm{K}^{p\circ})$$

is isomorphic to $(\mathbf{m}^{\dagger\circ})^*\mathcal{O}_{\mathbf{M}^{\circ}_{\mathfrak{p}}(\mathbf{V}^{\circ},\mathbf{K}^{p\circ})}(-(p+1)).$

Proof. By Theorem 5.1.5(4,5), we have that the normal bundle is isomorphic to

$$\mathcal{H}om\left(\omega_{\mathcal{A}^{\vee},\tau_{\infty}},\omega_{\mathcal{A}^{\vee},\tau_{\infty}^{\mathsf{c}}}^{\perp}/\omega_{\mathcal{A}^{\vee},\tau_{\infty}^{\mathsf{c}}}\right).$$

Thus, the claim follows from Theorem 5.2.4. We can also argue that the normal bundle of $m^{\dagger \bullet}$ is dual to the normal bundle of $m^{\dagger \circ}$ which is isomorphic to $(m^{\dagger \circ})^* \mathcal{O}_{M^{\circ}_{\mathfrak{p}}(V^{\circ}, K^{p \circ})}(p+1)$ by Theorem 5.2.4.

Construction 5.2.6. Let $K_{\mathfrak{q}}^{\circ}$ be the stabilizer of $\Lambda_{\mathfrak{q}}^{\circ}$ for every $\mathfrak{q} \mid p$; and put $K_{p}^{\circ} \coloneqq \prod_{\mathfrak{q} \mid p} K_{\mathfrak{q}}^{\circ}$. Similar to Construction 4.3.2, we may construct a *uniformization map*, *canonical* this time,

(5.4)
$$\upsilon^{\circ} \colon \mathrm{S}^{\circ}_{\mathfrak{p}}(\mathrm{V}^{\circ}, -)(\overline{\mathbb{F}}_{p}) \xrightarrow{\sim} \mathrm{Sh}(\mathrm{V}^{\circ}, -\mathrm{K}^{\circ}_{p}) \times \mathrm{T}_{\mathfrak{p}}(\overline{\mathbb{F}}_{p})$$

in $\operatorname{Fun}(\mathfrak{K}(V^{\circ})^{p} \times \mathfrak{T}, \operatorname{Set})_{/\operatorname{T}_{p}(\overline{\mathbb{F}}_{p})}$ which is an isomorphism, under which the induced action of $\operatorname{Gal}(\overline{\mathbb{F}}_{p}/\mathbb{F}_{p}^{\Phi})$ on the target is trivial on $\operatorname{Sh}(V^{\circ}, -\operatorname{K}_{p}^{\circ})$.

Moreover, similar to Construction 4.3.5 and Proposition 4.3.6, for every $g \in \mathrm{K}^{\circ}_{\mathfrak{p}} \setminus \mathrm{U}(\mathrm{V}^{\circ})(F^+_{\mathfrak{p}})/\mathrm{K}^{\circ}_{\mathfrak{p}}$, we may construct the Hecke correspondence

$$\mathrm{Hk}_g\colon \mathrm{S}^{\circ}_{\mathfrak{p}}(\mathrm{V}^{\circ},-)_g\to \mathrm{S}^{\circ}_{\mathfrak{p}}(\mathrm{V}^{\circ},-)\times \mathrm{S}^{\circ}_{\mathfrak{p}}(\mathrm{V}^{\circ},-)$$

as a morphism in $\operatorname{Fun}(\mathfrak{K}(V^{\circ})^{p} \times \mathfrak{T}, \operatorname{Sch}_{/\mathbb{F}_{p}^{\Phi}})_{/T_{\mathfrak{p}}}$ that is finite étale and compatible with the uniformization map.

5.3. Basic correspondence on ground stratum. In this subsection, we construct and study the basic correspondence on the ground stratum $M_{\mathfrak{p}}^{\bullet}(V^{\circ}, -)$. We assume $N \geq 2$.

Definition 5.3.1. We define a functor

$$\begin{split} \mathbf{S}^{\bullet}_{\mathfrak{p}}(\mathbf{V}^{\circ},-) \colon \mathfrak{K}(\mathbf{V}^{\circ})^{p} \times \mathfrak{T} &\to \mathsf{PSch}'_{/\mathbb{F}^{\Phi}_{p}} \\ \mathbf{K}^{p\circ} &\mapsto \mathbf{S}^{\bullet}_{\mathfrak{n}}(\mathbf{V}^{\circ},\mathbf{K}^{p\circ}) \end{split}$$

such that for every $S \in \mathsf{Sch}'_{/\mathbb{F}_p^{\Phi}}$, $S^{\bullet}_{\mathfrak{p}}(V^{\circ}, K^{p\circ})(S)$ is the set of equivalence classes of sextuples $(A_0, \lambda_0, \eta_0^p; A^{\bullet}, \lambda^{\bullet}, \eta^{p\bullet})$ where

O $(A_0, \lambda_0, \eta_0^p)$ is an element in $T_{\mathfrak{p}}(S)$;

- $O(A^{\bullet}, \lambda^{\bullet})$ is a unitary O_F -abelian scheme of signature type $N\Phi$ over S such that ker $\lambda^{\bullet}[p^{\infty}]$ is trivial (resp. contained in $A^{\bullet}[\mathfrak{p}]$ of rank p^2) if N is even (resp. odd);
- $\bigcirc \eta^{p\bullet}$ is, for a chosen geometric point s on every connected component of S, a $\pi_1(S, s)$ invariant $K^{p\bullet}$ -orbit of isomorphisms

$$\eta^{p\bullet} \colon \mathcal{V}^{\circ} \otimes_{\mathbb{Q}} \mathbb{A}^{\infty,p} \to \mathrm{Hom}_{F \otimes_{\mathbb{Q}} \mathbb{A}^{\infty,p}}^{\varpi\lambda_{0},\lambda^{\bullet}}(\mathcal{H}_{1}^{\mathrm{\acute{e}t}}(A_{0s},\mathbb{A}^{\infty,p}),\mathcal{H}_{1}^{\mathrm{\acute{e}t}}(A_{s}^{\bullet},\mathbb{A}^{\infty,p}))$$

of hermitian spaces over $F \otimes_{\mathbb{Q}} \mathbb{A}^{\infty,p} = F \otimes_{F^+} \mathbb{A}_{F^+}^{\infty,p}$.

The equivalence relation and the action of morphisms in $\mathfrak{K}(\mathcal{V}^{\circ})^{p} \times \mathfrak{T}$ are defined similarly as in Definition 4.1.2.

We have apparently the forgetful morphism

$$S^{\bullet}_{\mathfrak{p}}(V^{\circ}, -) \to T_{\mathfrak{p}}$$

in $\operatorname{Fun}(\mathfrak{K}(V^{\circ})^{p} \times \mathfrak{T}, \operatorname{\mathsf{PSch}}'_{\mathbb{F}^{\Phi}})$ which is represented by finite and étale schemes.

Now we take a point $s^{\bullet} = (A_0, \lambda_0, \eta_0^p; A^{\bullet}, \lambda^{\bullet}, \eta^{p^{\bullet}}) \in S^{\bullet}_{\mathfrak{p}}(V^{\circ}, K^{p^{\circ}})(\kappa)$ where κ is a perfect field containing \mathbb{F}_p^{Φ} . By Remark 3.4.10, the (κ, σ^{-1}) -linear Verschiebung map

$$\mathbb{V}\colon \mathrm{H}_{1}^{\mathrm{dR}}(A^{\bullet}/\kappa)_{\tau_{\infty}} \to \mathrm{H}_{1}^{\mathrm{dR}}(A^{\bullet}/\kappa)_{\sigma^{-1}\tau_{\infty}} = \mathrm{H}_{1}^{\mathrm{dR}}(A^{\bullet}/\kappa)_{\tau_{\infty}^{\mathsf{c}}},$$

is an isomorphism. Thus, we obtain a (κ, σ) -linear isomorphism

$$\mathbb{V}^{-1} \colon \mathrm{H}_{1}^{\mathrm{dR}}(A^{\bullet}/\kappa)_{\tau_{\infty}^{\mathsf{c}}} \to \mathrm{H}_{1}^{\mathrm{dR}}(A^{\bullet}/\kappa)_{\tau_{\infty}}$$

We define a pairing

$$\{ , \}_{s^{\bullet}} \colon \mathrm{H}_{1}^{\mathrm{dR}}(A^{\bullet}/\kappa)_{\tau_{\infty}^{\mathsf{c}}} \times \mathrm{H}_{1}^{\mathrm{dR}}(A^{\bullet}/\kappa)_{\tau_{\infty}^{\mathsf{c}}} \to \kappa$$

by the formula $\{x, y\}_{s^{\bullet}} := \langle V^{-1}x, y \rangle_{\lambda^{\bullet}, \tau_{\infty}}$ (Notation 3.4.7). To ease notation, we put

$$\mathscr{V}_{s^{\bullet}} \coloneqq \mathrm{H}_{1}^{\mathrm{dR}}(A^{\bullet}/\kappa)_{\tau_{\infty}^{\mathsf{c}}}$$

By the same proof of Lemma 4.2.2, we know that $(\mathscr{V}_{s^{\bullet}}, \{ , \}_{s^{\bullet}})$ is admissible. Thus, we have the Deligne–Lusztig variety $\mathrm{DL}_{s^{\bullet}}^{\bullet} := \mathrm{DL}^{\bullet}(\mathscr{V}_{s^{\bullet}}, \{ , \}_{s^{\bullet}})$ (Definition A.2.1). Moreover, $\dim_{\kappa} \mathscr{V}_{s^{\bullet}}^{\perp}$ is equal to 0 (resp. 1) when N is even (resp. odd).

Definition 5.3.2. We define a functor

$$B^{\bullet}_{\mathfrak{p}}(V^{\circ}, -) \colon \mathfrak{K}(V^{\circ})^{p} \times \mathfrak{T} \to \mathsf{PSch}'_{/\mathbb{F}_{p}^{\Phi}}$$
$$K^{p \circ} \mapsto B^{\bullet}_{\mathfrak{n}}(V^{\circ}, K^{p \circ})$$

such that for every $S \in \mathsf{Sch}'_{/\mathbb{F}_p^{\Phi}}$, $B^{\bullet}_{\mathfrak{p}}(\mathcal{V}^{\circ}, \mathcal{K}^{p\circ})(S)$ is the set of equivalence classes of decuples $(A_0, \lambda_0, \eta_0^p; A, \lambda, \eta^p; A^{\bullet}, \lambda^{\bullet}, \eta^{p\bullet}; \gamma)$ where

- O $(A_0, \lambda_0, \eta_0^p; A, \lambda, \eta^p)$ is an element of $\mathcal{M}^{\bullet}_{\mathfrak{p}}(\mathcal{V}^{\circ}, \mathcal{K}^{p\circ})(S);$
- O $(A_0, \lambda_0, \eta_0^p; A^{\bullet}, \lambda^{\bullet}, \eta^{p\bullet})$ is an element of $S^{\bullet}_{\mathfrak{p}}(V^{\circ}, K^{p\circ})(S);$
- $O \gamma: A \to A^{\bullet}$ is an O_F -linear quasi-*p*-isogeny (Definition 3.4.5) such that
 - (a) ker $\gamma[p^{\infty}]$ is contained in $A[\mathfrak{p}]$;
 - (b) $(\ker \gamma_{*,\tau_{\infty}})^{\perp}$ is contained in $\omega_{A^{\vee}/S,\tau_{\infty}^{c}}$;
 - (c) we have $\varpi \cdot \lambda = \gamma^{\vee} \circ \lambda^{\bullet} \circ \gamma$; and
 - (d) the K^{*p*°}-orbit of maps $v \mapsto \gamma_* \circ \eta^p(v)$ for $v \in V^\circ \otimes_{\mathbb{Q}} \mathbb{A}^{\infty,p}$ coincides with $\eta^{p\bullet}$.

The equivalence relation and the action of morphisms in $\mathfrak{K}(\mathcal{V}^{\circ})^{p} \times \mathfrak{T}$ are defined similarly as in Definition 4.2.3.

We obtain in the obvious way a correspondence

(5.5)
$$S^{\bullet}_{\mathfrak{p}}(V^{\circ}, -) \xleftarrow{\pi^{\bullet}} B^{\bullet}_{\mathfrak{p}}(V^{\circ}, -) \xrightarrow{\iota^{\bullet}} M^{\bullet}_{\mathfrak{p}}(V^{\circ}, -)$$

in $\operatorname{Fun}(\mathfrak{K}(V^{\circ})^{p} \times \mathfrak{T}, \operatorname{PSch}'_{/\mathbb{F}_{p}^{\Phi}})_{/\mathrm{T}_{\mathfrak{p}}}$.

Definition 5.3.3 (Basic correspondence). We refer to (5.5) as the *basic correspondence* on the ground stratum $M_{\mathfrak{p}}^{\bullet}(V^{\circ}, -)$, with $S_{\mathfrak{p}}^{\bullet}(V^{\circ}, -)$ being the *source* of the basic correspondence.

Theorem 5.3.4. In the diagram (5.5), take a point

$$s^{\bullet} = (A_0, \lambda_0, \eta_0^p; A^{\bullet}, \lambda^{\bullet}, \eta^{p \bullet}) \in \mathcal{S}^{\bullet}_{\mathfrak{p}}(\mathcal{V}^\circ, \mathcal{K}^{p \circ})(\kappa)$$

where κ is a perfect field containing \mathbb{F}_p^{Φ} . Put $\mathbf{B}_{s^{\bullet}}^{\bullet} \coloneqq \pi^{\bullet-1}(s^{\bullet})$, and denote by $(\mathcal{A}, \lambda, \eta^p; \gamma)$ the universal object over the fiber B_{s}^{\bullet} .

(1) The fiber $B^{\bullet}_{s^{\bullet}}$ is a smooth scheme over κ , whose tangent sheaf $\mathcal{T}_{B^{\bullet}_{s^{\bullet}}/\kappa}$ fits canonically into an exact sequence

$$0 \to \mathcal{H}om\left(\omega_{\mathcal{A}^{\vee},\tau_{\infty}}, \omega_{\mathcal{A}^{\vee},\tau_{\infty}}^{\perp}/\omega_{\mathcal{A}^{\vee},\tau_{\infty}}\right) \to \mathcal{T}_{\mathrm{B}^{\bullet}_{s^{\bullet}}/\kappa} \to \mathcal{H}om\left(\omega_{\mathcal{A}^{\vee},\tau_{\infty}^{\mathsf{c}}}/(\ker\gamma_{*,\tau_{\infty}})^{\perp}, \mathrm{Lie}_{\mathcal{A}^{\vee},\tau_{\infty}^{\mathsf{c}}}\right) \to 0.$$

(2) The restriction of ι_{κ}^{\bullet} to $B_{s^{\bullet}}^{\bullet}$ is locally on $B_{s^{\bullet}}^{\bullet}$ a closed immersion, with a canonical isomorphism for its normal sheaf

$$\mathcal{N}_{\iota_{\kappa}^{\bullet}|\mathcal{B}_{s^{\bullet}}^{\bullet}} \simeq \mathcal{H}om\left((\ker\gamma_{*,\tau_{\infty}})^{\perp}/\mathcal{H}_{1}^{\mathrm{dR}}(\mathcal{A})_{\tau_{\infty}}^{\perp}, \operatorname{Lie}_{\mathcal{A}^{\vee},\tau_{\infty}^{\mathsf{c}}}\right) \simeq (\operatorname{im}\gamma_{*,\tau_{\infty}}) \otimes_{\mathcal{O}_{\mathcal{B}_{s^{\bullet}}^{\bullet}}} \operatorname{Lie}_{\mathcal{A}^{\vee},\tau_{\infty}^{\mathsf{c}}}.$$

(3) The assignment sending $(A_0, \lambda_0, \eta_0^p; A, \lambda, \eta^p; A^{\bullet}, \lambda^{\bullet}, \eta^{p\bullet}; \gamma) \in B^{\bullet}_{s^{\bullet}}(S)$ to the subbundles

$$H_{1} \coloneqq ((\breve{\gamma}_{*,\tau_{\infty}})^{-1}\omega_{A^{\vee}/S,\tau_{\infty}})^{\perp} \subseteq \mathrm{H}_{1}^{\mathrm{dR}}(A^{\bullet}/S)_{\tau_{\infty}^{\mathsf{c}}} = \mathrm{H}_{1}^{\mathrm{dR}}(A^{\bullet}/\kappa)_{\tau_{\infty}^{\mathsf{c}}} \otimes_{\kappa} \mathcal{O}_{S} = (\mathscr{V}_{s^{\bullet}})_{S},$$
$$H_{2} \coloneqq \gamma_{*,\tau_{\infty}^{\mathsf{c}}}\omega_{A^{\vee}/S,\tau_{\infty}^{\mathsf{c}}} \subseteq \mathrm{H}_{1}^{\mathrm{dR}}(A^{\bullet}/S)_{\tau_{\infty}^{\mathsf{c}}} = \mathrm{H}_{1}^{\mathrm{dR}}(A^{\bullet}/\kappa)_{\tau_{\infty}^{\mathsf{c}}} \otimes_{\kappa} \mathcal{O}_{S} = (\mathscr{V}_{s^{\bullet}})_{S},$$

where $\check{\gamma}: A^{\bullet} \to A$ is the (unique) O_F -linear quasi-p-isogeny such that $\check{\gamma} \circ \gamma = \varpi \cdot \mathrm{id}_A$, induces an isomorphism

$$\zeta_{s^{\bullet}}^{\bullet} \colon \mathcal{B}_{s^{\bullet}}^{\bullet} \xrightarrow{\sim} \mathcal{DL}_{s^{\bullet}}^{\bullet} = \mathcal{DL}^{\bullet}(\mathscr{V}_{s^{\bullet}}, \{ \ , \ \}_{s^{\bullet}}).$$

In particular, $B^{\bullet}_{s^{\bullet}}$ is a geometrically irreducible projective smooth scheme in $\mathsf{Sch}_{/\kappa}$ of dimension $\lfloor \frac{N}{2} \rfloor$.

(4) If we denote by $(\mathcal{H}_{s^{\bullet}1}, \mathcal{H}_{s^{\bullet}2})$ the universal object over $\mathrm{DL}_{s^{\bullet}}^{\bullet}$, then there is a canonical isomorphism

$$\zeta_{s^{\bullet}}^{\bullet*}\left(\mathcal{H}_{s^{\bullet}1}^{\dashv}/\mathcal{H}_{s^{\bullet}2}\right) \simeq \iota^{\bullet*}\operatorname{Lie}_{\mathcal{A},\tau_{\infty}^{\mathsf{c}}}$$

of line bundles on $B^{\bullet}_{s\bullet}$.

Proof. For an object $(A_0, \lambda_0, \eta_0^p; A, \lambda, \eta^p; A^{\bullet}, \lambda^{\bullet}, \eta^{p \bullet}; \gamma) \in B^{\bullet}_{\mathfrak{p}}(V^{\circ}, K^{p \circ})(S)$, Definition 5.3.2(a) implies that there is a (unique) O_F -linear quasi-p-isogeny $\check{\gamma}: A^{\bullet} \to A$ such that $\check{\gamma} \circ \gamma = \varpi \cdot \mathrm{id}_A$ hence $\gamma \circ \breve{\gamma} = \varpi \cdot \mathrm{id}_{A^{\bullet}}$. Moreover, we have the following properties from Definition 5.3.2:

- (a') ker $\check{\gamma}[p^{\infty}]$ is contained in $A^{\bullet}[\mathfrak{p}]$;
- (b') $(\operatorname{im} \breve{\gamma}_{*,\tau_{\infty}})^{\perp}$ is contained in $\omega_{A^{\vee},\tau_{\infty}^{\circ}}$; (c') we have $\varpi \cdot \lambda^{\bullet} = \breve{\gamma}^{\vee} \circ \lambda \circ \breve{\gamma}$; and
- (d') the K^p-orbit of maps $v \mapsto \overline{\omega}^{-1} \check{\gamma}_* \circ \eta^{\bullet p}(v)$ for $v \in \mathcal{V}^{\circ} \otimes_{\mathbb{Q}} \mathbb{A}^{\infty,p}$ coincides with η^p .

First, we show (1). It is clear that $B_{s^{\bullet}}^{\bullet}$ is a scheme of finite type over κ . Consider a closed immersion $S \hookrightarrow \hat{S}$ in Sch'_{κ} defined by an ideal sheaf \mathcal{I} satisfying $\mathcal{I}^2 = 0$. Take a point x = $(A_0, \lambda_0, \eta_0^p; A, \lambda, \eta^p; A^{\bullet}, \lambda^{\bullet}, \eta^{p\bullet}; \gamma) \in B^{\bullet}_{\bullet}(S)$. To compute lifting of x to \hat{S} , we use the Serre-Tate and Grothendieck–Messing theories. Note that lifting γ is equivalent to lifting both γ and $\check{\gamma}$, satisfying (b,c,d) in Definition 5.3.2 and (b',c',d') above, respectively. Thus, by Proposition 3.4.8, to lift x to an \hat{S} -point is equivalent to lifting

- O $\omega_{A^{\vee}/S,\tau_{\infty}}$ to a subbundle $\hat{\omega}_{A^{\vee},\tau_{\infty}}$ of $\mathrm{H}_{1}^{\mathrm{cris}}(A/\hat{S})_{\tau_{\infty}}$ (of rank 1),
- O $\omega_{A^{\vee}/S,\tau_{\infty}^{c}}$ to a subbundle $\hat{\omega}_{A^{\vee},\tau_{\infty}^{c}}$ of $\mathrm{H}_{1}^{\mathrm{cris}}(A/\hat{S})_{\tau_{\infty}^{c}}$ (of rank N-1),

subject to the following requirements

- (a") $\hat{\omega}_{A^{\vee},\tau_{\infty}}$ and $\hat{\omega}_{A^{\vee},\tau_{\infty}}$ are orthogonal under $\langle , \rangle_{\lambda,\tau_{\infty}}^{\text{cris}}$ (3.3);
- (b") $(\check{\gamma}_{*,\tau_{\infty}} \mathrm{H}_{1}^{\mathrm{cris}}(A^{\bullet}/\hat{S})_{\tau_{\infty}})^{\perp}$ is contained in $\hat{\omega}_{A^{\vee},\tau_{\infty}^{\mathsf{c}}}$.

As $\check{\gamma}_{*,\tau_{\infty}} \mathrm{H}_{1}^{\mathrm{cris}}(A^{\bullet}/\hat{S})_{\tau_{\infty}} = \ker \gamma_{*,\tau_{\infty}} \subseteq \mathrm{H}_{1}^{\mathrm{cris}}(A/\hat{S})_{\tau_{\infty}}, \, (\mathrm{b}^{"})$ is equivalent to

(c") $(\ker \gamma_{*,\tau_{\infty}})^{\perp}$ is contained in $\hat{\omega}_{A^{\vee},\tau_{\infty}^{c}}$.

To summarize, lifting x to an \hat{S} -point is equivalent to lifting $\omega_{A^{\vee}/S,\tau_{\infty}^{c}}$ to a subbundle $\hat{\omega}_{A^{\vee},\tau_{\infty}^{c}}$ of $\mathrm{H}_{1}^{\mathrm{cris}}(A/\hat{S})_{\tau_{\infty}^{c}}$ containing $(\ker \gamma_{*,\tau_{\infty}})^{\perp}$, and then lifting $\omega_{A^{\vee}/S,\tau_{\infty}}$ to a subbundle $\hat{\omega}_{A^{\vee},\tau_{\infty}}$ of $\hat{\omega}_{A^{\vee},\tau_{\infty}^{c}}^{\perp}$. Thus, (1) follows.

Next, we show (2). By Theorem 5.1.5(4), the map $\mathcal{T}_{B^{\bullet}_{s^{\bullet}}/\kappa} \to \iota^{\bullet*} \mathcal{T}_{M^{\bullet}_{\mathfrak{p}}(V^{\circ}, K^{p^{\circ}})/\kappa}|_{B^{\bullet}_{s^{\bullet}}}$ is induced by the canonical map

$$\mathcal{H}om\left(\omega_{\mathcal{A}^{\vee},\tau_{\infty}^{c}}/(\ker\gamma_{*,\tau_{\infty}})^{\perp},\operatorname{Lie}_{\mathcal{A}^{\vee},\tau_{\infty}^{c}}\right)\to\mathcal{H}om\left(\omega_{\mathcal{A}^{\vee},\tau_{\infty}^{c}}/\operatorname{H}_{1}^{\mathrm{dR}}(\mathcal{A})_{\tau_{\infty}}^{\perp},\operatorname{Lie}_{\mathcal{A}^{\vee},\tau_{\infty}^{c}}\right).$$

It is clearly injective, whose cokernel is canonically isomorphic to

$$\mathcal{H}om\left((\ker\gamma_{*,\tau_{\infty}})^{\perp}/\mathrm{H}_{1}^{\mathrm{dR}}(\mathcal{A})_{\tau_{\infty}}^{\perp},\mathrm{Lie}_{\mathcal{A}^{\vee},\tau_{\infty}^{\mathsf{c}}}\right)$$
$$\simeq\mathcal{H}om\left((\operatorname{im}\gamma_{*,\tau_{\infty}})^{\vee},\mathrm{Lie}_{\mathcal{A}^{\vee},\tau_{\infty}^{\mathsf{c}}}\right)\simeq(\operatorname{im}\gamma_{*,\tau_{\infty}})\otimes_{\mathcal{O}_{\mathrm{B}_{s^{\bullet}}^{\bullet}}}\mathrm{Lie}_{\mathcal{A}^{\vee},\tau_{\infty}^{\mathsf{c}}}$$

We obtain (2).

(5.6)

Third, we show (3). We first show that $\zeta_{s^{\bullet}}^{\bullet}$ has the correct image, namely, we check

O rank_{\mathcal{O}_S} $H_1 = \lceil \frac{N}{2} \rceil$ and rank_{\mathcal{O}_S} $H_2 = \lceil \frac{N}{2} \rceil - 1$: By Lemma 3.4.13(2,3) and Definition 5.3.2, we have

$$\operatorname{rank}_{\mathcal{O}_{S}}(\ker \gamma_{*,\tau_{\infty}}) + \operatorname{rank}_{\mathcal{O}_{S}}(\ker \gamma_{*,\tau_{\infty}^{c}}) = 2\lfloor \frac{N}{2} \rfloor + 1,$$
$$\operatorname{rank}_{\mathcal{O}_{S}}(\ker \gamma_{*,\tau_{\infty}}) - \operatorname{rank}_{\mathcal{O}_{S}}(\ker \gamma_{*,\tau_{\infty}^{c}}) = 1,$$

which imply

$$\operatorname{rank}_{\mathcal{O}_S}(\ker \gamma_{*,\tau_{\infty}}) = \lceil \frac{N+1}{2} \rceil, \quad \operatorname{rank}_{\mathcal{O}_S}(\ker \gamma_{*,\tau_{\infty}^{\mathsf{c}}}) = \lceil \frac{N-1}{2} \rceil.$$

Thus, we obtain $\operatorname{rank}_{\mathcal{O}_S} H_1 = \lceil \frac{N}{2} \rceil$. Since $\ker \gamma_{*,\tau_{\infty}^c} \subseteq (\ker \gamma_{*,\tau_{\infty}})^{\perp} \subseteq \omega_{A^{\vee}/S,\tau_{\infty}^c}$, we have $H_2 = \gamma_{*,\tau_{\infty}^c} \omega_{A^{\vee}/S,\tau_{\infty}^c} \simeq \omega_{A^{\vee}/S,\tau_{\infty}^c} / \ker \gamma_{*,\tau_{\infty}^c}$. Thus, we obtain $\operatorname{rank}_{\mathcal{O}_S} H_2 = \lceil \frac{N}{2} \rceil - 1$.

O H^{dR}₁(A[•]/S)[⊥]_{τ∞} ⊆ H₂: By Definition 5.3.2(c) and the definition of $\check{\gamma}$, we have $\lambda \circ \check{\gamma} = \gamma^{\vee} \circ \lambda^{\bullet}$, which implies

(5.7)
$$(\ker \gamma_{*,\tau_{\infty}})^{\perp} = \gamma_{*,\tau_{\infty}}^{-1}(\mathrm{H}_{1}^{\mathrm{dR}}(A^{\bullet}/S)_{\tau_{\infty}}^{\perp}).$$

By Definition 5.3.2(b), H_2 contains $\gamma_{*,\tau_{\infty}^{c}}(\ker \gamma_{*,\tau_{\infty}})^{\perp}$ in which the latter coincides with $H_1^{dR}(A^{\bullet}/S)_{\tau_{\infty}}^{\perp}$ by (5.7). $O H_2 \subseteq H_1$: As $\lambda \circ \check{\gamma} = \gamma^{\vee} \circ \lambda^{\bullet}$, we have

$$\langle (\breve{\gamma}_{*,\tau_{\infty}})^{-1}\omega_{A^{\vee}/S,\tau_{\infty}},\gamma_{*,\tau_{\infty}^{\mathsf{c}}}\omega_{A^{\vee}/S,\tau_{\infty}^{\mathsf{c}}}\rangle_{\lambda^{\bullet},\tau_{\infty}} = \langle \breve{\gamma}_{*,\tau_{\infty}}(\breve{\gamma}_{*,\tau_{\infty}})^{-1}\omega_{A^{\vee}/S,\tau_{\infty}},\omega_{A^{\vee}/S,\tau_{\infty}^{\mathsf{c}}}\rangle_{\lambda,\tau_{\infty}} = 0.$$

Thus, we have $H_2 \subseteq H_1$.

 $O H_2 \subseteq H_1^{\dashv}$: Note that we have

$$\operatorname{im} \gamma_{*,\tau_{\infty}^{\mathsf{c}}} = \operatorname{ker} \check{\gamma}_{*,\tau_{\infty}^{\mathsf{c}}} = (\check{\gamma}_{*,\tau_{\infty}^{\mathsf{c}}})^{-1} (\mathsf{F}\omega_{A^{\vee}/S,\tau_{\infty}}^{(p)}) \subseteq \mathsf{F}((\check{\gamma}_{*,\tau_{\infty}})^{-1}\omega_{A^{\vee}/S,\tau_{\infty}}) = \mathsf{F}((H_{1}^{(p)})^{\perp}).$$

Thus, $(\mathbf{F}((H_1^{(p)})^{\perp}))^{\perp} \subseteq (\operatorname{im} \gamma_{*,\tau_{\infty}^{c}})^{\perp}$, which in turn implies $H_1^{(p)} \subseteq \mathbf{V}((\operatorname{im} \gamma_{*,\tau_{\infty}^{c}})^{\perp})$, which implies $\mathbf{V}^{-1}H_1^{(p)} \subseteq (\operatorname{im} \gamma_{*,\tau_{\infty}^{c}})^{\perp}$, which implies $\operatorname{im} \gamma_{*,\tau_{\infty}^{c}} \subseteq H_1^{\dashv}$. By comparing ranks via (5.6), we obtain

(5.8)
$$\operatorname{im} \gamma_{*,\tau_{\infty}^{\mathsf{c}}} = H_{1}^{-}$$

In particular, H_1^{\dashv} contains H_2 as im $\gamma_{*,\tau_{\infty}^c}$ does.

- $O H_1 \subseteq H_2^{\dashv}: \text{ Note that } H_2^{(p)} = \gamma_{*,\tau_{\infty}}(\mathbb{V}H_1^{d\mathbb{R}}(A/S)_{\tau_{\infty}^c}) = \mathbb{V}(\operatorname{im} \gamma_{*,\tau_{\infty}}) = \mathbb{V}(\ker \check{\gamma}_{*,\tau_{\infty}}) \subseteq \mathbb{V}(H_1^{\perp}).$ Thus, $\mathbb{V}^{-1}H_2^{(p)} \subseteq H_1^{\perp}$, which implies $H_1 \subseteq (\mathbb{V}^{-1}H_2^{(p)})^{\perp} = H_2^{\dashv}.$
- $\bigcirc H_1^{\dashv} \subseteq H_2^{\dashv}$: It follows from $H_2 \subseteq H_1$.

Since the target of $\zeta_{s^{\bullet}}^{\bullet}$ is smooth over κ by Proposition A.2.2, to see that $\zeta_{s^{\bullet}}^{\bullet}$ is an isomorphism, it suffices to check that for every algebraically closed field κ' containing κ

- (3.1) $\zeta_{s^{\bullet}}^{\bullet}$ induces a bijection on κ' -points; and
- (3.2) $\zeta_{s^{\bullet}}^{\bullet}$ induces an isomorphism on the tangent spaces at every κ -point.

To ease notation, we may assume $\kappa' = \kappa$.

For (3.1), we construct an inverse to the map $\zeta_{s^{\bullet}}^{\bullet}(\kappa)$. Take a point $y \in \mathrm{DL}_{s^{\bullet}}^{\bullet}(\kappa)$ represented by κ -linear subspaces

$$\mathrm{H}_{1}^{\mathrm{dR}}(A^{\bullet})_{\tau_{\infty}}^{\perp} \subseteq H_{2} \subseteq H_{1} \subseteq \mathscr{V}_{s^{\bullet}} = \mathrm{H}_{1}^{\mathrm{dR}}(A^{\bullet})_{\tau_{\infty}^{c}}.$$

We regard F and V as those sesquilinear maps in Remark 3.4.10. For every $\tau \in \Sigma_{\infty}$, we define a $W(\kappa)$ -submodule $\mathcal{D}_{A,\tau} \subseteq \mathcal{D}(A^{\bullet})_{\tau}$ as follows.

- O If $\tau \notin \{\tau_{\infty}, \tau_{\infty}^{\mathsf{c}}\}$, then $\mathcal{D}_{A,\tau} = \mathcal{D}(A^{\bullet})_{\tau}$.
- O We set $\mathcal{D}_{A,\tau_{\infty}} \coloneqq \mathbb{V}^{-1}\tilde{H}_2$, where \tilde{H}_2 is the preimage of H_2 under the reduction map $\mathcal{D}(A^{\bullet})_{\tau_{\infty}^{c}} \to \mathcal{D}(A^{\bullet})_{\tau_{\infty}^{c}}/p\mathcal{D}(A^{\bullet})_{\tau_{\infty}^{c}} = \mathrm{H}_{1}^{\mathrm{dR}}(A^{\bullet})_{\tau_{\infty}^{c}}.$
- O We set $\mathcal{D}_{A,\tau_{\infty}^{c}} \coloneqq \mathbb{F}\tilde{H}_{1}^{c}$, where \tilde{H}_{1}^{c} is the preimage of H_{1}^{\perp} under the reduction map $\mathcal{D}(A^{\bullet})_{\tau_{\infty}} \to \mathcal{D}(A^{\bullet})_{\tau_{\infty}}/p\mathcal{D}(A^{\bullet})_{\tau_{\infty}} = \mathrm{H}_{1}^{\mathrm{dR}}(A^{\bullet})_{\tau_{\infty}}.$

Finally, put $\mathcal{D}_A \coloneqq \bigoplus_{\tau \in \Sigma_\infty} \mathcal{D}_{A,\tau}$ as a $W(\kappa)$ -submodule of $\mathcal{D}(A^{\bullet})$. We show that it is stable under F and V. It suffices to show that both F and V stabilize $\mathcal{D}_{A,\tau_\infty} \oplus \mathcal{D}_{A,\tau_\infty^c}$, which breaks into checking that

- O $\mathcal{FD}_{A,\tau_{\infty}} \subseteq \mathcal{D}_{A,\tau_{\infty}^{c}}$, that is, $\mathcal{FV}^{-1}\tilde{H}_{2} \subseteq \mathcal{F}\tilde{H}_{1}^{c}$. It suffices to show that $\mathcal{V}^{-1}H_{2}$ (as a subspace of $\mathrm{H}_{1}^{\mathrm{dR}}(A^{\bullet})_{\tau_{\infty}}$) is contained in H_{1}^{\perp} , which follow from the relation $H_{1} \subseteq H_{2}^{\dashv}$.
- O $\mathcal{FD}_{A,\tau_{\infty}^{c}} \subseteq \mathcal{D}_{A,\tau_{\infty}}$, that is, $\mathcal{FF}\tilde{H}_{1}^{c} \subseteq \mathbb{V}^{-1}\tilde{H}_{2}$. It suffices to show $p\mathcal{F}\tilde{H}_{1}^{c} \subseteq \tilde{H}_{2}$, which obviously holds.
- O $\mathbb{V}\mathcal{D}_{A,\tau_{\infty}} \subseteq \mathcal{D}_{A,\tau_{\infty}^{c}}$, that is, $\mathbb{V}\mathbb{V}^{-1}\tilde{H}_{2} \subseteq \mathbb{F}\tilde{H}_{1}^{c}$. it suffices to show $H_{2} \subseteq \mathbb{F}H_{1}^{\perp}$, which follows from the identity $\mathbb{F}H_{1}^{\perp} = (\mathbb{V}^{-1}H_{1})^{\perp}$ and the relation $H_{2} \subseteq H_{1}^{\dashv}$.
- O $\mathbb{V}\mathcal{D}_{A,\tau_{\infty}^{c}} \subseteq \mathcal{D}_{A,\tau_{\infty}}$, that is, $\mathbb{V}F\tilde{H}_{1}^{c} \subseteq \mathbb{V}^{-1}\tilde{H}_{2}$. It is obvious as $\mathbb{V}^{-1}\tilde{H}_{2}$ contains $p\mathcal{D}(A^{\bullet})_{\tau_{\infty}}$.

Thus, $(\mathcal{D}_A, \mathbf{F}, \mathbf{V})$ is a Dieudonné module over $W(\kappa)$. By the Dieudonné theory, there is an O_F abelian scheme A over κ with $\mathcal{D}(A)_{\tau} = \mathcal{D}_{A,\tau}$ for every $\tau \in \Sigma_{\infty}$, and an O_F -linear isogeny $\gamma \colon A \to A^{\bullet}$ inducing the inclusion of Dieudonné modules $\mathcal{D}(A) = \mathcal{D}_A \subseteq \mathcal{D}(A^{\bullet})$. Moreover, since $\mathfrak{p}\mathcal{D}(A^{\bullet}) \subseteq$ $\mathcal{D}(A)$, we have ker $\gamma[p^{\infty}] \subseteq A[\mathfrak{p}]$. Now we check that $(\ker \gamma_{*,\tau_{\infty}})^{\perp}$ is contained in $\omega_{A^{\vee}/S,\tau_{\infty}^{c}}$, which is equivalent to that $p\mathcal{D}(A^{\bullet})_{\tau_{\infty}}^{\vee} \cap \mathcal{D}(A)_{\tau_{\infty}^{c}} \subseteq \mathbb{V}\mathcal{D}(A)_{\tau_{\infty}}$. However, as H_2 contains $\mathrm{H}_1^{\mathrm{dR}}(A^{\bullet})_{\tau_{\infty}}^{\perp}$, we have $p\mathcal{D}(A^{\bullet})_{\tau_{\infty}}^{\vee} \subseteq \tilde{H}_2 = \mathbb{V}\mathcal{D}(A)_{\tau_{\infty}}$.

Let $\lambda: A \to A^{\vee}$ be the unique quasi-polarization such that $\varpi \lambda = \gamma^{\vee} \circ \lambda^{\bullet} \circ \gamma$. We claim that $\lambda[p^{\infty}]$ is a polarization whose kernel is contained in $A[\mathfrak{p}]$ of rank p^2 . Since $H_2 \subseteq H_1$, we have $\langle \tilde{H}_1^c, \tilde{H}_2 \rangle_{\lambda^{\bullet}, \tau_{\infty}} \subseteq pW(\kappa)$, which implies $\langle \mathcal{D}(A)_{\tau_{\infty}}, \mathcal{D}(A)_{\tau_{\infty}^c} \rangle_{\lambda^{\bullet}, \tau_{\infty}} \subseteq pW(\kappa)$. It is enough to show that the inclusion $\mathcal{D}(A)_{\tau_{\infty}^c} \to \mathcal{D}(A)_{\tau_{\infty}}^{\vee}$ induced from $\langle , \rangle_{\lambda^{\bullet}, \tau_{\infty}}$ has cokernel of length N + 1. This

follows from the facts that the cokernel of $\mathcal{D}(A^{\bullet})_{\tau_{\infty}^{c}} \hookrightarrow \mathcal{D}(A^{\bullet})_{\tau_{\infty}}^{\vee}$ has length $N - 2\lfloor \frac{N}{2} \rfloor$, and the cokernel of $\mathcal{D}(A)_{\tau_{\infty}} \oplus \mathcal{D}(A)_{\tau_{\infty}^{c}} \hookrightarrow \mathcal{D}(A^{\bullet})_{\tau_{\infty}} \oplus \mathcal{D}(A^{\bullet})_{\tau_{\infty}^{c}}$ has length $2\lfloor \frac{N}{2} \rfloor + 1$.

It is an easy consequence of Lemma 3.4.13(2) that the O_F -abelian scheme A has signature type $N\Phi - \tau_{\infty} + \tau_{\infty}^{\mathsf{c}}$. Finally, let η^p be the unique K^p -level structure such that Definition 4.2.3(d) is satisfied. Putting together, we obtain a point $x = (A_0, \lambda_0, \eta_0^p; A, \lambda, \eta^p; A^{\bullet}, \lambda^{\bullet}, \eta^{p\bullet}; \gamma) \in \mathbf{B}_{s\bullet}^{\bullet}(\kappa)$ such that $\zeta_{s\bullet}^{\bullet}(x) = y$. It is easy to see that such assignment gives rise to an inverse of $\zeta_{s\bullet}^{\bullet}(\kappa)$; hence (3.1) follows immediately.

For (3.2), let \mathcal{T}_x and \mathcal{T}_y be the tangent spaces at x and y as in (3.1), respectively. By Proposition A.2.2 and the construction, the induced map $(\zeta_{s^{\bullet}})_* : \mathcal{T}_x \to \mathcal{T}_y$ fits into a commutative diagram

in $Mod(\kappa)$. The right vertical arrow is induced by maps

$$\omega_{A^{\vee},\tau_{\infty}^{\mathsf{c}}}/(\ker\gamma_{*,\tau_{\infty}})^{\perp} \xrightarrow{\gamma_{*,\tau_{\infty}^{\mathsf{c}}}} H_{2}/\mathscr{V}_{s^{\bullet}}^{\dashv}, \quad \operatorname{Lie}_{A^{\vee},\tau_{\infty}^{\mathsf{c}}} \simeq \operatorname{H}_{1}^{\mathrm{dR}}(A)_{\tau_{\infty}^{\mathsf{c}}}/\omega_{A^{\vee},\tau_{\infty}^{\mathsf{c}}} \xrightarrow{\gamma_{*,\tau_{\infty}^{\mathsf{c}}}} H_{1}^{\dashv}/H_{2}$$

which are both isomorphisms by (5.7) and (5.8), respectively. The left vertical arrow is the composition

$$\operatorname{Hom}_{\kappa}\left(\omega_{A^{\vee},\tau_{\infty}},\omega_{A^{\vee},\tau_{\infty}}^{\perp}/\omega_{A^{\vee},\tau_{\infty}}\right) \to \operatorname{Hom}_{\kappa}\left(H_{1}^{\perp}/\mathbb{V}^{-1}H_{2},H_{2}^{\perp}/H_{1}^{\perp}\right) \xrightarrow{\sim} \operatorname{Hom}_{\kappa}\left(H_{1}/H_{2},H_{2}^{\dashv}/H_{1}\right)$$

in which the first arrow is induced by maps

$$H_1^{\perp}/\mathbb{V}^{-1}H_2 \xrightarrow{\check{\gamma}_{*,\tau_{\infty}}} \omega_{A^{\vee},\tau_{\infty}}, \quad H_2^{\perp}/H_1^{\perp} \xrightarrow{\check{\gamma}_{*,\tau_{\infty}}} \omega_{A^{\vee},\tau_{\infty}}^{\perp}/\omega_{A^{\vee},\tau_{\infty}}$$

which are both isomorphisms as $\check{\gamma}_{*,\tau_{\infty}}(H_1^{\perp}) = \omega_{A^{\vee},\tau_{\infty}}, \, \check{\gamma}_{*,\tau_{\infty}}(\mathbb{V}^{-1}H_2) = 0$, and $\check{\gamma}_{*,\tau_{\infty}}(H_2^{\perp}) = \omega_{A^{\vee},\tau_{\infty}}^{\perp}$. Thus, $(\zeta_{s^{\bullet}})_*: \mathcal{T}_x \to \mathcal{T}_y$ is an isomorphism by the Five Lemma; hence (3.2) and (3) follow. Finally, (4) is a consequence of (5.8).

Remark 5.3.5. We have the following remarks concerning Theorem 5.3.4.

- (1) When $K^{p\circ}$ is sufficiently small, the restriction of ι^{\bullet}_{κ} to $B^{\bullet}_{s\bullet}$ is a closed immersion for every point $s^{\bullet} \in S^{\bullet}_{\mathfrak{p}}(V^{\circ}, K^{p\circ})(\kappa)$ and every perfect field κ containing \mathbb{F}^{Φ}_{p} .
- (2) In fact, one can show that the union of $M_{\mathfrak{p}}^{\dagger}(V^{\circ}, K^{p\circ})$ and the image of $\iota^{\bullet} \colon B_{\mathfrak{p}}^{\bullet}(V^{\circ}, K^{p\circ}) \to M_{\mathfrak{p}}^{\bullet}(V^{\circ}, K^{p\circ})$ is exactly the basic locus of $M_{\mathfrak{p}}^{\bullet}(V^{\circ}, K^{p\circ})$. In particular, as long as $N \geq 5$, the basic locus of $M_{\mathfrak{p}}^{\bullet}(V^{\circ}, K^{p\circ})$ is *not* equidimensional.

Construction 5.3.6. To construct a uniformization map for $S^{\bullet}_{\mathfrak{p}}(V^{\circ}, -)$, we need to choose an $O_{F_{\mathfrak{p}}}$ -lattice $\Lambda^{\bullet}_{\mathfrak{p}}$ in $V^{\circ} \otimes_F F_{\mathfrak{p}}$ satisfying

Ο
$$\Lambda_{\mathfrak{p}}^{\circ} \subseteq \Lambda_{\mathfrak{p}}^{\bullet} \subseteq p^{-1}\Lambda_{\mathfrak{p}}^{\circ}$$
, and
Ο $p\Lambda_{\mathfrak{p}}^{\bullet} \subseteq (\Lambda_{\mathfrak{p}}^{\bullet})^{\vee}$ such that $(\Lambda_{\mathfrak{p}}^{\bullet})^{\vee}/p\Lambda_{\mathfrak{p}}^{\bullet}$ has length 0 (resp. 1) if N is even (resp. odd)

Let $K_{\mathfrak{p}}^{\bullet}$ be the stabilizer of $\Lambda_{\mathfrak{p}}^{\bullet}$; and put $K_{p}^{\bullet} := K_{\mathfrak{p}}^{\bullet} \times \prod_{\mathfrak{q}|p,\mathfrak{q}\neq\mathfrak{p}} K_{\mathfrak{q}}^{\circ}$. Similar to Construction 4.3.2, we may construct a *uniformization map*

(5.9)
$$\upsilon^{\bullet} \colon \mathrm{S}_{\mathfrak{p}}^{\bullet}(\mathrm{V}^{\circ}, -)(\overline{\mathbb{F}}_{p}) \xrightarrow{\sim} \mathrm{Sh}(\mathrm{V}^{\circ}, -\mathrm{K}_{p}^{\bullet}) \times \mathrm{T}_{\mathfrak{p}}(\overline{\mathbb{F}}_{p})$$

in $\operatorname{Fun}(\mathfrak{K}(\mathcal{V}^{\circ})^{p} \times \mathfrak{T}, \operatorname{Set})_{/\operatorname{T}_{\mathfrak{p}}(\overline{\mathbb{F}}_{p})}$ which is an isomorphism, under which the induced action of $\operatorname{Gal}(\overline{\mathbb{F}}_{p}/\mathbb{F}_{p}^{\Phi})$ on the target is trivial on $\operatorname{Sh}(\mathcal{V}^{\circ}, -\mathcal{K}_{p}^{\bullet})$.

Moreover, similar to Construction 4.3.5 and Proposition 4.3.6, for every $g \in K^{\bullet}_{\mathfrak{p}} \setminus U(V^{\circ})(F^{+}_{\mathfrak{p}})/K^{\bullet}_{\mathfrak{p}}$ we may construct the Hecke correspondence

$$\mathrm{Hk}_g \colon \mathrm{S}^{\bullet}_{\mathfrak{p}}(\mathrm{V}^{\circ},-)_g \to \mathrm{S}^{\bullet}_{\mathfrak{p}}(\mathrm{V}^{\circ},-) \times \mathrm{S}^{\bullet}_{\mathfrak{p}}(\mathrm{V}^{\circ},-)$$

as a morphism in $\mathsf{Fun}(\mathfrak{K}(V^{\circ})^p \times \mathfrak{T}, \mathsf{Sch}_{/\mathbb{F}_p^{\Phi}})_{/\mathrm{T}_p}$ that is finite étale and compatible with the uniformization map.

5.4. Basic correspondence on link stratum. In this subsection, we construct and study the basic correspondence on the link stratum $M_{\mu}^{\dagger}(V^{\circ}, -)$. We also discuss its relation with the two previously constructed basic correspondences. We assume $N \geq 2$.

Definition 5.4.1. We define a functor

$$\begin{aligned} \mathbf{S}^{\dagger}_{\mathfrak{p}}(\mathbf{V}^{\circ},-) \colon \mathfrak{K}(\mathbf{V}^{\circ})^{p} \times \mathfrak{T} &\to \mathsf{PSch}_{/\mathbb{F}_{p}^{\Phi}}^{\prime} \\ \mathbf{K}^{p\circ} &\mapsto \mathbf{S}^{\dagger}_{\mathfrak{p}}(\mathbf{V}^{\circ},\mathbf{K}^{p\circ}) \end{aligned}$$

such that for every $S \in \mathsf{Sch}'_{/\mathbb{F}_p^{\Phi}}, \, \mathrm{S}^{\dagger}_{\mathfrak{p}}(\mathrm{V}^\circ, \mathrm{K}^{p\circ})(S)$ is the set of equivalence classes of decuples $(A_0, \lambda_0, \eta_0^p; A^{\circ}, \lambda^{\circ}, \eta^{p \circ}; A^{\bullet}, \lambda^{\bullet}, \eta^{p \bullet}; \psi)$ where

- $\begin{array}{l} \bigcirc \ (A_0,\lambda_0,\eta_0^p;A^\circ,\lambda^\circ,\eta^{p\circ}) \text{ is an element in } \mathbf{S}^\circ_{\mathfrak{p}}(\mathbf{V}^\circ,\mathbf{K}^{p\circ})(S); \\ \bigcirc \ (A_0,\lambda_0,\eta_0^p;A^\bullet,\lambda^\bullet,\eta^{p\bullet}) \text{ is an element in } \mathbf{S}^\circ_{\mathfrak{p}}(\mathbf{V}^\circ,\mathbf{K}^{p\circ})(S); \end{array}$
- $O \psi \colon A^{\circ} \to A^{\bullet}$ is an O_F -linear quasi-*p*-isogeny (Definition 3.4.5) such that
 - (a) ker $\psi[p^{\infty}]$ is contained in $A^{\circ}[\mathfrak{p}]$;
 - (b) we have $\overline{\omega} \cdot \lambda^{\circ} = \psi^{\vee} \circ \lambda^{\bullet} \circ \psi$; and
 - (c) the K^{po}-orbit of maps $v \mapsto \psi_* \circ \eta^{p\circ}(v)$ for $v \in V^{\circ} \otimes_{\mathbb{Q}} \mathbb{A}^{\infty,p}$ coincides with $\eta^{p\bullet}$.

The equivalence relation and the action of morphisms in $\mathfrak{K}(\mathcal{V}^{\circ})^{p} \times \mathfrak{T}$ are defined similarly as in Definition 4.2.3.

We have apparently the forgetful morphism

$$\mathrm{S}^{\dagger}_{\mathfrak{p}}(\mathrm{V}^{\circ},-) \to \mathrm{T}_{\mathfrak{p}}$$

in $\operatorname{Fun}(\mathfrak{K}(V^{\circ})^{p} \times \mathfrak{T}, \operatorname{\mathsf{PSch}}'_{/\mathbb{F}^{\Phi}})$ which is represented by finite and étale schemes.

By definition, we have the two forgetful morphisms

$$s^{\dagger\circ} \colon S^{\dagger}_{\mathfrak{p}}(V^{\circ}, -) \to S^{\circ}_{\mathfrak{p}}(V^{\circ}, -), \quad s^{\dagger\bullet} \colon S^{\dagger}_{\mathfrak{p}}(V^{\circ}, -) \to S^{\bullet}_{\mathfrak{p}}(V^{\circ}, -)$$

in $\operatorname{Fun}(\mathfrak{K}(V^{\circ})^{p} \times \mathfrak{T}, \operatorname{Sch}_{\mathbb{F}^{\Phi}})/_{\operatorname{T}_{\mathfrak{p}}}$.

Definition 5.4.2. We define $B^{\dagger}_{\mathfrak{p}}(V^{\circ}, -)$ to be the limit of the following diagram

$$\begin{split} \mathbf{S}_{\mathfrak{p}}^{\circ}(\mathbf{V}^{\circ},-) & \stackrel{\pi^{\circ}}{\longleftarrow} \mathbf{B}_{\mathfrak{p}}^{\circ}(\mathbf{V}^{\circ},-) \xrightarrow{\iota^{\circ}} \mathbf{M}_{\mathfrak{p}}^{\circ}(\mathbf{V}^{\circ},-) \\ & \uparrow^{\mathsf{s}^{\dagger\circ}} & \uparrow^{\mathsf{m}^{\dagger\circ}} \\ \mathbf{S}_{\mathfrak{p}}^{\dagger}(\mathbf{V}^{\circ},-) & \mathbf{M}_{\mathfrak{p}}^{\dagger}(\mathbf{V}^{\circ},-) \\ & \downarrow^{\mathsf{s}^{\dagger\bullet}} & \downarrow^{\mathsf{m}^{\dagger\bullet}} \\ \mathbf{S}_{\mathfrak{p}}^{\bullet}(\mathbf{V}^{\circ},-) & \stackrel{\tau^{\bullet}}{\longleftarrow} \mathbf{B}_{\mathfrak{p}}^{\bullet}(\mathbf{V}^{\circ},-) \xrightarrow{\iota^{\bullet}} \mathbf{M}_{\mathfrak{p}}^{\bullet}(\mathbf{V}^{\circ},-) \end{split}$$

in the category $\mathsf{Fun}(\mathfrak{K}(\mathcal{V}^\circ)^p \times \mathfrak{T}, \mathsf{Sch}_{/\mathbb{F}_n^\Phi})_{/\mathrm{T}_p}$.

From the definition above, we have the following commutative diagram

$$(5.10) \qquad S_{\mathfrak{p}}^{\circ}(V^{\circ}, -) \stackrel{\pi^{\circ}}{\underset{s^{\dagger}^{\circ}}{\overset{s^{\bullet}}{\overset{s^{\bullet}}{\overset{s^{\bullet}}{\overset{s^{\bullet}}{\overset{s^{\bullet}}{\overset{s^{\bullet}}}{\overset{s^{\bullet}}{\overset{s^{\bullet}}{\overset{s^{\bullet}}}{\overset{s^{\bullet}}{\overset{s^{\bullet}}}{\overset{s^{\bullet}}{\overset{s^{\bullet}}{\overset{s^{\bullet}}{\overset{s^{\bullet}}{\overset{s^{\bullet}}}{\overset{s^{\bullet}}{\overset{s^{\bullet}}{\overset{s^{\bullet}}}{\overset{s^{\bullet}}{\overset{s^{\bullet}}{\overset{s^{\bullet}}}{\overset{s^{\bullet}}{\overset{s^{\bullet}}}{\overset{s^{\bullet}}{\overset{s^{\bullet}}}{\overset{s^{\bullet}}{\overset{s^{\bullet}}}{\overset{s^{\bullet}}{\overset{s^{\bullet}}}{\overset{s^{\bullet}}{\overset{s^{\bullet}}}{\overset{s^{\bullet}}{\overset{s^{\bullet}}}{\overset{s^{\bullet}}}{\overset{s^{\bullet}}{\overset{s^{\bullet}}}{\overset{s^{\bullet}}}{\overset{s^{\bullet}}{\overset{s^{\bullet}}}{\overset{s^{\bullet}}{\overset{s^{\bullet}}}}}}}}}}}}}}}}}}}}}}}}$$

in $\operatorname{Fun}(\mathfrak{K}(V^{\circ})^{p} \times \mathfrak{T}, \operatorname{Sch}_{/\mathbb{F}_{p}^{\Phi}})_{/\mathrm{T}_{p}}$, together with the four new morphisms from $\operatorname{B}_{\mathfrak{p}}^{\dagger}(V^{\circ}, -)$ as indicated. It will be clear in Subsection 5.10 why we draw the diagram oblique.

Theorem 5.4.3. In the diagram (5.10), we have

(1) The square

$$\begin{split} B^{\dagger}_{\mathfrak{p}}(V^{\circ},-) & \stackrel{\iota^{\dagger}}{\longrightarrow} M^{\dagger}_{\mathfrak{p}}(V^{\circ},-) \\ & \downarrow_{b^{\dagger \bullet}} & \downarrow_{m^{\dagger \bullet}} \\ B^{\bullet}_{\mathfrak{p}}(V^{\circ},-) & \stackrel{\iota^{\bullet}}{\longrightarrow} M^{\bullet}_{\mathfrak{p}}(V^{\circ},-) \end{split}$$

is a Cartesian diagram.

(2) Take a point $s^{\dagger} = (A_0, \lambda_0, \eta_0^p; A^{\circ}, \lambda^{\circ}, \eta^{p \circ}; A^{\bullet}, \lambda^{\bullet}, \eta^{p \bullet}; \psi) \in S_{\mathfrak{p}}^{\dagger}(V^{\circ}, K^{p \circ})(\kappa)$ where κ is a perfect field containing \mathbb{F}_p^{Φ} . Put $B_{s^{\dagger}}^{\dagger} := \pi^{\dagger - 1}(s^{\dagger})$ and $\mathscr{V}_{s^{\dagger}} := (\operatorname{im} \psi_{*, \tau_{\infty}^{\circ}})/\mathrm{H}_1^{\mathrm{dR}}(A^{\bullet}/\kappa)_{\tau_{\infty}}^{\perp}$ which has dimension $\lfloor \frac{N}{2} \rfloor$. Then the assignment sending

$$((A_0, \lambda_0, \eta_0^p; A, \lambda, \eta^p; A^{\circ}, \lambda^{\circ}, \eta^{p^{\circ}}; \beta), (A_0, \lambda_0, \eta_0^p; A, \lambda, \eta^p; A^{\bullet}, \lambda^{\bullet}, \eta^{p^{\bullet}}; \gamma)) \in \mathcal{B}_{s^{\dagger}}^{\dagger}(S)$$

(with $\gamma = \psi \circ \beta$) to $(\gamma_{*, \tau_{\infty}^{c}} \omega_{A^{\vee}/S, \tau_{\infty}^{c}})/\mathcal{H}_1^{\mathrm{dR}}(A^{\bullet}/S)_{\tau_{\infty}}^{\perp}$ induces an isomorphism
 $\zeta_{s^{\dagger}}^{\dagger} \colon \mathcal{B}_{s^{\dagger}}^{\dagger} \xrightarrow{\sim} \mathbb{P}(\mathscr{V}_{s^{\dagger}}).$

Proof. For (1), unravelling all the definitions, it suffices to show that for every object

 $((A_0, \lambda_0, \eta_0^p; A, \lambda, \eta^p; A^{\circ}, \lambda^{\circ}, \eta^{p \circ}; \beta), (A_0, \lambda_0, \eta_0^p; A, \lambda, \eta^p; A^{\bullet}, \lambda^{\bullet}, \eta^{p \bullet}; \gamma))$

of $\mathrm{M}^{\dagger}_{\mathfrak{p}}(\mathrm{V}^{\circ},\mathrm{K}^{p\circ})(S) \times_{\mathrm{M}^{\bullet}_{\mathfrak{p}}(\mathrm{V}^{\circ},\mathrm{K}^{p\circ})(S)} \mathrm{B}^{\bullet}_{\mathfrak{p}}(\mathrm{V}^{\circ},\mathrm{K}^{p\circ})(S)$, the quasi-isogeny $\psi \coloneqq \gamma \circ \beta^{-1} \colon A^{\circ} \to A^{\bullet}$ is a quasi-*p*-isogeny. However, we know that $\beta_{*,\tau_{\infty}^{c}} \colon \mathrm{H}^{\mathrm{dR}}_{1}(A)_{\tau_{\infty}^{c}} \to \mathrm{H}^{\mathrm{dR}}_{1}(A^{\circ})_{\tau_{\infty}^{c}}$ is an isomorphism; and $\ker \beta_{*,\tau_{\infty}} = \omega_{A^{\vee},\tau_{\infty}}$. Thus, it suffices to show that $\omega_{A^{\vee},\tau_{\infty}}$ is contained in $\ker \gamma$, which is clear since $\omega_{A^{\bullet},\tau_{\infty}} = 0$.

For (2), we first show that for a point

$$x^{\bullet} = (A_0, \lambda_0, \eta_0^p; A, \lambda, \eta^p; A^{\bullet}, \lambda^{\bullet}, \eta^{p\bullet}; \gamma) \in \mathcal{B}^{\bullet}_{\mathfrak{p}}(\mathcal{V}^{\circ}, \mathcal{K}^{p\circ})(S),$$

 $\iota^{\bullet}(x^{\bullet}) \text{ belongs to } \mathrm{M}^{\dagger}_{\mathfrak{p}}(\mathrm{V}^{\circ},\mathrm{K}^{p\circ})(S) \text{ if and only if } H_{1} = H_{1}^{\dashv}, \text{ where we recall from Theorem 5.3.4} \\ \mathrm{that } H_{1} \coloneqq ((\check{\gamma}_{*,\tau_{\infty}})^{-1}\omega_{A^{\vee},\tau_{\infty}})^{\perp}. \text{ In fact, by Definition 5.1.3, } \iota^{\bullet}(x^{\bullet}) \in \mathrm{M}^{\dagger}_{\mathfrak{p}}(\mathrm{V}^{\circ},\mathrm{K}^{p\circ})(S) \text{ if and only if } \\ \omega_{A^{\vee},\tau_{\infty}} = \mathrm{H}^{1}_{\mathrm{dR}}(A)_{\tau_{\infty}^{\circ}}^{\perp}. \text{ In the proof of Theorem 5.3.4, we see im } \gamma_{*,\tau_{\infty}^{\circ}} = H_{1}^{\dashv}(5.8). \text{ As } \lambda \circ \check{\gamma} = \gamma^{\vee} \circ \lambda^{\bullet}, \\ \text{we have } (\mathrm{im } \gamma_{*,\tau_{\infty}^{\circ}})^{\perp} = (\check{\gamma}_{*,\tau_{\infty}})^{-1}\mathrm{H}^{1}_{\mathrm{dR}}(A)_{\tau_{\infty}^{\circ}}^{\perp}. \text{ Thus, if } \omega_{A^{\vee},\tau_{\infty}} = \mathrm{H}^{1}_{\mathrm{dR}}(A)_{\tau_{\infty}^{\circ}}^{\perp}, \text{ then } H_{1} = ((\mathrm{im } \gamma_{*,\tau_{\infty}^{\circ}})^{\perp})^{\perp} \\ \text{which equals im } \gamma_{*,\tau_{\infty}^{\circ}} = H_{1}^{\dashv}, \text{ as im } \gamma_{*,\tau_{\infty}^{\circ}} \text{ contains } \mathrm{H}^{\mathrm{dR}}_{1}(A^{\bullet})_{\tau_{\infty}}^{\perp}. \text{ On the other hand, if } H_{1} = H_{1}^{\dashv}, \text{ then } \\ (\check{\gamma}_{*,\tau_{\infty}})^{-1}\omega_{A^{\vee},\tau_{\infty}} = (\mathrm{im } \gamma_{*,\tau_{\infty}^{\circ}})^{\perp} = (\check{\gamma}_{*,\tau_{\infty}})^{-1}\mathrm{H}^{1}_{\mathrm{dR}}(A)_{\tau_{\infty}^{\circ}}^{\perp}, \text{ which implies easily that } \omega_{A^{\vee},\tau_{\infty}} = \mathrm{H}^{1}_{\mathrm{dR}}(A)_{\tau_{\infty}^{\circ}}^{\perp}. \end{cases}$

Second, we show $H_1 = \operatorname{im} \psi_{*,\tau_{\infty}^c}$ if $x^{\bullet} \in \mathrm{B}_{s^{\dagger}}^{\dagger}(S)$. Since $\gamma = \psi \circ \beta$, we have $\operatorname{im} \gamma_{*,\tau_{\infty}^c} \subseteq \operatorname{im} \psi_{*,\tau_{\infty}^c}$. As $\operatorname{im} \gamma_{*,\tau_{\infty}^c} = H_1^{\dashv} = H_1$, we have $H_1 \subseteq \operatorname{im} \psi_{*,\tau_{\infty}^c}$. On the other hand, it follows easily from Lemma 3.4.13(2,3) that $\operatorname{im} \psi_{*,\tau_{\infty}^c}$ has rank $\lceil \frac{N}{2} \rceil$. Thus, we must have $H_1 = \operatorname{im} \psi_{*,\tau_{\infty}^c}$. The above two claims together with Theorem 5.3.4(3) imply (2).

Remark 5.4.4. It follows from the proof of Theorem 5.4.3 that for every $s^{\dagger} \in S^{\dagger}_{\mathfrak{p}}(V^{\circ}, K^{p\circ})(\kappa)$, if we put $s^{\circ} \coloneqq s^{\dagger \circ}(s^{\dagger})$ and $s^{\bullet} \coloneqq s^{\dagger \bullet}(s^{\dagger})$, then

- (1) the morphism $\zeta_{s^{\circ}}^{\circ} \circ b^{\dagger \circ} \circ (\zeta_{s^{\dagger}}^{\dagger})^{-1}$ identifies $\mathbb{P}(\mathscr{V}_{s^{\dagger}})$ as a closed subscheme of $\mathbb{P}(\mathscr{V}_{s^{\circ}})$ induced by the obvious κ -linear (surjective) map $\mathscr{V}_{s^{\circ}} \to \mathscr{V}_{s^{\dagger}}$; and
- (2) the morphism $\zeta_{s^{\bullet}}^{\bullet} \circ b^{\dagger \bullet} \circ (\zeta_{s^{\dagger}}^{\dagger})^{-1}$ identifies $\mathbb{P}(\mathscr{V}_{s^{\dagger}})$ as a closed subscheme (of codimension one) of $\mathrm{DL}^{\bullet}(\mathscr{V}_{s^{\bullet}}, \{, \}_{s^{\bullet}})$ defined by the condition $H_1 = H_1^{\dashv}$.

Construction 5.4.5. Put $K_p^{\dagger} := K_p^{\circ} \cap K_p^{\bullet}$. Similar to Construction 4.3.2, we construct a *uniformization map*

(5.11)
$$\upsilon^{\dagger} \colon \mathrm{S}_{\mathfrak{p}}^{\dagger}(\mathrm{V}^{\circ}, -)(\overline{\mathbb{F}}_{p}) \xrightarrow{\sim} \mathrm{Sh}(\mathrm{V}^{\circ}, -\mathrm{K}_{p}^{\dagger}) \times \mathrm{T}_{\mathfrak{p}}(\overline{\mathbb{F}}_{p})$$

in $\operatorname{Fun}(\mathfrak{K}(\mathcal{V}^{\circ})^{p} \times \mathfrak{T}, \operatorname{Set})_{/\operatorname{T}_{\mathfrak{p}}(\overline{\mathbb{F}}_{p})}$ which is an isomorphism, under which the induced action of $\operatorname{Gal}(\overline{\mathbb{F}}_{p}/\mathbb{F}_{p}^{\Phi})$ on the target is trivial on $\operatorname{Sh}(\mathcal{V}^{\circ}, -\mathcal{K}_{p}^{\dagger})$.

5.5. Cohomology of link stratum. In this subsection, we study the cohomology of the link stratum. We assume $N \ge 2$.

We first construct certain Hecke correspondences for $B^{\circ}_{\mathfrak{p}}(V^{\circ}, -)$ extending Construction 5.2.6. Unlike the functor $S^{\circ}_{\mathfrak{p}}(V^{\circ}, -)$, the natural action of $K^{\circ}_{\mathfrak{p}} = U(\Lambda^{\circ}_{\mathfrak{p}})(O_{F^+_{\mathfrak{p}}})$ on the functor $B^{\circ}_{\mathfrak{p}}(V^{\circ}, -)$ is nontrivial. However, as we will see, such action factors through the quotient $U(\Lambda^{\circ}_{\mathfrak{p}})(O_{F^+_{\mathfrak{p}}}) \to U(\Lambda^{\circ}_{\mathfrak{p}})(\mathbb{F}_p)$. Let $K^{\circ}_{\mathfrak{p}1}$ be the kernel of the reduction map $K^{\circ}_{\mathfrak{p}} = U(\Lambda^{\circ}_{\mathfrak{p}})(O_{F^+_{\mathfrak{p}}}) \to U(\Lambda^{\circ}_{\mathfrak{p}})(\mathbb{F}_p)$.

Construction 5.5.1. We first define a functor

$$\begin{split} \mathbf{S}^{\circ}_{\mathfrak{p}1}(\mathbf{V}^{\circ},-) \colon \mathfrak{K}(\mathbf{V}^{\circ})^{p} \times \mathfrak{T} &\to \mathsf{PSch}'_{/\mathbb{F}^{\Phi}_{p}} \\ \mathbf{K}^{p\circ} &\mapsto \mathbf{S}^{\circ}_{\mathfrak{p}}(\mathbf{V}^{\circ},\mathbf{K}^{p\circ}) \end{split}$$

such that for every $S \in \mathsf{Sch}'_{\mathbb{F}^{\Phi}_{p}}, S^{\circ}_{\mathfrak{p}1}(\mathcal{V}^{\circ}, \mathcal{K}^{p\circ})(S)$ is the set of equivalence classes of septuples $(A_{0}, \lambda_{0}, \eta^{p}_{0}; A^{\circ}, \lambda^{\circ}, \eta^{p\circ}; \eta^{\circ}_{\mathfrak{p}})$ where

O $(A_0, \lambda_0, \eta_0^p; A^{\circ}, \lambda^{\circ}, \eta^{p \circ})$ is an element in $S^{\circ}_{\mathfrak{p}}(V^{\circ}, K^{p \circ})(S);$

O $\eta_{\mathfrak{p}}^{\circ}$ is, for a chosen geometric point s on every connected component of S, an isomorphism

$$\eta_{\mathfrak{p}}^{\circ} \colon \Lambda_{\mathfrak{p}}^{\circ} \otimes \mathbb{F}_{p} \to \operatorname{Hom}_{O_{F}}(A_{0s}[\mathfrak{p}], A_{s}^{\circ}[\mathfrak{p}])$$

of hermitian spaces over $O_{F_{\mathfrak{p}}} \otimes \mathbb{F}_p$, where $\operatorname{Hom}_{O_F}(A_{0s}[\mathfrak{p}], A_s^{\circ}[\mathfrak{p}])$ is equipped with the hermitian form constructed similarly as in Construction 3.4.4 with respect to $(\lambda_0, \lambda^{\circ})$.

The equivalence relation and the action of morphisms in $\mathfrak{K}(\mathcal{V}^\circ)^p \times \mathfrak{T}$ are defined similarly as in Definition 4.1.2. In fact, we have a further action of $U(\Lambda^\circ_{\mathfrak{p}})(\mathbb{F}_p)$ on $S^\circ_{\mathfrak{p}1}(\mathcal{V}^\circ, -)$. Moreover, similar to Construction 4.3.5 and Proposition 4.3.6, for every $g \in K^\circ_{\mathfrak{p}1} \setminus U(\mathcal{V}^\circ)(F^+_{\mathfrak{p}})/K^\circ_{\mathfrak{p}1}$, we may construct the Hecke correspondence

(5.12)
$$\operatorname{Hk}_{g} \colon \operatorname{S}^{\circ}_{\mathfrak{p}1}(\operatorname{V}^{\circ}, -)_{g} \to \operatorname{S}^{\circ}_{\mathfrak{p}1}(\operatorname{V}^{\circ}, -) \times \operatorname{S}^{\circ}_{\mathfrak{p}1}(\operatorname{V}^{\circ}, -)$$

as a morphism in $\mathsf{Fun}(\mathfrak{K}(\mathcal{V}^\circ)^p \times \mathfrak{T}, \mathsf{Sch}_{/\mathbb{F}_p^\Phi})_{/\mathrm{T}_p}$ that is finite étale.

On the other hand, Theorem 5.2.4 implies that we have a canonical isomorphism

$$\mathrm{B}^{\circ}_{\mathfrak{p}}(\mathrm{V}^{\circ},-)\simeq \mathrm{S}^{\circ}_{\mathfrak{p}1}(\mathrm{V}^{\circ},-) \overset{\mathrm{U}(\Lambda^{\circ}_{\mathfrak{p}})(\mathbb{F}_{p})}{\times} \mathbb{P}(\Lambda^{\circ}_{\mathfrak{p}}\otimes \mathbb{F}_{p})$$

in the category $\operatorname{\mathsf{Fun}}(\mathfrak{K}(\mathcal{V}^{\circ})^{p} \times \mathfrak{T}, \operatorname{\mathsf{Sch}}_{/\mathbb{F}_{p}^{\Phi}})_{/\mathrm{T}_{\mathfrak{p}}}$. Thus, for every $g \in \mathrm{K}_{\mathfrak{p}1}^{\circ} \setminus \mathrm{U}(\mathcal{V}^{\circ})(F_{\mathfrak{p}}^{+})/\mathrm{K}_{\mathfrak{p}1}^{\circ}$, we obtain from (5.12) the Hecke correspondence

$$\mathrm{Hk}_g \colon \mathrm{B}^{\circ}_{\mathfrak{p}}(\mathrm{V}^{\circ}, -)_g \to \mathrm{B}^{\circ}_{\mathfrak{p}}(\mathrm{V}^{\circ}, -) \times \mathrm{B}^{\circ}_{\mathfrak{p}}(\mathrm{V}^{\circ}, -)$$

as a morphism in $\operatorname{\mathsf{Fun}}(\mathfrak{K}(\mathcal{V}^\circ)^p \times \mathfrak{T}, \operatorname{\mathsf{Sch}}_{/\mathbb{F}_n^\Phi})_{/\mathrm{T}_p}$ that is finite étale.

Now we study cohomology.

Lemma 5.5.2. Consider a p-coprime coefficient ring L.

(1) If p + 1 is invertible in L, then the restriction map

$$(\mathbf{m}^{\dagger \circ})^* \colon \mathrm{H}^{i}_{\mathfrak{T}}(\overline{\mathrm{M}}^{\circ}_{\mathfrak{p}}(\mathrm{V}^{\circ}, \mathbf{-}), L) \to \mathrm{H}^{i}_{\mathfrak{T}}(\overline{\mathrm{M}}^{\dagger}_{\mathfrak{p}}(\mathrm{V}^{\circ}, \mathbf{-}), L)$$

is an isomorphism for every integer $i \notin \{N-2, 2N-2\}$. In particular, $\mathrm{H}^{i}_{\mathfrak{T}}(\overline{\mathrm{M}}^{\circ}_{\mathfrak{p}}(\mathrm{V}^{\circ}, -), L)$ and $\mathrm{H}^{i}_{\mathfrak{T}}(\overline{\mathrm{M}}^{\dagger}_{\mathfrak{p}}(\mathrm{V}^{\circ}, -), L)$ vanish if *i* is odd and different from N-2.

- (2) For every $i \in \mathbb{Z}$, both $\mathrm{H}^{i}_{\mathfrak{T}}(\overline{\mathrm{M}}^{\circ}_{\mathfrak{p}}(\mathrm{V}^{\circ}, -), L)$ and $\mathrm{H}^{i}_{\mathfrak{T}}(\overline{\mathrm{M}}^{\dagger}_{\mathfrak{p}}(\mathrm{V}^{\circ}, -), L)$ are free L-modules.
- (3) When N is even, the action of $\operatorname{Gal}(\overline{\mathbb{F}}_p/\mathbb{F}_p^{\Phi})$ on $\operatorname{H}^{N-2}_{\mathfrak{T}}(\overline{\operatorname{M}}_{\mathfrak{p}}^{\dagger}(V^{\circ}, -), L(\frac{N-2}{2}))$ is trivial.

Proof. By Theorem 5.2.4, for every $K^{p\circ} \in \mathfrak{K}(V^{\circ})^p$ and every $s^{\circ} \in S^{\circ}_{\mathfrak{p}}(V^{\circ}, K^{p\circ})(\overline{\mathbb{F}}_p)$, the restriction of $(\mathbf{m}^{\dagger \circ})^*$ to the fibers over s° is a morphism appeared in Lemma A.1.4.

Part (1) then follows from Lemma A.1.4(2). Part (2) follows from Lemma A.1.4(3). Part (3) follows from Lemma A.1.4(4) and Construction 5.2.6. \Box

Definition 5.5.3. Let $\xi \in \mathrm{H}^{2}_{\mathfrak{X}}(\overline{\mathrm{B}}^{\circ}_{\mathfrak{p}}(\mathrm{V}^{\circ}, -), L(1))$ be the first Chern class of the tautological quotient line bundle on $\overline{\mathrm{B}}^{\circ}_{\mathfrak{p}}(\mathrm{V}^{\circ}, -)$ (that is, in the situation of Theorem 5.2.4, the restriction of ξ to $\mathrm{B}^{\circ}_{s^{\circ}}$ is isomorphic to $\zeta^{\circ*}_{s^{\circ}}\mathcal{O}_{\mathbb{P}(\mathscr{V}_{s^{\circ}})}(1)$ for every $\mathrm{K}^{p^{\circ}} \in \mathfrak{K}(\mathrm{V}^{\circ})^{p}$ and every $s^{\circ} \in \mathrm{S}^{\circ}_{\mathfrak{p}}(\mathrm{V}^{\circ}, \mathrm{K}^{p^{\circ}})(\overline{\mathbb{F}}_{p})$). We define the *primitive cohomology* $\mathrm{H}^{\mathrm{prim}}(\overline{\mathrm{M}}^{\dagger}_{\mathfrak{p}}(\mathrm{V}^{\circ}, -), L(i))$ to be the kernel of the map

 $\cup (\mathbf{m}^{\dagger \circ *} \iota^{\circ}_{!} \xi) \colon \mathbf{H}^{N-2}_{\mathfrak{T}}(\overline{\mathbf{M}}^{\dagger}_{\mathfrak{p}}(\mathbf{V}^{\circ}, -), L(i)) \to \mathbf{H}^{N}_{\mathfrak{T}}(\overline{\mathbf{M}}^{\dagger}_{\mathfrak{p}}(\mathbf{V}^{\circ}, -), L(i+1)).$

Proposition 5.5.4. Take an object $K^{p\circ} \in \mathfrak{K}(V^{\circ})^{p}$ and a rational prime $\ell \neq p$. Let $\pi^{\infty,p}$ be an irreducible admissible representation of $U(V^{\circ})(\mathbb{A}_{F^{+}}^{\infty,p})$ with coefficients in $\overline{\mathbb{Q}}_{\ell}$ such that $(\pi^{\infty,p})^{K^{p\circ}}$ is a constituent of $H^{\text{prim}}(\overline{M}_{\mathfrak{p}}^{\dagger}(V^{\circ}, K^{p\circ}), \overline{\mathbb{Q}}_{\ell})$. Then one can complete $\pi^{\infty,p}$ to an automorphic representation $\pi = \pi^{\infty,p} \otimes \pi_{\infty} \otimes \prod_{\mathfrak{q}|p} \pi_{\mathfrak{q}}$ of $U(V^{\circ})(\mathbb{A}_{F^{+}})$ such that π_{∞} is trivial; $\pi_{\mathfrak{q}}$ is unramified for $\mathfrak{q} \neq \mathfrak{p}$; and

- (1) when N is even, π_{p} is a constituent of an unramified principal series;
- (2) when N is odd, $BC(\pi_p)$ is a constituent of an unramified principal series of $GL_N(F_p)$ whose Satake parameter contains $\{-p, -p^{-1}\}$.

Proof. Put $\mathrm{K}_{p1}^{\circ} \coloneqq \mathrm{K}_{\mathfrak{p}1}^{\circ} \times \prod_{\mathfrak{q}|p,\mathfrak{q}\neq\mathfrak{p}} \mathrm{K}_{\mathfrak{q}}^{\circ}$. By Construction 5.5.1, the cohomology $\mathrm{H}_{\mathfrak{T}}^{N-2}(\overline{\mathrm{M}}_{\mathfrak{p}}^{\dagger}(\mathrm{V}^{\circ},\mathrm{K}^{p\circ}),\overline{\mathbb{Q}}_{\ell})$ is an $\overline{\mathbb{Q}}_{\ell}[\mathrm{K}^{p\circ}\mathrm{K}_{p1}^{\circ}] \cdot \mathrm{V}(\mathrm{V}^{\circ})(\mathbb{A}_{F^{+}}^{\infty})/\mathrm{K}^{p\circ}\mathrm{K}_{p1}^{\circ}]$ -module for which $\mathrm{H}^{\mathrm{prim}}(\overline{\mathrm{M}}_{\mathfrak{p}}^{\dagger}(\mathrm{V}^{\circ},\mathrm{K}^{p\circ}),\overline{\mathbb{Q}}_{\ell})$ is a submodule.

In the uniformization map (5.4), we let $s_0 \in S^{\circ}_{\mathfrak{p}}(V^{\circ}, K^{p\circ})(\overline{\mathbb{F}}_p)$ be the point corresponding to the unit element on the right-hand side. Put

$$\mathrm{H}^{\mathrm{prim}}_{s_{0}}(\overline{\mathrm{M}}^{\dagger}_{\mathfrak{p}}(\mathrm{V}^{\circ},\mathrm{K}^{p\circ}),\overline{\mathbb{Q}}_{\ell}) \coloneqq \mathrm{H}^{\mathrm{prim}}(\overline{\mathrm{M}}^{\dagger}_{\mathfrak{p}}(\mathrm{V}^{\circ},\mathrm{K}^{p\circ}),\overline{\mathbb{Q}}_{\ell}) \bigcap \mathrm{H}^{N-2}(\overline{\mathrm{M}}^{\dagger}_{\mathfrak{p}}(\mathrm{V}^{\circ},\mathrm{K}^{p\circ})\cap \pi^{\circ}(s_{0}),\overline{\mathbb{Q}}_{\ell}).$$

Then $\mathrm{H}^{\mathrm{prim}}_{s_0}(\overline{\mathrm{M}}^{\dagger}_{\mathfrak{p}}(\mathrm{V}^{\circ},\mathrm{K}^{p\circ}),\overline{\mathbb{Q}}_{\ell})$ is a representation of $\mathrm{U}(\Lambda^{\circ}_{\mathfrak{p}})(\mathbb{F}_p) = \mathrm{K}^{\circ}_{\mathfrak{p}}/\mathrm{K}^{\circ}_{\mathfrak{p}1}$, which is (isomorphic to) the representation Ω_N studied in Subsection C.2. Then we may identify $\mathrm{H}^{\mathrm{prim}}(\overline{\mathrm{M}}^{\dagger}_{\mathfrak{p}}(\mathrm{V}^{\circ},\mathrm{K}^{p\circ}),\overline{\mathbb{Q}}_{\ell})$ with

(5.13)
$$\operatorname{Map}_{\mathrm{K}_{\mathfrak{p}}^{\circ}}\left(\mathrm{U}(\mathrm{V}^{\circ})(F^{+})\backslash\mathrm{U}(\mathrm{V}^{\circ})(\mathbb{A}_{F^{+}}^{\infty})/\mathrm{K}^{p\circ}\prod_{\mathfrak{q}\mid p,\mathfrak{q\neq p}}\mathrm{K}_{\mathfrak{q}}^{\circ},\Omega_{N}\right)$$

as $\overline{\mathbb{Q}}_{\ell}[\mathrm{K}^{p\circ}\mathrm{K}^{\circ}_{p1}\backslash\mathrm{U}(\mathrm{V}^{\circ})(\mathbb{A}^{\infty}_{F^{+}})/\mathrm{K}^{p\circ}\mathrm{K}^{\circ}_{p1}]$ -modules. It is well-known (see, for example, [HM78]) that the representation Ω_N is irreducible and determined by N up to isomorphism. Therefore, (5.13) is a submodule of Map(U(V^{\circ})(F^{+})\backslash\mathrm{U}(\mathrm{V}^{\circ})(\mathbb{A}^{\infty}_{F^{+}})/\mathrm{K}^{p\circ}\mathrm{K}^{\circ}_{p1},\overline{\mathbb{Q}}_{\ell}). In particular, we can complete $\pi^{\infty,p}$ to
an automorphic representation $\pi = \pi^{\infty,p} \otimes \pi_{\infty} \otimes \prod_{\mathfrak{q}|p} \pi_{\mathfrak{q}}$ of $U(V^{\circ})(\mathbb{A}_{F^+})$ such that π_{∞} is trivial; $\pi_{\mathfrak{q}}$ is unramified for $\mathfrak{q} \neq \mathfrak{p}$; and $\pi_{\mathfrak{p}}|_{K^{\circ}_{\mathfrak{p}}}$ contains Ω_N .

In case (1), by Proposition C.2.1(2), we know that Ω_N has nonzero Borel fixed vectors. Thus, $\pi_{\mathfrak{p}}$ is a constituent of an unramified principal series.

In case (2), we first consider the case where N = 3. As $\pi_{\mathfrak{p}}|_{K_{\mathfrak{p}}^{\circ}}$ contains Ω_3 , it has to be c-Ind_{K_3}^{U3} Ω_3 by Proposition C.2.1(3) and [MP96, Theorem 6.11(2)]. Thus, by [MP96, Proposition 6.6], $\pi_{\mathfrak{p}}|_{K_{\mathfrak{p}}^{\circ}}$ is irreducible supercuspidal, which is actually the unique supercuspidal unipotent representation of U(V°)($F_{\mathfrak{p}}^+$). In fact, c-Ind_{K_3}^{U3} Ω_3 is the representation $\pi^s(1)$ appeared in [Rog90, Proposition 13.1.3(d)], after identifying $\overline{\mathbb{Q}}_{\ell}$ with \mathbb{C} . By [Rog90, Proposition 13.2.2(c)], BC($\pi^s(1)$) is the tempered constituent of the unramified principal series of GL₃($F_{\mathfrak{p}}$) with the Satake parameter $\{-p, 1, -p^{-1}\}$. Now for general N = 2r + 1, as $\pi_{\mathfrak{p}}|_{K_{\mathfrak{p}}^{\circ}}$ contains Ω_N , by Proposition C.2.1(4) and [MP96, Theorem 6.11(2)], $\pi_{\mathfrak{p}}$ is a constituent the normalized parabolic induction of $\pi^s(1) \boxtimes \chi_1 \boxtimes \cdots \boxtimes \chi_{r-1}$ for some unramified characters $\chi_1, \ldots, \chi_{r-1}$ of F^{\times} . Therefore, by the compatibility of local base change and induction, BC($\pi_{\mathfrak{p}}$) is a constituent of an unramified principal series of GL_N($F_{\mathfrak{p}}$) whose Satake parameter contains $\{-p, -p^{-1}\}$.

The proposition is proved.

5.6. Intersection on ground stratum. In this subsection, we describe certain schemetheoretical intersection on the ground stratum, which will be used in the next subsection. We assume $N \ge 2$.

Take an object $\mathbf{K}^{p\circ} \in \mathfrak{K}(\mathbf{V}^{\circ})^{p}$. Given two (possibly same) points $s_{1}^{\bullet}, s_{2}^{\bullet} \in \mathbf{S}_{\mathfrak{p}}^{\bullet}(\mathbf{V}^{\circ}, \mathbf{K}^{p\circ})(\kappa)$ for a perfect field κ containing \mathbb{F}_{p}^{Φ} , we put

$$\mathbf{B}^{\bullet}_{s_{1}^{\bullet},s_{2}^{\bullet}} \coloneqq \mathbf{B}^{\bullet}_{s_{1}^{\bullet}} \times_{\mathbf{M}^{\bullet}_{\mathfrak{p}}(\mathbf{V}^{\circ},\mathbf{K}^{p\circ})_{\kappa}} \mathbf{B}^{\bullet}_{s_{2}^{\bullet}}$$

as the (possibly empty) fiber product of $\iota_{\kappa}^{\bullet} | B_{s_1^{\bullet}}^{\bullet}$ and $\iota_{\kappa}^{\bullet} | B_{s_2^{\bullet}}^{\bullet}$. To describe $B_{s_1^{\bullet},s_2^{\bullet}}^{\bullet}$, we need to use some particular cases of the Hecke correspondences introduced in Construction 5.3.6. We now give more details.

Definition 5.6.1. For every integer $0 \le j \le N$, we define a functor

$$\begin{aligned} \mathbf{S}^{\bullet}_{\mathfrak{p}}(\mathbf{V}^{\circ},-)_{j} \colon \mathfrak{K}(\mathbf{V}^{\circ})^{p} \times \mathfrak{T} &\to \mathsf{PSch}'_{/\mathbb{F}^{\Phi}_{p}} \\ \mathbf{K}^{p\circ} &\mapsto \mathbf{S}^{\bullet}_{\mathfrak{n}}(\mathbf{V}^{\circ},\mathbf{K}^{p\circ})_{j} \end{aligned}$$

such that for every $S \in \mathsf{Sch}'_{/\mathbb{F}_p^{\Phi}}, \ \mathsf{S}_p^{\bullet}(\mathsf{V}^\circ, \mathsf{K}^{p\circ})_j(S)$ is the set of equivalence classes of decuples $(A_0, \lambda_0, \eta_0^p; A_1^{\bullet}, \lambda_1^{\bullet}, \eta_1^{p\bullet}; A_2^{\bullet}, \lambda_2^{\bullet}, \eta_2^{p\bullet}; \phi^{\bullet})$ where

- O $(A_0, \lambda_0, \eta_0^p; A_i^{\bullet}, \lambda_i^{\bullet}, \eta_i^{p\bullet})$ for i = 1, 2 are two elements in $S_{\mathfrak{p}}^{\bullet}(V^{\circ}, K^{p\circ})(S)$; and
- $O \phi^{\bullet} \colon A_1^{\bullet} \to A_2^{\bullet}$ is an O_F -linear quasi-isogeny such that
 - (a) $p\phi^{\bullet} \circ \lambda_1^{\bullet-1}$ is a quasi-*p*-isogeny; and ker $(p\phi^{\bullet})[\mathfrak{p}]$ has rank $p^{2(N-j)}$;
 - (b) $\phi^{\bullet}[\mathfrak{q}^{\infty}]$ is an isomorphism for every prime \mathfrak{q} of F^{+} above p that is not \mathfrak{p} ;
 - (c) we have $\phi^{\bullet \vee} \circ \lambda_2^{\bullet} \circ \phi^{\bullet} = \lambda_1^{\bullet}$; and
 - (d) the K^{*p*°}-orbit of maps $v \mapsto \phi^{\bullet}_* \circ \eta_1^{p\bullet}(v)$ for $v \in V^{\circ} \otimes_{\mathbb{Q}} \mathbb{A}^{\infty,p}$ coincides with $\eta_2^{p\bullet}$.

The equivalence relation and the action of morphisms in $\mathfrak{K}(\mathcal{V}^{\circ})^{p} \times \mathfrak{T}$ are defined similarly as in Definition 4.2.3. Finally, we denote

$$\mathrm{Hk}_{j} \colon \mathrm{S}_{\mathfrak{p}}^{\bullet}(\mathrm{V}^{\circ},-)_{j} \to \mathrm{S}_{\mathfrak{p}}^{\bullet}(\mathrm{V}^{\circ},-) \times \mathrm{S}_{\mathfrak{p}}^{\bullet}(\mathrm{V}^{\circ},-)$$

the morphism in $\mathsf{Fun}(\mathfrak{K}(\mathcal{V}^{\circ})^{p} \times \mathfrak{T}, \mathsf{Sch}_{/\mathbb{F}_{p}^{\Phi}})_{/\mathrm{T}_{\mathfrak{p}}}$ induced by the assignment

 $(A_0,\lambda_0,\eta_0^p;A_1^{\bullet},\lambda_1^{\bullet},\eta_1^{p\bullet};A_2^{\bullet},\lambda_2^{\bullet},\eta_2^{p\bullet};\phi^{\bullet})\mapsto((A_0,\lambda_0,\eta_0^p;A_1^{\bullet},\lambda_1^{\bullet},\eta_1^{p\bullet}),(A_0,\lambda_0,\eta_0^p;A_2^{\bullet},\lambda_2^{\bullet},\eta_2^{p\bullet})).$

Remark 5.6.2. When $K^{p\circ}$ is sufficiently small, the morphism

$$\mathrm{Hk}_j \colon \mathrm{S}^{\bullet}_{\mathfrak{p}}(\mathrm{V}^{\circ}, \mathrm{K}^{p\circ})_j \to \mathrm{S}^{\bullet}_{\mathfrak{p}}(\mathrm{V}^{\circ}, \mathrm{K}^{p\circ}) \times \mathrm{S}^{\bullet}_{\mathfrak{p}}(\mathrm{V}^{\circ}, \mathrm{K}^{p\circ})$$

is a closed immersion for every j; and the images of Hk_i for all j are mutually disjoint.

Now we take a point $s^{\bullet} = (A_0, \lambda_0, \eta_0^p; A_1^{\bullet}, \lambda_1^{\bullet}, \eta_1^{p^{\bullet}}; A_2^{\bullet}, \lambda_2^{\bullet}, \eta_2^{p^{\bullet}}; \phi^{\bullet}) \in \mathcal{S}_{\mathfrak{p}}^{\bullet}(\mathcal{V}^{\circ}, \mathcal{K}^{p^{\circ}})_j(\kappa)$ where κ is a perfect field containing \mathbb{F}_p^{Φ} . By Definition 5.6.1(c), we have $(p\phi^{\bullet} \circ \lambda_1^{\bullet^{-1}})^{\vee} = p\phi^{\bullet^{-1}} \circ \lambda_2^{\bullet^{-1}}$. Thus, $p\phi^{\bullet^{-1}} \circ \lambda_2^{\bullet^{-1}}$ hence $p\phi^{\bullet^{-1}}$ are quasi-*p*-isogenies as well. In particular, for every $\tau \in \Sigma_{\infty}$, we may consider

$$\ker(p\phi^{\bullet})_{*,\tau} := \ker\left((p\phi^{\bullet})_{*,\tau} \colon \mathrm{H}_{1}^{\mathrm{dR}}(A_{1}^{\bullet}/\kappa)_{\tau} \to \mathrm{H}_{1}^{\mathrm{dR}}(A_{2}^{\bullet}/\kappa)_{\tau}\right),$$
$$\operatorname{im}(p\phi^{\bullet-1})_{*,\tau} := \operatorname{im}\left((p\phi^{\bullet-1})_{*,\tau} \colon \mathrm{H}_{1}^{\mathrm{dR}}(A_{2}^{\bullet}/\kappa)_{\tau} \to \mathrm{H}_{1}^{\mathrm{dR}}(A_{1}^{\bullet}/\kappa)_{\tau}\right).$$

Lemma 5.6.3. We have

(1) $\operatorname{im}(p\phi^{\bullet-1})_{*,\tau} \subseteq \operatorname{ker}(p\phi^{\bullet})_{*,\tau}$ for every $\tau \in \Sigma_{\infty}$; (2) $\operatorname{dim}_{\kappa} \operatorname{ker}(p\phi^{\bullet})_{*,\tau} = N - j$ for $\tau \in \{\tau_{\infty}, \tau_{\infty}^{\mathsf{c}}\}$; (3) $\operatorname{im}(p\phi^{\bullet-1})_{*,\tau} \cap \operatorname{H}_{1}^{\operatorname{dR}}(A_{1}^{\bullet}/\kappa)_{\tau^{\mathsf{c}}}^{\perp} = 0$ for $\tau \in \{\tau_{\infty}, \tau_{\infty}^{\mathsf{c}}\}$; (4) $(\operatorname{im}(p\phi^{\bullet-1})_{*,\tau})^{\perp} = \operatorname{ker}(p\phi^{\bullet})_{*,\tau^{\mathsf{c}}}$ for $\tau \in \{\tau_{\infty}, \tau_{\infty}^{\mathsf{c}}\}$; and (5) $\operatorname{dim}_{\kappa} \operatorname{im}(p\phi^{\bullet-1})_{*,\tau} = j$ for $\tau \in \{\tau_{\infty}, \tau_{\infty}^{\mathsf{c}}\}$. In particular, $\operatorname{S}_{p}^{\bullet}(\operatorname{V}^{\circ}, \operatorname{K}^{p\circ})_{j}$ is empty if $j > \lfloor \frac{N}{2} \rfloor$.

In particular, $S_{\mathfrak{p}}(\mathbf{V}, \mathbf{K})_j$ is empty if $j \ge \lfloor \frac{1}{2} \rfloor$.

Proof. For (1), it is obvious since $(p\phi^{\bullet}) \circ (p\phi^{\bullet-1}) = p^2$.

For (2), by Definition 5.6.1(a), we have $\dim_{\kappa} \ker(p\phi^{\bullet})_{*,\tau_{\infty}} + \dim_{\kappa} \ker(p\phi^{\bullet})_{*,\tau_{\infty}^{c}} = 2(N-j)$. Using the isomorphisms $\mathbb{V}: \operatorname{H}_{1}^{\mathrm{dR}}(A_{1}^{\bullet}/\kappa)_{\tau_{\infty}} \to \operatorname{H}_{1}^{\mathrm{dR}}(A_{1}^{\bullet}/\kappa)_{\tau_{\infty}^{c}}$ and $\mathbb{V}: \operatorname{H}_{1}^{\mathrm{dR}}(A_{2}^{\bullet}/\kappa)_{\tau_{\infty}} \to \operatorname{H}_{1}^{\mathrm{dR}}(A_{2}^{\bullet}/\kappa)_{\tau_{\infty}^{c}}$, we have $\dim_{\kappa} \ker(p\phi^{\bullet})_{*,\tau_{\infty}} = \dim_{\kappa} \ker(p\phi^{\bullet})_{*,\tau_{\infty}^{c}}$ hence both equal to N-j.

For (3), it suffices to consider $\tau = \tau_{\infty}$ due to the isomorphism V. Via ϕ^{\bullet} , we regard $\mathcal{D}(A_{2}^{\bullet})$ as a lattice in $\mathcal{D}(A_{1}^{\bullet})_{\mathbb{Q}}$. By Definition 5.6.1(a), we have $p\mathcal{D}(A_{2}^{\bullet})_{\tau_{\infty}} \subseteq \mathcal{D}(A_{1}^{\bullet})_{\tau_{\infty}} \subseteq \mathcal{D}(A_{2}^{\bullet})_{\tau_{\infty}^{\bullet}}^{\vee}$ (Notation 3.4.12). Suppose $\mathrm{H}_{1}^{\mathrm{dR}}(A_{1}^{\bullet}/\kappa)_{\tau_{\infty}^{\bullet}}^{\perp} \subseteq \mathrm{im}(p\phi^{\bullet-1})_{*,\tau_{\infty}}$. Then one can find $x_{2} \in \mathcal{D}(A_{2}^{\bullet})_{\tau_{\infty}}$ and $x_{1} \in$ $\mathcal{D}(A_{1}^{\bullet})_{\tau_{\infty}^{\bullet}}^{\vee} \setminus \mathcal{D}(A_{1}^{\bullet})_{\tau_{\infty}}$ such that $px_{1} = px_{2}$. It follows that $\langle x_{2}, \mathrm{V}x_{2} \rangle_{\lambda_{2}^{\bullet},\tau_{\infty}} = \langle x_{1}, \mathrm{V}x_{1} \rangle_{\lambda_{1}^{\bullet},\tau_{\infty}}$ does not belong to $W(\kappa)$, which is a contradiction. Here, we regard V as Verschiebung maps on for Dieudonné modules of A_{1}^{\bullet} and A_{2}^{\bullet} , which are isomorphisms.

For (4), as $\lambda_1^{\bullet} \circ \phi^{\bullet-1} = \phi^{\bullet \vee} \circ \lambda_2^{\bullet}$, we have for $\tau \in \{\tau_{\infty}, \tau_{\infty}^{\circ}\}$ that

$$(\mathrm{im}(p\phi^{\bullet-1})_{*,\tau})^{\perp} = ((p\phi^{\bullet})_{*,\tau^{c}})^{-1}\mathrm{H}_{1}^{\mathrm{dR}}(A_{1}^{\bullet}/\kappa)_{\tau}^{\perp},$$

which equals $\ker(p\phi^{\bullet})_{*,\tau^{c}}$ by (3).

For (5), by (2,3,4), we have $\dim_{\kappa} \operatorname{im}(p\phi^{\bullet-1})_{*,\tau} = j$ for $\tau \in \{\tau_{\infty}, \tau_{\infty}^{\mathsf{c}}\}$. The last claim follows from (1,2,5).

By Lemma 5.6.3(1,4), for $\tau \in \{\tau_{\infty}, \tau_{\infty}^{c}\}$, we may put

$$\mathrm{H}_{1}^{\mathrm{dR}}(\phi^{\bullet})_{\tau} \coloneqq \frac{\mathrm{ker}(p\phi^{\bullet})_{*,\tau}}{\mathrm{im}(p\phi^{\bullet-1})_{*,\tau}};$$

and we have the induced κ -bilinear pairing

$$\langle , \rangle_{\lambda_1^{\bullet}, \tau_{\infty}} \colon \mathrm{H}_1^{\mathrm{dR}}(\phi^{\bullet})_{\tau_{\infty}} \times \mathrm{H}_1^{\mathrm{dR}}(\phi^{\bullet})_{\tau_{\infty}^{\mathsf{c}}} \to \kappa.$$

On the other hand, the (κ, σ^{-1}) -linear Verschiebung map \mathbb{V} : $\mathrm{H}_{1}^{\mathrm{dR}}(A_{1}^{\bullet}/\kappa)_{\tau_{\infty}} \to \mathrm{H}_{1}^{\mathrm{dR}}(A_{1}^{\bullet}/\kappa)_{\tau_{\infty}^{c}}$ induces a (κ, σ^{-1}) -linear isomorphism \mathbb{V} : $\mathrm{H}_{1}^{\mathrm{dR}}(\phi^{\bullet})_{\tau_{\infty}} \to \mathrm{H}_{1}^{\mathrm{dR}}(\phi^{\bullet})_{\tau_{\infty}^{c}}$. We define a pairing

 $\{\ ,\ \}_{s^{\bullet}} \colon \mathrm{H}^{\mathrm{dR}}_{1}(\phi^{\bullet})_{\tau^{\mathsf{c}}_{\infty}} \times \mathrm{H}^{\mathrm{dR}}_{1}(\phi^{\bullet})_{\tau^{\mathsf{c}}_{\infty}} \to \kappa$

by the formula $\{x, y\}_{s^{\bullet}} := \langle \mathbb{V}^{-1}x, y \rangle_{\lambda_1^{\bullet}, \tau_{\infty}}$. To ease notation, we put

$$\mathscr{V}_{s^{\bullet}} \coloneqq \mathrm{H}_{1}^{\mathrm{dR}}(\phi^{\bullet})_{\tau_{\infty}^{\mathsf{c}}}.$$

Lemma 5.6.4. Suppose $j \leq \lfloor \frac{N}{2} \rfloor - 1$. The pair $(\mathscr{V}_{s^{\bullet}}, \{ , \}_{s^{\bullet}})$ is admissible of rank N - 2j(Definition A.1.1) satisfying $\dim_{\kappa} \mathscr{V}_{s^{\bullet}}^{\dashv} = N - 2\lfloor \frac{N}{2} \rfloor$. In particular, we have the geometrically irreducible smooth projective scheme $\mathrm{DL}^{\bullet}(\mathscr{V}_{s^{\bullet}}, \{ , \}_{s^{\bullet}}) \in \mathrm{Sch}_{/\kappa}$ of dimension $\lfloor \frac{N}{2} \rfloor - j$ as introduced in Definition A.2.1.

Proof. By Lemma 5.6.3(2,5), we have $\dim_{\kappa} \mathscr{V}_{s^{\bullet}} = N-2j$. By Lemma 5.6.3(3,4), we have $\dim_{\kappa} \mathscr{V}_{s^{\bullet}} = N-2\lfloor \frac{N}{2} \rfloor$. The lemma follows by Proposition A.2.2.

Now consider a connected scheme $S \in \mathsf{Sch}'_{\kappa}$ and a point $x \in \mathrm{B}^{\bullet}_{s_1,s_2^{\bullet}}(S)$ represented by a quattuordecuple $(A_0, \lambda_0, \eta_0^p; A, \lambda, \eta^p; A_1^{\bullet}, \lambda_1^{\bullet}, \eta_1^{p^{\bullet}}; \gamma_1; A_2^{\bullet}, \lambda_2^{\bullet}, \eta_2^{p^{\bullet}}; \gamma_2)$.

Lemma 5.6.5. There exists a unique integer j satisfying $0 \leq j \leq \lfloor \frac{N}{2} \rfloor - 1$ such that $s^{\bullet} := (A_0, \lambda_0, \eta_0^p; A_1^{\bullet}, \lambda_1^{\bullet}, \eta_1^{p\bullet}; A_2^{\bullet}, \lambda_2^{\bullet}, \eta_2^{p\bullet}; \phi^{\bullet})$ is an element in $S_{\mathfrak{p}}^{\bullet}(V^{\circ}, K^{p\circ})_j(S)$, where $\phi^{\bullet} := \gamma_2 \circ \gamma_1^{-1}$. Moreover, we have

(5.14)
$$\operatorname{im}(p\phi^{\bullet-1})_{*,\tau_{\infty}^{\mathsf{c}}} \subseteq H_2 \subseteq H_1 \subseteq \operatorname{ker}(p\phi^{\bullet})_{*,\tau_{\infty}^{\mathsf{c}}},$$

where $H_2 \subseteq H_1 \subseteq H_1^{\mathrm{dR}}(A_1^{\bullet}/S)_{\tau_{\infty}^{\mathsf{c}}}$ are subbundles in Theorem 5.3.4 for the image of x in $B_{s_1^{\bullet}}^{\bullet}(S)$.

Proof. First, by definition, we have $\ker(p\phi^{\bullet})[\mathfrak{p}] = \ker(\gamma_2 \circ \check{\gamma}_1)[\mathfrak{p}]$, which is an O_F -stable finite flat subgroup of $A_1^{\bullet}[\mathfrak{p}]$. Thus, as S is connected, there is a unique integer j satisfying $0 \le j \le N$ such that $\ker(p\phi^{\bullet})[\mathfrak{p}]$ has rank $p^{2(N-j)}$.

Second, we show that $p\phi^{\bullet} \circ \lambda_1^{\bullet-1}$ is a quasi-*p*-isogeny, that is, $\gamma_2 \circ \check{\gamma}_1 \circ \lambda_1^{\bullet-1}$ is a quasi-*p*-isogeny. By Theorem 5.3.4(3), $\gamma_{1*,\tau_{\infty}^{\bullet}} \omega_{A^{\vee}/S,\tau_{\infty}^{\bullet}}$ contains $\mathrm{H}_1^{\mathrm{dR}}(A_1^{\bullet})_{\tau_{\infty}}^{\perp}$, which implies $\check{\gamma}_{1*,\tau_{\infty}} \mathrm{H}_1^{\mathrm{dR}}(A_1^{\bullet})_{\tau_{\infty}}^{\perp} = 0$. In other words, ker $\lambda_1^{\bullet}[\mathfrak{p}^{\infty}]$ is contained in ker $\check{\gamma}_1[\mathfrak{p}^{\infty}]$. Thus, $\check{\gamma}_1 \circ \lambda_1^{\bullet-1}$ is already a quasi-*p*-isogeny; so is $p\phi^{\bullet} \circ \lambda_1^{\bullet-1}$.

Third, we show that j is at most $\lfloor \frac{N}{2} \rfloor - 1$. (Note that Lemma 5.6.3 already implies that $j \leq \lfloor \frac{N}{2} \rfloor$.) Theorem 5.3.4 implies $\operatorname{rank}_{\mathcal{O}_S} H_2 + 1 = \operatorname{rank}_{\mathcal{O}_S} H_1$ and $\operatorname{H}_1^{\mathrm{dR}}(A_1^{\bullet}/S)_{\tau_{\infty}}^{\perp} \subseteq H_2$. Lemma 5.6.3(3) implies $\operatorname{rank}_{\mathcal{O}_S} H_2 \geq \operatorname{rank}_{\mathcal{O}_S} \operatorname{im}(p\phi^{\bullet-1})_{*,\tau_{\infty}^{\circ}} + 1$. Thus, by Lemma 5.6.3(2,5) and (5.14), we have $(N-j)-j \geq 2$, that is, $j \leq \lfloor \frac{N}{2} \rfloor - 1$.

Definition 5.6.1(b,c,d) are obvious. Thus, it remains to check (5.14). On one hand, we have

$$\operatorname{im}(p\phi^{\bullet-1})_{*,\tau_{\infty}^{\mathsf{c}}} = \operatorname{im}(\gamma_{1} \circ \breve{\gamma}_{2})_{*,\tau_{\infty}^{\mathsf{c}}} = \gamma_{1*,\tau_{\infty}^{\mathsf{c}}}\breve{\gamma}_{2*,\tau_{\infty}^{\mathsf{c}}} \operatorname{H}_{1}^{\mathrm{dR}}(A_{2}^{\bullet}/S)_{\tau_{\infty}^{\mathsf{c}}}$$
$$= \gamma_{1*,\tau_{\infty}^{\mathsf{c}}}\breve{\gamma}_{2*,\tau_{\infty}^{\mathsf{c}}} \omega_{A_{2}^{\bullet\vee}/S,\tau_{\infty}^{\mathsf{c}}} \subseteq \gamma_{1*,\tau_{\infty}^{\mathsf{c}}} \omega_{A_{1}^{\bullet\vee}/S,\tau_{\infty}^{\mathsf{c}}} = H_{2}.$$

On the other hand, since $\check{\gamma}_{1*,\tau_{\infty}} \operatorname{im}(p\phi^{\bullet-1})_{*,\tau_{\infty}} = \check{\gamma}_{1*,\tau_{\infty}} \operatorname{im}(\gamma_{1} \circ \check{\gamma}_{2})_{*,\tau_{\infty}} = 0$, we have the inclusion $\operatorname{im}(p\phi^{\bullet-1})_{*,\tau_{\infty}} \subseteq (\check{\gamma}_{1*,\tau_{\infty}})^{-1}\omega_{A^{\vee},\tau_{\infty}}$. Thus, $H_{1} = ((\check{\gamma}_{1*,\tau_{\infty}})^{-1}\omega_{A^{\vee},\tau_{\infty}})^{\perp}$ is contained in $(\operatorname{im}(p\phi^{\bullet-1})_{*,\tau_{\infty}})^{\perp}$, which is $\operatorname{ker}(p\phi^{\bullet})_{*,\tau_{\infty}^{\circ}}$ by Lemma 5.6.3(4). The lemma is proved. \Box

Definition 5.6.6. By Lemma 5.6.5, we have a morphism

$$\mathbf{B}^{\bullet}_{s_{1}^{\bullet},s_{2}^{\bullet}} \to \coprod_{j=0}^{\lfloor \frac{N}{2} \rfloor - 1} \mathbf{H}\mathbf{k}_{j}^{-1}(s_{1}^{\bullet},s_{2}^{\bullet}).$$

For a point $s^{\bullet} \in \operatorname{Hk}_{j}^{-1}(s_{1}^{\bullet}, s_{2}^{\bullet})(\kappa)$ for some $0 \leq j \leq \lfloor \frac{N}{2} \rfloor - 1$, we denote by $B_{s^{\bullet}}^{\bullet}$ the inverse image under the above morphism, which is an open and closed subscheme of $B_{s_{1}^{\bullet}, s_{2}^{\bullet}}^{\bullet}$.

Theorem 5.6.7. Let $s_1^{\bullet}, s_2^{\bullet} \in S_{\mathfrak{p}}^{\bullet}(\mathcal{V}^{\circ}, \mathcal{K}^{p\circ})(\kappa)$ be two points for a perfect field κ containing \mathbb{F}_p^{Φ} . We have

$$\mathbf{B}_{s_{1}^{\bullet},s_{2}^{\bullet}}^{\bullet} = \prod_{j=0}^{\lfloor \frac{N}{2} \rfloor - 1} \prod_{s^{\bullet} \in \mathrm{Hk}_{j}^{-1}(s_{1}^{\bullet},s_{2}^{\bullet})(\kappa)} \mathbf{B}_{s^{\bullet}}^{\bullet}.$$

Take $s^{\bullet} = (A_0, \lambda_0, \eta_0^p; A_1^{\bullet}, \lambda_1^{\bullet}, \eta_1^{p^{\bullet}}; A_2^{\bullet}, \lambda_2^{\bullet}, \eta_2^{p^{\bullet}}; \phi^{\bullet}) \in \operatorname{Hk}_j^{-1}(s_1^{\bullet}, s_2^{\bullet})(\kappa) \text{ for some } 0 \le j \le \lfloor \frac{N}{2} \rfloor - 1.$

(1) Denote by \overline{H}_i the image of H_i in $\mathrm{H}_1^{\mathrm{dR}}(\phi^{\bullet})_{\tau_{\infty}^{\mathsf{c}}} \otimes_{\kappa} \mathcal{O}_S = (\mathscr{V}_{s^{\bullet}})_S$ for i = 1, 2. Then The assignment sending $(A_0, \lambda_0, \eta_0^p; A, \lambda, \eta^p; A_1^{\bullet}, \lambda_1^{\bullet}, \eta_1^{p^{\bullet}}; \gamma_1; A_2^{\bullet}, \lambda_2^{\bullet}, \eta_2^{p^{\bullet}}; \gamma_2) \in \mathrm{B}_{s^{\bullet}}^{\mathsf{s}}(S)$ to $(\overline{H}_1, \overline{H}_2)$ induces an isomorphism

$$\zeta_{s^{\bullet}}^{\bullet} \colon \mathcal{B}_{s^{\bullet}}^{\bullet} \to \mathcal{DL}^{\bullet}(\mathscr{V}_{s^{\bullet}}, \{ , \}_{s^{\bullet}})$$

(Definition A.2.1) in $Sch_{/\kappa}$.

(2) The cokernel of the map

$$\mathcal{T}_{\mathrm{B}^{\bullet}_{s^{\bullet}_{1}}/\kappa}\mid_{\mathrm{B}^{\bullet}_{s^{\bullet}}}\bigoplus \mathcal{T}_{\mathrm{B}^{\bullet}_{s^{\bullet}_{2}}/\kappa}\mid_{\mathrm{B}^{\bullet}_{s^{\bullet}}}\to \iota^{\bullet*}\mathcal{T}_{\mathrm{M}^{\bullet}_{\mathfrak{p}}(\mathrm{V}^{\circ},\mathrm{K}^{p\circ})/\kappa}\mid_{\mathrm{B}^{\bullet}_{s^{\bullet}}}$$

is canonically isomorphic to

$$\zeta_{s^{\bullet}}^{\bullet*}\left(\left(\sigma^{*}\bar{\mathcal{H}}_{s^{\bullet}2}\right)\otimes_{\mathcal{O}_{\mathrm{DL}^{\bullet}(\mathscr{V}_{s^{\bullet}},\{,\,\}_{s^{\bullet}})}}\left(\bar{\mathcal{H}}_{s^{\bullet}1}^{\dashv}/\bar{\mathcal{H}}_{s^{\bullet}2}\right)\right)$$

where
$$(\mathcal{H}_{s^{\bullet}1}, \mathcal{H}_{s^{\bullet}2})$$
 is the universal object over $\mathrm{DL}^{\bullet}(\mathscr{V}_{s^{\bullet}}, \{,\}_{s^{\bullet}})$

Proof. The decomposition of $B^{\bullet}_{s_1,s_2^{\bullet}}$ follows directly from the definition and the fact that $Hk_i^{-1}(s_1^{\bullet}, s_2^{\bullet})$ is isomorphic to a finite disjoint union of Spec κ .

Now we show (1). We first notice that Lemma 5.6.3 implies that (\bar{H}_1, \bar{H}_2) is an element in $DL^{\bullet}(\mathscr{V}_{s^{\bullet}}, \{, \}_{s^{\bullet}})(S)$.

Since the target of $\zeta_{s^{\bullet}}^{\bullet}$ is smooth over κ by Lemma 5.6.4, to see that $\zeta_{s^{\bullet}}^{\bullet}$ is an isomorphism, it suffices to check that for every algebraically closed field κ' containing κ

(1.1) $\zeta_{s^{\bullet}}^{\bullet}$ induces a bijection on κ' -points; and

(1.2) ζ_{s}^{\bullet} induces an isomorphism on the tangent spaces at every κ' -point.

To ease notation, we may assume $\kappa' = \kappa$.

For (1.1), we construct an inverse to the map $\zeta_{s^{\bullet}}^{\bullet}(\kappa)$. Take a point $y \in \mathrm{DL}^{\bullet}(\mathscr{V}_{s^{\bullet}}, \{ , \}_{s^{\bullet}})(\kappa)$ represented by κ -linear subspaces $\mathscr{V}_{s^{\bullet}}^{\dashv} \subseteq \bar{H}_2 \subseteq \bar{H}_1 \subseteq \mathscr{V}_{s^{\bullet}}$, or equivalently, subspaces

$$\operatorname{im}(p\phi^{\bullet-1})_{*,\tau_{\infty}^{\mathsf{c}}} \oplus \operatorname{H}_{1}^{\mathrm{dR}}(A_{1}^{\bullet}/\kappa)_{\tau_{\infty}}^{\perp} \subseteq H_{2} \subseteq H_{1} \subseteq \ker(p\phi^{\bullet})_{*,\tau_{\infty}^{\mathsf{c}}} \subseteq \operatorname{H}_{1}^{\mathrm{dR}}(A_{1}^{\bullet}/\kappa)_{\tau_{\infty}^{\mathsf{c}}}$$

These give rise to a point $y_1 \in DL^{\bullet}(\mathscr{V}_{s_1^{\bullet}}, \{, \}_{s_1^{\bullet}})(\kappa)$. By Theorem 5.3.4(3), we obtain a unique point $x_1 = (A_0, \lambda_0, \eta_0^p; A, \lambda, \eta^p; A_1^{\bullet}, \lambda_1^{\bullet}, \eta_1^{p^{\bullet}}; \gamma_1) \in B_{s_1^{\bullet}}^{\bullet}(\kappa)$ such that $\zeta_{s_1^{\bullet}}^{\bullet}(x_1) = y_1$. Put $\gamma_2 := \phi^{\bullet} \circ \gamma_1 \colon A \to A_2^{\bullet}$. We claim that γ_2 is a quasi-*p*-isogeny. In fact, as $\lambda \circ \check{\gamma}_1 = \gamma_1^{\vee} \circ \lambda_1^{\bullet}, \langle \operatorname{im} \gamma_{1*,\tau_{\infty}}, \operatorname{im} \gamma_{1*,\tau_{\infty}^{\circ}} \rangle_{\lambda_1^{\bullet},\tau_{\infty}} = 0$. Thus, we have

$$\operatorname{im} \gamma_{1*,\tau_{\infty}^{\mathsf{c}}} \subseteq (\operatorname{im} \gamma_{1*,\tau_{\infty}})^{\perp} = (\mathbb{V}^{-1}\gamma_{1*,\tau_{\infty}^{\mathsf{c}}}\omega_{A^{\vee},\tau_{\infty}^{\mathsf{c}}})^{\perp} = H_{2}^{\dashv} \subseteq \operatorname{ker}(p\phi^{\bullet})_{*,\tau_{\infty}^{\mathsf{c}}}.$$

By the isomorphisms $\mathbb{V}: \mathrm{H}_{1}^{\mathrm{dR}}(A_{1}^{\bullet}/\kappa)_{\tau_{\infty}} \to \mathrm{H}_{1}^{\mathrm{dR}}(A_{1}^{\bullet}/\kappa)_{\tau_{\infty}^{\circ}}$ and $\mathbb{V}: \mathrm{H}_{1}^{\mathrm{dR}}(A_{2}^{\bullet}/\kappa)_{\tau_{\infty}} \to \mathrm{H}_{1}^{\mathrm{dR}}(A_{2}^{\bullet}/\kappa)_{\tau_{\infty}^{\circ}}$, we obtain im $\gamma_{1*,\tau_{\infty}} \subseteq \ker(p\phi^{\bullet})_{*,\tau_{\infty}}$. In particular, $\operatorname{im}(p\phi^{\bullet} \circ \gamma_{1})_{*,\tau} = 0$ for every $\tau \in \Sigma_{\infty}$; in other words, γ_{2} is a quasi-*p*-isogeny. Now we show that $x_{2} \coloneqq (A_{0}, \lambda_{0}, \eta_{0}^{p}; A, \lambda, \eta^{p}; A_{2}^{\bullet}, \lambda_{2}^{\bullet}, \eta_{2}^{p\bullet}; \gamma_{2})$ satisfies Definition 5.3.2(a–d).

For (a), it suffices to show that $p\gamma_2^{-1}$ is a quasi-*p*-isogeny, equivalently, $\gamma_1^{-1} \circ (p\phi^{\bullet-1})$ is a quasi*p*-isogeny. However, we have $\operatorname{im}(p\phi^{\bullet-1})_{*,\tau_{\infty}} = \mathbb{V}^{-1}\operatorname{im}(p\phi^{\bullet-1})_{*,\tau_{\infty}^{c}} \subseteq \mathbb{V}^{-1}H_2 = \operatorname{im}\gamma_{1*,\tau_{\infty}}$, hence $\operatorname{im}(p\phi^{\bullet-1})_{*,\tau_{\infty}^{c}} \subseteq \operatorname{im}\gamma_{1*,\tau_{\infty}^{c}}$ using the action of \mathbb{V} , which together imply that $\gamma_1^{-1} \circ (p\phi^{\bullet-1})$ is a quasi-*p*-isogeny.

For (b), we identify $\mathcal{D}(A)$ as submodules of both $\mathcal{D}(A_1^{\bullet})$ and $\mathcal{D}(A_2^{\bullet})$ via γ_1 and γ_2 , respectively. Then we need to show that $p\mathcal{D}(A_2^{\bullet})_{\tau_{\infty}}^{\vee} \cap \mathcal{D}(A)_{\tau_{\infty}^{\circ}} \subseteq \mathbb{V}\mathcal{D}(A)_{\tau_{\infty}}$. As $p\phi^{\bullet-1} \circ \lambda_2^{\bullet-1}$ is a quasi-*p*-isogeny, we have $p\mathcal{D}(A_2^{\bullet})_{\tau_{\infty}}^{\vee} \subseteq \mathcal{D}(A_1^{\bullet})_{\tau_{\infty}^{c}}^{\vee}$. Moreover, the image of $p\mathcal{D}(A_2^{\bullet})_{\tau_{\infty}^{\vee}}^{\vee}$ in $\mathcal{D}(A_1^{\bullet})_{\tau_{\infty}^{c}}/p\mathcal{D}(A_1^{\bullet})_{\tau_{\infty}^{c}}^{\vee} = H_1^{\mathrm{dR}}(A_1^{\bullet})_{\tau_{\infty}^{c}}$ is contained in $\mathrm{im}(p\phi^{\bullet-1})_{*,\tau_{\infty}^{c}} \oplus H_1^{\mathrm{dR}}(A_1^{\bullet}/\kappa)_{\tau_{\infty}^{\vee}}^{\perp}$, which is further contained in H_2 . Thus, $p\mathcal{D}(A_2^{\bullet})_{\tau_{\infty}^{\vee}}^{\vee} \cap \mathcal{D}(A)_{\tau_{\infty}^{c}} \subseteq \mathcal{V}\mathcal{D}(A)_{\tau_{\infty}}$ as $\mathcal{V}\mathcal{D}(A)_{\tau_{\infty}}$ is the inverse image of H_2 in $\mathcal{D}(A_1^{\bullet})_{\tau_{\infty}^{c}}$.

For (c) and (d), they follow obviously.

To summarize, x_2 belongs to $B^{\bullet}_{s^{\bullet}}(\kappa)$; and $x \coloneqq (x_1, x_2)$ is an element in $B^{\bullet}_{s^{\bullet}}(\kappa)$ such that $\zeta^{\bullet}_{s^{\bullet}}(x) =$ y. It is easy to see that such assignment gives rise to an inverse of $\zeta_{s}^{\bullet}(\kappa)$; hence (1.1) follows immediately.

For (1.2), let \mathcal{T}_x and \mathcal{T}_y be the tangent spaces at x and y as in (1.1), respectively. By Theorem 5.3.4(1), we have a canonical short exact sequence

$$0 \to \operatorname{Hom}_{\kappa}\left(\omega_{A^{\vee},\tau_{\infty}}, \frac{\omega_{A^{\vee},\tau_{\infty}}^{\perp}}{\omega_{A^{\vee},\tau_{\infty}}}\right) \to \mathcal{T}_{x} \to \operatorname{Hom}_{\kappa}\left(\frac{\omega_{A^{\vee},\tau_{\infty}}}{(\ker\gamma_{1*,\tau_{\infty}})^{\perp} + (\ker\gamma_{2*,\tau_{\infty}})^{\perp}}, \operatorname{Lie}_{A^{\vee},\tau_{\infty}^{\mathsf{c}}}\right) \to 0.$$

Then by Proposition A.2.2 and the construction, the induced map $(\zeta_{s^{\bullet}})_{*} \colon \mathcal{T}_{x} \to \mathcal{T}_{y}$ fits into a commutative diagram

in $Mod(\kappa)$. The left vertical arrow is the composition

$$\operatorname{Hom}_{\kappa}\left(\omega_{A^{\vee},\tau_{\infty}},\omega_{A^{\vee},\tau_{\infty}}^{\perp}/\omega_{A^{\vee},\tau_{\infty}}\right) \to \operatorname{Hom}_{\kappa}\left(H_{1}^{\perp}/\mathbb{V}^{-1}H_{2},H_{2}^{\perp}/H_{1}^{\perp}\right) \\ \xrightarrow{\sim} \operatorname{Hom}_{\kappa}\left(H_{1}/H_{2},H_{2}^{\perp}/H_{1}\right) \simeq \operatorname{Hom}_{\kappa}\left(\bar{H}_{1}/\bar{H}_{2},\bar{H}_{2}^{\perp}/\bar{H}_{1}\right),$$

which is an isomorphism. The right vertical arrow is induced by maps

$$\frac{\omega_{A^{\vee},\tau_{\infty}^{\mathsf{c}}}}{(\ker\gamma_{1*,\tau_{\infty}})^{\perp} + (\ker\gamma_{2*,\tau_{\infty}})^{\perp}} \xrightarrow{\gamma_{1*,\tau_{\infty}^{\mathsf{c}}}} \frac{H_2}{\operatorname{im}(p\phi^{\bullet-1})_{*,\tau_{\infty}^{\mathsf{c}}} \oplus \mathrm{H}_1^{\mathrm{dR}}(A_1^{\bullet}/\kappa)_{\tau_{\infty}}^{\perp}} \simeq \bar{H}_2/\mathscr{V}_{s^{\bullet}}^{\dashv},$$
$$\mathrm{Lie}_{A^{\vee},\tau_{\infty}^{\mathsf{c}}} \simeq \mathrm{H}_1^{\mathrm{dR}}(A)_{\tau_{\infty}^{\mathsf{c}}}/\omega_{A^{\vee},\tau_{\infty}^{\mathsf{c}}} \xrightarrow{\gamma_{1*,\tau_{\infty}^{\mathsf{c}}}} H_1^{\dashv}/H_2 \simeq \bar{H}_1^{\dashv}/\bar{H}_2.$$

As ker $\gamma_{2*,\tau_{\infty}} = \operatorname{im} \breve{\gamma}_{2*,\tau_{\infty}}$, we have

$$\frac{\omega_{A^{\vee},\tau_{\infty}^{\mathsf{c}}}}{(\ker\gamma_{1*,\tau_{\infty}})^{\perp} + (\ker\gamma_{2*,\tau_{\infty}})^{\perp}} \simeq \frac{\gamma_{1*,\tau_{\infty}}\omega_{A^{\vee},\tau_{\infty}^{\mathsf{c}}}}{\operatorname{im}(\gamma_{1}\circ\breve{\gamma}_{2})_{*,\tau_{\infty}^{\mathsf{c}}} + \operatorname{H}_{1}^{\mathrm{dR}}(A_{1}^{\bullet}/\kappa)_{\tau_{\infty}}^{\perp}} = \frac{H_{2}}{\operatorname{im}(p\phi^{\bullet-1})_{*,\tau_{\infty}^{\mathsf{c}}} \oplus \operatorname{H}_{1}^{\mathrm{dR}}(A_{1}^{\bullet}/\kappa)_{\tau_{\infty}}^{\perp}},$$

which implies that the first map is an isomorphism. By Theorem 5.3.4(4), the second map is an isomorphism as well. Thus, $(\zeta_{s^{\bullet}})_* \colon \mathcal{T}_x \to \mathcal{T}_y$ is an isomorphism by the Five Lemma; hence (1.2) and (1) follow.

Then we show (2). Theorem 5.3.4(2) implies that the cokernel of the map

$$\mathcal{T}_{\mathrm{B}^{\bullet}_{s_{1}^{\bullet}}/\kappa}\mid_{\mathrm{B}^{\bullet}_{s^{\bullet}}}\bigoplus\mathcal{T}_{\mathrm{B}^{\bullet}_{s_{2}^{\bullet}}/\kappa}\mid_{\mathrm{B}^{\bullet}_{s^{\bullet}}}\to\iota^{\bullet*}\mathcal{T}_{\mathrm{M}^{\bullet}_{\mathfrak{p}}(\mathrm{V}^{\circ},\mathrm{K}^{p\circ})/\kappa}\mid_{\mathrm{B}^{\bullet}_{s^{\bullet}}}$$

is canonically isomorphic to

(5.15)
$$\mathcal{H}om\left((\ker\gamma_{1*,\tau_{\infty}} + \ker\gamma_{2*,\tau_{\infty}})^{\perp}/\mathrm{H}_{1}^{\mathrm{dR}}(\mathcal{A})_{\tau_{\infty}}^{\perp}, \mathrm{Lie}_{\mathcal{A}^{\vee},\tau_{\infty}^{\mathsf{c}}}\right).$$

As ker $\gamma_{2*,\tau_{\infty}} = \operatorname{im} \breve{\gamma}_{2*,\tau_{\infty}}$, we have

(5.16)
$$\frac{\mathrm{H}_{1}^{\mathrm{dR}}(\mathcal{A})_{\tau_{\infty}}}{\ker \gamma_{1*,\tau_{\infty}} + \ker \gamma_{2*,\tau_{\infty}}} \simeq \frac{\mathrm{im} \, \gamma_{1*,\tau_{\infty}}}{\mathrm{im}(\gamma_{1} \circ \breve{\gamma}_{2})_{*,\tau_{\infty}}} = \frac{\mathrm{im} \, \gamma_{1*,\tau_{\infty}}}{\mathrm{im}(p\phi^{\bullet-1})_{*,\tau_{\infty}}} \simeq \frac{\mathrm{V} \, \mathrm{im} \, \gamma_{1*,\tau_{\infty}}}{\mathrm{V} \, \mathrm{im}(p\phi^{\bullet-1})_{*,\tau_{\infty}}}.$$

However, we have $\mathbb{V} \operatorname{im} \gamma_{1*,\tau_{\infty}} = (\gamma_{1*,\tau_{\infty}^{c}} \omega_{\mathcal{A},\tau_{\infty}^{c}})^{(p)}$ and $\mathbb{V} \operatorname{im}(p\phi^{\bullet-1})_{*,\tau_{\infty}} = (\operatorname{im}(p\phi^{\bullet-1})_{*,\tau_{\infty}^{c}})^{(p)}$. Thus, (5.16) is isomorphic to $\sigma^{*}\bar{\mathcal{H}}_{s^{\bullet}2}$, hence

$$(5.15) \simeq \mathcal{H}om\left((\sigma^*\bar{\mathcal{H}}_{s^{\bullet}2})^{\vee}, \operatorname{Lie}_{\mathcal{A}^{\vee}, \tau_{\infty}^{\mathsf{c}}}\right) \simeq \left(\sigma^*\bar{\mathcal{H}}_{s^{\bullet}2}\right) \otimes_{\mathcal{O}_{\mathrm{DL}^{\bullet}(\mathscr{V}_{s^{\bullet}}, \{,\}_{s^{\bullet}})} \left(\bar{\mathcal{H}}_{s^{\bullet}1}^{\dashv}/\bar{\mathcal{H}}_{s^{\bullet}2}\right)$$

where we use Theorem 5.3.4(4) for the last isomorphism. We have proved (2) and the theorem. \Box

We also need a description for

$$\mathbf{B}_{s^{\bullet}}^{\dagger} \coloneqq \mathbf{B}_{s^{\bullet}}^{\bullet} \times_{\mathbf{M}_{\mathfrak{p}}^{\bullet}(\mathbf{V}^{\circ},\mathbf{K}^{p\circ})} \mathbf{M}_{\mathfrak{p}}^{\dagger}(\mathbf{V}^{\circ},\mathbf{K}^{p\circ})$$

for $s^{\bullet} \in \operatorname{Hk}_{j}^{-1}(s_{1}^{\bullet}, s_{2}^{\bullet})(\kappa)$. It is clear that if we put

$$\mathbf{B}_{s_{i}^{\bullet}}^{\dagger} \coloneqq \mathbf{B}_{s_{i}^{\bullet}}^{\bullet} \times_{\mathbf{M}_{\mathfrak{p}}^{\bullet}(\mathbf{V}^{\circ},\mathbf{K}^{p\circ})} \mathbf{M}_{\mathfrak{p}}^{\dagger}(\mathbf{V}^{\circ},\mathbf{K}^{p\circ})$$

for i = 1, 2, then

$$\mathbf{B}_{s^{\bullet}}^{\dagger} = \mathbf{B}_{s_{1}^{\bullet}}^{\dagger} \times_{\mathbf{M}_{\mathfrak{p}}^{\dagger}(\mathbf{V}^{\circ},\mathbf{K}^{p\circ})} \mathbf{B}_{s_{2}^{\bullet}}^{\dagger}.$$

By definition, for every $S \in \mathsf{Sch}_{\kappa}, B^{\dagger}_{s^{\bullet}}(S)$ is the set of equivalence classes of unvigintuples

 $(A_0, \lambda_0, \eta_0^p; A, \lambda, \eta^p; A^{\circ}, \lambda^{\circ}, \eta^{p^{\circ}}; A_1^{\bullet}, \lambda_1^{\bullet}, \eta_1^{p^{\bullet}}; A_2^{\bullet}, \lambda_2^{\bullet}, \eta_2^{p^{\bullet}}; \beta, \gamma_1, \gamma_2, \psi_1, \psi_2, \phi^{\bullet})$

rendering the diagram



commute. Here, the letters remain the same meaning as in our previous moduli problems. Put

 $S^{\dagger}_{s^{\bullet}} \coloneqq \{s^{\bullet}\} \times_{S^{\bullet}_{\mathfrak{p}}(V^{\circ}, K^{p^{\circ}}) \times S^{\bullet}_{\mathfrak{p}}(V^{\circ}, K^{p^{\circ}})} \left(S^{\dagger}_{\mathfrak{p}}(V^{\circ}, K^{p^{\circ}}) \times S^{\dagger}_{\mathfrak{p}}(V^{\circ}, K^{p^{\circ}})\right) \times S^{\circ}_{\mathfrak{p}}(V^{\circ}, K^{p^{\circ}}) \times S^{\circ}_{\mathfrak{p}}(V^{\circ}) \times S^{\circ}_{\mathfrak{p}}$

$$\pi_{s^{\bullet}}^{\dagger} \colon \mathcal{B}_{s^{\bullet}}^{\dagger} \to \mathcal{S}_{s}^{\dagger}$$

of κ -schemes by forgetting (A, λ, η^p) and related morphisms.

Theorem 5.6.8. Let $s_1^{\bullet}, s_2^{\bullet} \in S_p^{\bullet}(V^{\circ}, K^{p \circ})(\kappa)$ be two points for a perfect field κ containing \mathbb{F}_p^{Φ} . Take $s^{\bullet} \in \operatorname{Hk}_j^{-1}(s_1^{\bullet}, s_2^{\bullet})(\kappa)$ for some $0 \leq j \leq \lfloor \frac{N}{2} \rfloor - 1$. Then the scheme $S_{s^{\bullet}}^{\dagger}$ is a disjoint of $(p+1)(p^3+1)\cdots(p^{2\lfloor \frac{N}{2} \rfloor - 2j-1}+1)$ copies of Spec κ .

 $Take \ a \ point \ t^{\dagger} = (A_0, \lambda_0, \eta_0^p; A^{\circ}, \lambda^{\circ}, \eta^{p\circ}; A_1^{\bullet}, \lambda_1^{\bullet}, \eta_1^{p\bullet}; A_2^{\bullet}, \lambda_2^{\bullet}, \eta_2^{p\bullet}; \psi_1, \psi_2, \phi^{\bullet}) \in \mathbf{S}_{s^{\bullet}}^{\dagger}(\kappa).$

(1) The assignment sending

$$(A_{0}, \lambda_{0}, \eta_{0}^{p}; A, \lambda, \eta^{p}; A^{\circ}, \lambda^{\circ}, \eta^{p\circ}; A_{1}^{\bullet}, \lambda_{1}^{\bullet}, \eta_{1}^{p\bullet}; A_{2}^{\bullet}, \lambda_{2}^{\bullet}, \eta_{2}^{p\bullet}; \beta, \gamma_{1}, \gamma_{2}, \psi_{1}, \psi_{2}, \phi^{\bullet}) \in \mathbf{B}_{s\bullet}^{\dagger}(S)$$

to $H_{2}/(\operatorname{im}(p\phi^{\bullet-1})_{*,\tau_{\infty}^{\circ}} + \mathbf{H}_{1}^{\mathrm{dR}}(A^{\bullet}/S)_{\tau_{\infty}}^{\perp})$ induces an isomorphism
 $\zeta_{t^{\dagger}}^{\dagger}: (\pi_{s\bullet}^{\dagger})^{-1}(t^{\dagger}) \xrightarrow{\sim} \mathbb{P}(\mathscr{V}_{t^{\dagger}})$

where we put

$$\mathscr{V}_{t^{\dagger}} := \frac{\operatorname{im}(\psi_1)_{*,\tau_{\infty}^{\mathsf{c}}}}{\operatorname{im}(p\phi^{\bullet-1})_{*,\tau_{\infty}^{\mathsf{c}}} + \mathrm{H}_{1}^{\mathrm{dR}}(A^{\bullet}/S)_{\tau_{\infty}}^{\perp}}$$

which has dimension $\lfloor \frac{N}{2} \rfloor - j$. (2) The cokernel of the map

$$\mathcal{T}_{\mathrm{B}^{\dagger}_{s_{1}^{\bullet}}/\kappa}|_{(\pi^{\dagger}_{s_{\bullet}})^{-1}(t^{\dagger})} \bigoplus \mathcal{T}_{\mathrm{B}^{\dagger}_{s_{2}^{\bullet}}/\kappa}|_{(\pi^{\dagger}_{s_{\bullet}})^{-1}(t^{\dagger})} \to \iota^{\bullet*}\mathcal{T}_{\mathrm{M}^{\dagger}_{\mathfrak{p}}(\mathrm{V}^{\circ},\mathrm{K}^{p\circ})/\kappa}|_{(\pi^{\dagger}_{s_{\bullet}})^{-1}(t^{\dagger})}$$

is canonically isomorphic to

$$\zeta_{t^{\dagger}}^{\dagger*}\left((\sigma^{*}\mathcal{H}_{t^{\dagger}})\otimes_{\mathcal{O}_{\mathbb{P}(\mathscr{V}_{t^{\dagger}})}}\mathcal{O}_{\mathbb{P}(\mathscr{V}_{t^{\dagger}})}(1)\right)$$

where $\mathcal{H}_{t^{\dagger}}$ is the universal object, namely, the tautological bundle on $\mathbb{P}(\mathscr{V}_{t^{\dagger}})$.

Proof. In fact, the assignment sending $(A_0, \lambda_0, \eta_0^p; A^{\circ}, \lambda^{\circ}, \eta^{p^{\circ}}; A_1^{\bullet}, \lambda_1^{\bullet}, \eta_1^{p^{\bullet}}; A_2^{\bullet}, \lambda_2^{\bullet}, \eta_2^{p^{\bullet}}; \psi_1, \psi_2, \phi^{\bullet}) \in S_{s^{\bullet}}^{\dagger}(S)$ to $\operatorname{im}(\psi_1)_{*,\tau_{c}^{\infty}}$ induces a bijection from $S_{s^{\bullet}}^{\dagger}(S)$ to the subbundles $H \subseteq \operatorname{H}_1^{\operatorname{dR}}(A^{\bullet}/S)_{\tau_{\infty}^{\circ}}$ of rank $\lceil \frac{N}{2} \rceil$ satisfying $\operatorname{im}(p\phi^{\bullet-1})_{*,\tau_{\infty}^{\circ}} \otimes_{\kappa} \mathcal{O}_S \subseteq H \subseteq \operatorname{ker}(p\phi^{\bullet})_{*,\tau_{\infty}^{\circ}} \otimes_{\kappa} \mathcal{O}_S$ and $\langle V^{-1}H, H \rangle_{\tau_{\infty}^{\circ}} = 0$. Thus, we know that $S_{s^{\bullet}}^{\dagger}$ is a disjoint of $(p+1)(p^3+1)\cdots(p^{2\lfloor \frac{N}{2} \rfloor -2j-1}+1)$ copies of Spec κ .

For (1), we denote by s_1^{\dagger} the image of t^{\dagger} in $\tilde{S}_{\mathfrak{p}}^{\dagger}(V^{\circ}, K^{p\circ})(\kappa)$ in the first factor. Then a point $(A_0, \lambda_0, \eta_0^p; A, \lambda, \eta^p; A^{\circ}, \lambda^{\circ}, \eta^{p\circ}; A_1^{\bullet}, \lambda_1^{\bullet}, \eta_1^{p\bullet}; \beta, \gamma_1) \in B_{s_1^{\dagger}}^{\dagger}(S)$ belongs to $B_{s\bullet}^{\dagger}(S)$ if and only if H_2 contains $\operatorname{im}(p\phi^{\bullet-1})_{*,\tau_{\infty}^{\circ}} \otimes_{\kappa} \mathcal{O}_S$. Thus, (1) follows from Theorem 5.4.3(2).

For (2), it follows from Theorem 5.6.7(2) and the isomorphism

$$\left(\bar{\mathcal{H}}_{s^{\bullet}1}^{\dagger}/\bar{\mathcal{H}}_{s^{\bullet}2}\right)|_{\mathbb{P}(\mathscr{V}_{t^{\dagger}})} = \left(\bar{\mathcal{H}}_{s^{\bullet}1}/\bar{\mathcal{H}}_{s^{\bullet}2}\right)|_{\mathbb{P}(\mathscr{V}_{t^{\dagger}})} \simeq \mathcal{O}_{\mathbb{P}(\mathscr{V}_{t^{\dagger}})}(1).$$

5.7. Incidence maps on ground stratum. In this subsection, we define and study the incidence maps on ground stratum. We assume $N \ge 2$. In order to have a uniformization map for $S^{\bullet}_{\mathfrak{p}}(V^{\circ}, -)$, we also choose data as in Construction 5.3.6.

Definition 5.7.1. We denote

O $\mathbb{T}^{\circ}_{N,\mathfrak{p}}$ the Hecke algebra $\mathbb{Z}[\mathrm{K}^{\circ}_{\mathfrak{p}} \setminus \mathrm{U}(\mathrm{V}^{\circ})(F_{\mathfrak{p}}^{+})/\mathrm{K}^{\circ}_{\mathfrak{p}}];$ O $\mathbb{T}^{\bullet}_{N,\mathfrak{p}}$ the Hecke algebra $\mathbb{Z}[\mathrm{K}^{\circ}_{\mathfrak{p}} \setminus \mathrm{U}(\mathrm{V}^{\circ})(F_{\mathfrak{p}}^{+})/\mathrm{K}^{\circ}_{\mathfrak{p}}];$ O $\mathrm{T}^{\bullet\circ}_{N,\mathfrak{p}} \in \mathbb{Z}[\mathrm{K}^{\bullet}_{\mathfrak{p}} \setminus \mathrm{U}(\mathrm{V}^{\circ})(F_{\mathfrak{p}}^{+})/\mathrm{K}^{\circ}_{\mathfrak{p}}]$ the characteristic function of $\mathrm{K}^{\bullet}_{\mathfrak{p}}\mathrm{K}^{\circ}_{\mathfrak{p}};$ and O $\mathrm{T}^{\circ\circ}_{N,\mathfrak{p}} \in \mathbb{Z}[\mathrm{K}^{\circ}_{\mathfrak{p}} \setminus \mathrm{U}(\mathrm{V}^{\circ})(F_{\mathfrak{p}}^{+})/\mathrm{K}^{\circ}_{\mathfrak{p}}]$ the characteristic function of $\mathrm{K}^{\circ}_{\mathfrak{p}}\mathrm{K}^{\circ}_{\mathfrak{p}}.$

Moreover, we define the *intertwining Hecke operator* to be

$$\mathtt{I}_{N,\mathfrak{p}}^{\circ}\coloneqq\mathtt{T}_{N,\mathfrak{p}}^{\circ\bullet}\circ\mathtt{T}_{N,\mathfrak{p}}^{\bullet\circ}\in\mathbb{T}_{N,\mathfrak{p}}^{\circ}$$

where the composition is taken as composition of cosets.

Remark 5.7.2. We remind the readers that according to our convention, the unit elements of $\mathbb{Z}[\mathrm{K}^{\circ}_{\mathfrak{p}} \setminus \mathrm{U}(\mathrm{V}^{\circ})(F^{+}_{\mathfrak{p}})/\mathrm{K}^{\circ}_{\mathfrak{p}}]$ and $\mathbb{Z}[\mathrm{K}^{\circ}_{\mathfrak{p}} \setminus \mathrm{U}(\mathrm{V}^{\circ})(F^{+}_{\mathfrak{p}})/\mathrm{K}^{\circ}_{\mathfrak{p}}]$ are $\mathbb{1}_{\mathrm{K}^{\circ}_{\mathfrak{p}}}$ and $\mathbb{1}_{\mathrm{K}^{\circ}_{\mathfrak{p}}}$, respectively. However, when N is odd, $\mathrm{K}^{\circ}_{\mathfrak{p}}$ and $\mathrm{K}^{\circ}_{\mathfrak{p}}$ have different volumes under a common Haar measure on $\mathrm{U}(\mathrm{V}^{\circ})(F^{+}_{\mathfrak{p}})$; in other words, the convolution products on the two Hecke algebras are not induced by the same Haar measure on $\mathrm{U}(\mathrm{V}^{\circ})(F^{+}_{\mathfrak{p}})$.

Let L be a *p*-coprime coefficient ring. By Construction 5.2.6 and Construction 5.3.6, we have canonical isomorphisms

$$\begin{split} &L[\operatorname{Sh}(\mathrm{V}^{\circ},-\mathrm{K}_{p}^{\circ})]\simeq \mathrm{H}_{\mathfrak{T}}^{0}(\overline{\mathrm{S}}_{\mathfrak{p}}^{\circ}(\mathrm{V}^{\circ},-),L),\\ &L[\operatorname{Sh}(\mathrm{V}^{\circ},-\mathrm{K}_{p}^{\bullet})]\simeq \mathrm{H}_{\mathfrak{T}}^{0}(\overline{\mathrm{S}}_{\mathfrak{p}}^{\bullet}(\mathrm{V}^{\circ},-),L), \end{split}$$

in $\operatorname{Fun}(\mathfrak{K}(\mathcal{V}^{\circ})^{p}, \operatorname{\mathsf{Mod}}(L[\mathcal{K}_{\mathfrak{p}}^{\circ} \setminus \mathcal{U}(\mathcal{V}^{\circ})(F_{\mathfrak{p}}^{+})/\mathcal{K}_{\mathfrak{p}}^{\circ}]))$ and in $\operatorname{Fun}(\mathfrak{K}(\mathcal{V}^{\circ})^{p}, \operatorname{\mathsf{Mod}}(L[\mathcal{K}_{\mathfrak{p}}^{\bullet} \setminus \mathcal{U}(\mathcal{V}^{\circ})(F_{\mathfrak{p}}^{+})/\mathcal{K}_{\mathfrak{p}}^{\bullet}]))$, induced by v° (5.4) and v^{\bullet} (5.9), respectively.

Construction 5.7.3. Recall from Definition 5.5.3 the class $\xi \in H^2_{\mathfrak{L}}(\overline{B}^{\circ}_{\mathfrak{p}}(V^{\circ}, -), L(1))$, which is the first Chern class of the tautological quotient line bundle on $\overline{B}^{\circ}_{\mathfrak{p}}(V^{\circ}, -)$. Put $r \coloneqq \lfloor \frac{N}{2} \rfloor \ge 1$. We construct three pairs of maps in $\operatorname{Fun}(\mathfrak{K}(V^{\circ}_n)^p_{\operatorname{sp}}, \operatorname{Mod}(L))$ as follows:

$$\begin{cases} \operatorname{inc}_{!}^{\circ} : L[\operatorname{Sh}(\operatorname{V}^{\circ}, -\operatorname{K}^{\circ}_{p})] \xrightarrow{\sim} \operatorname{H}^{\circ}_{\mathfrak{A}}(\overline{\operatorname{S}}^{\circ}_{p}(\operatorname{V}^{\circ}, -), L) \xrightarrow{\pi^{\circ*}} \operatorname{H}^{0}_{\mathfrak{A}}(\overline{\operatorname{B}}^{\circ}_{p}(\operatorname{V}^{\circ}, -), L) \\ \xrightarrow{\cup \xi^{N-r-1}} \operatorname{H}^{2}_{\mathfrak{A}}(^{N-r-1})(\overline{\operatorname{H}}^{\circ}_{p}(\operatorname{V}^{\circ}, -), L(N-r-1)) \\ \xrightarrow{\iota^{\circ}_{!}} \operatorname{H}^{2}_{\mathfrak{A}}(^{N-r-1})(\overline{\operatorname{M}}^{\circ}_{p}(\operatorname{V}^{\circ}, -), L(N-r-1)) \\ \xrightarrow{\cup \xi^{N-r-1}} \operatorname{H}^{2}_{\mathfrak{A}}(^{N-1})(\overline{\operatorname{B}}^{\circ}_{p}(\operatorname{V}^{\circ}, -), L(N-r-1)) \\ \xrightarrow{\cup \xi^{N-r-1}} \operatorname{H}^{2}_{\mathfrak{A}}(^{N-1})(\overline{\operatorname{B}}^{\circ}_{p}(\operatorname{V}^{\circ}, -), L(N-1)) \\ \xrightarrow{\pi^{\circ}_{!}} \operatorname{H}^{0}_{\mathfrak{A}}(\overline{\operatorname{S}}^{\circ}_{p}(\operatorname{V}^{\circ}, -), L) \xrightarrow{\to} L[\operatorname{Sh}(\operatorname{V}^{\circ}, -\operatorname{K}^{\circ}_{p})]; \end{cases} \\ \begin{cases} \operatorname{inc}_{!}^{\dagger} : L[\operatorname{Sh}(\operatorname{V}^{\circ}, -\operatorname{K}^{\circ}_{p})] \xrightarrow{\sim} \operatorname{H}^{0}_{\mathfrak{A}}(\overline{\operatorname{S}}^{\circ}_{p}(\operatorname{V}^{\circ}, -), L) \xrightarrow{\pi^{\circ*}} \operatorname{H}^{0}_{\mathfrak{A}}(\overline{\operatorname{B}}^{\circ}_{p}(\operatorname{V}^{\circ}, -), L) \\ \xrightarrow{\cup \xi^{N-r-2}} \operatorname{H}^{2}_{\mathfrak{A}}(^{N-r-2})(\overline{\operatorname{H}}^{\circ}_{p}(\operatorname{V}^{\circ}, -), L(N-r-2)) \\ \xrightarrow{\pi^{\dagger}_{!}} \operatorname{H}^{2}_{\mathfrak{A}}(^{N-r-2})(\overline{\operatorname{M}}^{\circ}_{p}(\operatorname{V}^{\circ}, -), L(N-r-2)) \\ \xrightarrow{\pi^{\dagger}_{!}} \operatorname{H}^{2}_{\mathfrak{A}}(^{N-r-2})(\overline{\operatorname{M}}^{\circ}_{p}(\operatorname{V}^{\circ}, -), L(N-r-2)) \\ \xrightarrow{\pi^{\dagger}_{!}} \operatorname{H}^{2}_{\mathfrak{A}}(^{2(N-r-2)})(\overline{\operatorname{M}}^{\circ}_{p}(\operatorname{V}^{\circ}, -), L(N-r-1)), \\ \operatorname{inc}^{\dagger}_{!} : \operatorname{H}^{2}_{\mathfrak{A}}(\overline{\operatorname{M}}^{\circ}_{p}(\operatorname{V}^{\circ}, -), L(r)) \xrightarrow{\pi^{\dagger}_{!}} \operatorname{H}^{2}_{\mathfrak{A}}(^{2(N-r-2)})(\overline{\operatorname{M}}^{\circ}_{p}(\operatorname{V}^{\circ}, -), L(N-r-1)), \\ \operatorname{inc}^{\dagger}_{!} : \operatorname{H}^{2}_{\mathfrak{A}}(\overline{\operatorname{M}}^{\circ}_{p}(\operatorname{V}^{\circ}, -), L(r)) \xrightarrow{\pi^{\dagger}_{!}} \operatorname{H}^{2}_{\mathfrak{A}}(\overline{\operatorname{M}}^{\circ}_{p}(\operatorname{V}^{\circ}, -), L(r+1) \\ \xrightarrow{\cup \xi^{N-r-2}} \operatorname{H}^{2}_{\mathfrak{A}}(^{2(N-1)}(\overline{\operatorname{B}}^{\circ}_{p}(\operatorname{V}^{\circ}, -), L(N-1)) \\ \xrightarrow{\pi^{\dagger}_{!}} \operatorname{H}^{2}_{\mathfrak{A}}(\overline{\operatorname{S}}^{\circ}_{p}(\operatorname{V}^{\circ}, -), L(r+1) \\ \xrightarrow{\cup \xi^{N-r-2}} \operatorname{H}^{2}_{\mathfrak{A}}(\overline{\operatorname{M}}^{\circ}_{p}(\operatorname{V}^{\circ}, -), L(N-r-1)), \\ \operatorname{inc}^{\circ}_{!} : L[\operatorname{Sh}(\operatorname{V}^{\circ}, -\operatorname{K}^{\circ}_{p})] \xrightarrow{\cong} \operatorname{H}^{0}_{\mathfrak{A}}(\overline{\operatorname{S}}^{\circ}_{p}(\operatorname{V}^{\circ}, -), L(N-1)) \\ \xrightarrow{\pi^{\dagger}_{!}} \operatorname{H}^{2}_{\mathfrak{A}}(\overline{\operatorname{S}}^{\circ}_{p}(\operatorname{V}^{\circ}, -), L(N-r-1)), \\ \operatorname{inc}^{\circ}_{!} : \operatorname{H}^{2}_{\mathfrak{A}}(\overline{\operatorname{M}}^{\circ}_{p}(\operatorname{V}^{\circ}, -), L(r)) \xrightarrow{\pi^{\circ}_{!}} \operatorname{H}^{2}_{\mathfrak{A}}(\overline{\operatorname{M}}^{\circ}_{p}(\operatorname{V}^{\circ}, -), L(r)) \\ \operatorname{inc}^{\circ}_{!} : \operatorname{H}^{2}_{\mathfrak{A}}(\overline{\operatorname{M}}^{\circ}_{p}(\operatorname{V}^{\circ}, -), L(r)) \xrightarrow{\pi^{\circ}_{!}} \operatorname{H}^{2}_{\mathfrak{A}}(\overline{\operatorname{M}}^{\circ}_{p}($$

Note that the construction of the second pair only makes sense when $N \ge 3$; and when N = 2, we regard inc[†] and inc[†] as zero maps. In fact, the two maps in each pair are essentially Poincaré dual to each other.

Definition 5.7.4. Suppose N = 2r + 1 odd with $r \ge 1$. We define the *incidence map (on the ground stratum)* to be the map

$$\operatorname{inc} \colon L[\operatorname{Sh}(\mathrm{V}^{\circ}, -\mathrm{K}_{p}^{\circ})] \bigoplus L[\operatorname{Sh}(\mathrm{V}^{\circ}, -\mathrm{K}_{p}^{\bullet})] \to L[\operatorname{Sh}(\mathrm{V}^{\circ}, -\mathrm{K}_{p}^{\circ})] \bigoplus L[\operatorname{Sh}(\mathrm{V}^{\circ}, -\mathrm{K}_{p}^{\bullet})]$$

in $\operatorname{Fun}(\mathfrak{K}(\mathcal{V}^{\circ})^{p}, \operatorname{\mathsf{Mod}}(L))$ given by the matrix

$$\begin{pmatrix} \operatorname{inc}_{\dagger}^{*} \circ \operatorname{inc}_{!}^{\dagger} & \operatorname{inc}_{\dagger}^{*} \circ \operatorname{inc}_{!}^{\bullet} \\ \operatorname{inc}_{\bullet}^{*} \circ \operatorname{inc}_{!}^{\dagger} & \operatorname{inc}_{\bullet}^{*} \circ \operatorname{inc}_{!}^{\bullet} \end{pmatrix}$$

if we write elements in the column form.

Remark 5.7.5. The construction of the incidence map can be encoded in the following diagram



in $\operatorname{Fun}(\mathfrak{K}(\mathcal{V}^{\circ})^{p}, \operatorname{\mathsf{Mod}}(L)).$

Proposition 5.7.6. Suppose N = 2r + 1 odd with $r \ge 1$. Then the incidence map inc is given by the matrix

$$\begin{pmatrix} -(p+1)^2 & \mathsf{T}_{N,\mathfrak{p}}^{\bullet\bullet} \\ \mathsf{T}_{N,\mathfrak{p}}^{\bullet\circ} & \mathsf{T}_{N,\mathfrak{p}}^{\bullet} \end{pmatrix}$$

where

$$\mathsf{T}^ullet_{N,\mathfrak{p}}\coloneqq\sum_{\delta=0}^{r-1}\mathsf{d}^ullet_{r-\delta,p}\cdot\mathsf{T}^ullet_{N,\mathfrak{p};\delta}$$

in which the numbers $d^{\bullet}_{r-\delta,p}$ are introduced in Notation 1.3.2, and the Hecke operators $T^{\bullet}_{N,\mathfrak{p};\delta}$ are introduced in Notation B.2.1 (as $T^{\bullet}_{N;\delta}$).

Proof. Take an object $\mathbf{K}^{p\circ} \in \mathfrak{K}(\mathbf{V}^{\circ})^{p}$.

First, we show $\operatorname{inc}_{\dagger}^* \circ \operatorname{inc}_{!}^{\dagger} = -(p+1)^2$. Since $\operatorname{m}^{\dagger \circ *} \mathcal{O}_{\overline{\mathrm{M}}_{\mathfrak{p}}^\circ(\mathrm{V}^\circ,\mathrm{K}^{p\circ})}(1)$ has degree p+1, it follows from Corollary 5.2.5.

Second, we show $\operatorname{inc}_{\dagger}^* \circ \operatorname{inc}_{!}^{\bullet} = \mathsf{T}_{N,\mathfrak{p}}^{\circ\bullet}$ and $\operatorname{inc}_{\bullet}^* \circ \operatorname{inc}_{!}^{\dagger} = \mathsf{T}_{N,\mathfrak{p}}^{\circ\circ}$. However, these are consequences of Theorem 5.4.3 and Construction 5.4.5.

Finally, we show $\operatorname{inc}^{\bullet}_{\bullet} \circ \operatorname{inc}^{\bullet}_{!} = \mathsf{T}^{\bullet}_{N,\mathfrak{p}}$. By Theorem 5.6.7(1), it suffices to show that for every $s_{1}^{\bullet}, s_{2}^{\bullet} \in \mathrm{S}^{\bullet}_{\mathfrak{p}}(\mathrm{V}^{\circ}, \mathrm{K}^{p\circ})(\overline{\mathbb{F}}_{p})$ and every $s^{\bullet} \in \mathrm{Hk}_{j}^{-1}(s_{1}^{\bullet}, s_{2}^{\bullet})$, the intersection multiplicity of $\mathrm{B}^{\bullet}_{s_{1}^{\bullet}}$ and $\mathrm{B}^{\bullet}_{s_{2}^{\bullet}}$ at the component $\mathrm{B}^{\bullet}_{s^{\bullet}}$ equals $\mathrm{d}^{\bullet}_{r-j,p}$. However, this is true by Theorem 5.6.7(2), Proposition A.2.4(1), and the excessive intersection formula.

The proposition is proved.

Now we assume that N = 2r is even with $r \ge 2$. The reader may notice that the situation is different from Definition 5.7.4 since now $M^{\bullet}_{\mathfrak{p}}(V^{\circ}, -)$ has dimension 2r - 1 while $B^{\bullet}_{\mathfrak{p}}(V^{\circ}, -)$ still has dimension r. Thus to obtain a similar diagram as in Remark 5.7.5, we have to insert a map

$$\Theta \colon \mathrm{H}^{2r-2}_{\mathfrak{T}}(\overline{\mathrm{M}}^{\bullet}_{\mathfrak{p}}(\mathrm{V}^{\circ}, -), L(r-1)) \to \mathrm{H}^{2r}_{\mathfrak{T}}(\overline{\mathrm{M}}^{\bullet}_{\mathfrak{p}}(\mathrm{V}^{\circ}, -), L(r))$$

to obtain a diagram like



Definition 5.7.7. For every line bundle \mathcal{L} on $M_{p}^{\bullet}(V^{\circ}, -)$,⁷ we denote

$$\Theta_{\mathcal{L}} \colon \mathrm{H}^{2r-2}_{\mathfrak{T}}(\overline{\mathrm{M}}^{\bullet}_{\mathfrak{p}}(\mathrm{V}^{\circ}, -), L(r-1)) \to \mathrm{H}^{2r}_{\mathfrak{T}}(\overline{\mathrm{M}}^{\bullet}_{\mathfrak{p}}(\mathrm{V}^{\circ}, -), L(r))$$

the map by taking cup product with $c_1(\mathcal{L})$, and define the \mathcal{L} -incidence map (on the ground stratum) to be the map

$$\operatorname{inc}_{\mathcal{L}} \colon L[\operatorname{Sh}(\mathrm{V}^{\circ}, -\mathrm{K}_{p}^{\circ})] \bigoplus L[\operatorname{Sh}(\mathrm{V}^{\circ}, -\mathrm{K}_{p}^{\bullet})] \to L[\operatorname{Sh}(\mathrm{V}^{\circ}, -\mathrm{K}_{p}^{\circ})] \bigoplus L[\operatorname{Sh}(\mathrm{V}^{\circ}, -\mathrm{K}_{p}^{\bullet})]$$

in $\operatorname{Fun}(\mathfrak{K}(\mathcal{V}^{\circ})^{p}, \operatorname{\mathsf{Mod}}(L))$ given by the matrix

$$\begin{pmatrix} \operatorname{inc}_{\dagger}^{*} \circ \Theta_{\mathcal{L}} \circ \operatorname{inc}_{!}^{\dagger} & \operatorname{inc}_{\dagger}^{*} \circ \Theta_{\mathcal{L}} \circ \operatorname{inc}_{!}^{\bullet} \\ \operatorname{inc}_{\bullet}^{*} \circ \Theta_{\mathcal{L}} \circ \operatorname{inc}_{!}^{\dagger} & \operatorname{inc}_{\bullet}^{*} \circ \Theta_{\mathcal{L}} \circ \operatorname{inc}_{!}^{\bullet} \end{pmatrix}$$

if we write elements in the column form.

There are three natural choices of \mathcal{L} , which are $\mathcal{O}(M_{\mathfrak{p}}^{\dagger}(V^{\circ}, -))$, $\operatorname{Lie}_{\mathcal{A}, \tau_{\infty}^{\circ}}$, and $\omega_{\mathcal{A}^{\vee}, \tau_{\infty}}$. We now compute $\Theta_{\mathcal{L}}$ for the first two.⁸

Proposition 5.7.8. Suppose N = 2r with $r \ge 2$. Let L be a p-coprime coefficient ring. For $\mathcal{L} = \mathcal{O}(\mathrm{M}_{\mathfrak{p}}^{\dagger}(\mathrm{V}^{\circ}, -)), \text{ the incidence map inc}_{\mathcal{L}} \text{ is given by}$

$$\begin{pmatrix} (p+1)^3 & -(p+1)\mathsf{T}_{N,\mathfrak{p}}^{\circ \bullet} \\ -(p+1)\mathsf{T}_{N,\mathfrak{p}}^{\bullet \circ} & \mathsf{R}_{N,\mathfrak{p}}^{\bullet} \end{pmatrix}$$

where

$$\mathbf{R}_{N,\mathbf{p}}^{\bullet} \coloneqq \sum_{\delta=0}^{r-1} \frac{1 - (-p)^{r-\delta}}{p+1} (p+1)(p+3) \cdots (p^{2(r-\delta)-1}+1) \cdot \mathbf{T}_{N,\mathbf{p};\delta}^{\bullet}$$

in which the Hecke operators $T^{\bullet}_{N,\mathfrak{p};\delta}$ are introduced in Notation B.2.1 (as $T^{\bullet}_{N;\delta}$).

⁷A line bundle \mathcal{L} on $M^{\bullet}_{\mathfrak{p}}(V^{\circ}, -)$ is a collection of a line bundle $\mathcal{L}(K^{p\circ})$ on every $M^{\bullet}_{\mathfrak{p}}(V^{\circ}, K^{p\circ})$, compatible with respect to pullbacks.

⁸In fact, the third one is a linear combination of the first two.

Proof. Take an object $K^{p\circ} \in \mathfrak{K}(V^{\circ})^{p}$.

First, we show $\operatorname{inc}_{\dagger}^* \circ \Theta_{\mathcal{L}} \circ \operatorname{inc}_{!}^{\dagger} = (p+1)^3$. Since $\operatorname{m}^{\dagger \circ *} \mathcal{O}_{\overline{\mathrm{M}}_{\mathfrak{p}}^{\circ}(\mathrm{V}^{\circ},\mathrm{K}^{p\circ})}(1)$ has degree p+1, it follows from Corollary 5.2.5.

Second, we show $\operatorname{inc}_{\dagger}^* \circ \Theta_{\mathcal{L}} \circ \operatorname{inc}_{!}^{\bullet} = -(p+1) \mathsf{T}_{N,\mathfrak{p}}^{\circ \bullet}$ and $\operatorname{inc}_{\bullet}^* \circ \Theta_{\mathcal{L}} \circ \operatorname{inc}_{!}^{\dagger} = -(p+1) \mathsf{T}_{N,\mathfrak{p}}^{\circ \circ}$. However, these are consequences of Corollary 5.2.5, Theorem 5.4.3, and Construction 5.4.5.

It remains to compute $\operatorname{inc}_{\bullet}^{*} \circ \Theta_{\mathcal{L}} \circ \operatorname{inc}_{!}^{\bullet}$. By Theorem 5.6.7(1), it suffices to show that for every $s_{1}^{\bullet}, s_{2}^{\bullet} \in \mathrm{S}_{\mathfrak{p}}^{\bullet}(\mathrm{V}^{\circ}, \mathrm{K}^{p\circ})(\overline{\mathbb{F}}_{p})$ and every $s^{\bullet} \in \mathrm{Hk}_{j}^{-1}(s_{1}^{\bullet}, s_{2}^{\bullet})$, the intersection multiplicity of $\mathrm{B}_{s_{1}^{\bullet}}^{\dagger}$ and $\mathrm{B}_{s_{2}^{\bullet}}^{\dagger}$ at the component B_{s}^{\dagger} equals

$$\frac{1 - (-p)^{r-j}}{p+1}(p+1)(p+3)\cdots(p^{2(r-j)-1}+1).$$

By Theorem 5.6.8 and the excessive intersection formula, such intersection multiplicity equals

$$\sum_{t^{\dagger} \in \mathbf{S}_{s^{\bullet}}^{\dagger}(\overline{\mathbb{F}}_{p})} \int_{\mathbb{P}(\mathscr{Y}_{t^{\dagger}})} c_{r-j-1} \left((\sigma^{*}\mathcal{H}_{t^{\dagger}}) \otimes_{\mathcal{O}_{\mathbb{P}(\mathscr{Y}_{t^{\dagger}})}} \mathcal{O}_{\mathbb{P}(\mathscr{Y}_{t^{\dagger}})}(1) \right).$$

A simple exercise shows that

$$\int_{\mathbb{P}(\mathscr{V}_{t^{\dagger}})} c_{r-j-1}\left(\left(\sigma^{*}\mathcal{H}_{t^{\dagger}}\right) \otimes_{\mathcal{O}_{\mathbb{P}(\mathscr{V}_{t^{\dagger}})}} \mathcal{O}_{\mathbb{P}(\mathscr{V}_{t^{\dagger}})}(1)\right) = \frac{1 - (-p)^{r-j}}{p+1}$$

for every $t^{\dagger} \in S_{s}^{\dagger}(\overline{\mathbb{F}}_{p})$. Thus, the claim follows from Theorem 5.6.8.

Proposition 5.7.9. Suppose N = 2r with $r \ge 2$. Let L be a p-coprime coefficient ring. For $\mathcal{L} = \operatorname{Lie}_{\mathcal{A}, \tau_{\infty}^{c}}$, the incidence map $\operatorname{inc}_{\mathcal{L}}$ is given by

$$\begin{pmatrix} -(p+1)^2 & \mathsf{T}_{N,\mathfrak{p}}^{\bullet\bullet} \\ \mathsf{T}_{N,\mathfrak{p}}^{\bullet\circ} & \mathsf{T}_{N,\mathfrak{p}}^{\bullet} \end{pmatrix}$$

where

$$\mathbf{T}_{N,\mathfrak{p}}^{\bullet} \coloneqq \sum_{\delta=0}^{r-1} \mathbf{d}_{r-\delta,p}^{\bullet} \cdot \mathbf{T}_{N,\mathfrak{p};\delta}^{\bullet}$$

in which the numbers $\mathbf{d}_{r-\delta,p}^{\bullet}$ are introduced in Notation 1.3.2, and the Hecke operators $\mathbf{T}_{N,\mathfrak{p};\delta}^{\bullet}$ are introduced in Notation B.2.1 (as $\mathbf{T}_{N;\delta}^{\bullet}$).

Proof. Take an object $K^{p\circ} \in \mathfrak{K}(V^{\circ})^{p}$. By Theorem 5.2.4, we have an isomorphism

(5.17)
$$\iota^{\bullet*} \operatorname{Lie}_{\mathcal{A}, \tau^{\mathsf{c}}_{\infty}} \simeq \mathcal{O}_{\mathrm{M}^{\dagger}_{\mathfrak{p}}(\mathrm{V}^{\circ}, \mathrm{K}^{p \circ})}(1)$$

of line bundles on $M^{\dagger}_{\mathfrak{p}}(V^{\circ}, K^{p\circ})$.

First, we show $\operatorname{inc}_{\dagger}^* \circ \Theta_{\mathcal{L}} \circ \operatorname{inc}_{!}^{\dagger} = -(p+1)^2$. This is a consequence of (5.17), Corollary 5.2.5 and the fact that $\operatorname{m}^{\dagger \circ *} \mathcal{O}_{\overline{\mathrm{M}}_{p}^{\circ}(\mathrm{V}^{\circ},\mathrm{K}^{p\circ})}(1)$ has degree p+1.

Second, we show $\operatorname{inc}_{\dagger}^* \circ \Theta_{\mathcal{L}} \circ \operatorname{inc}_{!}^{\bullet} = \mathsf{T}_{N,\mathfrak{p}}^{\circ\bullet}$ and $\operatorname{inc}_{\bullet}^* \circ \Theta_{\mathcal{L}} \circ \operatorname{inc}_{!}^{\dagger} = \mathsf{T}_{N,\mathfrak{p}}^{\circ\circ}$. However, these are consequences of (5.17) and Corollary 5.2.5, Theorem 5.4.3, and Construction 5.4.5.

It remains to compute $\operatorname{inc}_{\bullet}^{*} \circ \Theta_{\mathcal{L}} \circ \operatorname{inc}_{!}^{\bullet}$. By Theorem 5.6.7 and the excessive intersection formula, it suffices to show that for every $s_{1}^{\bullet}, s_{2}^{\bullet} \in \mathrm{S}_{\mathfrak{p}}^{\bullet}(\mathrm{V}^{\circ}, \mathrm{K}^{p\circ})(\overline{\mathbb{F}}_{p})$ and every $s^{\bullet} \in \mathrm{Hk}_{j}^{-1}(s_{1}^{\bullet}, s_{2}^{\bullet})$, we have

(5.18)
$$\int_{\mathrm{DL}^{\bullet}(\mathscr{V}_{s^{\bullet}},\{\,,\,\}_{s^{\bullet}})} c_{r-1}\left(\left(\sigma^{*}\bar{\mathcal{H}}_{s^{\bullet}2}\right) \otimes_{\mathcal{O}_{\mathrm{DL}^{\bullet}(\mathscr{V}_{s^{\bullet}},\{\,,\,\}_{s^{\bullet}})}} \left(\bar{\mathcal{H}}_{s^{\bullet}1}^{\dashv}/\bar{\mathcal{H}}_{s^{\bullet}2}\right)\right) \cdot c_{1}\left(\left(\zeta_{s^{\bullet}}^{\bullet}\right)_{*}\mathrm{Lie}_{\mathcal{A},\tau_{\infty}^{c}}\right) = \mathsf{d}_{r,p}^{\bullet}$$

where $(\bar{\mathcal{H}}_{s^{\bullet}1}, \bar{\mathcal{H}}_{s^{\bullet}2})$ is the universal object over $DL^{\bullet}(\mathscr{V}_{s^{\bullet}}, \{, \}_{s^{\bullet}})$. However, by Theorem 5.3.4(4), we have $(\zeta_{s^{\bullet}}^{\bullet})_* \operatorname{Lie}_{\mathcal{A}, \tau_{\infty}^{c}} \simeq \bar{\mathcal{H}}_{s^{\bullet}1}^{\dashv}/\bar{\mathcal{H}}_{s^{\bullet}2}$. Thus, (5.18) follows from Proposition A.2.4(2). The proposition is proved.

5.8. Weight spectral sequence. In this subsection, we study the weight spectral sequence associated to $\mathbf{M}_{\mathfrak{p}}(\mathcal{V}^{\circ}, -)$. We keep the setup in Subsection 5.7. In particular, N is an integer at least 2 with $r \coloneqq \lfloor \frac{N}{2} \rfloor \ge 1$, and L is a *p*-coprime coefficient ring. To ease notation, we put $X_N^? \coloneqq X_{\mathfrak{p}}^?(\mathcal{V}^{\circ}, -)$ for meaningful pairs $(X, ?) \in \{\mathbf{M}, \mathbf{M}, \mathbf{B}, \mathbf{S}\} \times \{ , \circ, \bullet, \dagger \}$.

Construction 5.8.1. By Theorem 5.1.5(1), we have the weight spectral sequence $(\mathbf{E}_{s}^{p,q}, \mathbf{d}_{s}^{p,q})$, with terms in the category $L[\operatorname{Gal}(\overline{\mathbb{F}}_{p}/\mathbb{F}_{p}^{\Phi})]$, abutting to the cohomology $\mathrm{H}_{\mathfrak{T}}^{p+q}(\overline{\mathrm{M}}_{N}, \mathrm{R}\Psi L(r))$. In particular, we have

$$\mathbf{E}_{1}^{0,2d} = \mathbf{H}_{\mathfrak{T}}^{2d}(\overline{\mathbf{M}}_{N}^{\circ}, L(r)) \bigoplus \mathbf{H}_{\mathfrak{T}}^{2d}(\overline{\mathbf{M}}_{N}^{\bullet}, L(r)).$$

Thus, the six maps in Construction 5.7.3 give rise to another six maps

$$\begin{cases} \operatorname{Inc}_{!}^{\circ} \colon L[\operatorname{Sh}(\operatorname{V}^{\circ}, -\operatorname{K}_{\mathfrak{p}}^{\circ})] \to \operatorname{E}_{1}^{0,2(N-r-1)}(N-2r-1), \\ \operatorname{Inc}_{!}^{\dagger} \colon L[\operatorname{Sh}(\operatorname{V}^{\circ}, -\operatorname{K}_{\mathfrak{p}}^{\circ})] \to \operatorname{E}_{1}^{0,2(N-r-1)}(N-2r-1), \\ \operatorname{Inc}_{!}^{\bullet} \colon L[\operatorname{Sh}(\operatorname{V}^{\circ}, -\operatorname{K}_{\mathfrak{p}}^{\bullet})] \to \operatorname{E}_{1}^{0,2(N-r-1)}(N-2r-1), \\ \operatorname{Inc}_{\circ}^{\bullet} \colon \operatorname{E}_{1}^{0,2r} \to L[\operatorname{Sh}(\operatorname{V}^{\circ}, -\operatorname{K}_{\mathfrak{p}}^{\circ})], \\ \operatorname{Inc}_{\dagger}^{*} \colon \operatorname{E}_{1}^{0,2r} \to L[\operatorname{Sh}(\operatorname{V}^{\circ}, -\operatorname{K}_{\mathfrak{p}}^{\circ})], \\ \operatorname{Inc}_{\bullet}^{*} \colon \operatorname{E}_{1}^{0,2r} \to L[\operatorname{Sh}(\operatorname{V}^{\circ}, -\operatorname{K}_{\mathfrak{p}}^{\circ})], \end{cases} \end{cases}$$

in $\operatorname{Fun}(\mathfrak{K}(\mathcal{V}^{\circ})^p, \operatorname{\mathsf{Mod}}(L)).$

In the future, we will have to study the composite maps

$$\begin{pmatrix} \operatorname{Inc}_{\circ}^{\circ} \\ \operatorname{Inc}_{\dagger}^{*} \\ \operatorname{Inc}_{\bullet}^{*} \end{pmatrix} \begin{pmatrix} \operatorname{Inc}_{!}^{\circ} & \operatorname{Inc}_{!}^{\dagger} & \operatorname{Inc}_{!}^{\bullet} \end{pmatrix}, \quad \begin{pmatrix} \operatorname{Inc}_{\circ}^{*} \\ \operatorname{Inc}_{\dagger}^{*} \\ \operatorname{Inc}_{\bullet}^{*} \end{pmatrix} \circ \operatorname{d}_{1}^{-1,2r} \circ \operatorname{d}_{1}^{0,2r-2}(-1) \circ \left(\operatorname{Inc}_{!}^{\circ} & \operatorname{Inc}_{!}^{\dagger} & \operatorname{Inc}_{!}^{\bullet} \right)$$

when N is odd and even, respectively. In the next two lemmas, we will study the spectral sequence and prove two formulae related to the above maps, according to the parity of N.

Lemma 5.8.2. Suppose N = 2r + 1 odd with $r \ge 1$.

(1) The first page of $E_s^{p,q}$ is as follows:

$$\begin{split} q &\geq 2r+2 & \cdots \longrightarrow \cdots \longrightarrow \cdots \\ q &= 2r+1 & H_{\mathfrak{T}}^{2r-1}(\overline{\mathrm{M}}_{N}^{\dagger}, L(r-1))^{\overset{\mathbf{d}_{1}^{-1,2r+1}}{\longrightarrow}} H_{\mathfrak{T}}^{2r+1}(\overline{\mathrm{M}}_{N}^{\circ}, L(r)) \oplus H_{\mathfrak{T}}^{2r+1}(\overline{\mathrm{M}}_{N}^{\bullet}, L(r))^{\overset{\mathbf{d}_{1}^{0,2r+1}}{\longrightarrow}} H_{\mathfrak{T}}^{2r+1}(\overline{\mathrm{M}}_{N}^{\dagger}, L(r)) \\ q &= 2r & H_{\mathfrak{T}}^{2r-2}(\overline{\mathrm{M}}_{N}^{\dagger}, L(r-1)) \xrightarrow{\overset{\mathbf{d}_{1}^{-1,2r}}{\longrightarrow}} H_{\mathfrak{T}}^{2r}(\overline{\mathrm{M}}_{N}^{\circ}, L(r)) \oplus H_{\mathfrak{T}}^{2r}(\overline{\mathrm{M}}_{N}^{\bullet}, L(r)) \xrightarrow{\overset{\mathbf{d}_{1}^{0,2r}}{\longrightarrow}} H_{\mathfrak{T}}^{2r}(\overline{\mathrm{M}}_{N}^{\dagger}, L(r)) \\ q &= 2r-1 & H_{\mathfrak{T}}^{2r-3}(\overline{\mathrm{M}}_{N}^{\dagger}, L(r-1))^{\overset{\mathbf{d}_{1}^{-1,2r-1}}{\longrightarrow}} H_{\mathfrak{T}}^{2r-1}(\overline{\mathrm{M}}_{N}^{\circ}, L(r)) \oplus H_{\mathfrak{T}}^{2r-1}(\overline{\mathrm{M}}_{N}^{\bullet}, L(r)) \xrightarrow{\overset{\mathbf{d}_{1}^{0,2r-1}}{\longrightarrow}} H_{\mathfrak{T}}^{2r-1}(\overline{\mathrm{M}}_{N}^{\dagger}, L(r)) \\ q &\leq 2r-2 & \cdots & \longrightarrow & \cdots \\ E_{1}^{p,q} & p &= -1 & p &= 0 & p &= 1 \end{split}$$

with $d_1^{-1,i} = (m_!^{\dagger \circ}, -m_!^{\dagger \bullet}), \ d_1^{0,i} = (m^{\dagger \circ})^* - (m^{\dagger \bullet})^*$ for every $i \in \mathbb{Z}$; and $E_1^{p,q} = 0$ if |p| > 1.

(2) We have

$$\begin{pmatrix} \operatorname{Inc}^*_{\circ} \\ \operatorname{Inc}^*_{\dagger} \\ \operatorname{Inc}^*_{\bullet} \end{pmatrix} \begin{pmatrix} \operatorname{Inc}^{\circ}_{!} & \operatorname{Inc}^{\dagger}_{!} & \operatorname{Inc}^{\bullet}_{!} \end{pmatrix} = \begin{pmatrix} 1 & 0 & 0 \\ 0 & -(p+1)^2 & \mathsf{T}^{\circ\bullet}_{N,\mathfrak{p}} \\ 0 & \mathsf{T}^{\bullet\circ}_{N,\mathfrak{p}} & \mathsf{T}^{\bullet}_{N,\mathfrak{p}} \end{pmatrix}.$$

(3) We have $(\mathbb{T}_{N,\mathfrak{p}}^{\bullet\circ} \circ \operatorname{Inc}_{\dagger}^* + (p+1)^2 \operatorname{Inc}_{\bullet}^*) \circ d_1^{-1,2r} = 0.$

Proof. Part (1) is immediate. Part (2) is a consequence of Proposition 5.7.6.

For (3), note that under the composite isomorphism

$$i\colon L[\operatorname{Sh}(\operatorname{V}^{\circ}, -\operatorname{K}^{\circ}_{\mathfrak{p}})] \xrightarrow{\sim} \operatorname{H}^{0}_{\mathfrak{T}}(\overline{\operatorname{S}}^{\circ}_{N}, L) \xrightarrow{\pi^{\circ*}} \operatorname{H}^{0}_{\mathfrak{T}}(\overline{\operatorname{B}}^{\circ}_{N}, L) \xrightarrow{\cup \xi^{r-1}} \operatorname{H}^{2r-2}_{\mathfrak{T}}(\overline{\operatorname{B}}^{\circ}_{N}, L(r-1)) \xrightarrow{\iota^{\circ}_{\mathfrak{T}}} \operatorname{H}^{2r-2}_{\mathfrak{T}}(\overline{\operatorname{M}}^{\circ}_{N}, L(r-1)) \xrightarrow{\operatorname{m}^{\dagger \circ*}} \operatorname{H}^{2r-2}_{\mathfrak{T}}(\overline{\operatorname{M}}^{\dagger}_{N}, L(r-1)) = \operatorname{E}^{-1,2r}_{1},$$

the map $d_1^{-1,2r} \circ i$: $L[Sh(V^\circ, -K_p^\circ)] \to E_1^{0,2r}$ coincides with $(p+1)Inc_!^\circ - Inc_!^\dagger$. Thus, (3) follows by (2) as we have

$$\begin{pmatrix} 0 & \mathsf{T}_{N,\mathfrak{p}}^{\bullet\circ} & (p+1)^2 \end{pmatrix} \begin{pmatrix} 1 & 0 & 0 \\ 0 & -(p+1)^2 & \mathsf{T}_{N,\mathfrak{p}}^{\circ\bullet} \\ 0 & \mathsf{T}_{N,\mathfrak{p}}^{\bullet\circ} & \mathsf{T}_{N,\mathfrak{p}}^{\bullet} \end{pmatrix} \begin{pmatrix} p+1 \\ -1 \\ 0 \end{pmatrix} = 0.$$

The lemma is proved.

For N even, we first recall that there is an (increasing) monodromy filtration $F_{\bullet}R\Psi L(r)$ of $R\Psi L(r)$. Such filtration induces a filtration $F_{\bullet}H^{i}_{\mathfrak{T}}(\overline{M}_{N}, R\Psi L(r))$ of the cohomology $H^{i}_{\mathfrak{T}}(\overline{M}_{N}, R\Psi L(r))$, and a corresponding filtration $F_{\bullet}H^{1}(I_{\mathbb{Q}_{p}^{\Phi}}, H^{i}_{\mathfrak{T}}(\overline{M}_{N}, R\Psi L(r)))$ of the quotient module $H^{1}(I_{\mathbb{Q}_{p}^{\Phi}}, H^{i}_{\mathfrak{T}}(\overline{M}_{N}, R\Psi L(r)))$.

Lemma 5.8.3. Suppose N = 2r even with $r \ge 1$.

(1) The first page of $E_s^{p,q}$ is as follows:

$$\begin{split} q &\geq 2r+1 \qquad \cdots \longrightarrow \cdots \longrightarrow \cdots \longrightarrow \cdots \\ q &= 2r \qquad H_{\mathfrak{T}}^{2r-2}(\overline{\mathrm{M}}_{N}^{\dagger}, L(r-1)) \xrightarrow{\mathrm{d}_{1}^{-1,2r}} \mathrm{H}_{\mathfrak{T}}^{2r}(\overline{\mathrm{M}}_{N}^{\circ}, L(r)) \oplus \mathrm{H}_{\mathfrak{T}}^{2r}(\overline{\mathrm{M}}_{N}^{\bullet}, L(r)) \xrightarrow{\mathrm{d}_{1}^{0,2r}} \mathrm{H}_{\mathfrak{T}}^{2r}(\overline{\mathrm{M}}_{N}^{\dagger}, L(r)) \\ q &= 2r-1 \qquad 0 \xrightarrow{\qquad} \mathrm{H}_{\mathfrak{T}}^{2r-1}(\overline{\mathrm{M}}_{N}^{\bullet}, L(r)) \longrightarrow 0 \\ q &= 2r-2 \qquad \mathrm{H}_{\mathfrak{T}}^{2r-4}(\overline{\mathrm{M}}_{N}^{\dagger}, L(r-1))^{\mathrm{d}_{1}^{-1,2r-2}} \mathrm{H}_{\mathfrak{T}}^{2r-2}(\overline{\mathrm{M}}_{N}^{\circ}, L(r)) \oplus \mathrm{H}_{\mathfrak{T}}^{2r-2}(\overline{\mathrm{M}}_{N}^{\bullet}, L(r)) \xrightarrow{\mathrm{d}_{1}^{0,2r-2}} \mathrm{H}_{\mathfrak{T}}^{2r-2}(\overline{\mathrm{M}}_{N}^{\dagger}, L(r)) \\ q &\leq 2r-3 \qquad \cdots \longrightarrow \cdots \\ \mathrm{E}_{1}^{p,q} \qquad p &= -1 \qquad p = 0 \qquad p = 1 \end{split}$$

with $d_1^{-1,i} = (m_!^{\dagger \circ}, -m_!^{\dagger \bullet}), d_1^{0,i} = (m^{\dagger \circ})^* - (m^{\dagger \bullet})^*$ for every $i \in \mathbb{Z}$; and $E_1^{p,q} = 0$ if |p| > 1. (2) The spectral sequence $E_s^{p,q}$ degenerates at the second page.

(3) In the (three-step) filtration $F_{\bullet}H_{\mathfrak{T}}^{2r-1}(\overline{M}_N, R\Psi L(r))$, we have canonical isomorphisms

$$\begin{cases} F_{-1}H_{\mathfrak{T}}^{2r-1}(\overline{M}_{N}, R\Psi L(r)) \simeq E_{2}^{1,2r-2} = \operatorname{coker} d_{1}^{0,2r-2}, \\ \frac{F_{0}H_{\mathfrak{T}}^{2r-1}(\overline{M}_{N}, R\Psi L(r))}{F_{-1}H_{\mathfrak{T}}^{2r-1}(\overline{M}_{N}, R\Psi L(r))} \simeq E_{2}^{0,2r-1} = H_{\mathfrak{T}}^{2r-1}(\overline{M}_{N}^{\bullet}, L(r)), \\ \frac{H_{\mathfrak{T}}^{2r-1}(\overline{M}_{N}, R\Psi L(r))}{F_{0}H_{\mathfrak{T}}^{2r-1}(\overline{M}_{N}, R\Psi L(r))} \simeq E_{2}^{-1,2r} = \ker d_{1}^{-1,2r-2}, \end{cases}$$

in $\operatorname{Fun}(\mathfrak{K}(\mathcal{V}^{\circ})^{p}, \operatorname{\mathsf{Mod}}(L[\operatorname{Gal}(\overline{\mathbb{F}}_{p}/\mathbb{F}_{p}^{\Phi})])).$

(4) The monodromy map on $\mathrm{H}_{\mathfrak{T}}^{2r-1}(\overline{\mathrm{M}}_N, \mathrm{R}\Psi L(r))$ factors through $\mathrm{F}_0\mathrm{H}_{\mathfrak{T}}^{2r-1}(\overline{\mathrm{M}}_N, \mathrm{R}\Psi L(r))$ and is given by the composite map

$$\mathbf{E}_{2}^{-1,2r} \xrightarrow{\mu} \mathbf{E}_{2}^{1,2r-2} \hookrightarrow \mathbf{H}_{\mathfrak{T}}^{2r-1}(\overline{\mathbf{M}}_{N}, \mathbf{R}\Psi L(r))$$

where μ is the map induced from the identity map on $\mathrm{H}_{\mathfrak{T}}^{2r-2}(\overline{\mathrm{M}}_{N}^{\dagger}, L(r-1))$. (5) We have a canonical isomorphism

$$\mathbf{F}_{-1}\mathbf{H}^{1}(\mathbf{I}_{\mathbb{Q}_{p}^{\Phi}},\mathbf{H}_{\mathfrak{T}}^{2r-1}(\overline{\mathbf{M}}_{N},\mathbf{R}\Psi L(r))) \simeq \left(\frac{\mathbf{E}_{2}^{1,2r-2}}{\mu\mathbf{E}_{2}^{-1,2r}}\right)(-1)$$

in $\operatorname{Fun}(\mathfrak{K}(\mathcal{V}^{\circ})^{p}, \operatorname{Mod}(L[\operatorname{Gal}(\overline{\mathbb{F}}_{p}/\mathbb{F}_{p}^{\Phi})]));$ and the map $\operatorname{d}_{1}^{-1,2r}$ induces an isomorphism

$$\left(\frac{\mathbf{E}_{2}^{1,2r-2}}{\mu\mathbf{E}_{2}^{-1,2r}}\right)(-1) \simeq \frac{\operatorname{im} \mathbf{d}_{1}^{-1,2r}}{\operatorname{im}(\mathbf{d}_{1}^{-1,2r} \circ \mathbf{d}_{1}^{0,2r-2}(-1))}$$

in $\operatorname{Fun}(\mathfrak{K}(\mathcal{V}^{\circ})^{p}, \operatorname{\mathsf{Mod}}(L[\operatorname{Gal}(\overline{\mathbb{F}}_{p}/\mathbb{F}_{p}^{\Phi})])).$

(6) If $p^2 - 1$ is invertible in L, then we have a canonical short exact sequence

 $0 \longrightarrow \mathcal{F}_{-1}\mathcal{H}^{1}(\mathcal{I}_{\mathbb{Q}_{p}^{\Phi}}, \mathcal{H}_{\mathfrak{T}}^{2r-1}(\overline{\mathcal{M}}_{N}, \mathcal{R}\Psi L(r))) \longrightarrow \mathcal{H}^{1}_{\mathrm{sing}}(\mathbb{Q}_{p}^{\Phi}, \mathcal{H}_{\mathfrak{T}}^{2r-1}(\overline{\mathcal{M}}_{N}, \mathcal{R}\Psi L(r))) \longrightarrow \mathcal{H}^{2r-1}_{\mathfrak{T}}(\overline{\mathcal{M}}_{N}^{\bullet}, L(r-1))^{\mathrm{Gal}(\overline{\mathbb{F}}_{p}/\mathbb{F}_{p}^{\Phi})} \to 0$

 $in \operatorname{Fun}(\mathfrak{K}(\mathcal{V}^{\circ})^{p}, \operatorname{\mathsf{Mod}}(L)).$

(7) The composite map

$$\begin{pmatrix} \operatorname{Inc}_{\circ}^{*} \\ \operatorname{Inc}_{\dagger}^{*} \\ \operatorname{Inc}_{\bullet}^{*} \end{pmatrix} \circ d_{1}^{-1,2r} \circ d_{1}^{0,2r-2}(-1) \circ \left(\operatorname{Inc}_{!}^{\circ} \operatorname{Inc}_{!}^{\dagger} \operatorname{Inc}_{!}^{\bullet} \right)$$

coincides with

$$\begin{pmatrix} p+1 & (p+1)^2 & -\mathsf{T}_{N,\mathfrak{p}}^{\circ\bullet} \\ (p+1)^2 & (p+1)^3 & -(p+1)\mathsf{T}_{N,\mathfrak{p}}^{\circ\bullet} \\ -\mathsf{T}_{N,\mathfrak{p}}^{\bullet\circ} & -(p+1)\mathsf{T}_{N,\mathfrak{p}}^{\bullet\circ} & \mathsf{R}_{N,\mathfrak{p}}^{\bullet} \end{pmatrix}, \quad \begin{pmatrix} p+1 & 0 & -\mathsf{T}_{N,\mathfrak{p}}^{\circ\bullet} \\ 0 & 0 & 0 \\ -\mathsf{T}_{N,\mathfrak{p}}^{\bullet\circ} & 0 & \mathsf{R}_{N,\mathfrak{p}}^{\bullet} \end{pmatrix}$$

when $N \ge 4$ and when N = 2, respectively. (8) The image of the map

$$(\mathbf{T}_{N,\mathfrak{p}}^{\bullet\circ} \circ \operatorname{Inc}_{\circ}^{*} + (p+1)\operatorname{Inc}_{\bullet}^{*}) \circ \mathbf{d}_{1}^{-1,2r} \circ \mathbf{d}_{1}^{0,2r-2}(-1) \circ (\operatorname{Inc}_{!}^{\circ} + \operatorname{Inc}_{!}^{\dagger} + \operatorname{Inc}_{!}^{\bullet}):$$

$$L[\operatorname{Sb}(\mathcal{V}^{\circ} - \mathcal{K}^{\circ})]^{\oplus 2} \bigoplus L[\operatorname{Sb}(\mathcal{V}^{\circ} - \mathcal{K}^{\bullet})] \rightarrow L[\operatorname{Sb}(\mathcal{V}^{\circ} - \mathcal{K}^{\bullet})]$$

$$L[\operatorname{Sn}(V, -\mathbf{K}_{\mathfrak{p}})]^{\oplus^{\bullet}} \bigoplus L[\operatorname{Sn}(V, -\mathbf{K}_{\mathfrak{p}})] \to L[\operatorname{Sn}(V, -\mathbf{K}_{\mathfrak{p}})]$$

is exactly $((p+1)\mathbf{R}^{\bullet}_{N,\mathfrak{p}} - \mathsf{T}^{\bullet\circ}_{N,\mathfrak{p}} \circ \mathsf{T}^{\circ\bullet}_{N,\mathfrak{p}})L[Sh(\mathbf{V}^{\circ}, -\mathbf{K}^{\bullet}_{\mathfrak{p}})]$, where $\mathbf{R}^{\bullet}_{N,\mathfrak{p}}$ is introduced in Proposition 5.7.8.

Proof. For (1), note that by Lemma 5.5.2(1), both $\mathrm{H}^{i}_{\mathfrak{T}}(\overline{\mathrm{M}}^{\dagger}_{N}, L)$ and $\mathrm{H}^{i}_{\mathfrak{T}}(\overline{\mathrm{M}}^{\circ}_{N}, L)$ vanish for *i* odd. Thus, (1) follows.

Parts (2–4) follow directly from the description of $E_1^{p,q}$ and [Sai03, Corollary 2.8](2) for the description of the monodromy map⁹. Part (5) follows from (1–4).

For (6), by Lemma 5.5.2(3), we know that the action of $\operatorname{Gal}(\overline{\mathbb{F}}_p/\mathbb{F}_p^{\Phi})$ on $\operatorname{E}_2^{1,2r-2}(-1)$ is trivial. As $p^2 - 1$ is invertible in L, we further have $\operatorname{E}_2^{-1,2r}(-1)^{\operatorname{Gal}(\overline{\mathbb{F}}_p/\mathbb{F}_p^{\Phi})} = 0$ and

$$\mathrm{H}^{1}(\mathrm{Gal}(\overline{\mathbb{F}}_{p}/\mathbb{F}_{p}^{\Phi}),\mathrm{F}_{-1}\mathrm{H}^{1}(\mathrm{I}_{\mathbb{Q}_{p}^{\Phi}},\mathrm{H}^{2r-1}_{\mathfrak{T}}(\overline{\mathrm{M}}_{N},\mathrm{R}\Psi L(r))))=0.$$

In particular, we have the isomorphism

$$H^{1}_{\text{sing}}(\mathbb{Q}_{p}^{\Phi}, \mathrm{H}^{2r-1}_{\mathfrak{T}}(\overline{\mathrm{M}}_{N}, \mathrm{R}\Psi L(r))) \simeq \mathrm{H}^{1}(\mathrm{I}_{\mathbb{Q}_{p}^{\Phi}}, \mathrm{H}^{2r-1}_{\mathfrak{T}}(\overline{\mathrm{M}}_{N}, \mathrm{R}\Psi L(r)))^{\mathrm{Gal}(\mathbb{F}_{p}/\mathbb{F}_{p}^{\Phi})} \\ \simeq \mathrm{F}_{0}\mathrm{H}^{1}(\mathrm{I}_{\mathbb{Q}_{p}^{\Phi}}, \mathrm{H}^{2r-1}_{\mathfrak{T}}(\overline{\mathrm{M}}_{N}, \mathrm{R}\Psi L(r)))^{\mathrm{Gal}(\overline{\mathbb{F}}_{p}/\mathbb{F}_{p}^{\Phi})},$$

and that (6) follows from the induced long exact sequence.

For (7), when $N \ge 4$ (that is, $r \ge 2$), it follows from Theorem 5.2.4(2) and Proposition 5.7.8. When N = 2, it follows from a direct computation.

For (8), we have the identity

$$\begin{pmatrix} \mathsf{T}_{N,\mathfrak{p}}^{\bullet\circ} & 0 & p+1 \end{pmatrix} \begin{pmatrix} \mathrm{Inc}_{\circ}^{*} \\ \mathrm{Inc}_{\dagger}^{*} \\ \mathrm{Inc}_{\bullet}^{*} \end{pmatrix} \circ \mathbf{d}_{1}^{-1,2r} \circ \mathbf{d}_{1}^{0,2r-2}(-1) \circ \left(\mathrm{Inc}_{!}^{\circ} & \mathrm{Inc}_{!}^{\dagger} & \mathrm{Inc}_{!}^{\bullet} \right)$$
$$= \begin{pmatrix} 0 & 0 & (p+1)\mathsf{R}_{N,\mathfrak{p}}^{\bullet} - \mathsf{T}_{N,\mathfrak{p}}^{\bullet\circ} \circ \mathsf{T}_{N,\mathfrak{p}}^{\circ\bullet} \end{pmatrix}$$

by (7), which implies (8).

The lemma is all proved.

Construction 5.8.4. We construct

(1) when N = 2r + 1 is odd, the map

$$\nabla^1 \colon \mathrm{E}_2^{0,2r} \to L[\mathrm{Sh}(\mathrm{V}_N^\circ,\mathrm{K}_N^\circ)]$$

to be restriction of the map

$$\mathbb{T}_{N,\mathfrak{p}}^{\bullet\circ} \circ \operatorname{Inc}_{\dagger}^* + (p+1)^2 \operatorname{Inc}_{\bullet}^* \colon \mathcal{E}_1^{0,2r-1} \to L[\operatorname{Sh}(\mathcal{V}_N^\circ, \mathcal{K}_N^\bullet)],$$

to ker $d_1^{0,2r-1}$, which factors through $E_2^{0,2r}$ by Lemma 5.8.2(3), composed with the map $T_{N,\mathfrak{p}}^{\circ\bullet}: L[Sh(V_N^{\circ}, K_N^{\circ})] \to L[Sh(V_N^{\circ}, K_N^{\circ})];$

(2) when N = 2r is even, the map

$$\nabla^0 \colon \ker \mathrm{d}_1^{0,2r} \to L[\mathrm{Sh}(\mathrm{V}_N^\circ,\mathrm{K}_N^\circ)]$$

to be restriction of the map

$$\mathbf{T}_{N,\mathfrak{p}}^{\bullet\circ} \circ \operatorname{Inc}_{\circ}^{*} + (p+1)\operatorname{Inc}_{\bullet}^{*} \colon \mathrm{E}_{1}^{0,2r} \to L[\operatorname{Sh}(\mathrm{V}_{N}^{\circ},\mathrm{K}_{N}^{\bullet})]$$

in Lemma 5.8.3(8) to ker $d_1^{0,2r}$, composed with the map $T_{N,\mathfrak{p}}^{\circ\bullet}: L[Sh(V_N^\circ, K_N^\bullet)] \to L[Sh(V_N^\circ, K_N^\circ)].$

Remark 5.8.5. By the descriptions of the Galois actions in Construction 5.2.6 and Construction 5.3.6, the map ∇^1 factors through the quotient map $E_2^{0,2r} \to (E_2^{0,2r})_{\text{Gal}(\overline{\mathbb{F}}_n/\mathbb{F}_n^{\Phi})}$.

To temporarily end the discussion on weight spectral sequences, we record the following easy lemma, which will be used later.

⁹The description of the monodromy map does not require that the scheme is proper over the base.

Lemma 5.8.6. Suppose $N \geq 3$. The following diagram

$$\begin{array}{ccc} \mathbf{E}_{1}^{0,2r} & \xrightarrow{(\mathrm{Inc}_{\mathfrak{p}}^{*},\mathrm{Inc}_{\mathfrak{p}}^{*})} & L[\mathrm{Sh}(\mathrm{V}^{\circ},-\mathrm{K}_{\mathfrak{p}}^{\circ})]^{\oplus 2} \bigoplus L[\mathrm{Sh}(\mathrm{V}^{\circ},-\mathrm{K}_{\mathfrak{p}}^{\bullet})] \\ \stackrel{\mathrm{d}_{1}^{0,2r}}{\overset{|}{\underset{1}}} & & \downarrow^{(p+1,-1,0)} \\ & \mathbf{E}_{1}^{1,2r} & \xrightarrow{} & L[\mathrm{Sh}(\mathrm{V}^{\circ},-\mathrm{K}_{\mathfrak{p}}^{\circ})] \end{array}$$

is commutative, where the lower arrow is the composite map

$$\begin{aligned} \mathrm{H}^{2r}_{\mathfrak{T}}(\overline{\mathrm{M}}^{\dagger}_{\mathfrak{p}}(\mathrm{V}^{\circ},-),L(r)) \xrightarrow{\mathrm{m}^{\circ}_{1}} \mathrm{H}^{2(r+1)}_{\mathfrak{T}}(\overline{\mathrm{M}}^{\circ}_{\mathfrak{p}}(\mathrm{V}^{\circ},-),L(r+1)) \xrightarrow{\iota^{\circ*}} \mathrm{H}^{2(r+1)}_{\mathfrak{T}}(\overline{\mathrm{B}}^{\circ}_{\mathfrak{p}}(\mathrm{V}^{\circ},-),L(r+1)) \\ \xrightarrow{\cup \xi^{N-r-2}} \mathrm{H}^{2(N-1)}_{\mathfrak{T}}(\overline{\mathrm{B}}^{\circ}_{\mathfrak{p}}(\mathrm{V}^{\circ},-),L(N-1)) \xrightarrow{\pi^{\circ}_{1}} \mathrm{H}^{0}_{\mathfrak{T}}(\overline{\mathrm{S}}^{\circ}_{\mathfrak{p}}(\mathrm{V}^{\circ},-),L) \xrightarrow{\sim} L[\mathrm{Sh}(\mathrm{V}^{\circ},-\mathrm{K}^{\circ}_{p})], \end{aligned}$$

which is an isomorphism.

Proof. The commutativity of the diagram follows from the formula $d_1^{0,2r} = (m^{\dagger \circ})^* - (m^{\dagger \bullet})^*$, and the fact that $M_{\mathfrak{p}}^{\dagger}(V^{\circ}, -)$ is a hypersurface in $M_{\mathfrak{p}}^{\circ}(V^{\circ}, -)$ of degree p + 1 by Theorem 5.2.4 and Lemma A.1.4(1). By Lemma 5.5.2 and Poincaré duality, the lower arrow is an isomorphism. \Box

5.9. Special results in the rank 3 case. In this subsection, we study some special properties of the ground stratum $M_{\mathfrak{p}}^{\bullet}(V^{\circ}, -)$ when N = 3. The results here will only be used in the situation (b) of Lemma 8.1.4 and are only necessary for the main theorems in Subsection 1.1 in the case where n = 2 and $F^+ = \mathbb{Q}$, so readers may skip this subsection at this moment.

To begin with, we recall the following definition.

Definition 5.9.1 ([FK88, Chapter I. Definition 3.7 & Note 3.10]). A proper morphism $f: X \to Y$ of schemes of characteristic p is *purely inseparable* if the following two equivalent conditions hold:

- (1) For every (scheme-theoretical) point y of Y, there lies exactly one point x of X, and the residue field extension is purely inseparable.
- (2) For every algebraically field κ of characteristic p, the induced map $f(\kappa): X(\kappa) \to Y(\kappa)$ is a bijection.

We now assume $\dim_F V^\circ = N = 3$.

Definition 5.9.2. We define a functor

$$\begin{split} \mathbf{M}'_{\mathfrak{p}}(\mathbf{V}^{\circ},-) \colon \mathfrak{K}(\mathbf{V}^{\circ})^{p} \times \mathfrak{T} &\to \mathsf{PSch}'_{\mathbb{Z}_{p}^{\Phi}}\\ \mathbf{K}^{p \circ} &\mapsto \mathbf{M}'_{\mathfrak{n}}(\mathbf{V}^{\circ},\mathbf{K}^{p \circ}) \end{split}$$

such that for every $S \in \mathsf{Sch}'_{\mathbb{Z}_p^{\Phi}}, \mathbf{M}'_{\mathfrak{p}}(\mathcal{V}^\circ, \mathcal{K}^{p\circ})(S)$ is the set of equivalence classes of sextuples $(A_0, \lambda_0, \eta_0^p; A', \lambda', \eta^{p'})$ where

- O $(A_0, \lambda_0, \eta_0^p)$ is an element in $\mathbf{T}_{\mathfrak{p}}(S)$;
- O (A', λ') is a unitary O_F -abelian scheme of signature type $3\Phi 2\tau_{\infty} + 2\tau_{\infty}^{c}$ over S (Definitions 3.4.2 and 3.4.3) such that λ' is p-principal;
- $O \eta^{p'}$ is a $K^{p\circ}$ -level structure, that is, for a chosen geometric point s on every connected component of S, a $\pi_1(S, s)$ -invariant $K^{p\circ}$ -orbit of isomorphisms

$$\eta^{p'} \colon \mathcal{V}^{\circ} \otimes_{\mathbb{Q}} \mathbb{A}^{\infty, p} \to \operatorname{Hom}_{F \otimes_{\mathbb{Q}} \mathbb{A}^{\infty, p}}^{\lambda_{0}, \lambda'}(\mathcal{H}_{1}^{\operatorname{\acute{e}t}}(A'_{0s}, \mathbb{A}^{\infty, p}), \mathcal{H}_{1}^{\operatorname{\acute{e}t}}(A'_{s}, \mathbb{A}^{\infty, p}))$$

of hermitian spaces over $F \otimes_{\mathbb{Q}} \mathbb{A}^{\infty,p} = F \otimes_{F^+} \mathbb{A}^{\infty,p}_{F^+}$. See Construction 3.4.4 (with $\square = \{\infty, p\}$) for the right-hand side.

The equivalence relation and the action of morphisms in $\mathfrak{K}(\mathcal{V}^{\circ})^{p} \times \mathfrak{T}$ are defined similarly as in Definition 4.1.2.

We have apparently the forgetful morphism

$$(5.19) \mathbf{M}'_{\mathfrak{p}}(\mathbf{V}^{\circ}, -) \to \mathbf{T}_{\mathfrak{p}}$$

in $\operatorname{Fun}(\mathfrak{K}(V^{\circ})^{p} \times \mathfrak{T}, \operatorname{PSch}_{\mathbb{Z}_{p}^{\Phi}})$. By a similar proof of Theorem 4.1.3, the morphism (5.19) is represented by quasi-projective smooth schemes of relative dimension 2. We denote by the base change of (5.19) to \mathbb{F}_{p}^{Φ} by $\operatorname{M}_{\mathfrak{p}}'(V^{\circ}, -) \to \operatorname{T}_{\mathfrak{p}}$, which is a morphism in $\operatorname{Fun}(\mathfrak{K}(V^{\circ})^{p} \times \mathfrak{T}, \operatorname{Sch}_{\mathbb{F}_{p}^{\Phi}})$.

Definition 5.9.3. We define a functor

$$N_{\mathfrak{p}}(V^{\circ}, -) \colon \mathfrak{K}(V^{\circ})^{p} \times \mathfrak{T} \to \mathsf{PSch}'_{/\mathbb{F}_{p}^{\Phi}}$$
$$K^{p \circ} \mapsto N_{\mathfrak{p}}(V^{\circ}, K^{p \circ})$$

such that for every $S \in \mathsf{Sch}'_{/\mathbb{F}_p^{\Phi}}$, $N_{\mathfrak{p}}(V^{\circ}, K^{p^{\circ}})(S)$ is the set of equivalence classes of decuples $(A_0, \lambda_0, \eta_0^p; A, \lambda, \eta^p; A', \lambda', \eta^{p'}; \delta)$ where

- **O** $(A_0, \lambda_0, \eta_0^p; A, \lambda, \eta^p)$ is an element of $M_{\mathfrak{p}}(V^{\circ}, K^{p \circ})(S);$
- O $(A_0, \lambda_0, \eta_0^p; A', \lambda', \eta^{p'})$ is an element of $M'_{\mathfrak{p}}(V^{\circ}, K^{p \circ})(S);$
- $O \delta: A \to A'$ is an O_F -linear quasi-*p*-isogeny (Definition 3.4.5) such that
 - (a) ker $\delta[p^{\infty}]$ is contained in $A[\mathfrak{p}]$;
 - (b) we have $\lambda = \delta^{\vee} \circ \lambda' \circ \delta$; and
 - (c) the K^{*p*°}-orbit of maps $v \mapsto \delta_* \circ \eta^p(v)$ for $v \in V^\circ \otimes_{\mathbb{Q}} \mathbb{A}^{\infty,p}$ coincides with $\eta^{p'}$.

The equivalence relation and the action of morphisms in $\mathfrak{K}(\mathcal{V}^{\circ})^{p} \times \mathfrak{T}$ are defined similarly as in Definition 4.2.3.

By definition, we have the following two obvious forgetful morphisms.



in $\operatorname{Fun}(\mathfrak{K}(V^{\circ})^{p} \times \mathfrak{T}, \operatorname{Sch}_{/\mathbb{F}_{p}^{\Phi}})$. By the extension property of isogeny, it is clear that both μ and μ' are proper. We apply the Stein factorization to the morphism μ' and obtain the following diagram



in $\operatorname{Fun}(\mathfrak{K}(\mathcal{V}^{\circ})^{p} \times \mathfrak{T}, \operatorname{Sch}_{/\mathbb{F}_{p}^{\Phi}})$. For every $\mathcal{K}^{p\circ} \in \mathfrak{K}(\mathcal{V}^{\circ})^{p}$ a perfect field κ containing \mathbb{F}_{p}^{Φ} , we say that a point $(A_{0}, \lambda_{0}, \eta_{0}^{p}; A', \lambda', \eta^{p\prime}) \in \operatorname{M}'_{\mathfrak{p}}(\mathcal{V}^{\circ}, \mathcal{K}^{p\circ})(\kappa)$ is *special* if we have $\operatorname{FH}_{1}^{\mathrm{dR}}(A'/\kappa)_{\tau_{\infty}} = \operatorname{VH}_{1}^{\mathrm{dR}}(A'/\kappa)_{\tau_{\infty}}$. We denote by $\operatorname{M}'_{\mathfrak{p}}(\mathcal{V}^{\circ}, \mathcal{K}^{p\circ})_{\mathrm{sp}}$ the locus of special points in $\operatorname{M}'_{\mathfrak{p}}(\mathcal{V}^{\circ}, \mathcal{K}^{p\circ})$, regarded as a Zariski closed subset, and by $\operatorname{N}'_{\mathfrak{p}}(\mathcal{V}^{\circ}, \mathcal{K}^{p\circ})_{\mathrm{sp}}$ the (set-theoretical) inverse image of $\operatorname{M}'_{\mathfrak{p}}(\mathcal{V}^{\circ}, \mathcal{K}^{p\circ})_{\mathrm{sp}}$ under ν' . An easy deformation argument shows that $\operatorname{M}'_{\mathfrak{p}}(\mathcal{V}^{\circ}, \mathcal{K}^{p\circ})_{\mathrm{sp}}$ is of dimension zero.

Proposition 5.9.4. In (5.20), for every $K^{p\circ} \in \mathfrak{K}(V^{\circ})^{p}$, we have

- (1) The morphism $\mu \colon N_{\mathfrak{p}}(V^{\circ}, K^{p\circ}) \to M_{\mathfrak{p}}(V^{\circ}, K^{p\circ})$ induces a purely inseparable morphism onto its image which is $M_{\mathfrak{p}}^{\mathfrak{p}}(V^{\circ}, K^{p\circ})$.
- (2) The morphism $\nu' \colon N'_{\mathfrak{p}}(V^{\circ}, K^{p \circ}) \to M'_{\mathfrak{p}}(V^{\circ}, K^{p \circ})$ is purely inseparable.
- (3) The morphism $\nu \colon \mathrm{N}_{\mathfrak{p}}^{\bullet}(\mathrm{V}^{\circ},\mathrm{K}^{p\circ}) \to \mathrm{N}_{\mathfrak{p}}^{\prime}(\mathrm{V}^{\circ},\mathrm{K}^{p\circ})$ is the blow-up along $\mathrm{N}_{\mathfrak{p}}^{\prime}(\mathrm{V}^{\circ},\mathrm{K}^{p\circ})_{\mathrm{sp}}$.¹⁰

¹⁰Note that blow-up along a zero-dimensional closed subscheme Z of a regular scheme depends only on the underlying closed subset of Z.

Proof. For (1), it suffices to show that for every algebraically closed field κ containing \mathbb{F}_p^{Φ} , $\mu(\kappa)$ is an isomorphism from $N_{\mathfrak{p}}(V^{\circ}, K^{p\circ})(\kappa)$ to $M_{\mathfrak{p}}^{\bullet}(V^{\circ}, K^{p\circ})(\kappa)$.

We first show that the image of $\mu(\kappa)$ is contained in $M^{\bullet}_{\mathfrak{p}}(V^{\circ}, K^{p\circ})(\kappa)$. Take a point $y = (A_0, \lambda_0, \eta_0^p; A, \lambda, \eta^p; A', \lambda', \eta^{p'}; \delta) \in N_{\mathfrak{p}}(V^{\circ}, K^{p\circ})(\kappa)$. By Lemma 3.4.13(2,4) and the relation $\lambda = \delta^{\vee} \circ \lambda' \circ \delta$, we know that $\delta_{*,\tau} : H_1^{dR}(A/\kappa)_{\tau} \to H_1^{dR}(A'/\kappa)_{\tau}$ is an isomorphism if $\tau \neq \tau^{\circ}_{\infty}$; and ker $\delta_{*,\tau^{\circ}_{\infty}}$ has dimension 1. Moreover, since λ' is *p*-principal, we have ker $\delta_{*,\tau^{\circ}_{\infty}} = H_1^{dR}(A/\kappa)_{\tau^{\circ}_{\infty}}$. By the signature condition again, we also have ker $\delta_{*,\tau^{\circ}_{\infty}} \subseteq \omega_{A^{\vee},\tau^{\circ}_{\infty}}$. Thus, $\mu(y)$ belongs to $M^{\bullet}_{\mathfrak{p}}(V^{\circ}, K^{p\circ})(\kappa)$ by Definition 5.1.3.

We now consider (2) and (3) simultaneously. Let $N_{\mathfrak{p}}(V^{\circ}, K^{p\circ})_{sp}$ be the inverse image of $M'_{\mathfrak{p}}(V^{\circ}, K^{p\circ})_{sp}$ under μ' . By Lemma 5.9.5(1) below, the induced morphism

$$\mu' \colon \mathcal{N}_{\mathfrak{p}}(\mathcal{V}^{\circ},\mathcal{K}^{p \circ}) \setminus \mathcal{N}_{\mathfrak{p}}(\mathcal{V}^{\circ},\mathcal{K}^{p \circ})_{sp} \to \mathcal{M}'_{\mathfrak{p}}(\mathcal{V}^{\circ},\mathcal{K}^{p \circ}) \setminus \mathcal{M}'_{\mathfrak{p}}(\mathcal{V}^{\circ},\mathcal{K}^{p \circ})_{sp}$$

is purely inseparable. Thus, the induced morphism

$$\nu \colon \mathcal{N}_{\mathfrak{p}}(\mathcal{V}^{\circ},\mathcal{K}^{p\circ}) \setminus \mathcal{N}_{\mathfrak{p}}(\mathcal{V}^{\circ},\mathcal{K}^{p\circ})_{sp} \to \mathcal{N}'_{\mathfrak{p}}(\mathcal{V}^{\circ},\mathcal{K}^{p\circ}) \setminus \mathcal{N}'_{\mathfrak{p}}(\mathcal{V}^{\circ},\mathcal{K}^{p\circ})_{sp}$$

is an isomorphism, and the induced morphism

$$\nu' \colon \mathcal{N}'_{\mathfrak{p}}(\mathcal{V}^{\circ},\mathcal{K}^{p\circ}) \setminus \mathcal{N}'_{\mathfrak{p}}(\mathcal{V}^{\circ},\mathcal{K}^{p\circ})_{sp} \to \mathcal{M}'_{\mathfrak{p}}(\mathcal{V}^{\circ},\mathcal{K}^{p\circ}) \setminus \mathcal{M}'_{\mathfrak{p}}(\mathcal{V}^{\circ},\mathcal{K}^{p\circ})_{sp}$$

is purely inseparable. Since $N_{\mathfrak{p}}(V^{\circ}, K^{p\circ})$ is quasi-projective, ν is projective. Thus, ν is a projective birational morphism, which has to be the blow-up along a subset Z of $N'_{\mathfrak{p}}(V^{\circ}, K^{p\circ})_{sp}$ (see, for example, [Liu02, Theorem 8.1.24]). Now we take a point x' of $M'_{\mathfrak{p}}(V^{\circ}, K^{p\circ})_{sp}$ with the residue field κ , which is a finite extension of \mathbb{F}_p^{Φ} . Since ν' is a finite morphism, the inverse image of x' consists of finitely many points y'_1, \ldots, y'_n of $N'_{\mathfrak{p}}(V^{\circ}, K^{p\circ})_{sp}$ with residue fields $\kappa_1, \ldots, \kappa_n$, respectively. By Lemma 5.9.5(2) below, the residue field extension κ_i/κ is trivial for every $1 \leq i \leq n$; and moreover, Z has nonempty intersection with $\{y'_1, \ldots, y'_n\}$. Thus, $\mu'^{-1}(x')$ has cardinality at least $|\kappa| + n$. But we know that $\mu'^{-1}(x')$ has cardinality exactly $|\kappa| + 1$. Therefore, we must have n = 1. We immediately have both (2) and (3).

Lemma 5.9.5. Consider an element $x' \in M'_{\mathfrak{p}}(V^{\circ}, K^{p\circ})(\kappa)$ for some $K^{p\circ} \in \mathfrak{K}(V^{\circ})^{p}$ and a perfect field κ containing \mathbb{F}_{p}^{Φ} . We have

- (1) If the image of x' is not special, then $\mu'^{-1}(x')$ is a singleton.
- (2) If the image of x' is special, then $\mu'^{-1}(x')$ is isomorphic to $\mathbb{P}^{1}(\kappa)$.

Proof. Write $x' = (A_0, \lambda_0, \eta_0^p; A', \lambda', \eta^{p'})$. By the Dieudonné theory and Lemma 3.4.13(2,4), we see that $\mu'^{-1}(x')$ is bijective to Dieudonné submodules $\mathcal{D}_A \subseteq \mathcal{D}(A')$ satisfying $\mathcal{D}_{A,\tau} = \mathcal{D}(A')_{\tau}$ for $\tau \neq \tau_{\infty}^c$, and that $\mathcal{D}(A')_{\tau_{\infty}^c}/\mathcal{D}_{A,\tau_{\infty}^c}$ is a vector space over κ of dimension 1. This amounts to the subspaces of $\mathrm{H}_1^{\mathrm{dR}}(A'/\kappa)_{\tau_{\infty}^c}$ of dimension 2 containing $\mathrm{FH}_1^{\mathrm{dR}}(A'/\kappa)_{\tau_{\infty}} + \mathrm{VH}_1^{\mathrm{dR}}(A'/\kappa)_{\tau_{\infty}}$. Since both $\mathrm{FH}_1^{\mathrm{dR}}(A'/\kappa)_{\tau_{\infty}}$ and $\mathrm{VH}_1^{\mathrm{dR}}(A'/\kappa)_{\tau_{\infty}}$ have dimension 1, the lemma follows by the definition of special points.

Remark 5.9.6. In fact, one can show that μ induces an isomorphism from $N_{\mathfrak{p}}(V^{\circ}, -)$ to $M^{\bullet}_{\mathfrak{p}}(V^{\circ}, -)$; and ν' is purely inseparable of degree p. But we do not need these facts.

5.10. Functoriality under special morphisms. In this subsection, we study the behavior of various moduli schemes under the special morphisms, which is closely related to the Rankin–Selberg motives for $GL_n \times GL_{n+1}$.

We start from the datum $(V_n^{\circ}, \{\Lambda_{n,\mathfrak{q}}^{\circ}\}_{\mathfrak{q}|p})$ as in the beginning of Subsection 5.1, but with V_n° of rank $n \geq 2$. (See Remark 5.10.15 below for the case n = 1.) We then have the induced datum

$$(\mathbf{V}_{n+1}^{\circ},\{\boldsymbol{\Lambda}_{n+1,\mathfrak{q}}^{\circ}\}_{\mathfrak{q}|p})\coloneqq((\mathbf{V}_{n}^{\circ})_{\sharp},\{(\boldsymbol{\Lambda}_{n,\mathfrak{q}}^{\circ})_{\sharp}\}_{\mathfrak{q}|p})$$

of rank n + 1 by Definition 3.1.7. For $N \in \{n, n + 1\}$, we let $K_{N,q}^{\circ}$ be the stabilizer of $\Lambda_{N,q}^{\circ}$, and put $K_{N,p}^{\circ} \coloneqq \prod_{\mathfrak{q}|p} K_{N,\mathfrak{q}}^{\circ}$. Recall the category $\mathfrak{K}(V_{n}^{\circ})_{sp}^{p}$ and functors -, -, + from Definition 3.1.11. To unify notation, we put $-_{n} \coloneqq -_{\flat}$ and $-_{n+1} \coloneqq -_{\sharp}$. Similar to the case of smooth moduli schemes considered in Subsection 4.4, there are five stages of functoriality we will consider.

The first stage concerns Shimura varieties.

Notation 5.10.1. We choose an indefinite uniformization datum $(V'_n, \mathbf{j}_n, \{\Lambda'_{n,\mathfrak{q}}\}_{\mathfrak{q}|p})$ for V°_n as in Definition 5.1.6. Put $V'_{n+1} \coloneqq (V'_n)_{\sharp}$, $\mathbf{j}_{n+1} \coloneqq (\mathbf{j}_n)_{\sharp}$, and $\Lambda'_{n+1,\mathfrak{q}} \coloneqq (\Lambda'_{n,\mathfrak{q}})_{\sharp}$. Then $(V'_{n+1}, \mathbf{j}_{n+1}, \{\Lambda'_{n+1,\mathfrak{q}}\}_{\mathfrak{q}|p})$ is an indefinite uniformization datum for V°_{n+1} . For $N \in \{n, n+1\}$, we let $K'_{N,\mathfrak{q}}$ be the stabilizer of $\Lambda'_{N,\mathfrak{q}}$, and put $K'_{N,p} \coloneqq \prod_{\mathfrak{q}|p} K'_{N,\mathfrak{q}}$.

We obtain a morphism

$$\mathrm{sh}_{\uparrow}': \mathrm{Sh}(\mathrm{V}_n', \mathtt{j}_n - \mathtt{k}_{n,p}') \to \mathrm{Sh}(\mathrm{V}_{n+1}', \mathtt{j}_{n+1} - \mathtt{k}_{n+1,p}')$$

in $\operatorname{Fun}(\mathfrak{K}(\mathcal{V}_n^\circ)_{\operatorname{sp}}^p, \operatorname{Sch}_{/F}).$

For the second stage of functoriality, we have a morphism

(5.21)
$$\mathbf{m}_{\uparrow} \colon \mathbf{M}_{\mathfrak{p}}(\mathbf{V}_{n}^{\circ}, -_{n}) \to \mathbf{M}_{\mathfrak{p}}(\mathbf{V}_{n+1}^{\circ}, -_{n+1})$$

in $\operatorname{Fun}(\mathfrak{K}(\operatorname{V}_{n}^{\circ})_{\operatorname{sp}}^{p} \times \mathfrak{T}, \operatorname{Sch}_{/\mathbb{Z}_{p}^{\Phi}})_{/\mathbf{T}_{\mathfrak{p}}}$ sending an object $(A_{0}, \lambda_{0}, \eta_{0}^{p}; A, \lambda, \eta^{p}) \in \mathbf{M}_{\mathfrak{p}}(\operatorname{V}_{n}^{\circ}, \operatorname{K}_{n}^{p\circ})(S)$ to the object $(A_{0}, \lambda_{0}, \eta_{0}^{p}; A \times A_{0}, \lambda \times \lambda_{0}, \eta^{p} \oplus (\operatorname{id}_{A_{0}})_{*}) \in \mathbf{M}_{\mathfrak{p}}(\operatorname{V}_{n+1}^{\circ}, \operatorname{K}_{n+1}^{p\circ})(S)$. It is clear that \mathbf{m}_{\uparrow} restricts to three morphisms

(5.22)
$$\begin{cases} \mathbf{m}^{\circ}_{\uparrow} \colon \mathbf{M}^{\circ}_{\mathfrak{p}}(\mathbf{V}^{\circ}_{n},-_{n}) \to \mathbf{M}^{\circ}_{\mathfrak{p}}(\mathbf{V}^{\circ}_{n+1},-_{n+1}), \\ \mathbf{m}^{\dagger}_{\uparrow} \colon \mathbf{M}^{\dagger}_{\mathfrak{p}}(\mathbf{V}^{\circ}_{n},-_{n}) \to \mathbf{M}^{\dagger}_{\mathfrak{p}}(\mathbf{V}^{\circ}_{n+1},-_{n+1}), \\ \mathbf{m}^{\bullet}_{\uparrow} \colon \mathbf{M}^{\bullet}_{\mathfrak{p}}(\mathbf{V}^{\circ}_{n},-_{n}) \to \mathbf{M}^{\bullet}_{\mathfrak{p}}(\mathbf{V}^{\circ}_{n+1},-_{n+1}). \end{cases}$$

Moreover, we have the following commutative diagram

in $\mathsf{Fun}(\mathfrak{K}(\mathcal{V}_n^\circ)_{\mathrm{sp}}^p \times \mathfrak{T}, \mathsf{Sch}_{/\mathbb{Q}_p^\Phi})_{/\mathbf{T}_{\mathfrak{p}}^\eta}.$

At the third stage of functoriality, we study the basic correspondence diagram (5.10) for N = n, n+1 under the special morphisms. We will complete a commutative diagram in $\operatorname{Fun}(\mathfrak{K}(\mathbf{V}_n^\circ)_{sp}^p \times \mathfrak{T}, \operatorname{Sch}_{/\mathbb{F}_p^\Phi})_{/\mathrm{T}_p}$ as follows

(5.24)



in which the bottom (resp. top) layer is the basic correspondence diagram (5.10) for $M_{\mathfrak{p}}(V_n^{\circ}, -_n)$ (resp. $M_{\mathfrak{p}}(V_{n+1}^{\circ}, -_{n+1})$).

First, we consider the basic correspondences on the balloon strata, that is, the back layer of the diagram (5.24).

We define $s_{\uparrow}^{\circ}: S_{\mathfrak{p}}^{\circ}(V_{n}^{\circ}, -_{n}) \to S_{\mathfrak{p}}^{\circ}(V_{n+1}^{\circ}, -_{n+1})$ to be the morphism sending an object $(A_0, \lambda_0, \eta_0^p; A^\circ, \lambda^\circ, \eta^{p\circ}) \in \mathcal{S}^\circ_{\mathfrak{n}}(\mathcal{V}^\circ_n, \mathcal{K}^{p\circ}_n)(S)$

to the object

$$(A_0, \lambda_0, \eta_0^p; A^{\circ} \times A_0, \lambda^{\circ} \times \lambda_0, \eta^{p \circ} \oplus (\mathrm{id}_{A_0})_*) \in \mathrm{S}^{\circ}_{\mathfrak{p}}(\mathrm{V}^{\circ}_{n+1}, \mathrm{K}^{p \circ}_{n+1})(S).$$

Remark 5.10.2. The canonical inclusions

$$\mathcal{V}_n^{\circ} \hookrightarrow \mathcal{V}_{n+1}^{\circ}, \quad {\{\Lambda_{n,\mathfrak{q}}^{\circ} \hookrightarrow \Lambda_{n+1,\mathfrak{q}}^{\circ}\}_{\mathfrak{q}|p}}$$

induce a morphism

$$\operatorname{sh}^{\circ}_{\uparrow}: \operatorname{Sh}(\operatorname{V}^{\circ}_{n}, -_{n}\operatorname{K}^{\circ}_{n,p}) \to \operatorname{Sh}(\operatorname{V}^{\circ}_{n+1}, -_{n+1}\operatorname{K}^{\circ}_{n+1,p})$$

in $\mathsf{Fun}(\mathfrak{K}(\mathbf{V}_n^\circ)_{sp}^p,\mathsf{Set})$. It is clear that the following diagram

in $\operatorname{\mathsf{Fun}}(\mathfrak{K}(\operatorname{V}_n^\circ)_{\operatorname{sp}}^p,\operatorname{\mathsf{Set}})_{/\operatorname{T}_{\mathfrak{p}}(\overline{\mathbb{F}}_p)}$ commutes, where υ_{n+1}° and υ_n° are uniformization maps in Construction 5.2.6.

We define $b^{\circ}_{\uparrow}: B^{\circ}_{\mathfrak{p}}(V^{\circ}_n, -_n) \to B^{\circ}_{\mathfrak{p}}(V^{\circ}_{n+1}, -_{n+1})$ to be the morphism sending an object $(A_0, \lambda_0, \eta_0^p; A, \lambda, \eta^p; A^\circ, \lambda^\circ, \eta^{p\circ}; \beta) \in \mathcal{B}^\circ_{\mathfrak{n}}(\mathcal{V}^\circ_n, \mathcal{K}^{p\circ}_n)(S)$

to the object

$$(A_0, \lambda_0, \eta_0^p; A \times A_0, \lambda \times \lambda_0, \eta^p \oplus (\mathrm{id}_{A_0})_*; A^\circ \times A_0, \lambda^\circ \times \lambda_0, \eta^{p\circ} \oplus (\mathrm{id}_{A_0})_*; \beta \times \mathrm{id}_{A_0}) \in \mathrm{B}^\circ_{\mathfrak{p}}(\mathrm{V}_{n+1}^\circ, \mathrm{K}_{n+1}^{p\circ})(S).$$

Second, we consider the basic correspondences on the ground strata, that is, the front layer of the diagram (5.24).

Definition 5.10.3. We define a functor

$$\begin{split} \mathbf{S}^{\bullet}_{\mathfrak{p}}(\mathbf{V}^{\circ}_{n},-)_{\mathrm{sp}} \colon \mathfrak{K}(\mathbf{V}^{\circ}_{n})^{p}_{\mathrm{sp}} \times \mathfrak{T} &\to \mathsf{PSch}'_{/\mathbb{F}^{\Phi}_{p}} \\ \mathbf{K}^{p\circ} &\mapsto \mathbf{S}^{\bullet}_{\mathfrak{p}}(\mathbf{V}^{\circ}_{n},\mathbf{K}^{p\circ})_{\mathrm{sp}} \end{split}$$

such that for every $S \in \mathsf{Sch}'_{\mathbb{F}^{\Phi}_n}$, $S^{\bullet}_{\mathfrak{p}}(V^{\circ}_n, K^{p\circ})_{\mathrm{sp}}(S)$ is the set of equivalence classes of decuples $(A_0, \lambda_0, \eta_0^p; A^{\bullet}, \lambda^{\bullet}, \eta^{p \bullet}; A^{\bullet}_{h}, \lambda^{\bullet}_{h}, \eta^{p \bullet}_{h}; \delta^{\bullet})$ where

- $O (A_0, \lambda_0, \eta_0^p; A^{\bullet}, \lambda^{\bullet}, \eta^{p \bullet})$ is an element in $S^{\bullet}_{\mathfrak{p}}(V_n^{\circ}, K_n^{p \circ})(S);$
- O $(A_0, \lambda_0, \eta_0^p; A_{\natural}^{\bullet}, \lambda_{\natural}^{\bullet}, \eta_{\natural}^{p\bullet})$ is an element in S[•]_p(V^o_{n+1}, K^{po}_{n+1})(S); O $\delta^{\bullet}: A^{\bullet} \times A_0 \to A_{\natural}^{\bullet}$ is an O_F-linear quasi-p-isogeny (Definition 3.4.5) such that
 - (a) ker $\delta^{\bullet}[p^{\infty}]$ is contained in $(A^{\bullet} \times A_0)[\mathfrak{p}]$;
 - (b) we have $\lambda^{\bullet} \times \varpi \lambda_0 = \delta^{\bullet \vee} \circ \lambda_{\flat}^{\bullet} \circ \delta^{\bullet}$; and
 - (c) the K_{n+1}^p -orbit of maps $v \mapsto \delta_*^{\bullet} \circ (\eta^{p \bullet} \oplus (\mathrm{id}_{A_0})_*)(v)$ for $v \in \mathrm{V}_{\sharp}^{\circ} \otimes_{\mathbb{Q}} \mathbb{A}^{\infty,p}$ coincides with $\eta^{p\bullet}_{h}$.

The equivalence relation and the action of morphisms in $\mathfrak{K}(\mathcal{V}_n^\circ)^p_{sp} \times \mathfrak{T}$ are defined similarly as in Definition 4.2.3.

We have apparently the forgetful morphism

$$\mathrm{S}^{\bullet}_{\mathfrak{p}}(\mathrm{V}^{\circ}_{n},-)_{\mathrm{sp}} \to \mathrm{T}_{\mathfrak{p}}$$

in $\operatorname{Fun}(\mathfrak{K}(\operatorname{V}_n^{\circ})_{\operatorname{sp}}^p \times \mathfrak{T}, \operatorname{PSch}'_{/\mathbb{F}_p^{\Phi}})$ which is represented by finite and étale schemes. By definition, we have the two forgetful morphisms

$$\mathbf{s}_{\downarrow}^{\bullet} \colon \mathbf{S}_{\mathfrak{p}}^{\bullet}(\mathbf{V}_{n}^{\circ}, -)_{\mathrm{sp}} \to \mathbf{S}_{\mathfrak{p}}^{\bullet}(\mathbf{V}_{n}^{\circ}, -_{n}), \quad \mathbf{s}_{\uparrow}^{\bullet} \colon \mathbf{S}_{\mathfrak{p}}^{\bullet}(\mathbf{V}_{n}^{\circ}, -)_{\mathrm{sp}} \to \mathbf{S}_{\mathfrak{p}}^{\bullet}(\mathbf{V}_{n+1}^{\circ}, -_{n+1})$$

in $\mathsf{Fun}(\mathfrak{K}(\mathcal{V}_n^\circ)_{\mathrm{sp}}^p \times \mathfrak{T}, \mathsf{Sch}_{/\mathbb{F}_p^\Phi})_{/\mathrm{T}_{\mathfrak{p}}}.$

Lemma 5.10.4. We have the following properties concerning s_{\perp}^{\bullet} .

(1) When n is even, s_{\perp}^{\bullet} is an isomorphism, and the morphism

$$\mathbf{s}^{\bullet}_{\uparrow} \circ \mathbf{s}^{\bullet-1}_{\downarrow} \colon \mathrm{S}^{\bullet}_{\mathfrak{p}}(\mathrm{V}^{\circ}_{n},-_{n}) \to \mathrm{S}^{\bullet}_{\mathfrak{p}}(\mathrm{V}^{\circ}_{n+1},-_{n+1})$$

is given by the assignment

$$(A_0, \lambda_0, \eta_0^p; A^{\bullet}, \lambda^{\bullet}, \eta^{p \bullet}) \mapsto (A_0, \lambda_0, \eta_0^p; A^{\bullet} \times A_0, \lambda^{\bullet} \times \varpi \lambda_0, \eta^{p \bullet} \times (\mathrm{id}_{A_0})_*)$$

(2) When n is odd, s^{\bullet}_{\downarrow} is finite étale of degree p + 1.

Proof. The proof is very similar to Lemma 4.4.2, which we leave to readers.

Definition 5.10.5. We define $B^{\bullet}_{\mathfrak{p}}(V^{\circ}_n, -)_{sp}$ to be the fiber product indicated in the following Cartesian diagram

in $\operatorname{\mathsf{Fun}}(\mathfrak{K}(\operatorname{V}_n^\circ)_{\operatorname{sp}}^p \times \mathfrak{T}, \operatorname{\mathsf{Sch}}_{/\mathbb{F}_p^\Phi})_{/\mathrm{T}_p}$. We define $\mathrm{b}_{\uparrow}^{\bullet} \colon \mathrm{B}_{\mathfrak{p}}^{\bullet}(\operatorname{V}_n^\circ, -)_{\operatorname{sp}} \to \mathrm{B}_{\mathfrak{p}}^{\bullet}(\operatorname{V}_{n+1}^\circ, -_{n+1})$ to be the morphism sending an object

$$((A_0, \lambda_0, \eta_0^p; A, \lambda, \eta^p; A^{\bullet}, \lambda^{\bullet}, \eta^{p\bullet}; \gamma), (A_0, \lambda_0, \eta_0^p; A^{\bullet}, \lambda^{\bullet}, \eta^{p\bullet}; A^{\bullet}_{\natural}, \lambda^{\bullet}_{\natural}, \eta^{p\bullet}_{\natural}; \delta^{\bullet})) \in \mathcal{B}^{\bullet}_{\mathfrak{p}}(\mathcal{V}_n^{\circ}, \mathcal{K}^{p\circ})_{\mathrm{sp}}(S)$$

to $(A_0, \lambda_0, \eta_0^p; A \times A_0, \lambda \times \lambda_0, \eta^p \oplus (\mathrm{id}_{A_0})_*; A_{\natural}^{\bullet}, \lambda_{\natural}^{\bullet}, \eta_{\natural}^{p \bullet}; \delta^{\bullet} \circ (\gamma \times \mathrm{id}_{A_0}))$, which is an object of $\mathrm{B}^{\bullet}_{\mathfrak{p}}(\mathrm{V}^{\circ}_{n+1}, \mathrm{K}^{p \circ}_{n+1})(S)$ by a similar argument of Lemma 4.4.4.

We have the following result.

Proposition 5.10.6. When n is odd, the square

$$\begin{split} \mathbf{B}_{\mathfrak{p}}^{\bullet}(\mathbf{V}_{n+1}^{\circ},-_{n+1}) & \xrightarrow{\iota_{n+1}^{\bullet}} \mathbf{M}_{\mathfrak{p}}^{\bullet}(\mathbf{V}_{n+1}^{\circ},-_{n+1}) \\ & \stackrel{\mathbf{b}_{\uparrow}^{\bullet} \uparrow}{\overset{\uparrow}{\underset{\mathbf{B}_{\mathfrak{p}}^{\bullet}(\mathbf{V}_{n}^{\circ},-_{)_{\mathrm{Sp}}}{\overset{\iota_{n}^{\bullet} \circ \mathbf{b}_{\downarrow}^{\bullet}}{\overset{\bullet}{\underset{\mathbf{M}_{p}^{\bullet}(\mathbf{V}_{n}^{\circ},-_{n})}}} \mathbf{M}_{\mathfrak{p}}^{\bullet}(\mathbf{V}_{n}^{\circ},-_{n}) \end{split}$$

extracted from the diagram (5.24) is Cartesian.

Proof. The proof is very similar to Proposition 4.4.5, which we leave to readers.

Third, we consider the basic correspondences on the link strata, that is, the middle (vertical) layer of the diagram (5.24).

Definition 5.10.7. We define $S_{\mathfrak{p}}^{\dagger}(V_n^{\circ}, -)_{sp}$ to be the fiber product indicated in the following Cartesian diagram



in $\operatorname{Fun}(\mathfrak{K}(\operatorname{V}_{n}^{\circ})_{\operatorname{sp}}^{p} \times \mathfrak{T}, \operatorname{Sch}_{/\mathbb{F}_{p}^{\Phi}})_{/\mathrm{T}_{p}}$. By Lemma 5.10.4, we know that $\operatorname{s}_{\downarrow}^{\dagger}$ is an isomorphism (resp. finite étale of degree p + 1) when n is even (resp. odd). We define $\operatorname{s}_{\uparrow}^{\dagger} \colon \operatorname{S}_{\mathfrak{p}}^{\dagger}(\operatorname{V}_{n}^{\circ}, -)_{\operatorname{sp}} \to \operatorname{S}_{\mathfrak{p}}^{\dagger}(\operatorname{V}_{n+1}^{\circ}, -_{n+1})$ to be the morphism sending an object

$$((A_0,\lambda_0,\eta_0^p;A^\circ,\lambda^\circ,\eta^{p\circ};A^\bullet,\lambda^\bullet,\eta^{p\bullet};\psi),(A_0,\lambda_0,\eta_0^p;A^\bullet,\lambda^\bullet,\eta^{p\bullet};A^\bullet_{\natural},\lambda^\bullet_{\natural},\eta^{p\bullet}_{\natural};\delta^\bullet)) \in \mathcal{S}^{\dagger}_{\mathfrak{p}}(\mathcal{V}^\circ_n,\mathcal{K}^{p\circ})_{\mathrm{sp}}(S)$$

to the object

$$(A_0, \lambda_0, \eta_0^p; A^{\circ} \times A_0, \lambda^{\circ} \times \lambda_0, \eta^{p^{\circ}} \oplus (\mathrm{id}_{A_0})_*; A_{\natural}^{\bullet}, \lambda_{\natural}^{\bullet}, \eta_{\natural}^{p^{\bullet}}; \delta^{\bullet} \circ (\psi \times \mathrm{id}_{A_0})) \in \mathrm{S}_{\mathfrak{p}}^{\dagger}(\mathrm{V}_{n+1}^{\circ}, \mathrm{K}_{n+1}^{p^{\circ}})(S).$$

Lemma 5.10.8. We have

(1) When n is even, the square

$$\begin{split} \mathbf{S}_{\mathfrak{p}}^{\dagger}(\mathbf{V}_{n+1}^{\circ},-_{n+1}) & \xrightarrow{\mathbf{s}_{n+1}^{\dagger}} \mathbf{S}_{\mathfrak{p}}^{\bullet}(\mathbf{V}_{n+1}^{\circ},-_{n+1}) \\ & \uparrow \\ \mathbf{s}_{\uparrow}^{\dagger} & & \uparrow \\ \mathbf{S}_{\mathfrak{p}}^{\dagger}(\mathbf{V}_{n}^{\circ},-)_{\mathrm{sp}} & \xrightarrow{\mathbf{s}_{\mathrm{sp}}^{\dagger\bullet}} \mathbf{S}_{\mathfrak{p}}^{\bullet}(\mathbf{V}_{n}^{\circ},-)_{\mathrm{sp}} \end{split}$$

extracted from (5.24) is a Cartesian diagram.

(2) When n is odd, the square

$$\begin{split} \mathbf{S}_{\mathfrak{p}}^{\circ}(\mathbf{V}_{n+1}^{\circ},-_{n+1}) & \stackrel{\mathbf{s}_{n+1}^{\dagger\circ}}{\longleftrightarrow} \mathbf{S}_{\mathfrak{p}}^{\dagger}(\mathbf{V}_{n+1}^{\circ},-_{n+1}) \\ & \uparrow \\ \mathbf{s}_{\uparrow}^{\circ} & \uparrow \\ \mathbf{S}_{\mathfrak{p}}^{\circ}(\mathbf{V}_{n}^{\circ},-_{n}) & \stackrel{\mathbf{s}_{n}^{\dagger\circ}\circ\mathbf{s}_{\downarrow}^{\dagger}}{\longleftarrow} \mathbf{S}_{\mathfrak{p}}^{\dagger}(\mathbf{V}_{n}^{\circ},-)_{\mathrm{sp}} \end{split}$$

extracted from (5.24) is a Cartesian diagram.

Proof. Let $S^{\ddagger}_{\mathfrak{p}}(V_n^{\circ}, -)_{sp}$ be the actual fiber product in both cases. Take an object $K^{p\circ} \in \mathfrak{K}(V_n^{\circ})_{sp}^p$. We have to show that the natural morphism $s^{\ddagger} \colon S^{\dagger}_{\mathfrak{p}}(V_n^{\circ}, K^{p\circ})_{sp} \to S^{\ddagger}_{\mathfrak{p}}(V_n^{\circ}, K^{p\circ})_{sp}$ is an isomorphism. Since s^{\ddagger} is a morphism of étale schemes over \mathbb{F}_p^{Φ} , it suffices to show that $s^{\ddagger}(\kappa)$ is an isomorphism for every perfect field κ containing \mathbb{F}_p^{Φ} .

For (1), by Lemma 5.10.4(1), an object in $S^{\ddagger}_{\mathfrak{p}}(V_n^{\circ}, K^{p\circ})_{sp}(S)$ is given by a pair of objects:

$$(A_0, \lambda_0, \eta_0^p; A^{\bullet}, \lambda^{\bullet}, \eta^{p\bullet}; A^{\bullet} \times A_0, \lambda^{\bullet} \times \varpi \lambda_0, \eta^{p\bullet} \times (\mathrm{id}_{A_0})_*) \in \mathrm{S}^{\bullet}_{\mathfrak{p}}(\mathrm{V}_n^{\circ}, \mathrm{K}^{p\circ})_{\mathrm{sp}}(\kappa),$$
$$(A_0, \lambda_0, \eta_0^p; A^{\circ}_{\natural}, \lambda^{\circ}_{\natural}, \eta^{p\circ}_{\natural}; A^{\bullet} \times A_0, \lambda^{\bullet} \times \varpi \lambda_0, \eta^{p\bullet} \times (\mathrm{id}_{A_0})_*; \psi_{\natural}) \in \mathrm{S}^{\dagger}_{\mathfrak{p}}(\mathrm{V}_{n+1}^{\circ}, \mathrm{K}_{n+1}^{p\circ})(\kappa).$$

Let A° be the cokernel of the kernel of the composite map $A^{\circ}_{\natural} \xrightarrow{\psi_{\natural}} A^{\bullet} \times A_{0} \to A^{\bullet}$, and $\psi \colon A^{\circ} \to A^{\bullet}$ the induced map. Let λ° be the unique quasi-polarization of A° satisfying $\varpi \cdot \lambda^{\circ} = \psi^{\vee} \circ \lambda^{\bullet} \circ \psi$. Since $\lambda^{\circ}_{\natural}$ is *p*-principal and we have $\varpi \cdot \lambda^{\circ}_{\natural} = \psi^{\vee}_{\natural} \circ (\lambda^{\bullet} \times \varpi \cdot \lambda_{0}) \circ \psi_{\natural}$, the composite map $A^{\circ}_{\natural} \xrightarrow{\psi_{\natural}} A^{\bullet} \times A_{0} \to A_{0}$ splits. Thus, the natural map $A^{\circ}_{\natural} \to A^{\circ} \times A_0$ is an isomorphism. Then λ° is *p*-principal, and we obtain an object

$$(A_0, \lambda_0, \eta_0^p; A^{\circ}, \lambda^{\circ}, \eta^{p \circ}; A^{\bullet}, \lambda^{\bullet}, \eta^{p \bullet}; \psi) \in \mathcal{S}^{\dagger}_{\mathfrak{p}}(\mathcal{V}_n^{\circ}, \mathcal{K}_n^{p \circ})(\kappa) = \mathcal{S}^{\dagger}_{\mathfrak{p}}(\mathcal{V}_n^{\circ}, \mathcal{K}^{p \circ})_{\mathrm{sp}}(\kappa),$$

where $\eta^{p\circ}$ is chosen so that Definition 5.4.1(c) is satisfied. In other words, we obtain a morphism from $S^{\ddagger}_{\mathfrak{p}}(V_n^{\circ}, K^{p\circ})_{sp}(\kappa)$ to $S^{\dagger}_{\mathfrak{p}}(V_n^{\circ}, K^{p\circ})_{sp}(\kappa)$. It is straightforward to check that it is an inverse to the morphism $s^{\ddagger}(\kappa)$.

For (2), an object in $S^{\ddagger}_{\mathfrak{p}}(V_n^{\circ}, K^{p\circ})_{sp}(\kappa)$ is given by a pair of objects:

$$(A_0, \lambda_0, \eta_0^p; A^{\circ}, \lambda^{\circ}, \eta^{p \circ}) \in \mathcal{S}^{\circ}_{\mathfrak{p}}(\mathcal{V}^{\circ}_n, \mathcal{K}^{p \circ}_n)(\kappa),$$
$$(A_0, \lambda_0, \eta_0^p; A^{\circ} \times A_0, \lambda^{\circ} \times \lambda_0, \eta^{p \circ} \times (\mathrm{id}_{A_0})_*; A^{\bullet}_{\natural}, \lambda^{\bullet}_{\natural}, \eta^{p \bullet}_{\natural}; \psi_{\natural}) \in \mathcal{S}^{\dagger}_{\mathfrak{p}}(\mathcal{V}^{\circ}_{n+1}, \mathcal{K}^{p \circ}_{n+1})(\kappa).$$

Let $A^{\bullet\vee}$ be the cokernel of the kernel of the composite map $A_{\natural}^{\bullet\vee} \xrightarrow{\psi_{\natural}^{\vee}} A^{\circ\vee} \times A_{0}^{\vee} \to A^{\circ\vee}$, and $\psi^{\vee} \colon A^{\circ\vee} \to A^{\bullet\vee}$ the induced map. Taking dual, we obtain a map $\psi \colon A^{\circ} \to A^{\bullet}$ and an induced map $\delta^{\bullet} \colon A^{\bullet} \times A_{0} \to A_{\natural}^{\bullet}$. Let λ^{\bullet} be the unique quasi-polarization of A^{\bullet} satisfying $\varpi \cdot \lambda^{\circ} = \psi^{\vee} \circ \lambda^{\bullet} \circ \psi$. Since $\lambda_{\natural}^{\bullet}$ is *p*-principal and we have $\lambda^{\bullet} \times \varpi \cdot \lambda_{0} = \delta^{\bullet\vee} \circ \lambda_{\natural}^{\bullet} \circ \delta^{\bullet}$, we know that ker $\lambda^{\bullet}[p^{\infty}]$ is contained in $A^{\bullet}[\mathfrak{p}]$ of rank p^{2} , and we obtain an object

$$\left((A_0,\lambda_0,\eta_0^p;A^\circ,\lambda^\circ,\eta^{p\circ};A^\bullet,\lambda^\bullet,\eta^{p\bullet};\psi),(A_0,\lambda_0,\eta_0^p;A^\bullet,\lambda^\bullet,\eta^{p\bullet};A^\bullet_{\natural},\lambda^\bullet_{\natural},\eta^{p\bullet}_{\natural};\delta^\bullet)\right)\in\mathcal{S}^{\dagger}_{\mathfrak{p}}(\mathcal{V}_n^\circ,\mathcal{K}^{p\circ})_{\mathrm{sp}}(\kappa),$$

where $\eta^{p\bullet}$ is chosen so that Definition 5.4.1(c) is satisfied. In other words, we obtain a morphism from $S^{\ddagger}_{\mathfrak{p}}(V_n^{\circ}, K^{p\circ})_{sp}(\kappa)$ to $S^{\dagger}_{\mathfrak{p}}(V_n^{\circ}, K^{p\circ})_{sp}(\kappa)$. It is straightforward to check that it is an inverse to the morphism $s^{\ddagger}(\kappa)$.

Definition 5.10.9. We define $B_{\mathfrak{p}}^{\dagger}(V_n^{\circ}, -)_{sp}$ to be the fiber product indicated in the following Cartesian diagram

$$\begin{array}{c|c} \mathbf{B}_{\mathfrak{p}}^{\dagger}(\mathbf{V}_{n}^{\circ},-)_{\mathrm{sp}} & \xrightarrow{\pi_{\mathrm{sp}}^{\dagger}} \mathbf{S}_{\mathfrak{p}}^{\dagger}(\mathbf{V}_{n}^{\circ},-)_{\mathrm{sp}} \\ & & \downarrow_{s\downarrow}^{\dagger} \\ & & \downarrow_{s\downarrow}^{\dagger} \\ \mathbf{B}_{\mathfrak{p}}^{\dagger}(\mathbf{V}_{n}^{\circ},-_{n}) & \xrightarrow{\pi_{n}^{\dagger}} \mathbf{S}_{\mathfrak{p}}^{\dagger}(\mathbf{V}_{n}^{\circ},-_{n}) \end{array}$$

in $\mathsf{Fun}(\mathfrak{K}(\mathcal{V}_n^\circ)_{\mathrm{sp}}^p \times \mathfrak{T}, \mathsf{Sch}_{/\mathbb{F}_p^\Phi})_{/\mathrm{T}_{\mathfrak{p}}}.$

By the universal property of Cartesian diagrams, we obtain a unique morphism

$$\mathbf{b}_{\mathrm{sp}}^{\dagger\bullet} \colon \mathbf{B}_{\mathfrak{p}}^{\dagger}(\mathbf{V}_{n}^{\circ}, -)_{\mathrm{sp}} \to \mathbf{B}_{\mathfrak{p}}^{\bullet}(\mathbf{V}_{n}^{\circ}, -)_{\mathrm{sp}}$$

rendering the front lower-left cube of (5.24) commute. Finally, an easy diagram chasing indicates that we have a unique morphism

$$\mathbf{b}^{\dagger}_{\uparrow} \colon \mathbf{B}^{\dagger}_{\mathfrak{p}}(\mathbf{V}^{\circ}_{n}, \mathbf{-})_{\mathrm{sp}} \to \mathbf{B}^{\dagger}_{\mathfrak{p}}(\mathbf{V}^{\circ}_{n+1}, \mathbf{-}_{n+1})$$

rendering the entire diagram (5.24) commute. Thus, we obtain our desired diagram (5.24).

Remark 5.10.10. By Proposition 5.10.6 and Theorem 5.4.3(1), one can show that when n is odd, the square

$$\begin{array}{ccc}
\mathbf{B}_{\mathfrak{p}}^{\dagger}(\mathbf{V}_{n+1}^{\circ},-n+1) & \xrightarrow{\iota_{n+1}^{\dagger}} & \mathbf{M}_{\mathfrak{p}}^{\dagger}(\mathbf{V}_{n+1}^{\circ},-n+1) \\ & & \mathbf{b}_{\uparrow}^{\dagger} & & & \mathbf{M}_{\mathfrak{p}}^{\dagger} \\ & & \mathbf{B}_{\mathfrak{p}}^{\dagger}(\mathbf{V}_{n}^{\circ},-)_{\mathrm{sp}} & \xrightarrow{\iota_{n}^{\dagger} \circ \mathbf{b}_{\downarrow}^{\dagger}} & & \mathbf{M}_{\mathfrak{p}}^{\dagger}(\mathbf{V}_{n}^{\circ},-n) \\ \end{array}$$

extracted from the diagram (5.24) is Cartesian.

Remark 5.10.11. By Lemma 5.10.4(1), Definition 5.10.5, Definition 5.10.7, and Definition 5.10.9, the four downward arrows in the diagram (5.24) are isomorphisms when n is even.

At the fourth stage of functoriality, we compare the special morphisms for basic correspondences and for Deligne–Lusztig varieties. Take a point $s^{\dagger} \in S^{\dagger}_{\mathfrak{p}}(V_n^{\circ}, K^{p\circ})_{sp}(\kappa)$ for a perfect field κ containing \mathbb{F}_p^{Φ} . Put

$$\begin{split} s_n^{\dagger} &\coloneqq \mathbf{s}_{\downarrow}^{\dagger}(s^{\dagger}), \quad s_{n+1}^{\dagger} \coloneqq \mathbf{s}_{\uparrow}^{\dagger}(s^{\dagger}); \\ s_n^{\circ} &\coloneqq \mathbf{s}_n^{\dagger \circ}(s_n^{\dagger}), \quad s_{n+1}^{\circ} \coloneqq \mathbf{s}_{n+1}^{\dagger \circ}(s_{n+1}^{\dagger}); \\ s^{\bullet} &\coloneqq \mathbf{s}_{\mathrm{sp}}^{\dagger \bullet}(s^{\dagger}), \quad s_n^{\bullet} \coloneqq \mathbf{s}_n^{\dagger \bullet}(s_n^{\dagger}), \quad s_{n+1}^{\bullet} \coloneqq \mathbf{s}_{n+1}^{\dagger \bullet}(s_{n+1}^{\dagger}). \end{split}$$

Denote by $B_{s^{\dagger}}^{\dagger}$, $B_{s_{n}^{\dagger}}^{\dagger}$, $B_{s_{n+1}^{\dagger}}^{\dagger}$, $B_{s_{n}^{\circ}}^{\circ}$, $B_{s_{n+1}^{\circ}}^{\circ}$, $B_{s^{\bullet}}^{\bullet}$, $B_{s^{\bullet}}^{\bullet}$, and $B_{s_{n+1}^{\bullet}}^{\bullet}$ their preimages under π_{sp}^{\dagger} , π_{n}^{\dagger} , π_{n+1}^{\dagger} , π_{n}° , π_{n+1}° , π_{sp}° , π_{n}^{\bullet} , and π_{n+1}^{\bullet} , respectively.

Proposition 5.10.12. Let the notation be as above. The following diagram



in Sch_{κ} commutes, where

- O $\zeta_{s_n^{\circ}}^{\circ}$ and $\zeta_{s_{n+1}^{\circ}}^{\circ}$ are the isomorphisms in Theorem 5.2.4;
- $O \zeta_{s_n}^{\bullet}$ and $\zeta_{s_{n+1}}^{\bullet}$ are the isomorphisms in Theorem 5.3.4(3);
- O $\zeta_{s_n^{\dagger}}^{\dagger}$ and $\zeta_{s_{n+1}^{\dagger}}^{\dagger}$ are the isomorphisms in Theorem 5.4.3(2);
- $\begin{array}{l} O \ \mathbb{P}(\mathscr{V}_{s_{n}^{\dagger}}) \xrightarrow{s_{n+1}} \\ O \ \mathbb{P}(\mathscr{V}_{s_{n}^{\dagger}}) \xrightarrow{\mathcal{P}}(\mathscr{V}_{s_{n}^{\circ}}) \ and \ \mathbb{P}(\mathscr{V}_{s_{n+1}^{\dagger}}) \xrightarrow{\mathcal{P}}(\mathscr{V}_{s_{n+1}^{\circ}}) \ are \ closed \ embeddings \ in \ Remark \ 5.4.4(1); \\ O \ \mathbb{P}(\mathscr{V}_{s_{n}^{\dagger}}) \xrightarrow{\mathcal{P}}(\mathbb{V}_{s_{n}^{\bullet}}) \ = \ \mathrm{DL}^{\bullet}(\mathscr{V}_{s_{n}^{\bullet}}, \{ \ , \ \}_{s_{n}^{\bullet}}) \ and \ \mathbb{P}(\mathscr{V}_{s_{n+1}^{\dagger}}) \xrightarrow{\mathcal{P}}(\mathbb{D}_{s_{n+1}^{\bullet}}) \ = \ \mathrm{DL}^{\bullet}(\mathscr{V}_{s_{n+1}^{\bullet}}, \{ \ , \ \}_{s_{n+1}^{\bullet}}) \ are \ \mathcal{P}(\mathbb{V}_{s_{n+1}^{\dagger}}) \xrightarrow{\mathcal{P}}(\mathbb{V}_{s_{n+1}^{\bullet}}) \ \mathcal{P}(\mathbb{V}_{s_{n+1}^{\bullet}}) \ \mathcal{P}(\mathbb{V}_{s_{n+1}^{\bullet}}) \ \mathcal{P}(\mathbb{V}_{s_{n+1}^{\bullet}}) \ ds \ \mathcal{P}(\mathbb{$ closed embeddings in Remark 5.4.4(2);
- $O \mathbb{P}(\mathscr{V}_{s_n^{\circ}}) \to \mathbb{P}(\mathscr{V}_{s_{n+1}^{\circ}})$ is the morphism induced by the obvious κ -linear (surjective) map $\mathscr{V}_{s_{n+1}^{\circ}} \to \mathscr{V}_{s_{n}^{\circ}};$
- $O \ \delta_{s^{\bullet\uparrow}}$ is the morphism in Construction A.2.3 with respect to the map $\delta_{s^{\bullet}}: \mathscr{V}_{s^{\bullet}_{n,\sharp}} \to \mathscr{V}_{s^{\bullet}_{n+1}}$ induced by $\delta^{\bullet} \colon A^{\bullet} \times A_0 \to A^{\bullet}_{\natural}$; and
- $O \mathbb{P}(\mathscr{V}_{s_n^{\dagger}}) \to \mathbb{P}(\mathscr{V}_{s_{n+1}^{\dagger}})$ is the restriction of $\delta_{s^{\bullet\uparrow}}$, in view of Remark 5.4.4(2).

In particular, $\mathbf{b}^{\bullet}_{\uparrow} \colon \mathbf{B}^{\bullet}_{s^{\bullet}} \to \mathbf{B}^{\bullet}_{s^{\bullet}_{n+1}}$ is an isomorphism when n is even.

Proof. The proof is very similar to Proposition 4.4.6, which we leave to readers. The last assertion follows as $b^{\bullet}_{\downarrow} \colon B^{\bullet}_{s_{\bullet}} \to B^{\bullet}_{s_{n}}$ is always an isomorphism, and $\delta_{s^{\bullet\uparrow}}$ is an isomorphism when *n* is even. \Box

At the final stage of functoriality, we relate the special morphisms for sources of basic correspondences to Shimura sets under the uniformization maps v° (5.4), v^{\bullet} (5.9), and v^{\dagger} (5.11). Recall that we have data $(\mathcal{V}_n^\circ, \{\Lambda_{n,\mathfrak{g}}^\circ\}_{\mathfrak{q}|p})$ and $(\mathcal{V}_{n+1}^\circ, \{\Lambda_{n+1,\mathfrak{g}}^\circ\}_{\mathfrak{q}|p})$.

Notation 5.10.13. As in Construction 5.3.6, we choose a lattice chain $\Lambda_{n,\mathfrak{p}}^{\circ} \subseteq \Lambda_{n,\mathfrak{p}}^{\bullet} \subseteq p^{-1} \Lambda_{n,\mathfrak{p}}^{\circ}$ of $V_n^{\circ} \otimes_F F_{\mathfrak{p}}$, and a lattice chain $\Lambda_{n+1,\mathfrak{p}}^{\circ} \subseteq \Lambda_{n+1,\mathfrak{p}}^{\bullet} \subseteq p^{-1} \Lambda_{n+1,\mathfrak{p}}^{\circ}$ of $V_{n+1}^{\circ} \otimes_F F_{\mathfrak{p}}$, for which we assume that $(\Lambda_{n,\mathfrak{p}}^{\bullet})_{\sharp} \subseteq \Lambda_{n+1,\mathfrak{p}}^{\bullet} \subseteq p^{-1} (\Lambda_{n,\mathfrak{p}}^{\bullet})_{\sharp}^{\vee}$ holds. We now introduce various open compact subgroups at p.

- **O** For $N \in \{n, n+1\}$, we have $K_{N,p}^{\circ}$ from Construction 5.2.6, $K_{N,p}^{\bullet}$ from Construction 5.3.6, and $K_{N,p}^{\dagger} = K_{N,p}^{\circ} \cap K_{N,p}^{\bullet}$ from Construction 5.4.5. O Put $K_{sp,\mathfrak{p}}^{\bullet} \coloneqq K_{n,\mathfrak{p}}^{\bullet} \cap K_{n+1,\mathfrak{p}}^{\bullet}$ (as a subgroup of $K_{n,\mathfrak{p}}^{\bullet}$) and $K_{sp,p}^{\bullet} \coloneqq K_{sp,\mathfrak{p}}^{\bullet} \times \prod_{\mathfrak{q}|p,\mathfrak{q}\neq\mathfrak{p}} K_{n,\mathfrak{q}}^{\circ}$. O Put $K_{sp,p}^{\dagger} \coloneqq K_{sp,p}^{\bullet} \cap K_{n,p}^{\circ}$.

For later use, we also introduce natural maps

$$\begin{cases} \operatorname{sh}_{\uparrow}^{\circ} \colon \operatorname{Sh}(\operatorname{V}_{n}^{\circ}, -_{n}\operatorname{K}_{n,p}^{\circ}) \to \operatorname{Sh}(\operatorname{V}_{n+1}^{\circ}, -_{n+1}\operatorname{K}_{n+1,p}^{\circ}), \\ \operatorname{sh}_{\uparrow}^{\circ} \colon \operatorname{Sh}(\operatorname{V}_{n}^{\circ}, -_{n}\operatorname{K}_{n,p}^{\bullet}) \to \operatorname{Sh}(\operatorname{V}_{n}^{\circ}, -_{n}\operatorname{K}_{\operatorname{sp},p}^{\bullet}), \\ \operatorname{sh}_{\downarrow}^{\circ} \colon \operatorname{Sh}(\operatorname{V}_{n}^{\circ}, -_{n}\operatorname{K}_{\operatorname{sp},p}^{\dagger}) \to \operatorname{Sh}(\operatorname{V}_{n+1}^{\circ}, -_{n+1}\operatorname{K}_{n+1,p}^{\bullet}), \\ \operatorname{sh}_{\uparrow}^{\dagger} \colon \operatorname{Sh}(\operatorname{V}_{n}^{\circ}, -_{n}\operatorname{K}_{n,p}^{\dagger}) \to \operatorname{Sh}(\operatorname{V}_{n}^{\circ}, -_{n}\operatorname{K}_{\operatorname{sp},p}^{\dagger}), \\ \operatorname{sh}_{\downarrow}^{\dagger} \colon \operatorname{Sh}(\operatorname{V}_{n}^{\circ}, -_{n}\operatorname{K}_{n,p}^{\dagger}) \to \operatorname{Sh}(\operatorname{V}_{n+1}^{\circ}, -_{n+1}\operatorname{K}_{n+1,p}^{\dagger}), \\ \operatorname{sh}_{n}^{\dagger\circ} \colon \operatorname{Sh}(\operatorname{V}_{n}^{\circ}, -_{n}\operatorname{K}_{n,p}^{\dagger}) \to \operatorname{Sh}(\operatorname{V}_{n}^{\circ}, -_{n}\operatorname{K}_{n,p}^{\circ}), \\ \operatorname{sh}_{n}^{\dagger\circ} \colon \operatorname{Sh}(\operatorname{V}_{n}^{\circ}, -_{n}\operatorname{K}_{n,p}^{\dagger}) \to \operatorname{Sh}(\operatorname{V}_{n}^{\circ}, -_{n}\operatorname{K}_{n,p}^{\circ}), \\ \operatorname{sh}_{n+1}^{\dagger\circ} \colon \operatorname{Sh}(\operatorname{V}_{n+1}^{\circ}, -_{n+1}\operatorname{K}_{n+1,p}^{\dagger}) \to \operatorname{Sh}(\operatorname{V}_{n+1}^{\circ}, -_{n+1}\operatorname{K}_{n+1,p}^{\circ}) \\ \operatorname{sh}_{n+1}^{\dagger\circ} \colon \operatorname{Sh}(\operatorname{V}_{n}^{\circ}, -_{n}\operatorname{K}_{sp,p}^{\dagger}) \to \operatorname{Sh}(\operatorname{V}_{n}^{\circ}, -_{n}\operatorname{K}_{sp,p}^{\bullet}), \\ \operatorname{sh}_{sp}^{\dagger\circ} \colon \operatorname{Sh}(\operatorname{V}_{n}^{\circ}, -_{n}\operatorname{K}_{sp,p}^{\dagger}) \to \operatorname{Sh}(\operatorname{V}_{n}^{\circ}, -_{n}\operatorname{K}_{sp,p}^{\bullet}), \\ \operatorname{sh}_{sp}^{\circ\circ} : \operatorname{Sh}(\operatorname{V}_{n}^{\circ}, -_{n}\operatorname{K}_{sp,p}^{\dagger}) \to \operatorname{Sh}(\operatorname{V}_{n}^{\circ}, -_{n}\operatorname{K}_{sp,p}^{\bullet}), \\ \operatorname{sh}_{sp}^{\circ\circ} : \operatorname{Sh}(\operatorname{Sh}(\operatorname{Sh}_{n}^{\circ}, -_{n}\operatorname{Sh}_{sp,p}^{\circ}) \to \operatorname{Sh}(\operatorname{Sh}(\operatorname{Sh}_{n}^{\circ}, -_{n}\operatorname{Sh}_{sp,p}^{\circ}), \\ \operatorname{Sh}_{sh}^{\circ\circ} : \operatorname{Sh}(\operatorname{Sh}(\operatorname{Sh}_{n}^{\circ}, -_{n}\operatorname{Sh}_{sp,p}^{\circ}) \to \operatorname{Sh}(\operatorname{Sh}(\operatorname{Sh}_{n}^{\circ}, -_{n}\operatorname{Sh}_{sp,p}^{\circ}), \\ \operatorname{Sh}_{sh}^{\circ\circ} : \operatorname{Sh}(\operatorname{Sh}(\operatorname{Sh}_{n}^{\circ}, -_{n}\operatorname{Sh}_{sp,p}^{\circ}) \to \operatorname{Sh}(\operatorname{Sh}(\operatorname{Sh}_{n}^{\circ}, -_{n}\operatorname{Sh}_{sp,p}^{\circ}), \\ \operatorname{Sh}_{sh}^{\circ\circ} : \operatorname{Sh}(\operatorname{Sh}(\operatorname{Sh}_{n}^{\circ}, -_{n}\operatorname{Sh}_{sh}^{\circ\circ}) \to \operatorname{Sh}(\operatorname{Sh}(\operatorname{Sh}_{n}^{\circ}, -_{n}\operatorname{Sh}_{sh}^{\circ\circ}), \\ \operatorname{Sh}_{sh}^{\circ\circ} : \operatorname{Sh}(\operatorname{Sh}(\operatorname{Sh}_{n}$$

,

in $\operatorname{Fun}(\mathfrak{K}(V^{\circ})_{sp}^{p}, \operatorname{Set})$. Note that $\operatorname{sh}^{\circ}_{\uparrow}$ has already appeared in Remark 5.10.2.

Similar to Construction 4.3.2, we may construct two uniformization maps

(5.25)
$$\upsilon_{\rm sp}^{\bullet}\colon {\rm S}_{\mathfrak{p}}^{\bullet}({\rm V}_n,-)_{\rm sp}(\overline{\mathbb{F}}_p)\to {\rm Sh}({\rm V}_n^{\circ},-_n{\rm K}_{{\rm sp},p}^{\bullet})\times{\rm T}_{\mathfrak{p}}(\overline{\mathbb{F}}_p)$$

(5.26)
$$\upsilon_{\rm sp}^{\dagger}\colon {\rm S}_{\mathfrak{p}}^{\dagger}({\rm V}_n,-)_{\rm sp}(\overline{\mathbb{F}}_p)\to {\rm Sh}({\rm V}_n^{\circ},-_n{\rm K}_{{\rm sp},p}^{\dagger})\times{\rm T}_{\mathfrak{p}}(\overline{\mathbb{F}}_p)$$

in $\operatorname{Fun}(\mathfrak{K}(\operatorname{V}_n^\circ)_{\operatorname{sp}}^p \times \mathfrak{T}, \operatorname{Set})_{/\operatorname{T}_n(\overline{\mathbb{F}}_n)}$, which are isomorphisms. We leave the details to readers.

Proposition 5.10.14. The following diagram



in $\operatorname{Fun}(\mathfrak{K}(V_n^{\circ})_{\operatorname{sp}}^p \times \mathfrak{T}, \operatorname{Set})_{/\operatorname{T}_{\mathfrak{p}}(\overline{\mathbb{F}}_p)}$ commutes (in which all uniformization maps are isomorphisms). Moreover, the induced actions of $\operatorname{Gal}(\overline{\mathbb{F}}_p/\mathbb{F}_p^{\Phi})$ on all terms on the right-hand side factor through the projection to the factor $\operatorname{T}_{\mathfrak{p}}(\overline{\mathbb{F}}_p)$.

Proof. It follows from Constructions 5.2.6, 5.3.6, and 5.4.5.

Remark 5.10.15. When n = 1, we have the diagram (5.24) in which all terms not in the top or back layers are empty. Propositions 5.10.12 and 5.10.14 can be modified in the obvious way.

5.11. First geometric reciprocity law. In this subsection, we state and prove a theorem we call first geometric reciprocity law, which can be regarded a geometric template for the first explicit reciprocity law studied in Subsection 7.2 once throw the automorphic input.

We maintain the setup in Subsection 5.10. However, we allow - = (-n, -n+1) to be an object of $\mathfrak{K}(\mathcal{V}_n^\circ)^p \times \mathfrak{K}(\mathcal{V}_{n+1}^\circ)^p$, rather than $\mathfrak{K}(\mathcal{V}_n^\circ)_{sp}^p$. Denote by n_0 and n_1 the unique even and odd numbers in $\{n, n+1\}$, respectively. Write $n_0 = 2r_0$ and $n_1 = 2r_1 + 1$ for unique integers $r_0, r_1 \ge 1$. In particular, we have $n = r_0 + r_1$. Let L be a p-coprime coefficient ring.

To ease notation, we put $X_{n_{\alpha}}^{?} \coloneqq X_{\mathfrak{p}}^{?}(V_{n_{\alpha}}^{\circ}, -n_{\alpha})$ for meaningful triples $(X, ?, \alpha) \in \{\mathbf{M}, \mathbf{M}, \mathbf{B}, \mathbf{S}\} \times \mathbb{C}$ $\{ , \eta, \circ, \bullet, \dagger \} \times \{0, 1\}.$

Notation 5.11.1. We introduce following objects.

- (1) Put $\mathbf{P} \coloneqq \mathbf{M}_{n_0} \times_{\mathbf{T}_p} \mathbf{M}_{n_1}$.
- (2) For $(?_0, ?_1) \in \{\circ, \bullet, \dagger\}^2$, put $P^{?_0, ?_1} := M_{n_0}^{?_0} \times_{T_p} M_{n_1}^{?_1}$, which is a closed subscheme of P.¹¹ (3) Let $\sigma: \mathbf{Q} \to \mathbf{P}$ be the blow-up along the subscheme $P^{\circ, \circ}$, which is a morphism in $\mathsf{Fun}(\mathfrak{K}(\mathrm{V}_n^{\circ})^p \times \mathfrak{K}(\mathrm{V}_{n+1}^{\circ})^p \times \mathfrak{T}, \mathsf{Sch}_{/\mathbb{Z}_p^{\Phi}})_{/\mathbf{T}_{\mathfrak{p}}}.$
- (4) For $(?_0, ?_1) \in \{\circ, \bullet, \dagger\}^2$, let $Q^{?_0, ?_1}$ be the strict transform of $P^{?_0, ?_1}$ under σ , which is a closed subscheme of Q.
- (5) Let $\gamma_{?'_0,?'_1}^{?_0,?_1} \colon \mathbb{P}^{?_0,?'_1} \to \mathbb{P}^{?'_0,?'_1}$ be the closed embedding if $\mathbb{P}^{?_0,?_1}$ is contained in $\mathbb{P}^{?'_0,?'_1}$, and $\delta^{?_0,?_1}_{?'_0,?'_1} : \mathbf{Q}^{?_0,?_1} \to \mathbf{Q}^{?'_0,?'_1}$ the closed embedding if $\mathbf{Q}^{?_0,?_1}$ is contained in $\mathbf{Q}^{?'_0,?'_1}$.

Suppose that – is taken in the subcategory $\Re(V_n^\circ)_{sp}^p$

- (6) Let \mathbf{P}_{\triangle} be the graph of $\mathbf{m}_{\uparrow} \colon \mathbf{M}_n \to \mathbf{M}_{n+1}$ (5.21) over $\mathbf{T}_{\mathfrak{p}}$ in $\mathsf{Fun}(\mathfrak{K}(\mathbf{V}_n^{\circ})_{\mathrm{sp}}^p \times \mathfrak{T}, \mathsf{Sch}_{\mathbb{Z}_n^{\Phi}})/\mathbf{T}_{\mathfrak{p}},$ as a closed subscheme of **P**.
- (7) For $? = \bullet, \circ$, let $P^?_{\bigtriangleup}$ be the graph of $m^?_{\uparrow}: M^?_n \to M^?_{n+1}$ (5.22) over $T_{\mathfrak{p}}$ in $\mathsf{Fun}(\mathfrak{K}(V^{\circ}_n)^p_{\mathrm{sp}} \times$ $\mathfrak{T}, \mathsf{Sch}_{/\mathbb{F}_n^{\Phi}})_{/T_{\mathfrak{p}}}, \text{ as a closed subscheme of P}^{?,?}.$
- (8) Let \mathbf{Q}_{Δ} be the strict transform of \mathbf{P}_{Δ} under σ , which is a closed subscheme of \mathbf{Q} .

Lemma 5.11.2. The two specialization maps

$$\begin{split} \mathrm{H}^{i}_{\mathfrak{T},c}(\mathbf{Q}\otimes_{\mathbb{Z}_{p^{2}}}\overline{\mathbb{Q}}_{p},L) &\to \mathrm{H}^{i}_{\mathfrak{T},c}(\overline{\mathrm{Q}},\mathrm{R}\Psi L),\\ \mathrm{H}^{i}_{\mathfrak{T}}(\mathbf{Q}\otimes_{\mathbb{Z}_{p^{2}}}\overline{\mathbb{Q}}_{p},L) &\to \mathrm{H}^{i}_{\mathfrak{T}}(\overline{\mathrm{Q}},\mathrm{R}\Psi L), \end{split}$$

are both isomorphisms.

Proof. When \mathbf{Q} is proper, this is simply the proper base change. When \mathbf{Q} is not proper, this again follows from [LS18, Corollary 5.20].

Lemma 5.11.3. The scheme **Q** (valued at any object of $\mathfrak{K}(\mathcal{V}_n^\circ)_{sp}^p$) is strictly semistable over \mathbb{Z}_p^Φ of relative dimension 2n-1. Moreover, we have

¹¹Recall from Notation 3.3.6(5) that P is $\mathbf{P} \otimes_{\mathbb{Z}_p^{\Phi}} \mathbb{F}_p^{\Phi}$.

(1) The reduction graph of \mathbf{Q} is as follows



so that

$$\begin{split} \left\{ \begin{aligned} \mathbf{Q}^{(0)} &= \mathbf{Q}^{\circ,\circ} \coprod \mathbf{Q}^{\circ,\bullet} \coprod \mathbf{Q}^{\bullet,\bullet} \coprod \mathbf{Q}^{\bullet,\circ}, \\ \mathbf{Q}^{(1)} &= \mathbf{Q}^{\circ,\dagger} \coprod \mathbf{Q}^{\dagger,\bullet} \coprod \mathbf{Q}^{\bullet,\bullet} \coprod \mathbf{Q}^{\bullet,\circ} \coprod \mathbf{Q}^{\dagger,\circ} \coprod \mathbf{Q}^{\dagger,\circ}, \\ \mathbf{Q}^{(2)} &= (\mathbf{Q}^{\bullet,\circ} \cap \mathbf{Q}^{\dagger,\dagger}) \coprod (\mathbf{Q}^{\circ,\bullet} \cap \mathbf{Q}^{\dagger,\dagger}), \\ \mathbf{Q}^{(c)} &= \emptyset, \text{ for } c \geq 3. \end{split} \end{split}$$

Here, $Q^{(c)}$ denotes the union of the strata of Q of codimension c.

- (2) For the morphism σ , we have that
 - O the induced morphism $\sigma: \mathbb{Q}^{?_0,?_1} \to \mathbb{P}^{?_0,?_1}$ is an isomorphism if $?_0 \neq ?_1$;
 - O the induced morphism $\sigma: \mathbb{Q}^{?_0,?_1} \to \mathbb{P}^{?_0,?_1}$ is the blow-up along $\mathbb{P}^{\dagger,\dagger}$ if $(?_0,?_1) \in \{(\circ,\circ),(\bullet,\bullet)\};$
 - O the induced morphism $\sigma: \mathbb{Q}^{\dagger,\dagger} \to \mathbb{P}^{\dagger,\dagger}$ is a trivial \mathbb{P}^1 -bundle;
 - O the induced morphisms $\sigma: Q^{\bullet,\circ} \cap Q^{\dagger,\dagger} \to P^{\dagger,\dagger}$ and $\sigma: Q^{\circ,\bullet} \cap Q^{\dagger,\dagger} \to P^{\dagger,\dagger}$ are both isomorphisms.
- (3) The natural map

$$\sigma^* \colon \mathrm{H}^{i}_{\mathfrak{T}}(\overline{\mathrm{P}}^{?_0,?_1},O_{\lambda}) \to \mathrm{H}^{i}_{\mathfrak{T}}(\overline{\mathrm{Q}}^{?_0,?_1},O_{\lambda})$$

is injective, and moreover an isomorphism if $?_0 \neq ?_1$. (4) For $(?_0, ?_1) \in \{(\circ, \circ), (\bullet, \bullet)\}$, the map

$$(\delta_{?_0,?_1}^{\dagger,\dagger})_! \circ \sigma^* \colon \mathrm{H}^{i-2}_{\mathfrak{T}}(\overline{\mathrm{P}}^{\dagger,\dagger}, O_{\lambda}(-1)) \to \mathrm{H}^{i}_{\mathfrak{T}}(\overline{\mathrm{Q}}^{?_0,?_1}, O_{\lambda})$$

is injective; and we have

$$\mathrm{H}^{i}_{\mathfrak{T}}(\overline{\mathrm{Q}}^{?_{0},?_{1}},O_{\lambda}) = \sigma^{*}\mathrm{H}^{i}_{\mathfrak{T}}(\overline{\mathrm{P}}^{?_{0},?_{1}},O_{\lambda}) \bigoplus (\delta^{\dagger,\dagger}_{?_{0},?_{1}})_{!}\sigma^{*}\mathrm{H}^{i-2}_{\mathfrak{T}}(\overline{\mathrm{P}}^{\dagger,\dagger},O_{\lambda}(-1)).$$

(5) If we denote by $\mathfrak{f} \in \mathrm{H}^{2}_{\mathfrak{T}}(\overline{\mathrm{Q}}^{\dagger,\dagger},O_{\lambda}(1))$ the cycle class of an arbitrary \mathfrak{T} -orbit of fibers of the trivial \mathbb{P}^{1} -fibration $\overline{\sigma} \colon \overline{\mathrm{Q}}^{\dagger,\dagger} \to \overline{\mathrm{P}}^{\dagger,\dagger}$, then the map

$$(\mathfrak{f}\cup)\circ\sigma^*\colon\mathrm{H}^{i-2}_{\mathfrak{T}}(\overline{\mathrm{P}}^{\dagger,\dagger},O_{\lambda}(-1))\to\mathrm{H}^{i}_{\mathfrak{T}}(\overline{\mathrm{Q}}^{\dagger,\dagger},O_{\lambda})$$

is injective; and we have

$$\mathrm{H}^{i}_{\mathfrak{T}}(\overline{\mathbf{Q}}^{\dagger,\dagger},O_{\lambda})=\sigma^{*}\mathrm{H}^{i}_{\mathfrak{T}}(\overline{\mathbf{P}}^{\dagger,\dagger},O_{\lambda})\bigoplus\mathfrak{f}\cup\sigma^{*}\mathrm{H}^{i-2}_{\mathfrak{T}}(\overline{\mathbf{P}}^{\dagger,\dagger},O_{\lambda}(-1)).$$

Proof. Parts (1,2) follow from a standard computation of blow-up. Parts (3-5) follow from (2).

Let $(\mathbb{E}_{s}^{p,q}, \mathbf{d}_{s}^{p,q})$ be the weight spectral sequence¹² abutting to the cohomology $\mathrm{H}_{\mathfrak{T}}^{p+q}(\overline{\mathbf{Q}}, \mathrm{R}\Psi O_{\lambda}(n))$, whose first page is as follows:

(5.27)



with $\mathbb{E}_1^{p,q} = 0$ if |p| > 2. The following lemma will be used later.

Construction 5.11.4. For $\alpha = 0, 1$, let $\xi_{\alpha} \in \mathrm{H}^{2}_{\mathfrak{T}}(\overline{\mathrm{B}}^{\circ}_{n_{\alpha}}, L(1))$ be the first Chern class of the tautological quotient line bundle on $\overline{\mathrm{B}}^{\circ}_{n_{\alpha}}$. We construct four new pairs of maps in $\mathsf{Fun}(\mathfrak{K}(\mathrm{V}^{\circ}_{n})^{p} \times$

¹²Strictly speaking, the differential maps $d_s^{p,q}$ depend on the choice of the ordering of (types of) irreducible components of Q, which we choose to be the clockwise order $Q^{\circ,\circ} < Q^{\circ,\bullet} < Q^{\bullet,\bullet} < Q^{\bullet,\circ}$.

 $\mathfrak{K}(\mathrm{V}_{n+1}^{\circ})^p,\mathsf{Mod}(L))$ as follows:

$$\begin{cases} \operatorname{inc}_{!}^{\circ,\dagger} \colon L[\operatorname{Sh}(\operatorname{V}_{n_{0}}^{\circ},-n_{0}\operatorname{K}_{n_{0}}^{\circ},p)] \otimes_{L} L[\operatorname{Sh}(\operatorname{V}_{n_{1}}^{\circ},-n_{1}\operatorname{K}_{n_{1},p}^{\circ})] \\ \xrightarrow{\sim} \operatorname{H}_{2}^{0}(\overline{\operatorname{S}}_{n_{0}}^{\circ},L) \otimes_{L} \operatorname{H}_{2}^{0}(\overline{\operatorname{S}}_{n_{1}}^{\circ},L) = \operatorname{H}_{2}^{0}(\overline{\operatorname{S}}_{n_{0}}^{\circ} \times \overline{\operatorname{T}_{p}} \overline{\operatorname{S}}_{n_{1}}^{\circ},L) \\ \xrightarrow{(\pi_{n_{0}}^{\circ} \times \pi_{n_{1}}^{\circ})^{*}} \operatorname{H}_{2}^{0}(\overline{\operatorname{B}}_{n_{0}}^{\circ} \times \overline{\operatorname{T}_{p}} \overline{\operatorname{B}}_{n_{1}}^{\circ},L) \\ \xrightarrow{(\underline{\cup}_{0}^{\varepsilon_{0}^{\circ-1}} \cup \underline{\zeta}_{1}^{\varepsilon_{1}^{-1}})} \operatorname{H}_{2}^{2(n-2)}(\overline{\operatorname{R}}_{n_{0}}^{\circ} \times \overline{\operatorname{T}_{p}} \overline{\operatorname{B}}_{n_{1}}^{\circ},L(n-2)) \\ \xrightarrow{(id \times m_{n_{1}}^{\dagger})^{*}} \operatorname{H}_{2}^{2(n-2)}(\overline{\operatorname{M}}_{n_{0}}^{\circ} \times \overline{\operatorname{T}_{p}} \overline{\operatorname{M}}_{n_{1}}^{\dagger},L(n-2)) \\ \xrightarrow{(id \times m_{n_{1}}^{\dagger})^{*}} \operatorname{H}_{2}^{2(n-2)}(\overline{\operatorname{M}}_{n_{0}}^{\circ} \times \overline{\operatorname{T}_{p}} \overline{\operatorname{M}}_{n_{1}}^{\bullet},L(n-1)) \\ \xrightarrow{(id \times m_{n_{1}}^{\dagger})^{*}} \operatorname{H}_{2}^{2(n-2)}(\overline{\operatorname{M}}_{n_{0}}^{\circ} \times \overline{\operatorname{T}_{p}} \overline{\operatorname{M}}_{n_{1}}^{\bullet},L(n-1)) \\ \xrightarrow{(id \times m_{n_{1}}^{\dagger})^{*}} \operatorname{H}_{2}^{2(n}(\overline{\operatorname{M}}_{n_{0}}^{\circ} \times \overline{\operatorname{T}_{p}} \overline{\operatorname{M}}_{n_{1}}^{\bullet},L(n)) \\ \xrightarrow{(id \times m_{n_{1}}^{\dagger})^{*}} \operatorname{H}_{2}^{2(n}(\overline{\operatorname{M}}_{n_{0}^{\circ} \times \overline{\operatorname{T}_{p}} \overline{\operatorname{M}}_{n_{1}}^{\circ},L(n+1)) \\ \xrightarrow{(id \times m_{n_{1}}^{\dagger})^{*}} \operatorname{H}_{2}^{2(n-2)}(\overline{\operatorname{M}}_{n_{0}^{\circ} \times \overline{\operatorname{T}_{p}} \overline{\operatorname{M}}_{n_{1}}^{\circ},L(n+1)) \\ \xrightarrow{(id \times m_{n_{1}}^{\dagger})^{*}} \operatorname{H}_{2}^{2(n-2)}(\overline{\operatorname{M}}_{n_{0}^{\circ} \times \overline{\operatorname{T}_{p}} \overline{\operatorname{M}}_{n$$

$$\begin{cases} \operatorname{inc}_{!}^{\circ,\bullet} \colon L[\operatorname{Sh}(\operatorname{V}_{n_{0}}^{\circ},-n_{0}\operatorname{K}_{n_{0},p}^{\circ})] \otimes_{L} L[\operatorname{Sh}(\operatorname{V}_{n_{1}}^{\circ},-n_{1}\operatorname{K}_{n_{1},p}^{\bullet})] \\ \xrightarrow{\sim} \operatorname{H}_{\mathfrak{T}}^{0}(\overline{\operatorname{S}}_{n_{0}}^{\circ},L) \otimes_{L} \operatorname{H}_{\mathfrak{T}}^{0}(\overline{\operatorname{S}}_{n_{1}}^{\bullet},L) = \operatorname{H}_{\mathfrak{T}}^{0}(\overline{\operatorname{S}}_{n_{0}}^{\circ} \times_{\overline{\operatorname{T}}_{p}} \overline{\operatorname{S}}_{n_{1}}^{\bullet},L) \\ \xrightarrow{(\pi_{n_{0}}^{\circ} \times \pi_{n_{1}}^{\bullet})^{*}} \operatorname{H}_{\mathfrak{T}}^{0}(\overline{\operatorname{B}}_{n_{0}}^{\circ} \times_{\overline{\operatorname{T}}_{p}} \overline{\operatorname{B}}_{n_{1}}^{\bullet},L) \\ \xrightarrow{(\psi_{n_{0}}^{\circ} \times \psi_{n_{1}}^{\bullet})^{\circ}} \operatorname{H}_{\mathfrak{T}}^{2(r_{0}-1)}(\overline{\operatorname{B}}_{n_{0}}^{\circ} \times_{\overline{\operatorname{T}}_{p}} \overline{\operatorname{B}}_{n_{1}}^{\bullet},L(r_{0}-1)) \\ \xrightarrow{(\psi_{n_{0}}^{\circ} \times \psi_{n_{1}}^{\bullet})^{\circ}} \operatorname{H}_{\mathfrak{T},c}^{2(n-1)}(\overline{\operatorname{M}}_{n_{0}}^{\circ} \times_{\overline{\operatorname{T}}_{p}} \overline{\operatorname{M}}_{n_{1}}^{\bullet},L(n-1)) = \operatorname{H}_{\mathfrak{T},c}^{2(n-1)}(\overline{\operatorname{P}}^{\circ,\bullet},L(n-1)), \\ \operatorname{inc}_{\circ,\bullet}^{*} \colon \operatorname{H}_{\mathfrak{T}}^{2n}(\overline{\operatorname{P}}^{\circ,\bullet},L(n)) = \operatorname{H}_{\mathfrak{T}}^{2n}(\overline{\operatorname{M}}_{n_{0}}^{\circ} \times_{\overline{\operatorname{T}}_{p}} \overline{\operatorname{M}}_{n_{1}}^{\bullet},L(n)) \\ \xrightarrow{(\psi_{n_{0}}^{\circ} \times \psi_{n_{1}}^{\bullet})^{*}} \operatorname{H}_{\mathfrak{T}}^{2n}(\overline{\operatorname{B}}_{n_{0}}^{\circ} \times_{\overline{\operatorname{T}}_{p}} \overline{\operatorname{B}}_{n_{1}}^{\bullet},L(n)) \\ \xrightarrow{(\psi_{n_{0}}^{\circ} \times \psi_{n_{1}}^{\bullet})^{*}} \operatorname{H}_{\mathfrak{T}}^{2(n-1+r_{1})}(\overline{\operatorname{B}}_{n_{0}}^{\circ} \times_{\overline{\operatorname{T}}_{p}} \overline{\operatorname{B}}_{n_{1}}^{\bullet},L(n_{0}-1+r_{1})) \\ \xrightarrow{(\pi_{n_{0}}^{\circ} \times \pi_{n_{1}}^{\bullet})^{*}} \operatorname{H}_{\mathfrak{T}}^{2(n-1+r_{1})$$

$$\begin{cases} \operatorname{inc}_{!}^{\bullet,\dagger} : L[\operatorname{Sh}(\mathbb{V}_{n_{0}}^{\circ}, -n_{0}\mathbb{K}_{n_{0},p}^{\circ})] \otimes_{L} L[\operatorname{Sh}(\mathbb{V}_{n_{1}}^{\circ}, -n_{1}\mathbb{K}_{n_{1},p}^{\circ})] \\ \xrightarrow{\longrightarrow} \operatorname{H}_{2}^{0}(\overline{S}_{n_{0}}^{\circ}, L) \otimes_{L} \operatorname{H}_{2}^{0}(\overline{S}_{n_{1}}^{\circ}, L) = \operatorname{H}_{2}^{0}(\overline{S}_{n_{0}}^{\circ} \times \overline{T_{p}} \overline{S}_{n_{1}}^{\circ}, L) \\ \xrightarrow{(\pi_{0}^{\circ} \times \pi_{n}^{\circ})^{*}} \operatorname{H}_{2}^{0}(\overline{B}_{n_{0}}^{\circ} \times \overline{T_{p}} \overline{B}_{n_{1}}^{\circ}, L(r_{1} - 1)) \\ \xrightarrow{(\operatorname{id} \times \iota_{n_{1}}^{\circ})^{\circ}} \operatorname{H}_{2}^{2r_{1}-2}(\overline{B}_{n_{0}}^{\circ} \times \overline{T_{p}} \overline{M}_{n}^{\circ}, L(r_{1} - 1)) \\ \xrightarrow{(\operatorname{id} \times \iota_{n_{1}}^{\circ})^{\circ}} \operatorname{H}_{2}^{2r_{1}-2}(\overline{B}_{n_{0}}^{\circ} \times \overline{T_{p}} \overline{M}_{n}^{\circ}, L(r_{1} - 1)) \\ \xrightarrow{(\operatorname{id} \times \iota_{n_{1}}^{\circ})^{\circ}} \operatorname{H}_{2}^{2r_{1}-2}(\overline{B}_{n_{0}}^{\circ} \times \overline{T_{p}} \overline{M}_{n}^{\circ}, L(r_{1} - 1)) \\ \xrightarrow{(\operatorname{id} \times \iota_{n_{1}}^{\circ})^{\circ}} \operatorname{H}_{2}^{2r_{1}-2}(\overline{B}_{n_{0}}^{\circ} \times \overline{T_{p}} \overline{M}_{n}^{\circ}, L(r_{1} - 1)) \\ \xrightarrow{(\operatorname{id} \times \iota_{n_{1}}^{\circ})^{\circ}} \operatorname{H}_{2}^{2r_{1}-2}(\overline{B}_{n_{0}}^{\circ} \times \overline{T_{p}} \overline{M}_{n}^{\circ}, L(r_{1} - 1)) \\ \xrightarrow{(\operatorname{id} \times \iota_{n_{1}}^{\circ})^{\circ}} \operatorname{H}_{2}^{2r_{1}-2}(\overline{B}_{n_{0}}^{\circ} \times \overline{T_{p}} \overline{M}_{n}^{\circ}, L(n - 1)) = \operatorname{H}_{2,c}^{2(n-1)}(\overline{P}^{\bullet,\bullet}, L(n - 1)), \\ \operatorname{inc}_{\bullet,\dagger,\dagger}^{\bullet,\dagger} : \operatorname{H}_{2}^{2n}(\overline{P}^{\bullet,\bullet}, L(n)) = \operatorname{H}_{2}^{2n}(\overline{M}_{n_{0}}^{\circ} \times \overline{T_{p}} \overline{M}_{n}^{\circ}, L(n - 1)) \\ \xrightarrow{(\operatorname{id} \times \iota_{n_{1}}^{\circ})^{\circ}} \operatorname{H}_{2}^{2r_{1}-2}(\overline{B}_{n_{0}}^{\circ} \times \overline{T_{p}} \overline{M}_{n}^{\circ}, L(n + 1)) \\ \xrightarrow{(\operatorname{id} \times \iota_{n_{1}}^{\circ})^{\circ}} \operatorname{H}_{2}^{2n+2}(\overline{B}_{n_{0}}^{\circ} \times \overline{T_{p}} \overline{M}_{n}^{\circ}, L(n + 1)) \\ \xrightarrow{(\operatorname{id} \times \iota_{n_{1}}^{\circ})^{\circ}} \operatorname{H}_{2}^{2(n-1)}(\overline{B}_{n_{0}}^{\circ} \times \overline{T_{p}} \overline{B}_{n}^{\circ}, L(n + 1)) \\ \xrightarrow{(\operatorname{id} \times \iota_{n_{1}}^{\circ})^{\circ}} \operatorname{H}_{2}^{2}(\overline{S}_{n_{0}}^{\circ} \times \overline{T_{p}} \overline{S}_{n}^{\circ}, L) = \operatorname{H}_{2}^{0}(\overline{S}_{n_{0}}^{\circ}, L) \otimes_{L} \operatorname{H}_{2}^{0}(\overline{S}_{n_{1}}^{\circ}, L) \\ \xrightarrow{\to} L[\operatorname{Sh}(\mathbb{V}_{n_{0}}^{\circ}, -n_{0}\mathbb{K}_{n_{0}p})] \otimes_{L} L[\operatorname{Sh}(\mathbb{V}_{n_{1}}^{\circ}, -n_{1}\mathbb{K}_{n_{1}p}^{\circ})]; \\ \xrightarrow{\to} L[\operatorname{Sh}(\mathbb{V}_{n_{0}}^{\circ}, -n_{0}\mathbb{K}_{n_{0}p}^{\circ})] \otimes_{L} L[\operatorname{Sh}(\mathbb{V}_{n_{1}}^{\circ}, -n_{1}\mathbb{K}_{n_{1}p}^{\circ})] \\ \xrightarrow{\to} L[\operatorname{Sh}(\mathbb{V}_{n_{0}}^{\circ}, -n_{0}\mathbb{K}_{n_{0}}^{\circ} \times \overline{T_{p}} \overline{B}_{n_{1}}^{\circ}, L) = \operatorname{H}_{2}^{0}(\overline{S}_{n_{0}}^{\circ} \times \overline{T_{p}} \overline{$$

$$\operatorname{Inc}_{\bullet,\bullet}^{*} \colon \operatorname{H}_{\mathfrak{T}}^{2n}(\overline{\mathbf{P}}^{\bullet,\bullet}, L(n)) = \operatorname{H}_{\mathfrak{T}}^{2n}(\overline{\mathrm{M}}_{n_{0}}^{\bullet} \times_{\overline{\mathrm{T}}_{\mathfrak{p}}} \overline{\mathrm{M}}_{n_{1}}^{\bullet}, L(n))$$

$$\xrightarrow{(\iota_{n_{0}}^{\bullet} \times \iota_{n_{1}}^{\bullet})^{*}} \operatorname{H}_{\mathfrak{T}}^{2n}(\overline{\mathrm{B}}_{n_{0}}^{\bullet} \times_{\overline{\mathrm{T}}_{\mathfrak{p}}} \overline{\mathrm{B}}_{n_{1}}^{\bullet}, L(n))$$

$$\xrightarrow{(\pi_{n_{0}}^{\bullet} \times \pi_{n_{1}}^{\bullet})!} \operatorname{H}_{\mathfrak{T}}^{0}(\overline{\mathrm{S}}_{n_{0}}^{\bullet} \times_{\overline{\mathrm{T}}_{\mathfrak{p}}} \overline{\mathrm{S}}_{n_{1}}^{\bullet}, L) = \operatorname{H}_{\mathfrak{T}}^{0}(\overline{\mathrm{S}}_{n_{0}}^{\bullet}, L) \otimes_{L} \operatorname{H}_{\mathfrak{T}}^{0}(\overline{\mathrm{S}}_{n_{1}}^{\bullet}, L)$$

$$\xrightarrow{\sim} L[\operatorname{Sh}(\operatorname{V}_{n_{0}}^{\circ}, -_{n_{0}}\operatorname{K}_{n_{0}, p}^{\bullet})] \otimes_{L} L[\operatorname{Sh}(\operatorname{V}_{n_{1}}^{\circ}, -_{n_{1}}\operatorname{K}_{n_{1}, p}^{\bullet})].$$

In fact, the two maps in each pair are Poincaré dual to each other.

Theorem 5.11.5 (First geometric reciprocity law). Take an object $K^{p\circ} \in \mathfrak{K}(V_n^\circ)_{sp}^p$. For the class $\operatorname{cl}(\mathbb{P}^{\bullet}_{\Delta}) \in \operatorname{H}^{2n}_{\mathfrak{T}}(\overline{\mathbb{P}}^{\bullet,\bullet}, L(n))$, we have (1) For $f \in L[\operatorname{Sh}(V_{n_0}^\circ, K_{n_0}^{p\circ} K_{n_0,p}^\bullet)] \otimes_L L[\operatorname{Sh}(V_{n_1}^\circ, K_{n_1}^{p\circ} K_{n_1,p}^\circ)]$, the identity

$$\int_{\overline{\mathbf{P}}^{\bullet,\bullet}}^{\mathfrak{L}} \mathrm{cl}(\mathbf{P}^{\bullet}_{\bigtriangleup}) \cup \mathrm{inc}_{!}^{\bullet,\dagger}(f) = \sum_{s \in \mathrm{Sh}(\mathbf{V}^{\circ}_{n},\mathbf{K}^{p\circ}_{n}\mathbf{K}^{\bullet}_{\mathrm{sp},p})} (\mathbf{T}^{\bullet\circ}_{n_{1},\mathfrak{p}}f)(\mathrm{sh}^{\bullet}_{\downarrow}(s), \mathrm{sh}^{\bullet}_{\uparrow}(s))$$

holds. (2) For $f \in L[\operatorname{Sh}(\operatorname{V}_{n_{0}}^{\circ}, \operatorname{K}_{n_{0}}^{p\circ}\operatorname{K}_{n_{0}, p}^{\bullet})] \otimes_{L} L[\operatorname{Sh}(\operatorname{V}_{n_{1}}^{\circ}, \operatorname{K}_{n_{1}}^{p\circ}\operatorname{K}_{n_{1}, p}^{\bullet})], \text{ the identity}$ $\int_{\overline{P}^{\bullet, \bullet}}^{\mathfrak{T}} \operatorname{cl}(\operatorname{P}_{\Delta}^{\bullet}) \cup \operatorname{inc}_{!}^{\bullet, \bullet}(f) = \sum_{s \in \operatorname{Sh}(\operatorname{V}_{n}^{\circ}, \operatorname{K}_{n}^{p\circ}\operatorname{K}_{sp, p}^{\bullet})} (\operatorname{T}_{n_{1}, \mathfrak{p}}^{\bullet}f)(\operatorname{sh}_{\downarrow}^{\bullet}(s), \operatorname{sh}_{\uparrow}^{\bullet}(s))$ holds. (3) For $f \in L[\operatorname{Sh}(\operatorname{V}_{n_{0}}^{\circ}, \operatorname{K}_{n_{0}}^{p\circ}\operatorname{K}_{n_{0}, p}^{\circ})] \otimes_{L} L[\operatorname{Sh}(\operatorname{V}_{n_{1}}^{\circ}, \operatorname{K}_{n_{1}}^{p\circ}\operatorname{K}_{n_{1}, p}^{\circ})], \text{ the identity}$ $\int_{\overline{P}^{\bullet, \bullet}}^{\mathfrak{T}} \operatorname{cl}(\operatorname{P}_{\Delta}^{\bullet}) \cup \left(\operatorname{inc}_{!}^{\bullet, \dagger}(\operatorname{T}_{n_{0}, \mathfrak{p}}^{\circ\circ} \otimes \operatorname{I}_{n_{1}, \mathfrak{p}}^{\circ}f) + (p+1)^{2}\operatorname{inc}_{!}^{\bullet, \bullet}(\operatorname{T}_{n_{0}, \mathfrak{p}}^{\bullet\circ} \otimes \operatorname{T}_{n_{1}, \mathfrak{p}}^{\bullet, \bullet}f)\right)$ $= \sum_{s \in \operatorname{Sh}(\operatorname{V}_{n}^{\circ}, \operatorname{K}_{n}^{p\circ}\operatorname{K}_{n}^{\circ}, p)} (\operatorname{I}_{n_{0}, \mathfrak{p}}^{\circ} \otimes \operatorname{T}_{n_{1}, \mathfrak{p}}^{\circ}f)(s, \operatorname{sh}_{\uparrow}^{\circ}(s))$

holds.

Here, $\int_{\overline{P}^{\bullet,\bullet}}^{\mathfrak{T}}$ denotes the \mathfrak{T} -trace map in Definition 3.5.8; and $\mathrm{sh}^{\circ}_{\uparrow}$, $\mathrm{sh}^{\bullet}_{\uparrow}$, and $\mathrm{sh}^{\bullet}_{\downarrow}$ are maps in Notation 5.10.13.

The intersection number in (3) is the actual one that is responsible for the first explicit reciprocity law which will be discussed in Subsection 7.2.

Proof. We first show (3) assuming (1) and (2). By (1), (2), and Lemma B.4.4, we have for $f \in L[\operatorname{Sh}(\operatorname{V}_{n_0}^\circ, \operatorname{K}_{n_0}^{p\circ}\operatorname{K}_{n_0,p}^\circ)] \otimes_L L[\operatorname{Sh}(\operatorname{V}_{n_1}^\circ, \operatorname{K}_{n_1}^{p\circ}\operatorname{K}_{n_1,p}^\circ)],$

$$\begin{split} &\int_{\overline{\mathbf{P}}\bullet,\bullet}^{\overline{\mathbf{x}}} \mathrm{cl}(\mathbf{P}^{\bullet}_{\Delta}) \cup \left(\mathrm{inc}_{!}^{\bullet,\dagger}(\mathbf{T}^{\bullet\circ}_{n_{0},\mathfrak{p}}\otimes\mathbf{I}^{\circ}_{n_{1},\mathfrak{p}}f) + (p+1)^{2}\mathrm{inc}_{!}^{\bullet,\bullet}(\mathbf{T}^{\bullet\circ}_{n_{0},\mathfrak{p}}\otimes\mathbf{T}^{\bullet\circ}_{n_{1},\mathfrak{p}}f)\right) \\ &= \sum_{s\in\mathrm{Sh}(\mathbf{V}^{\circ}_{n},\mathbf{K}^{p\circ}_{n}\mathbf{K}^{\bullet}_{\mathrm{sp},p})} (\mathbf{T}^{\bullet\circ}_{n_{0},\mathfrak{p}}\otimes(\mathbf{T}^{\bullet\circ}_{n_{1},\mathfrak{p}}\circ\mathbf{I}^{\circ}_{n_{1},\mathfrak{p}})f)(\mathrm{sh}^{\bullet}_{\downarrow}(s),\mathrm{sh}^{\bullet}_{\uparrow}(s)) \\ &+ \sum_{s\in\mathrm{Sh}(\mathbf{V}^{\circ}_{n},\mathbf{K}^{p\circ}_{n}\mathbf{K}^{\bullet}_{\mathrm{sp},p})} (\mathbf{T}^{\bullet\circ}_{n_{0},\mathfrak{p}}\otimes((p+1)^{2}\mathbf{T}^{\bullet}_{n_{1},\mathfrak{p}}\circ\mathbf{T}^{\bullet\circ}_{n_{1},\mathfrak{p}})f)(\mathrm{sh}^{\bullet}_{\downarrow}(s),\mathrm{sh}^{\bullet}_{\uparrow}(s)) \\ &= \sum_{s\in\mathrm{Sh}(\mathbf{V}^{\circ}_{n},\mathbf{K}^{p\circ}_{n}\mathbf{K}^{\bullet}_{\mathrm{sp},p})} (\mathbf{T}^{\bullet\circ}_{n_{0},\mathfrak{p}}\otimes(\mathbf{T}^{\bullet\circ}_{n_{1},\mathfrak{p}}\circ\mathbf{I}^{\circ}_{n_{1},\mathfrak{p}})f)(\mathrm{sh}^{\bullet}_{\downarrow}(s),\mathrm{sh}^{\bullet}_{\uparrow}(s)) \\ &+ \sum_{s\in\mathrm{Sh}(\mathbf{V}^{\circ}_{n},\mathbf{K}^{p\circ}_{n}\mathbf{K}^{\bullet}_{\mathrm{sp},p})} (\mathbf{T}^{\bullet\circ}_{n_{0},\mathfrak{p}}\otimes(\mathbf{T}^{\bullet\circ}_{n_{1},\mathfrak{p}}\circ\mathbf{T}^{\circ}_{n_{1},\mathfrak{p}}-\mathbf{T}^{\bullet\circ}_{n_{1},\mathfrak{p}}\circ\mathbf{I}^{\circ}_{n_{1},\mathfrak{p}})f)(\mathrm{sh}^{\bullet}_{\downarrow}(s),\mathrm{sh}^{\bullet}_{\uparrow}(s)) \\ &= \sum_{s\in\mathrm{Sh}(\mathbf{V}^{\circ}_{n},\mathbf{K}^{p\circ}_{n}\mathbf{K}^{\bullet}_{\mathrm{sp},p})} (\mathbf{T}^{\bullet\circ}_{n_{0},\mathfrak{p}}\otimes(\mathbf{T}^{\bullet\circ}_{n_{1},\mathfrak{p}}\circ\mathbf{T}^{\circ}_{n_{1},\mathfrak{p}})f)(\mathrm{sh}^{\bullet}_{\downarrow}(s),\mathrm{sh}^{\bullet}_{\uparrow}(s))) \\ &= \sum_{s\in\mathrm{Sh}(\mathbf{V}^{\circ}_{n},\mathbf{K}^{p\circ}_{n}\mathbf{K}^{\bullet}_{\mathrm{sp},p})} (\mathbf{T}^{\bullet\circ}_{n_{0},\mathfrak{p}}\otimes(\mathbf{T}^{\bullet\circ}_{n_{1},\mathfrak{p}}\circ\mathbf{T}^{\circ}_{n_{1},\mathfrak{p}})f)(\mathrm{sh}^{\bullet}_{\downarrow}(s),\mathrm{sh}^{\bullet}_{\uparrow}(s))) \end{aligned}$$

which, by Lemma 5.11.6 below, equals

$$\sum_{s\in \mathrm{Sh}(\mathrm{V}_n^\circ,\mathrm{K}_n^{\mathrm{p}^\circ}\mathrm{K}_{n,p}^\circ)}(\mathtt{I}_{n_0,\mathfrak{p}}^\circ\otimes \mathtt{T}_{n_1,\mathfrak{p}}^\circ f)(s,\mathrm{sh}^\circ_\uparrow(s)).$$

Thus, (3) is proved.

Now we consider (1) and (2) simultaneously. Similar to the maps $\operatorname{inc}_{!}^{\bullet}$ and $\operatorname{inc}_{!}^{\dagger}$ in Construction 5.7.3, we have maps

$$\operatorname{inc}_{\alpha}^{\bullet} \colon L[\operatorname{Sh}(\operatorname{V}_{n_{\alpha}}^{\circ}, \operatorname{K}_{n_{\alpha}}^{p_{\alpha}}\operatorname{K}_{n_{\alpha}, p}^{\bullet})] \to \operatorname{H}_{\mathfrak{T}, c}^{2(r_{\alpha}+\alpha-1)}(\overline{\operatorname{M}}_{n_{\alpha}}^{\bullet}, L(r_{\alpha}+\alpha-1)), \\ \operatorname{inc}_{\alpha}^{\dagger} \colon L[\operatorname{Sh}(\operatorname{V}_{n_{\alpha}}^{\circ}, \operatorname{K}_{n_{\alpha}}^{p_{\alpha}}\operatorname{K}_{n_{\alpha}, p}^{\circ})] \to \operatorname{H}_{\mathfrak{T}, c}^{2(r_{\alpha}+\alpha-1)}(\overline{\operatorname{M}}_{n_{\alpha}}^{\bullet}, L(r_{\alpha}+\alpha-1)),$$

for $\alpha = 0, 1$. Note that we now take $H_{\mathfrak{T},c}$ for the target of the maps rather than $H_{\mathfrak{T}}$. Moreover, in the calculation below, we will frequently use the following formula for intersection number pairings:

for a finite morphism $i: X \to Y$ of smooth schemes over an algebraically closed field, and proper smooth subschemes X' of X and Y' of Y, we have

$$\langle X_{\triangle}, X' \times Y' \rangle_{X \times Y} = \langle X'_{\triangle}, X' \times Y' \rangle_{X' \times Y} = \langle i_* X', Y' \rangle_Y$$

where X_{\triangle} and X'_{\triangle} denote by the graphs of *i* and *i* | X', respectively. The proof for (1) and (2) differs by the parity of *n*.

We first consider the case where $n = n_0$ is even. By Lemma 5.10.4(1) and Proposition 5.10.14, $\mathrm{sh}^{\bullet}_{\downarrow}$ is an isomorphism. Take a point $s^{\bullet}_n \in \mathrm{Sh}(\mathrm{V}^{\circ}_n, \mathrm{K}^{p\circ}_n \mathrm{K}^{\bullet}_{n,p})$. Let s^{\bullet} be the unique element in $\mathrm{Sh}(\mathrm{V}^{\circ}_n, \mathrm{K}^{p\circ}_n \mathrm{K}^{\bullet}_{\mathrm{sp},p})$ such that $s^{\bullet}_n = \mathrm{sh}^{\bullet}_{\downarrow}(s^{\bullet})$, and put $s^{\bullet}_{n+1} \coloneqq \mathrm{sh}^{\bullet}_{\uparrow}(s^{\bullet})$. By (the last assertion in) Proposition 5.10.12, we have

$$\mathrm{m}^{\bullet}_{\uparrow !}\mathrm{inc}^{\bullet}_{0}(1_{s^{\bullet}_{n}}) = \mathrm{inc}^{\bullet}_{1}(1_{s^{\bullet}_{n+1}})$$

For (1), we have for every $s'_{n+1} \in \operatorname{Sh}(\operatorname{V}_{n+1}^{\circ}, \operatorname{K}_{n+1}^{p\circ}\operatorname{K}_{n+1,p}^{\circ})$ the identity

$$\begin{split} \int_{\overline{\mathbf{P}}^{\bullet,\bullet}}^{\mathfrak{T}} \mathrm{cl}(\mathbf{P}^{\bullet}_{\bigtriangleup}) \cup \mathrm{inc}_{!}^{\bullet,\dagger}(\mathbf{1}_{(s_{n}^{\bullet},s_{n+1}^{\prime})}) &= \int_{\overline{\mathrm{M}}_{n+1}}^{\mathfrak{T}} \left(\mathrm{m}_{\uparrow !}^{\bullet}\mathrm{inc}_{0}^{\bullet}(\mathbf{1}_{s_{n}^{\bullet}}) \right) \cup \mathrm{inc}_{1}^{\dagger}(\mathbf{1}_{s_{n+1}^{\prime}}) \\ &= \int_{\overline{\mathrm{M}}_{n+1}}^{\mathfrak{T}} \mathrm{inc}_{0}^{\bullet}(\mathbf{1}_{s_{n+1}^{\bullet}}) \cup \mathrm{inc}_{1}^{\dagger}(\mathbf{1}_{s_{n+1}^{\prime}}). \end{split}$$

Thus, (1) follows from Proposition 5.7.6. For (2), we have for every $s'_{n+1} \in Sh(V_{n+1}^{\circ}, K_{n+1}^{p\circ}K_{n+1,p}^{\bullet})$ the identity

$$\int_{\overline{\mathbf{P}}^{\bullet,\bullet}}^{\mathfrak{T}} \operatorname{cl}(\mathbf{P}^{\bullet}_{\Delta}) \cup \operatorname{inc}_{!}^{\bullet,\bullet}(\mathbf{1}_{(s_{n}^{\bullet},s_{n+1}')}) = \int_{\overline{\mathbf{M}}_{n+1}}^{\mathfrak{T}} \left(\operatorname{m}^{\bullet}_{\uparrow !} \operatorname{inc}_{0}^{\bullet}(\mathbf{1}_{s_{n}^{\bullet}}) \right) \cup \operatorname{inc}_{1}^{\bullet}(\mathbf{1}_{s_{n+1}'})$$
$$= \int_{\overline{\mathbf{M}}_{n+1}}^{\mathfrak{T}} \operatorname{inc}_{0}^{\bullet}(\mathbf{1}_{s_{n+1}^{\bullet}}) \cup \operatorname{inc}_{1}^{\bullet}(\mathbf{1}_{s_{n+1}'}).$$

Thus, (2) follows from Proposition 5.7.6.

We then consider the case where $n = n_1$ is odd. Take a point $s_{n+1}^{\bullet} \in \text{Sh}(V_{n+1}^{\circ}, K_{n+1}^{p\circ}K_{n+1,p}^{\bullet})$. By Proposition 5.10.6, Proposition 5.10.12, and Proposition 5.10.14, we have

$$\mathbf{n}^{\bullet\ast}_{\uparrow} \mathrm{inc}^{\bullet}_{0}(1_{s^{\bullet}_{n+1}}) = \mathrm{inc}^{\bullet}_{1}(\mathrm{sh}^{\bullet}_{\downarrow} \mathrm{sh}^{\bullet\ast}_{\uparrow} 1_{s^{\bullet}_{n+1}}).$$

For (1), we have for every $s'_n \in Sh(\mathcal{V}_n^\circ, \mathcal{K}_n^{p\circ}\mathcal{K}_{n,p}^\circ)$ the identity

r

$$\begin{split} \int_{\overline{\mathbf{P}}^{\bullet,\bullet}}^{\mathfrak{T}} \mathrm{cl}(\mathbf{P}^{\bullet}_{\bigtriangleup}) \cup \mathrm{inc}_{!}^{\bullet,\dagger}(\mathbf{1}_{(s_{n+1}^{\bullet},s_{n}')}) &= \int_{\overline{\mathbf{M}}_{n}^{\bullet}}^{\mathfrak{T}} \left(\mathbf{m}^{\bullet*}_{\uparrow}\mathrm{inc}_{0}^{\bullet}(\mathbf{1}_{s_{n+1}})\right) \cup \mathrm{inc}_{1}^{\dagger}(\mathbf{1}_{s_{n}'}) \\ &= \int_{\overline{\mathbf{M}}_{n}^{\bullet}}^{\mathfrak{T}} \mathrm{inc}_{1}^{\bullet}(\mathrm{sh}^{\bullet}_{\downarrow!}\mathrm{sh}^{\bullet*}_{\uparrow}\mathbf{1}_{s_{n+1}^{\bullet}}) \cup \mathrm{inc}_{1}^{\dagger}(\mathbf{1}_{s_{n}'}). \end{split}$$

Thus, (1) follows from Proposition 5.7.6. For (2), we have for every $s'_n \in \operatorname{Sh}(\operatorname{V}^{\circ}_n, \operatorname{K}^{p\circ}_n \operatorname{K}^{\bullet}_{n,p})$ the identity

$$\begin{split} \int_{\overline{\mathbf{P}}^{\bullet,\bullet}}^{\mathfrak{T}} \mathrm{cl}(\mathbf{P}^{\bullet}_{\bigtriangleup}) \cup \mathrm{inc}_{!}^{\bullet,\bullet}(\mathbf{1}_{(s_{n+1}^{\bullet},s_{n}')}) &= \int_{\overline{\mathbf{M}}_{n}^{\bullet}}^{\mathfrak{T}} \left(\mathbf{m}^{\bullet*}_{\uparrow}\mathrm{inc}_{0}^{\bullet}(\mathbf{1}_{s_{n+1}^{\bullet}})\right) \cup \mathrm{inc}_{1}^{\bullet}(\mathbf{1}_{s_{n}'}) \\ &= \int_{\overline{\mathbf{M}}_{n}^{\bullet}}^{\mathfrak{T}} \mathrm{inc}_{1}^{\bullet}(\mathrm{sh}^{\bullet}_{\downarrow!}\mathrm{sh}^{\bullet*}_{\uparrow}\mathbf{1}_{s_{n+1}^{\bullet}}) \cup \mathrm{inc}_{1}^{\bullet}(\mathbf{1}_{s_{n}'}). \end{split}$$

Thus, (1) follows from Proposition 5.7.6.

The theorem is proved.

s

Lemma 5.11.6. For every $f \in L[Sh(V_{n_0}^\circ, K_{n_0}^{p\circ}K_{n_0,p}^{\bullet})] \otimes_L L[Sh(V_{n_1}^\circ, K_{n_1}^{p\circ}K_{n_1,p}^{\circ})]$, we have

$$\sum_{\in \mathrm{Sh}(\mathrm{V}_n^\circ,\mathrm{K}_n^{p^\circ}\mathrm{K}_{\mathrm{sp},p}^{\bullet})} (\mathrm{T}_{n_1,\mathfrak{p}}^{\bullet\circ}f)(\mathrm{sh}_{\downarrow}^{\bullet}(s),\mathrm{sh}_{\uparrow}^{\bullet}(s)) = \sum_{s\in \mathrm{Sh}(\mathrm{V}_n^\circ,\mathrm{K}_n^{p^\circ}\mathrm{K}_{n,p}^{\circ})} (\mathrm{T}_{n_0,\mathfrak{p}}^{\circ\bullet}f)(s,\mathrm{sh}_{\uparrow}^{\circ}(s)).$$

Proof. There are two cases.

When n is even, by Lemma 5.10.8(1) and Proposition 5.10.14, we have

$$\begin{split} \sum_{s \in \operatorname{Sh}(\mathcal{V}_n^\circ, \mathcal{K}_n^{p^\circ}\mathcal{K}_{\operatorname{sp}, p}^\bullet)} (\mathcal{T}_{n_1, \mathfrak{p}}^{\bullet\circ} f)(\operatorname{sh}_{\downarrow}^{\bullet}(s), \operatorname{sh}_{\uparrow}^{\bullet}(s)) &= \sum_{s \in \operatorname{Sh}(\mathcal{V}_n^\circ, \mathcal{K}_n^{p^\circ}\mathcal{K}_{\operatorname{sp}, p}^\dagger)} f(\operatorname{sh}_{\downarrow}^{\dagger}(\operatorname{sh}_n^{\dagger\bullet}(s)), \operatorname{sh}_{n+1}^{\dagger\circ}(\operatorname{sh}_{\uparrow}^{\dagger}(s))) \\ &= \sum_{s \in \operatorname{Sh}(\mathcal{V}_n^\circ, \mathcal{K}_n^{p^\circ}\mathcal{K}_{\operatorname{sp}, p}^\dagger)} f(\operatorname{sh}_{\downarrow}^{\dagger}(\operatorname{sh}_n^{\dagger\bullet}(s)), \operatorname{sh}_{\uparrow}^{\circ}(\operatorname{sh}_n^{\dagger\circ}(\operatorname{sh}_{\downarrow}^{\dagger}(s)))), \end{split}$$

which, by Lemma 5.10.4(1), Definition 5.10.7, and Proposition 5.10.14, equals

$$\sum_{s \in \operatorname{Sh}(\operatorname{V}_n^\circ, \operatorname{K}_n^{p\circ}\operatorname{K}_{\operatorname{sp}, p}^\dagger)} f(\operatorname{sh}_n^{\dagger \bullet}(s), \operatorname{sh}_{\uparrow}^\circ(\operatorname{sh}_n^{\dagger \circ}(s))) = \sum_{s \in \operatorname{Sh}(\operatorname{V}_n^\circ, \operatorname{K}_n^{p\circ}\operatorname{K}_{n, p}^\circ)} (\operatorname{T}_{n_0, \mathfrak{p}}^{\circ \bullet} f)(s, \operatorname{sh}_{\uparrow}^\circ(s)).$$

When n is odd, by Definition 5.10.7 and Proposition 5.10.14, we have

$$\sum_{s \in \mathrm{Sh}(\mathrm{V}_{n}^{\circ},\mathrm{K}_{n}^{p\circ}\mathrm{K}_{\mathrm{sp},p}^{\bullet})} (\mathrm{T}_{n_{1},\mathfrak{p}}^{\bullet\circ}f)(\mathrm{sh}_{\downarrow}^{\bullet}(s),\mathrm{sh}_{\uparrow}^{\bullet}(s)) = \sum_{s \in \mathrm{Sh}(\mathrm{V}_{n}^{\circ},\mathrm{K}_{n}^{p\circ}\mathrm{K}_{\mathrm{sp},p}^{\dagger})} f(\mathrm{sh}_{n}^{\dagger\circ}(\mathrm{sh}_{\downarrow}^{\dagger}(s)),\mathrm{sh}_{\uparrow}^{\bullet}(\mathrm{sh}_{\mathrm{sp}}^{\dagger}(s))) = \sum_{s \in \mathrm{Sh}(\mathrm{V}_{n}^{\circ},\mathrm{K}_{n}^{p\circ}\mathrm{K}_{\mathrm{sp},p}^{\dagger})} f(\mathrm{sh}_{n}^{\dagger\circ}(\mathrm{sh}_{\downarrow}^{\dagger}(s)),\mathrm{sh}_{n+1}^{\bullet}(\mathrm{sh}_{\uparrow}^{\dagger}(s)))$$

which, by Lemma 5.10.8(2) and Proposition 5.10.14, equals

$$\sum_{s\in \mathrm{Sh}(\mathrm{V}_n^\circ,\mathrm{K}_n^{p\circ}\mathrm{K}_{n,p}^\circ)} (\mathtt{T}_{n_0,\mathfrak{p}}^{\circ\bullet}f)(s,\mathrm{sh}^\circ_\uparrow(s)).$$

The lemma is proved.

Construction 5.11.7. We constructs maps

$$\begin{cases} \operatorname{Inc}_{\circ,\dagger}^{*} \colon \operatorname{H}_{\mathfrak{T}}^{2n}(\overline{\mathbf{Q}}^{(0)}, L(n)) \to \operatorname{H}_{\mathfrak{T}}^{2n}(\overline{\mathbf{Q}}^{\circ,\bullet}, L(n)) \xrightarrow{\sigma_{!}} \operatorname{H}_{\mathfrak{T}}^{2n}(\overline{\mathbf{P}}^{\circ,\bullet}, L(n)) \\ \xrightarrow{\operatorname{inc}_{\circ,\dagger}^{*}} L[\operatorname{Sh}(\operatorname{V}_{n_{0}}^{\circ}, -_{n_{0}}\operatorname{K}_{n_{0},p}^{\circ})] \otimes_{L} L[\operatorname{Sh}(\operatorname{V}_{n_{1}}^{\circ}, -_{n_{1}}\operatorname{K}_{n_{1},p}^{\circ})], \\ \operatorname{Inc}_{\circ,\bullet}^{*} \colon \operatorname{H}_{\mathfrak{T}}^{2n}(\overline{\mathbf{Q}}^{(0)}, L(n)) \to \operatorname{H}_{\mathfrak{T}}^{2n}(\overline{\mathbf{Q}}^{\circ,\bullet}, L(n)) \xrightarrow{\sigma_{!}} \operatorname{H}_{\mathfrak{T}}^{2n}(\overline{\mathbf{P}}^{\circ,\bullet}, L(n)) \\ \xrightarrow{\operatorname{inc}_{\circ,\bullet}^{*}} L[\operatorname{Sh}(\operatorname{V}_{n_{0}}^{\circ}, -_{n_{0}}\operatorname{K}_{n_{0},p}^{\circ})] \otimes_{L} L[\operatorname{Sh}(\operatorname{V}_{n_{1}}^{\circ}, -_{n_{1}}\operatorname{K}_{n_{1},p}^{\bullet})], \\ \operatorname{Inc}_{\bullet,\dagger}^{*} \colon \operatorname{H}_{\mathfrak{T}}^{2n}(\overline{\mathbf{Q}}^{(0)}, L(n)) \to \operatorname{H}_{\mathfrak{T}}^{2n}(\overline{\mathbf{Q}}^{\bullet,\bullet}, L(n)) \xrightarrow{\sigma_{!}} \operatorname{H}_{\mathfrak{T}}^{2n}(\overline{\mathbf{P}}^{\bullet,\bullet}, L(n)) \\ \xrightarrow{\operatorname{inc}_{\bullet,\dagger}^{*}} L[\operatorname{Sh}(\operatorname{V}_{n_{0}}^{\circ}, -_{n_{0}}\operatorname{K}_{n_{0},p}^{\bullet})] \otimes_{L} L[\operatorname{Sh}(\operatorname{V}_{n_{1}}^{\circ}, -_{n_{1}}\operatorname{K}_{n_{1},p}^{\circ})], \\ \operatorname{Inc}_{\bullet,\bullet}^{*} \coloneqq \operatorname{H}_{\mathfrak{T}}^{2n}(\overline{\mathbf{Q}}^{(0)}, L(n)) \to \operatorname{H}_{\mathfrak{T}}^{2n}(\overline{\mathbf{Q}}^{\bullet,\bullet}, L(n)) \xrightarrow{\sigma_{!}} \operatorname{H}_{\mathfrak{T}}^{2n}(\overline{\mathbf{P}}^{\bullet,\bullet}, L(n)) \\ \xrightarrow{\operatorname{inc}_{\bullet,\bullet}^{*}} L[\operatorname{Sh}(\operatorname{V}_{n_{0}}^{\circ}, -_{n_{0}}\operatorname{K}_{n_{0},p}^{\bullet})] \otimes_{L} L[\operatorname{Sh}(\operatorname{V}_{n_{1}}^{\circ}, -_{n_{1}}\operatorname{K}_{n_{1},p}^{\circ})]. \end{cases}$$

Define the map

$$\nabla \colon \mathrm{H}^{2n}_{\mathfrak{T}}(\overline{\mathrm{Q}}^{(0)}, L(n)) \to L[\mathrm{Sh}(\mathrm{V}^{\circ}_{n_{0}}, -_{n_{0}}\mathrm{K}^{\circ}_{n_{0}, p})] \otimes_{L} L[\mathrm{Sh}(\mathrm{V}^{\circ}_{n_{1}}, -_{n_{1}}\mathrm{K}^{\circ}_{n_{1}, p})]$$

to be the sum of the following four maps

$$(\mathbf{I}_{n_{0},\mathfrak{p}}^{\circ}\otimes\mathbf{I}_{n_{1},\mathfrak{p}}^{\circ})\circ\mathrm{Inc}_{\circ,\dagger}^{*},\quad (p+1)^{2}(\mathbf{I}_{n_{0},\mathfrak{p}}^{\circ}\otimes\mathbf{T}_{n_{1},\mathfrak{p}}^{\circ\bullet})\circ\mathrm{Inc}_{\circ,\bullet}^{*},\\(p+1)(\mathbf{T}_{n_{0},\mathfrak{p}}^{\circ\bullet}\otimes\mathbf{I}_{n_{1},\mathfrak{p}}^{\circ})\circ\mathrm{Inc}_{\bullet,\dagger}^{*},\quad (p+1)^{3}(\mathbf{T}_{n_{0},\mathfrak{p}}^{\circ\bullet}\otimes\mathbf{T}_{n_{1},\mathfrak{p}}^{\circ\bullet})\circ\mathrm{Inc}_{\bullet,\bullet}^{*}.$$

At last, we recall the construction of potential map from [Liu19, Section 2.2]. For $r \in \mathbb{Z}$, put

$$B^{r}(\mathbf{Q},L) := \ker \left(\delta_{0}^{*} \colon \mathbf{H}_{\mathfrak{T}}^{2r}(\overline{\mathbf{Q}}^{(0)},L(r)) \to \mathbf{H}_{\mathfrak{T}}^{2r}(\overline{\mathbf{Q}}^{(1)},L(r)) \right)$$

and

$$B_r(\mathbf{Q},L) \coloneqq \operatorname{coker}\left(\delta_{1!} \colon \operatorname{H}^{2(2n-r-2)}_{\mathfrak{T}}(\overline{\mathbf{Q}}^{(1)}, L(2n-r-2)) \to \operatorname{H}^{2(2n-r-1)}_{\mathfrak{T}}(\overline{\mathbf{Q}}^{(0)}, L(2n-r-1))\right).$$

We define $B^r(\mathbf{Q}, L)^0$ and $B_{2n-r-1}(\mathbf{Q}, L)_0$ to be the kernel and the cokernel of the tautological map

$$B^r(\mathbf{Q}, L) \to B_{2n-r-1}(\mathbf{Q}, L),$$

respectively. By [Liu19, Lemma 2.4], the composite map

$$\mathrm{H}^{2(r-1)}_{\mathfrak{T}}(\overline{\mathrm{Q}}^{(0)}, L(r-1)) \xrightarrow{\delta_{0}^{*}} \mathrm{H}^{2(r-1)}_{\mathfrak{T}}(\overline{\mathrm{Q}}^{(1)}, L(r-1)) \xrightarrow{\delta_{1!}} \mathrm{H}^{2r}_{\mathfrak{T}}(\overline{\mathrm{Q}}^{(0)}, L(r))$$

factors through a unique map

$$B_{2n-r}(\mathbf{Q},L)_0 \to B^r(\mathbf{Q},L)^0$$

in $\operatorname{Fun}(\mathfrak{K}(\operatorname{V}_n^\circ)^p \times \mathfrak{K}(\operatorname{V}_{n+1}^\circ)^p, \operatorname{\mathsf{Mod}}(L[\operatorname{Gal}(\overline{\mathbb{F}}_p/\mathbb{F}_p^\Phi)]))$. Put

 $C^{r}(\mathbf{Q},L) \coloneqq B^{r}(\mathbf{Q},L)^{0}_{\mathrm{Gal}(\overline{\mathbb{F}}_{p}/\mathbb{F}_{p}^{\Phi})}, \quad C_{r}(\mathbf{Q},L) \coloneqq B_{r}(\mathbf{Q},L)^{\mathrm{Gal}(\overline{\mathbb{F}}_{p}/\mathbb{F}_{p}^{\Phi})}_{0}.$

Then we obtain the *potential map*

(5.28)
$$\Delta^r \colon C_{2n-r}(\mathbf{Q}, L) \to C^r(\mathbf{Q}, L)$$

in $\operatorname{Fun}(\mathfrak{K}(\operatorname{V}_n^\circ)^p \times \mathfrak{K}(\operatorname{V}_{n+1}^\circ)^p, \operatorname{\mathsf{Mod}}(L))$.¹³ We will be most interested in the case where r = n.

Remark 5.11.8. By the descriptions of the Galois actions in Construction 5.2.6 and Construction 5.3.6, the map ∇ in Construction 5.11.7 factors through the quotient map

$$\mathrm{H}^{2n}_{\mathfrak{T}}(\overline{\mathbf{Q}}^{(0)}, L(n)) \to \mathrm{H}^{2n}_{\mathfrak{T}}(\overline{\mathbf{Q}}^{(0)}, L(n))_{\mathrm{Gal}(\overline{\mathbb{F}}_p/\mathbb{F}_p^{\Phi})}$$

hence restricts to a map

$$\nabla \colon C^{n}(\mathbf{Q},L) \to L[\mathrm{Sh}(\mathbf{V}_{n_{0}}^{\circ},-_{n_{0}}\mathbf{K}_{n_{0},p}^{\circ})] \otimes_{L} L[\mathrm{Sh}(\mathbf{V}_{n_{1}}^{\circ},-_{n_{1}}\mathbf{K}_{n_{1},p}^{\circ})]$$

in $\operatorname{\mathsf{Fun}}(\mathfrak{K}(\operatorname{V}_n^\circ)^p \times \mathfrak{K}(\operatorname{V}_{n+1}^\circ)^p, \operatorname{\mathsf{Mod}}(L))$, via the canonical map $C^n(\operatorname{Q}, L) \to \operatorname{H}^{2n}_{\mathfrak{T}}(\overline{\operatorname{Q}}^{(0)}, L(n))_{\operatorname{Gal}(\overline{\mathbb{F}}_p/\mathbb{F}_n^\Phi)}$.

6. TATE CLASSES AND ARITHMETIC LEVEL RAISING

In this section, we study two important arithmetic properties of semistable moduli schemes introduced in Section 5. The first is the existence of Tate cycles when the rank is odd, studied in Subsection 6.2. The second is the arithmetic level raising when the rank is even, studied in Subsections 6.3 and 6.4. In Subsection 6.1, we collect some preliminaries on automorphic representations and their motives.

Let N > 2 be an integer with $r := \lfloor \frac{N}{2} \rfloor$.

6.1. Preliminaries on automorphic representations. In this subsection, we consider

- O a relevant representation Π of $\operatorname{GL}_N(\mathbb{A}_F)$ (Definition 1.1.3),
- **O** a strong coefficient field $E \subseteq \mathbb{C}$ of Π (Definition 3.2.5),
- O a finite set Σ_{\min}^+ of nonarchimedean places of F^+ containing Σ_{Π}^+ (Notation 3.1.4), O a (possibly empty) finite set Σ_{\ln}^{+14} of nonarchimedean places of F^+ that are inert in F, strongly disjoint from Σ_{\min}^+ (Definition 1.3.3),
- O a finite set Σ^+ of nonarchimedean places of F^+ containing $\Sigma^+_{\min} \cup \Sigma^+_{lr}$.

¹³In [Liu19], $C^r(\mathbf{Q}, L)$ and $C_r(\mathbf{Q}, L)$ are denoted by $A^r(\mathbf{Q}, L)^0$ and $A_r(\mathbf{Q}, L)_0$, respectively.

¹⁴Here, the subscript "lr" standards for "level-raising".
We then have, by Construction 3.1.10, the homomorphism

$$\phi_{\Pi} \colon \mathbb{T}_N^{\Sigma^+} \to O_E$$

For every prime λ of E, we have a continuous homomorphism

$$\rho_{\Pi,\lambda} \colon \Gamma_F \to \mathrm{GL}_N(E_\lambda)$$

from Proposition 3.2.4(2) and Definition 3.2.5, such that $\rho_{\Pi,\lambda}^{c}$ and $\rho_{\Pi,\lambda}^{\vee}(1-N)$ are conjugate. We choose

- O a finite place λ of E (with the underlying rational prime ℓ) satisfying $\ell \nmid ||v||(||v||^2 1)$ for every $v \in \Sigma_{lr}^+$,
- **O** a positive integer m,
- O a standard definite hermitian space V_N° of rank N over F, together with a self-dual $\prod_{v \notin \Sigma_{\min}^+ \cup \Sigma_{\ln}^+} O_{F_v}$ -lattice Λ_N° in $V_N^{\circ} \otimes_F \mathbb{A}_F^{\Sigma_{\infty}^+ \cup \Sigma_{\ln}^+}$, satisfying that $(V_N^{\circ})_v$ is not split for $v \in \Sigma_{\ln}^+$ when N is even,
- O an object $\mathcal{K}_N^{\circ} \in \mathfrak{K}(\mathcal{V}_N^{\circ})$ of the form

$$\mathbf{K}_{N}^{\circ} = \prod_{v \in \Sigma_{\min}^{+} \cup \Sigma_{\mathrm{lr}}^{+}} (\mathbf{K}_{N}^{\circ})_{v} \times \prod_{v \notin \Sigma_{\infty}^{+} \cup \Sigma_{\min}^{+} \cup \Sigma_{\mathrm{lr}}^{+}} \mathbf{U}(\Lambda_{N}^{\circ})(O_{F_{v}^{+}})_{v}$$

satisfying that when N is even, $(K_N^{\circ})_v$ is a transferable open compact subgroup of $U(V_N^{\circ})(F_v^+)$ (Definition D.2.1)¹⁵ for $v \in \Sigma_{\min}^+$ and is a special maximal subgroup of $U(V_N^{\circ})(F_v^+)$ for $v \in \Sigma_{\ln}^+$,

- O a special inert prime (Definition 3.3.4) \mathfrak{p} of F^+ (with the underlying rational prime p) satisfying
 - (P1): Σ^+ does not contain *p*-adic places;
 - (P2): ℓ does not divide $p(p^2 1)$;
 - (P3): there exists a CM type Φ containing τ_{∞} as in the initial setup of Section 5 satisfying $\mathbb{Q}_p^{\Phi} = \mathbb{Q}_{p^2};$
 - (P4): if N is even, then $P_{\alpha(\Pi_{\mathfrak{p}})} \mod \lambda^m$ is level-raising special at \mathfrak{p} (Definition 3.1.5);
 - if N is odd, then $P_{\alpha(\Pi_{\mathfrak{p}})} \mod \lambda$ is Tate generic at \mathfrak{p} (Definition 3.1.5);
 - (P5): $P_{\alpha(\Pi_p)} \mod \lambda$ is intertwining generic at \mathfrak{p} (Definition 3.1.5);
 - (P6): if N is even, the natural map

$$\frac{O_E/\lambda^m[\operatorname{Sh}(\mathcal{V}_N^\circ, \mathcal{K}_N^\circ)]}{\mathbb{T}_N^{\Sigma^+ \cup \Sigma_p^+} \cap \ker \phi_{\Pi}} \to \frac{O_E/\lambda^m[\operatorname{Sh}(\mathcal{V}_N^\circ, \mathcal{K}_N^\circ)]}{\ker \phi_{\Pi}}$$

is an isomorphism;

- (So we can and will adopt the setup in Section 5 to the datum $(V_N^{\circ}, \{\Lambda_{N,\mathfrak{q}}^{\circ}\}|_{\mathfrak{q}|p})$.)
- **O** remaining data in the initial setup of Section 5 with $\mathbb{Q}_p^{\Phi} = \mathbb{Q}_{p^2}$;
- O a definite uniformization datum as in Construction 5.3.6; and

O an indefinite uniformization datum $(V'_N, j_N, \{\Lambda'_{\mathfrak{q},N}\}_{\mathfrak{q}|p})$ for V°_N as in Definition 5.1.6.

Put $\mathcal{K}_N^{p\circ} \coloneqq (\mathcal{K}_N^{\circ})^p$ and $\mathcal{K}_N^{\bullet} \coloneqq \mathcal{K}_N^{p\circ} \times \mathcal{K}_p^{\bullet}$. Like in Subsection 5.8, we put $\mathcal{X}_N^? \coloneqq \mathcal{X}_p^?(\mathcal{V}_N^{\circ}, \mathcal{K}_N^{p\circ})$ for meaningful pairs $(\mathcal{X}, ?) \in {\mathbf{M}, \mathcal{M}, \mathcal{B}, \mathcal{S}} \times { , \eta, \circ, \bullet, \dagger}$. Let $(\mathcal{E}_s^{p,q}, \mathcal{d}_s^{p,q})$ be the weight spectral sequence abutting to the cohomology $\mathcal{H}_{\mathfrak{T}}^{p+q}(\overline{\mathcal{M}}_N, \mathcal{R}\Psi O_\lambda(r))$ from Subsection 5.8.

Remark 6.1.1. By Construction 3.1.10 and (P2) (namely, $\ell \neq p$), we know that $P_{\alpha(\Pi_{\mathfrak{p}})}$ is a polynomial with coefficients in O_{λ} .

Remark 6.1.2. Note that when N = 2, (P2) and (P4) together imply (P5).

¹⁵By Lemma D.2.2(3), every sufficiently small $(K_N^{\circ})_v$ is transferable. So the readers may ignore this technical requirement.

Notation 6.1.3. We introduce the following ideas of $\mathbb{T}_N^{\Sigma^+ \cup \Sigma_p^+}$

$$\begin{cases} \mathfrak{m} \coloneqq \mathbb{T}_{N}^{\Sigma^{+} \cup \Sigma_{p}^{+}} \cap \ker \left(\mathbb{T}_{N}^{\Sigma^{+}} \xrightarrow{\phi_{\Pi}} O_{E} \to O_{E} / \lambda \right), \\ \mathfrak{n} \coloneqq \mathbb{T}_{N}^{\Sigma^{+} \cup \Sigma_{p}^{+}} \cap \ker \left(\mathbb{T}_{N}^{\Sigma^{+}} \xrightarrow{\phi_{\Pi}} O_{E} \to O_{E} / \lambda^{m} \right). \end{cases}$$

We then introduce following assumptions.

Assumption 6.1.4. We have $\mathrm{H}^{i}_{\mathfrak{T}}(\overline{\mathrm{M}}_{N}, \mathrm{R}\Psi O_{\lambda})_{\mathfrak{m}} = 0$ for $i \neq N-1$, and that $\mathrm{H}^{N-1}_{\mathfrak{T}}(\overline{\mathrm{M}}_{N}, \mathrm{R}\Psi O_{\lambda})_{\mathfrak{m}}$ is a finite free O_{λ} -module.

Assumption 6.1.5. The Galois representation $\rho_{\Pi,\lambda}$ is residually absolutely irreducible.

Remark 6.1.6. Under Assumption 6.1.5, we obtain a homomorphism

$$\bar{\rho}_{\Pi,\lambda} \colon \Gamma_F \to \mathrm{GL}_N(O_\lambda/\lambda)$$

from the residual homomorphism of $\rho_{\Pi,\lambda}$, which is unique to to conjugation, absolutely irreducible, and (1 - N)-polarizable (Definition 2.4.7). From Construction 2.4.8 or Lemma E.1.3(3), we then have an extension

$$\bar{\rho}_{\Pi,\lambda,+} \colon \Gamma_{F^+} \to \mathscr{G}_N(O_\lambda/\lambda)$$

of $\bar{\rho}_{\Pi,\lambda}$. For a different extension $\bar{\rho}_{\Pi,\lambda,+'}$, there exist elements $g \in \operatorname{GL}_N(O_\lambda/\lambda)$ and $a \in (O_\lambda/\lambda)^{\times}$ such that $\bar{\rho}_{\Pi,\lambda,+'}(x) = g\bar{\rho}_{\Pi,\lambda,+}(x)g^{-1}$ for $x \in \Gamma_F$, and $\bar{\rho}_{\Pi,\lambda,+'}(\mathbf{c}) = (aB,\mu,\mathbf{c})$ if $\bar{\rho}_{\Pi,\lambda,+}(\mathbf{c}) = (B,\mu,\mathbf{c})$. The discussion below on the extension $\bar{\rho}_{\Pi,\lambda,+}$ is independent of such ambiguity.

We now fix an isomorphism $\iota_{\ell} \colon \mathbb{C} \simeq \overline{\mathbb{Q}}_{\ell}$ that induces the place λ of E, till the end of this section.

Definition 6.1.7. Let π be an automorphic representation of $U(V_N^\circ)(\mathbb{A}_{F^+})$. We say that π is Π -congruent (outside $\Sigma^+ \cup \Sigma_p^+$) if π_∞ is trivial, and for every nonarchimedean place v of F^+ not in $\Sigma^+ \cup \Sigma_p^+ \cup \Sigma_\ell^+$, the two homomorphisms $\iota_\ell \phi_{\alpha(\mathrm{BC}(\pi_v))}$ and $\iota_\ell \phi_{\alpha(\Pi_v)}$ from $\mathbb{T}_{N,v}$ to $\overline{\mathbb{Q}}_\ell$, which in fact take values in $\overline{\mathbb{Z}}_\ell$, coincide in $\overline{\mathbb{F}}_\ell$.

Lemma 6.1.8. The two maps

$$\begin{split} \mathbf{T}_{N,\mathfrak{p}}^{\bullet\circ} \colon O_E[\mathrm{Sh}(\mathbf{V}_N^\circ,\mathbf{K}_N^\circ)]_{\mathfrak{m}} &\to O_E[\mathrm{Sh}(\mathbf{V}_N^\circ,\mathbf{K}_N^\bullet)]_{\mathfrak{m}} \\ \mathbf{T}_{N,\mathfrak{p}}^{\circ\bullet} \colon O_E[\mathrm{Sh}(\mathbf{V}_N^\circ,\mathbf{K}_N^\bullet)]_{\mathfrak{m}} &\to O_E[\mathrm{Sh}(\mathbf{V}_N^\circ,\mathbf{K}_N^\circ)]_{\mathfrak{m}} \end{split}$$

are both isomorphisms, where $T_{N,\mathfrak{p}}^{\bullet\bullet}$ and $T_{N,\mathfrak{p}}^{\bullet\bullet}$ are introduced in Definition 5.7.1.

Proof. By Proposition B.4.3(1) (resp. Proposition B.3.5(1)) when N is odd (resp. even) and (P5), we know that the endomorphism $I_{N,\mathfrak{p}}^{\circ} = T_{N,\mathfrak{p}}^{\circ \bullet} \circ T_{N,\mathfrak{p}}^{\circ \circ}$ of $O_E[Sh(V_N^{\circ}, K_N^{\circ})]_{\mathfrak{m}}$ is an isomorphism. Thus, it suffices to show that the free O_{λ} -modules $O_E[Sh(V_N^{\circ}, K_N^{\circ})]_{\mathfrak{m}}$ and $O_E[Sh(V_N^{\circ}, K_N^{\bullet})]_{\mathfrak{m}}$ have the same rank. We show that $O_E[Sh(V_N^{\circ}, K_N^{\circ})]_{\mathfrak{m}} \otimes_{O_{\lambda}} \overline{\mathbb{Q}}_{\ell}$ and $O_E[Sh(V_N^{\circ}, K_N^{\bullet})]_{\mathfrak{m}} \otimes_{O_{\lambda}} \overline{\mathbb{Q}}_{\ell}$ have the same dimension. We have

$$O_E[\operatorname{Sh}(\operatorname{V}_N^\circ, \operatorname{K}_N^\circ)]_{\mathfrak{m}} \otimes_{O_{\lambda}} \overline{\mathbb{Q}}_{\ell} \simeq \bigoplus_{\pi} m(\pi) \cdot \pi^{\operatorname{K}_N^\circ},$$
$$O_E[\operatorname{Sh}(\operatorname{V}_N^\circ, \operatorname{K}_N^\bullet)]_{\mathfrak{m}} \otimes_{O_{\lambda}} \overline{\mathbb{Q}}_{\ell} \simeq \bigoplus_{\pi} m(\pi) \cdot \pi^{\operatorname{K}_N^\bullet},$$

where π runs over all automorphic representations of $U(V_N^{\circ})(\mathbb{A}_{F^+})$ (with coefficients in $\overline{\mathbb{Q}}_{\ell}$) that are Π -congruent; and $m(\pi)$ denotes the automorphic multiplicity of π .¹⁶ It suffices to show that if in the second direct sum $\pi_{\mathfrak{p}}^{K^{\bullet}_N} \neq \{0\}$, which has to be of dimension one since K_N^{\bullet} is special

¹⁶Although we know that $m(\pi) = 1$ by Proposition C.3.1(2), we do not need this fact here.

maximal, then $\pi_{\mathfrak{p}}^{K_N^{\circ}} \neq \{0\}$ as well. Since $\pi_{\mathfrak{p}}$ is semistable, then its Satake parameter α does not contain the pair $\{-1, -1\}$ (resp. $\{-p, -p^{-1}\}$) when N is even (resp. odd) by (P5). Let $\pi'_{\mathfrak{p}}$ be the unique constituent of the principal series of α such that $(\pi'_{\mathfrak{p}})^{K_N^{\circ}} \neq \{0\}$, then by Proposition B.4.3(1) (resp. Proposition B.3.5(1)) when N is odd (resp. even) again, we see that $(\pi'_{\mathfrak{p}})^{K_N^{\circ}} \neq \{0\}$. Thus, we must have $\pi_{\mathfrak{p}} = \pi'_{\mathfrak{p}}$ as K_N° is special maximal. The lemma follows.

Lemma 6.1.9. Let π be an automorphic representation of $U(V_N^\circ)(\mathbb{A}_{F^+})$ that is Π -congruent. If Assumption 6.1.5 holds, then π is stable.

Proof. By Proposition 3.2.8, we know that $BC(\pi)$ exists as (V_N°, π) is a relevant pair. Let $\rho_{BC(\pi),\iota_\ell} \colon \Gamma_F \to GL_N(\overline{\mathbb{Q}}_\ell)$ be the associated Galois representation from Proposition 3.2.3(2), which is the direct sum of the associated Galois representation of each isobaric factors. If $BC(\pi)$ is not cuspidal, then $\rho_{BC(\pi),\iota_\ell}$ is decomposable. Since π is Π -congruent, by the Chebotarev density theorem, $\rho_{BC(\pi),\iota_\ell}$ admits a lattice whose residual representation is isomorphic to $\bar{\rho}_{\Pi,\lambda} \otimes_{O_{\lambda/\lambda}} \overline{\mathbb{F}}_\ell$, which is irreducible. This is a contradiction. Thus, the lemma follows.

Lemma 6.1.10. Assume Assumption 6.1.5. Then the natural maps

$$\begin{split} \mathrm{H}^{i}_{\mathrm{\acute{e}t},c}(\mathrm{Sh}(\mathrm{V}'_{N},\mathbf{j}_{N}\mathrm{K}^{p\circ}_{N}\mathrm{K}'_{p,N})_{\overline{F}},O_{\lambda})_{\mathfrak{m}} &\to \mathrm{H}^{i}_{\mathrm{\acute{e}t}}(\mathrm{Sh}(\mathrm{V}'_{N},\mathbf{j}_{N}\mathrm{K}^{p\circ}_{N}\mathrm{K}'_{p,N})_{\overline{F}},O_{\lambda})_{\mathfrak{m}} \\ \mathrm{H}^{i}_{\mathfrak{T},c}(\overline{\mathrm{M}}^{\bullet}_{N},O_{\lambda})_{\mathfrak{m}} &\to \mathrm{H}^{i}_{\mathfrak{T}}(\overline{\mathrm{M}}^{\bullet}_{N},O_{\lambda})_{\mathfrak{m}}, \end{split}$$

are both isomorphisms for every $i \in \mathbb{Z}$.

Proof. By Lemma 5.1.7, and the description of the weight spectral sequence $(E_s^{p,q}, d_s^{p,q})$ in Lemma 5.8.2 (for N odd) and Lemma 5.8.3 (for N even), it suffices to show that the natural map

(6.1)
$$\mathrm{H}^{i}_{\mathrm{\acute{e}t},c}(\mathrm{Sh}(\mathrm{V}'_{N}, \mathbf{j}_{N}\mathrm{K}^{p\circ}_{N}\mathrm{K}'_{p,N})_{\overline{F}}, O_{\lambda})_{\mathfrak{m}} \to \mathrm{H}^{i}_{\mathrm{\acute{e}t}}(\mathrm{Sh}(\mathrm{V}'_{N}, \mathbf{j}_{N}\mathrm{K}^{p\circ}_{N}\mathrm{K}'_{p,N})_{\overline{F}}, O_{\lambda})_{\mathfrak{m}}$$

is an isomorphism for every $i \in \mathbb{Z}$. This is trivial when $\operatorname{Sh}(V'_N, j_N K^{p\circ}_N K'_{p,N})$ is proper.

If $\operatorname{Sh}(V'_N, \mathbf{j}_N \mathbf{K}_N^{p\circ} \mathbf{K}'_{p,N})$ is not proper, then the Witt index of \mathbf{V}'_N is 1; and $\operatorname{Sh}(\mathbf{V}'_N, \mathbf{j}_N \mathbf{K}_N^{p\circ} \mathbf{K}'_{p,N})$ has a *canonical* toroidal compactification $\widetilde{\operatorname{Sh}}(\mathbf{V}'_N, \mathbf{j}_N \mathbf{K}_N^{p\circ} \mathbf{K}'_{p,N})$, which is smooth over F. As $\mathbf{j}_N \mathbf{K}_N^{p\circ} \mathbf{K}'_{p,N}$ is neat, the boundary $Z \coloneqq \widetilde{\operatorname{Sh}}(\mathbf{V}'_N, \mathbf{j}_N \mathbf{K}_N^{p\circ} \mathbf{K}'_{p,N}) \setminus \operatorname{Sh}(\mathbf{V}'_N, \mathbf{j}_N \mathbf{K}_N^{p\circ} \mathbf{K}'_{p,N})$ is geometrically isomorphic to a disjoint union of abelian varieties (of dimension N-2). In particular, $\operatorname{H}^i_{\operatorname{\acute{e}t}}(Z_{\overline{F}}, O_{\lambda})$ is a free O_{λ} -module (of finite rank). Let π'^{∞} be an irreducible admissible representation of $\operatorname{U}(\mathbf{V}'_N)(\mathbb{A}_{F^+})$ that appears in $\operatorname{H}^i_{\operatorname{\acute{e}t}}(Z_{\overline{F}}, O_{\lambda}) \otimes_{O_{\lambda,\ell}} \mathbb{C}$. Then π'^{∞} extends to an automorphic representation π' of $\operatorname{U}(\mathbf{V}'_N)(\mathbb{A}_{F^+})$ that is a subquotient of the parabolic induction of a cuspidal automorphic representation of $L(\mathbb{A}_{F^+})$ where L is the unique proper Levi subgroup of $\operatorname{U}(\mathbf{V}'_N)$ up to conjugation. In particular, π' is not stable. Thus, by (the same argument of) Lemma 6.1.9, we have $\operatorname{H}^i_{\operatorname{\acute{e}t}}(Z_{\overline{F}}, O_{\lambda})_{\mathfrak{m}} = 0$ for every $i \in \mathbb{Z}$. This implies that (6.1) is an isomorphism. \Box

6.2. Tate classes in the odd rank case. In this section, we assume that N = 2r + 1 is odd with $r \ge 1$. We study the properties of the localized spectral sequence $E_{s,\mathfrak{m}}^{p,q}$, after Lemma 5.8.2.

Lemma 6.2.1. We have

$$\mathrm{H}^{i}_{\mathfrak{T}}(\overline{\mathrm{M}}^{\dagger}_{N}, O_{\lambda})_{\mathfrak{m}} = 0$$

for every odd integer i.

Proof. For $i \neq 2r - 1$, it follows from Lemma 5.5.2(1). Now we assume i = 2r - 1.

Let $\pi^{\infty,p}$ be an irreducible admissible representation of $U(V_N^{\circ})(\mathbb{A}_{F^+}^{\infty,p})$ that appears in the cohomology $H^{2r-1}_{\mathfrak{T}}(\overline{M}_N^{\dagger}, O_{\lambda})_{\mathfrak{m}} \otimes_{O_{\lambda,\ell}} \mathbb{C}$. By Proposition 5.5.4, we may complete $\pi^{\infty,p}$ to an automorphic representation π of $U(V_N^{\circ})(\mathbb{A}_{F^+})$ as in that proposition, such that π is Π -congruent and such that $BC(\pi_{\mathfrak{p}})$ is a constituent of an unramified principal series of $GL_N(F_{\mathfrak{p}})$, whose Satake parameter contains $\{-p, -p^{-1}\}$ which is different from $\boldsymbol{\alpha}(\Pi_{\mathfrak{p}})$ in $\overline{\mathbb{F}}_{\ell}$ by (P5).

On the other hand, by the Chebotarev density theorem, both $\rho_{BC(\pi),\iota_{\ell}}$ and $\rho_{\Pi,\lambda} \otimes_{E_{\lambda}} \mathbb{Q}_{\ell}$ each admits a lattice such that their reductions are isomorphic. However, this is not possible by Proposition C.3.1(2) and Proposition 3.2.4(2). Therefore, we must have $H_{\mathfrak{T}}^{2r-1}(\overline{M}_{N}^{\dagger}, O_{\lambda})_{\mathfrak{m}} = 0$. The lemma is proved.

Lemma 6.2.2. Assume Assumption 6.1.4. We have

- (1) $E_{1,m}^{p,q} = 0$ if q is odd;
- (2) $E_{1,\mathfrak{m}}^{p,q}$ is a free O_{λ} -module for every $(p,q) \in \mathbb{Z}^2$;
- (3) $E_{2,\mathfrak{m}}^{p,q} = 0$ unless (p,q) = (0,2r);
- (4) $E_{2,\mathfrak{m}}^{0,2r}$ is canonically isomorphic to $H_{\mathfrak{T}}^{2r}(\overline{M}_N, \mathbb{R}\Psi O_{\lambda}(r))_{\mathfrak{m}}$, which is a free O_{λ} -module;
- (5) $\mathbf{E}_{s,\mathfrak{m}}^{0,2r}$ degenerates at the second page.

Proof. Part (1) follows from Lemma 6.2.1 and Assumption 6.1.4. Part (3) follows since $d_1^{-1,2r}$ is injective and $d_1^{0,2r}$ is surjective. The remaining parts are immediate consequences of (1) and Assumption 6.1.4.

Theorem 6.2.3. The map

$$\nabla^1_{\mathfrak{m}} \colon \mathrm{E}^{0,2r}_{2,\mathfrak{m}} \to O_{\lambda}[\mathrm{Sh}(\mathrm{V}^{\circ}_N,\mathrm{K}^{\circ}_N)]_{\mathfrak{m}}$$

(Construction 5.8.4) is surjective. Moreover, if we assume Assumptions 6.1.4, 6.1.5, and Hypothesis 3.2.9 for N, then we have

- (1) The generalized Frobenius eigenvalues of the $O_{\lambda}/\lambda[\operatorname{Gal}(\overline{\mathbb{F}}_p/\mathbb{F}_{p^2})]$ -module $\operatorname{E}_{2,\mathfrak{m}}^{0,2r} \otimes_{O_{\lambda}} O_{\lambda}/\lambda$ is contained in the set of roots of $P_{\alpha(\Pi_{\mathfrak{p}})} \mod \lambda$ in a finite extension of O_{λ}/λ .
- (2) The $O_{\lambda}[\operatorname{Gal}(\overline{\mathbb{F}}_p/\mathbb{F}_{p^2})]$ -module $\operatorname{E}_{2,\mathfrak{m}}^{0,2r}$ is weakly semisimple (Definition 2.1.2).
- (3) The map $\nabla^1_{\mathfrak{m}}$ induces an isomorphism

$$\nabla^{1}_{\mathfrak{m}} \colon (\mathrm{E}^{0,2r}_{2,\mathfrak{m}})_{\mathrm{Gal}(\overline{\mathbb{F}}_{p}/\mathbb{F}_{p^{2}})} \xrightarrow{\sim} O_{\lambda}[\mathrm{Sh}(\mathrm{V}^{\circ}_{N},\mathrm{K}^{\circ}_{N})]_{\mathfrak{m}}.$$

By Remark 5.8.5, the map $\nabla^1_{\mathfrak{m}}$ always factors through the quotient map $E^{0,2r}_{2,\mathfrak{m}} \to (E^{0,2r}_{2,\mathfrak{m}})_{\operatorname{Gal}(\overline{\mathbb{F}}_p/\mathbb{F}_{p^2})}$.

Proof. We first show that $\nabla^1_{\mathfrak{m}}$ is surjective. From Construction 5.8.1, we have a map

$$(\operatorname{Inc}_{!}^{\circ}, \operatorname{Inc}_{!}^{\dagger}, \operatorname{Inc}_{!}^{\bullet} \circ \mathsf{T}_{\mathfrak{p}}^{\bullet \circ}) \coloneqq O_{\lambda}[\operatorname{Sh}(\operatorname{V}_{N}^{\circ}, \operatorname{K}_{N}^{\circ})]^{\oplus 3} \to \operatorname{E}_{1}^{0, 2r}$$

which induces a map

$$\ker\left(\mathrm{d}_{1}^{0,2r}\circ\left(\mathrm{Inc}_{!}^{\circ},\mathrm{Inc}_{!}^{\dagger},\mathrm{Inc}_{!}^{\bullet}\circ\mathsf{T}_{\mathfrak{p}}^{\bullet\circ}\right)\right)\to\ker\mathrm{d}_{1}^{0,2r}$$

However, by Lemma 5.8.6, the former kernel is simply the kernel of the map

$$\begin{pmatrix} p+1 & -1 & 0 \end{pmatrix} \begin{pmatrix} \operatorname{Inc}^*_{\mathfrak{o}} \\ \operatorname{Inc}^*_{\mathfrak{o}} \\ \operatorname{Inc}^*_{\bullet} \end{pmatrix} \begin{pmatrix} \operatorname{Inc}^\circ_! & \operatorname{Inc}^\dagger_! & \operatorname{Inc}^\bullet_! \circ \mathsf{T}^{\bullet\circ}_{\mathfrak{p}} \end{pmatrix}.$$

Now since (p+1, -1, 0) and $(0, T_{\mathfrak{p}}^{\circ \bullet} \circ T_{\mathfrak{p}}^{\circ \circ}, (p+1)^2 T_{\mathfrak{p}}^{\circ \bullet}) \otimes O_{\lambda}$ are linearly independent, by Nakayama's lemma, $\nabla_{\mathfrak{m}}^1$ is surjective if the following matrix

$$\begin{pmatrix} \operatorname{Inc}_{\circ}^{\circ} \\ \operatorname{Inc}_{\dagger}^{*} \\ \mathsf{T}_{\mathfrak{p}}^{\circ\bullet} \circ \operatorname{Inc}_{\bullet}^{*} \end{pmatrix} \left(\operatorname{Inc}_{!}^{\circ} \operatorname{Inc}_{!}^{\dagger} \operatorname{Inc}_{!}^{\bullet} \circ \mathsf{T}_{\mathfrak{p}}^{\bullet \circ} \right)$$

in $\mathbb{T}^{\circ}_{N,\mathfrak{p}}$ is nondegenerate modulo \mathfrak{m} . However, by Lemma 5.8.2(2), the above matrix equals

$$\begin{pmatrix} 1 & 0 & 0 \\ 0 & -(p+1)^2 & \mathbf{I}_{N,\mathfrak{p}}^{\circ} \\ 0 & \mathbf{I}_{N,\mathfrak{p}}^{\circ} & \mathbf{T}_{N,\mathfrak{p}}^{\circ\bullet} \circ \mathbf{T}_{N,\mathfrak{p}}^{\bullet,\mathfrak{p}} \circ \mathbf{T}_{N,\mathfrak{p}}^{\bullet\circ} \end{pmatrix},$$

whose non-degeneracy modulo **m** follows from Lemma B.4.2, Proposition B.4.3, and (P4,P5).

Now we consider the three remaining assertions. By Lemma 5.1.7 and Lemma 6.2.2, we have an isomorphism

$$\mathbf{E}_{2,\mathfrak{m}}^{0,2r} \simeq \mathbf{H}_{\mathrm{\acute{e}t}}^{2r}(\mathrm{Sh}(\mathbf{V}',\mathbf{j}_{N}\mathbf{K}_{N}^{p\circ}\mathbf{K}'_{p,N})_{\overline{F}},O_{\lambda}(r))_{\mathfrak{m}}$$

of $O_{\lambda}[\operatorname{Gal}(\overline{\mathbb{Q}}_p/\mathbb{Q}_{p^2})]$ -modules. By Lemma 6.1.10, Proposition C.3.1(2), and Hypothesis 3.2.9, we have

$$\mathrm{H}^{2r}_{\mathrm{\acute{e}t}}(\mathrm{Sh}(\mathrm{V}', \mathbf{j}_{N}\mathrm{K}^{p\circ}_{N}\mathrm{K}'_{p,N})_{\overline{F}}, O_{\lambda}(r))_{\mathfrak{m}} \otimes_{O_{\lambda}} \overline{\mathbb{Q}}_{\ell} \simeq \bigoplus_{\pi'} \rho^{\mathtt{c}}_{\mathrm{BC}(\pi'),\iota_{\ell}}(r)^{\oplus d(\pi')}$$

of representations of Γ_F with coefficients in $\overline{\mathbb{Q}}_{\ell}$, where $d(\pi') \coloneqq \dim(\pi'^{\infty,p})^{j_N K_N^{p^{\circ}}}$, and the direct sum is taken over all stable automorphic representations π' of $U(V')(\mathbb{A}_{F^+})$ that is Π -congruent and such that $\pi'_{\underline{\tau}_{\infty}}$ is a holomorphic discrete series representation of $U(V')(F_{\underline{\tau}_{\infty}}^+)$ with the Harish-Chandra parameter $\{r, r - 1, \ldots, 1 - r, -r\}$; and $\pi'_{\underline{\tau}}$ is trivial for every archimedean place $\underline{\tau} \neq \underline{\tau}_{\infty}$.

For the proof of (1–3), we may replace E_{λ} by a finite extension inside $\overline{\mathbb{Q}}_{\ell}$ such that $\rho_{\mathrm{BC}(\pi'),\iota_{\ell}}$ is defined over E_{λ} for every π' appeared in the previous direct sum. Now we regard $\rho_{\mathrm{BC}(\pi'),\iota_{\ell}}$ as a representation over E_{λ} . Then $\rho_{\mathrm{BC}(\pi'),\iota_{\ell}}(r)$ admits a Γ_F -stable O_{λ} -lattice $\mathrm{R}_{\mathrm{BC}(\pi')}$, unique up to homothety, whose reduction $\overline{\mathrm{R}}_{\mathrm{BC}(\pi')}$ is isomorphic to $\rho_{\Pi,\lambda}(r)$. Moreover, we have an inclusion

$$\mathbf{E}_{2,\mathfrak{m}}^{0,2r} \simeq \mathbf{H}_{\mathrm{\acute{e}t}}^{2r}(\mathrm{Sh}(\mathbf{V}',\mathbf{j}_{N}\mathbf{K}_{N}^{p\circ}\mathbf{K}'_{p,N})_{\overline{F}},O_{\lambda}(r))_{\mathfrak{m}} \subseteq \bigoplus_{\pi'}(\mathbf{R}_{\mathrm{BC}(\pi')}^{\mathtt{c}})^{\oplus d(\pi')}$$

of $O_{\lambda}[\operatorname{Gal}(\overline{\mathbb{F}}_p/\mathbb{F}_{p^2})]$ -modules. This already implies (1).

By (P4), we know that $\bar{\rho}_{\Pi,\lambda}^{c}(r)$ is weakly semisimple and

$$\dim_{O_{\lambda}/\lambda} \bar{\rho}_{\Pi,\lambda}^{\mathsf{c}}(r)^{\operatorname{Gal}(\overline{\mathbb{F}}_{p}/\mathbb{F}_{p^{2}})} = 1.$$

On the other hand, we have

$$\dim_{E_{\lambda}} \rho_{\mathrm{BC}(\pi'),\iota_{\ell}}^{\mathsf{c}}(r)^{\mathrm{Gal}(\overline{\mathbb{F}}_{p}/\mathbb{F}_{p^{2}})} \geq 1.$$

Thus by Lemma 2.1.5, for every π' in the previous direct sum, $R^{c}_{BC(\pi')}$ is weakly semisimple, and

$$\dim_{E_{\lambda}} \rho_{\mathrm{BC}(\pi'),\iota_{\ell}}^{\mathsf{c}}(r)^{\mathrm{Gal}(\overline{\mathbb{F}}_{p}/\mathbb{F}_{p^{2}})} = 1$$

This implies (2) by Lemma 2.1.4(1).

The above discussion also implies that, for (3), it suffices to show

$$\sum_{\pi'} d(\pi') \leq \dim_{E_{\lambda}} O_{\lambda}[\operatorname{Sh}(\operatorname{V}_{N}^{\circ}, \operatorname{K}_{N}^{\circ})]_{\mathfrak{m}} \otimes_{O_{\lambda}} E_{\lambda}$$

where π' is taken over the same set as in the previous direct sum. However, this follows from Corollary C.3.3 and Lemma 6.1.8. The theorem is proved.

6.3. Arithmetic level raising in the even rank case. In this subsection, we assume that N = 2r is even with $r \ge 1$. We study the properties of the localized spectral sequence $\mathbb{E}_{s,\mathfrak{m}}^{p,q}$, after Lemma 5.8.3.

Proposition 6.3.1. Suppose $N \ge 4$. Assume Assumptions 6.1.4, 6.1.5, and Hypothesis 3.2.9 for N. Then we have

(1) The map

$$(\operatorname{Inc}_{!}^{\circ} + \operatorname{Inc}_{!}^{\dagger} + \operatorname{Inc}_{!}^{\bullet})_{\mathfrak{m}} \colon O_{\lambda}[\operatorname{Sh}(\operatorname{V}_{N}^{\circ}, \operatorname{K}_{N}^{\circ})]_{\mathfrak{m}}^{\oplus 2} \bigoplus O_{\lambda}[\operatorname{Sh}(\operatorname{V}_{N}^{\circ}, \operatorname{K}_{N}^{\bullet})]_{\mathfrak{m}} \to \operatorname{E}_{1,\mathfrak{m}}^{0,2r-2}(-1)$$

from Construction 5.8.1 is an isomorphism.

(2) The map

$$(\mathrm{Inc}^*_{\circ}, \mathrm{Inc}^*_{\dagger}, \mathrm{Inc}^*_{\bullet})_{\mathfrak{m}} \colon \mathrm{E}^{0,2r}_{1,\mathfrak{m}} \to O_{\lambda}[\mathrm{Sh}(\mathrm{V}^{\circ}_{N}, \mathrm{K}^{\circ}_{N})]_{\mathfrak{m}}^{\oplus 2} \bigoplus O_{\lambda}[\mathrm{Sh}(\mathrm{V}^{\circ}_{N}, \mathrm{K}^{\bullet}_{N})]_{\mathfrak{m}}$$

from Construction 5.8.1 is surjective, whose kernel is the O_{λ} -torsion of $\mathrm{H}_{\mathfrak{T}}^{2r}(\overline{\mathrm{M}}_{N}^{\bullet},O_{\lambda}(r))_{\mathfrak{m}}$.

- (3) The map $\nabla^0_{\mathfrak{m}} \colon \mathrm{d}^{0,2r}_{1,\mathfrak{m}} \to O_{\lambda}[\mathrm{Sh}(\mathrm{V}^{\circ}_{N},\mathrm{K}^{\circ}_{N})]_{\mathfrak{m}}$ (Construction 5.8.4) is surjective. (4) The map $\nabla^0_{\mathfrak{m}} \circ \mathrm{d}^{-1,2r}_{1,\mathfrak{m}} \circ \mathrm{d}^{0,2r-2}_{1,\mathfrak{m}}(-1)$ induces a map

$$\mathbf{F}_{-1}\mathbf{H}^{1}(\mathbf{I}_{\mathbb{Q}_{p^{2}}},\mathbf{H}_{\mathfrak{T}}^{2r-1}(\overline{\mathbf{M}}_{N},\mathbf{R}\Psi O_{\lambda}(r))_{\mathfrak{m}})\to O_{\lambda}[\mathrm{Sh}(\mathbf{V}_{N}^{\circ},\mathbf{K}_{N}^{\circ})]_{\mathfrak{m}}/((p+1)\mathbf{R}_{N,\mathfrak{p}}^{\circ}-\mathbf{I}_{N,\mathfrak{p}}^{\circ})$$

which is surjective, whose kernel is canonically the O_{λ} -torsion of $\mathrm{H}_{\mathfrak{T}}^{2r}(\overline{\mathrm{M}}_{N}^{\bullet},O_{\lambda}(r))_{\mathfrak{m}}$.

Proof. We first claim that the map

$$(\mathrm{inc}_{!}^{\dagger} + \mathrm{inc}_{!}^{\bullet} \circ \mathsf{T}_{N,\mathfrak{p}}^{\bullet\circ})_{\mathfrak{m}} \colon O_{\lambda}[\mathrm{Sh}(\mathrm{V}_{N}^{\circ}, \mathrm{K}_{N}^{\circ})]_{\mathfrak{m}}^{\oplus 2} \to \mathrm{H}_{\mathfrak{T}}^{2r-2}(\overline{\mathrm{M}}_{N}^{\bullet}, O_{\lambda}(r-1))_{\mathfrak{m}}$$

is an isomorphism. In fact, by Lemma 6.3.2 below, it suffices to find a line bundle \mathcal{L} as in Definition 5.7.7 such that $(inc_{\mathcal{L}})_{\mathfrak{m}}$ is surjective, where

$$\operatorname{inc}_{\mathcal{L}} \coloneqq (\operatorname{inc}_{\dagger}^{*}, \mathsf{T}_{N, \mathfrak{p}}^{\circ \bullet} \circ \operatorname{inc}_{\bullet}^{*}) \circ \Theta_{\mathcal{L}} \circ (\operatorname{inc}_{!}^{\dagger} + \operatorname{inc}_{!}^{\bullet} \circ \mathsf{T}_{N, \mathfrak{p}}^{\bullet \circ}).$$

We take \mathcal{L} to be $\mathcal{O}(\mathcal{M}_N^{\dagger})^{\otimes 2} \otimes (\operatorname{Lie}_{\mathcal{A},\tau_{\infty}^{c}})^{\otimes p+1}$. Then by Proposition 5.7.8 and Proposition 5.7.9, the endomorphism $inc_{\mathcal{L}}$ is given the matrix

$$\begin{pmatrix} (p+1)^3 & (p+1)\mathbf{I}_{N,\mathfrak{p}}^{\circ} \\ (p+1)\mathbf{I}_{N,\mathfrak{p}}^{\circ} & \mathbf{T}_{N,\mathfrak{p}}^{\circ\bullet} \circ (\mathbf{R}_{N,\mathfrak{p}}^{\bullet} + (\mathbf{R}_{N,\mathfrak{p}}^{\bullet} + (p+1)\mathbf{T}_{N,\mathfrak{p}}^{\bullet})) \circ \mathbf{T}_{N,\mathfrak{p}}^{\circ\circ} \end{pmatrix}$$

in $\mathbb{T}^{\circ}_{N,\mathfrak{p}}$. Now, by Lemma B.3.6 and Proposition B.3.5, the determinant of the above matrix mod \mathfrak{m} is equal to

$$-p^{r^{2}}\prod_{i=1}^{r}\left(\alpha_{i}+\frac{1}{\alpha_{i}}+2\right)\cdot\left((p+1)^{2}p^{r^{2}}\prod_{i=1}^{r}\left(\alpha_{i}+\frac{1}{\alpha_{i}}-p-\frac{1}{p}\right)+(p+1)^{3}\left(p^{r^{2}+1}-p^{r^{2}-1}\right)\sum_{j=1}^{r}\prod_{\substack{i=1\\i\neq j}}^{r}\left(\alpha_{i}+\frac{1}{\alpha_{i}}-p-\frac{1}{p}\right)\right)$$

where $\{\alpha_r, \ldots, \alpha_1, \alpha_1^{-1}, \ldots, \alpha_r^{-1}\}$ are the roots of $P_{\alpha(\Pi_{N,\mathfrak{p}})} \mod \lambda$ in a finite extension of O_{λ}/λ . By (P2), we have

$$p^{r^2}(p+1)^3 \left(p^{r^2+1} - p^{r^2-1}\right) \not\equiv 0 \mod \lambda;$$

by (P4), we have

$$\prod_{i=1}^{r} \left(\alpha_i + \frac{1}{\alpha_i} - p - \frac{1}{p} \right) \equiv 0 \mod \lambda, \quad \sum_{j=1}^{r} \prod_{\substack{i=1\\i \neq j}}^{r} \left(\alpha_i + \frac{1}{\alpha_i} - p - \frac{1}{p} \right) \not\equiv 0 \mod \lambda;$$

and by (P5), we have

$$\prod_{i=1}^{r} \left(\alpha_i + \frac{1}{\alpha_i} + 2 \right) \not\equiv 0 \mod \lambda.$$

In particular, the matrix representing $inc_{\mathcal{L}}$ is nondegenerate modulo \mathfrak{m} ; hence the claim follows from Nakayama's lemma.

Part (1) follows immediately from the above claim and Lemma 6.1.8. Part (2) follows from (1) by Poincaré duality, together with Lemma 6.1.10.

For (3), by definition, $\Delta^0_{\mathfrak{m}}$ is the composition of

$$(\mathsf{T}^{\bullet\circ}_{N,\mathfrak{p}}\mathrm{Inc}^*_{\circ}, \mathsf{T}^{\bullet\circ}_{N,\mathfrak{p}} \circ \mathrm{Inc}^*_{\dagger}, \mathrm{Inc}^*_{\bullet})_{\mathfrak{m}} \colon \mathrm{E}^{0,2r}_{1,\mathfrak{m}} \to O_{\lambda}[\mathrm{Sh}(\mathrm{V}^{\circ}_{N}, \mathrm{K}^{\bullet}_{N})]^{\oplus 3}_{\mathfrak{m}},$$

which is surjective by (2) and Lemma 6.1.8, and the obviously surjective map

$$(1,0,p+1)\colon O_{\lambda}[\operatorname{Sh}(\operatorname{V}_{N}^{\circ},\operatorname{K}_{N}^{\bullet})]_{\mathfrak{m}}^{\oplus 3} \to O_{\lambda}[\operatorname{Sh}(\operatorname{V}_{N}^{\circ},\operatorname{K}_{N}^{\bullet})]_{\mathfrak{m}}.$$

Thus, (3) follows.

Now we consider (4). Let $(E_{1,\mathfrak{m}}^{0,2r})_{\mathrm{fr}}$ be the free O_{λ} -quotient of $E_{1,\mathfrak{m}}^{0,2r}$, which is simply the quotient by the O_{λ} -torsion $(\mathrm{H}_{\mathfrak{T}}^{2r}(\overline{\mathrm{M}}_{N}^{\bullet},O_{\lambda}(r))_{\mathfrak{m}})_{\mathrm{tor}}$ of $\mathrm{H}_{\mathfrak{T}}^{2r}(\overline{\mathrm{M}}_{N}^{\bullet},O_{\lambda}(r))_{\mathfrak{m}}$. Thus by (2), we obtain an isomorphism $(\mathrm{Inc}_{\circ}^{*},\mathrm{Inc}_{\dagger}^{*},\mathrm{Inc}_{\bullet}^{*})_{\mathfrak{m}} \colon (E_{1,\mathfrak{m}}^{0,2r})_{\mathrm{fr}} \xrightarrow{\sim} O_{\lambda}[\mathrm{Sh}(\mathrm{V}_{N}^{\circ},\mathrm{K}_{N}^{\circ})]_{\mathfrak{m}}^{\oplus 2} \bigoplus O_{\lambda}[\mathrm{Sh}(\mathrm{V}_{N}^{\circ},\mathrm{K}_{N}^{\bullet})]_{\mathfrak{m}}$

through which we identify the two sides. If we let $(\ker d_{1,\mathfrak{m}}^{0,2r})_{\mathrm{fr}}$ be the free O_{λ} -quotient of $\ker d_{1,\mathfrak{m}}^{0,2r}$, then by Lemma 5.8.6, the above isomorphism maps the submodule $(\ker d_{1,\mathfrak{m}}^{0,2r})_{\mathrm{fr}}$ to the kernel of the map

$$(p+1,-1,0)\colon O_{\lambda}[\operatorname{Sh}(\operatorname{V}_{N}^{\circ},\operatorname{K}_{N}^{\circ})]_{\mathfrak{m}}^{\oplus 2} \bigoplus O_{\lambda}[\operatorname{Sh}(\operatorname{V}_{N}^{\circ},\operatorname{K}_{N}^{\bullet})]_{\mathfrak{m}} \to O_{\lambda}[\operatorname{Sh}(\operatorname{V}_{N}^{\circ},\operatorname{K}_{N}^{\circ})]_{\mathfrak{m}}$$

By Assumption 6.1.4, we have $\operatorname{im} d_{1,\mathfrak{m}}^{-1,2r} = \ker d_{1,\mathfrak{m}}^{0,2r}$. Combining Lemma 5.8.3(5), we see that the map $d_{1,\mathfrak{m}}^{-1,2r}$ induces a canonical isomorphism

$$F_{-1}H^{1}(I_{\mathbb{Q}_{p^{2}}}, H_{\mathfrak{T}}^{2r-1}(\overline{M}_{N}, \mathbb{R}\Psi O_{\lambda}(r))_{\mathfrak{m}}) \simeq \frac{\operatorname{im} d_{1,\mathfrak{m}}^{-1,2r}}{\operatorname{im}(d_{1,\mathfrak{m}}^{-1,2r} \circ d_{1,\mathfrak{m}}^{0,2r-2}(-1))} = \frac{\operatorname{ker} d_{1,\mathfrak{m}}^{0,2r}}{\operatorname{im}(d_{1,\mathfrak{m}}^{-1,2r} \circ d_{1,\mathfrak{m}}^{0,2r-2}(-1))}$$

induced by $d_{1,\mathfrak{m}}^{-1,2r}$. Thus, we have a canonical surjective map

$$\mathbf{F}_{-1}\mathbf{H}^{1}(\mathbf{I}_{\mathbb{Q}_{p^{2}}},\mathbf{H}_{\mathfrak{T}}^{2r-1}(\overline{\mathbf{M}}_{N},\mathbf{R}\Psi O_{\lambda}(r))_{\mathfrak{m}}) \to \frac{(\ker \mathbf{d}_{1,\mathfrak{m}}^{0,2r})_{\mathrm{fr}}}{\mathrm{im}(\mathbf{d}_{1,\mathfrak{m}}^{-1,2r} \circ \mathbf{d}_{1,\mathfrak{m}}^{0,2r-2}(-1))}$$

whose kernel is

$$\frac{(\mathrm{H}_{\mathfrak{T}}^{2r}(\mathrm{M}_{N}^{\bullet},O_{\lambda}(r))_{\mathfrak{m}})_{\mathrm{tor}}}{(\mathrm{H}_{\mathfrak{T}}^{2r}(\overline{\mathrm{M}}_{N}^{\bullet},O_{\lambda}(r))_{\mathfrak{m}})_{\mathrm{tor}}\cap\mathrm{im}(\mathrm{d}_{1,\mathfrak{m}}^{-1,2r}\circ\mathrm{d}_{1,\mathfrak{m}}^{0,2r-2}(-1))}.$$

By Lemma 6.1.8 and Lemma 5.8.3(7), we see that $(\ker d_{1,\mathfrak{m}}^{0,2r})_{\mathrm{fr}} \cap \ker \nabla_{\mathfrak{m}}^{0}$ is contained in the image $d_{1,\mathfrak{m}}^{-1,2r} \circ d_{1,\mathfrak{m}}^{0,2r-2}(-1)$, as modules of $(\mathrm{E}_{1,\mathfrak{m}}^{0,2r})_{\mathrm{fr}}$. Thus, by (3), the map $\nabla_{\mathfrak{m}}^{0}$ induces an isomorphism

$$\frac{(\ker \mathrm{d}_{1,\mathfrak{m}}^{0,2r})_{\mathrm{fr}}}{\mathrm{im}(\mathrm{d}_{1,\mathfrak{m}}^{-1,2r} \circ \mathrm{d}_{1,\mathfrak{m}}^{0,2r-2}(-1))} \xrightarrow{\sim} \frac{O_{\lambda}[\mathrm{Sh}(\mathrm{V}_{N}^{\circ},\mathrm{K}_{N}^{\circ})]_{\mathfrak{m}}}{\mathrm{im}(\nabla_{\mathfrak{m}}^{0} \circ \mathrm{d}_{1,\mathfrak{m}}^{-1,2r} \circ \mathrm{d}_{1,\mathfrak{m}}^{0,2r-2}(-1))}$$

By Lemma 5.8.3(8), $\operatorname{im}(\nabla^0_{\mathfrak{m}} \circ d_{1,\mathfrak{m}}^{-1,2r} \circ d_{1,\mathfrak{m}}^{0,2r-2}(-1))$ coincides with the submodule

$$\left(\mathsf{T}_{N,\mathfrak{p}}^{\circ\bullet}\circ((p+1)\mathsf{R}_{N,\mathfrak{p}}^{\bullet}-\mathsf{T}_{N,\mathfrak{p}}^{\circ\circ}\circ\mathsf{T}_{N,\mathfrak{p}}^{\circ\circ})\circ\mathsf{T}_{N,\mathfrak{p}}^{\circ\circ}\right).O_{\lambda}[\mathrm{Sh}(\mathsf{V}_{N}^{\circ},\mathsf{K}_{N}^{\circ})]_{\mathfrak{m}}$$

Note that, by Lemma B.3.6, we have

$$\mathbf{T}_{N,\mathfrak{p}}^{\circ\bullet} \circ \left((p+1)\mathbf{R}_{N,\mathfrak{p}}^{\bullet} - \mathbf{T}_{N,\mathfrak{p}}^{\circ\circ} \circ \mathbf{T}_{N,\mathfrak{p}}^{\circ\circ}\right) \circ \mathbf{T}_{N,\mathfrak{p}}^{\circ\circ} = \mathbf{I}_{N,\mathfrak{p}}^{\circ} \left((p+1)\mathbf{R}_{N,\mathfrak{p}}^{\circ} - \mathbf{I}_{N,\mathfrak{p}}^{\circ}\right).$$

Thus, to conclude (4), it remains to show that

(6.2)
$$(\mathrm{H}^{2r}_{\mathfrak{T}}(\overline{\mathrm{M}}^{\bullet}_{N}, O_{\lambda}(r))_{\mathfrak{m}})_{\mathrm{tor}} \cap \mathrm{im}(\mathrm{d}^{-1,2r}_{1,\mathfrak{m}} \circ \mathrm{d}^{0,2r-2}_{1,\mathfrak{m}}(-1)) = 0$$

By Lemma 5.1.7, Hypothesis 3.2.9, and Proposition C.3.1(2), we know that the $\overline{\mathbb{Q}}_{\ell}[\Gamma_F]$ -module $\mathrm{H}^{2r-1}_{\mathfrak{T}}(\overline{\mathrm{M}}_N, \mathrm{R}\Psi O_{\lambda}(r))_{\mathfrak{m}} \otimes_{O_{\lambda}} \overline{\mathbb{Q}}_{\lambda}$ is isomorphic to a direct sum of $\rho_{\Pi',\iota_{\ell}}(r)$ for some relevant representations Π' of $\mathrm{GL}_N(\mathbb{A}_F)$. By Proposition 3.2.4(1) and [TY07, Lemma 1.4(3)], we know that $\rho_{\Pi',\iota_{\ell}}(r)$ is pure of weight -1 at \mathfrak{p} (Definition 2.4.4). In particular, we have $\mathrm{H}^1(\mathbb{Q}_{p^2},\rho_{\Pi',\iota_{\ell}}(r)) = 0$ by [Nek07, Proposition 4.2.2(1)], hence that both sides of the inclusion

$$\mathbf{F}_{-1}\mathbf{H}^{1}(\mathbf{I}_{\mathbb{Q}_{p^{2}}},\mathbf{H}_{\mathfrak{T}}^{2r-1}(\overline{\mathbf{M}}_{N},\mathbf{R}\Psi O_{\lambda}(r))_{\mathfrak{m}}) \subseteq \mathbf{H}^{1}_{\operatorname{sing}}(\mathbb{Q}_{p^{2}},\mathbf{H}_{\mathfrak{T}}^{2r-1}(\overline{\mathbf{M}}_{N},\mathbf{R}\Psi O_{\lambda}(r))_{\mathfrak{m}})$$

are torsion O_{λ} -modules. Thus, the O_{λ} -rank of $\operatorname{im}(d_{1,\mathfrak{m}}^{-1,2r} \circ d_{1,\mathfrak{m}}^{0,2r-2}(-1))$ is equal to the O_{λ} -rank of $\operatorname{ker} d_{1,\mathfrak{m}}^{0,2r}$, which in turn is equal to the sum of O_{λ} -ranks of $O_{\lambda}[\operatorname{Sh}(\operatorname{V}_{N}^{\circ},\operatorname{K}_{N}^{\circ})]_{\mathfrak{m}}$ and $O_{\lambda}[\operatorname{Sh}(\operatorname{V}_{N}^{\circ},\operatorname{K}_{N}^{\bullet})]_{\mathfrak{m}}$. However, the source of the map $d_{1,\mathfrak{m}}^{-1,2r} \circ d_{1,\mathfrak{m}}^{0,2r-2}(-1)$, which is $\operatorname{E}_{1,\mathfrak{m}}^{0,2r-2}/\operatorname{im} d_{1,\mathfrak{m}}^{-1,2r}$, is also a free O_{λ} -module of the same rank. Therefore, we must have (6.2). Part (4) is proved. **Lemma 6.3.2.** Suppose $N \ge 4$. Assume Assumptions 6.1.4, 6.1.5, and Hypothesis 3.2.9 for N. Then $\mathrm{H}^{2r-2}_{\mathfrak{T}}(\overline{\mathrm{M}}^{\bullet}_{N}, O_{\lambda})_{\mathfrak{m}}$ is a free O_{λ} -module; and its rank over O_{λ} is at most twice the rank of the (free) O_{λ} -module $O_{\lambda}[Sh(V_{N}^{\circ}, K_{N}^{\circ})]_{\mathfrak{m}}$.

Proof. By Assumption 6.1.4, Lemma 5.8.3(2), and Lemma 5.5.2(2), we have an injective map

$$\mathrm{H}^{2r-2}_{\mathfrak{T}}(\overline{\mathrm{M}}^{\bullet}_{N},O_{\lambda})_{\mathfrak{m}} \hookrightarrow \mathrm{H}^{2r-2}_{\mathfrak{T}}(\overline{\mathrm{M}}^{\dagger}_{N},O_{\lambda})_{\mathfrak{m}}$$

induced by $d_1^{0,2r-2}$. For the target, we have an isomorphism

$$\mathrm{H}^{2r-2}_{\mathfrak{T}}(\overline{\mathrm{M}}_{N}^{\dagger},O_{\lambda})_{\mathfrak{m}} \simeq O_{\lambda}[\mathrm{Sh}(\mathrm{V}_{N}^{\circ},\mathrm{K}_{N}^{\circ})]_{\mathfrak{m}} \oplus \mathrm{H}^{\mathrm{prim}}(\overline{\mathrm{M}}_{N}^{\dagger},O_{\lambda})_{\mathfrak{m}}.$$

In particular, $\mathrm{H}^{2r-2}_{\mathfrak{T}}(\overline{\mathrm{M}}^{\dagger}_{N}, O_{\lambda})_{\mathfrak{m}}$ hence $\mathrm{H}^{2r-2}_{\mathfrak{T}}(\overline{\mathrm{M}}^{\bullet}_{N}, O_{\lambda})_{\mathfrak{m}}$ are free O_{λ} -modules. Let $\pi^{\infty, p}$ be an irreducible admissible representation of $\mathrm{U}(\mathrm{V}^{\circ}_{N})(\mathbb{A}^{\infty, p}_{F^{+}})$ that appears in $\mathrm{H}^{2r-2}_{\mathfrak{T}}(\overline{\mathrm{M}}^{\bullet}_{N}, O_{\lambda})_{\mathfrak{m}} \otimes_{O_{\lambda}, \iota} \mathbb{C}$. Then, by Proposition 5.5.4, one can complete $\pi^{\infty, p}$ to an automorphic representation $\pi = \pi^{\infty,p} \otimes \pi_{\infty} \otimes \prod_{\mathfrak{q}|p} \pi_{\mathfrak{q}}$ such that π_{∞} is trivial; $\pi_{\mathfrak{q}}$ is unramified for $\mathfrak{q} \neq \mathfrak{p}$; and $\pi_{\mathfrak{p}}$ is a constituent of an unramified principal series. Moreover, π is Π -congruent. By Assumption 6.1.5 and Lemma 6.1.9, we know that π is stable.

To prove the lemma, it suffices to show that for such π as above, we have

(6.3)
$$\dim_{\overline{\mathbb{Q}}_{\ell}} \mathrm{H}^{2r-2}_{\mathfrak{T}}(\overline{\mathrm{M}}^{\bullet}_{N}, \overline{\mathbb{Q}}_{\ell})[\iota_{\ell}\pi^{\infty}] \leq 2 \dim_{\overline{\mathbb{Q}}_{\ell}} \overline{\mathbb{Q}}_{\ell}[\mathrm{Sh}(\mathrm{V}^{\circ}_{N}, \mathrm{K}^{\circ}_{N})][\iota_{\ell}\pi^{\infty}].$$

Note that as in the proof of Proposition 5.5.4, we have an isomorphism

(6.4)
$$\iota_{\ell}^{-1}\mathrm{H}^{\mathrm{prim}}(\overline{\mathrm{M}}_{N}^{\dagger},\overline{\mathbb{Q}}_{\ell})\simeq\mathrm{Map}_{\mathrm{K}_{N,\mathfrak{p}}^{\circ}}\left(\mathrm{U}(\mathrm{V}_{N}^{\circ})(F^{+})\backslash\mathrm{U}(\mathrm{V}_{N}^{\circ})(\mathbb{A}_{F^{+}}^{\infty})/\mathrm{K}_{N}^{p\circ}\prod_{\mathfrak{q}\mid p,\mathfrak{q\neq\mathfrak{p}}}\mathrm{K}_{N,\mathfrak{q}}^{\circ},\Omega_{N}\right)$$

By Proposition C.3.1(2), we have $BC(\pi_p) \simeq BC(\pi)_p$. Let $\rho_{BC(\pi),\iota_\ell} \colon \Gamma_F \to GL_N(\overline{\mathbb{Q}}_\ell)$ be the associated Galois representation. Since π is Π -congruent, by the Chebotarev density theorem, $\rho_{\mathrm{BC}(\pi),\iota_{\ell}}$ admits a lattice whose residual representation is isomorphic to $\bar{\rho}_{\Pi,\lambda} \otimes_{O_{\lambda}/\lambda} \overline{\mathbb{F}}_{\ell}$, which is irreducible by Assumption 6.1.5. Thus, by Proposition 3.2.4(2), $\alpha(BC(\pi_{\mathfrak{p}}))$ does not contain $\{-1, -1\}$ due to (P5) and contains $\{p, p^{-1}\}$ with multiplicity at most one by (P4). We now have three cases.

Case 1: $\pi_{\mathfrak{p}}$ is unramified. Then (6.3) follows by (6.4) and the fact that the multiplicity of Ω_N in $\pi_{\mathfrak{p}}|_{\mathbf{K}_{N,\mathfrak{p}}^{\circ}}$ is at most 1 by Proposition C.2.1(2).

Case 2: $\pi_{\mathfrak{p}}$ is not unramified and $\pi_{\mathfrak{p}} \notin \mathcal{S}$. Then by Lemma C.2.3(1), $\pi_{\mathfrak{p}}|_{K_{N_{\mathfrak{p}}}^{\circ}}$ does not contain Ω_N . Thus, both sides of (6.3) are zero by (6.4).

Case 3: $\pi_{\mathfrak{p}}$ belongs to \mathcal{S} . Then we have $\overline{\mathbb{Q}}_{\ell}[\operatorname{Sh}(V_{N}^{\circ}, K_{N}^{\circ})][\iota_{\ell}\pi^{\infty}] = 0$, hence an inclusion

$$(6.5) \quad \iota_{\ell}^{-1} \mathrm{H}^{2r-2}_{\mathfrak{T}}(\overline{\mathrm{M}}^{\bullet}_{N}, \overline{\mathbb{Q}}_{\ell})[\pi^{\infty}] \hookrightarrow \mathrm{Map}_{\mathrm{K}^{\circ}_{N,\mathfrak{p}}} \left(\mathrm{U}(\mathrm{V}^{\circ}_{N})(F^{+}) \backslash \mathrm{U}(\mathrm{V}^{\circ}_{N})(\mathbb{A}^{\infty}_{F^{+}}) / \mathrm{K}^{p\circ}_{N} \prod_{\mathfrak{q} \mid p, \mathfrak{q} \neq \mathfrak{p}} \mathrm{K}^{\circ}_{N,\mathfrak{q}}, \Omega_{N} \right) [\pi^{\infty}]$$

by (6.4). Note that, by Proposition C.2.1(2), the multiplicity of Ω_N in $\pi_{\mathfrak{p}}|_{\mathcal{K}^{\circ}_{N,\mathfrak{p}}}$ is one, hence we have

$$\operatorname{Map}_{\mathcal{K}_{N,\mathfrak{p}}^{\circ}}\left(\mathcal{U}(\mathcal{V}_{N}^{\circ})(F^{+})\backslash\mathcal{U}(\mathcal{V}_{N}^{\circ})(\mathbb{A}_{F^{+}}^{\infty})/\mathcal{K}_{N}^{p\circ}\prod_{\mathfrak{q}\mid p,\mathfrak{q}\neq\mathfrak{p}}\mathcal{K}_{N,\mathfrak{q}}^{\circ},\Omega_{N}\right)[\pi^{\infty}]\simeq(\pi^{\infty,p})^{\mathcal{K}_{N}^{p\circ}}$$

by Proposition C.3.1(2).

On the other hand, by Lemma 6.1.10, Proposition C.3.1(2), Corollary C.3.2, and Hypothesis 3.2.9, we know that the $\mathbb{Q}_{\ell}[\Gamma_F]$ -module

$$\mathrm{H}^{2r-1}_{\mathrm{\acute{e}t}}(\mathrm{Sh}(\mathrm{V}'_{N}, \mathbf{j}_{N}\mathrm{K}^{p\circ}_{N}\mathrm{K}'_{p,N})_{\overline{F}}, \overline{\mathbb{Q}}_{\ell})[\iota_{\ell}\pi^{\infty, p}]$$

is isomorphic to $\dim(\pi^{\infty,p})^{K_N^{p\circ}}$ copies of $\rho_{\mathrm{BC}(\pi),\iota_\ell}^{\mathsf{c}}$. By Proposition 3.2.4(2), we know that $\rho_{BC(\pi),\iota_{\ell}}^{c}|_{Gal(\overline{\mathbb{Q}}_{p}/\mathbb{Q}_{n^{2}})}$ has nontrivial monodromy action. Thus, by Lemma 5.1.7 and the spectral sequence $E_s^{p,q}$, the cokernel of (6.5) has dimension $\dim(\pi^{\infty,p})^{K_N^{p\circ}}$, which forces the target of (6.5) to vanish. In particular, (6.3) holds.

The lemma is proved.

Remark 6.3.3. Following the well-known computation of level raising of Shimura curves (see, for example, Step 5 of the proof of [Liu19, Proposition 3.32], we know that Proposition 6.3.1(4) also holds when N = 2. Moreover, as M_2^{\bullet} is a disjoint union of projective lines, the kernel of the map is trivial hence the map is an isomorphism.

Recall that we have fixed a positive integer m at the beginning of Subsection 6.1, and introduced the ideal \mathfrak{n} in Notation 6.1.3.

Theorem 6.3.4. Assume Assumptions 6.1.4, 6.1.5, and Hypothesis 3.2.9 for N. Moreover, if $N \geq 4$, we further assume that

- (a) $\ell \geq 2(N+1)$ and ℓ is unramified in F;
- (b) $\bar{\rho}_{\Pi,\lambda,+}$ (Remark 6.1.5) is rigid for $(\Sigma_{\min}^+, \Sigma_{\ln}^+)$ (Definition E.7.1), and $\bar{\rho}_{\Pi,\lambda}|_{\mathrm{Gal}(\overline{F}/F(\zeta_{\ell}))}$ is absolutely irreducible;
- (c) the composite homomorphism $\mathbb{T}_N^{\Sigma^+} \xrightarrow{\phi_{\Pi}} O_E \to O_E/\lambda$ is cohomologically generic (Definition D.1.1; and
- (d) $O_{\lambda}[Sh(V_N^{\circ}, K_N^{\circ})]_{\mathfrak{m}}$ is nontrivial.

Then we have

- (1) $\operatorname{H}^{i}_{\mathfrak{T}}(\overline{\operatorname{M}}^{\bullet}_{N}, O_{\lambda})_{\mathfrak{m}}$ is a free O_{λ} -module for every $i \in \mathbb{Z}$. (2) $\operatorname{E}^{p,q}_{2,\mathfrak{m}}$ is a free O_{λ} -module, and vanishes if $(p,q) \notin \{(-1,2r), (0,2r-1), (1,2r-2)\}$.
- (3) If we denote by $\{\alpha_1^{\pm 1}, \ldots, \alpha_r^{\pm 1}\}$ the roots of $P_{\alpha(\Pi_{\mathfrak{p}})} \mod \lambda$ in a finite extension of O_{λ}/λ , then the generalized Frobenius eigenvalues of the $O_{\lambda}/\lambda[\operatorname{Gal}(\overline{\mathbb{F}}_p/\mathbb{F}_{p^2})]$ -module $\mathrm{H}^{2r-1}_{\mathfrak{T}}(\overline{\mathrm{M}}^{\bullet}_{N},O_{\lambda}(r))_{\mathfrak{m}}\otimes_{O_{\lambda}}O_{\lambda}/\lambda \text{ is contained in } \{p\alpha_{1}^{\pm 1},\ldots,p\alpha_{r}^{\pm 1}\}\setminus\{1,p^{2}\}.$
- (4) The map in Proposition 6.3.1(4) (see Remark 6.3.3 for N = 2) factors through a map

 $\nabla^0_{/\mathfrak{n}} \colon \mathrm{F}_{-1}\mathrm{H}^1(\mathrm{I}_{\mathbb{O}_{-2}},\mathrm{H}^{2r-1}_{\mathfrak{T}}(\overline{\mathrm{M}}_N,\mathrm{R}\Psi O_{\lambda}(r))/\mathfrak{n}) \to O_{\lambda}[\mathrm{Sh}(\mathrm{V}_N^\circ,\mathrm{K}_N^\circ)]/\mathfrak{n}$

which is an isomorphism. The map from Lemma 5.8.3(6) induces a canonical isomorphism

$$\mathrm{F}_{-1}\mathrm{H}^{1}(\mathrm{I}_{\mathbb{Q}_{p^{2}}},\mathrm{H}^{2r-1}_{\mathfrak{T}}(\overline{\mathrm{M}}_{N},\mathrm{R}\Psi O_{\lambda}(r))/\mathfrak{n})\to\mathrm{H}^{1}_{\mathrm{sing}}(\mathbb{Q}_{p^{2}},\mathrm{H}^{2r-1}_{\mathfrak{T}}(\overline{\mathrm{M}}_{N},\mathrm{R}\Psi O_{\lambda}(r))/\mathfrak{n}).$$

(5) There exist finitely many positive integers m_1, \ldots, m_{μ} at most m such that we have an isomorphism

$$\mathrm{H}^{2r-1}_{\mathrm{\acute{e}t}}(\mathrm{Sh}(\mathrm{V}'_{N}, \mathbf{j}_{N}\mathrm{K}^{p\circ}_{N}\mathrm{K}'_{p,N})_{\overline{F}}, O_{\lambda}(r))/\mathfrak{n} \simeq \bigoplus_{i=1}^{\mu} \bar{\mathrm{R}}^{(m_{i})\mathtt{c}}$$

of $O_{\lambda}[\Gamma_{F}]$ -modules, where R is the Γ_{F} -stable O_{λ} -lattice in $\rho_{\Pi,\lambda}(r)$, unique up to homothety.

Remark 6.3.5. In fact, from the proof, one sees that when $N \ge 4$, we can take $m_1 = \cdots = m_\mu = m$ due to our strong extra assumptions.

6.4. Proof of Theorem 6.3.4. The proof differs according to N = 2 or $N \ge 4$ as we can see from the extra assumptions. We start from the much more difficult case where $N \ge 4$.

We apply the discussion of Section E to the pair (\bar{r}, χ) as in Subsection E.7, where

$$\bar{r} \coloneqq \bar{\rho}_{\Pi,\lambda,+} \colon \Gamma_{F^+} \to \mathscr{G}_N(O_\lambda/\lambda),$$

and the similitude character $\chi = \eta_{F/F^+}^{\mu} \epsilon_{\ell}^{1-N} \colon \Gamma_{F^+} \to O_{\lambda}^{\times}$ for some $\mu \in \mathbb{Z}/2\mathbb{Z}$.¹⁷ Then \bar{r} is rigid for $(\Sigma_{\min}^+, \Sigma_{lr}^+)$, and also for $(\Sigma_{\min}^+, \Sigma_{lr}^+ \cup \{\mathfrak{p}\})$ by (P4).

¹⁷In fact, it will follow from Theorem E.7.3(3) that $\mu = 0$; but we do not need this fact a priori.

For ? = mix, unr, ram, consider a global deformation problem (Definition E.2.6)

$$\mathscr{S}^{?} \coloneqq (\bar{r}, \eta_{F/F^{+}}^{\mu} \epsilon_{\ell}^{1-N}, \Sigma_{\min}^{+} \cup \Sigma_{\mathrm{lr}}^{+} \cup \{\mathfrak{p}\} \cup \Sigma_{\ell}^{+}, \{\mathscr{D}_{v}\}_{v \in \Sigma_{\min}^{+} \cup \Sigma_{\mathrm{lr}}^{+} \cup \{\mathfrak{p}\} \cup \Sigma_{\ell}^{+})}$$

where

- O for $v \in \Sigma_{\min}^+$, \mathscr{D}_v is the local deformation problem classifying all liftings of \bar{r}_v ;
- O for $v \in \Sigma_{\mathrm{lr}}^+$, \mathscr{D}_v is the local deformation problem $\mathscr{D}^{\mathrm{ram}}$ of \bar{r}_v from Definition E.6.1;
- O for $v = \mathfrak{p}, \mathscr{D}_v$ is the local deformation problem $\mathscr{D}^?$ of \bar{r}_v from Definition E.6.1;
- **O** for $v \in \Sigma_{\ell}^+$, \mathscr{D}_v is the local deformation problem \mathscr{D}^{FL} of \bar{r}_v from Definition E.3.6.

Then we have the global universal deformation ring $\mathsf{R}^{\mathrm{univ}}_{\mathscr{S}^?}$ from Proposition E.2.7. Put $\mathsf{R}^? \coloneqq \mathsf{R}^{\mathrm{univ}}_{\mathscr{S}^?}$ for short. Then we have canonical surjective homomorphisms $\mathsf{R}^{\mathrm{mix}} \to \mathsf{R}^{\mathrm{unr}}$ and $\mathsf{R}^{\mathrm{mix}} \to \mathsf{R}^{\mathrm{ram}}$ of O_{λ} -algebras. Finally, put

$$\mathsf{R}^{\mathrm{cong}} \coloneqq \mathsf{R}^{\mathrm{unr}} \otimes_{\mathsf{R}^{\mathrm{mix}}} \mathsf{R}^{\mathrm{ram}}$$

We fix a universal lifting

$$r_{\min} \colon \Gamma_{F^+} \to \mathscr{G}_N(\mathsf{R}^{\min})$$

of \bar{r} , which induce a continuous homomorphism

$$r_{\min}^{\natural} \colon \Gamma_F \to \operatorname{GL}_N(\mathsf{R}^{\min})$$

by restriction (Notation E.1.2). By pushforward, R^{cong} also induces homomorphisms

$$r_{\mathrm{unr}} \colon \Gamma_{F^+} \to \mathscr{G}_N(\mathsf{R}^{\mathrm{unr}}), \quad r_{\mathrm{ram}} \colon \Gamma_{F^+} \to \mathscr{G}_N(\mathsf{R}^{\mathrm{ram}})$$

Denote by $P_{F_p^+}$ the maximal closed subgroup of the inertia subgroup $I_{F_p^+} \subseteq \Gamma_{F_p^+}$ of pro-order coprime to ℓ , as in Subsection E.5. Then $\Gamma_{F_p^+}/P_{F_p^+} \simeq t^{\mathbb{Z}_\ell} \rtimes \phi_p^{\widehat{\mathbb{Z}}}$ is a *p*-tame group (Definition E.4.1). By definition, the homomorphism r_{\min}^{\natural} is trivial on $P_{F_p^+}$. Let \overline{v} and \overline{v}' be eigenvectors in $(O_\lambda/\lambda)^{\oplus N}$ for $\overline{r}^{\natural}(\phi_p^2)$ with eigenvalues p^{-2r} and p^{-2r+2} , respectively. By Hensel's lemma, \overline{v} and \overline{v}' lift to eigenvectors v and v' in $(\mathbb{R}^{\min})^{\oplus N}$ for $r_{\min}^{\natural}(\phi_p^2)$, with eigenvalues \mathbf{s} and \mathbf{s}' in \mathbb{R}^{\min} lifting p^{-2r} and p^{-2r+2} , respectively. Let $\mathbf{x} \in \mathbb{R}^{\min}$ be the unique element such that $r_{\min}^{\natural}(t)\mathbf{v}' = \mathbf{x}\mathbf{v} + \mathbf{v}'$. Then we must have $\mathbf{x}(\mathbf{s} - p^{-2r}) = 0$. By Definition E.6.1, we have

$$\mathsf{R}^{\mathrm{unr}} = \mathsf{R}^{\mathrm{mix}}/(\mathsf{x}), \quad \mathsf{R}^{\mathrm{ram}} = \mathsf{R}^{\mathrm{mix}}/(\mathsf{s} - p^{-2r}), \quad \mathsf{R}^{\mathrm{cong}} = \mathsf{R}^{\mathrm{mix}}/(\mathsf{s} - p^{-2r}, \mathsf{x}).$$

Let $\mathsf{T}^{\mathrm{unr}}$ be the image of $\mathbb{T}_N^{\Sigma^+}$ in $\operatorname{End}_{O_\lambda}(O_\lambda[\operatorname{Sh}(\mathsf{V}_N^\circ,\mathsf{K}_N^\circ)])$. By (d) in Theorem 6.3.4, we know that $\mathsf{T}^{\mathrm{unr}}_{\mathfrak{m}} \neq 0$. Thus by Proposition E.7.3, we have a canonical isomorphism $\mathsf{R}^{\mathrm{unr}} \xrightarrow{\sim} \mathsf{T}^{\mathrm{unr}}_{\mathfrak{m}}$ so that $O_\lambda[\operatorname{Sh}(\mathsf{V}_N^\circ,\mathsf{K}_N^\circ)]_{\mathfrak{m}}$ is canonically a free $\mathsf{R}^{\mathrm{unr}}$ -module of rank $d_{\mathrm{unr}} > 0$. We may write the eigenvalues of $r^{\natural}_{\mathrm{unr}}(\phi_p^2)$ by $\{p^{-2r+1}\alpha_1^{\pm 1},\ldots,p^{-2r+1}\alpha_{r-1}^{\pm 1},\mathsf{s},\mathsf{s}'=p^{-4r+2}\mathsf{s}^{-1}\}$ with $\alpha_1,\ldots,\alpha_{r-1}$ in certain finite flat extension of $\mathsf{R}^{\mathrm{unr}}$ that are not congruent to p or p^{-1} in O_λ/λ . By Proposition B.3.5(2), we have

$$((p+1)\mathfrak{R}^{\circ}_{N,\mathfrak{p}}-\mathfrak{l}^{\circ}_{N,\mathfrak{p}}).O_{\lambda}[\mathrm{Sh}(\mathrm{V}^{\circ}_{N},\mathrm{K}^{\circ}_{N})]_{\mathfrak{m}}=(\mathfrak{s}-p^{-2r}).O_{\lambda}[\mathrm{Sh}(\mathrm{V}^{\circ}_{N},\mathrm{K}^{\circ}_{N})]_{\mathfrak{m}}.$$

In particular, we have

$$O_{\lambda}[\operatorname{Sh}(\operatorname{V}_{N}^{\circ},\operatorname{K}_{N}^{\circ})]_{\mathfrak{m}}/((p+1)\mathfrak{R}_{N,\mathfrak{p}}^{\circ}-\operatorname{I}_{N,\mathfrak{p}}^{\circ})=O_{\lambda}[\operatorname{Sh}(\operatorname{V}_{N}^{\circ},\operatorname{K}_{N}^{\circ})]_{\mathfrak{m}}\otimes_{\mathsf{R}^{\operatorname{unr}}}\mathsf{R}^{\operatorname{cong}},$$

which is a free R^{cong} -module of rank d_{unr} .

On the other hand, let $\mathsf{T}^{\mathrm{ram}}$ be the image of $\mathbb{T}_N^{\Sigma^+ \cap \Sigma_p^+}$ in $\mathrm{End}_{O_\lambda}(\mathrm{H}_{\mathfrak{T}}^{2r-1}(\overline{\mathrm{M}}_N, \mathrm{R}\Psi O_\lambda))$. By Proposition 6.3.1(4) and Lemma 5.8.3(6), we know that $\mathsf{T}_{\mathfrak{m}}^{\mathrm{ram}} \neq 0$. Thus by Lemma 5.1.7 and Theorem E.7.3 (with $(\Sigma_{\min}^+, \Sigma_{\mathrm{lr}}^+, \Sigma^+)$ replaced by $(\Sigma_{\min}^+, \Sigma_{\mathrm{lr}}^+ \cup \{\mathfrak{p}\}, \Sigma^+ \cup \Sigma_p^+)$), we have a canonical isomorphism $\mathsf{R}^{\mathrm{ram}} \xrightarrow{\sim} \mathsf{T}_{\mathfrak{m}}^{\mathrm{ram}}$ so that $\mathsf{H}_{\mathfrak{T}}^{2r-1}(\overline{\mathrm{M}}_N, \mathrm{R}\Psi O_\lambda)_{\mathfrak{m}}$ is canonically a free $\mathsf{R}^{\mathrm{ram}}$ -module. Define the $\mathsf{R}^{\mathrm{ram}}$ -module

$$\mathsf{H} \coloneqq \operatorname{Hom}_{\Gamma_F} \left((\mathsf{R}^{\operatorname{ram}})^{\oplus N}, \mathrm{H}^{2r-1}_{\mathfrak{T}}(\overline{\mathrm{M}}_N, \mathrm{R}\Psi O_{\lambda})_{\mathfrak{m}} \right)$$

where Γ_F acts on $(\mathsf{R}^{\mathrm{ram}})^{\oplus N}$ via the homomorphism $r_{\mathrm{ram}}^{\sharp,\mathsf{c}}$. By the same argument for [Sch18, Theorem 5.6] (using Proposition C.3.1 and Hypothesis 3.2.9, here), we have a canonical isomorphism

$$\mathrm{H}^{2r-1}_{\mathfrak{T}}(\overline{\mathrm{M}}_{N},\mathrm{R}\Psi O_{\lambda})_{\mathfrak{m}}\simeq\mathsf{H}\otimes_{\mathsf{R}^{\mathrm{ram}}}(\mathsf{R}^{\mathrm{ram}})^{\oplus l}$$

of $\mathsf{R}^{\mathrm{ram}}[\Gamma_F]$ -modules. Since $\mathsf{R}^{\mathrm{ram}}$ is a local ring, H is a free $\mathsf{R}^{\mathrm{ram}}$ -module, say of rank d_{ram} . If we still denote by v and v' for their projection in $(\mathsf{R}^{\mathrm{ram}})^{\oplus N}$, then it is easy to see that

$$\mathrm{H}^{1}_{\mathrm{sing}}(\mathbb{Q}_{p^{2}},(\mathsf{R}^{\mathrm{ram}})^{\oplus N}(r)) = \mathsf{R}^{\mathrm{ram}}\mathsf{v}/\mathsf{x}\mathsf{v} \simeq \mathsf{R}^{\mathrm{ram}}/(\mathsf{x}) = \mathsf{R}^{\mathrm{cong}}$$

Thus, we obtain

$$\mathrm{H}^{1}_{\mathrm{sing}}(\mathbb{Q}_{p^{2}},\mathrm{H}^{2r-1}_{\mathfrak{T}}(\overline{\mathrm{M}}_{N},\mathrm{R}\Psi O_{\lambda}(r))_{\mathfrak{m}})\simeq \mathsf{H}\otimes_{\mathsf{R}^{\mathrm{ram}}}\mathrm{H}^{1}_{\mathrm{sing}}(\mathbb{Q}_{p^{2}},(\mathsf{R}^{\mathrm{ram}})^{\oplus N}(r))\simeq \mathsf{H}\otimes_{\mathsf{R}^{\mathrm{ram}}}\mathsf{R}^{\mathrm{cong}},$$

which is a free $\mathsf{R}^{\mathrm{cong}}$ -module of rank $d_{\mathrm{ram}}>0$.

Proposition 6.4.1. Under the assumptions of Theorem 6.3.4, we have $d_{unr} = d_{ram}$. In particular, the two canonical maps

$$\begin{split} & \mathcal{F}_{-1}\mathcal{H}^{1}(\mathcal{I}_{\mathbb{Q}_{p^{2}}},\mathcal{H}^{2r-1}_{\mathfrak{T}}(\overline{\mathcal{M}}_{N},\mathcal{R}\Psi O_{\lambda}(r))_{\mathfrak{m}}) \to O_{\lambda}[\mathrm{Sh}(\mathcal{V}_{N}^{\circ},\mathcal{K}_{N}^{\circ})]_{\mathfrak{m}}/((p+1)\mathcal{R}_{N,\mathfrak{p}}^{\circ}-\mathfrak{I}_{N,\mathfrak{p}}^{\circ}), \\ & \mathcal{F}_{-1}\mathcal{H}^{1}(\mathcal{I}_{\mathbb{Q}_{p^{2}}},\mathcal{H}^{2r-1}_{\mathfrak{T}}(\overline{\mathcal{M}}_{N},\mathcal{R}\Psi O_{\lambda}(r))_{\mathfrak{m}}) \to \mathcal{H}^{1}_{\mathrm{sing}}(\mathbb{Q}_{p^{2}},\mathcal{H}^{2r-1}_{\mathfrak{T}}(\overline{\mathcal{M}}_{N},\mathcal{R}\Psi O_{\lambda}(r))_{\mathfrak{m}}), \end{split}$$

from Proposition 6.3.1(4) and Lemma 5.8.3(6), respectively, are both isomorphisms.

Proof. By Proposition 6.3.1(4), the first map is surjective. By Lemma 5.8.3(6), the second map is injective. Thus, we must have $d_{\text{ram}} \ge d_{\text{unr}} > 0$ by the previous discussion.

Take a geometric point $\eta_1 \in (\operatorname{Spec} \mathsf{R}^{\operatorname{unr}})(\overline{\mathbb{Q}}_{\ell})$ in the support of $O_{\lambda}[\operatorname{Sh}(V_N^\circ, \mathsf{K}_N^\circ)]_{\mathfrak{m}}$, which corresponds to a relevant representation Π_1 of $\operatorname{GL}_N(\mathbb{A}_F)$ such that ρ_{Π_1,ι_ℓ} is residually isomorphic to $\bar{\rho}_{\Pi,\lambda} \otimes_{O_{\lambda}/\lambda} \overline{\mathbb{F}}_{\ell}$. Then we have

$$d_{\rm unr} = \dim \overline{\mathbb{Q}}_{\ell} [\operatorname{Sh}(\mathcal{V}_N^\circ, \mathcal{K}_N^\circ)] [\iota_{\ell} \phi_{\Pi_1}].$$

Take a geometric point $\eta_2 \in (\operatorname{Spec} \mathsf{R}^{\operatorname{ram}})(\overline{\mathbb{Q}}_{\ell})$ in the support of $\mathrm{H}^{2r-1}_{\mathfrak{T}}(\overline{\mathrm{M}}_N, \mathrm{R}\Psi O_{\lambda})_{\mathfrak{m}}$, which corresponds to a relevant representation Π_2 of $\mathrm{GL}_N(\mathbb{A}_F)$ such that ρ_{Π_2,ι_ℓ} is residually isomorphic to $\bar{\rho}_{\Pi,\lambda} \otimes_{O_{\lambda}/\lambda} \overline{\mathbb{F}}_{\ell}$. Then we have

$$d_{\mathrm{ram}} = \dim \mathrm{H}^{2r-1}_{\mathfrak{T}}(\overline{\mathrm{M}}_{N}, \mathrm{R}\Psi\overline{\mathbb{Q}}_{\ell})[\iota_{\ell}\phi_{\Pi_{2}}] = \dim \mathrm{H}^{2r-1}_{\mathrm{\acute{e}t}}(\mathrm{Sh}(\mathrm{V}'_{N}, \mathbf{j}_{N}\mathrm{K}^{p\circ}_{N}\mathrm{K}'_{p,N})_{\overline{F}}, \overline{\mathbb{Q}}_{\ell})[\iota_{\ell}\phi_{\Pi_{2}}]$$

by Lemma 5.1.7. By Proposition D.2.3 and Lemma 6.4.2 below, we have $d_{unr} = d_{ram}$. The proposition follows.

Lemma 6.4.2. Let Π_1 and Π_2 be two relevant representations of $\operatorname{GL}_N(\mathbb{A}_F)$ such that the associated Galois representations ρ_{Π_1,ι_ℓ} and ρ_{Π_2,ι_ℓ} are both residually isomorphic to $\bar{\rho}_{\Pi,\lambda} \otimes_{O_{\lambda}/\lambda} \overline{\mathbb{F}}_{\ell}$. For every $v \in \Sigma_{\min}^+$ (so that every lifting of $\bar{\rho}_{\Pi,\lambda,+,v}$ is minimally ramified), if we realize $\Pi_{1,v}$ and $\Pi_{2,v}$ on vector spaces V_1 and V_2 , respectively, then there exist normalized intertwining operators $A_{\Pi_{1,v}}$ and $A_{\Pi_{2,v}}$ for $\Pi_{1,v}$ and $\Pi_{2,v}$ [Shi11, Section 4.1], respectively, such that we have an $\operatorname{GL}_N(O_{F_v})$ -equivariant isomorphism $i: V_1 \xrightarrow{\sim} V_2$ satisfying $i \circ A_{\Pi_{1,v}} = A_{\Pi_{2,v}} \circ i$.

Proof. We will give the proof when v does not split in F, and leave the other similar case to the readers. Let w be the unique place of F above v.

By Proposition 3.2.4(1), both $\Pi_{1,w}$ and $\Pi_{2,w}$ are tempered. Thus by the Bernstein–Zelevinsky classification, for $\alpha = 1, 2$, we can write

$$\Pi_{\alpha,w} = \mathrm{I}_{P_{\alpha}}^{\mathrm{GL}_{N}(F_{w})}\left(\sigma_{\alpha,-t_{\alpha}}\boxtimes\cdots\boxtimes\sigma_{\alpha,-1}\boxtimes\sigma_{\alpha,0}\boxtimes\sigma_{\alpha,1}\boxtimes\cdots\boxtimes\sigma_{\alpha,t_{\alpha}}\right)$$

for some integer $t_{\alpha} \geq 0$, some standard parabolic subgroup $P_{\alpha} \subseteq \operatorname{GL}_{N}(F_{w})$, and some (unitary) discrete series representations $\{\sigma_{\alpha,-t_{\alpha}},\ldots,\sigma_{\alpha,t_{\alpha}}\}$ satisfying $\sigma_{\alpha,-i} \simeq \sigma_{\alpha,i}^{\vee c}$. See Subsection C.1 for the notation on parabolic induction.

By Proposition E.5.11(3) and [BLGGT14, Lemma 1.3.4(2)], we know that $\rho_{\Pi_1,\iota_\ell}|_{I_{Fw}}$ and $\rho_{\Pi_2,\iota_\ell}|_{I_{Fw}}$ are conjugate. Thus, by [Yao, Lemma 3.6], we have $P_1 = P_2$ (say P) and $t_1 = t_2$ (say t), and we assume that there are unramified (unitary) characters $\{\chi_{-t}, \ldots, \chi_t\}$ of F_w^{\times} satisfying $\chi_{-i} \simeq \chi_i^{-1}$ such that $\sigma_{2,i} = \sigma_{1,i} \otimes \chi_i$. For every i, we choose a vector space W_i on which $\sigma_{1,i}$ realizes (and also realize $\sigma_{1,i}^{\vee c}$ on W_i via $g \mapsto {}^t g^{-1,c}$), and fix a linear map $A_i \colon W_i \to W_{-i}$ intertwining σ_i and $\sigma_{-i}^{\vee c}$ satisfying $A_{-i} \circ A_i = \mathrm{id}_{W_i}$. Put $\sigma \coloneqq \boxtimes_{i=-t}^t \sigma_{1,i}$ regarded as a representation of P by inflation, which realizes on the space $W \coloneqq \bigotimes_{i=-t}^t W_i$; and put $A_{\sigma} \coloneqq \bigotimes_{i=-t}^t A_i \in \mathrm{End}(W)$. Choose an element $w \in \mathrm{GL}_N(F_w)$ satisfying $w = {}^t w^c$, that $w P w^{-1} \cap P$ is the standard Levi subgroup of P, and that for $(a_{-t}, \ldots, a_t) \in w P w^{-1} \cap P$, we have $w(a_{-t}, \ldots, a_t) w^{-1} = (a_t, \ldots, a_{-t})$.

We realize $\Pi_{1,w}$ on the space

$$V_1 \coloneqq \{ f \colon \operatorname{GL}_n(F_w) \to W \mid f(pg) = \delta_P^{1/2}(p)\sigma(p)f(g), p \in P, g \in \operatorname{GL}_n(F_w) \}$$

Define a linear map $A_{\Pi_{1,w}}: V_1 \to V_1$ by the formula

$$\left(A_{\Pi_{1,w}}(f)\right)(g) = A_{\sigma}\left(f(w^{\mathsf{t}}g^{-1,\mathsf{c}})\right).$$

Then it is clear that $A_{\Pi_{1,w}}$ is a intertwining operator for $\Pi_{1,w}$ satisfying $A^2_{\Pi_{1,w}} = 1$. Similarly, we realize $\Pi_{2,w}$ on the space

$$V_2 \coloneqq \{f \colon \operatorname{GL}_n(F_w) \to W \mid f(pg) = \delta_P^{1/2}(p)\chi(p)\sigma(p)f(g), p \in P, g \in \operatorname{GL}_n(F_w)\},\$$

where we put $\chi := \boxtimes_{i=-t}^{t} \chi_i$ regarded as a character of P. We define $A_{\Pi_{2,w}} \colon V_2 \to V_2$ by the same formula, which is a normalized intertwining operator for $\Pi_{2,w}$. The desired isomorphism i is the map sending $f \in V_1$ to the unique function i(f) such that i(f)(g) = f(g) for $g \in \operatorname{GL}_N(O_{F_w})$. The lemma is proved.

Now we can prove Theorem 6.3.4 when $N \ge 4$.

Proof of Theorem 6.3.4 when $N \ge 4$. For (1), Assumption 6.1.4, Lemma 5.5.2, and the spectral sequence in Lemma 5.8.3 imply that $\mathrm{H}^{i}_{\mathfrak{T}}(\overline{\mathrm{M}}^{\bullet}_{N}, O_{\lambda})_{\mathfrak{m}}$ is O_{λ} -torsion free for $i \neq 2r - 1, 2r$. By Proposition 6.3.1(4) and Proposition 6.4.1, we know that $\mathrm{H}^{2r}_{\mathfrak{T}}(\overline{\mathrm{M}}^{\bullet}_{N}, O_{\lambda})_{\mathfrak{m}}$ is O_{λ} -torsion free. It remains to show that $\mathrm{H}^{2r-1}_{\mathfrak{T}}(\overline{\mathrm{M}}^{\bullet}_{N}, O_{\lambda})_{\mathfrak{m}}$ is O_{λ} -torsion free as well.

By definition, the universal homomorphism $r_{\text{ram}}^{\natural} \colon \Gamma_F \to \text{GL}_N(\mathbb{R}^{\text{ram}})$ has a direct sum decomposition $(\mathbb{R}^{\text{ram}})^{\oplus N} = \mathbb{R}_1 \oplus \mathbb{R}_2$ in which \mathbb{R}_1 is a free \mathbb{R}^{ram} -submodule of rank N-2, and \mathbb{R}_2 is the free \mathbb{R}^{ram} -submodule of rank 2 generated by (the image of) v and v' . We have

$$\mathbf{F}_{-1}\mathbf{H}^{2r-1}_{\mathfrak{T}}(\overline{\mathbf{M}}_N, \mathbf{R}\Psi O_{\lambda})_{\mathfrak{m}} \subseteq \mathsf{H} \otimes_{\mathsf{R}^{\mathrm{ram}}} \mathsf{R}^{\mathrm{ram}}\mathsf{v}$$

On the other hand, we have

$$\mathrm{H}^{2r-1}_{\mathfrak{T}}(\overline{\mathrm{M}}^{\bullet}_{N}, O_{\lambda})_{\mathfrak{m}} = \frac{\mathrm{F}_{0}\mathrm{H}^{2r-1}_{\mathfrak{T}}(\overline{\mathrm{M}}_{N}, \mathrm{R}\Psi O_{\lambda})_{\mathfrak{m}}}{\mathrm{F}_{-1}\mathrm{H}^{2r-1}_{\mathfrak{T}}(\overline{\mathrm{M}}_{N}, \mathrm{R}\Psi O_{\lambda})_{\mathfrak{m}}},$$

and that the quotient $\frac{F_1H_{\mathfrak{T}}^{2r-1}(\overline{M}_N,\mathbb{R}\Psi O_\lambda)_{\mathfrak{m}}}{F_0H_{\mathfrak{T}}^{2r-1}(\overline{M}_N,\mathbb{R}\Psi O_\lambda)_{\mathfrak{m}}}$ is torsion free by Lemma 5.5.2. Thus, the O_λ -torsion of $H_{\mathfrak{T}}^{2r-1}(\overline{M}_N,O_\lambda)_{\mathfrak{m}}$ coincides with

$$\mathsf{H} \otimes_{\mathsf{R}^{\mathrm{ram}}} \mathsf{R}^{\mathrm{ram}} \mathsf{v} / \mathrm{F}_{-1} \mathrm{H}_{\mathfrak{T}}^{2r-1} (\overline{\mathrm{M}}_{N}, \mathrm{R} \Psi O_{\lambda})_{\mathfrak{m}}$$

which is fixed by $\operatorname{Gal}(\overline{\mathbb{F}}_p/\mathbb{F}_{p^2})$. However, by Lemma 5.8.3(6) and Proposition 6.4.1, the O_{λ} -torsion of $\operatorname{H}^{2r-1}_{\mathfrak{T}}(\overline{\operatorname{M}}^{\bullet}_{N}, O_{\lambda})_{\mathfrak{m}}$ has to vanish. Part (1) is proved.

Part (2) is an immediate consequence of (1), Assumption 6.1.4, Lemma 5.5.2, and the spectral sequence in Lemma 5.8.3.

Part (3) is a consequence of (1) and (P4) that $P_{\alpha(\Pi_{\mathfrak{p}})} \mod \lambda^m$ is level-raising special at \mathfrak{p} . In fact, we have an isomorphism

$$\mathrm{H}^{2r-1}_{\mathfrak{T}}(\overline{\mathrm{M}}^{\bullet}_{N}, O_{\lambda}(r)) \simeq \mathsf{H} \otimes_{\mathsf{R}^{\mathrm{ram}}} \mathsf{R}_{1}(r)$$

of $O_{\lambda}[\operatorname{Gal}(\overline{\mathbb{F}}_p/\mathbb{F}_{p^2})]$ -modules.

For (4), by Proposition 6.4.1, it suffices to show that the two natural maps

$$\begin{split} & \mathcal{F}_{-1}\mathcal{H}^{1}(\mathcal{I}_{\mathbb{Q}_{p^{2}}},\mathcal{H}_{\mathfrak{T}}^{2r-1}(\overline{\mathcal{M}}_{N},\mathcal{R}\Psi O_{\lambda}(r))_{\mathfrak{m}})/\mathfrak{n} \to \mathcal{F}_{-1}\mathcal{H}^{1}(\mathcal{I}_{\mathbb{Q}_{p^{2}}},\mathcal{H}_{\mathfrak{T}}^{2r-1}(\overline{\mathcal{M}}_{N},\mathcal{R}\Psi O_{\lambda}(r))/\mathfrak{n}), \\ & \mathcal{H}^{1}_{\mathrm{sing}}(\mathbb{Q}_{p^{2}},\mathcal{H}_{\mathfrak{T}}^{2r-1}(\overline{\mathcal{M}}_{N},\mathcal{R}\Psi O_{\lambda}(r))_{\mathfrak{m}})/\mathfrak{n} \to \mathcal{H}^{1}_{\mathrm{sing}}(\mathbb{Q}_{p^{2}},\mathcal{H}_{\mathfrak{T}}^{2r-1}(\overline{\mathcal{M}}_{N},\mathcal{R}\Psi O_{\lambda}(r))/\mathfrak{n}), \end{split}$$

are both isomorphisms. Note that we have a short exact sequence

$$0 \to \mathcal{F}_{-1}\mathcal{H}^{1}(\mathcal{I}_{\mathbb{Q}_{p^{2}}}, \mathcal{H}^{2r-1}_{\mathfrak{T}}(\overline{\mathcal{M}}_{N}, \mathcal{R}\Psi O_{\lambda}(r))_{\mathfrak{m}}) \to \mathcal{H}^{1}(\mathcal{I}_{\mathbb{Q}_{p^{2}}}, \mathcal{H}^{2r-1}_{\mathfrak{T}}(\overline{\mathcal{M}}_{N}, \mathcal{R}\Psi O_{\lambda}(r))_{\mathfrak{m}}) \to \frac{\mathcal{H}^{2r-1}_{\mathfrak{T}}(\overline{\mathcal{M}}_{N}, \mathcal{R}\Psi O_{\lambda}(r))_{\mathfrak{m}}}{\mathcal{F}_{-1}\mathcal{H}^{2r-1}_{\mathfrak{T}}(\overline{\mathcal{M}}_{N}, \mathcal{R}\Psi O_{\lambda}(r))_{\mathfrak{m}}} \to 0$$

of $\mathbb{T}_N^{\Sigma^+ \cup \Sigma_p^+}$ -modules, which is split by considering $\operatorname{Gal}(\overline{\mathbb{F}}_p/\mathbb{F}_{p^2})$ actions and (3). Thus, the first isomorphism is confirmed. The second one is also confirmed as, by (3), one can replace $\operatorname{Gal}(\overline{\mathbb{F}}_p/\mathbb{F}_{p^2})$ -invariants by $\operatorname{Gal}(\overline{\mathbb{F}}_p/\mathbb{F}_{p^2})$ -coinvariants. Part (4) is proved.

For (5), we have

$$\mathrm{H}^{2r-1}_{\mathrm{\acute{e}t}}(\mathrm{Sh}(\mathrm{V}'_{N}, \mathfrak{j}_{N}\mathrm{K}^{p\circ}_{N}\mathrm{K}'_{p,N})_{\overline{F}}, O_{\lambda}(r))/\mathfrak{n} \simeq \mathsf{H} \otimes_{\mathsf{R}^{\mathrm{ram}}/\mathfrak{n}} (\mathsf{R}^{\mathrm{ram}}/\mathfrak{n})^{\oplus N}(r)$$

by Lemma 5.1.7. Here, we regard \mathfrak{n} as its image in $\mathsf{T}^{\mathrm{ram}}_{\mathfrak{m}}$, where the latter is canonically isomorphic to $\mathsf{R}^{\mathrm{ram}}$. We claim that $O_{\lambda}/\lambda^m = \mathsf{R}^{\mathrm{ram}}/\mathfrak{n}$ and $(\mathsf{R}^{\mathrm{ram}}/\mathfrak{n})^{\oplus N}(r) \simeq \overline{\mathsf{R}}^{(m)\mathsf{c}}$ as $O_{\lambda}/\lambda^m[\Gamma_F]$ -modules, where we recall that Γ_F acts on $(\mathsf{R}^{\mathrm{ram}}/\mathfrak{n})^{\oplus N}$ via $r_{\mathrm{ram}}^{\natural,\mathsf{c}}$. By definition, \mathfrak{n} is the kernel of homomorphism

$$\mathbb{T}_N^{\Sigma^+ \cup \Sigma_p^+} \xrightarrow{\phi_{\Pi}} O_E \to O_E / \lambda^m,$$

which satisfies $\mathfrak{n} \cap O_{\lambda} = \lambda^m O_{\lambda}$. Thus, the structure homomorphism $O_{\lambda} \to \mathsf{R}^{\mathrm{ram}}$ induces an equality $O_{\lambda}/\lambda^m = \mathsf{R}^{\mathrm{ram}}/\mathfrak{n}$. Now by the Chebotarev density theorem, and a result of Mazur and Carayol (see [Kis09, Theorem 1.4.1]), we know that the two liftings $(\mathsf{R}^{\mathrm{ram}}/\mathfrak{n})^{\oplus N}(r)$ and $\bar{\mathsf{R}}^{(m)c}$ of $\bar{\rho}^{\mathsf{c}}_{\Pi,\lambda}(r)$ to O_{λ}/λ^m have to be isomorphic.

Theorem 6.3.4 is all proved when $N \ge 4$.

Proof of Theorem 6.3.4 when N = 2. Part (1) is trivial since $\overline{\mathrm{M}}_{2}^{\bullet}$ is a disjoint union of projective lines.

Part (2) follows from (1) by the same reason as for $N \ge 4$.

Part (3) is trivial as $H^1_{\mathfrak{T}}(M^{\bullet}_2, O_{\lambda}(1)) = 0.$

For (4), from Remark 6.3.3, we know that the natural map

$$\mathbf{F}_{-1}\mathbf{H}^{1}(\mathbf{I}_{\mathbb{Q}_{p^{2}}},\mathbf{H}^{1}_{\mathfrak{T}}(\overline{\mathbf{M}}_{2},\mathbf{R}\Psi O_{\lambda}(1))_{\mathfrak{m}})\to O_{\lambda}[\mathbf{Sh}(\mathbf{V}_{2}^{\circ},\mathbf{K}_{2}^{\circ})]_{\mathfrak{m}}/((p+1)\mathbf{R}_{2,\mathfrak{p}}^{\circ}-\mathbf{I}_{2,\mathfrak{p}}^{\circ})$$

is an isomorphism. By (3) and Lemma 5.8.3(6), the natural map

$$\mathrm{F}_{-1}\mathrm{H}^{1}(\mathrm{I}_{\mathbb{Q}_{p^{2}}},\mathrm{H}^{1}_{\mathfrak{T}}(\overline{\mathrm{M}}_{2},\mathrm{R}\Psi O_{\lambda}(1))_{\mathfrak{m}})\to\mathrm{H}^{1}_{\mathrm{sing}}(\mathbb{Q}_{p^{2}},\mathrm{H}^{1}_{\mathfrak{T}}(\overline{\mathrm{M}}_{2},\mathrm{R}\Psi O_{\lambda}(1))_{\mathfrak{m}})$$

is an isomorphism as well. Passing to the quotient by \mathfrak{n} follows from the same argument as for $N \geq 4$.

For (5), let $\mathsf{T}^{\mathrm{ram}}$ be the image of $\mathbb{T}_2^{\Sigma^+ \cap \Sigma_p^+}$ in $\mathrm{End}_{O_\lambda}(\mathrm{H}^1_{\mathfrak{T}}(\overline{\mathrm{M}}_2, \mathrm{R}\Psi O_\lambda(1)))$. Then by the same argument for [Sch18, Theorem 5.6], one have an isomorphism

$$\mathrm{H}^{1}_{\mathfrak{T}}(\overline{\mathrm{M}}_{2},\mathrm{R}\Psi O_{\lambda}(1))_{\mathfrak{m}}\simeq \mathsf{H}\otimes_{\mathsf{T}^{\mathrm{ram}}_{\mathfrak{m}}}(\mathsf{T}^{\mathrm{ram}}_{\mathfrak{m}})^{\oplus 2}$$

of $\mathsf{T}^{\mathrm{ram}}_{\mathfrak{m}}[\Gamma_F]$ -modules, where H is a (finitely generated) $\mathsf{T}^{\mathrm{ram}}_{\mathfrak{m}}$ -module, and Γ_F acts on the factor $(\mathsf{T}^{\mathrm{ram}}_{\mathfrak{m}})^{\oplus 2}$ by some continuous homomorphism which lifts $\bar{\rho}^{\mathrm{c}}_{\mathrm{II},\lambda}(r)$ (from $O_{\lambda}/\lambda = \mathsf{T}^{\mathrm{ram}}_{\mathfrak{m}}/\mathfrak{m}$ to $\mathsf{T}^{\mathrm{ram}}_{\mathfrak{m}}$).

Clearly, the natural homomorphism $O_{\lambda}/\lambda^m \to \mathsf{T}^{\mathrm{ram}}_{\mathfrak{m}}/\mathfrak{n}$ is an isomorphism. Then as an O_{λ}/λ^m module, $\mathsf{H} \otimes_{\mathsf{T}^{\mathrm{ram}}_{\mathfrak{m}}} \mathsf{T}^{\mathrm{ram}}_{\mathfrak{m}}/\mathfrak{n}$ is isomorphic to $\bigoplus_{i=1}^{\mu} O_{\lambda}/\lambda^{m_i}$ for some positive integers m_1, \ldots, m_{μ} at most *m*. Thus, it remains to show that $(\mathsf{T}_{\mathfrak{m}}^{ram}/\mathfrak{n})^{\oplus 2}$ and $\bar{\mathsf{R}}^{(m)c}$ are isomorphic as deformations of $\bar{\rho}_{\Pi\lambda}^{c}(r)$. But this is a consequence of the Eichler–Shimura relation for the unitary Shimura curve $Sh(V'_2, j_2K_2^{p\circ}K'_{p,2})$ [Liu, Corollary D.9], the Chebotarev density theorem, and a result of Mazur and Carayol (see [Kis09, Theorem 1.4.1]).

Theorem 6.3.4 is all proved when N = 2.

7. Explicit reciprocity laws for Rankin–Selberg motives

In this section, we state and prove the two explicit reciprocity laws for automorphic Rankin– Selberg motives. In Subsection 7.1, we setup the stage for automorphic Rankin–Selberg motives, which will be used until the end of the next section. In Subsections 7.2 and 7.3, we state and prove our first and second explicit reciprocity law, respectively.

7.1. Setup for automorphic Rankin–Selberg motives. Let $n \geq 2$ be an integer. We denote by n_0 and n_1 the unique even and odd numbers in $\{n, n+1\}$, respectively. Write $n_0 = 2r_0$ and $n_1 = 2r_1 + 1$ for unique integers $r_0, r_1 \ge 1$. In particular, we have $n = r_0 + r_1$. In this and the next sections, we consider

O for $\alpha = 0, 1$, a relevant representation Π_{α} of $\operatorname{GL}_{n_{\alpha}}(\mathbb{A}_{F})$ (Definition 1.1.3),

O a strong coefficient field $E \subseteq \mathbb{C}$ of both Π_0 and Π_1 (Definition 3.2.5).

Put $\Sigma_{\min}^+ \coloneqq \Sigma_{\Pi_0}^+ \cup \Sigma_{\Pi_1}^+$ (Notation 3.1.4). We then have the homomorphism

$$\phi_{\Pi_{\alpha}} \colon \mathbb{T}_{n_{\alpha}}^{\Sigma_{\min}^{+}} \to O_{E}$$

for $\alpha = 0, 1$. For $\alpha = 0, 1$ and every prime λ of E, we have a continuous homomorphism

$$\rho_{\Pi_{\alpha},\lambda} \colon \Gamma_F \to \operatorname{GL}_{n_{\alpha}}(E_{\lambda})$$

from Proposition 3.2.4(2) and Definition 3.2.5, such that $\rho_{\Pi_{\alpha},\lambda}^{\mathsf{c}}$ and $\rho_{\Pi_{\alpha},\lambda}^{\vee}(1-n_{\alpha})$ are conjugate.

Assumption 7.1.1. For $\alpha = 0, 1$, the Galois representation $\rho_{\Pi_{\alpha,\lambda}}$ is residually absolutely irreducible.

7.2. First explicit reciprocity law. We start by choosing

- O a finite place λ of E (with the underlying rational prime ℓ),
- **O** a positive integer m,
- O a (possibly empty) finite set $\Sigma_{\text{lr,I}}^+$ ¹⁸ of nonarchimedean places of F^+ that are inert in F, strongly disjoint from Σ_{\min}^+ (Definition 1.3.3), satisfying $\ell \nmid ||v|| (||v||^2 - 1)$ for $v \in \Sigma_{lr,I}^+$,
- O a finite set $\Sigma_{\rm I}^+$ of nonarchimedean places of F^+ containing $\Sigma_{\rm min}^+ \cup \Sigma_{\rm lr,I}^+$, O a standard definite hermitian space V_n° of rank *n* over *F*, together with a self-dual $\prod_{v \notin \Sigma_{\infty}^+ \cup \Sigma_{\rm lr,I}^+} O_{F_v}$ -lattice Λ_n° in $V_n^{\circ} \otimes_F A_F^{\Sigma_{\infty}^+ \cup \Sigma_{\rm lr,I}^+}$ (and put $V_{n+1}^{\circ} \coloneqq (V_n^{\circ})_{\sharp}$ and $\Lambda_{n+1}^{\circ} \coloneqq (\Lambda_n^{\circ})_{\sharp}$, satisfying that $(V_{n_0}^{\circ})_v$ is not split for $v \in \Sigma_{\mathrm{lr},\mathrm{I}}^+$,

¹⁸Here, the subscript "I" (Roman number one) stands for the "first". In the next subsection, we will have $\Sigma_{lr,II}^+$ for the second reciprocity law.

O an object $K_n^{\circ} \in \mathfrak{K}(V_n^{\circ})$ and an object $(K_{sp}^{\circ}, K_{n+1}^{\circ}) \in \mathfrak{K}(V_n^{\circ})_{sp}$ of the forms

$$\begin{split} \mathbf{K}_{n}^{\circ} &= \prod_{v \in \Sigma_{\min}^{+} \cup \Sigma_{\mathrm{lr},\mathrm{I}}^{+}} (\mathbf{K}_{n}^{\circ})_{v} \times \prod_{v \not\in \Sigma_{\infty}^{+} \cup \Sigma_{\min}^{+} \cup \Sigma_{\mathrm{lr},\mathrm{I}}^{+}} \mathbf{U}(\Lambda_{n}^{\circ})(O_{F_{v}^{+}}), \\ \mathbf{K}_{\mathrm{sp}}^{\circ} &= \prod_{v \in \Sigma_{\min}^{+} \cup \Sigma_{\mathrm{lr},\mathrm{I}}^{+}} (\mathbf{K}_{\mathrm{sp}}^{\circ})_{v} \times \prod_{v \not\in \Sigma_{\infty}^{+} \cup \Sigma_{\min}^{+} \cup \Sigma_{\mathrm{lr},\mathrm{I}}^{+}} \mathbf{U}(\Lambda_{n}^{\circ})(O_{F_{v}^{+}}), \\ \mathbf{K}_{n+1}^{\circ} &= \prod_{v \in \Sigma_{\min}^{+} \cup \Sigma_{\mathrm{lr},\mathrm{I}}^{+}} (\mathbf{K}_{n+1}^{\circ})_{v} \times \prod_{v \not\in \Sigma_{\infty}^{+} \cup \Sigma_{\min}^{+} \cup \Sigma_{\mathrm{lr},\mathrm{I}}^{+}} \mathbf{U}(\Lambda_{n+1}^{\circ})(O_{F_{v}^{+}}), \end{split}$$

satisfying $(\mathcal{K}_{sp}^{\circ})_v = (\mathcal{K}_n^{\circ})_v$ for $v \in \Sigma_{\min}^+$, $(\mathcal{K}_{sp}^{\circ})_v \subseteq (\mathcal{K}_n^{\circ})_v$ for $v \in \Sigma_{\mathrm{lr},\mathrm{I}}^+$, and that $(\mathcal{K}_{n_0}^{\circ})_v$ is a transferable open compact subgroup (Definition D.2.1) of $U(\mathcal{V}_{n_0}^{\circ})(F_v^+)$ for $v \in \Sigma_{\min}^+$ and is a special maximal subgroup of $U(\mathcal{V}_{n_0}^{\circ})(F_v^+)$ for $v \in \Sigma_{\mathrm{lr},\mathrm{I}}^+$,

- O a special inert prime (Definition 3.3.4) \mathfrak{p} of F^+ (with the underlying rational prime p) satisfying
 - (PI1): $\Sigma_{\rm I}^+$ does not contain *p*-adic places;
 - (PI2): ℓ does not divide $p(p^2 1)$;
 - (PI3): there exists a CM type Φ containing τ_{∞} as in the initial setup of Section 5 satisfying $\mathbb{Q}_p^{\Phi} = \mathbb{Q}_{p^2};$
 - (PI4): $P_{\alpha(\Pi_{0,\mathfrak{p}})} \mod \lambda^m$ is level-raising special at \mathfrak{p} (Definition 3.1.5);

 $P_{\alpha(\Pi_{1,\mathfrak{p}})} \mod \lambda$ is Tate generic at \mathfrak{p} (Definition 3.1.5);

(PI5): $P_{\alpha(\Pi_{\alpha,\mathfrak{p}})} \mod \lambda$ is intertwining generic at \mathfrak{p} (Definition 3.1.5) for $\alpha = 0, 1$; (PI6): the natural map

$$\frac{O_E/\lambda^m[\operatorname{Sh}(\operatorname{V}_{n_\alpha}^\circ,\operatorname{K}_{n_\alpha}^\circ)]}{\mathbb{T}_{n_\alpha}^{\Sigma_1^+\cup\Sigma_p^+}\cap \ker \phi_{\Pi_\alpha}} \to \frac{O_E/\lambda^m[\operatorname{Sh}(\operatorname{V}_{n_\alpha}^\circ,\operatorname{K}_{n_\alpha}^\circ)]}{\mathbb{T}_{n_\alpha}^{\Sigma_1^+}\cap \ker \phi_{\Pi_\alpha}}$$

is an isomorphism of *nontrivial* O_E/λ^m -modules for $\alpha = 0, 1$;

(PI7): $P_{\alpha(\Pi_{0,\mathfrak{p}})\otimes\alpha(\Pi_{1,\mathfrak{p}})} \mod \lambda^m$ is level-raising special at \mathfrak{p} (Definition 3.1.5);

(So we can and will adopt the setup in Subsection 5.10 to the datum $(V_n^{\circ}, \{\Lambda_{n,\mathfrak{g}}^{\circ}\}|_{\mathfrak{g}|p})$.)

- O remaining data in the initial setup of Section 5 with $\mathbb{Q}_p^{\Phi} = \mathbb{Q}_{p^2}$; and
- O a definite uniformization datum as in Notation 5.10.13.

Put $K_{sp}^{p\circ} \coloneqq (K_{sp}^{\circ})^p$ and $K_{sp}^{\bullet} \coloneqq K_{sp}^{p\circ} \times K_{n_0,p}^{\bullet}$; put $K_{n_\alpha}^{p\circ} \coloneqq (K_{n_\alpha}^{\circ})^p$ and $K_{n_\alpha}^{\bullet} \coloneqq K_{n_\alpha,p}^{p\circ}$ for $\alpha = 0, 1$. Like in Subsection 5.11, we put $X_{n_\alpha}^? \coloneqq X_{\mathfrak{p}}^?(V_{n_\alpha}^{\circ}, K_{n_\alpha}^{p\circ})$ for meaningful triples $(X, ?, \alpha) \in \{\mathbf{M}, \mathbf{M}, \mathbf{B}, \mathbf{S}\} \times \{, \eta, \circ, \bullet, \dagger\} \times \{0, 1\}$. For $\alpha = 0, 1$, let $({}^{\alpha}\mathbf{E}_s^{p,q}, {}^{\alpha}\mathbf{d}_s^{p,q})$ be the weight spectral sequence abutting to the cohomology $\mathrm{H}^{p+q}_{\mathfrak{T}}(\overline{\mathrm{M}}_{n_\alpha}, \mathrm{R}\Psi O_{\lambda}(r_\alpha))$ from Subsection 5.8.

Notation 7.2.1. We introduce the following ideas of $\mathbb{T}_{n_{\alpha}}^{\Sigma_{1}^{+}\cup\Sigma_{p}^{+}}$, for $\alpha = 0, 1$

$$\begin{cases} \mathfrak{m}_{\alpha} \coloneqq \mathbb{T}_{n_{\alpha}}^{\Sigma_{1}^{+} \cup \Sigma_{p}^{+}} \cap \ker \left(\mathbb{T}_{n_{\alpha}}^{\Sigma^{+}} \xrightarrow{\phi_{\Pi_{\alpha}}} O_{E} \to O_{E} / \lambda \right), \\ \mathfrak{n}_{\alpha} \coloneqq \mathbb{T}_{n_{\alpha}}^{\Sigma_{1}^{+} \cup \Sigma_{p}^{+}} \cap \ker \left(\mathbb{T}_{n_{\alpha}}^{\Sigma^{+}} \xrightarrow{\phi_{\Pi_{\alpha}}} O_{E} \to O_{E} / \lambda^{m} \right). \end{cases}$$

We then introduce following assumptions.

Assumption 7.2.2. For $\alpha = 0, 1$, we have $\mathrm{H}^{i}_{\mathfrak{T}}(\overline{\mathrm{M}}_{n_{\alpha}}, \mathrm{R}\Psi O_{\lambda})_{\mathfrak{m}_{\alpha}} = 0$ for $i \neq n_{\alpha} - 1$, and that $\mathrm{H}^{n_{\alpha}-1}_{\mathfrak{T}}(\overline{\mathrm{M}}_{n_{\alpha}}, \mathrm{R}\Psi O_{\lambda})_{\mathfrak{m}_{\alpha}}$ is a finite free O_{λ} -module.

Assumption 7.2.3. Under Assumption 7.1.1, if $n_0 \ge 4$, then

(a) $\ell \geq 2(n_0 + 1)$ and ℓ is unramified in F;

- (b) $\bar{\rho}_{\Pi_0,\lambda,+}$ (Remark 6.1.5) is rigid for $(\Sigma_{\min}^+, \Sigma_{\mathrm{lr},\mathrm{I}}^+)$ (Definition E.7.1), and $\bar{\rho}_{\Pi_0,\lambda}|_{\mathrm{Gal}(\overline{F}/F(\zeta_{\ell}))}$ is absolutely irreducible; and
- (c) the composite homomorphism $\mathbb{T}_{n_0}^{\Sigma_{\min}^+} \xrightarrow{\phi_{\Pi_0}} O_E \to O_E/\lambda$ is cohomologically generic (Definition D.1.1).

Now we apply constructions in Subsection 5.11, evaluating on the object $(\mathbf{K}_n^{p\circ}, \mathbf{K}_{n+1}^{p\circ})$ of $\mathfrak{K}(\mathbf{V}_n^{\circ})^p \times \mathfrak{K}(\mathbf{V}_{n+1}^{\circ})^p$. In particular, we have the blow-up morphism $\sigma : \mathbf{Q} \to \mathbf{P}$ from Notation 5.11.1, and the localized spectral sequence $(\mathbb{E}_{s,(\mathfrak{m}_0,\mathfrak{m}_1)}^{p,q}, \mathbf{d}_{s,(\mathfrak{m}_0,\mathfrak{m}_1)}^{p,q})$ from (5.27).

Lemma 7.2.4. Assume Assumptions 7.1.1, 7.2.3, 7.2.2, and Hypothesis 3.2.9 for both n and n+1. Then

(1) For $(?_0,?_1) \in \{\circ,\bullet,\dagger\}^2$ and $i \in \mathbb{Z}$, we have a canonical isomorphism

$$\mathrm{H}^{i}_{\mathfrak{T}}(\overline{\mathrm{P}}^{?_{0},?_{1}},O_{\lambda})_{(\mathfrak{m}_{0},\mathfrak{m}_{1})} \simeq \bigoplus_{i_{0}+i_{1}=i} \mathrm{H}^{i_{0}}_{\mathfrak{T}}(\overline{\mathrm{M}}^{?_{0}}_{n_{0}},O_{\lambda})_{\mathfrak{m}_{0}} \otimes_{O_{\lambda}} \mathrm{H}^{i_{1}}_{\mathfrak{T}}(\overline{\mathrm{M}}^{?_{1}}_{n_{1}},O_{\lambda})_{\mathfrak{m}_{2}}$$

in $\mathsf{Mod}(\mathrm{Gal}(\overline{\mathbb{F}}_p/\mathbb{F}_{p^2}), O_\lambda)_{\mathrm{fr}}.$

(2) We have $\mathbb{E}_{2,(\mathfrak{m}_0,\mathfrak{m}_1)}^{p,q} = 0$ if $(p,q) \notin \{(-1,2n), (0,2n-1), (1,2n-2)\}$, and canonical isomorphisms

$$\begin{cases} \mathbb{E}_{2,(\mathfrak{m}_{0},\mathfrak{m}_{1})}^{-1,2n} \simeq {}^{0}\mathrm{E}_{2,\mathfrak{m}_{0}}^{-1,2r_{0}} \otimes_{O_{\lambda}} {}^{1}\mathrm{E}_{2,\mathfrak{m}_{1}}^{0,2r_{1}}, \\ \mathbb{E}_{2,(\mathfrak{m}_{0},\mathfrak{m}_{1})}^{0,2n-1} \simeq {}^{0}\mathrm{E}_{2,\mathfrak{m}_{0}}^{0,2r_{0}-1} \otimes_{O_{\lambda}} {}^{1}\mathrm{E}_{2,\mathfrak{m}_{1}}^{0,2r_{1}}, \\ \mathbb{E}_{2,(\mathfrak{m}_{0},\mathfrak{m}_{1})}^{1,2n-2} \simeq {}^{0}\mathrm{E}_{2,\mathfrak{m}_{0}}^{1,2r_{0}-2} \otimes_{O_{\lambda}} {}^{1}\mathrm{E}_{2,\mathfrak{m}_{1}}^{0,2r_{1}}, \end{cases}$$

in $\operatorname{\mathsf{Mod}}(\operatorname{Gal}(\overline{\mathbb{F}}_p/\mathbb{F}_{p^2}), O_\lambda)_{\operatorname{fr}}$.

- (3) If $\mathbb{E}_{2,(\mathfrak{m}_0,\mathfrak{m}_1)}^{i,2n-1-i}(-1)$ has a nontrivial subquotient on which $\operatorname{Gal}(\overline{\mathbb{F}}_p/\mathbb{F}_{p^2})$ acts trivially, then i=1.
- (4) For $(?_0,?_1) \in \{\circ,\bullet,\dagger\}^2$ and $i \in \mathbb{Z}$, both $\mathrm{H}^{2i}_{\mathfrak{T}}(\overline{\mathrm{P}}^{?_0,?_1},O_{\lambda}(i))_{(\mathfrak{m}_0,\mathfrak{m}_1)}$ and $\mathrm{H}^{2i}_{\mathfrak{T}}(\overline{\mathrm{Q}}^{?_0,?_1},O_{\lambda}(i))_{(\mathfrak{m}_0,\mathfrak{m}_1)}$ are weakly semisimple.
- (5) We have $\mathrm{H}^{i}_{\mathfrak{T}}(\overline{\mathbb{Q}}, \mathrm{R}\Psi O_{\lambda})_{(\mathfrak{m}_{0},\mathfrak{m}_{1})} = 0$ for $i \neq 2n-1$.
- (6) The canonical map $\mathrm{H}^{i}_{\mathfrak{T},c}(\overline{\mathrm{Q}}^{(c)},O_{\lambda})_{(\mathfrak{m}_{0},\mathfrak{m}_{1})} \to \mathrm{H}^{i}_{\mathfrak{T}}(\overline{\mathrm{Q}}^{(c)},O_{\lambda})_{(\mathfrak{m}_{0},\mathfrak{m}_{1})}$ is an isomorphism for every integers c and i.

Proof. For (1), by Lemma 5.5.2, Lemma 6.2.2(2), Theorem 6.3.4(1), we know that $\mathrm{H}^{i_{\alpha}}_{\mathfrak{T}}(\overline{\mathrm{M}}^{?_{\alpha}}_{n_{\alpha}}, O_{\lambda})_{\mathfrak{m}_{\alpha}}$ is a finitely generated free O_{λ} -module for $\alpha = 0, 1$ and every $i_{\alpha} \in \mathbb{Z}$. Thus, (1) follows from Lemma 6.1.10 and the Künneth formula.

For (2), we first show that $\mathbb{E}_{s,(\mathfrak{m}_0,\mathfrak{m}_1)}^{p,q}$ degenerates at the second page. By (1), Lemma 5.11.3(2), Lemma 5.5.2, and Lemma 6.2.1, the composition of $d_{1,(\mathfrak{m}_0,\mathfrak{m}_1)}^{-2,q}$ and the natural projection

$$\mathbb{E}_{1,(\mathfrak{m}_{0},\mathfrak{m}_{1})}^{-1,q} \to \mathrm{H}_{\mathfrak{T}}^{q-2}(\overline{\mathbf{Q}}^{\dagger,\dagger},O_{\lambda}(n-1)) \bigoplus \mathrm{H}_{\mathfrak{T}}^{q-2}(\overline{\mathbf{Q}}^{\dagger,\circ},O_{\lambda}(n-1))$$

is injective for every $q \in \mathbb{Z}$. Thus, $d_{1,(\mathfrak{m}_0,\mathfrak{m}_1)}^{-2,q}$ is injective, which implies $\mathbb{E}_{2,(\mathfrak{m}_0,\mathfrak{m}_1)}^{-2,q} = 0$ for every $q \in \mathbb{Z}$. By a dual argument, we have $\mathbb{E}_{2,(\mathfrak{m}_0,\mathfrak{m}_1)}^{2,q} = 0$ for every $q \in \mathbb{Z}$ as well. For the degeneration, it suffices to show that $d_{1,(\mathfrak{m}_0,\mathfrak{m}_1)}^{-1,q}$ is injective and $d_{1,(\mathfrak{m}_0,\mathfrak{m}_1)}^{0,q}$ is surjective for q odd. By Lemma 5.11.3(2), Lemma 5.5.2, and Lemma 6.2.2(1), we have $\mathbb{H}_{\mathfrak{T}}^{-2}(\overline{Q}^{(1)}, O_{\lambda}(n-1)) = \mathbb{H}_{\mathfrak{T}}^{q-2}(\overline{Q}^{\bullet,\dagger}, O_{\lambda}(n-1))$ for q odd, which easily implies the injectivity of $d_{1,(\mathfrak{m}_0,\mathfrak{m}_1)}^{-1,q}$. By a dual argument, $d_{1,(\mathfrak{m}_0,\mathfrak{m}_1)}^{0,q}$ is surjective for q odd.

Now for every $q \in \mathbb{Z}$, the morphism σ induces a map

$$\sigma_1^* \colon \bigoplus_{q_0+q_1=q} {}^0 \mathrm{E}_{1,\mathfrak{m}_0}^{*,q_0} \otimes_{O_{\lambda}} {}^1 \mathrm{E}_{1,\mathfrak{m}_1}^{*,q_1} \to \mathbb{E}_{1,(\mathfrak{m}_0,\mathfrak{m}_1)}^{*,q}$$

of complexes of $O_{\lambda}[\operatorname{Gal}(\mathbb{F}/\mathbb{F}_{p^2})]$ -modules, hence a map

$$\sigma_2^* \colon \bigoplus_{p_0+p_1=p} \bigoplus_{q_0+q_1=q} {}^0 \mathrm{E}_{2,\mathfrak{m}_0}^{p_0,q_0} \otimes_{O_{\lambda}} {}^1 \mathrm{E}_{2,\mathfrak{m}_1}^{p_1,q_1} \to \mathbb{E}_{2,(\mathfrak{m}_0,\mathfrak{m}_1)}^{p,q}$$

of $O_{\lambda}[\operatorname{Gal}(\overline{\mathbb{F}}/\mathbb{F}_{p^2})]$ -modules for $(p,q) \in \mathbb{Z}^2$. By Lemma 6.2.2 and Theorem 6.3.4(2), to show (2), it suffices to show that σ_2^* is an isomorphism, or the natural map

$$\bigoplus_{i_0+i_1=i} \mathrm{H}^{i_0}_{\mathfrak{T}}(\overline{\mathrm{M}}_{n_0}, \mathrm{R}\Psi O_{\lambda}(r_0))_{\mathfrak{m}_0} \otimes_{O_{\lambda}} \mathrm{H}^{i_1}_{\mathfrak{T}}(\overline{\mathrm{M}}_{n_1}, \mathrm{R}\Psi O_{\lambda}(r_1))_{\mathfrak{m}_1} \to \mathrm{H}^{i}_{\mathfrak{T}}(\overline{\mathrm{Q}}, \mathrm{R}\Psi O_{\lambda}(n))_{(\mathfrak{m}_0,\mathfrak{m}_1)}$$

induced by σ is an isomorphism for every $i \in \mathbb{Z}$. By Lemma 5.1.7 and Lemma 5.11.2, the above map is identified with

$$\bigoplus_{i_0+i_1=i} \mathrm{H}^{i_0}_{\mathfrak{T}}(\mathbf{M}^{\eta}_{n_0} \otimes_{\mathbb{Q}_{p^2}} \overline{\mathbb{Q}}_p, O_{\lambda}(r_0))_{\mathfrak{m}_0} \otimes_{O_{\lambda}} \mathrm{H}^{i_1}_{\mathfrak{T}}(\mathbf{M}^{\eta}_{n_1} \otimes_{\mathbb{Q}_{p^2}} \overline{\mathbb{Q}}_p, O_{\lambda}(r_1))_{\mathfrak{m}_1} \to \mathrm{H}^{i}_{\mathfrak{T}}(\mathbf{Q}^{\eta} \otimes_{\mathbb{Q}_{p^2}} \overline{\mathbb{Q}}_p, O_{\lambda}(n))_{(\mathfrak{m}_0,\mathfrak{m}_1)},$$

which is an isomorphism by Lemma 6.1.10, and the Künneth formula. Thus, (2) follows.

For (3), let $\{\alpha_{0,1}^{\pm 1}, \ldots, \alpha_{0,r_0}^{\pm 1}\}$ and $\{\alpha_{1,1}^{\pm 1}, \ldots, \alpha_{1,r_1}^{\pm 1}, 1\}$ be the roots of $P_{\alpha(\Pi_{0,p})} \mod \lambda$ and $P_{\alpha(\Pi_{1,p})} \mod \lambda$ in a finite extension of O_{λ}/λ , respectively. By (PI4), we may assume $\alpha_{0,r_0} = p$. By (2), Theorem 6.2.3(1), and Theorem 6.3.4(3), the generalized Frobenius eigenvalues of the $O_{\lambda}/\lambda[\operatorname{Gal}(\overline{\mathbb{F}}_p/\mathbb{F}_{p^2})]$ -modules $\mathbb{E}_{2,(\mathfrak{m}_0,\mathfrak{m}_1)}^{-1,2n}(-1) \otimes_{O_{\lambda}} O_{\lambda}/\lambda$ and $\mathbb{E}_{2,(\mathfrak{m}_0,\mathfrak{m}_1)}^{0,2n-1}(-1) \otimes_{O_{\lambda}} O_{\lambda}/\lambda$ are contained in $\{p^{-2}\alpha_{1,1}^{\pm 1}, \ldots, p^{-2}\alpha_{1,r_1}^{\pm 1}, p^{-2}\}$ and $\{p^{-1}\alpha_{0,1}^{\pm 1}\alpha_{1,1}^{\pm 1}, \ldots, p^{-1}\alpha_{0,r_0-1}^{\pm 1}\alpha_{1,r_1}^{\pm 1}\} \cup \{p^{-1}\alpha_{0,1}^{\pm 1}, \ldots, p^{-1}\alpha_{0,r_0-1}^{\pm 1}\}$, respectively. By (PI2), we have $p^2 \neq 1$ in O_{λ}/λ . By (PI7), we have $\alpha_{1,i_1} \notin \{p^2, p^{-2}\}$ for $1 \leq i_1 \leq r_1$, which implies $1 \notin \{p^{-2}\alpha_{1,1}^{\pm 1}, \ldots, p^{-2}\alpha_{1,r_1}^{\pm 1}, p^{-2}\}$. Again by (PI7), we have $\alpha_{0,i_0}\alpha_{1,i_1} \notin \{p, p^{-1}\}$ for $1 \leq i_0 < r_0$ and $1 \leq i_1 \leq r_1$, which implies $1 \notin \{p^{-1}\alpha_{0,1}^{\pm 1}\alpha_{1,1}^{\pm 1}, \ldots, p^{-1}\alpha_{0,r_0-1}^{\pm 1}\alpha_{1,r_1}^{\pm 1}\}$. By (PI4), we have $\alpha_{0,i_0} \notin \{p, p^{-1}\}$ for $1 \leq i_0 < r_0$, which implies $1 \notin \{p^{-1}\alpha_{0,1}^{\pm 1}, \ldots, p^{-1}\alpha_{0,r_0-1}^{\pm 1}\}$. Thus, (3) follows.

For (4), by Lemma 5.11.3 (3–5) and Lemma 2.1.4(1), it suffices to show that $H_{\mathfrak{T}}^{2i}(\overline{\mathbb{P}}^{?_0,?_1}, O_{\lambda}(i))_{(\mathfrak{m}_0,\mathfrak{m}_1)}$ is weakly semisimple. By (1) and Lemma 6.2.2(1), it suffices to show that $H_{\mathfrak{T}}^{2i_0}(\overline{M}_{n_0}^{?_0}, O_{\lambda}(i_0))_{\mathfrak{m}_0} \otimes_{O_{\lambda}} H_{\mathfrak{T}}^{2i_1}(\overline{M}_{n_1}^{?_1}, O_{\lambda}(i_1))_{\mathfrak{m}_1}$ is weakly semisimple for $i_0, i_1 \in \mathbb{Z}$. By Lemma 5.5.2, the action of $\operatorname{Gal}(\overline{\mathbb{F}}_p/\mathbb{F}_{p^2})$ on $H_{\mathfrak{T}}^{2i_\alpha}(\overline{M}_{n_\alpha}^{?}, O_{\lambda}(i_\alpha))_{\mathfrak{m}_\alpha}$ is trivial for $\alpha = 0, 1, ? = \circ, \dagger,$ and every $i_\alpha \in \mathbb{Z}$. On the other hand, it is a consequence of Theorem 6.3.4(2) (for i_0) and Lemma 6.2.2(3) (for i_1) that the action of $\operatorname{Gal}(\overline{\mathbb{F}}_p/\mathbb{F}_{p^2})$ on $H_{\mathfrak{T}}^{2i_\alpha}(\overline{M}_{n_\alpha}^{\bullet}, O_{\lambda}(i_\alpha))_{\mathfrak{m}_\alpha}$ is trivial if $i_0 \notin \{r_0 - 1, r_0\}$ or $i_1 \neq r_1$. By Proposition 6.3.1(1,2) and Theorem 6.3.4(1), the actions of $\operatorname{Gal}(\overline{\mathbb{F}}_p/\mathbb{F}_{p^2})$ on both $H_{\mathfrak{T}}^{2r_0-2}(\overline{M}_{n_0}^{\bullet}, O_{\lambda}(r_0 - 1))_{\mathfrak{m}_0}$ and $H_{\mathfrak{T}}^{2r_0}(\overline{M}_{n_0}^{\bullet}, O_{\lambda}(r_0))_{\mathfrak{m}_0}$ are also trivial. Thus, by Lemma 2.1.4(1), it remains to show that $H_{\mathfrak{T}}^{2r_1}(\overline{M}_{n_1}^{\bullet}, O_{\lambda}(r_1))_{\mathfrak{m}_1}$ is weakly semisimple, which follows from Theorem 6.2.3(2) as it is isomorphic to the direct sum of ${}^{1}\mathrm{E}_{2,\mathfrak{m}^{1}}^{0,2r_1}$ and $H_{\mathfrak{T}}^{2r_1}(\overline{M}_{n_1}^{\bullet}, O_{\lambda}(r_1))_{\mathfrak{m}_1}$.

Part (5) is a direct consequence of (2).

Part (6) follows from (1), Lemma 6.1.10, and Lemma 5.11.3(3-5).

Remark 7.2.5. In fact, Lemma 7.2.4(5) holds under only Assumption 7.2.2; and Lemma 7.2.4(6) holds under only Assumption 7.1.1.

Lemma 7.2.4(5) induces a coboundary map

$$\mathrm{AJ}_{\mathbf{Q}} \colon \mathrm{Z}^{n}_{\mathfrak{T}}(\mathbf{Q}^{\eta}) \to \mathrm{H}^{1}(\mathbb{Q}_{p^{2}}, \mathrm{H}^{2n-1}_{\mathfrak{T}}(\overline{\mathbf{Q}}, \mathrm{R}\Psi O_{\lambda}(n))_{(\mathfrak{m}_{0},\mathfrak{m}_{1})}).$$

We also recall the singular quotient map

$$\partial \colon \mathrm{H}^{1}(\mathbb{Q}_{p^{2}}, \mathrm{H}^{2n-1}_{\mathfrak{T}}(\overline{\mathrm{Q}}, \mathrm{R}\Psi O_{\lambda}(n))_{(\mathfrak{m}_{0},\mathfrak{m}_{1})}) \to \mathrm{H}^{1}_{\mathrm{sing}}(\mathbb{Q}_{p^{2}}, \mathrm{H}^{2n-1}_{\mathfrak{T}}(\overline{\mathrm{Q}}, \mathrm{R}\Psi O_{\lambda}(n))_{(\mathfrak{m}_{0},\mathfrak{m}_{1})}).$$

By our choice of K_n° and $(K_{sp}^{\circ}, K_{n+1}^{\circ})$, we obtain a morphism

$$\mathbf{M}_{\mathfrak{p}}(\mathbf{V}_{n}^{\circ},\mathbf{K}_{\mathrm{sp}}^{\circ})\to\mathbf{P}$$

which is finite. Denote by \mathbf{P}_{sp} the corresponding cycle; and let \mathbf{Q}_{sp} be the strict transform of \mathbf{P}_{sp} under σ , which is a $\mathbf{T}_{\mathfrak{p}}$ -invariant cycle of \mathbf{Q} . Our main goal is to compute $\partial \operatorname{AJ}_{\mathbf{Q}}(\mathbf{Q}_{sp}^{\eta})$ in $\operatorname{H}^{1}_{sing}(\mathbb{Q}_{p^{2}}, \operatorname{H}^{2n-1}_{\mathfrak{T}}(\overline{\mathbb{Q}}, \mathbb{R}\Psi O_{\lambda}(n))/(\mathfrak{n}_{0}, \mathfrak{n}_{1}))$. Recall the map Δ^{n} (5.28); the cycle Q_{sp} gives rise to a class $\operatorname{cl}(\operatorname{Q}_{sp}) \in C^{n}(\mathbb{Q}, L)$ (see Subsection 5.11 for the target).

Proposition 7.2.6. Assume Assumptions 7.1.1, 7.2.3, 7.2.2, and Hypothesis 3.2.9 for both n and n+1. There is a canonical isomorphism

$$\mathrm{H}^{1}_{\mathrm{sing}}(\mathbb{Q}_{p^{2}},\mathrm{H}^{2n-1}_{\mathfrak{T}}(\overline{\mathbb{Q}},\mathrm{R}\Psi O_{\lambda}(n))_{(\mathfrak{m}_{0},\mathfrak{m}_{1})})\simeq\operatorname{coker}\Delta^{n}_{(\mathfrak{m}_{0},\mathfrak{m}_{1})}$$

under which $\partial \operatorname{AJ}_{\mathbf{Q}}(\mathbf{Q}_{\operatorname{sp}}^{\eta})$ coincides with the image of $\operatorname{cl}(\mathbf{Q}_{\operatorname{sp}})$ in $\operatorname{coker} \Delta_{(\mathfrak{m}_0,\mathfrak{m}_1)}^n$.

Proof. By [Liu19, Theorem 2.16 and Theorem 2.18],¹⁹ it suffices to show that O_{λ} is a very nice coefficient for $\mathbb{E}_{s,(\mathfrak{m}_{0},\mathfrak{m}_{1})}^{p,q}$ in the sense of [Liu19, Definition 2.15]. In fact, in [Liu19, Definition 2.15], (N1) is satisfied due to Lemma 7.2.4(2); (N2) is satisfied due to Lemma 7.2.4(3); and (N3) is satisfied due to Lemma 7.2.4(4) and Lemma 2.1.4(2).

The proposition is proved.

By Construction 5.11.7 and Remark 5.11.8, we have a map

$$\nabla \colon C^{n}(\mathbf{Q}, O_{\lambda}) \to O_{\lambda}[\mathrm{Sh}(\mathbf{V}_{n_{0}}^{\circ}, \mathbf{K}_{n_{0}}^{\circ})] \otimes_{O_{\lambda}} O_{\lambda}[\mathrm{Sh}(\mathbf{V}_{n_{1}}^{\circ}, \mathbf{K}_{n_{1}}^{\circ})]$$

Theorem 7.2.7 (First explicit reciprocity law). Assume Assumptions 7.1.1, 7.2.3, 7.2.2, and Hypothesis 3.2.9 for both n and n + 1.

- (1) The image of the composite map $\nabla_{(\mathfrak{m}_0,\mathfrak{m}_1)} \circ \Delta^n_{(\mathfrak{m}_0,\mathfrak{m}_1)}$ is contained in $\mathfrak{n}_0.O_{\lambda}[Sh(V^{\circ}_{n_0}, K^{\circ}_{n_0})]_{\mathfrak{m}_0} \otimes_{O_{\lambda}} O_{\lambda}[Sh(V^{\circ}_{n_1}, K^{\circ}_{n_1})]_{\mathfrak{m}_1}.$
- (2) In view of (1), the induced map

$$\nabla_{\mathfrak{m}_1/\mathfrak{n}_0} \colon \operatorname{coker} \Delta^n_{\mathfrak{m}_1}/\mathfrak{n}_0 \to O_{\lambda}[\operatorname{Sh}(\operatorname{V}_{n_0}^\circ, \operatorname{K}_{n_0}^\circ)]/\mathfrak{n}_0 \otimes_{O_{\lambda}} O_{\lambda}[\operatorname{Sh}(\operatorname{V}_{n_1}^\circ, \operatorname{K}_{n_1}^\circ)]_{\mathfrak{m}_1}$$

is an isomorphism.

(3) Under the natural pairing

 $O_{\lambda}[\operatorname{Sh}(\operatorname{V}_{n_{0}}^{\circ},\operatorname{K}_{n_{0}}^{\circ})]/\mathfrak{n}_{0}\otimes_{O_{\lambda}}O_{\lambda}[\operatorname{Sh}(\operatorname{V}_{n_{1}}^{\circ},\operatorname{K}_{n_{1}}^{\circ})]_{\mathfrak{m}_{1}}\times O_{\lambda}/\lambda^{m}[\operatorname{Sh}(\operatorname{V}_{n_{0}}^{\circ},\operatorname{K}_{n_{0}}^{\circ})][\mathfrak{n}_{0}]\otimes_{O_{\lambda}}O_{\lambda}[\operatorname{Sh}(\operatorname{V}_{n_{1}}^{\circ},\operatorname{K}_{n_{1}}^{\circ})]_{\mathfrak{m}_{1}}\to O_{\lambda}/\lambda^{m}[\operatorname{Sh}(\operatorname{V}_{n_{0}}^{\circ},\operatorname{K}_{n_{0}}^{\circ})][\mathfrak{n}_{0}]\otimes_{O_{\lambda}}O_{\lambda}[\operatorname{Sh}(\operatorname{V}_{n_{1}}^{\circ},\operatorname{K}_{n_{1}}^{\circ})]_{\mathfrak{m}_{1}}\to O_{\lambda}/\lambda^{m}[\operatorname{Sh}(\operatorname{V}_{n_{0}}^{\circ},\operatorname{K}_{n_{0}}^{\circ})][\mathfrak{n}_{0}]\otimes_{O_{\lambda}}O_{\lambda}[\operatorname{Sh}(\operatorname{V}_{n_{1}}^{\circ},\operatorname{K}_{n_{1}}^{\circ})]_{\mathfrak{m}_{1}}\to O_{\lambda}/\lambda^{m}[\operatorname{Sh}(\operatorname{V}_{n_{0}}^{\circ},\operatorname{K}_{n_{0}}^{\circ})][\mathfrak{n}_{0}]\otimes_{O_{\lambda}}O_{\lambda}[\operatorname{Sh}(\operatorname{V}_{n_{1}}^{\circ},\operatorname{K}_{n_{1}}^{\circ})]_{\mathfrak{m}_{1}}\to O_{\lambda}/\lambda^{m}[\operatorname{Sh}(\operatorname{V}_{n_{0}}^{\circ},\operatorname{K}_{n_{0}}^{\circ})][\mathfrak{n}_{0}]\otimes_{O_{\lambda}}O_{\lambda}[\operatorname{Sh}(\operatorname{V}_{n_{1}}^{\circ},\operatorname{K}_{n_{1}}^{\circ})]_{\mathfrak{m}_{1}}\to O_{\lambda}/\lambda^{m}[\operatorname{Sh}(\operatorname{V}_{n_{0}}^{\circ},\operatorname{K}_{n_{0}}^{\circ})][\mathfrak{n}_{0}]\otimes_{O_{\lambda}}O_{\lambda}[\operatorname{Sh}(\operatorname{V}_{n_{1}}^{\circ},\operatorname{K}_{n_{1}}^{\circ})]_{\mathfrak{m}_{{0}}}\to O_{\lambda}/\lambda^{m}[\operatorname{Sh}(\operatorname{V}_{n_{0}}^{\circ},\operatorname{K}_{n_{0}}^{\circ})][\mathfrak{n}_{0}]\otimes_{O_{\lambda}}O_{\lambda}[\operatorname{Sh}(\operatorname{V}_{n_{0}}^{\circ},\operatorname{K}_{n_{0}}^{\circ})]]_{\mathfrak{m}_{{0}}}\to O_{\lambda}/\lambda^{m}[\operatorname{Sh}(\operatorname{V}_{n_{0}}^{\circ},\operatorname{K}_{n_{0}}^{\circ})][\mathfrak{n}_{0}]\otimes_{O_{\lambda}}O_{\lambda}[\operatorname{Sh}(\operatorname{V}_{n_{0}}^{\circ},\operatorname{K}_{n_{0}}^{\circ})]]]_{\mathfrak{m}_{{0}}}\to O_{\lambda}/\lambda^{m}[\operatorname{Sh}(\operatorname{V}_{n_{0}}^{\circ},\operatorname{K}_{n_{0}}^{\circ})]]$

obtained by taking inner product, the pairing of $\nabla_{/(\mathfrak{n}_0,\mathfrak{n}_1)}(\partial \operatorname{AJ}_{\mathbf{Q}}(\mathbf{Q}^{\eta}_{\Delta}))$ and every function $f \in O_{\lambda}/\lambda^m[\operatorname{Sh}(\operatorname{V}^{\circ}_{n_0},\operatorname{K}^{\circ}_{n_0})][\mathfrak{n}_0] \otimes_{O_{\lambda}} O_{\lambda}/\lambda^m[\operatorname{Sh}(\operatorname{V}^{\circ}_{n_1},\operatorname{K}^{\circ}_{n_1})][\mathfrak{n}_1]$ is equal to

$$(p+1) \cdot \phi_{\Pi_0}(\mathbf{I}_{n_0,\mathfrak{p}}^{\circ}) \cdot \phi_{\Pi_1}(\mathbf{T}_{n_1,\mathfrak{p}}^{\circ}) \cdot \sum_{s \in \operatorname{Sh}(\mathbf{V}_n^{\circ},\mathbf{K}_{\operatorname{sp}}^{\circ})} f(s,\operatorname{sh}_{\uparrow}^{\circ}(s)).$$

Here, we regard $\partial AJ_{\mathbf{Q}}(\mathbf{Q}_{sp}^{\eta})$ as an element in coker $\Delta_{(\mathfrak{m}_0,\mathfrak{m}_1)}^n$ (hence in coker $\Delta_{\mathfrak{m}_1}^n/\mathfrak{n}_0$) via the canonical isomorphism in Proposition 7.2.6.

Proof. We first consider (1). By Lemma 5.11.3(3,4), we have

$$\begin{aligned} \mathbf{H}_{\mathfrak{T}}^{2(n-1)}(\overline{\mathbf{Q}}^{(0)},O_{\lambda}(n-1))_{(\mathfrak{m}_{0},\mathfrak{m}_{1})} &= \bigoplus_{\substack{(?_{0},?_{1})\in\{\circ,\bullet\}^{2}}} \sigma^{*}\mathbf{H}_{\mathfrak{T}}^{2(n-1)}(\overline{\mathbf{P}}^{?_{0},?_{1}},O_{\lambda}(n-1))_{(\mathfrak{m}_{0},\mathfrak{m}_{1})} \\ & \bigoplus_{\substack{(\delta_{\circ,\circ}^{\dagger,\dagger}):\sigma^{*}\mathbf{H}_{\mathfrak{T}}^{2(n-2)}(\overline{\mathbf{P}}^{\dagger,\dagger},O_{\lambda}(n-2))_{(\mathfrak{m}_{0},\mathfrak{m}_{1})} \bigoplus_{\substack{(\delta_{\bullet,\bullet}^{\dagger,\dagger}):\sigma^{*}\mathbf{H}_{\mathfrak{T}}^{2(n-2)}(\overline{\mathbf{P}}^{\dagger,\dagger},O_{\lambda}(n-2))_{(\mathfrak{m}_{0},\mathfrak{m}_{1})}. \end{aligned}$$

Thus, it suffices to show that

(1a) The image of $\sigma^* \mathcal{H}^{2(n-1)}_{\mathfrak{T}}(\overline{\mathcal{P}}^{\circ,\bullet}, O_{\lambda}(n-1))_{(\mathfrak{m}_0,\mathfrak{m}_1)} \bigoplus \sigma^* \mathcal{H}^{2(n-1)}_{\mathfrak{T}}(\overline{\mathcal{P}}^{\bullet,\bullet}, O_{\lambda}(n-1))_{(\mathfrak{m}_0,\mathfrak{m}_1)}$ under the map $(\nabla \circ \delta_{1!} \circ \delta_0^*)_{(\mathfrak{m}_0,\mathfrak{m}_1)}$ is contained in $\mathfrak{n}_0.O_{\lambda}[Sh(\mathcal{V}_{n_0}^{\circ}, \mathcal{K}_{n_0}^{\circ})]_{\mathfrak{m}_0} \otimes_{O_{\lambda}} O_{\lambda}[Sh(\mathcal{V}_{n_1}^{\circ}, \mathcal{K}_{n_1}^{\circ})]_{\mathfrak{m}_1}.$

¹⁹Although it is assumed that the underlying strictly semistable scheme X is proper over the base in [Liu19], the proof of relevant results works without change in our case even when \mathbf{Q} is not proper in view of Lemma 7.2.4(6).

- (1b) The image of $\sigma^* \mathcal{H}^{2(n-1)}_{\mathfrak{T}}(\overline{\mathcal{P}}^{\circ,\circ}, O_{\lambda}(n-1))_{(\mathfrak{m}_0,\mathfrak{m}_1)} \oplus \sigma^* \mathcal{H}^{2(n-1)}_{\mathfrak{T}}(\overline{\mathcal{P}}^{\bullet,\circ}, O_{\lambda}(n-1))_{(\mathfrak{m}_0,\mathfrak{m}_1)}$ under the map $(\nabla \circ \delta_{1!} \circ \delta_0^*)_{(\mathfrak{m}_0,\mathfrak{m}_1)}$ is zero.
- (1c) The image of $(\delta_{0,0}^{\dagger,\dagger})!\sigma^* \mathcal{H}_{\mathfrak{T}}^{2(n-2)}(\overline{\mathcal{P}}^{\dagger,\dagger}, O_{\lambda}(n-2))_{(\mathfrak{m}_0,\mathfrak{m}_1)}$ under the map $(\nabla \circ \delta_{1!} \circ \delta_0^*)_{(\mathfrak{m}_0,\mathfrak{m}_1)}$ is zero.
- (1d) The image of $(\delta_{\bullet,\bullet}^{\dagger,\dagger})_! \sigma^* \mathcal{H}^{2(n-2)}_{\mathfrak{T}}(\overline{\mathcal{P}}^{\dagger,\dagger}, O_{\lambda}(n-2))_{(\mathfrak{m}_0,\mathfrak{m}_1)}$ under the map $(\nabla \circ \delta_{1!} \circ \delta_0^*)_{(\mathfrak{m}_0,\mathfrak{m}_1)}$ is zero.
- For (1a), we have a commutative diagram

in which

O the upper horizontal arrow is the map

$$\begin{aligned} & \mathrm{H}^{2(n-1)}_{\mathfrak{T}}(\overline{\mathrm{P}}^{\circ,\bullet},O_{\lambda}(n-1))_{(\mathfrak{m}_{0},\mathfrak{m}_{1})} \bigoplus \mathrm{H}^{2(n-1)}_{\mathfrak{T}}(\overline{\mathrm{P}}^{\bullet,\bullet},O_{\lambda}(n-1))_{(\mathfrak{m}_{0},\mathfrak{m}_{1})} \\ & \to \mathrm{H}^{2(r_{0}-1)}_{\mathfrak{T}}(\overline{\mathrm{M}}^{\circ}_{n_{0}},O_{\lambda}(r_{0}-1))_{\mathfrak{m}_{0}} \otimes_{O_{\lambda}} \mathrm{H}^{2r_{1}}_{\mathfrak{T}}(\overline{\mathrm{M}}^{\bullet}_{n_{1}},O_{\lambda}(r_{1}))_{\mathfrak{m}_{1}} \\ & \bigoplus \mathrm{H}^{2(r_{0}-1)}_{\mathfrak{T}}(\overline{\mathrm{M}}^{\bullet}_{n_{0}},O_{\lambda}(r_{0}-1))_{\mathfrak{m}_{0}} \otimes_{O_{\lambda}} \mathrm{H}^{2r_{1}}_{\mathfrak{T}}(\overline{\mathrm{M}}^{\bullet}_{n_{1}},O_{\lambda}(r_{1}))_{\mathfrak{m}_{1}} \\ & = {}^{0}\mathrm{E}^{0,2r_{0}-2}_{1,\mathfrak{m}_{0}} \otimes_{O_{\lambda}} \mathrm{H}^{2r_{1}}_{\mathfrak{T}}(\overline{\mathrm{M}}^{\bullet}_{n_{1}},O_{\lambda}(r_{1}))_{\mathfrak{m}_{1}} \end{aligned}$$

given by Lemma 7.2.4(1) and the Künneth formula;

O the right vertical arrow is

$$(\nabla^0 \circ {}^0\mathrm{d}_1^{-1,2r_0} \circ {}^0\mathrm{d}_1^{0,2r_0-2}(-1))_{\mathfrak{m}_0} \otimes (\mathrm{I}_{n_1,\mathfrak{p}}^\circ \circ \mathrm{inc}_{\dagger}^* + (p+1)^2 \mathrm{T}_{n_1,\mathfrak{p}}^{\circ \bullet} \circ \mathrm{inc}_{\bullet}^*)_{\mathfrak{m}_1};$$

and

O the lower horizontal arrow is $(\nabla \circ \delta_{1!} \circ \delta_0^*)_{(\mathfrak{m}_0,\mathfrak{m}_1)}$.

For (1a), by Proposition B.3.5(2) and (PI4), we have

$$((p+1)\mathfrak{R}^{\circ}_{N,\mathfrak{p}}-\mathfrak{I}^{\circ}_{N,\mathfrak{p}}).O_{\lambda}[\mathrm{Sh}(\mathrm{V}^{\circ}_{n_{0}},\mathrm{K}^{\circ}_{n_{0}})]_{\mathfrak{m}_{0}}\subseteq\mathfrak{n}_{0}.O_{\lambda}[\mathrm{Sh}(\mathrm{V}^{\circ}_{n_{0}},\mathrm{K}^{\circ}_{n_{0}})]_{\mathfrak{m}_{0}}.$$

Thus, (1a) follows from Proposition 6.3.1(4) and Lemma 5.11.3(3).

For (1b) and (1c), both images are actually contained in the sum of

$$(\mathbf{I}_{n_{1},\mathfrak{p}}^{\circ}\circ\mathrm{inc}_{\circ,\dagger}^{*}+(p+1)^{2}\mathbf{T}_{n_{1},\mathfrak{p}}^{\circ\bullet}\circ\mathrm{inc}_{\circ,\bullet}^{*})(\gamma_{\circ,\bullet}^{\circ,\dagger})_{!}\mathbf{H}_{\mathfrak{T}}^{2(n-1)}(\overline{\mathbf{P}}^{\circ,\dagger},O_{\lambda}(n-1))_{(\mathfrak{m}_{0},\mathfrak{m}_{1})}$$

and

$$(\mathbf{I}_{n_{1},\mathfrak{p}}^{\circ}\circ\mathrm{inc}_{\circ,\dagger}^{*}+(p+1)^{2}\mathbf{T}_{n_{1},\mathfrak{p}}^{\circ\bullet}\circ\mathrm{inc}_{\bullet,\bullet}^{*})(\gamma_{\bullet,\bullet}^{\bullet,\dagger})_{!}\mathbf{H}_{\mathfrak{T}}^{2(n-1)}(\overline{\mathbf{P}}^{\bullet,\dagger},O_{\lambda}(n-1))_{(\mathfrak{m}_{0},\mathfrak{m}_{1})},$$

which by Lemma 7.2.4(1) coincide with

 $\mathrm{H}^{2r_{0}}_{\mathfrak{T}}(\overline{\mathrm{M}}^{\circ}_{n_{0}},O_{\lambda}(r_{0}))_{\mathfrak{m}_{0}}\otimes_{O_{\lambda}}\left((\mathrm{I}^{\circ}_{n_{1},\mathfrak{p}}\circ\mathrm{Inc}^{*}_{\dagger}+(p+1)^{2}\mathrm{T}^{\circ\bullet}_{n_{1},\mathfrak{p}}\circ\mathrm{Inc}^{*}_{\bullet})^{1}\mathrm{d}^{-1,2r_{1}}_{1}\mathrm{H}^{2(r_{1}-1)}_{\mathfrak{T}}(\overline{\mathrm{M}}^{\dagger}_{n_{1}},O_{\lambda}(r_{1}-1))_{\mathfrak{m}_{1}}\right)$ and

$$\mathbf{H}_{\mathfrak{T}}^{2r_{0}}(\overline{\mathbf{M}}_{n_{0}}^{\bullet},O_{\lambda}(r_{0}))_{\mathfrak{m}_{0}}\otimes_{O_{\lambda}}\left((\mathbf{I}_{n_{1},\mathfrak{p}}^{\circ}\circ\operatorname{Inc}_{\dagger}^{*}+(p+1)^{2}\mathbf{T}_{n_{1},\mathfrak{p}}^{\circ\circ}\circ\operatorname{Inc}_{\bullet}^{*})^{1}\mathbf{d}_{1}^{-1,2r_{1}}\mathbf{H}_{\mathfrak{T}}^{2(r_{1}-1)}(\overline{\mathbf{M}}_{n_{1}}^{\dagger},O_{\lambda}(r_{1}-1))_{\mathfrak{m}_{1}}\right),$$

respectively. However, they vanish by Lemma 5.8.2(3). Thus, (1b) and (1c) follow.

For (1d), by [Liu19, Lemma 2.4], it follows from (1c). Thus, (1) is proved.

Now we consider (2). We claim that the map $\nabla_{(\mathfrak{m}_0,\mathfrak{m}_1)}$ (with domain $C^n(\mathbf{Q},O_\lambda)_{(\mathfrak{m}_0,\mathfrak{m}_1)}$) is surjective. In fact, consider the submodule

$$\ker {}^{0}\mathrm{d}^{0,2r_{0}}_{1,\mathfrak{m}_{0}} \otimes_{O_{\lambda}} \ker {}^{1}\mathrm{d}^{0,2r_{1}}_{1,\mathfrak{m}_{1}} \subseteq \bigoplus_{(?_{0},?_{1}) \in \{\circ,\bullet\}^{2}} \mathrm{H}^{2(n-1)}_{\mathfrak{T}}(\overline{\mathrm{P}}^{?_{0},?_{1}},O_{\lambda}(n-1))_{(\mathfrak{m}_{0},\mathfrak{m}_{1})}$$

in view of Lemma 7.2.4(1). Then $\sigma^* \left(\ker^0 d_{1,\mathfrak{m}_0}^{0,2r_0} \otimes_{O_\lambda} \ker^1 d_{1,\mathfrak{m}_1}^{0,2r_1} \right)$ is contained in $C^n(\mathbf{Q}, O_\lambda)_{(\mathfrak{m}_0,\mathfrak{m}_1)}$. On the other hand, the map $\nabla_{(\mathfrak{m}_0,\mathfrak{m}_1)} \circ \sigma^*$ (with domain $\ker^0 d_{1,\mathfrak{m}_0}^{0,2r_0} \otimes_{O_\lambda} \ker^1 d_{1,\mathfrak{m}_1}^{0,2r_1}$) coincides with $\nabla_{\mathfrak{m}_0}^{\mathfrak{m}_0} \otimes \nabla_{\mathfrak{m}_1}^{\mathfrak{m}_1}$, which is surjective by Proposition 6.3.1(3) and Theorem 6.2.3. The claim follows.

Thus, it remains to show that the domain and the target of $\nabla_{\mathfrak{m}_1/\mathfrak{n}_0}$ have the same cardinality. By Proposition 7.2.6, we have an isomorphism

(7.1)
$$\operatorname{coker} \Delta^{n}_{\mathfrak{m}_{1}}/\mathfrak{n}_{0} = \operatorname{coker} \Delta^{n}_{(\mathfrak{m}_{0},\mathfrak{m}_{1})}/\mathfrak{n}_{0} \simeq \mathrm{H}^{1}_{\mathrm{sing}}(\mathbb{Q}_{p^{2}}, \mathrm{H}^{2n-1}_{\mathfrak{T}}(\overline{\mathrm{Q}}, \mathrm{R}\Psi O_{\lambda}(n))_{(\mathfrak{m}_{0},\mathfrak{m}_{1})})/\mathfrak{n}_{0}$$

of O_{λ}/λ^m -modules. By Lemma 7.2.4(2,3) and Theorem 6.2.3(2), we have

$$\mathrm{H}^{1}_{\mathrm{sing}}(\mathbb{Q}_{p^{2}},\mathrm{H}^{2n-1}_{\mathfrak{T}}(\overline{\mathrm{Q}},\mathrm{R}\Psi O_{\lambda}(n))_{(\mathfrak{m}_{0},\mathfrak{m}_{1})})\simeq\mathrm{H}^{1}_{\mathrm{sing}}(\mathbb{Q}_{p^{2}},\mathrm{H}^{2r_{0}-1}_{\mathfrak{T}}(\overline{\mathrm{M}}_{n_{0}},\mathrm{R}\Psi O_{\lambda}(r_{0}))_{\mathfrak{m}_{0}})\otimes_{O_{\lambda}}({}^{1}\mathrm{E}^{0,2r_{1}}_{2,\mathfrak{m}_{1}})^{\mathrm{Gal}(\overline{\mathbb{F}}_{p}/\mathbb{F}_{p^{2}})}.$$

Then by Theorem 6.2.3(3) and Theorem 6.3.4(4), we have

(

$$7.1) \simeq O_{\lambda}[\operatorname{Sh}(\operatorname{V}_{n_0}^{\circ}, \operatorname{K}_{n_0}^{\circ})]/\mathfrak{n}_0 \otimes_{O_{\lambda}} O_{\lambda}[\operatorname{Sh}(\operatorname{V}_{n_1}^{\circ}, \operatorname{K}_{n_1}^{\circ})]_{\mathfrak{m}_1}]$$

Thus, (2) is proved.

Finally we consider (3). As Q_{sp} does not intersect with $Q^{\circ \bullet}$, we have

$$\operatorname{cl}(\mathbf{Q}_{\operatorname{sp}}) = \operatorname{cl}(\mathbf{Q}^{\bullet}_{\Delta}) \in \mathrm{H}^{2n}_{\mathfrak{T}}(\overline{\mathbf{Q}}^{\bullet,\bullet}, O_{\lambda}(n)),$$

and by Construction 5.11.7,

$$\nabla(\mathrm{cl}(\mathbf{Q}_{\mathrm{sp}})) = \left((p+1)(\mathsf{T}_{n_0,\mathfrak{p}}^{\circ\bullet} \otimes \mathsf{I}_{n_1,\mathfrak{p}}^{\circ}) \circ \mathrm{inc}_{\bullet,\dagger}^* + (p+1)^3 (\mathsf{T}_{n_0,\mathfrak{p}}^{\circ\bullet} \otimes \mathsf{T}_{n_1,\mathfrak{p}}^{\circ\bullet}) \circ \mathrm{inc}_{\bullet,\bullet}^* \right) (\mathrm{cl}(\mathbf{P}_{\mathrm{sp}}^{\bullet})).$$

Applying Theorem 5.11.5(3) to the object $(K_{sp}^{\circ}, K_{n+1}^{\circ}) \in \mathfrak{K}(V_n^{\circ})_{sp}$ followed by pushforward, we know that the pairing between $\nabla_{\mathfrak{m}_1/\mathfrak{n}_0}(\mathrm{cl}(Q_{sp}))$ and any function

$$f \in O_{\lambda}/\lambda^{m}[\operatorname{Sh}(\operatorname{V}_{n_{0}}^{\circ},\operatorname{K}_{n_{0}}^{\circ})][\mathfrak{n}_{0}] \otimes_{O_{\lambda}} O_{\lambda}/\lambda^{m}[\operatorname{Sh}(\operatorname{V}_{n_{1}}^{\circ},\operatorname{K}_{n_{1}}^{\circ})][\mathfrak{n}_{1}]$$

is given by the formula

$$(p+1) \cdot \phi_{\Pi_0}(\mathtt{I}_{n_0,\mathfrak{p}}^{\circ}) \cdot \phi_{\Pi_1}(\mathtt{T}_{n_1,\mathfrak{p}}^{\circ}) \cdot \sum_{s \in \operatorname{Sh}(\mathrm{V}_n^{\circ},\mathrm{K}_{\operatorname{sp}}^{\circ})} f(s, \operatorname{sh}^{\circ}_{\uparrow}(s))$$

in view of (PI6). We then obtain (3) by Proposition 7.2.6.

The theorem is proved.

We state a corollary for later application. We choose an indefinite uniformization datum as in Notation 5.10.1, and put $\mathrm{Sh}'_{n_{\alpha}} \coloneqq \mathrm{Sh}(\mathrm{V}'_{n_{\alpha}}, \mathfrak{j}_{n_{\alpha}}\mathrm{K}^{p\circ}_{n_{\alpha}}\mathrm{K}'_{n_{\alpha},p})$ for $\alpha = 0, 1$.

Assume Assumptions 7.1.1, 7.2.2. By Lemma 6.1.10, Lemma 5.1.7, and the Künneth formula, we have $\mathrm{H}^{i}_{\mathrm{\acute{e}t}}((\mathrm{Sh}'_{n_{0}} \times_{\mathrm{Spec}\,F} \mathrm{Sh}'_{n_{1}})_{\overline{F}}, O_{\lambda})_{(\mathfrak{m}_{0},\mathfrak{m}_{1})} = 0$ if $i \neq 2n-1$. In particular, we obtain the Abel–Jacobi map

$$\mathrm{AJ}\colon \mathrm{Z}^{n}(\mathrm{Sh}_{n_{0}}^{\prime}\times_{\mathrm{Spec}\,F}\mathrm{Sh}_{n_{1}}^{\prime})\to\mathrm{H}^{1}(F,\mathrm{H}^{2n-1}((\mathrm{Sh}_{n_{0}}^{\prime}\times_{\mathrm{Spec}\,F}\mathrm{Sh}_{n_{1}}^{\prime})_{\overline{F}},O_{\lambda}(n))/(\mathfrak{n}_{0},\mathfrak{n}_{1})).$$

Let $\operatorname{Sh}'_{\operatorname{sp}}$ be the cycle given by the finite morphism $\operatorname{Sh}(V'_n, j_n \operatorname{K}^{p\circ}_{\operatorname{sp}} \operatorname{K}'_{n,p}) \to \operatorname{Sh}'_n \times_{\operatorname{Spec} F} \operatorname{Sh}'_{n+1}$, which is an element in $\operatorname{Z}^n(\operatorname{Sh}'_{n_0} \times_{\operatorname{Spec} F} \operatorname{Sh}'_{n_1})$.

Corollary 7.2.8. Assume Assumptions 7.1.1, 7.2.3, 7.2.2, and Hypothesis 3.2.9 for both n and n+1. Then we have

$$\exp_{\lambda} \left(\partial_{\mathfrak{p}} \mathrm{loc}_{\mathfrak{p}} \operatorname{AJ}(\mathrm{Sh}'_{\mathrm{sp}}), \operatorname{H}^{1}_{\mathrm{sing}}(F_{\mathfrak{p}}, \operatorname{H}^{2n-1}((\operatorname{Sh}'_{n_{0}} \times_{\operatorname{Spec} F} \operatorname{Sh}'_{n_{1}})_{\overline{F}}, O_{\lambda}(n)) / (\mathfrak{n}_{0}, \mathfrak{n}_{1})) \right)$$
$$= \exp_{\lambda} \left(\mathbbm{1}_{\operatorname{Sh}(\operatorname{V}^{\circ}_{n}, \operatorname{K}^{\circ}_{\mathrm{sp}})}, O_{\lambda}[\operatorname{Sh}(\operatorname{V}^{\circ}_{n_{0}}, \operatorname{K}^{\circ}_{n_{0}}) \times \operatorname{Sh}(\operatorname{V}^{\circ}_{n_{1}}, \operatorname{K}^{\circ}_{n_{1}})] / (\mathfrak{n}_{0}, \mathfrak{n}_{1}) \right)$$

where \exp_{λ} is introduced in Definition 2.1.6. Here, we regard $\mathbb{1}_{\operatorname{Sh}(\operatorname{V}_{n}^{\circ},\operatorname{K}_{\operatorname{sp}}^{\circ})}$ as the pushforward of the characteristic function along the map $\operatorname{Sh}(\operatorname{V}_{n}^{\circ},\operatorname{K}_{\operatorname{sp}}^{\circ}) \to \operatorname{Sh}(\operatorname{V}_{n}^{\circ},\operatorname{K}_{n}^{\circ}) \times \operatorname{Sh}(\operatorname{V}_{n+1}^{\circ},\operatorname{K}_{n+1}^{\circ}).$

Proof. Note that the isomorphism (5.2) induces a map

$$\mathrm{H}^{2n-1}((\mathrm{Sh}'_{n_0} \times_{\mathrm{Spec}\,F} \mathrm{Sh}'_{n_1})_{\overline{F}}, O_{\lambda}(n))_{(\mathfrak{m}_0,\mathfrak{m}_1)} \to \mathrm{H}^{2n-1}_{\mathfrak{T}}(\overline{\mathrm{Q}}, \mathrm{R}\Psi O_{\lambda}(n))_{(\mathfrak{m}_0,\mathfrak{m}_1)}$$

of $O_{\lambda}[\operatorname{Gal}(\overline{\mathbb{Q}}_p/\mathbb{Q}_{p^2})]$ -modules, which is an isomorphism by Lemma 5.11.2. Combining with the diagram (5.23), we have

$$\exp_{\lambda} \left(\partial_{\mathfrak{p}} \mathrm{loc}_{\mathfrak{p}} \operatorname{AJ}(\mathrm{Sh}'_{\mathrm{sp}}), \operatorname{H}^{1}_{\mathrm{sing}}(F_{\mathfrak{p}}, \operatorname{H}^{2n-1}((\operatorname{Sh}'_{n_{0}} \times_{\operatorname{Spec} F} \operatorname{Sh}'_{n_{1}})_{\overline{F}}, O_{\lambda}(n)) / (\mathfrak{n}_{0}, \mathfrak{n}_{1})) \right)$$
$$= \exp_{\lambda} \left(\partial \operatorname{AJ}_{\mathbf{Q}}(\mathbf{Q}^{\eta}_{\mathrm{sp}}), \operatorname{H}^{1}_{\mathrm{sing}}(\mathbb{Q}_{p^{2}}, \operatorname{H}^{2n-1}_{\mathfrak{T}}(\overline{\mathbb{Q}}, \operatorname{R}\Psi O_{\lambda}(n)) / (\mathfrak{n}_{0}, \mathfrak{n}_{1})) \right).$$

Now Theorem 7.2.7 implies

=

$$\exp_{\lambda} \left(\partial \operatorname{AJ}_{\mathbf{Q}}(\mathbf{Q}_{\operatorname{sp}}^{\eta}), \operatorname{H}^{1}_{\operatorname{sing}}(\mathbb{Q}_{p^{2}}, \operatorname{H}^{2n-1}_{\mathfrak{T}}(\overline{\mathbb{Q}}, \operatorname{R}\Psi O_{\lambda}(n))/(\mathfrak{n}_{0}, \mathfrak{n}_{1})) \right)$$

$$= \exp_{\lambda} \left((p+1)\phi_{\Pi_{0}}(\mathfrak{l}_{n_{0}, \mathfrak{p}}^{\circ})\phi_{\Pi_{1}}(\mathfrak{T}_{n_{1}, \mathfrak{p}}^{\circ}) \mathbb{1}_{\operatorname{Sh}(\operatorname{V}_{n}^{\circ}, \operatorname{K}_{\operatorname{sp}}^{\circ})}, O_{\lambda}[\operatorname{Sh}(\operatorname{V}_{n_{0}}^{\circ}, \operatorname{K}_{n_{0}}^{\circ})]/\mathfrak{n}_{0} \otimes_{O_{\lambda}} O_{\lambda}[\operatorname{Sh}(\operatorname{V}_{n_{1}}^{\circ}, \operatorname{K}_{n_{1}}^{\circ})]/\mathfrak{n}_{1} \right).$$

Note that (p+1) is invertible in O_{λ} by (PI2); $\phi_{\Pi_0}(\mathfrak{I}_{n_0,\mathfrak{p}}^\circ)$ is invertible in O_{λ} by (PI5) and Proposition B.3.5(1); and $\phi_{\Pi_1}(T^{\circ}_{n_1,\mathfrak{p}})$ is invertible in O_{λ} by (PI4) and Proposition B.4.3(2). Thus, the corollary follows.

7.3. Second explicit reciprocity law. We start by choosing

- O a finite place λ of E (with the underlying rational prime ℓ),
- **O** a positive integer m,
- O a (possibly empty) finite set $\Sigma^+_{lr,II}$ of nonarchimedean places of F^+ that are inert in F, strongly disjoint from Σ_{\min}^+ (Definition 1.3.3), satisfying $\ell \nmid ||v|| (||v||^2 - 1)$ for $v \in \Sigma_{lr,II}^+$,
- **O** a finite set Σ_{II}^+ of nonarchimedean places of F^+ containing $\Sigma_{\min}^+ \cup \Sigma_{\text{lr,II}}^+$,
- O a standard indefinite hermitian space V_n of rank n over F, together with a self-dual $\prod_{v \notin \Sigma_{\infty}^{+} \cup \Sigma_{\min}^{+} \cup \Sigma_{\Gamma,\Pi}^{+}} O_{F_{v}} \text{-lattice } \Lambda_{n} \text{ in } V_{n} \otimes_{F} \mathbb{A}_{F}^{\Sigma_{\infty}^{+} \cup \Sigma_{\min}^{+} \cup \Sigma_{\Gamma,\Pi}^{+}} \text{ (and put } V_{n+1} \coloneqq (V_{n})_{\sharp} \text{ and}$

 $\Lambda_{n+1} := (\Lambda_n)_{\sharp}$, satisfying that $(V_{n_0})_v$ is not split for $v \in \Sigma^+_{lr,II}$,

O an object $K_n \in \mathfrak{K}(V_n)$ and an object $(K_{sp}, K_{n+1}) \in \mathfrak{K}(V_n)_{sp}$ of the forms

$$K_{n} = \prod_{v \in \Sigma_{\min}^{+} \cup \Sigma_{\mathrm{lr,II}}^{+}} (K_{n})_{v} \times \prod_{v \notin \Sigma_{\infty}^{+} \cup \Sigma_{\min}^{+} \cup \Sigma_{\mathrm{lr,II}}^{+}} U(\Lambda_{n})(O_{F_{v}^{+}}),$$

$$K_{\mathrm{sp}} = \prod_{v \in \Sigma_{\min}^{+} \cup \Sigma_{\mathrm{lr,II}}^{+}} (K_{\mathrm{sp}})_{v} \times \prod_{v \notin \Sigma_{\infty}^{+} \cup \Sigma_{\min}^{+} \cup \Sigma_{\mathrm{lr,II}}^{+}} U(\Lambda_{n})(O_{F_{v}^{+}}),$$

$$K_{n+1} = \prod_{v \in \Sigma_{\min}^{+} \cup \Sigma_{\mathrm{lr,II}}^{+}} (K_{n+1})_{v} \times \prod_{v \notin \Sigma_{\infty}^{+} \cup \Sigma_{\min}^{+} \cup \Sigma_{\mathrm{lr,II}}^{+}} U(\Lambda_{n+1})(O_{F_{v}^{+}}).$$

satisfying $(K_{sp})_v = (K_n)_v$ for $v \in \Sigma_{min}^+$, $(K_{sp})_v \subseteq (K_n)_v$ for $v \in \Sigma_{lr,II}^+$, and that $(K_{n_0}^\circ)_v$ is a transferable open compact subgroup (Definition D.2.1) of $U(V_{n_0}^{\circ})(F_v^+)$ for $v \in \Sigma_{\min}^+$ and is a special maximal subgroup of $U(V_{n_0}^{\circ})(F_v^+)$ for $v \in \Sigma_{lr,II}^+$,

- O a special inert prime (Definition 3.3.4) \mathfrak{p} of F^+ (with the underlying rational prime p) satisfying
 - (PII1): Σ_{II}^+ does not contain *p*-adic places;
 - (PII2): ℓ does not divide $p(p^2 1)$;
 - (PII3): there exists a CM type Φ containing τ_{∞} as in the initial setup of Section 5 satisfying $\mathbb{Q}_p^{\Phi} = \mathbb{Q}_{p^2};$
 - (PII4): $P_{\alpha(\Pi_{0,\mathfrak{p}})} \mod \lambda^m$ is level-raising special at \mathfrak{p} (Definition 3.1.5);

 $P_{\boldsymbol{\alpha}(\Pi_{1,\mathfrak{p}})} \mod \lambda$ is Tate generic at \mathfrak{p} (Definition 3.1.5);

- (PII7): $P_{\alpha(\Pi_{0,\mathfrak{p}})\otimes\alpha(\Pi_{1,\mathfrak{p}})} \mod \lambda^m$ is level-raising special at \mathfrak{p} (Definition 3.1.5);
- (So we can and will adopt the setup in Subsection 4.4 to the datum $(V_n, \{\Lambda_{n,\mathfrak{q}}\}|_{\mathfrak{q}|p})$.)

- O remaining data in the initial setup of Section 4 with $\mathbb{Q}_p^{\Phi} = \mathbb{Q}_{p^2}$; and
- O a definite uniformization datum as in Notation 4.4.7.

Put $K_{sp}^{\star} \coloneqq (i_n K_{sp}^p) \times K_{n,p}^{\star}$, and $K_{n_{\alpha}}^{\star} \coloneqq (i_{n_{\alpha}} K_{n_{\alpha}}^p) \times K_{n_{\alpha},p}^{\star}$ for $\alpha = 0, 1$. Put $K_{sp,sp}^{\star} \coloneqq (i_n K_{sp}^p) \times K_{sp,p}^{\star}$ and $K_{n,sp}^{\star} \coloneqq (i_n K_n^p) \times K_{sp,p}^{\star}$. Like in Subsection 4.5, we put $X_{n_{\alpha}}^? \coloneqq X_p^?(V_{n_{\alpha}}, K_{n_{\alpha}}^p)$ for meaningful triples $(X, ?, \alpha) \in \{\mathbf{M}, \mathbf{M}, \mathbf{B}, \mathbf{S}\} \times \{, \eta\} \times \{0, 1\}$.

Notation 7.3.1. We introduce the following ideas of $\mathbb{T}_{n_{\alpha}}^{\Sigma_{\Pi}^+ \cup \Sigma_p^+}$, for $\alpha = 0, 1$

$$\begin{cases} \mathfrak{m}_{\alpha} \coloneqq \mathbb{T}_{n_{\alpha}}^{\Sigma_{\Pi}^{+} \cup \Sigma_{p}^{+}} \cap \ker \left(\mathbb{T}_{n_{\alpha}}^{\Sigma^{+}} \xrightarrow{\phi_{\Pi_{\alpha}}} O_{E} \to O_{E} / \lambda \right), \\ \mathfrak{n}_{\alpha} \coloneqq \mathbb{T}_{n_{\alpha}}^{\Sigma_{\Pi}^{+} \cup \Sigma_{p}^{+}} \cap \ker \left(\mathbb{T}_{n_{\alpha}}^{\Sigma^{+}} \xrightarrow{\phi_{\Pi_{\alpha}}} O_{E} \to O_{E} / \lambda^{m} \right). \end{cases}$$

We then introduce following assumption.

Assumption 7.3.2. For $\alpha = 0, 1$, we have $\mathrm{H}^{i}_{\mathfrak{T}}(\overline{\mathrm{M}}_{n_{\alpha}}, O_{\lambda})_{\mathfrak{m}_{\alpha}} = 0$ for $i \neq n_{\alpha} - 1$, and that $\mathrm{H}^{n_{\alpha}-1}_{\mathfrak{T}}(\overline{\mathrm{M}}_{n_{\alpha}}, O_{\lambda})_{\mathfrak{m}_{\alpha}}$ is a finite free O_{λ} -module.

Lemma 7.3.3. Assume Assumptions 7.1.1, 7.3.2, and Hypothesis 3.2.9 for n_1 .

(1) The $O_{\lambda}[\operatorname{Gal}(\overline{\mathbb{F}}_p/\mathbb{F}_{p^2})]$ -module $\operatorname{H}^{2r_1}_{\mathfrak{T}}(\overline{\operatorname{M}}_{n_1}, O_{\lambda}(r_1))_{\mathfrak{m}_1}$ is weakly semisimple (Definition 2.1.2). (2) The map

$$\pi_{n_1!} \circ \iota_{n_1}^* \colon (\mathrm{H}^{2r_1}_{\mathfrak{T}}(\overline{\mathrm{M}}_{n_1}, O_{\lambda}(r_1))_{\mathfrak{m}_1})_{\mathrm{Gal}(\overline{\mathbb{F}}_{p}/\mathbb{F}_{p^2})} \to \mathrm{H}^0_{\mathfrak{T}}(\overline{\mathrm{S}}_{n_1}, O_{\lambda})_{\mathfrak{m}_1}$$

is an isomorphism.

Proof. The proof of the lemma is similar to Theorem 6.2.3. For readers' convenience, we reproduce the details under the current setup.

For (1), by Lemma 5.1.7, we have an isomorphism

$$\mathrm{H}^{2r_{1}}_{\mathfrak{T}}(\overline{\mathrm{M}}_{n_{1}},O_{\lambda}(r_{1}))_{\mathfrak{m}_{1}}\simeq\mathrm{H}^{2r_{1}}_{\mathrm{\acute{e}t}}(\mathrm{Sh}(\mathrm{V}_{n_{1}},\mathrm{K}_{n_{1}})_{\overline{F}},O_{\lambda}(r_{1}))_{\mathfrak{m}_{1}}$$

of $O_{\lambda}[\operatorname{Gal}(\overline{\mathbb{Q}}_p/\mathbb{Q}_{p^2})]$ -modules. By Lemma 6.1.10, Proposition C.3.1(2), and Hypothesis 3.2.9, we have an isomorphism

$$\mathrm{H}^{2r_{1}}_{\mathrm{\acute{e}t}}(\mathrm{Sh}(\mathrm{V}_{n_{1}},\mathrm{K}_{n_{1}})_{\overline{F}},O_{\lambda}(r_{1}))_{\mathfrak{m}_{1}}\otimes_{O_{\lambda}}\overline{\mathbb{Q}}_{\ell}\simeq\bigoplus_{\pi_{1}}\rho_{\mathrm{BC}(\pi_{1}),\iota_{\ell}}^{\mathtt{c}}(r_{1})^{\oplus d(\pi_{1})}$$

of representations of Γ_F with coefficients in $\overline{\mathbb{Q}}_{\ell}$, where $d(\pi_1) \coloneqq \dim(\pi_1^{\infty,p})^{\mathbf{K}_{n_1}^p}$. Here, the direct sum is taken over all stable automorphic representations π_1 of $\mathrm{U}(\mathrm{V}_{n_1})(\mathbb{A}_{F^+})$ that is Π_1 -congruent and such that $\pi_{1_{\underline{\tau}_{\infty}}}$ is a holomorphic discrete series representation of $\mathrm{U}(\mathrm{V}_{n_1})(\mathbb{A}_{F^+})$ with the Harish-Chandra parameter $\{r_1, r_1 - 1, \ldots, 1 - r_1, -r_1\}$; and $\pi_{1_{\underline{\tau}}}$ is trivial for every archimedean place $\underline{\tau} \neq \underline{\tau}_{\infty}$. We may replace E_{λ} by a finite extension inside $\overline{\mathbb{Q}}_{\ell}$ such that $\rho_{\mathrm{BC}(\pi_1),\iota_{\ell}}$ is defined over E_{λ} for every π_1 appeared in the previous direct sum. Now we regard $\rho_{\mathrm{BC}(\pi_1),\iota_{\ell}}$ as a representation over E_{λ} . Then $\rho_{\mathrm{BC}(\pi_1),\iota_{\ell}}(r_1)$ admits a Γ_F -stable O_{λ} -lattice $\mathrm{R}_{\mathrm{BC}(\pi_1)}$, unique up to homothety, whose reduction $\overline{\mathrm{R}}_{\mathrm{BC}(\pi_1)}$ is isomorphic to $\rho_{\Pi_1,\lambda}(r_1)$. Moreover, we have an inclusion

$$\mathrm{H}^{2r_{1}}_{\mathrm{\acute{e}t}}(\mathrm{Sh}(\mathrm{V}_{n_{1}},\mathrm{K}_{n_{1}})_{\overline{F}},O_{\lambda}(r_{1}))_{\mathfrak{m}_{1}}\subseteq\bigoplus_{\pi_{1}}(\mathrm{R}^{\mathtt{c}}_{\mathrm{BC}(\pi_{1})})^{\oplus d(\pi_{1})}$$

of $O_{\lambda}[\operatorname{Gal}(\overline{\mathbb{F}}_p/\mathbb{F}_{p^2})]$ -modules. By (PII4), we know that $\bar{\rho}_{\Pi_1,\lambda}^{\mathsf{c}}(r_1)$ is weakly semisimple and

$$\dim_{O_{\lambda}/\lambda} \bar{\rho}_{\Pi_{1},\lambda}^{\mathsf{c}}(r_{1})^{\operatorname{Gal}(\mathbb{F}_{p}/\mathbb{F}_{p^{2}})} = 1.$$

On the other hand, we have

$$\dim_{E_{\lambda}} \rho_{\mathrm{BC}(\pi_1),\iota_{\ell}}^{\mathsf{c}}(r_1)^{\mathrm{Gal}(\mathbb{F}_p/\mathbb{F}_{p^2})} \ge 1$$

Thus by Lemma 2.1.5, for every π_1 in the previous direct sum, $\mathrm{R}^{\mathsf{c}}_{\mathrm{BC}(\pi_1)}$ is weakly semisimple. Thus, $\mathrm{H}^{2r_1}_{\mathfrak{T}}(\overline{\mathrm{M}}_{n_1}, O_{\lambda}(r_1))_{\mathfrak{m}_1}$ is weakly semisimple by Lemma 2.1.4(1). Thus, (1) follows.

For (2), we note that in (1) we have also proved that $(\mathrm{H}_{\mathfrak{T}}^{2r_1}(\overline{\mathrm{M}}_{n_1}, O_{\lambda}(r_1))_{\mathfrak{m}_1})_{\mathrm{Gal}(\overline{\mathbb{F}}_{p}/\mathbb{F}_{p^2})}$ is a free O_{λ} -module of rank $\sum_{\pi_1} d(\pi_1)$. By Theorem 4.3.10, Proposition B.4.3(2), and (PII4), we know that $\pi_{n_1!} \circ \iota_{n_1}^*$ is surjective. Thus, it remains to show that

$$\sum_{\pi_1} d(\pi_1) \leq \dim_{E_{\lambda}} \mathrm{H}^{0}_{\mathfrak{T}}(\overline{\mathrm{S}}_{n_1}, O_{\lambda})_{\mathfrak{m}_1} \otimes_{O_{\lambda}} E_{\lambda}$$

However, the above inequality is a consequence of Proposition 4.3.4 and Corollary C.3.3.

The lemma is proved.

We have a finite morphism $\operatorname{Sh}(V_n, K_{\operatorname{sp}}) \to \operatorname{Sh}(V_n, K_n) \times_{\operatorname{Spec} F} \operatorname{Sh}(V_{n+1}, K_{n+1})$, which gives rise to a class

$$[\operatorname{Sh}(\mathbf{V}_n, \mathbf{K}_{\operatorname{sp}})] \in \mathrm{H}^{2n}_{\operatorname{\acute{e}t}}(\operatorname{Sh}(\mathbf{V}_{n_0}, \mathbf{K}_{n_0}) \times_{\operatorname{Spec} F} \operatorname{Sh}(\mathbf{V}_{n_1}, \mathbf{K}_{n_1}), O_{\lambda}(n))$$

by the absolute cycle class map.

Theorem 7.3.4 (Second explicit reciprocity law). Assume Assumptions 7.1.1, 7.3.2, and Hypothesis 3.2.9 for both n and n + 1. Then we have

$$\exp_{\lambda}\left(\operatorname{loc}_{\mathfrak{p}}([\operatorname{Sh}(\mathbf{V}_{n},\mathbf{K}_{\operatorname{sp}})]),\operatorname{H}^{2n}_{\operatorname{\acute{e}t}}((\operatorname{Sh}(\mathbf{V}_{n_{0}},\mathbf{K}_{n_{0}})\times_{\operatorname{Spec} F}\operatorname{Sh}(\mathbf{V}_{n_{1}},\mathbf{K}_{n_{1}}))_{F_{\mathfrak{p}}},O_{\lambda}(n))/(\mathfrak{n}_{0},\mathfrak{n}_{1})\right)$$
$$\leq \exp_{\lambda}\left(\mathbbm{1}_{\operatorname{Sh}(\mathbf{V}_{n}^{\star},\mathbf{K}_{\operatorname{sp},\operatorname{sp}}^{\star})},O_{\lambda}[\operatorname{Sh}(\mathbf{V}_{n_{0}}^{\star},\mathbf{K}_{n_{0}}^{\star})\times\operatorname{Sh}(\mathbf{V}_{n_{1}}^{\star},\mathbf{K}_{n_{1}}^{\star})]/(\mathfrak{n}_{0},\mathfrak{n}_{1})\right)$$

where loc_p is introduced in Construction 4.5.1; exp_{λ} is introduced in Definition 2.1.6; and the element $\mathbb{1}_{Sh(V_n^{\star}, K_{sp,sp}^{\star})}$ is regarded as the pushforward of the characteristic function along the map $Sh(V_n^{\star}, K_{sp,sp}^{\star}) \to Sh(V_n^{\star}, K_n^{\star}) \times Sh(V_{n+1}^{\star}, K_{n+1}^{\star}).$

Proof. We claim that

- (1) the action of $T_{n_1,\mathfrak{p}}^{\star}$ on $H_{\mathfrak{T}}^{2r_0}(M_{n_0} \times_{T_{\mathfrak{p}}} S_{n_1}, O_{\lambda}(r_0))_{(\mathfrak{m}_0,\mathfrak{m}_1)}$ is invertible; and
- (2) the composite map

$$(\mathrm{id} \times \pi_{n_1})_! \circ (\mathrm{id} \times \iota_{n_1})^* \colon \mathrm{H}^{2n}_{\mathfrak{T}}(\mathrm{M}_{n_0} \times_{\mathrm{T}_{\mathfrak{p}}} \mathrm{M}_{n_1}, O_{\lambda}(n))_{(\mathfrak{m}_0,\mathfrak{m}_1)} \to \mathrm{H}^{2r_0}_{\mathfrak{T}}(\mathrm{M}_{n_0} \times_{\mathrm{T}_{\mathfrak{p}}} \mathrm{S}_{n_1}, O_{\lambda}(r_0))_{(\mathfrak{m}_0,\mathfrak{m}_1)}$$

is an isomorphism.

We prove the theorem assuming these two claims. Take a uniformizer λ_0 of E_{λ} . Suppose $\lambda_0^e \mathbb{1}_{\mathrm{Sh}(\mathrm{V}_n^\star,\mathrm{K}_{\mathrm{sp,sp}}^\star)} = 0$ in $O_{\lambda}[\mathrm{Sh}(\mathrm{V}_{n_0}^\star,\mathrm{K}_{n_0}^\star) \times \mathrm{Sh}(\mathrm{V}_{n_1}^\star,\mathrm{K}_{n_1}^\star)]/(\mathfrak{n}_0,\mathfrak{n}_1)$ for some integer $e \geq 0$. Applying Theorem 4.5.2 to the object $(\mathrm{K}_{\mathrm{sp}},\mathrm{K}_{n+1}^\star) \in \mathfrak{K}(\mathrm{V}_n)_{\mathrm{sp}}$ followed by pushforward, we have

$$\lambda_0^e \mathsf{T}_{n_1,\mathfrak{p}}^{\star}.(\mathrm{id} \times \pi_{n_1})!(\mathrm{id} \times \iota_{n_1})^* \mathrm{loc}_{\mathfrak{p}}'([\mathrm{Sh}(\mathrm{V}_n, \mathrm{K}_{\mathrm{sp}})]) = 0$$

in $\mathrm{H}^{2n}_{\mathfrak{T}}(\mathrm{M}_{n_0} \times_{\mathrm{T}_{\mathfrak{p}}} \mathrm{S}_{n_1}, O_{\lambda}(n)) / (\mathfrak{n}_0, \mathfrak{n}_1)$. By the above two claims, we must have

$$\lambda_0^e \text{loc}'_{\mathfrak{p}}([\text{Sh}(\mathbf{V}_n, \mathbf{K}_{\text{sp}})]) = 0$$

in $\mathrm{H}^{2n}_{\mathfrak{T}}(\mathrm{M}_{n_0} \times_{\mathrm{T}_{\mathfrak{p}}} \mathrm{M}_{n_1}, O_{\lambda}(n))/(\mathfrak{n}_0, \mathfrak{n}_1)$. Thus, we have

$$\lambda_0^e \operatorname{loc}_{\mathfrak{p}}([\operatorname{Sh}(\mathbf{V}_n, \mathbf{K}_{\operatorname{sp}})]) = 0$$

as the map $\mathrm{H}^{2n}_{\mathrm{\acute{e}t}}((\mathrm{Sh}(\mathrm{V}_{n_0}, \mathrm{K}_{n_0}) \times_{\mathrm{Spec}\,F} \mathrm{Sh}(\mathrm{V}_{n_1}, \mathrm{K}_{n_1}))_{F_{\mathfrak{p}}}, O_{\lambda}(n)) \to \mathrm{H}^{2n}_{\mathfrak{T}}(\mathrm{M}_{n_0} \times_{\mathrm{T}_{\mathfrak{p}}} \mathrm{M}_{n_1}, O_{\lambda}(n))$ is an isomorphism. The theorem follows.

Now we consider the two claims. By the Hochschild–Serre spectral sequence, we have a short exact sequence

$$0 \longrightarrow \mathrm{H}^{1}(\mathbb{F}_{p^{2}}, \mathrm{H}^{2n-1}_{\mathfrak{T}}(\overline{\mathrm{M}}_{n_{0}} \times_{\overline{\mathrm{T}_{\mathfrak{p}}}} \overline{\mathrm{M}}_{n_{1}}, O_{\lambda}(n))_{(\mathfrak{m}_{0},\mathfrak{m}_{1})}) \longrightarrow \mathrm{H}^{2n}_{\mathfrak{T}}(\mathrm{M}_{n_{0}} \times_{\mathrm{T}_{\mathfrak{p}}} \mathrm{M}_{n_{1}}, O_{\lambda}(n))_{(\mathfrak{m}_{0},\mathfrak{m}_{1})} \longrightarrow \mathrm{H}^{2n}_{\mathfrak{T}}(\overline{\mathrm{M}}_{n_{0}} \times_{\overline{\mathrm{T}_{\mathfrak{p}}}} \overline{\mathrm{M}}_{n_{1}}, O_{\lambda}(n))_{(\mathfrak{m}_{0},\mathfrak{m}_{1})}) \longrightarrow \mathrm{H}^{2n}_{\mathfrak{T}}(\mathrm{M}_{n_{0}} \times_{\mathrm{T}_{\mathfrak{p}}} \mathrm{M}_{n_{1}}, O_{\lambda}(n))_{(\mathfrak{m}_{0},\mathfrak{m}_{1})} \longrightarrow \mathrm{H}^{2n}_{\mathfrak{T}}(\mathrm{M}_{n_{0}} \times_{\overline{\mathrm{T}_{\mathfrak{p}}}} \overline{\mathrm{M}}_{n_{1}}, O_{\lambda}(n))_{(\mathfrak{m}_{0},\mathfrak{m}_{1})}) \longrightarrow \mathrm{H}^{2n}_{\mathfrak{T}}(\mathrm{M}_{n_{0}} \times_{\mathrm{T}_{\mathfrak{p}}} \mathrm{M}_{n_{1}}, O_{\lambda}(n))_{(\mathfrak{m}_{0},\mathfrak{m}_{1})}) \longrightarrow \mathrm{H}^{2n}_{\mathfrak{T}}(\mathrm{M}_{n_{0}} \times_{\mathrm{T}_{\mathfrak{p}}} \mathrm{M}_{n_{0}} \times_{\mathrm{T}_{\mathfrak{p}}} \mathrm{M}_{n_{0}} \times_{\mathrm{T}_{\mathfrak{p}}} \mathrm{M}_{n_{0}} \times_{\mathrm{M}_{\mathfrak{p}}} \mathrm{M}_{\mathfrak{p}})$$

of O_{λ} -modules. By the Künneth formula and (an analog of) Lemma 6.1.10, we have

$$\mathrm{H}^{i}_{\mathfrak{T}}(\overline{\mathrm{M}}_{n_{0}}\times_{\overline{\mathrm{T}}_{\mathfrak{p}}}\overline{\mathrm{M}}_{n_{1}},O_{\lambda})_{(\mathfrak{m}_{0},\mathfrak{m}_{1})}\simeq\bigoplus_{i_{0}+i_{1}=i}\mathrm{H}^{i_{0}}_{\mathfrak{T}}(\overline{\mathrm{M}}_{n_{0}},O_{\lambda})\otimes_{O_{\lambda}}\mathrm{H}^{i_{1}}_{\mathfrak{T}}(\overline{\mathrm{M}}_{n_{1}},O_{\lambda})$$

for every $i \in \mathbb{Z}$. This implies $\mathrm{H}^{2n}_{\mathfrak{T}}(\overline{\mathrm{M}}_{n_0} \times_{\overline{\mathrm{T}}_{\mathfrak{p}}} \overline{\mathrm{M}}_{n_1}, O_{\lambda}(n))_{(\mathfrak{m}_0,\mathfrak{m}_1)} = 0$ and

$$\mathrm{H}^{2n-1}_{\mathfrak{T}}(\overline{\mathrm{M}}_{n_{0}}\times_{\overline{\mathrm{T}}_{\mathfrak{p}}}\overline{\mathrm{M}}_{n_{1}},O_{\lambda}(n))_{(\mathfrak{m}_{0},\mathfrak{m}_{1})}\simeq\mathrm{H}^{2r_{0}-1}_{\mathfrak{T}}(\overline{\mathrm{M}}_{n_{0}},O_{\lambda}(r_{0}))_{\mathfrak{m}_{0}}\otimes_{O_{\lambda}}\mathrm{H}^{2r_{1}}_{\mathfrak{T}}(\overline{\mathrm{M}}_{n_{1}},O_{\lambda}(r_{1}))_{\mathfrak{m}_{1}}$$

In particular, we have a canonical isomorphism

$$\mathrm{H}^{2n}_{\mathfrak{T}}(\mathrm{M}_{n_{0}}\times_{\mathrm{T}\mathfrak{p}}\mathrm{M}_{n_{1}},O_{\lambda}(n))_{(\mathfrak{m}_{0},\mathfrak{m}_{1})}\simeq\mathrm{H}^{1}(\mathbb{F}_{p^{2}},\mathrm{H}^{2r_{0}-1}_{\mathfrak{T}}(\overline{\mathrm{M}}_{n_{0}},O_{\lambda}(r_{0}))_{\mathfrak{m}_{0}}\otimes_{O_{\lambda}}\mathrm{H}^{2r_{1}}_{\mathfrak{T}}(\overline{\mathrm{M}}_{n_{1}},O_{\lambda}(r_{1}))_{\mathfrak{m}_{1}}).$$

Similarly, we have

$$(7.3) \qquad \mathbf{H}_{\mathfrak{T}}^{2r_{0}}(\mathbf{M}_{n_{0}} \times_{\mathbf{T}_{\mathfrak{p}}} \mathbf{S}_{n_{1}}, O_{\lambda}(r_{0}))_{(\mathfrak{m}_{0},\mathfrak{m}_{1})} \simeq \mathbf{H}^{1}(\mathbb{F}_{p^{2}}, \mathbf{H}_{\mathfrak{T}}^{2r_{0}-1}(\overline{\mathbf{M}}_{n_{0}}, O_{\lambda}(r_{0}))_{\mathfrak{m}_{0}} \otimes_{O_{\lambda}} \mathbf{H}_{\mathfrak{T}}^{0}(\overline{\mathbf{S}}_{n_{1}}, O_{\lambda})_{\mathfrak{m}_{1}}) = \mathbf{H}^{1}(\mathbb{F}_{p^{2}}, \mathbf{H}_{\mathfrak{T}}^{2r_{0}-1}(\overline{\mathbf{M}}_{n_{0}}, O_{\lambda}(r_{0}))_{\mathfrak{m}_{0}}) \otimes_{O_{\lambda}} \mathbf{H}_{\mathfrak{T}}^{0}(\overline{\mathbf{S}}_{n_{1}}, O_{\lambda})_{\mathfrak{m}_{1}}.$$

For claim (1), note that the action of $\mathbb{T}_{n_1,\mathfrak{p}}$ on $\mathrm{H}^{2r_0}_{\mathfrak{T}}(\mathrm{M}_{n_0} \times_{\mathrm{T}_{\mathfrak{p}}} \mathrm{S}_{n_1}, O_{\lambda}(r_0))_{(\mathfrak{m}_0,\mathfrak{m}_1)}$ factors through the second factor under the isomorphism (7.3). By Proposition B.4.3(2) and (PII4), we know that the action of $\mathrm{T}^{\star}_{n_1,\mathfrak{p}}$ on $\mathrm{H}^0_{\mathfrak{T}}(\overline{\mathrm{S}}_{n_1}, O_{\lambda})_{\mathfrak{m}_1}$ is invertible. Thus, (1) follows.

For claim (2), by (PII7) and a similar argument for the proof of Lemma 7.2.4(3), we know that the $O_{\lambda}[\text{Gal}(\overline{\mathbb{F}}_{p}/\mathbb{F}_{p^{2}})]$ -module

$$\mathrm{H}^{2r_{0}-1}_{\mathfrak{T}}(\overline{\mathrm{M}}_{n_{0}},O_{\lambda}(r_{0}))_{\mathfrak{m}_{0}}\otimes_{O_{\lambda}}\mathrm{ker}\left((\mathrm{H}^{2r_{1}}_{\mathfrak{T}}(\overline{\mathrm{M}}_{n_{1}},O_{\lambda}(r_{1}))_{\mathfrak{m}_{1}})\to(\mathrm{H}^{2r_{1}}_{\mathfrak{T}}(\overline{\mathrm{M}}_{n_{1}},O_{\lambda}(r_{1}))_{\mathfrak{m}_{1}})_{\mathrm{Gal}(\overline{\mathbb{F}}_{p}/\mathbb{F}_{p^{2}})}\right)$$

has zero $\operatorname{Gal}(\mathbb{F}_p/\mathbb{F}_{p^2})$ -coinvariants. Combining with Lemma 7.3.3, we obtain an isomorphism

$$\mathrm{H}^{2n}_{\mathfrak{T}}(\mathrm{M}_{n_{0}}\times_{\mathrm{T}_{\mathfrak{P}}}\mathrm{M}_{n_{1}},O_{\lambda}(n))_{(\mathfrak{m}_{0},\mathfrak{m}_{1})}\simeq\mathrm{H}^{1}(\mathbb{F}_{p^{2}},\mathrm{H}^{2r_{0}-1}_{\mathfrak{T}}(\overline{\mathrm{M}}_{n_{0}},O_{\lambda}(r_{0}))_{\mathfrak{m}_{0}})\otimes_{O_{\lambda}}(\mathrm{H}^{2r_{1}}_{\mathfrak{T}}(\overline{\mathrm{M}}_{n_{1}},O_{\lambda}(r_{1}))_{\mathfrak{m}_{1}})_{\mathrm{Gal}(\overline{\mathbb{F}}_{p}/\mathbb{F}_{p^{2}})}$$

from (7.2), under which the map $(id \times \pi_{n_1})_! \circ (id \times \iota_{n_1})^*$ coincides with $id \otimes (\pi_{n_1!} \circ \iota_{n_1}^*)$. Thus, (2) follows.

The theorem is proved.

8. Proof of main theorems

In the section, we prove our main theorems on bounding Selmer groups. In Subsection 8.1, we introduce the notation of (weakly) admissible primes for the coefficient field, and make some additional preparation for the main theorem. In Subsections 8.2 and 8.3, we prove our main theorems in the (Selmer) rank 0 and 1 cases, respectively.

8.1. Admissible primes for coefficient field. We keep the setup in Subsection 7.1.

Definition 8.1.1. We introduce following assumptions on a prime λ of E with the underlying rational prime ℓ (and the ring of integers O_{λ} of E_{λ}):

- (L1): $\ell > 4n$ and is unramified in F;
- (L2): Σ_{\min}^+ does not contain ℓ -adic places;
- (L3): the Galois representation $\rho_{\Pi_0,\lambda} \otimes_{E_\lambda} \rho_{\Pi_1,\lambda}$ is absolutely irreducible;
- (L4): Assumption 7.1.1 is satisfied, that is, $\rho_{\Pi_0,\lambda}$ and $\rho_{\Pi_1,\lambda}$ are both residually absolutely irreducible;

- (L5): under (L4), for $\alpha = 0, 1$, we have a Γ_F -stable O_{λ} -lattice R_{α} in $\rho_{\Pi_{\alpha},\lambda}(r_{\alpha})$, unique up to homothety, that is $(1-\alpha)$ -polarizable, for which we choose a $(1-\alpha)$ -polarization $\Xi_{\alpha} \colon R_{\alpha}^{c} \xrightarrow{\sim} R_{\alpha}^{\vee}(1-\alpha)$ and an isomorphism $R_{\alpha} \simeq O_{\lambda}^{\oplus n_{\alpha}}$ of O_{λ} -modules.²⁰ After adopting the notation in Subsection 2.6, we have
 - (L5-1): either one of the two assumptions in Lemma 2.3.4 is satisfied;
 - (L5-2): Lemma 2.6.1(3) holds with $F' = F_{\text{rflx}}^+$ (Definition 3.3.2) and $\mathscr{P}(T) = T^2 1$ (see Remark 8.1.2 below for a more explicit description);
- (L6): under (L4), the homomorphism $\bar{\rho}_{\Pi_0,\lambda,+}$ (Remark 6.1.5) is rigid for $(\Sigma_{\min}^+, \emptyset)$ (Definition E.7.1), and $\bar{\rho}_{\Pi_0,\lambda}|_{\text{Gal}(\overline{F}/F(\zeta_\ell))}$ is absolutely irreducible;
- (L7): for $\alpha = 0, 1$, the composite homomorphism $\mathbb{T}_{n_{\alpha}}^{\Sigma_{\min}^+} \xrightarrow{\phi_{\Pi_{\alpha}}} O_E \to O_E/\lambda$ is cohomologically generic (Definition D.1.1).

Finally, we say that

- (1) λ is weakly admissible (with respect to (Π_0, Π_1)) if (L1–L5) are satisfied;
- (2) λ is *admissible* (with respect to (Π_0, Π_1)) if (L1–L7) are satisfied.

Remark 8.1.2. In Definition 8.1.1, (L5-2) is equivalent to the following assertion: the image of the restriction of the homomorphism

$$(\bar{\rho}_{0+},\bar{\rho}_{1+},\bar{\epsilon}_{\ell})\colon\Gamma_{F^+}\to\mathscr{G}_{n_0}(O_{\lambda}/\lambda)\times\mathscr{G}_{n_1}(O_{\lambda}/\lambda)\times(O_{\lambda}/\lambda)^{\times}$$

(see Notation 2.5.1 for the notation) to $\operatorname{Gal}(\overline{F}/F_{rflx}^+)$ contains an element $(\gamma_0, \gamma_1, \xi)$ satisfying

- (a) $\xi^2 1 \neq 0;$
- (b) for $\alpha = 0, 1, \gamma_{\alpha}$ belongs to $(\operatorname{GL}_{n_{\alpha}}(O_{\lambda}/\lambda) \times (O_{\lambda}/\lambda)^{\times})\mathfrak{c}$ with order coprime to ℓ ;
- (c) 1 appears in the eigenvalues of each of h_{γ_0} , h_{γ_1} , and $h_{\gamma_0} \otimes h_{\gamma_1}$ (Notation 2.5.2) with multiplicity one;
- (d) h_{γ_0} does not have an eigenvalue that is equal to -1 in O_{λ}/λ ;
- (e) h_{γ_1} does not have an eigenvalue that is equal to $-\xi$ in O_{λ}/λ .

Lemma 8.1.3. The representation $\rho_{\Pi_0,\lambda} \otimes_{E_\lambda} \rho_{\Pi_1,\lambda}(n)$ is pure of weight -1 at every nonarchimedean place w of F not above ℓ (Definition 2.4.4).

Proof. This is a consequence of Proposition
$$3.2.4(1)$$
 and [TY07, Lemma $1.4(3)$].

Lemma 8.1.4. Assume Hypothesis 3.2.9 for n_1 . Let V_{n_1} be a standard indefinite hermitian space of rank n_1 over F, Λ_{n_1} a self-dual $\prod_{v \notin \Sigma_{\infty}^+ \cup \Sigma_{\min}^+} O_{F_v}$ -lattice in $V_{n_1} \otimes_F \mathbb{A}_F^{\Sigma_{\infty}^+ \cup \Sigma_{\min}^+}$, and λ a prime of E. Consider a finite set \mathfrak{P} of special inert primes of F^+ whose underlying rational primes are distinct and coprime to Σ_{\min}^+ , and an object $K_{n_1} \in \mathfrak{K}(V_{n_1})$ of the form $(K_{n_1})_{\Sigma_{\min}^+} \times \prod_{v \notin \Sigma_{\infty}^+ \cup \Sigma_{\min}^+} U(\Lambda_{n_1})(O_{F_v^+})$. Put

$$\mathfrak{m}_{1} \coloneqq \mathbb{T}_{n_{1}}^{\Sigma_{\min}^{+} \cup \Sigma_{\mathfrak{P}}^{+}} \cap \ker \left(\mathbb{T}_{n_{1}}^{\Sigma_{\min}^{+}} \xrightarrow{\phi_{\Pi_{1}}} O_{E} \to O_{E} / \lambda \right)$$

where $\Sigma_{\mathfrak{P}}^+$ is the union of Σ_p^+ for all underlying rational primes p of \mathfrak{P} . Suppose that $P_{\alpha(\Pi_{1,\mathfrak{p}})} \mod \lambda$ is intertwining generic (Definition 3.1.5) for every $\mathfrak{p} \in \mathfrak{P}$; and that

- (a) either the composite homomorphism $\mathbb{T}_{n_1}^{\Sigma_{\min}^+} \xrightarrow{\phi_{\Pi_1}} O_E \to O_E/\lambda$ is cohomologically generic;
- (b) or $n_1 = 3$ and $\operatorname{H}^i_{\operatorname{\acute{e}t}}(\operatorname{Sh}(\operatorname{V}_3, \operatorname{K}_3), O_E/\lambda)_{\mathfrak{m}_1} = 0$ if $i \neq 2$.

Then for every special maximal subgroup $K'_{n_1,\mathfrak{P}}$ of $\prod_{\mathfrak{p}\in\mathfrak{P}} U(V_{n_1})(F_{\mathfrak{p}}^+)$ and every $i \in \mathbb{Z}$, we have an isomorphism

$$\mathrm{H}^{i}_{\mathrm{\acute{e}t}}(\mathrm{Sh}(\mathrm{V}_{n_{1}},\mathrm{K}_{n_{1}}),O_{\lambda})_{\mathfrak{m}_{1}}\simeq\mathrm{H}^{i}_{\mathrm{\acute{e}t}}(\mathrm{Sh}(\mathrm{V}_{n_{1}},\mathrm{K}^{\mathfrak{P}}_{n_{1}}\mathrm{K}'_{n_{1},\mathfrak{P}}),O_{\lambda})_{\mathfrak{m}_{1}}$$

²⁰In fact, (L5) does not depend on the choice of Ξ_{α} and the basis, since Ξ_{α} is unique up to units in O_{λ} and the basis is unique up to conjugation in $\operatorname{GL}_{n_{\alpha}}(O_{\lambda})$.

of $O_{\lambda}[\Gamma_F]$ -modules.

Note that (a) is stronger than (b) when $n_1 = 3$.

Proof. We first note that for every $\mathfrak{p} \in \mathfrak{P}$, $U(V_{n_1})(F_{\mathfrak{p}}^+)$ has two special maximal subgroups up to conjugation, exact one of which is hyperspecial maximal.

Take an element $\mathfrak{p} \in \mathfrak{P}$, a special maximal subgroup $K_{n_1,\mathfrak{P}}^{\prime\mathfrak{p}}$ of $\prod_{\mathfrak{p}'\in\mathfrak{P}\setminus\{\mathfrak{p}\}} U(V_{n_1})(F_{\mathfrak{p}'}^+)$, a hyperspecial maximal subgroup $K_{n_1,\mathfrak{p}}^{\circ}$ of $U(V_{n_1})(F_{\mathfrak{p}}^+)$, and a non-hyperspecial special maximal subgroup $K_{n_1,\mathfrak{p}}^{\bullet}$ of $U(V_{n_1})(F_{\mathfrak{p}}^+)$. We claim that if $H^i_{\text{ét}}(Sh(V_{n_1}, K_{n_1}^{\mathfrak{P}}K_{n_1,\mathfrak{P}}^{\circ}K_{n_1,\mathfrak{p}}^{\circ}), O_E/\lambda)_{\mathfrak{m}_1} = 0$ for $i \neq 2r_1$, then there is an isomorphism

$$\mathrm{H}^{i}_{\mathrm{\acute{e}t}}(\mathrm{Sh}(\mathrm{V}_{n_{1}},\mathrm{K}^{\mathfrak{P}}_{n_{1}}\mathrm{K}^{\prime\mathfrak{p}}_{n_{1},\mathfrak{P}}\mathrm{K}^{\circ}_{n_{1},\mathfrak{p}}),O_{\lambda})_{\mathfrak{m}_{1}}\simeq\mathrm{H}^{i}_{\mathrm{\acute{e}t}}(\mathrm{Sh}(\mathrm{V}_{n_{1}},\mathrm{K}^{\mathfrak{P}}_{n_{1}}\mathrm{K}^{\prime\mathfrak{p}}_{n_{1},\mathfrak{P}}\mathrm{K}^{\bullet}_{n_{1},\mathfrak{p}}),O_{\lambda})_{\mathfrak{m}_{2}}$$

of $O_{\lambda}[\Gamma_F]$ -modules for every $i \in \mathbb{Z}$.

Fix an embedding $E_{\lambda} \hookrightarrow \overline{\mathbb{Q}}_{\ell}$. We first show that there is an isomorphism

(8.1) $H^{i}_{\text{\acute{e}t}}(\mathrm{Sh}(\mathrm{V}_{n_{1}},\mathrm{K}^{\mathfrak{P}}_{n_{1}}\mathrm{K}'^{\mathfrak{p}}_{n_{1},\mathfrak{P}}\mathrm{K}^{\circ}_{n_{1},\mathfrak{p}}),O_{\lambda})_{\mathfrak{m}_{1}}\otimes_{O_{\lambda}}\overline{\mathbb{Q}}_{\ell}\simeq \mathrm{H}^{i}_{\text{\acute{e}t}}(\mathrm{Sh}(\mathrm{V}_{n_{1}},\mathrm{K}^{\mathfrak{P}}_{n_{1}}\mathrm{K}'^{\mathfrak{p}}_{n_{1},\mathfrak{P}}\mathrm{K}^{\bullet}_{n_{1},\mathfrak{p}}),O_{\lambda})_{\mathfrak{m}_{1}}\otimes_{O_{\lambda}}\overline{\mathbb{Q}}_{\ell}$

of $\overline{\mathbb{Q}}_{\ell}[\Gamma_F]$ -modules for every $i \in \mathbb{Z}$. Let $\Lambda_{n_1,\mathfrak{p}}^{\circ}$ be the self-dual $O_{F_{\mathfrak{p}}}$ -lattice in $V_{n_1} \otimes_F F_{\mathfrak{p}}$ whose stabilizer is $K_{n_1,\mathfrak{p}}^{\circ}$. Without lost of generality, we may assume that $K_{n_1,\mathfrak{p}}^{\bullet}$ is the stabilizer of a lattice $\Lambda_{n_1,\mathfrak{p}}^{\bullet}$ satisfying $\Lambda_{n_1,\mathfrak{p}}^{\circ} \subseteq \Lambda_{n_1,\mathfrak{p}}^{\bullet}$ and $(\Lambda_{n_1,\mathfrak{p}}^{\bullet})^{\vee}/\mathfrak{p}\Lambda_{n_1,\mathfrak{p}}^{\bullet} \simeq \mathbb{F}_{p^2}$. To show (8.1), it suffices to show that for every (necessarily cuspidal) automorphic representation π_1 of $U(V_{n_1})(\mathbb{A}_{F^+})$ that appears in either side of (8.1), the maps

$$\mathbf{T}_{n_{1},\mathfrak{p}}^{\bullet\circ} \colon \pi_{1,\mathfrak{p}}^{\mathbf{K}_{n_{1},\mathfrak{p}}^{\circ}} \to \pi_{1,\mathfrak{p}}^{\mathbf{K}_{n_{1},\mathfrak{p}}^{\bullet}}, \quad \mathbf{T}_{n_{1},\mathfrak{p}}^{\circ\bullet} \colon \pi_{1,\mathfrak{p}}^{\mathbf{K}_{n_{1},\mathfrak{p}}^{\bullet}} \to \pi_{1,\mathfrak{p}}^{\mathbf{K}_{n_{1},\mathfrak{p}}^{\circ}}$$

are both isomorphisms. Hypothesis 3.2.9 and the Chebotarev density theorem imply that $\rho_{\mathrm{BC}(\pi_1),\iota_{\ell}}$ and $\rho_{\Pi_1,\lambda} \otimes_{E_{\lambda}} \overline{\mathbb{Q}}_{\ell}$ have the isomorphic (irreducible) residual representations. In particular, the Satake parameter of $\mathrm{BC}(\pi_1)_{\mathfrak{p}}$ does not contain $\{-p, -p^{-1}\}$ by Proposition 3.2.4(2) and the assumption that $P_{\alpha(\Pi_{1,\mathfrak{p}})} \mod \lambda$ is intertwining generic. Thus, we obtain the isomorphism (8.1) by Proposition B.4.3(2).

To prove the claim it suffices to show that $\mathrm{H}^{i}_{\mathrm{\acute{e}t}}(\mathrm{Sh}(\mathrm{V}_{n_{1}}, \mathrm{K}^{\mathfrak{P}}_{n_{1}}, \mathrm{K}^{\prime \mathfrak{p}}_{n_{1}, \mathfrak{p}}, \mathbb{C}_{\lambda})_{\mathfrak{m}_{1}}$ is a free O_{λ} -module for every $i \in \mathbb{Z}$. If we assume (a), then this follows immediately. If we assume (b), then this follows from Proposition 5.9.4, Lemma 6.2.1, and a straightforward computation on the spectral sequence in Lemma 5.8.2.

The lemma follows immediately from the above claim by induction on the number of primes $\mathfrak{p} \in \mathfrak{P}$ for which $K'_{n_1,\mathfrak{P}}$ is not hyperspecial maximal at \mathfrak{p} . Note that the initial induction hypothesis is satisfied by either (a) or (b).

Proposition 8.1.5. Suppose $E = \mathbb{Q}$ and that there are two elliptic curves A_0 and A_1 over F^+ such that for every rational prime ℓ of E and $\alpha = 0, 1$, we have $\rho_{\Pi_{\alpha},\ell} \simeq \operatorname{Sym}^{n_{\alpha}-1} \operatorname{H}^{1}_{\operatorname{\acute{e}t}}(A_{\alpha\overline{F}}, \mathbb{Q}_{\ell})|_{\Gamma_{F}}$. If $A_{0\overline{F}}$ and $A_{1\overline{F}}$ are not isogenous to each other and $\operatorname{End}(A_{0\overline{F}}) = \operatorname{End}(A_{1\overline{F}}) = \mathbb{Z}$, then all but finitely many rational primes ℓ are weakly admissible; and when $[F^+ : \mathbb{Q}] > 1$, all but finitely many rational primes ℓ are admissible.

Proof. We need to show that every condition in Definition 8.1.1 excludes only finitely many ℓ (for (L7) we assume $[F^+:\mathbb{Q}] > 1$). By [Ser72, Théorème 6], for sufficiently large ℓ , the homomorphisms

$$\Gamma_{F^+} \to \mathrm{GL}(\mathrm{H}^1_{\mathrm{\acute{e}t}}(A_{\alpha \overline{F}}, \mathbb{F}_\ell)) \simeq \mathrm{GL}_2(\mathbb{F}_\ell)$$

are both surjective for $\alpha = 0, 1$. Thus, we may assume that this is the case.

For (L1) and (L2), this is trivial.

For (L3), (L4), and (L5), this has been proved in Proposition 2.6.3.

For (L6), by Proposition E.5.12, the condition that $\bar{\rho}_{\Pi_0,\lambda,+}$ is rigid for $(\Sigma_{\min}^+, \emptyset)$ excludes only finitely many ℓ . It is clear that the remaining two conditions also exclude only finitely many ℓ .

For (L7), this follows from Corollary D.1.4.

8.2. Main theorem in the Selmer rank 0 case.

Theorem 8.2.1. Keep the setup in Subsection 7.1. Assume Hypothesis 3.2.9 for both n and n+1. If $L(\frac{1}{2}, \Pi_0 \times \Pi_1) \neq 0$, then for all admissible primes λ of E, and for all but finitely many weakly admissible primes λ of E when n = 2, we have

$$\mathrm{H}_{f}^{1}(F,\rho_{\Pi_{0},\lambda}\otimes_{E_{\lambda}}\rho_{\Pi_{1},\lambda}(n))=0$$

Proof. By Lemma 8.2.2 below, we may fix the choices of V_n° , Λ_n° , $(K_n^\circ, K_{n+1}^\circ)$ in that lemma such that

$$\sum_{\mathrm{ESh}(\mathrm{V}_n^\circ,\mathrm{K}_n^\circ)} f(s,\mathrm{sh}_{\uparrow}(s)) \neq 0$$

for some $f \in O_E[Sh(V_{n_0}^{\circ}, K_{n_0}^{\circ})][\ker \phi_{\Pi_0}] \otimes_{O_E} O_E[Sh(V_{n_1}^{\circ}, K_{n_1}^{\circ})][\ker \phi_{\Pi_1}]$. Moreover, by Lemma D.2.2(3), we may assume that $(K_{n_0}^{\circ})_v$ is transferable (Definition D.2.1) for $v \in \Sigma_{\min}^+$.

We take a prime λ of E with the underlying rational prime ℓ . We adopt notation in Subsection 2.6 with the initial data in Definition 8.1.1. Define two nonnegative integers m_{per} and m_{lat} as follows.

(1) Let m_{per} be the largest (nonnegative) integer such that

 $s \in$

$$\sum_{\in \operatorname{Sh}(\mathcal{V}_n^\circ, \mathcal{K}_n^\circ)} f(s, \operatorname{sh}_{\uparrow}(s)) \in \lambda^{m_{\operatorname{per}}} O_E$$

for every $f \in O_E[\operatorname{Sh}(\operatorname{V}_{n_0}^\circ, \operatorname{K}_{n_0}^\circ)][\ker \phi_{\Pi_0}] \otimes_{O_E} O_E[\operatorname{Sh}(\operatorname{V}_{n_1}^\circ, \operatorname{K}_{n_1}^\circ)][\ker \phi_{\Pi_1}].$

(2) We choose a standard *indefinite* hermitian space V_{n_1} over F of rank n_1 , together with an identification $U((V_{n_1}^{\circ})^{\infty}) \simeq U(V_{n_1}^{\infty})$ of reductive groups over $\mathbb{A}_{F^+}^{\infty}$.²¹ In particular, we have the Shimura variety $Sh(V_{n_1}, K_{n_1}^{\circ})$. By Hypothesis 3.2.9, we have an isomorphism

$$\mathrm{H}^{2r_1}_{\mathrm{\acute{e}t}}(\mathrm{Sh}(\mathrm{V}_{n_1},\mathrm{K}^{\circ}_{n_1})_{\overline{F}},E_{\lambda}(r_1))/\ker\phi_{\Pi_1}\simeq (\mathrm{R}^{\mathsf{c}}_1\otimes_{O_{\lambda}}E_{\lambda})^{\oplus\mu}$$

of $E_{\lambda}[\Gamma_F]$ -modules for some integer $\mu_1 > 0$. We fix a map

 $\mathrm{H}^{2r_{1}}_{\mathrm{\acute{e}t}}(\mathrm{Sh}(\mathrm{V}_{n_{1}},\mathrm{K}^{\circ}_{n_{1}})_{\overline{F}},O_{\lambda}(r_{1}))/\ker\phi_{\Pi_{1}}\to(\mathrm{R}^{\mathtt{c}}_{1})^{\oplus\mu_{1}}$

of $O_{\lambda}[\Gamma_F]$ -modules whose kernel and cokernel are both O_{λ} -torsion. Then we let m_{lat} be the smallest nonnegative integer such that both the kernel and the cokernel are annihilated by $\lambda^{m_{\text{lat}}}$.

Now we assume that either λ is admissible, or n = 2 and λ is weakly admissible and satisfies $\mathrm{H}^{i}_{\mathrm{\acute{e}t}}(\mathrm{Sh}(\mathrm{V}_{3},\mathrm{K}^{\circ}_{3})_{\overline{F}},O_{E}/\lambda)/\ker\phi_{\Pi_{1}}=0$ for $i\neq 2$ (which only excludes finitely many primes). Note that in both cases we have $\mathrm{H}^{i}_{\mathrm{\acute{e}t}}(\mathrm{Sh}(\mathrm{V}_{n_{1}},\mathrm{K}^{\circ}_{n_{1}})_{\overline{F}},O_{E}/\lambda)/\ker\phi_{\Pi_{1}}=0$ if $i\neq 2r_{1}$.

We start to prove the theorem by contradiction, hence assume

$$\dim_{E_{\lambda}} \mathrm{H}^{1}_{f}(F, \rho_{\Pi_{0}, \lambda} \otimes_{E_{\lambda}} \rho_{\Pi_{1}, \lambda}(n)) \geq 1.$$

Take a sufficiently large positive integer m which will be determined later. By Lemma 8.1.3, we may apply Proposition 2.4.6 by taking Σ to be the set of places of F above $\Sigma_{\min}^+ \cup \Sigma_{\ell}^+$. Then we obtain a submodule S of $\mathrm{H}^1_{f,\mathrm{R}}(F, \bar{\mathrm{R}}^{(m)})$ that is free of rank 1 over $O_{\lambda}/\lambda^{m-m_{\Sigma}}$ such that $\mathrm{loc}_w|_S = 0$ for every nonarchimedean place $w \in \Sigma$ not above ℓ . Now we apply the discussion in Subsection 2.3 to the submodule $S \subseteq \mathrm{H}^1(F, \bar{\mathrm{R}}^{(m)})$. By (L5-1) and Lemma 2.3.6, we obtain an injective map

$$\theta_S \colon \operatorname{Gal}(F_S/F_{\bar{\rho}^{(m)}}) \to \operatorname{Hom}_{O_{\lambda}}(S, \mathbb{R}^{(m)})$$

²¹There are many choices of such V_{n_1} and the isomorphism. We choose one only to get some control on the discrepancy of the integral cohomology of Shimura varieties and the lattice coming from Galois representations.

whose image generates an O_{λ} -submodule containing $\lambda^{\mathbf{r}_{\bar{\mathbf{R}}(m)}}$ Hom $_{O_{\lambda}}(S, \bar{\mathbf{R}}^{(m)})$, which further contains $\lambda^{\mathbf{r}_{\mathbf{R}}}$ Hom $_{O_{\lambda}}(S, \bar{\mathbf{R}}^{(m)})$ by Lemma 2.3.3 and (L3). By (L5-2) and Lemma 2.6.1, we may choose an element $(\gamma_1, \gamma_2, \xi)$ in the image of $(\bar{\rho}_{1+}^{(m)}, \bar{\rho}_{2+}^{(m)}, \bar{\epsilon}_{\ell}^{(m)})|_{\mathrm{Gal}(\bar{F}/F_{\mathrm{rflx}}^+)}$ satisfying Lemma 2.6.1(2). It gives rise to an element $\gamma \in (\mathrm{GL}_{n_0n_1}(O_{\lambda}/\lambda^m) \times (O_{\lambda}/\lambda^m)^{\times})\mathfrak{c}$ as in Notation 2.5.2 such that $(\bar{\mathbf{R}}^{(m)})^{h_{\gamma}}$ is a free O_{λ}/λ^m -module of rank 1 by Lemma 2.6.2 and (2.4). Now we apply the discussion in Subsection 2.5. By Proposition 2.5.5 (with $m_0 = m_{\Sigma}, r_{\gamma} = 1, r_S = 1$), we may fix an (S, γ) -abundant element $\Psi \in G_{S,\gamma}$ (Definition 2.5.6).

We apply the discussion and notation in Subsection 7.2 to our situation with λ , m, $\Sigma_{\text{lr,I}}^+ = \emptyset$, $\Sigma_{\text{I}}^+ = \Sigma_{\min}^+$, $(\mathcal{V}_n^\circ, \Lambda_n^\circ)$, \mathcal{K}_n° and $(\mathcal{K}_n^\circ, \mathcal{K}_{n+1}^\circ)$. By the Chebotarev density theorem, we can choose a γ -abundant place $w_+^{(m)}$ of $F_+^{(m)}$ satisfying $\Psi_{w^{(m)}} = \Psi$ and whose underlying prime \mathfrak{p} of F^+ (and the underlying rational prime p) is a special inert prime satisfying (PI1)–(PI7) and

(PI8): the natural map

$$\operatorname{H}_{\operatorname{\acute{e}t}}^{2r_1}(\operatorname{Sh}(\operatorname{V}_{n_1},\operatorname{K}_{n_1}^\circ)_{\overline{F}},O_{\lambda}(r_1))/(\mathbb{T}_{n_1}^{\Sigma_1^+\cup\Sigma_p^+}\cap\ker\phi_{\Pi_1})\to\operatorname{H}_{\operatorname{\acute{e}t}}^{2r_1}(\operatorname{Sh}(\operatorname{V}_{n_1},\operatorname{K}_{n_1}^\circ)_{\overline{F}},O_{\lambda}(r_1))/\ker\phi_{\Pi_1}$$

is an isomorphism.

We also choose remaining data in the initial setup of Section 5 with $\mathbb{Q}_p^{\Phi} = \mathbb{Q}_{p^2}$, a definite uniformization datum as in Notation 5.10.13, and an indefinite uniformization datum as in Notation 5.10.1. By the definition of m_{per} , we have

(8.2)
$$\exp_{\lambda}\left(\mathbb{1}_{\mathrm{Sh}(\mathrm{V}_{n}^{\circ},\mathrm{K}_{\mathrm{sp}}^{\circ})},O_{E}[\mathrm{Sh}(\mathrm{V}_{n_{0}}^{\circ},\mathrm{K}_{n_{0}}^{\circ})\times\mathrm{Sh}(\mathrm{V}_{n_{1}}^{\circ},\mathrm{K}_{n_{1}}^{\circ})]/(\mathfrak{n}_{0},\mathfrak{n}_{1})\right)\geq m-m_{\mathrm{per}},$$

where we recall that

$$\mathbf{n}_{\alpha} = \mathbb{T}_{n_{\alpha}}^{\Sigma_{\mathbf{I}}^{+} \cup \Sigma_{p}^{+}} \cap \ker \left(\mathbb{T}_{n_{\alpha}}^{\Sigma_{\min}^{+}} \xrightarrow{\phi_{\Pi_{\alpha}}} O_{E} \to O_{E} / \lambda^{m} \right)$$

for $\alpha = 0, 1$. Here, $\mathbb{1}_{\operatorname{Sh}(V_n^\circ, K_{\operatorname{sp}}^\circ)}$ is nothing but the characteristic function of the graph $\Delta \operatorname{Sh}(V_n^\circ, K_n^\circ)$ of the map $\operatorname{Sh}(V_n^\circ, K_n^\circ) \to \operatorname{Sh}(V_{n+1}^\circ, K_{n+1}^\circ)$.

We claim that there exists an element $c_1 \in \mathrm{H}^1(F, \bar{\mathrm{R}}^{(m_1)c})$ for some positive integer $m_1 \leq m$ satisfying

(8.3)
$$\exp_{\lambda}\left(\partial_{\mathfrak{p}}\mathrm{loc}_{\mathfrak{p}}(c_{1}),\mathrm{H}^{1}_{\mathrm{sing}}(F_{\mathfrak{p}},\bar{\mathrm{R}}^{(m_{1})\mathsf{c}})\right) \geq m-m_{\mathrm{per}}-m_{\mathrm{lat}};$$

and such that for every nonarchimedean place w of F not above $\Sigma^+ \cup \{\mathfrak{p}\},\$

(8.4)
$$\operatorname{loc}_{w}(c_{1}) \in \operatorname{H}^{1}_{f}(F_{w}, \overline{\operatorname{R}}^{(m_{1})c})$$

t

holds.

We first prove the theorem assuming the existence of such c_1 . Fix a generator of the submodule $S \subseteq \mathrm{H}^1_{f,\mathrm{R}}(F, \bar{\mathrm{R}}^{(m)})$ and denote by its image in $\mathrm{H}^1(F, \bar{\mathrm{R}}^{(m_1)})$ by s_1 . We also identify $\bar{\mathrm{R}}^{(m_1)c}$ with $(\bar{\mathrm{R}}^{(m_1)})^*$ via the polarization Ξ . Now we compute the local Tate pairing $\langle s_1, c_1 \rangle_w$ (2.1) for every nonarchimedean place w of F.

- O Suppose that w is above Σ_{\min}^+ . Then we have $loc_w(s_1) = 0$ by our choice of S. Thus, $\langle s_1, c_1 \rangle_w = 0$.
- O Suppose that w is above Σ_{ℓ}^+ . Then by (L2), $\mathbb{R}_{\mathbb{Q}}$ is crystalline with Hodge–Tate weights in [-n, n-1]. Thus, we have $\mathrm{loc}_w(s_1) \in \mathrm{H}^1_f(F_w, \mathbb{R}^{(m_1)})$ by Lemma 2.4.3(2) and (L1). By (8.4), Lemma 2.2.6 and (L1), we have $\lambda^{m_{\mathrm{dif}}} \langle s_1, c_1 \rangle_w = 0$ where $\mathfrak{d}_{\lambda} = \lambda^{m_{\mathrm{dif}}} \subseteq O_{\lambda}$ is the different ideal of $E_{\lambda}/\mathbb{Q}_{\ell}$.
- O Suppose that w is not above $\Sigma_{\min}^+ \cup \Sigma_{\ell}^+ \cup \{\mathfrak{p}\}$. Then by (L2), R is unramified. Thus, we have $\operatorname{loc}_w(s_1) \in \operatorname{H}^1_f(F_w, \overline{R}^{(m_1)})$ by Lemma 2.4.3(1). By (8.4) and Lemma 2.2.3, we have $\langle s_1, c_1 \rangle_w = 0$.

O Suppose that w is the unique place above \mathfrak{p} . Then since $w_+^{(m)}$ is abundant, we have

$$\exp_{\lambda}\left(\operatorname{loc}_{w}(s_{1}), \operatorname{H}_{f}^{1}(F_{w}, \bar{\operatorname{R}}^{(m_{1})})\right) \geq m_{1} - m_{\Sigma} - \mathfrak{r}_{\mathrm{R}}.$$

By (8.3) and Lemma 2.2.3 again, we have

$$\exp_{\lambda}\left(\langle s_1, c_1 \rangle_w, O_{\lambda} / \lambda^{m_1}\right) \ge m - m_{\text{per}} - m_{\text{lat}} - m_{\Sigma} - \mathfrak{r}_{\text{R}}.$$

Therefore, as long as we take m such that $m > m_{\text{per}} + m_{\text{lat}} + m_{\Sigma} + \mathfrak{r}_{\text{R}} + m_{\text{dif}}$, we will have a contradiction to the relation

$$\sum_{w} \langle s_1, c_1 \rangle_w = 0$$

where the sum is taken over all nonarchimedean places w of F. The theorem is proved.

Now we consider the claim. By (L4), (L6), and Theorem 6.3.4(5), we have an isomorphism

(8.5)
$$\mathrm{H}^{2r_0-1}_{\mathrm{\acute{e}t}}((\mathrm{Sh}(\mathrm{V}'_{n_0},\mathbf{j}_{n_0}\mathrm{K}^{p\circ}_{n_0}\mathrm{K}'_{n_0,p})_{\overline{F}},O_{\lambda}(r_0))/\mathfrak{n}_0 \xrightarrow{\sim} \bigoplus_{i=1}^{\mu_0} \bar{\mathrm{R}}^{(m_i)c}_0$$

of $O_{\lambda}[\Gamma_F]$ -modules, for finitely many positive integers m_1, \ldots, m_{μ_0} at most m. Assumption 7.2.2 for $\alpha = 0$ is satisfied by (L7) if $n \geq 3$ and by (L4) if n = 2.

By Lemma 8.1.4, we have an isomorphism

$$\mathrm{H}^{i}_{\mathrm{\acute{e}t}}(\mathrm{Sh}(\mathrm{V}_{n_{1}},\mathrm{K}^{\circ}_{n_{1}})_{\overline{F}},O_{\lambda})_{\mathfrak{m}_{1}}\simeq\mathrm{H}^{i}_{\mathrm{\acute{e}t}}(\mathrm{Sh}(\mathrm{V}'_{n_{1}},\mathrm{j}_{n_{1}}\mathrm{K}^{p\circ}_{n_{1}}\mathrm{K}'_{n_{1},p})_{\overline{F}},O_{\lambda})_{\mathfrak{m}_{1}}$$

of $O_{\lambda}[\Gamma_F]$ -modules. Assumption 7.2.2 for $\alpha = 1$ is satisfied by (L7) if $n \ge 3$ and by (PI8) if n = 2. Moreover, by (PI8), we may fix a map

$$\mathrm{H}^{2r_{1}}_{\mathrm{\acute{e}t}}(\mathrm{Sh}(\mathrm{V}_{n_{1}}', \mathbf{j}_{n_{1}}\mathrm{K}_{n_{1}}^{p\circ}\mathrm{K}_{n_{1},p}')_{\overline{F}}, O_{\lambda}(r_{1}))/(\mathbb{T}_{n_{1}}^{\Sigma_{1}^{+}\cup\Sigma_{p}^{+}}\cap \ker \phi_{\Pi_{1}}) \to (\mathrm{R}_{1}^{\mathsf{c}})^{\oplus\mu_{1}}$$

of $O_{\lambda}[\Gamma_F]$ -modules whose kernel and cokernel are both annihilated by $\lambda^{m_{\text{lat}}}$. Taking quotient by λ^m , we obtain a map

(8.6)
$$\mathrm{H}_{\mathrm{\acute{e}t}}^{2r_{1}}(\mathrm{Sh}(\mathrm{V}_{n_{1}}', \mathbf{j}_{n_{1}}\mathrm{K}_{n_{1}}^{p_{0}}\mathrm{K}_{n_{1},p}')_{\overline{F}}, O_{\lambda}(r_{1}))/\mathfrak{n}_{1} \to (\bar{\mathrm{R}}_{1}^{(m)\mathtt{c}})^{\oplus\mu_{1}}$$

of $O_{\lambda}[\Gamma_F]$ -modules whose kernel and cokernel are both annihilated by $\lambda^{m_{\text{lat}}}$.

To continue, we adopt the notational abbreviation prior to Corollary 7.2.8. By Lemma 6.1.10 and the Künneth formula, we obtain a map

(8.7)
$$\Upsilon \colon \mathrm{H}^{2n-1}_{\mathrm{\acute{e}t}}((\mathrm{Sh}'_{n_0} \times_{\mathrm{Spec}\,F} \mathrm{Sh}'_{n_1})_{\overline{F}}, O_{\lambda}(n))/(\mathfrak{n}_0, \mathfrak{n}_1) \to \bigoplus_{i=1}^{\mu} \bar{\mathrm{R}}^{(m_i)\mathsf{c}}$$

of $O_{\lambda}[\Gamma_F]$ -modules whose kernel and cokernel are both annihilated by $\lambda^{m_{\text{lat}}}$, from (8.5) and (8.6). Here, we have re-indexed μ_1 copies of $\{m_1, \ldots, m_{\mu_0}\}$ into $\mu \coloneqq \mu_0 \mu_1$ positive integers at most m. Recall that we have a class

$$\mathrm{AJ}(\mathrm{Sh}'_{\mathrm{sp}}) \in \mathrm{H}^{1}(F, \mathrm{H}^{2n-1}_{\mathrm{\acute{e}t}}((\mathrm{Sh}'_{n_{0}} \times_{\mathrm{Spec}\,F} \mathrm{Sh}'_{n_{1}})_{\overline{F}}, O_{\lambda}(n))/(\mathfrak{n}_{0}, \mathfrak{n}_{1}))$$

where $\operatorname{Sh}'_{\operatorname{sp}}$ is nothing but the graph of the morphism $\operatorname{Sh}'_n \to \operatorname{Sh}'_{n+1}$. By Corollary 7.2.8 and (8.2), we have

(8.8)
$$\exp_{\lambda}\left(\partial_{\mathfrak{p}}\mathrm{loc}_{\mathfrak{p}}\operatorname{AJ}(\mathrm{Sh}'_{\mathrm{sp}}), \mathrm{H}^{1}_{\mathrm{sing}}(F_{\mathfrak{p}}, \mathrm{H}^{2n-1}_{\mathrm{\acute{e}t}}((\mathrm{Sh}'_{n_{0}} \times_{\mathrm{Spec}\,F} \mathrm{Sh}'_{n_{1}})_{\overline{F}}, O_{\lambda}(n))/(\mathfrak{n}_{0}, \mathfrak{n}_{1}))\right) \geq m - m_{\mathrm{per}}.$$

For every $1 \leq i \leq \mu$, let

$$\Upsilon_i \colon \mathrm{H}^{2n-1}_{\mathrm{\acute{e}t}}((\mathrm{Sh}'_{n_0} \times_{\operatorname{Spec} F} \mathrm{Sh}'_{n_1})_{\overline{F}}, O_{\lambda}(n))/(\mathfrak{n}_0, \mathfrak{n}_1) \to \bar{\mathrm{R}}^{(m_i)\mathsf{c}}$$

be the composition of Υ (8.7) with the projection to the *i*-th factor; and put

$$c_i \coloneqq \mathrm{H}^1(F, \Upsilon_i)(\mathrm{AJ}(\mathrm{Sh}'_{\mathrm{sp}})) \in \mathrm{H}^1(F, \mathrm{R}^{(m_i)c})$$

Then (8.8) implies

$$\max_{1 \le i \le \mu} \exp_{\lambda} \left(\partial_{\mathfrak{p}} \mathrm{loc}_{\mathfrak{p}}(c_i), \mathrm{H}^{1}_{\mathrm{sing}}(F_{\mathfrak{p}}, \bar{\mathrm{R}}^{(m_i)\mathsf{c}}) \right) \ge m - m_{\mathrm{per}} - m_{\mathrm{lat}}.$$

Without lost of generality, we obtain (8.3). On the other hand, as both Sh'_n and Sh'_{n+1} have smooth models over O_{F_w} for which (an analogue of) Lemma 4.1.4 holds, we obtain (8.4).

Lemma 8.2.2. Let Π_0 and Π_1 be as in Theorem 8.2.1 such that for $\alpha = 0, 1, \rho_{\Pi_{\alpha},\lambda}$ is absolutely irreducible for some prime λ of E. Suppose $L(\frac{1}{2}, \Pi_0 \times \Pi_1) \neq 0$. Then there exist

O a standard definite hermitian space V_n° of rank n over F, together with a self-dual $\prod_{v \notin \Sigma_{\infty}^+ \cup \Sigma_{\min}^+} O_{F_v}$ -lattice Λ_n° in $V_n^{\circ} \otimes_F \mathbb{A}_F^{\Sigma_{\infty}^+ \cup \Sigma_{\min}^+}$ (and put $V_{n+1}^{\circ} \coloneqq (V_n^{\circ})_{\sharp}$ and $\Lambda_{n+1}^{\circ} \coloneqq (\Lambda_n^{\circ})_{\sharp}$), O an object $(K_n^{\circ}, K_{n+1}^{\circ}) \in \mathfrak{K}(V_n^{\circ})_{sp}$ in which $K_{n_{\alpha}}^{\circ}$ is of the form

$$\mathbf{K}^{\circ}_{n_{\alpha}} = \prod_{v \in \Sigma^{+}_{\min}} (\mathbf{K}^{\circ}_{n_{\alpha}})_{v} \times \prod_{v \notin \Sigma^{+}_{\infty} \cup \Sigma^{+}_{\min}} \mathbf{U}(\Lambda^{\circ}_{n_{\alpha}})(O_{F^{+}_{v}})$$

for $\alpha = 0, 1$,

such that

$$\sum_{\in \operatorname{Sh}(\mathcal{V}_n^\circ, \mathcal{K}_n^\circ)} f(s, \operatorname{sh}_{\uparrow}(s)) \neq 0$$

for some element $f \in O_E[\operatorname{Sh}(\operatorname{V}_{n_0}^\circ, \operatorname{K}_{n_0}^\circ)][\ker \phi_{\Pi_0}] \otimes_{O_E} O_E[\operatorname{Sh}(\operatorname{V}_{n_1}^\circ, \operatorname{K}_{n_1}^\circ)][\ker \phi_{\Pi_1}].$

Proof. This follows from the direction $(1) \Rightarrow (2)$ of [BPLZZ, Theorem 1.7], together with [BPLZ-Z, Remark 4.15]. Note that since our Π_0 and Π_1 are relevant representations of $\operatorname{GL}_{n_0}(\mathbb{A}_F)$ and $\operatorname{GL}_{n_1}(\mathbb{A}_F)$, respectively, both members in the pair of hermitian spaces in (2) of [BPLZZ, Theorem 1.7] have to be standard definite.

Corollary 8.2.3. Keep the setup in Subsection 7.1. Suppose that

- (a) there exists a very special inert prime \mathfrak{p} of F^+ (Definition 3.3.4) such that $\Pi_{0,\mathfrak{p}}$ is Steinberg, and $\Pi_{1,\mathfrak{p}}$ is unramified whose Satake parameter contains 1 exactly once;
- (b) for $\alpha = 0, 1$, there exists a nonarchimedean place w_{α} of F such that $\Pi_{\alpha, w_{\alpha}}$ is supercuspidal; (c) $[F^+ : \mathbb{Q}] > 1$ if $n \ge 3$.
- If $L(\frac{1}{2}, \Pi_0 \times \Pi_1) \neq 0$, then for all but finitely many primes λ of E, we have

$$\mathrm{H}^{1}_{f}(F,\rho_{\Pi_{0},\lambda}\otimes_{E_{\lambda}}\rho_{\Pi_{1},\lambda}(n))=0.$$

Proof. By Theorem 8.2.1, it suffices to show that all but finitely many primes λ of E are admissible (or weakly admissible if n = 2). It suffices to show that each of conditions (L1–L6) in Definition 8.1.1 excludes only finitely many λ , and also for (L7) if $n \neq 3$.

For (L1) and (L2), this is trivial.

- For (L4), this follows from Lemma E.8.3 by (b).
- For (L3), this follows from Lemma 8.2.4 below by (L4) and (a).
- For (L6), this follows from Theorem E.8.4 by (a) and (b).
- For (L7), this follows from Corollary D.1.4 by (c).

For (L5-1), let λ be a prime of E satisfying (L4) and (L6), whose underlying rational prime is at least 2n(n+1)-1. Then by (a), $\bar{\rho}_{\Pi_0,\lambda}$ and $\bar{\rho}_{\Pi_1,\lambda}$ satisfy the assumptions in Lemma 8.2.4 below, with $k = O_{\lambda}/\lambda$ and $\Gamma = \Gamma_F$. Thus, by Lemma 8.2.4(2), assumption (b) of Lemma 2.3.4 hence (L5-1) hold.

For (L5-2), take an arithmetic Frobenius element $\phi_{\mathfrak{p}} \in \Gamma_{F_{\mathfrak{p}}^+}$. By Definition 3.3.4, $\phi_{\mathfrak{p}}$ belongs to $\operatorname{Gal}(\overline{F}/F_{\operatorname{rflx}}^+)$. For $\alpha = 0, 1$, put $r_{\alpha} \coloneqq \lfloor \frac{n_{\alpha}}{2} \rfloor$ as always. By (a), the Satake parameter of $\Pi_{0,\mathfrak{p}}$ is

 \square

 $\{p^{\pm 1}, \ldots, p^{\pm (2r_0-1)}\}$; and we may write the Satake parameter of $\Pi_{1,\mathfrak{p}}$ as $\{1, \alpha_1^{\pm 1}, \ldots, \alpha_{r_1}^{\pm 1}\}$ in which α_i is an algebraic number other than 1 for $1 \leq i \leq r_1$. For our purpose, we may replace E by a finite extension in \mathbb{C} so that $\alpha_i \in E$ for $1 \leq i \leq r_1$. By Proposition 3.2.4(1), we have $|\alpha_i| = 1$ for $1 \leq i \leq r_1$. Therefore, for all but finitely many prime λ of E, we have

- $O \{p, \alpha_1, \ldots, \alpha_{r_1}\}$ is contained in O_{λ}^{\times} ;
- $O \{p^{\pm 1} \mod \lambda, \dots, p^{\pm (2r_0-1)} \mod \lambda\} \text{ consists of distinct elements and does not contain } -1;$
- O { $\alpha_i \mod \lambda \mid 1 \le i \le r_1$ } is disjoint from { $1, -p, -p^{-1}$ };
- $O \{ p^{\pm 1}\alpha_i \mod \lambda, \dots, p^{\pm (2r_0 1)}\alpha_i \mod \lambda \mid 1 \le i \le r_1 \} \text{ is disjoint from } \{p, p^{-1}\}.$

Then for every prime λ satisfying (L4) and the above properties, (L5-2) is satisfied by taking the element $(\bar{\rho}_{0+}, \bar{\rho}_{1+}, \bar{\epsilon}_{\ell})(\phi_{\mathfrak{p}})$ in Lemma 2.6.1(3).

The corollary follows.

Lemma 8.2.4. Let Γ be a group, and k a field of characteristic either zero or at least 2n(n+1)-1. Let $\rho_0: \Gamma \to \operatorname{GL}_{n_0}(k)$ and $\rho_1: \Gamma \to \operatorname{GL}_{n_1}(k)$ be two homomorphisms that are absolutely irreducible. Suppose that there exists an element $t \in \Gamma$ such that $\rho_0(t) = 1 + J_{n_0}$ and $\rho_1(t) = 1$. Then we have

- (1) $\rho_0 \otimes \rho_1$ is absolutely irreducible;
- (2) $\rho_0 \otimes \rho_1$ is not a subquotient of $\operatorname{ad}(\rho_0 \otimes \rho_1)$.

Proof. We may assume that k is algebraically closed. For $\alpha = 0, 1$, let $V_i = k^{\oplus n_i}$ be the space which Γ acts on through ρ_{α} . By [Ser94, Corollaire 1], we know that both $\rho_0 \otimes \rho_1$ and $\operatorname{ad}(\rho_0 \otimes \rho_1)$ are semisimple.

For (1), thanks to the element t, every subrepresentation of $V_0 \otimes_k V_1$ is of the form $V_0 \otimes_k V'_1$ where V'_1 is a subrepresentation of V_1 . Since ρ_1 is irreducible, it follows that $\rho_0 \otimes \rho_1$ is irreducible.

For (2), note that $(\rho_0 \otimes \rho_1)(t)$ is conjugate to $(1 + J_{n_0})^{\oplus n_1}$. On the other hand, $\operatorname{ad}(\rho_0 \otimes \rho_1)(t)$ is conjugate to

$$\bigoplus_{i=1}^{n_0} (1+J_{2i-1})^{\oplus n_1^2}.$$

Since n_0 is even and $1, 3, \ldots, 2n_0 - 1$ are odd, $\rho_0 \otimes \rho_1$ is not a subquotient of $ad(\rho_0 \otimes \rho_1)$ as $ad(\rho_0 \otimes \rho_1)$ is semisimple.

The lemma is proved.

Corollary 8.2.5. Let $n \ge 2$ be an integer and denote by n_0 and n_1 the unique even and odd numbers in $\{n, n+1\}$, respectively. Let A_0 and A_1 be two modular elliptic curves over F^+ such that $\operatorname{End}(A_{0\overline{F}}) = \operatorname{End}(A_{1\overline{F}}) = \mathbb{Z}$. Suppose that

- (a) $A_{0\overline{F}}$ and $A_{1\overline{F}}$ are not isogenous to each other;
- (b) both $\operatorname{Sym}^{n_0-1} A_0$ and $\operatorname{Sym}^{n_1-1} A_1$ are modular; and
- (c) $[F^+:\mathbb{Q}] > 1$ if $n \ge 3$.

If the (central critical) L-value $L(n, \operatorname{Sym}^{n_0-1} A_{0F} \times \operatorname{Sym}^{n_1-1} A_{1F})$ does not vanish, then we have

$$\mathrm{H}^{1}_{f}(F, \mathrm{Sym}^{n_{0}-1} \mathrm{H}^{1}_{\mathrm{\acute{e}t}}(A_{0\overline{F}}, \mathbb{Q}_{\ell}) \otimes_{\mathbb{Q}_{\ell}} \mathrm{Sym}^{n_{1}-1} \mathrm{H}^{1}_{\mathrm{\acute{e}t}}(A_{1\overline{F}}, \mathbb{Q}_{\ell})(n)) = 0$$

for all but finitely many rational primes ℓ .

Proof. By (b) and [AC89], both $\operatorname{Sym}^{n_0-1} A_{0F}$ and $\operatorname{Sym}^{n_1-1} A_{1F}$ are modular. Thus, we may let Π_{α} be the (cuspidal) automorphic representation of $\operatorname{GL}_{n_{\alpha}}(\mathbb{A}_F)$ associated to $\operatorname{Sym}^{n_{\alpha}-1} A_{\alpha F}$ for $\alpha = 0, 1$, which is a relevant representation (Definition 1.1.3). We also have the identify

$$L(n+s, \operatorname{Sym}^{n_0-1} A_{0F} \times \operatorname{Sym}^{n_1-1} A_{1F}) = L(\frac{1}{2} + s, \Pi_0 \times \Pi_1)$$

of *L*-functions, and that the representation of Γ_F on $\operatorname{Sym}^{n_{\alpha}-1} \operatorname{H}^1_{\operatorname{\acute{e}t}}(A_{\alpha \overline{F}}, \mathbb{Q}_{\ell})$ is isomorphic to $\rho_{\Pi_{\alpha,\ell}}$ for $\alpha = 0, 1$. By Proposition 3.2.10 and (c), Hypothesis 3.2.9 is known in this case. Then the

corollary follows immediately from Theorem 8.2.1 and Proposition 8.1.5 (where we use (a) and (c)) with $E = \mathbb{Q}$.

Remark 8.2.6. In this remark, we summarize the current knowledge on the modularity of symmetric powers of elliptic curves, namely, condition (a) in Corollary 8.2.5. Let A be a modular elliptic curve over F^+ . We have

- O $\operatorname{Sym}^2 A$ is modular by [GJ76];
- O $\operatorname{Sym}^3 A$ is modular by [KS02];
- O $\operatorname{Sym}^4 A$ is modular by [Kim03];
- O Sym⁵ A and Sym⁶ A are modular if F^+ is linearly disjoint from $\mathbb{Q}(\zeta_5)$ over \mathbb{Q} ;
- O Sym⁷ A is modular if F^+ is linearly disjoint from $\mathbb{Q}(\zeta_{35})$ over \mathbb{Q} ;
- O Sym⁸ A is modular if F^+ is linearly disjoint from $\mathbb{Q}(\zeta_7)$ over \mathbb{Q} ;

in which the last three cases are obtained in a series of recent work [CT14, CT15, CT17] of Clozel and Thorne.

8.3. Main theorem in the Selmer rank 1 case. We state the following weak version of the arithmetic Gan–Gross–Prasad conjecture.

Conjecture 8.3.1. Suppose $L(\frac{1}{2}, \Pi_0 \times \Pi_1) = 0$ but $L'(\frac{1}{2}, \Pi_0 \times \Pi_1) \neq 0$. Then there exist

O a standard indefinite hermitian space V_n of rank n over F, together with a self-dual $\prod_{v \notin \Sigma_{\infty}^+ \cup \Sigma_{\min}^+} O_{F_v}$ -lattice Λ_n in $V_n \otimes_F \mathbb{A}_F^{\Sigma_{\infty}^+ \cup \Sigma_{\min}^+}$ (and put $V_{n+1} \coloneqq (V_n)_{\sharp}$ and $\Lambda_{n+1} \coloneqq (\Lambda_n)_{\sharp}$), O an object $(K_n, K_{n+1}) \in \mathfrak{K}(V_n)_{sp}$ in which $K_{n_{\alpha}}$ is of the form

$$\mathbf{K}_{n_{\alpha}} = \prod_{v \in \Sigma_{\min}^{+}} (\mathbf{K}_{n_{\alpha}})_{v} \times \prod_{v \notin \Sigma_{\infty}^{+} \cup \Sigma_{\min}^{+}} \mathbf{U}(\Lambda_{n_{\alpha}})(O_{F_{v}^{+}})$$

for $\alpha = 0, 1$,

such that for every prime λ of E, the graph $\Delta \operatorname{Sh}(V_n, K_n)$ of the morphism $\operatorname{sh}_{\uparrow} \colon \operatorname{Sh}(V_n, K_n) \to \operatorname{Sh}(V_{n+1}, K_{n+1})$ (4.6) is nonvanishing in the quotient Chow group

 $\operatorname{CH}^{n}(\operatorname{Sh}(\operatorname{V}_{n_{0}},\operatorname{K}_{n_{0}})\times_{\operatorname{Spec} F}\operatorname{Sh}(\operatorname{V}_{n_{1}},\operatorname{K}_{n_{1}}))_{E}/(\ker\phi_{\Pi_{0}},\ker\phi_{\Pi_{1}}).$

In the situation of the above conjecture, since both Π_0 and Π_1 are cuspidal, we have

$$\mathrm{H}^{i}_{\mathrm{\acute{e}t}}((\mathrm{Sh}(\mathrm{V}_{n_{0}},\mathrm{K}_{n_{0}})\times_{\mathrm{Spec}\,F}\mathrm{Sh}(\mathrm{V}_{n_{1}},\mathrm{K}_{n_{1}}))_{\overline{F}},E_{\lambda})/(\ker\phi_{\Pi_{0}},\ker\phi_{\Pi_{1}})=0$$

if $i \neq 2n - 1$. In particular, the Hochschild–Serre spectral sequence gives rise to a coboundary map

$$\begin{aligned} \operatorname{AJ}_{\lambda}^{\Pi_{0},\Pi_{1}} \colon \operatorname{Z}^{n}(\operatorname{Sh}(\operatorname{V}_{n_{0}},\operatorname{K}_{n_{0}}) \times_{\operatorname{Spec} F} \operatorname{Sh}(\operatorname{V}_{n_{1}},\operatorname{K}_{n_{1}})) \\ & \to \operatorname{H}^{1}(F,\operatorname{H}_{\operatorname{\acute{e}t}}^{2n-1}((\operatorname{Sh}(\operatorname{V}_{n_{0}},\operatorname{K}_{n_{0}}) \times_{\operatorname{Spec} F} \operatorname{Sh}(\operatorname{V}_{n_{1}},\operatorname{K}_{n_{1}}))_{\overline{F}}, E_{\lambda}(n))/(\ker \phi_{\Pi_{0}},\ker \phi_{\Pi_{1}})). \end{aligned}$$

Theorem 8.3.2. Keep the setup in Subsection 7.1. Assume Hypothesis 3.2.9 for both n and n+1. Let λ be a prime of E for which there exist

O a standard indefinite hermitian space V_n of rank n over F, together with a self-dual $\prod_{v \notin \Sigma_{\infty}^+ \cup \Sigma_{\min}^+} O_{F_v}$ -lattice Λ_n in $V_n \otimes_F \mathbb{A}_F^{\Sigma_{\infty}^+ \cup \Sigma_{\min}^+}$ (and put $V_{n+1} \coloneqq (V_n)_{\sharp}$ and $\Lambda_{n+1} \coloneqq (\Lambda_n)_{\sharp}$), O an object $(K_n, K_{n+1}) \in \mathfrak{K}(V_n)_{sp}$ in which $K_{n_{\alpha}}$ is of the form

$$\mathbf{K}_{n_{\alpha}} = \prod_{v \in \Sigma_{\min}^{+}} (\mathbf{K}_{n_{\alpha}})_{v} \times \prod_{v \notin \Sigma_{\infty}^{+} \cup \Sigma_{\min}^{+}} \mathbf{U}(\Lambda_{n_{\alpha}})(O_{F_{v}^{+}})$$

for $\alpha = 0, 1$, satisfying that $(K_{n_0})_v$ is a transferable open compact subgroup (Definition D.2.1) of $U(V_{n_0}^{\circ})(F_v^+)$ for $v \in \Sigma_{\min}^+$,

such that

(8.9)
$$\operatorname{AJ}_{\lambda}^{\Pi_{0},\Pi_{1}}(\bigtriangleup \operatorname{Sh}(\operatorname{V}_{n},\operatorname{K}_{n})) \neq 0$$

- -

If we further assume that either λ is admissible, or n = 2 and λ is weakly admissible and satisfies $H^i_{\text{\'et}}(Sh(V_3, K_3)_{\overline{F}}, O_E/\lambda) / \ker \phi_{\Pi_1} = 0$ for $i \neq 2$, then we have

 $\dim_{E_{\lambda}} \mathrm{H}^{1}_{f}(F, \rho_{\Pi_{0}, \lambda} \otimes_{E_{\lambda}} \rho_{\Pi_{1}, \lambda}(n)) = 1.$

Remark 8.3.3. In fact, (8.9) already implies that the global epsilon factor of $\Pi_0 \times \Pi_1$ is -1.

Proof of Theorem 8.3.2. We take a prime λ of E for which we may choose data V_n , Λ_n , (K_n, K_{n+1}) as in the statement of the theorem such that $AJ_{\lambda}^{\Pi_0,\Pi_1}(\Delta \operatorname{Sh}(V_n, K_n)) \neq 0$. We assume that λ satisfies either (a) or (b) of Lemma 8.1.4. Lemma 8.1.3 and (L2) imply that $AJ_{\lambda}^{\Pi_0,\Pi_1}(\Delta \operatorname{Sh}(V_n, K_n))$ belongs to the subspace

$$\mathrm{H}_{f}^{1}(F, \mathrm{H}_{\mathrm{\acute{e}t}}^{2n-1}((\mathrm{Sh}(\mathrm{V}_{n_{0}}, \mathrm{K}_{n_{0}}) \times_{\mathrm{Spec}\,F} \mathrm{Sh}(\mathrm{V}_{n_{1}}, \mathrm{K}_{n_{1}}))_{\overline{F}}, E_{\lambda}(n))/(\ker \phi_{\Pi_{0}}, \ker \phi_{\Pi_{1}}))$$

hence to the submodule

$$\mathrm{H}_{f}^{1}(F, \mathrm{H}_{\mathrm{\acute{e}t}}^{2n-1}((\mathrm{Sh}(\mathrm{V}_{n_{0}}, \mathrm{K}_{n_{0}}) \times_{\mathrm{Spec}\,F} \mathrm{Sh}(\mathrm{V}_{n_{1}}, \mathrm{K}_{n_{1}}))_{\overline{F}}, O_{\lambda}(n))/(\ker \phi_{\Pi_{0}}, \ker \phi_{\Pi_{1}}))$$

by Definition 2.4.2.

We adopt notation in Subsection 2.6 with the initial data in Definition 8.1.1. Define two nonnegative integers m_{per} and m_{lat} as follows.

(1) By Hypothesis 3.2.9, we may choose a map

$$\mathrm{H}^{2n-1}_{\mathrm{\acute{e}t}}((\mathrm{Sh}(\mathrm{V}_{n_{0}},\mathrm{K}_{n_{0}})\times_{\mathrm{Spec}\,F}\mathrm{Sh}(\mathrm{V}_{n_{1}},\mathrm{K}_{n_{1}}))_{\overline{F}},O_{\lambda}(n))/(\ker\phi_{\Pi_{0}},\ker\phi_{\Pi_{1}})\to\mathrm{R}^{\mathtt{c}}$$

of $O_{\lambda}[\Gamma_{F}]$ -modules such that the induced image of $AJ_{\lambda}^{\Pi_{0},\Pi_{1}}(\Delta \operatorname{Sh}(V_{n}, K_{n}))$ in $H_{f}^{1}(F, \mathbb{R}^{c})$, denoted by s^{c} , is non-torsion. Let $s \in H_{f}^{1}(F, \mathbb{R})$ be the element corresponding to s^{c} under the isomorphism in Lemma 2.4.5. We put

$$m_{\mathrm{per}} \coloneqq \mathrm{ord}_{\lambda}\left(s, \mathrm{H}_{f}^{1}(F, \mathrm{R})/\mathrm{H}_{f}^{1}(F, \mathrm{R})_{\mathrm{tor}}\right)$$

(Definition 2.1.6), which is a nonnegative integer.

(2) By Hypothesis 3.2.9, we have an isomorphism

$$\mathrm{H}^{2r_{1}}_{\mathrm{\acute{e}t}}(\mathrm{Sh}(\mathrm{V}_{n_{1}},\mathrm{K}_{n_{1}})_{\overline{F}},E_{\lambda}(r_{1}))/\ker\phi_{\Pi_{1}}\simeq(\mathrm{R}_{1}^{\mathsf{c}}\otimes_{O_{\lambda}}E_{\lambda})^{\oplus\mu_{1}}$$

of $E_{\lambda}[\Gamma_F]$ -modules for some integer $\mu_1 > 0$. We fix a map

$$\mathrm{H}^{2r_1}_{\mathrm{\acute{e}t}}(\mathrm{Sh}(\mathrm{V}_{n_1},\mathrm{K}_{n_1})_{\overline{F}},O_\lambda(r_1))/\ker\phi_{\Pi_1}\to (\mathrm{R}_1^{\mathsf{c}})^{\oplus\mu_1}$$

of $O_{\lambda}[\Gamma_F]$ -modules whose kernel and cokernel are both O_{λ} -torsion. Then we let m_{lat} be the smallest nonnegative integer such that both the kernel and the cokernel are annihilated by $\lambda^{m_{\text{lat}}}$.

Note that in (1), we obtain an element $s \in \mathrm{H}^{1}_{f}(F, \mathbb{R})_{\mathbb{Q}} = \mathrm{H}^{1}_{f}(F, \mathbb{R}_{\mathbb{Q}}) = \mathrm{H}^{1}_{f}(F, \rho_{\Pi_{0},\lambda} \otimes_{E_{\lambda}} \rho_{\Pi_{1},\lambda}(n))$ that is nonzero. In particular, we have $\dim_{E_{\lambda}} \mathrm{H}^{1}_{f}(F, \rho_{\Pi_{0},\lambda} \otimes_{E_{\lambda}} \rho_{\Pi_{1},\lambda}(n)) \geq 1$.

We start to prove the theorem by contradiction, hence assume

$$\dim_{E_{\lambda}} \mathrm{H}^{1}_{f}(F, \rho_{\Pi_{0}, \lambda} \otimes_{E_{\lambda}} \rho_{\Pi_{1}, \lambda}(n)) \geq 2.$$

Take a sufficiently large positive integer m which will be determined later. We fix a uniformizer λ_0 of E_{λ} . By Lemma 8.1.3, we may apply Proposition 2.4.6 by taking Σ to be the set of places of F above $\Sigma_{\min}^+ \cup \Sigma_{\ell}^+$. Then we obtain a submodule S of $\mathrm{H}^1_{f,\mathrm{R}}(F, \bar{\mathrm{R}}^{(m)})$ containing (the image of) $\lambda_0^{m_{\Sigma}-m_{\mathrm{per}}}s$ of order $0,^{22}$ that is free of rank 2 over $O_{\lambda}/\lambda^{m-m_{\Sigma}}$, and such that $\mathrm{loc}_w|_S = 0$ for every

²²Here, $\lambda_0^{-m_{\text{per}}}s$ is any element in $\mathrm{H}^1_f(F, \mathbf{R})$ satisfying $\lambda_0^{m_{\text{per}}}(\lambda_0^{-m_{\text{per}}}s) = s$.

nonarchimedean place $w \in \Sigma$ not above ℓ . Now we apply the discussion in Subsection 2.3 to the submodule $S \subseteq H^1(F, \overline{R}^{(m)})$. By (L5-1) and Lemma 2.3.6, we obtain an injective map

$$\theta_S \colon \operatorname{Gal}(F_S/F_{\bar{\rho}^{(m)}}) \to \operatorname{Hom}_{O_\lambda}(S, \mathbb{R}^{(m)})$$

whose image generates an O_{λ} -submodule containing $\lambda^{4\mathbf{r}_{\bar{\mathbf{R}}(m)}}$ Hom $_{O_{\lambda}}(S, \bar{\mathbf{R}}^{(m)})$, which further contains $\lambda^{4\mathbf{r}_{\mathbf{R}}}$ Hom $_{O_{\lambda}}(S, \bar{\mathbf{R}}^{(m)})$ by Lemma 2.3.3 and (L3). By (L5-2) and Lemma 2.6.1, we may choose an element $(\gamma_1, \gamma_2, \xi)$ in the image of $(\bar{\rho}_{1+}^{(m)}, \bar{\rho}_{2+}^{(m)}, \bar{\epsilon}_{\ell}^{(m)})|_{\mathrm{Gal}(\bar{F}/F_{\mathrm{rffx}}^+)}$ satisfying Lemma 2.6.1(2). It gives rise to an element $\gamma \in (\mathrm{GL}_{n_0n_1}(O_{\lambda}/\lambda^m) \times (O_{\lambda}/\lambda^m)^{\times})\mathfrak{c}$ as in Notation 2.5.2 such that $(\bar{\mathbf{R}}^{(m)})^{h_{\gamma}}$ is a free O_{λ}/λ^m -module of rank 1 by Lemma 2.6.2 and (2.4). Now we apply the discussion in Subsection 2.5. By Proposition 2.5.5 (with $m_0 = m_{\Sigma}, r_{\gamma} = 1, r_S = 2$), we may fix an (S, γ) -abundant pair $(\Psi_1, \Psi_2) \in G_{S,\gamma}^2$ (Definition 2.5.6). By the definition of a (S, γ) -abundant pair, we may choose a basis $\{s_1, s_2\}$ of S such that $\theta_S(\Psi_1)(s_2) = \theta_S(\Psi_2)(s_1) = 0$, and

(8.10)
$$\exp_{\lambda}\left(\theta_{S}(\Psi_{j})(s_{j}), \bar{\mathbf{R}}^{(m),h_{\gamma}}\right) \geq m - m_{\Sigma} - 4\mathfrak{r}_{R}$$

for j = 1, 2. Moreover, without lost of generality, we may assume $\lambda_0^{m_{\Sigma}-m_{\text{per}}}s = a_1s_1 + a_2s_2$ in which $a_1 \in O_{\lambda}^{\times}$.

First, we apply the discussion and notation in Subsection 7.3 to our situation with λ , m, $\Sigma_{\text{lr,II}}^+ = \emptyset$, $\Sigma_{\text{II}}^+ = \Sigma_{\min}^+$, (V_n, Λ_n) , K_n and (K_n, K_{n+1}) . By the Chebotarev density theorem, we can choose a γ -abundant place $w_{1+}^{(m)}$ of $F_+^{(m)}$ satisfying $\Psi_{w_1^{(m)}} = \Psi_1$ and whose underlying prime \mathfrak{p}_1 of F^+ (and the underlying rational prime p_1) is a special inert prime satisfying (PII1)–(PII7) and (**PII8**): the natural map

$$\mathrm{H}^{2r_{1}}_{\mathrm{\acute{e}t}}(\mathrm{Sh}(\mathrm{V}_{n_{1}},\mathrm{K}_{n_{1}})_{\overline{F}},O_{\lambda}(r_{1}))/(\mathbb{T}^{\Sigma_{\Pi}^{+}\cup\Sigma_{p_{1}}^{+}}_{n_{1}}\cap\ker\phi_{\Pi_{1}}) \to \mathrm{H}^{2r_{1}}_{\mathrm{\acute{e}t}}(\mathrm{Sh}(\mathrm{V}_{n_{1}},\mathrm{K}_{n_{1}})_{\overline{F}},O_{\lambda}(r_{1}))/\ker\phi_{\Pi_{1}})$$

is an isomorphism.

We also choose remaining data in the initial setup of Section 4 with $\mathbb{Q}_{p_1}^{\Phi} = \mathbb{Q}_{p_1^2}$, a definite uniformization datum $(V_{n_{\alpha}}^{\star}, \mathbf{i}_{n_{\alpha}, \mathfrak{q}}, \{\Lambda_{n_{\alpha}, \mathfrak{q}}^{\star}\}_{\mathfrak{q}|p_1})$ for $\alpha = 0, 1$ as in Notation 4.4.7. By (2.4), (8.10), and our choice of S, we have

$$\exp_{\lambda}\left(s, \mathrm{H}_{f}^{1}(F_{w_{1}}, \bar{\mathrm{R}}^{(m)})\right) \geq m - m_{\mathrm{per}} - 4\mathfrak{r}_{\mathrm{R}},$$

which implies that

$$\exp_{\lambda}\left(\operatorname{loc}_{\mathfrak{p}_{1}}([\bigtriangleup\operatorname{Sh}(\operatorname{V}_{n},\operatorname{K}_{n})]),\operatorname{H}^{2n}_{\operatorname{\acute{e}t}}((\operatorname{Sh}(\operatorname{V}_{n_{0}},\operatorname{K}_{n_{0}})\times_{\operatorname{Spec} F}\operatorname{Sh}(\operatorname{V}_{n_{1}},\operatorname{K}_{n_{1}}))_{F_{\mathfrak{p}_{1}}},L(n))/(\mathfrak{n}_{0},\mathfrak{n}_{1})\right)\geq m-m_{\operatorname{per}}-4\mathfrak{r}_{\operatorname{R}}.$$

Here, we recall that

$$\mathfrak{n}_{\alpha} = \mathbb{T}_{n_{\alpha}}^{\Sigma_{\mathrm{II}}^{+} \cup \Sigma_{p_{1}}^{+}} \cap \ker \left(\mathbb{T}_{n_{\alpha}}^{\Sigma_{\mathrm{min}}^{+}} \xrightarrow{\phi_{\Pi_{\alpha}}} O_{E} \to O_{E} / \lambda^{m} \right)$$

for $\alpha = 0, 1$. Note that Assumption 7.3.2 for $\alpha = 0$ is satisfied by (L7) if $n \ge 3$ and by (L4) if n = 2; Assumption 7.3.2 for $\alpha = 1$ is satisfied by (L7) if $n \ge 3$ and by (PII8) if n = 2. Thus, we may apply Theorem 7.3.4 hence obtain

(8.11)
$$\exp_{\lambda}\left(\mathbbm{1}_{\mathrm{Sh}(\mathrm{V}_{n}^{\star},\mathrm{K}_{\mathrm{sp}}^{\star})},O_{E}[\mathrm{Sh}(\mathrm{V}_{n_{0}}^{\star},\mathrm{K}_{n_{0}}^{\star})\times\mathrm{Sh}(\mathrm{V}_{n_{1}}^{\star},\mathrm{K}_{n_{1}}^{\star})]/(\mathfrak{n}_{0},\mathfrak{n}_{1})\right)\geq m-m_{\mathrm{per}}-4\mathfrak{r}_{\mathrm{R}}.$$

Second, we apply the discussion and notation in Subsection 7.2 to our situation with λ , m, $\Sigma_{\text{lr,I}}^+ = \{\mathfrak{p}_1\}, \Sigma_{\text{I}}^+ = \Sigma_{\min}^+ \cup \Sigma_{p_1}^+, V_n^\circ = V_n^\star, K_n^\circ = K_n^\star \text{ and } (K_{\text{sp}}^\circ, K_{n+1}^\circ) = (K_{\text{sp}}^\star, K_{n+1}^\star).$ By the Chebotarev density theorem, we can choose a γ -abundant place $w_{2+}^{(m)}$ of $F_+^{(m)}$ satisfying $\Psi_{w_2^{(m)}} = \Psi_2$ and whose underlying prime \mathfrak{p}_2 of F^+ (and the underlying rational prime p_2) is a special inert prime satisfying (PI1)–(PI7), $p_2 \neq p_1$, and

(PI8): the natural map

$$\mathrm{H}^{2r_{1}}_{\mathrm{\acute{e}t}}(\mathrm{Sh}(\mathrm{V}_{n_{1}},\mathrm{K}_{n_{1}})_{\overline{F}},O_{\lambda}(r_{1}))/(\mathbb{T}_{n_{1}}^{\Sigma_{1}^{+}\cup\Sigma_{p_{2}}^{+}}\cap\ker\phi_{\Pi_{1}})\to\mathrm{H}^{2r_{1}}_{\mathrm{\acute{e}t}}(\mathrm{Sh}(\mathrm{V}_{n_{1}},\mathrm{K}_{n_{1}})_{\overline{F}},O_{\lambda}(r_{1}))/\ker\phi_{\Pi_{1}})$$

is an isomorphism.

We claim that there exists an element $c_2 \in \mathrm{H}^1(F, \bar{\mathrm{R}}^{(m_2)c})$ for some positive integer $m_2 \leq m$ satisfying

(8.12)
$$\exp_{\lambda}\left(\partial_{\mathfrak{p}_{2}}\mathrm{loc}_{\mathfrak{p}_{2}}(c_{2}),\mathrm{H}^{1}_{\mathrm{sing}}(F_{\mathfrak{p}_{2}},\bar{\mathrm{R}}^{(m_{2})\mathfrak{c}})\right) \geq m-m_{\mathrm{per}}-4\mathfrak{r}_{\mathrm{R}}-m_{\mathrm{lat}};$$

and such that for every nonarchimedean place w of F not above $\Sigma^+ \cup \{\mathfrak{p}_1, \mathfrak{p}_2\},\$

$$\log_w(c_2) \in \mathrm{H}^1_f(F_w, \bar{\mathrm{R}}^{(m_2)\mathsf{c}})$$

holds.

By Remark 4.3.8 and Remark 4.4.8, we know that there exists an isomorphism $U((V_{n_1}^{\circ})^{\infty}) \simeq U(V_{n_1}^{\infty})$ sending $K_{n_1}^{\circ}$ to K_{n_1} . Then the claim can be proved by the exactly same argument for the similar claim in the proof of Theorem 8.2.1, using (8.11) and the fact that $\bar{\rho}_{\Pi_0,\lambda,+}$ is rigid for $(\Sigma_{\min}^+, \Sigma_{\ln}^+)^{23}$

Now we deduce a contradiction. Replace s_2 by its image in $\mathrm{H}^1_f(F, \bar{\mathrm{R}}^{(m_2)})$. We also identify $\bar{\mathrm{R}}^{(m_2)c}$ with $(\bar{\mathrm{R}}^{(m_2)})^*$ via the polarization Ξ . Now we compute the local Tate pairing $\langle s_2, c_2 \rangle_w$ (2.1) for every nonarchimedean place w of F.

- O Suppose that w is above Σ_{\min}^+ . Then we have $loc_w(s_2) = 0$ by our choice of S. Thus, $\langle s_2, c_2 \rangle_w = 0$.
- O Suppose that w is above Σ_{ℓ}^+ . Then by (L2), $\mathbb{R}_{\mathbb{Q}}$ is crystalline with Hodge–Tate weights in [1-n,n]. Thus, we have $\mathrm{loc}_w(s_2) \in \mathrm{H}^1_f(F_w, \mathbb{R}^{(m_2)})$ by Lemma 2.4.3(2) and (L1). By (8.13), Lemma 2.2.6 and (L1), we have $\lambda^{m_{\mathrm{dif}}} \langle s_2, c_2 \rangle_w = 0$ where $\mathfrak{d}_{\lambda} = \lambda^{m_{\mathrm{dif}}} \subseteq O_{\lambda}$ is the different ideal of $E_{\lambda}/\mathbb{Q}_{\ell}$.
- O Suppose that w is not above $\Sigma_{\min}^+ \cup \Sigma_{\ell}^+ \cup \{\mathfrak{p}_1, \mathfrak{p}_2\}$. Then by (L2), R is unramified. Thus, we have $\operatorname{loc}_w(s_2) \in \operatorname{H}^1_f(F_w, \overline{\mathsf{R}}^{(m_2)})$ by Lemma 2.4.3(1). By (8.13) and Lemma 2.2.3, we have $\langle s_2, c_2 \rangle_w = 0$.
- O Suppose that w is the unique place above \mathfrak{p}_1 . Then we have $loc_w(s_2) = 0$ by our choice of the basis $\{s_1, s_2\}$. Thus, we have $\langle s_2, c_2 \rangle_w = 0$.
- O Suppose that w is the unique place above \mathfrak{p}_2 . Then we have

$$\exp_{\lambda}\left(\operatorname{loc}_{w}(s_{2}), \operatorname{H}_{f}^{1}(F_{w}, \bar{\operatorname{R}}^{(m_{2})})\right) \geq m_{2} - m_{\Sigma} - 4\mathfrak{r}_{R}.$$

By (8.12) and Lemma 2.2.3 again, we have

$$\exp_{\lambda}\left(\langle s_2, c_2 \rangle_w, O_{\lambda}/\lambda^{m_2}\right) \ge m - m_{\text{per}} - m_{\text{lat}} - m_{\Sigma} - 8\mathfrak{r}_{\text{R}}$$

Therefore, as long as we take m such that $m > m_{per} + m_{lat} + m_{\Sigma} + 8\mathfrak{r}_{R} + m_{dif}$, we will have a contradiction to the relation

$$\sum_{w} \langle s_2, c_2 \rangle_w = 0,$$

where the sum is taken over all nonarchimedean places w of F. The theorem is proved.

²³In fact, one needs to use the additional fact that when $[F^+ : \mathbb{Q}] > 1$, both Shimura varieties Sh'_{n_0} and Sh'_{n_1} have proper smooth reduction at every place w of F above $\Sigma^+_{p_1} \setminus \{\mathfrak{p}_1\}$. See Remark 5.1.8.

APPENDIX A. UNITARY DELIGNE-LUSZTIG VARIETIES

In this appendix, we study some unitary Deligne–Lusztig varieties in Subsections A.1 and A.2 for those used in Sections 4 and 5, respectively.

We fix a rational prime p. Let κ be a field containing \mathbb{F}_{p^2} . Recall from Subsection 1.3 that we denote by $\sigma: S \to S$ the absolute p-power Frobenius morphism for schemes S in characteristic p.

A.1. Unitary Deligne–Lusztig varieties in the smooth case. In this subsection, we introduce certain Deligne–Lusztig varieties that appear in the special fiber of the smooth integral model studied in Section 4.

Consider a pair $(\mathscr{V}, \{,\})$ in which \mathscr{V} is a finite dimensional κ -linear space, and $\{,\}: \mathscr{V} \times \mathscr{V} \to \kappa$ is a (not necessarily non-degenerate) pairing that is (κ, σ) -linear in the first variable and κ -linear in the second variable. For every κ -scheme S, put $\mathscr{V}_S := \mathscr{V} \otimes_{\kappa} \mathcal{O}_S$. Then there is a unique pairing $\{,\}_S: \mathscr{V}_S \times \mathscr{V}_S \to \mathcal{O}_S$ extending $\{,\}$ that is (\mathcal{O}_S, σ) -linear in the first variable and \mathcal{O}_S -linear in the second variable. For a subbundle $H \subseteq \mathscr{V}_S$, we denote by $H^{\dashv} \subseteq \mathscr{V}_S$ its *right* orthogonal complement under $\{,\}_S$.

Definition A.1.1. We say that a pair $(\mathcal{V}, \{,\})$ is *admissible* if there exists an \mathbb{F}_{p^2} -linear subspace $\mathcal{V}_0 \subseteq \mathcal{V}_{\overline{\kappa}}$ such that the induced map $\mathcal{V}_0 \otimes_{\mathbb{F}_{p^2}} \overline{\kappa} \to \mathcal{V}_{\overline{\kappa}}$ is an isomorphism, and $\{x, y\} = -\{y, x\}^{\sigma}$ for every $x, y \in \mathcal{V}_0$.

Definition A.1.2. For a pair $(\mathcal{V}, \{,\})$ and an integer h, we define a presheaf

$$\mathrm{DL}(\mathscr{V}, \{,\}, h)$$

on $\operatorname{Sch}_{\kappa}$ such that for every $S \in \operatorname{Sch}_{\kappa}$, $\operatorname{DL}(\mathscr{V}, \{, \}, h)(S)$ is the set of subbundles H of \mathscr{V}_S of rank h such that $H^{\dashv} \subseteq H$. We call $\operatorname{DL}(\mathscr{V}, \{, \}, h)$ the *(unitary) Deligne-Lusztig variety* (see Proposition A.1.3 below) attached to $(\mathscr{V}, \{, \})$ of rank h.

Proposition A.1.3. Consider an admissible pair $(\mathcal{V}, \{,\})$. Put $N \coloneqq \dim_{\kappa} \mathcal{V}$ and $d \coloneqq \dim_{\kappa} \mathcal{V}^{\dashv}$.

- (1) If 2h < N + d or h > N, then $DL(\mathscr{V}, \{ , \}, h)$ is empty.
- (2) If $N + d \le 2h \le 2N$, then $DL(\mathscr{V}, \{, \}, h)$ is represented by a projective smooth scheme over κ of dimension (2h N d)(N h) with a canonical isomorphism for its tangent sheaf

$$\mathcal{T}_{\mathrm{DL}(\mathscr{V},\{\,,\,\},h)/\kappa} \simeq \mathcal{H}om\left(\mathcal{H}/\mathcal{H}^{\dashv},\mathscr{V}_{\mathrm{DL}(\mathscr{V},\{\,,\,\},h)}/\mathcal{H}\right)$$

where $\mathcal{H} \subseteq \mathscr{V}_{\mathrm{DL}(\mathscr{V}, \{,\},h)}$ is the universal subbundle.

(3) If $N + d < 2h \leq 2N$, then $DL(\mathcal{V}, \{,\}, h)$ is geometrically irreducible.

Proof. Part (1) is obvious from the definitions.

For (2), $DL(\mathscr{V}, \{, \}, h)$ is a closed sub-presheaf of the Grassmannian scheme $Gr(\mathscr{V}, h)$ classifying subbundles of \mathscr{V} of rank h, hence is represented by a projective scheme over κ . Now we compute the tangent sheaf. Consider a closed immersion $S \hookrightarrow \hat{S}$ in $\mathsf{Sch}_{/\kappa}$ defined by an ideal sheaf \mathcal{I} with $\mathcal{I}^2 = 0$. Take an object $H \subseteq \mathscr{V}_S$ in $DL(\mathscr{V}, \{, \}, h)(S)$. Let D_H and G_H be the subset of $DL(\mathscr{V}, \{, \}, h)(\hat{S})$ and $Gr(\mathscr{V}, h)(\hat{S})$ of elements that reduce to H, respectively. It is well-known that G_H is canonically a torsor over $\operatorname{Hom}_{\mathcal{O}_S}(H, (\mathscr{V}_S/H) \otimes_{\mathcal{O}_S} \mathcal{I})$. Since $\mathcal{I}^p = 0$, the right orthogonal complement \hat{H}^{\dashv} depends only on H for every $\hat{H} \in G_H$. In particular, the subset D_H is canonically a torsor over the subgroup $\operatorname{Hom}_{\mathcal{O}_S}(H/H^{\dashv}, (\mathscr{V}_S/H) \otimes_{\mathcal{O}_S} \mathcal{I})$ of $\operatorname{Hom}_{\mathcal{O}_S}(H, (\mathscr{V}_S/H) \otimes_{\mathcal{O}_S} \mathcal{I})$. Thus, $DL(\mathscr{V}, \{, \}, h)$ is smooth; and we have a canonical isomorphism for the tangent sheaf

$$\mathcal{T}_{\mathrm{DL}(\mathscr{V},\{\,,\,\},h)/\kappa}\simeq\mathcal{H}om\left(\mathcal{H}/\mathcal{H}^{\dashv},\mathscr{V}_{\mathrm{DL}(\mathscr{V},\{\,,\,\},h)}/\mathcal{H}
ight)$$

where \mathcal{H} is the universal subbundle. Note that this is a locally free $\mathcal{O}_{\mathrm{DL}(\mathscr{V}, \{,\},h)}$ -module of rank (2h - N - d)(N - h).
For (3), we may assume that κ is algebraically closed. By Definitions A.1.1 and A.1.2, we have a canonical isomorphism $\mathrm{DL}(\mathscr{V}, \{ \ , \ \}, h) \simeq \mathrm{DL}(\mathscr{V}_0, \{ \ , \ \}_0, h) \otimes_{\mathbb{F}_{p^2}} \kappa$, where $\{ \ , \ \}_0$ denotes the restriction of $\{ \ , \ \}$ to \mathscr{V}_0 . Suppose d = 0. Then $\{ \ , \ \}_0$ is non-degenerate. By [BR06, Theorem 1], we know that $\mathrm{DL}(\mathscr{V}_0, \{ \ , \ \}_0, h)$ is geometrically irreducible. In general, we consider $\mathscr{V}_0' \coloneqq \mathscr{V}_0/\mathscr{V}_0^{\dashv}$ equipped with a pairing $\{ \ , \ \}_0'$ induced from $\{ \ , \ \}_0$. Then it is clear that the morphism $\mathrm{DL}(\mathscr{V}_0, \{ \ , \ \}_0, h) \to \mathrm{DL}(\mathscr{V}_0', \{ \ , \ \}_0', h)$ sending a point $H \in \mathrm{DL}(\mathscr{V}_0, \{ \ , \ \}_0, h)(S)$ to $H/\mathscr{V}_{0S}^{\dashv}$ is an isomorphism. Thus, $\mathrm{DL}(\mathscr{V}_0, \{ \ , \ \}_0, h)$ is geometrically irreducible by the previous case. The proposition is proved.

Lemma A.1.4. Consider a pair $(\mathcal{V}, \{,\})$ with $\dim_{\kappa} \mathcal{V} = N \geq 2$ and $\dim_{\kappa} \mathcal{V}^{+} = 0$, and a *p*-coprime coefficient ring *L*. Suppose that p+1 is invertible in *L*.

- (1) The subscheme $DL(\mathcal{V}, \{,\}, N-1)$ is a hypersurface in $\mathbb{P}(\mathcal{V})$ of degree p+1.
- (2) The restriction map

 $\mathrm{H}^{i}_{\mathrm{\acute{e}t}}(\mathbb{P}(\mathscr{V})_{\overline{\kappa}}, L) \to \mathrm{H}^{i}_{\mathrm{\acute{e}t}}(\mathrm{DL}(\mathscr{V}, \{ , \}, N-1)_{\overline{\kappa}}, L)$

induced by the obvious inclusion $DL(\mathcal{V}, \{, \}, N-1) \to \mathbb{P}(\mathcal{V})$ is an isomorphism for $i \notin \{N-2, 2N-2\}.$

- (3) For every $i \in \mathbb{Z}$, $\mathrm{H}^{i}_{\mathrm{\acute{e}t}}(\mathrm{DL}(\mathscr{V}, \{,\}, N-1)_{\overline{\kappa}}, L)$ is a free L-module.
- (4) When N is even, the action of $\operatorname{Gal}(\overline{\kappa}/\kappa)$ on $\operatorname{H}^{N-2}_{\operatorname{\acute{e}t}}(\operatorname{DL}(\mathscr{V}, \{,\}, N-1)_{\overline{\kappa}}, L(\frac{N-2}{2}))$ is trivial.

Proof. The lemma is trivial if N = 2. Now we assume $N \ge 3$. Then $S := DL(\mathcal{V}, \{, \}, N-1)$ is a geometrically connected smooth hypersurface in $\mathbb{P}(\mathcal{V})$ by Proposition A.1.3.

Part (1) follows since S is defined by a homogenous polynomial of degree p+1, by its definition. For (2), by the Lefschetz hyperplane theorem, the restriction map $\mathrm{H}^{i}_{\mathrm{\acute{e}t}}(\mathbb{P}(\mathscr{V})_{\overline{\kappa}}, L) \to \mathrm{H}^{i}_{\mathrm{\acute{e}t}}(S_{\overline{\kappa}}, L)$ is an isomorphism for $0 \leq i \leq N-3$; and the Gysin map $\mathrm{H}^{i}_{\mathrm{\acute{e}t}}(S_{\overline{\kappa}}, L) \to \mathrm{H}^{i+2}_{\mathrm{\acute{e}t}}(\mathbb{P}(\mathscr{V})_{\overline{\kappa}}, L(1))$ is an isomorphism for $N-1 \leq i \leq 2(N-2)$. By (1), the composite map

$$\mathrm{H}^{i}_{\mathrm{\acute{e}t}}(\mathbb{P}(\mathscr{V})_{\overline{\kappa}},L) \to \mathrm{H}^{i}_{\mathrm{\acute{e}t}}(S_{\overline{\kappa}},L) \to \mathrm{H}^{i+2}_{\mathrm{\acute{e}t}}(\mathbb{P}(\mathscr{V})_{\overline{\kappa}},L(1))$$

is given by the cup product with $c_1(\mathcal{O}_{\mathbb{P}(\mathscr{V})_{\overline{\kappa}}}(p+1))$, which is an isomorphism for $i \neq 2N-2$ since p+1 is invertible in L. Thus, (2) follows.

Part (3) is an immediate consequence of (2).

For (4), it suffices to consider the case where $L = \mathbb{Q}_{\ell}$ for some $\ell \neq p$ by (3). Then it is well-known that $\mathrm{H}_{\mathrm{\acute{e}t}}^{N-2}(\mathrm{DL}(\mathscr{V}, \{, \}, N-1)_{\overline{\kappa}}, \mathbb{Q}_{\ell}(\frac{N-2}{2}))$ is spanned by Tate cycles over κ (see, for example, [HM78]). In particular, (4) follows.

Now we construct the special morphisms between Deligne–Lusztig varieties when rank increases.

Construction A.1.5. Let $(\mathscr{V}, \{ , \})$ be an admissible pair with $\dim_{\kappa} \mathscr{V} = n \geq 1$ satisfying $\dim \mathscr{V}^{\dashv} = n + 1 - 2\lfloor \frac{n+1}{2} \rfloor$. We put $\mathscr{V}_{\sharp} := \mathscr{V} \oplus \kappa 1$ and extend $\{ , \}$ to a pairing $\{ , \}_{\sharp}$ on \mathscr{V}_{\sharp} with $\{1, 1\}_{\sharp} = 0$. Suppose that we have another admissible pair $(\mathscr{V}_{\natural}, \{ , \}_{\natural})$ with $\dim_{\kappa} \mathscr{V}_{\natural} = n + 1$ satisfying $\dim \mathscr{V}_{\natural}^{\dashv} = n - 2\lfloor \frac{n}{2} \rfloor$, together with a κ -linear map $\delta \colon \mathscr{V}_{\sharp} \to \mathscr{V}_{\natural}$ of corank $\dim \mathscr{V}^{\dashv}$ such that $\{\delta(x), \delta(y)\}_{\natural} = \{x, y\}_{\sharp}$ for every $x, y \in \mathscr{V}_{\sharp}$. We construct a morphism

$$f_{\uparrow} \colon \mathrm{DL}(\mathscr{V}, \{ \ , \ \}, \lceil \frac{n+1}{2} \rceil) \to \mathrm{DL}(\mathscr{V}_{\natural}, \{ \ , \ \}_{\natural}, \lceil \frac{n+2}{2} \rceil)$$

by sending $H \in \mathrm{DL}(\mathscr{V}, \{,\}, h\lceil \frac{n+1}{2} \rceil)(S)$ to $\delta(H \oplus \mathcal{O}_S)$. We call δ_{\uparrow} a special morphism.

Proposition A.1.6. The morphism δ_{\uparrow} is well-defined, and is a regular embedding.

Proof. When n is odd, δ is an isomorphism, which implies that δ_{\uparrow} is well-defined an is an isomorphism.

When *n* is even, δ is of corank 1. The identity $\{\delta(x), \delta(y)\}_{\sharp} = \{x, y\}_{\sharp}$ for every $x, y \in \mathscr{V}_{\sharp}$ implies $\ker \delta \subset \mathscr{V}_{\sharp}^{\dashv} = \mathscr{V}^{\sharp} \oplus \kappa 1$. Take $S \in \mathsf{Sch}_{/\kappa}$. For $H \in \mathrm{DL}(\mathscr{V}, \{,\}, \lceil \frac{n+1}{2} \rceil)(S)$, $H \oplus \mathcal{O}_S$ must contain

 $\mathscr{V}_{\sharp}^{\dashv}$ hence $(\ker \delta)_S$. It follows that $\delta(H \oplus \mathcal{O}_S)$ has the same rank as H, which is $\lceil \frac{n+1}{2} \rceil = \lceil \frac{n+2}{2} \rceil$. Since $\{ , \}_{\natural}$ is nondegenerate, we have $\delta 1 \neq 0$ hence $(\delta \kappa 1)^{\dashv} = \delta \mathscr{V}_{\sharp}$. In particular, $(\delta (H \oplus \mathcal{O}_{S}))^{\dashv}$ is contained in $(\delta \mathscr{V}_{\sharp})_{S}$, which implies that $(\delta (H \oplus \mathcal{O}_{S}))^{\dashv} \subseteq \delta (H \oplus \mathcal{O}_{S})$. In other words, δ_{\uparrow} is well-defined. On the other hand, for $H_{\natural} \in \mathrm{DL}(\mathscr{V}_{\natural}, \{, \}_{\natural}, \lceil \frac{n+2}{2} \rceil)(S)$, whether $(\delta \kappa 1)_S \subseteq H \subseteq (\delta \mathscr{V}_{\sharp})_S$ holds is a closed condition; and once it does, there is a unique $H \in DL(\mathcal{V}, \{,\}, \lceil \frac{n+1}{2} \rceil)(S)$ such that $H_{\natural} = \delta(H \oplus \mathcal{O}_S)$. Thus, δ_{\uparrow} is a regular embedding by Proposition A.1.3(2).

The proposition is proved.

A.2. Unitary Deligne–Lusztig varieties in the semistable case. In this subsection, we introduce certain Deligne–Lusztig varieties that appear in the special fiber of the semistable integral model studied in Section 5. We keep the notation from the previous subsection.

Definition A.2.1. For a pair $(\mathcal{V}, \{,\})$ with $\dim_{\kappa} \mathcal{V} = N$, we define a presheaf

 $DL^{\bullet}(\mathscr{V}, \{,\})$

on Sch_{κ} such that for every $S \in \mathsf{Sch}_{\kappa}$, $\mathsf{DL}^{\bullet}(\mathscr{V}, \{,\})(S)$ is the set of pairs (H_1, H_2) of subbundles of \mathscr{V}_S of ranks $\lceil \frac{N}{2} \rceil$ and $\lceil \frac{N}{2} \rceil - 1$, respectively, satisfying the following inclusion relations



of subbundles of \mathscr{V}_S .

Proposition A.2.2. Consider an admissible pair $(\mathcal{V}, \{,\})$. Put $N \coloneqq \dim_{\kappa} \mathcal{V}$ and $d \coloneqq \dim_{\kappa} \mathcal{V}^{\dashv}$.

- (1) If $d \ge \lceil \frac{N}{2} \rceil$, then $\mathrm{DL}^{\bullet}(\mathscr{V}, \{ , \})$ is empty.
- (2) If $d \leq \left\lceil \frac{N}{2} \right\rceil 1$, then $\mathrm{DL}^{\bullet}(\mathscr{V}, \{,\})$ is represented by a projective smooth scheme over κ , whose tangent sheaf fits canonically into a sequence

$$0 \to \mathcal{H}om\left(\mathcal{H}_1/\mathcal{H}_2, \mathcal{H}_2^{\dashv}/\mathcal{H}_1\right) \to \mathcal{T}_{\mathrm{DL}^{\bullet}(\mathscr{V}, \{,\})/\kappa} \to \mathcal{H}om(\mathcal{H}_2/\mathscr{V}_{\mathrm{DL}^{\bullet}(\mathscr{V}, \{,\})}, \mathcal{H}_1^{\dashv}/\mathcal{H}_2) \to$$

- where $\mathscr{V}_{\mathrm{DL}^{\bullet}(\mathscr{V},\{,\})}^{\dashv} \subseteq \mathscr{H}_2 \subseteq \mathscr{H}_1 \subseteq \mathscr{V}_{\mathrm{DL}^{\bullet}(\mathscr{V},\{,\})}$ are the universal subbundles.
- (3) If $N \ge 2$ and $d = N 2\lfloor \frac{N}{2} \rfloor$, then $\mathrm{DL}^{\bullet}(\mathscr{V}, \{,\})$ is geometrically irreducible of dimension $\left|\frac{N}{2}\right|$.

Proof. Part (1) is obvious from the definitions.

For (2), let $\operatorname{Gr}(\mathscr{V}, r)$ denote by the Grassmannian variety that classifies subspaces of \mathscr{V} of dimension r. Then $DL^{\bullet}(\mathscr{V}, \{,\})$ is a closed sub-presheaf of $Gr(\mathscr{V}, \lceil \frac{N}{2} \rceil) \times Gr(\mathscr{V}, \lceil \frac{N}{2} \rceil - 1)$, hence it is represented by a projective scheme over κ . Now we prove that $DL^{\bullet}(\mathscr{V}, \{,\})$ is smooth and compute its tangent sheaf. Consider a closed immersion $S \hookrightarrow \hat{S}$ in $\mathsf{Sch}_{/\kappa}$ defined by an ideal sheaf \mathcal{I} with $\mathcal{I}^2 = 0$. Take an object $\mathscr{V}_S^{\dashv} \subseteq H_2 \subseteq H_1 \subseteq \mathscr{V}_S$ in $\mathrm{DL}^{\bullet}(\mathscr{V}, \{,\})(S)$. To lift (H_1, H_2) to a pair $(\hat{H}_1, \hat{H}_2) \in DL^{\bullet}(\mathscr{V}, \{,\})(\hat{S})$, we first lift H_2 , which is canonically a torsor under the group $\operatorname{Hom}_{\mathcal{O}_S}(H_2/\mathscr{V}_S^{\dashv}, (H_1^{\dashv}/H_2) \otimes_{\mathcal{O}_S} \mathcal{I})$ as \hat{H}_1^{\dashv} depends only on H_1^{\dashv} . Once such a lift \hat{H}_2 is given, the possible lifts of H_1 form a torsor under the group $\operatorname{Hom}_{\mathcal{O}_S}(H_1/H_2, (H_2^{\dashv}/H_1) \otimes_{\mathcal{O}_S} \mathcal{I})$. In particular, Zariski locally, there is no obstruction to lifting (H_1, H_2) , hence $DL^{\bullet}(\mathscr{V}, \{,\})$ is smooth. The statement on the tangent bundle of $\mathrm{DL}^{\bullet}(\mathscr{V}, \{ \ , \ \})$ follows immediately from the discussion by considering the universal object on $DL^{\bullet}(\mathcal{V}, \{,\})$.

For (3), similar to the argument for Proposition A.1.3(3), we may assume N even this time. Then the statement follows again by [BR06, Theorem 1].

Construction A.2.3. Let $(\mathcal{V}, \{,\})$ be an admissible pair with $\dim_{\kappa} \mathcal{V} = n \geq 2$ satisfying $\dim_{\kappa} \mathcal{V}^{\dashv} = n - 2\lfloor \frac{n}{2} \rfloor$. We put $\mathcal{V}_{\sharp} := \mathcal{V} \oplus \kappa 1$ and extend $\{,\}$ to a pairing $\{,\}_{\sharp}$ on \mathcal{V}_{\sharp} with $\{1,1\}_{\sharp} = 0$. Suppose that we have another admissible pair $(\mathcal{V}_{\natural}, \{,\}_{\natural})$ with $\dim_{\kappa} \mathcal{V}_{\natural} = n + 1$ satisfying $\dim \mathcal{V}_{\natural}^{\dashv} = n + 1 - 2\lfloor \frac{n+1}{2} \rfloor$, together with a κ -linear map $\delta \colon \mathcal{V}_{\sharp} \to \mathcal{V}_{\natural}$ of corank $\dim \mathcal{V}^{\dashv}$ such that $\{\delta(x), \delta(y)\}_{\natural} = \{x, y\}_{\sharp}$ for every $x, y \in \mathcal{V}_{\sharp}$. Then similar to Construction A.1.5 and Proposition A.1.6, we have a morphism

$$\delta_{\uparrow} \colon \mathrm{DL}^{\bullet}(\mathscr{V}, \{ \ , \ \}) \to \mathrm{DL}^{\bullet}(\mathscr{V}_{\natural}, \{ \ , \ \}_{\natural})$$

by sending $(H_1, H_2) \in \mathrm{DL}^{\bullet}(\mathscr{V}, \{,\})(S)$ to $(\delta(H_1 \oplus \mathcal{O}_S), \delta(H_2 \oplus \mathcal{O}_S)) \in \mathrm{DL}^{\bullet}(\mathscr{V}_{\natural}, \{,\}_{\natural})(S)$, which is a regular embedding.

Proposition A.2.4. Suppose κ algebraically closed. Consider an admissible pair $(\mathcal{V}, \{,\})$ over κ . Let $(\mathcal{H}_1, \mathcal{H}_2)$ be the universal object over $DL^{\bullet}(\mathcal{V}, \{,\})$.

(1) Suppose $\dim_{\kappa} \mathscr{V} = 2r + 1$ for some integer $r \geq 1$ and $\dim_{\kappa} \mathscr{V}^{\dashv} = 1$. Then we have

$$\int_{\mathrm{DL}^{\bullet}(\mathscr{V},\{\,,\,\})} c_r\left((\sigma^*\mathcal{H}_2) \otimes_{\mathcal{O}_{\mathrm{DL}^{\bullet}(\mathscr{V},\{\,,\,\})}} \left(\mathcal{H}_1^{\dashv}/\mathcal{H}_2\right)\right) = \mathsf{d}_{r,p}^{\bullet}$$

(2) Suppose $\dim_{\kappa} \mathscr{V} = 2r$ for some integer $r \geq 1$ and $\dim_{\kappa} \mathscr{V}^{\dashv} = 0$. Then we have

$$\int_{\mathrm{DL}^{\bullet}(\mathscr{V},\{,\})} c_{r-1}\left((\sigma^{*}\mathcal{H}_{2}) \otimes_{\mathcal{O}_{\mathrm{DL}^{\bullet}(\mathscr{V},\{,\})}} \left(\mathcal{H}_{1}^{\dashv}/\mathcal{H}_{2}\right)\right) \cdot c_{1}\left(\mathcal{H}_{1}^{\dashv}/\mathcal{H}_{2}\right) = \mathsf{d}_{r,p}^{\bullet}$$

Here, $\mathbf{d}_{r,p}^{\bullet}$ is the number introduced in Notation 1.3.2.

Note that $DL^{\bullet}(\mathcal{V}, \{,\})$ is irreducible of dimension r, by Proposition A.2.2.

Proof. For (1), we let $\bar{\mathscr{V}}$ be the quotient space $\mathscr{V}/\mathscr{V}^{\dashv}$, equipped with the induced pairing, which we still denote by $\{, \}$. Then we have a canonical isomorphism $\mathrm{DL}^{\bullet}(\mathscr{V}, \{, \}) \xrightarrow{\sim} \mathrm{DL}^{\bullet}(\bar{\mathscr{V}}, \{, \})$ by sending a pair (H_1, H_2) to $(H_1/\mathscr{V}^{\dashv}, H_2/\mathscr{V}^{\dashv})$. If we denote by $(\bar{\mathcal{H}}_1, \bar{\mathcal{H}}_2)$ the universal object over $\mathrm{DL}^{\bullet}(\bar{\mathscr{V}}, \{, \})$. Then we have

$$c_r\left((\sigma^*\mathcal{H}_2)\otimes_{\mathcal{O}_{\mathrm{DL}^{\bullet}(\mathscr{V},\{,\})}}\left(\mathcal{H}_1^{\dashv}/\mathcal{H}_2\right)\right) = c_{r-1}\left(\left(\sigma^*\bar{\mathcal{H}}_2\right)\otimes_{\mathcal{O}_{\mathrm{DL}^{\bullet}(\check{\mathscr{V}},\{,\})}}\left(\bar{\mathcal{H}}_1^{\dashv}/\bar{\mathcal{H}}_2\right)\right) \cdot c_1\left(\bar{\mathcal{H}}_1^{\dashv}/\bar{\mathcal{H}}_2\right)$$

under the above isomorphism. Therefore, (1) follows from (2).

For (2), consider $\mathscr{V}_{\sharp} := \mathscr{V} \oplus \kappa 1$ and extend $\{ , \}$ to a pairing $\{ , \}_{\sharp}$ on \mathscr{V}_{\sharp} with $\{1, 1\}_{\sharp} = 1$. Then we have Deligne–Lusztig varieties $DL(\mathscr{V}_{\sharp}, \{ , \}_{\sharp}, h)$. In what follows, we only need to study the one with h = r + 1, and will simply write $DL(\mathscr{V}_{\sharp})$ for $DL(\mathscr{V}_{\sharp}, \{ , \}_{\sharp}, r + 1)$. Since we will work with two spaces, we will denote by (\vdash, \dashv) for the (left, right) orthogonal complement for \mathscr{V} , and (\vDash, \dashv) for the (left, right) orthogonal complement for \mathscr{V}_{\sharp} .

We now define a correspondence

$$\mathrm{DL}(\mathscr{V}_{\sharp}) \xleftarrow{\pi} \widetilde{\mathrm{DL}}(\mathscr{V}) \xrightarrow{\pi^{\bullet}} \mathrm{DL}^{\bullet}(\mathscr{V})$$

of schemes over κ . For every κ -scheme S,

- O $DL(\mathscr{V})(S)$ is the set of pairs (H, H_2) where H is an element in $DL(\mathscr{V}_{\sharp})(S)$ and H_2 is a subbundle of H^{\models} of rank r-1 that is contained in \mathscr{V}_S ;
- O π sends $(H, H_2) \in DL(\mathscr{V})(S)$ to $H \in DL(\mathscr{V})(S)$; and
- O π^{\bullet} sends $(H, H_2) \in DL(\mathscr{V})(S)$ to $(H_1, H_2) \in DL^{\bullet}(\mathscr{V})(S)$ where $H_1 := (H \cap \mathscr{V}_S)^{\vdash}$.

It needs to show that π^{\bullet} is well-defined, which amounts to the following four statements:

- O H_1 is a subbundle of \mathscr{V}_S : It suffices to show that the composite map $H \to \mathscr{V}_{\sharp S} \to \mathcal{O}_S$ is surjective, where the latter map is induced by the projection $\mathscr{V}_{\sharp} \to \kappa 1$. If not, then there exists a geometric point s of S such that H_s is contained in \mathscr{V}_s , which contradicts the inclusion $H_s^{\dashv} \subseteq H_s$.
- O $H_2 \subseteq H_1$: As $H^{\ddagger} \subseteq H$ by the definition of DL(\mathscr{V}_{\sharp}), we have $H^{\models} \subseteq H$ and $\{H^{\models}, H\}_{\sharp} = 0$. Thus, $\{H^{\models} \cap \mathscr{V}_S, H \cap \mathscr{V}_S\} = 0$, which implies $H_2 \subseteq H^{\models} \cap \mathscr{V}_S \subseteq (H \cap \mathscr{V}_S)^{\vdash} = H_1$.
- O $H_1 \subseteq H_2^{\vdash}$: As $H^{\models} \subseteq H$, we have that $H_1^{\dashv} = H \cap \mathscr{V}_S$ contains H_2 , which implies $H_1 = (H_1^{\dashv})^{\vdash} \subseteq H_2^{\vdash}$.
- $\begin{array}{l} \bigcirc \ H_1 \subseteq H_2^{\dashv}: \mbox{ As } H^{\dashv} \subseteq H, \mbox{ we have } (H^{\models})^{\dashv \dashv} \cap \mathscr{V}_S \subseteq H \cap \mathscr{V}_S, \mbox{ which is equivalent to } (H^{\models} \cap \mathscr{V}_S)^{\dashv \dashv} \subseteq H \cap \mathscr{V}_S. \mbox{ As } H_2 \mbox{ is contained in } H^{\models} \cap \mathscr{V}_S, \mbox{ we have } H_2^{\dashv \dashv} \subseteq H \cap \mathscr{V}_S = H_1^{\dashv}, \mbox{ which implies } H_1 \subseteq H_2^{\dashv}. \end{array}$

We denote by \mathcal{H} , $(\widetilde{\mathcal{H}}, \widetilde{\mathcal{H}}_2)$, and $(\mathcal{H}_1, \mathcal{H}_2)$ the universal objects over $\mathrm{DL}(\mathscr{V}_{\sharp})$, $\widetilde{\mathrm{DL}}(\mathscr{V})$, and $\mathrm{DL}^{\bullet}(\mathscr{V})$, respectively. By definition, we have $\widetilde{\mathcal{H}} = \pi^* \mathcal{H}$ and $\widetilde{\mathcal{H}}_2 = \pi^{\bullet*} \mathcal{H}_2$.

We first study the morphism π . We say that a point $s \in \mathrm{DL}(\mathscr{V}_{\sharp})(\kappa)$ represented by H_s is special if H_s^{\vDash} is a maximal isotropic subspace of \mathscr{V} satisfying $H_s^{\dashv} = H_s^{\vDash}$. Then there are exactly $(p+1)(p^3+1)\cdots(p^{2r-1}+1)$ special points. Let $\mathrm{DL}(\mathscr{V}_{\sharp})'$ be the locus of special points. It is clear that for every morphism $S \to \mathrm{DL}(\mathscr{V}_{\sharp}) \setminus \mathrm{DL}(\mathscr{V}_{\sharp})'$, $\pi^{-1}(S)$ is a singleton; and for a special point s, we have $\pi^{-1}(s) = \mathbb{P}(H_s^{\vDash}) \simeq \mathbb{P}_{\kappa}^{r-1}$. In particular, π is a blow-up along $\mathrm{DL}(\mathscr{V}_{\sharp})'$, for which we denote by $E \subseteq \widetilde{\mathrm{DL}}(\mathscr{V})$ the exceptional divisor. In particular, π is projective. Moreover, E is exactly the zero locus of the canonical projection map

$$\widetilde{\mathcal{H}}^{\models}/\widetilde{\mathcal{H}}_{2} \to \mathcal{O}_{\widetilde{\mathrm{DL}}(\mathscr{V})} \cdot 1 \subseteq \mathcal{O}_{\widetilde{\mathrm{DL}}(\mathscr{V})} \otimes_{\kappa} \mathscr{V}_{\sharp},$$

which implies

(A.1)
$$\widetilde{\mathcal{H}}^{\models}/\widetilde{\mathcal{H}}_{2} \simeq \mathcal{O}_{\widetilde{\mathrm{DL}}(\mathscr{V})}(-E).$$

Next we study the morphism π^{\bullet} . We claim that π^{\bullet} is generically finite of degree p + 1. Take a point $s \in \mathrm{DL}^{\bullet}(\mathscr{V})(\kappa)$ represented by (H_{1s}, H_{2s}) . Then by construction, for every scheme S over $\{s\} \times_{\mathrm{DL}^{\bullet}(\mathscr{V})} \widetilde{\mathrm{DL}}(\mathscr{V}), \widetilde{\mathrm{DL}}(\mathscr{V})(S)$ consists of subbundles $H \subseteq \mathscr{V}_{\sharp} \otimes_{\kappa} \mathcal{O}_{S}$ satisfying $H_{2s} \otimes_{\kappa} \mathcal{O}_{S} \subseteq H^{\models} \subseteq$ $H_{1s} \otimes_{\kappa} \mathcal{O}_{S} \oplus \mathcal{O}_{S} 1$ and $H^{\models} \subseteq H$. Note that we have an induced pairing

$$\{\ ,\ \}_s \colon \frac{H_{1s} \oplus \kappa 1}{H_{2s}} \times \frac{H_{1s} \oplus \kappa 1}{H_{2s}} \to \kappa$$

that is σ -linear in the first variable and linear in the second variable. Then it is clear that when $\{ , \}_s$ is perfect, $\{s\} \times_{\mathrm{DL}^{\bullet}(\mathscr{V})} \widetilde{\mathrm{DL}}(\mathscr{V})$ is isomorphic to the union of p+1 copies of Spec κ . However, $\{ , \}_s$ fails to be perfect if and only if $H_1^{\dashv} = H_1$. Thus, the locus where $\{ , \}_s$ fails to be perfect is a finite union of $\mathbb{P}_{\kappa}^{r-1}$. Therefore, π^{\bullet} is generically finite of degree p+1.

To proceed, we introduce two more bundles

$$\mathcal{E} \coloneqq \left(\sigma^* \mathcal{H}^{\vDash}\right) \otimes_{\mathrm{DL}(\mathscr{V}_{\sharp})} \left(\mathcal{H}/\mathcal{H}^{\vDash}\right), \quad \mathcal{E}^{\bullet} \coloneqq \left(\sigma^* \mathcal{H}_2\right) \otimes_{\mathrm{DL}^{\bullet}(\mathscr{V})} \left(\mathcal{H}_1^{\dashv}/\mathcal{H}_2\right)$$

on $DL(\mathscr{V}_{\sharp})$ and $DL^{\bullet}(\mathscr{V})$ of ranks r and r-1, respectively.

We claim that

(A.2)
$$\mathcal{L} \coloneqq \pi^{\bullet*} \left(\mathcal{H}_1^{\dashv} / \mathcal{H}_2 \right) \simeq \mathcal{O}_{\widetilde{\mathrm{DL}}(\mathscr{V})}(-E) \otimes_{\mathcal{O}_{\widetilde{\mathrm{DL}}(\mathscr{V})}} \left(\widetilde{\mathcal{H}} / \widetilde{\mathcal{H}}^{\vDash} \right).$$

In fact, we have

$$\mathcal{L} = \left(\widetilde{\mathcal{H}} \cap \mathscr{V}_{\widetilde{\mathrm{DL}}(\mathscr{V})}\right) / \widetilde{\mathcal{H}}_2$$

by definition. Thus, the claim follows from the following injective map



of short exact sequences of coherent sheaves on $\widetilde{\mathrm{DL}}(\mathscr{V})$ by (A.1) and the Snake Lemma. By (A.1) and (A.2), we have

$$\begin{aligned} \pi^* \left(c_r(\mathcal{E}) \right) \\ &= c_r \left(\pi^* \mathcal{E} \right) \\ &= c_{r-1} \left(\left(\sigma^* \widetilde{\mathcal{H}}_2 \right) \otimes_{\mathcal{O}_{\widetilde{\mathrm{DL}}(\mathcal{V})}} \left(\mathcal{H}/\mathcal{H}^{\models} \right) \right) \cdot c_1 \left(\mathcal{O}_{\widetilde{\mathrm{DL}}(\mathcal{V})} (-pE) \otimes_{\mathcal{O}_{\widetilde{\mathrm{DL}}(\mathcal{V})}} \left(\mathcal{H}/\mathcal{H}^{\models} \right) \right) \\ &= c_{r-1} \left(\left(\sigma^* \widetilde{\mathcal{H}}_2 \right) \otimes_{\mathcal{O}_{\widetilde{\mathrm{DL}}(\mathcal{V})}} \mathcal{L}(E) \right) \cdot c_1 (\mathcal{L}((1-p)E)) \\ &= c_{r-1} \left(\pi^{\bullet*} \mathcal{E}^{\bullet} \otimes_{\mathcal{O}_{\widetilde{\mathrm{DL}}(\mathcal{V})}} \mathcal{O}_{\widetilde{\mathrm{DL}}(\mathcal{V})} (E) \right) \cdot c_1 (\mathcal{L}((1-p)E)) \\ &= \left(c_{r-1} \left(\pi^{\bullet*} \mathcal{E}^{\bullet} \right) + \sum_{i=1}^{r-1} c_1(E)^i c_{r-i-1} \left(\pi^{\bullet*} \mathcal{E}^{\bullet} \right) \right) \cdot (c_1(\mathcal{L}) + (1-p)c_1(E)) \\ &= c_{r-1} \left(\pi^{\bullet*} \mathcal{E}^{\bullet} \right) \cdot c_1(\mathcal{L}) + \sum_{i=1}^{r-1} c_1(E)^i c_1(\mathcal{L}) c_{r-i-1} \left(\pi^{\bullet*} \mathcal{E}^{\bullet} \right) + (1-p) \sum_{i=1}^r c_1(E)^i c_{r-i} \left(\pi^{\bullet*} \mathcal{E}^{\bullet} \right) \\ &= \pi^{\bullet*} \left(c_{r-1}(\mathcal{E}^{\bullet}) \cdot c_1 \left(\mathcal{H}_1^{\dashv} / \mathcal{H}_2 \right) \right) + \sum_{i=1}^{r-1} c_1(E)^i c_1(\mathcal{L}) c_{r-i-1} \left(\pi^{\bullet*} \mathcal{E}^{\bullet} \right) + (1-p) \sum_{i=1}^r c_1(E)^i c_{r-i} \left(\pi^{\bullet*} \mathcal{E}^{\bullet} \right) \end{aligned}$$

In particular, since π and π^{\bullet} are generically finite of degrees 1 and p+1, respectively, we have

(A.3)
$$(p+1) \int_{\mathrm{DL}^{\bullet}(\mathscr{V})} c_{r-1}(\mathscr{E}) \cdot c_1 \left(\mathcal{H}_1^{\dagger}/\mathcal{H}_2\right) - \int_{\mathrm{DL}(\mathscr{V}_\sharp)} c_r(\mathscr{E})$$
$$= (p-1) \sum_{i=1}^r \int_{\widetilde{\mathrm{DL}}(\mathscr{V})} c_1(E)^i c_{r-i} \left(\pi^{\bullet*}\mathscr{E}^{\bullet}\right) - \sum_{i=1}^{r-1} \int_{\widetilde{\mathrm{DL}}(\mathscr{V})} c_1(E)^i c_1(\mathscr{L}) c_{r-i-1} \left(\pi^{\bullet*}\mathscr{E}^{\bullet}\right)$$
$$= (p-1) \sum_{i=0}^{r-1} \int_E (-\eta)^i c_{r-i-1} \left(\pi^{\bullet*}\mathscr{E}^{\bullet}|_E\right) - \sum_{i=0}^{r-2} \int_E (-\eta)^i c_1(\mathscr{L}|_E) c_{r-i-2} \left(\pi^{\bullet*}\mathscr{E}^{\bullet}|_E\right)$$

where $\eta \coloneqq c_1(\mathcal{O}_E(1))$. As $\widetilde{\mathcal{H}}/\widetilde{\mathcal{H}}^{\vDash} = \pi^* \left(\mathcal{H}/\mathcal{H}^{\vDash}\right)$, we have $\mathcal{L}|_E \simeq \mathcal{O}_E(-E) = \mathcal{O}_E(1)$. On the other hand, $\widetilde{\mathcal{H}}_2|_E$ is the tautological subbundle (of rank r-1), which satisfies the short exact sequence

$$0 \to \widetilde{\mathcal{H}}_2|_E \to \mathcal{O}_E^{\oplus r} \to \mathcal{O}_E(1) \to 0.$$

Thus, $\mathcal{F} \coloneqq \pi^{\bullet *} \mathcal{E}^{\bullet}|_{E}$, which equals $(\sigma^{*} \widetilde{\mathcal{H}}_{2}|_{E}) \otimes_{\mathcal{O}_{E}} (\mathcal{L}|_{E})$, satisfies the short exact sequence

$$0 \to \mathcal{F} \to \mathcal{O}_E(1)^{\oplus r} \to \mathcal{O}_E(p+1) \to 0.$$

Therefore, we have

(A.4)

$$(A.3) = p \sum_{i=0}^{r-1} \int_{E} (-\eta)^{i} c_{r-i-1}(\mathcal{F}) - \int_{E} c_{r-1}(\mathcal{F})$$

$$= p \int_{E} c_{r-1}(\mathcal{F}(-1)) - \int_{E} c_{r-1}(\mathcal{F})$$

$$= p \int_{E} (-p)^{r-1} \eta^{r-1} - \int_{E} \frac{1 - (-p)^{r}}{p+1} \eta^{r-1}$$

$$= \frac{(-p)^{r+1} - 1}{p+1} \int_{E} \eta^{r-1}$$

$$= \frac{(-p)^{r+1} - 1}{p+1} \cdot |\operatorname{DL}(\mathscr{V}_{\sharp})'(\kappa)|$$

$$= \frac{(-p)^{r+1} - 1}{p+1} (p+1)(p^{3}+1) \cdots (p^{2r-1}+1).$$

By [XZ, Proposition 9.3.10], we have

(A.5)
$$\int_{\mathrm{DL}(\mathscr{V}_{\sharp})} c_r(\mathscr{E}) = \mathsf{d}_{r,p}$$

where $d_{r,p}$ is the number introduced in Notation 1.3.2. Thus, (2) follows from (A.3), (A.4) and (A.5). The proposition is proved.

APPENDIX B. COMPUTATION IN HECKE ALGEBRAS

In this appendix, we compute several explicit formulae on the evaluation of certain Hecke elements. In Subsection B.1, we prove some combinatorial formulae on characters of the dual group (of a unitary group). In Subsection B.2, we introduce the two unitary Hecke algebras and prove a formula for an intertwining operator between the two Hecke algebras. In Subsections B.3 and B.4, we evaluate certain Hecke operators under a Satake parameter in the even and odd rank cases, respectively.

B.1. Characters of the dual group. Let $N \ge 1$ be an integer with $r := \lfloor \frac{N}{2} \rfloor$. We let GL_N be the group of automorphism of the \mathbb{Z} -module $\mathbb{Z}^{\oplus N}$, which is a group scheme over \mathbb{Z} . Let $T_N \subseteq GL_N$ be the subgroup of diagonal matrices. The group of homomorphisms from T_N to \mathbb{G}_m , denoted by \mathbb{X}_N^* , is a free abelian group generated by $\{\mu_1, \ldots, \mu_N\}$ where μ_i is the projection to the *i*-th factor. For $\mu \in \mathbb{X}_N^*$, we denote by $[\mu]$ the corresponding element in $\mathbb{Z}[\mathbb{X}_N^*]$. For $1 \le i \le r$, we put

$$\boldsymbol{\mu}_i \coloneqq [\mu_i - \mu_{N+1-i}] + [\mu_{N+1-i} - \mu_i] \in \mathbb{Z}[\mathbb{X}_N^*]$$

For $0 \leq \delta \leq r$, let $\mathfrak{s}_{\delta} \in \mathbb{Z}[\mathbb{X}_{N}^{*}]$ be the elementary symmetric polynomial in μ_{1}, \ldots, μ_{r} of degree δ . Finally, we denote by $\mathbb{Z}[\mathbb{X}_{N}^{*}]^{\text{sym}}$ the subring of $\mathbb{Z}[\mathbb{X}_{N}^{*}]$ generated by $\{\mathfrak{s}_{1}, \ldots, \mathfrak{s}_{r}\}$ over \mathbb{Z} .

Now we consider $\operatorname{GL}_N^{\operatorname{ext}} := \operatorname{GL}_N \rtimes \{1, \sigma\}$ in which the involution σ sends $A \in \operatorname{GL}_N$ to

$$\begin{pmatrix} & & & 1 \\ & & -1 & \\ & & \ddots & & \\ & & (-1)^{N-2} & & & \\ (-1)^{N-1} & & & & \end{pmatrix}^{t} A^{-1} \begin{pmatrix} & & & & 1 \\ & & & -1 & \\ & & & \ddots & & \\ & & (-1)^{N-2} & & & \\ (-1)^{N-1} & & & & & \end{pmatrix}^{-1}$$

For every algebraic representation ρ of $\operatorname{GL}_N^{\operatorname{ext}}$ (over \mathbb{Z}), we denote by $\chi(\rho)$ the restriction of the character of ρ to $\operatorname{T}_N \sigma$, regarded as an element in $\mathbb{Z}[\mathbb{X}_N^*]$. Let $\rho_{N,\operatorname{std}}$ be the standard representation of GL_N and $\rho_{N,\operatorname{std}}^{\vee}$ its dual. We let $\{\varepsilon_1, \ldots, \varepsilon_N\}$ be the standard basis of $\rho_{N,\operatorname{std}}$ and $\{\varepsilon_1^{\vee}, \ldots, \varepsilon_N^{\vee}\}$

the dual basis of $\rho_{N,\text{std}}^{\vee}$. For a subset $I \subseteq \{1, \ldots, N\}$, we put $\langle I \rangle \coloneqq \sum_{i \in I} i, I^{\vee} \coloneqq \{N+1-i \mid i \in I\}$, $\varepsilon_I \coloneqq \wedge_{i \in I} \varepsilon_i$ and $\varepsilon_I^{\vee} \coloneqq \wedge_{i \in I} \varepsilon_i^{\vee}$ (in the increasing order of the indices). For $0 \leq \delta \leq r$, put

$$\rho_{N;\delta} \coloneqq \left(\bigwedge^{\delta} \rho_{N,\mathrm{std}}\right) \otimes \left(\bigwedge^{\delta} \rho_{N,\mathrm{std}}^{\vee}\right),$$

which extends uniquely to a representation of $\operatorname{GL}_N^{\operatorname{ext}}$ such that σ sends $\varepsilon_I \otimes \varepsilon_{J^{\vee}}^{\vee}$ to $(-1)^{\langle I \rangle + \langle J \rangle} \varepsilon_J \otimes \varepsilon_{I^{\vee}}^{\vee}$.

Remark B.1.1. In the next subsection, we will study the unramified unitary group $U(V_N)$ over nonarchimedean local fields. Then $\operatorname{GL}_N^{\operatorname{ext}}(\mathbb{C})$ is simply the Langlands dual group of $U(V_N)$, and we have $\mathbb{Z}[\mathbb{X}_N^*]^{\operatorname{sym}} \simeq \mathbb{Z}[\mathbb{X}^*(\widehat{U(V_N)})^{\sigma}]^{W_N}$.

Lemma B.1.2. We have

$$\chi(\rho_{N;\delta}) = \begin{cases} \sum_{i=0}^{\delta} \binom{r-\delta+i}{\lfloor\frac{i}{2}\rfloor} \cdot \mathfrak{s}_{\delta-i}, & \text{if } N \text{ is odd;} \\ \sum_{j=0}^{\lfloor\frac{\delta}{2}\rfloor} \binom{r-\delta+2j}{j} \cdot \mathfrak{s}_{\delta-2j}, & \text{if } N \text{ is even.} \end{cases}$$

In particular, $\chi(\rho_{N;\delta})$ belongs to $\mathbb{Z}[\mathbb{X}_N^*]^{\text{sym}}$.

Proof. Note that for every $t \in T_N$, $t\sigma$ sends $\varepsilon_I \otimes \varepsilon_{J^{\vee}}^{\vee}$ to

$$(-1)^{\langle I \rangle + \langle J \rangle} \prod_{i \in I^{\vee}} \mu_i(t)^{-1} \prod_{j \in J} \mu_j(t) \cdot \varepsilon_J \otimes \varepsilon_{I^{\vee}}^{\vee}.$$

Thus, we have

$$\chi(\rho_{N,\delta})(t\sigma) = \sum_{I \subseteq \{1,\dots,r\}, |I|=\delta} \prod_{i \in I^{\vee}} \mu_i(t)^{-1} \prod_{i \in I} \mu_i(t)$$
$$= \sum_{I \subseteq \{1,\dots,r\}, |I|=\delta} \prod_{i \in I} \mu_i(t) \mu_{N+1-i}(t)^{-1}.$$

To evaluate the above sum, we consider $i := |I \cap I^{\vee}|$, which has to be even when N is even. It is easy to see that for fixed $0 \le i \le \delta$ (that is even if N is even), the contribution from those subsets I to the above sum is

$$\binom{r-\delta+i}{\lfloor \frac{i}{2} \rfloor} \cdot \mathfrak{s}_{\delta-i}(t).$$

Thus, the lemma follows.

Lemma B.1.3. Suppose that N = 2r is even.

(1) We have

$$\prod_{i=1}^{r} \left(\lambda + \lambda^{-1} + \boldsymbol{\mu}_{i} \right) = \chi(\rho_{N;r}) + \sum_{\delta=1}^{r} \chi(\rho_{N;r-\delta})(\lambda^{\delta} + \lambda^{-\delta})$$
in $\mathbb{Z}[\mathbb{X}_{N}^{*}]^{\text{sym}} \otimes \mathbb{Z}[\lambda, \lambda^{-1}].$
(2) We have

$$\sum_{j=1}^{r} \prod_{\substack{i=1\\i\neq j}}^{r} \left(\lambda + \lambda^{-1} + \boldsymbol{\mu}_{i} \right) = \sum_{\delta=1}^{r} \delta \cdot \chi(\rho_{N;r-\delta}) \frac{\lambda^{\delta} - \lambda^{-\delta}}{\lambda - \lambda^{-1}}$$
in $\mathbb{Z}[\mathbb{X}_{N}^{*}]^{\text{sym}} \otimes \mathbb{Z}[\lambda, \lambda^{-1}].$

Proof. Part (1) is follows from Lemma B.1.2 by comparing coefficients of powers of λ . Part (2) follows from (1) by taking derivative with respect to λ and dividing both sides of the resulted equality by $1 - \lambda^{-2}$.

Lemma B.1.4. Suppose that N = 2r + 1 is odd. We have

$$\prod_{i=1}^{r} \left(\lambda + \lambda^{-1} + \boldsymbol{\mu}_i \right) = \sum_{\delta=0}^{r} \chi(\rho_{N;r-\delta}) \frac{\lambda^{\delta+1} + \lambda^{-\delta}}{\lambda+1}$$

in $\mathbb{Z}[\mathbb{X}_N^*]^{\text{sym}} \otimes \mathbb{Z}[\lambda, \lambda^{-1}].$

Proof. By Lemma B.1.2, the right-hand side of the desired identity equals

$$\sum_{\delta=0}^{r} \frac{\lambda^{\delta+1} + \lambda^{-\delta}}{\lambda+1} \sum_{i=0}^{r-\delta} \binom{\delta+i}{\lfloor \frac{i}{2} \rfloor} \cdot \mathfrak{s}_{r-\delta-i},$$

which coincides with

$$\sum_{i=0}^{r} \left(\sum_{\delta=0}^{r-i} \frac{\lambda^{\delta+1} + \lambda^{-\delta}}{\lambda+1} \binom{r-i}{\lfloor \frac{r-i-\delta}{2} \rfloor} \right) \mathfrak{s}_{i}$$

by substituting i by $r - \delta - 1$. Thus, it remains to show that

$$\sum_{\delta=0}^{k} \frac{\lambda^{\delta+1} + \lambda^{-\delta}}{\lambda+1} \binom{k}{\lfloor \frac{k-\delta}{2} \rfloor} = (\lambda + \lambda^{-1})^{k}$$

for $0 \le k \le r$. However, we have

$$\begin{split} &\sum_{\lambda=0}^{k} \frac{\lambda^{\delta+1} + \lambda^{-\delta}}{\lambda + 1} \binom{k}{\lfloor \frac{k-\delta}{2} \rfloor} \\ &= \binom{k}{0} \left(\frac{\lambda^{k+1} + \lambda^{-k}}{\lambda + 1} + \frac{\lambda^{k} + \lambda^{-(k-1)}}{\lambda + 1} \right) + \binom{k}{1} \left(\frac{\lambda^{k-1} + \lambda^{-(k-2)}}{\lambda + 1} + \frac{\lambda^{k-2} + \lambda^{-(k-3)}}{\lambda + 1} \right) + \cdots \\ &= \binom{k}{0} (\lambda^{k} + \lambda^{-k}) + \binom{k}{1} (\lambda^{k-1} + \lambda^{-(k-1)}) + \cdots \\ &= (\lambda + \lambda^{-1})^{k}. \end{split}$$

The lemma follows.

B.2. Two Hecke algebras. From now to the end of this section, we fix an unramified quadratic extension F/F^+ of nonarchimedean local fields. Let q be the residue cardinality of F^+ and \mathfrak{p} the maximal ideal of O_F .

Let $N \ge 1$ be an integer with $r \coloneqq \lfloor \frac{N}{2} \rfloor$. Consider a hermitian space V_N over F (with respect to F/F^+) of rank N together with a basis $\{e_{-r}, \ldots, e_r\}$ (with e_0 omitted if N is even) such that $(e_{-i}, e_j)_{V_N} = \delta_{ij}$ for $0 \le i, j \le r$. Via this basis, we identify $U(V_N)$ as a closed subgroup of $\operatorname{Res}_{F/F^+} \operatorname{GL}_N$. We study two lattices

(B.1)
$$\Lambda_N^{\circ} = O_F e_{-r} \oplus \cdots \oplus O_F e_r, \quad \Lambda_N^{\bullet} = \mathfrak{p}^{-1} e_{-r} \oplus \cdots \oplus \mathfrak{p}^{-1} e_{-1} \oplus O_F e_0 \oplus \cdots \oplus O_F e_r$$

of V_N . We have $(\Lambda_N^{\circ})^{\vee} = \Lambda_N^{\circ}$, $\mathfrak{p}\Lambda_N^{\bullet} \subseteq (\Lambda_N^{\bullet})^{\vee}$, and that the O_F -module $(\Lambda_N^{\bullet})^{\vee}/\mathfrak{p}\Lambda_N^{\bullet}$ has length N - 2r. Let K_N° and K_N^{\bullet} be the stabilizers of Λ_N° and Λ_N^{\bullet} , respectively, which are subgroups of $U(V_N)(F^+)$. It is clear that K_N° is hyperspecial maximal; K_N^{\bullet} is special maximal and is hyperspecial if and only if N is even. We have two *commutative* Hecke algebras

$$\mathbb{T}_N^{\circ} \coloneqq \mathbb{Z}[\mathrm{K}_N^{\circ} \setminus \mathrm{U}(\mathrm{V}_N)(F^+)/\mathrm{K}_N^{\circ}], \quad \mathbb{T}_N^{\bullet} \coloneqq \mathbb{Z}[\mathrm{K}_N^{\bullet} \setminus \mathrm{U}(\mathrm{V}_N)(F^+)/\mathrm{K}_N^{\bullet}]$$

Recall that by our convention in Subsection 1.3, the units in \mathbb{T}_N° and \mathbb{T}_N^{\bullet} are $\mathbb{1}_{K_N^{\circ}}$ and $\mathbb{1}_{K_N^{\bullet}}$, respectively. Let $A_N(F^+)$ (resp. $A_N(O_{F^+})$) be the subgroup of $U(V_N)(F^+)$ that acts on e_i by a scalar in F^+ (resp. O_{F^+}) for every $-r \leq i \leq r$.

Notation B.2.1. For each element $\mathbf{t} = (t_1, \ldots, t_N) \in \mathbb{Z}^N$ satisfying $t_i + t_{N+1-i} = 0$ and $a \in F^{\times}$, we have an element $a^t \in A_N(F^+)$ such that $a^t \cdot e_{-i} = a^{t_{r+1-i}} e_{-i}$ for $1 \leq i \leq r$. For $0 \leq \delta \leq r$, put $\mathbf{t}_{\delta} := (1^{\delta}, 0^{N-2\delta}, (-1)^{\delta})$. We let $\mathsf{T}_{N;t}^{\circ}$ (resp. $\mathsf{T}_{N;t}^{\bullet}$) be the element in \mathbb{T}_N° (resp. \mathbb{T}_N^{\bullet}) corresponding to the double coset $\mathsf{K}_N^{\circ} \varpi^t \mathsf{K}_N^{\circ}$ (resp. $\mathsf{K}_N^{\bullet} \varpi^t \mathsf{K}_N^{\bullet}$) for some uniformizer ϖ of F; and simply write $\mathsf{T}_{N;\delta}^{\circ}$ (resp. $\mathsf{T}_{N;\delta}^{\bullet}$) for $\mathsf{T}_{N;t_{\delta}}^{\circ}$ (resp. $\mathsf{T}_{N;t_{\delta}}^{\bullet}$).

Remark B.2.2. The elements $T_{N;t}^{\circ} \in \mathbb{T}_{N}^{\circ}$ and $T_{N;t}^{\bullet} \in \mathbb{T}_{N}^{\circ}$ do not depend on the choice of the basis $\{e_{-r}, \ldots, e_{r}\}$ satisfying (B.1).

Now we recall Satake isomorphisms. Denote by W_N the Weyl group of $A_N(F^+)$ in $U(V_N)(F^+)$, which preserves $A_N(O_{F^+})$; and we have the two Satake isomorphisms²⁴

$$\operatorname{Sat}_{N}^{\circ} \colon \mathbb{T}_{N}^{\circ} \xrightarrow{\sim} \mathbb{Z}[\operatorname{A}_{N}(F^{+})/\operatorname{A}_{N}(O_{F^{+}})]^{\operatorname{W}_{N}},$$
$$\operatorname{Sat}_{N}^{\bullet} \colon \mathbb{T}_{N}^{\bullet} \xrightarrow{\sim} \mathbb{Z}[\operatorname{A}_{N}(F^{+})/\operatorname{A}_{N}(O_{F^{+}})]^{\operatorname{W}_{N}}.$$

In addition, we have an isomorphism

$$\mathbb{Z}[\mathcal{A}_N(F^+)/\mathcal{A}_N(O_{F^+})]^{\mathcal{W}_N} \simeq \mathbb{Z}[\mathbb{X}_N^*]^{\text{sym}}$$

of rings under which \mathfrak{s}_{δ} corresponds to the characteristic function of the W_N-orbit of $p^{t_{\delta}}$ for every $0 \leq \delta \leq r$. In what follows, we will regard $\mathbb{Z}[\mathbb{X}_{N}^{*}]^{\text{sym}}$ as the target of both Satake isomorphisms $\operatorname{Sat}_{N}^{\bullet}$ and $\operatorname{Sat}_{N}^{\bullet}$.

Notation B.2.3. Let $\mathbb{Z}[\mathbb{X}_N^*]'$ be the subring of $\mathbb{Z}[\mathbb{X}_N^*]$ generated by $\{\boldsymbol{\mu}_1, \ldots, \boldsymbol{\mu}_r\}$ over \mathbb{Z} , sending $\boldsymbol{\mu}_i$ to $\alpha_i + \alpha_i^{-1}$ for $1 \leq i \leq r$. For every ring L and every tuple $\boldsymbol{\alpha} = (\alpha_1, \ldots, \alpha_N) \in L^N$ satisfying $\alpha_i \alpha_{N+1-i} = 1$, we have a homomorphism $\phi'_{\boldsymbol{\alpha}} : \mathbb{Z}[\mathbb{X}_N^*]' \to L$, and denote

$$\begin{split} \phi^{\circ}_{\alpha} \colon \mathbb{T}_{N}^{\circ} \xrightarrow{\operatorname{Sat}_{N}^{\circ}} \mathbb{Z}[\mathbb{X}_{N}^{*}]^{\operatorname{sym}} &\subseteq \mathbb{Z}[\mathbb{X}_{N}^{*}]' \xrightarrow{\phi'_{\alpha}} L, \\ \phi^{\bullet}_{\alpha} \colon \mathbb{T}_{N}^{\bullet} \xrightarrow{\operatorname{Sat}_{N}^{\bullet}} \mathbb{Z}[\mathbb{X}_{N}^{*}]^{\operatorname{sym}} &\subseteq \mathbb{Z}[\mathbb{X}_{N}^{*}]' \xrightarrow{\phi'_{\alpha}} L, \end{split}$$

the composite homomorphisms.

Definition B.2.4. We denote

- O Lat^{\circ}_N the set of all self-dual lattices in V_N;
- O Lat_N the set of all lattices L in V_N satisfying $\mathfrak{p}L \subseteq L^{\vee}$ and that $L^{\vee}/\mathfrak{p}L$ has length $N-2\lfloor \frac{N}{2} \rfloor$;
- O $T_N^{\bullet\circ} \in \mathbb{Z}[K_N^{\bullet} \setminus U(V_N)(F^+)/K_N^{\circ}]$ the characteristic function of $K_N^{\bullet}K_N^{\circ}$; and
- $O T_N^{\circ \bullet} \in \mathbb{Z}[K_N^{\circ} \setminus U(V_N)(F^+)/K_N^{\bullet}]$ the characteristic function of $K_N^{\circ}K_N^{\bullet}$.

Moreover, we define the *intertwining Hecke operator*

$$\mathbf{I}_N^{\circ} \coloneqq \mathbf{T}_N^{\circ \bullet} \circ \mathbf{T}_N^{\bullet \circ} \in \mathbb{T}_N^{\circ}$$

where the composition is taken as composition of cosets.

Note that we have canonical injective homomorphisms

$$\mathbb{T}_N^{\circ} \to \operatorname{End}_{\mathbb{Z}}(\mathbb{Z}[\operatorname{Lat}_N^{\circ}]), \quad \mathbb{T}_N^{\bullet} \to \operatorname{End}_{\mathbb{Z}}(\mathbb{Z}[\operatorname{Lat}_N^{\bullet}])$$

sending $T_{N;t}^{?}$ to the endomorphism that takes $f \in \mathbb{Z}[\operatorname{Lat}_{N}^{?}]$ to the function $T_{N;t}^{?}f$ satisfying $(T_{N;t}^{?}f)(L) = \sum f(L')$ where the sum is taken over all $L' \in \operatorname{Lat}_{N}^{?}$ such that L' and L have relative position ϖ^{t} for $? = \circ, \bullet$.

²⁴Strictly speaking, we need to choose a square root of q^2 , which we take to be q.

Lemma B.2.5. We have the identity

$$\mathbf{I}_{N}^{\circ} = \begin{cases} \mathbf{T}_{N;r}^{\circ} + (q+1)\mathbf{T}_{N;r-1}^{\circ} + (q+1)(q^{3}+1)\mathbf{T}_{N;r-2}^{\circ} + \dots + \prod_{i=1}^{r} (q^{2i-1}+1)\mathbf{T}_{N;0}^{\circ}, & \text{if } N = 2r; \\ \mathbf{T}_{N;r}^{\circ} + (q^{3}+1)\mathbf{T}_{N;r-1}^{\circ} + (q^{3}+1)(q^{5}+1)\mathbf{T}_{N;r-2}^{\circ} + \dots + \prod_{i=1}^{r} (q^{2i+1}+1)\mathbf{T}_{N;0}^{\circ}, & \text{if } N = 2r+1 \\ \text{in } \mathbb{T}_{N}^{\circ}. \end{cases}$$

Proof. To compute I_N° , it suffices to compute its induced endomorphism on $\mathbb{Z}[\operatorname{Lat}_N^{\circ}]$. Now we take an element $f \in \mathbb{Z}[\operatorname{Lat}_N^{\circ}]$. Then

$$(\mathbf{T}_N^{\circ\bullet}(\mathbf{T}_N^{\bullet\circ}f))(\mathbf{L}_1^{\circ}) = \sum_{\substack{\mathbf{L}^{\bullet} \in \operatorname{Lat}_N^{\bullet} \\ \mathbf{L}_1^{\circ} \subseteq \mathbf{L}^{\bullet} \subseteq \mathfrak{p}^{-1}\mathbf{L}_1^{\circ}}} (\mathbf{T}_N^{\bullet\circ}f)(\mathbf{L}^{\bullet}) = \sum_{\substack{\mathbf{L}^{\bullet} \in \operatorname{Lat}_N^{\bullet} \\ \mathbf{L}_1^{\circ} \subseteq \mathbf{L}^{\bullet} \subseteq \mathfrak{p}^{-1}\mathbf{L}_1^{\circ}}} \sum_{\substack{\mathbf{L}_2^{\circ} \in \operatorname{Lat}_N^{\circ} \\ \mathbf{L}_1^{\circ} \subseteq \mathbf{L}^{\bullet} \subseteq \mathfrak{p}^{-1}\mathbf{L}_1^{\circ}}} f(\mathbf{L}_2^{\circ})$$

for every $L_1^{\circ} \in Lat_N^{\circ}$. Note that for pairs $(L_1^{\circ}, L_2^{\circ}) \in (Lat_N^{\circ})^2$ appearing in the formula above, we have $\mathfrak{p}L_2^{\circ} \subseteq L_1^{\circ} \subset \mathfrak{p}^{-1}L_2^{\circ}$ and $[L_1^{\circ}: L_2^{\circ}] \coloneqq [L_1^{\circ}: L_1^{\circ} \cap L_2^{\circ}] + [L_2^{\circ}: L_1^{\circ} \cap L_2^{\circ}] \in \{0, 2, \ldots, 2r\}$. Now for a pair $(L_1^{\circ}, L_2^{\circ}) \in (Lat_N^{\circ})^2$ satisfying $\mathfrak{p}L_2^{\circ} \subseteq L_1^{\circ} \subset \mathfrak{p}^{-1}L_2^{\circ}$, we consider the set

 $\operatorname{Lat}_{N}^{\bullet}(\operatorname{L}_{1}^{\circ},\operatorname{L}_{2}^{\circ}) \coloneqq \{\operatorname{L}^{\bullet} \in \operatorname{Lat}_{N}^{\bullet} | \operatorname{L}_{1}^{\circ} \subseteq \operatorname{L}^{\bullet} \subseteq \mathfrak{p}^{-1}\operatorname{L}_{1}^{\circ}, \operatorname{L}_{2}^{\circ} \subseteq \operatorname{L}^{\bullet} \subseteq \mathfrak{p}^{-1}\operatorname{L}_{2}^{\circ}\}.$

It is easy to see that the cardinality of $\operatorname{Lat}_{N}^{\bullet}(\operatorname{L}_{1}^{\circ}, \operatorname{L}_{2}^{\circ})$ depends only on $[\operatorname{L}_{1}^{\circ}: \operatorname{L}_{2}^{\circ}]$. For $0 \leq \delta \leq r$, we denote by $c_{N,\delta}$ the cardinality of $\operatorname{Lat}_{N}^{\bullet}(\operatorname{L}_{1}^{\circ}, \operatorname{L}_{2}^{\circ})$ with $[\operatorname{L}_{1}^{\circ}: \operatorname{L}_{2}^{\circ}] = 2\delta$. Then the lemma is equivalent to showing that $c_{N,r} = 1$ and

$$c_{N,\delta} = \begin{cases} \prod_{i=1}^{r-\delta} (q^{2i-1}+1), & 0 \le \delta < r, & \text{when } N = 2r; \\ \prod_{i=1}^{r-\delta} (q^{2i+1}+1), & 0 \le \delta < r, & \text{when } N = 2r+1. \end{cases}$$

Without lost of generality, we may assume $L_1^\circ = \Lambda_N^\circ$ and

$$\mathcal{L}_{2}^{\circ} = \mathfrak{p}^{-1}e_{-r} \oplus \cdots \oplus \mathfrak{p}^{-1}e_{-r+\delta-1} \oplus O_{F}e_{-r+\delta} \oplus \cdots \oplus O_{F}e_{r-\delta} \oplus \mathfrak{p}O_{F}e_{r-\delta+1} \oplus \cdots \oplus \mathfrak{p}O_{F}e_{r}.$$

When $\delta = r$, Λ_N^{\bullet} is the only element in $\operatorname{Lat}_N^{\bullet}(L_1^{\circ}, L_2^{\circ})$. Thus, we have $c_{N,r} = 1$. For $0 \leq \delta < r$, we have $c_{N,\delta} = c_{N-2\delta,0}$. Thus, it suffices to show

$$c_{N,0} = \begin{cases} \prod_{i=1}^{r} (q^{2i-1}+1) = (q+1) \cdots (q^{2r-1}+1), & \text{when } N = 2r; \\ \prod_{i=1}^{r} (q^{2i+1}+1) = (q^3+1) \cdots (q^{2r+1}+1), & \text{when } N = 2r+1. \end{cases}$$

However, $c_{N,0}$ is nothing but the number of maximal isotropic subspaces of the hermitian space $\Lambda_N^\circ \otimes_{O_F} O_F/\mathfrak{p}$ over O_F/\mathfrak{p} of dimension N, which is given by the above formula. Thus, the lemma is proved.

The following three lemmas will be used in later computation.

Lemma B.2.6. We have the identity

$$q^{\delta(N-\delta)}\chi(\rho_{N,\delta}) = \sum_{i=0}^{\delta} \begin{bmatrix} N-2i\\ \delta-i \end{bmatrix}_{-q} \operatorname{Sat}_{N}^{\circ}(\mathsf{T}_{N;i}^{\circ})$$

in $\mathbb{Z}[\mathbb{X}_N^*]^{\text{sym}}$ for $0 \leq \delta \leq r$.

Proof. This is [XZ, Lemma 9.2.4].

Lemma B.2.7. For every integer $k \ge 1$, we have

$$\sum_{\delta=-k}^{k} q^{\delta^2} {2k \brack k-\delta}_{-q} = (q+1)(q^3+1)\cdots(q^{2k-1}+1).$$

Proof. For every integer $k \ge 1$, we have the Gauss polynomial identity

$$\sum_{\delta=0}^{2k} (-1)^{\delta} \begin{bmatrix} 2k\\ \delta \end{bmatrix}_{\lambda} = (1-\lambda)(1-\lambda^3)\cdots(1-\lambda^{2k-1})$$

in $\mathbb{Z}[\lambda]$.²⁵ Now we specialize the identity to $\lambda = -q^{-1}$. Then we get

$$\sum_{\delta=0}^{2k} (-1)^{\delta} (-q)^{-(2k-1)-(2k-3)-\dots-(2k-2\delta+1)} \begin{bmatrix} 2k\\ \delta \end{bmatrix}_{-q} = q^{-k^2} (q+1)(q^3+1)\cdots(q^{2k-1}+1).$$

The lemma then follows by changing δ to $k - \delta$.

Lemma B.2.8. For every integer $k \ge 1$, we have

$$\sum_{\delta=-k-1}^{k} (-1)^{\delta} \delta q^{\delta^2 + \delta} \begin{bmatrix} 2k+1\\k-\delta \end{bmatrix}_{-q} - \sum_{\delta=-k}^{k} (-1)^{\delta} \delta q^{\delta^2 + \delta} \begin{bmatrix} 2k\\k-\delta \end{bmatrix}_{-q} = (-q)^k (q+1)(q^3+1) \cdots (q^{2k-1}+1).$$

Proof. In fact, we have

$$\sum_{\delta=-k-1}^{k} (-1)^{\delta} \delta q^{\delta^{2}+\delta} \begin{bmatrix} 2k+1\\k-\delta \end{bmatrix}_{-q} - \sum_{\delta=-k}^{k} (-1)^{\delta} \delta q^{\delta^{2}+\delta} \begin{bmatrix} 2k\\k-\delta \end{bmatrix}_{-q}$$
$$= \sum_{\delta=-k-1}^{k} (-1)^{\delta} \delta q^{\delta^{2}+\delta} (-q)^{k+\delta+1} \begin{bmatrix} 2k\\k-\delta - 1 \end{bmatrix}_{-q}$$
$$= (-1)^{k+1} q^{k} \sum_{\delta=-k}^{k} (\delta-1) q^{\delta^{2}} \begin{bmatrix} 2k\\k-\delta \end{bmatrix}_{-q}$$

which, by Lemma B.2.7, equals

$$(-q)^{k}(q+1)(q^{3}+1)\cdots(q^{2k-1}+1) + (-1)^{k+1}q^{k}\sum_{\delta=-k}^{k}\delta q^{\delta^{2}} \begin{bmatrix} 2k\\k-\delta \end{bmatrix}_{-q}.$$

The lemma follows since

$$\sum_{\delta=-k}^{k} \delta q^{\delta^2} \begin{bmatrix} 2k\\ k-\delta \end{bmatrix}_{-q} = 0.$$

B.3. Enumeration of Hecke operators in the even rank case. In this subsection, we assume that N = 2r is even.

Lemma B.3.1. We have the identity

$$q^{r^{2}} \prod_{i=1}^{r} (\boldsymbol{\mu}_{i} + 2) = \operatorname{Sat}_{N}^{\circ}(\mathsf{T}_{N;r}^{\circ}) + \sum_{\delta=1}^{r} (q+1)(q^{3}+1) \cdots (q^{2\delta-1}+1) \cdot \operatorname{Sat}_{N}^{\circ}(\mathsf{T}_{N;r-\delta}^{\circ})$$

in $\mathbb{Z}[\mathbb{X}_N^*]^{\text{sym}}$.

²⁵A proof can be found at http://mathworld.wolfram.com/GausssPolynomialIdentity.html.

Proof. By Lemma B.1.3(1) and Lemma B.2.6, we have

$$q^{r^{2}} \prod_{i=1}^{r} (\boldsymbol{\mu}_{i} + 2) = q^{r^{2}} \chi(\rho_{N;r}) + q^{r^{2}} \sum_{\delta=1}^{r} 2\chi(\rho_{N;r-\delta})$$

$$= \sum_{i=0}^{r} \begin{bmatrix} 2r - 2i \\ r - i \end{bmatrix}_{-q} \operatorname{Sat}_{N}^{\circ}(\mathsf{T}_{N;i}^{\circ}) + \sum_{\delta=1}^{r} 2q^{\delta^{2}} \sum_{i=0}^{r-\delta} \begin{bmatrix} 2r - 2i \\ r - \delta - i \end{bmatrix}_{-q} \operatorname{Sat}_{N}^{\circ}(\mathsf{T}_{N;i}^{\circ})$$

$$= \sum_{i=0}^{r} \left(\sum_{\delta=-(r-i)}^{r-i} q^{\delta^{2}} \begin{bmatrix} 2r - 2i \\ r - \delta - i \end{bmatrix}_{-q} \right) \operatorname{Sat}_{N}^{\circ}(\mathsf{T}_{N;i}^{\circ}),$$

which equals

$$\operatorname{Sat}_{N}^{\circ}(\mathsf{T}_{N;r}^{\circ}) + \sum_{\delta=1}^{r} (q+1)(q^{3}+1)\cdots(q^{2\delta-1}+1)\cdot\operatorname{Sat}_{N}^{\circ}(\mathsf{T}_{N;r-\delta}^{\circ})$$

by Lemma B.2.7. The lemma is proved.

Lemma B.3.2. We have the identity

$$q^{r^{2}} \prod_{i=1}^{r} \left(\boldsymbol{\mu}_{i} - q - q^{-1} \right) = \operatorname{Sat}_{N}^{\circ}(\operatorname{T}_{N;r}^{\circ}) + \sum_{\delta=1}^{r} (-q)^{\delta} (q+1)(q^{3}+1) \cdots (q^{2\delta-1}+1) \cdot \operatorname{Sat}_{N}^{\circ}(\operatorname{T}_{N;r-\delta}^{\circ})$$

in $\mathbb{Z}[\mathbb{X}_N^*]^{\text{sym}}$.

Proof. By Lemma B.1.3(1) and Lemma B.2.6, we have

$$\begin{split} q^{r^{2}} \prod_{i=1}^{r} \left(\boldsymbol{\mu}_{i} - q - q^{-1} \right) \\ &= q^{r^{2}} \chi(\rho_{N;r}) + q^{r^{2}} \sum_{\delta=1}^{r} \left((-q)^{\delta} + (-q)^{-\delta} \right) \chi(\rho_{N;r-\delta}) \\ &= \sum_{i=0}^{r} \begin{bmatrix} 2r - 2i \\ r - i \end{bmatrix}_{-q} \operatorname{Sat}_{N}^{\circ}(\mathsf{T}_{N;i}^{\circ}) + \sum_{\delta=1}^{r} \sum_{i=0}^{r-\delta} q^{\delta^{2}} \left((-q)^{\delta} + (-q)^{-\delta} \right) \begin{bmatrix} 2r - 2i \\ r - \delta - i \end{bmatrix}_{-q} \operatorname{Sat}_{N}^{\circ}(\mathsf{T}_{N;i}^{\circ}) \\ &= \sum_{i=0}^{r} \left(\begin{bmatrix} 2r - 2i \\ r - i \end{bmatrix}_{-q} + \sum_{\delta=1}^{r-i} (-1)^{\delta} \left(q^{\delta^{2}+\delta} + q^{\delta^{2}-\delta} \right) \begin{bmatrix} 2r - 2i \\ r - \delta - i \end{bmatrix}_{-q} \right) \operatorname{Sat}_{N}^{\circ}(\mathsf{T}_{N;i}^{\circ}) \\ &= \sum_{i=0}^{r} \left(\sum_{\delta=-(r-i)}^{r-i} (-1)^{\delta} q^{\delta^{2}+\delta} \begin{bmatrix} 2r - 2i \\ r - \delta - i \end{bmatrix}_{-q} \right) \operatorname{Sat}_{N}^{\circ}(\mathsf{T}_{N;i}^{\circ}). \end{split}$$

Thus, the lemma follows from Lemma B.3.3 below by comparing coefficients.

Lemma B.3.3. For every integer $k \ge 1$, we have

$$\sum_{\delta=-k}^{k} (-1)^{\delta} q^{\delta^{2}+\delta} {2k \brack k-\delta}_{-q} = (-q)^{k} (q+1)(q^{3}+1) \cdots (q^{2k-1}+1).$$

Proof. By Lemma B.2.7, the lemma is equivalent to the identity

$$(-q)^k \sum_{\delta=-k}^k q^{\delta^2} \begin{bmatrix} 2k\\ k-\delta \end{bmatrix}_{-q} = \sum_{\delta=-k}^k (-1)^{\delta} q^{\delta^2+\delta} \begin{bmatrix} 2k\\ k-\delta \end{bmatrix}_{-q}.$$

However, we have

$$\begin{split} &(-q)^k \sum_{\delta=-k}^k q^{\delta^2} {2k \brack k-\delta}_{-q} - \sum_{\delta=-k}^k (-1)^{\delta} q^{\delta^2+\delta} {2k \brack k-\delta}_{-q} \\ &= \sum_{\delta=-k}^k (-1)^{\delta} q^{\delta^2+\delta} \left((-q)^{k-\delta} - 1 \right) {2k \brack k-\delta}_{-q} \\ &= \sum_{\delta=-k}^k (-1)^{\delta} q^{\delta^2+\delta} \left((-q)^{2k} - 1 \right) {2k-1 \brack k-\delta}_{-q} \\ &= \left((-q)^{2k} - 1 \right) \sum_{\delta=-k}^k (-1)^{\delta} q^{\delta^2+\delta} {2k-1 \brack k-\delta}_{-q}. \end{split}$$

Note that in the last summation, the term of δ and the term of $1 - \delta$ cancel with each other for $-k < \delta \leq k$; and the term of -k is zero. Thus, the above summation is zero; and the lemma follows.

Lemma B.3.4. We have the identity

$$\left(q^{r^{2}+1} - q^{r^{2}-1}\right) \sum_{j=1}^{r} \prod_{\substack{i=1\\i \neq j}}^{r} \left(\boldsymbol{\mu}_{i} - q - q^{-1}\right)$$

= $\sum_{\delta=1}^{r} \left((-q)^{\delta}(q+1)(q^{3}+1)\cdots(q^{2\delta-1}+1) - \sum_{i=0}^{\delta}(-1)^{i}(2i+1)q^{i^{2}+i} \begin{bmatrix} 2\delta+1\\\delta-i \end{bmatrix}_{-q} \right) \operatorname{Sat}_{N}^{\circ}(\mathsf{T}_{N;r-\delta}^{\circ})$

in $\mathbb{Z}[\mathbb{X}_N^*]^{\text{sym}}$.

Proof. By Lemma B.1.3(2) and Lemma B.2.6, we have

$$\begin{split} & \left(q^{r^2+1}-q^{r^2-1}\right)\sum_{j=1}^r\prod_{i\neq j}\left(\boldsymbol{\mu}_i-q-q^{-1}\right) \\ &=q^{r^2}\sum_{\delta=1}^r(-1)^{\delta-1}\delta(q^{\delta}-q^{-\delta})\cdot\chi(\rho_{N;r-\delta}) \\ &=\sum_{\delta=1}^r(-1)^{\delta-1}q^{\delta^2}(\delta q^{\delta}-\delta q^{-\delta})\sum_{i=0}^{r-\delta}\left[\frac{2r-2i}{r-\delta-i}\right]_{-q}\operatorname{Sat}_N^\circ(\mathsf{T}_{N;i}^\circ) \\ &=\sum_{i=0}^{r-1}\left(\sum_{\delta=1}^{r-i}(-1)^{\delta-1}q^{\delta^2}(\delta q^{\delta}-\delta q^{-\delta})\left[\frac{2r-2i}{r-\delta-i}\right]_{-q}\right)\operatorname{Sat}_N^\circ(\mathsf{T}_{N;i}^\circ). \end{split}$$

Thus the lemma is equivalent to the identity

$$\sum_{\delta=0}^{k} (-1)^{\delta} (2\delta+1) q^{\delta^{2}+\delta} {2k+1 \brack k-\delta}_{-q} - \sum_{\delta=1}^{k} (-1)^{\delta} q^{\delta^{2}} (\delta q^{\delta} - \delta q^{-\delta}) {2k \brack k-\delta}_{-q} = (-q)^{k} (q+1) (q^{3}+1) \cdots (q^{2k-1}+1)$$

for every integer $k \geq 1$. In fact, we have

$$\sum_{\delta=0}^{k} (-1)^{\delta} (2\delta+1) q^{\delta^{2}+\delta} \begin{bmatrix} 2k+1\\k-\delta \end{bmatrix}_{-q} - \sum_{\delta=1}^{k} (-1)^{\delta} q^{\delta^{2}} (\delta q^{\delta} - \delta q^{-\delta}) \begin{bmatrix} 2k\\k-\delta \end{bmatrix}_{-q}$$
$$= \sum_{\delta=-k-1}^{k} (-1)^{\delta} \delta q^{\delta^{2}+\delta} \begin{bmatrix} 2k+1\\k-\delta \end{bmatrix}_{-q} - \sum_{\delta=-k}^{k} (-1)^{\delta} q^{\delta^{2}} \delta q^{\delta} \begin{bmatrix} 2k\\k-\delta \end{bmatrix}_{-q}$$
$$= (-q)^{k} (q+1) (q^{3}+1) \cdots (q^{2k-1}+1)$$

by Lemma B.2.8. The lemma follows.

Proposition B.3.5. Let *L* be a ring. Consider an *N*-tuple $\boldsymbol{\alpha} = (\alpha_1, \ldots, \alpha_N) \in L^N$ satisfying $\alpha_i \alpha_{N+1-i} = 1$, which determines a homomorphism $\phi_{\boldsymbol{\alpha}}^\circ \colon \mathbb{T}_N^\circ \to L$ as in Notation B.2.3.

(1) We have

$$\phi^{\circ}_{\alpha}(\mathbf{I}^{\circ}_{N}) = q^{r^{2}} \prod_{i=1}^{r} \left(\alpha_{i} + \frac{1}{\alpha_{i}} + 2 \right).$$

(2) We have

$$\phi_{\alpha}^{\circ}\left((q+1)\mathbf{R}_{N}^{\circ}-\mathbf{I}_{N}^{\circ}\right)=-q^{r^{2}}\prod_{i=1}^{r}\left(\alpha_{i}+\frac{1}{\alpha_{i}}-q-\frac{1}{q}\right)$$

where

$$\mathbf{R}_{N}^{\circ} \coloneqq \sum_{\delta=0}^{r-1} \frac{1 - (-q)^{r-\delta}}{q+1} (q+1)(q+3) \cdots (q^{2(r-\delta)-1}+1) \cdot \mathbf{T}_{N,\delta}^{\circ}$$

(3) We have

$$\phi_{\alpha}^{\circ}\left(\mathbf{R}_{N}^{\circ} + (q+1)\mathbf{T}_{N}^{\circ}\right) = -\left(q^{r^{2}+1} - q^{r^{2}-1}\right)\sum_{j=1}^{r}\prod_{\substack{i=1\\i\neq j}}^{r}\left(\alpha_{i} + \frac{1}{\alpha_{i}} - q - \frac{1}{q}\right)$$

where

$$\mathbf{T}_{N}^{\circ} \coloneqq \sum_{\delta=0}^{r-1} \mathbf{d}_{r-\delta,q}^{\bullet} \cdot \mathbf{T}_{N;\delta}^{\circ}$$

in which the numbers $\mathbf{d}_{r-\delta,q}^{\bullet}$ are introduced in Notation 1.3.2.

Proof. Part (1) follows from Lemma B.2.5 and Lemma B.3.1. Part (2) follows from Lemma B.2.5 and Lemma B.3.2. Part (3) follows from Lemma B.3.4. \Box

Lemma B.3.6. We have

 $\mathsf{T}_N^{\bullet\circ}\circ\mathsf{R}_N^\circ=\mathsf{R}_N^\bullet\circ\mathsf{T}_N^{\bullet\circ},\quad\mathsf{T}_N^{\bullet\circ}\circ\mathsf{T}_N^\circ=\mathsf{T}_N^\bullet\circ\mathsf{T}_N^{\bullet\circ}$

in $\mathbb{Z}[K_N^{\bullet} \setminus U(V_N)(F^+)/K_N^{\circ}]$, where \mathbb{R}_N° and \mathbb{T}_N° are defined in Proposition B.3.5 (2) and (3), respectively, and

$$\begin{cases} \mathsf{R}_{N}^{\bullet} \coloneqq \sum_{\delta=0}^{r-1} \frac{1 - (-q)^{r-\delta}}{q+1} (q+1)(q+3) \cdots (q^{2(r-\delta)-1}+1) \cdot \mathsf{T}_{N;\delta}^{\bullet}, \\ \mathsf{T}_{N}^{\bullet} \coloneqq \sum_{\delta=0}^{r-1} \mathrm{d}_{r-\delta,q}^{\bullet} \cdot \mathsf{T}_{N;\delta}^{\bullet}. \end{cases}$$

Proof. In fact, by the same lattice counting argument as for Lemma B.2.5, we have

$$\mathsf{T}_N^{ullet\circ}\circ\mathsf{T}_{N;\delta}^\circ=\mathsf{T}_{N;\delta}^ullet\circ\mathsf{T}_N^{ullet\circ}$$

for every $0 \leq \delta \leq r$. Then the lemma follows immediately.

157

B.4. Enumeration of Hecke operators in the odd rank case. In this subsection, we assume that N = 2r + 1 is odd.

Lemma B.4.1. We have the identity

$$q^{r^{2}+r} \prod_{i=1}^{r} \left(\boldsymbol{\mu}_{i} + q + q^{-1} \right) = \operatorname{Sat}_{N}^{\circ}(\mathsf{T}_{N;r}^{\circ}) + \sum_{\delta=1}^{r} (q^{3}+1)(q^{5}+1) \cdots (q^{2\delta+1}+1) \cdot \operatorname{Sat}_{N}^{\circ}(\mathsf{T}_{N;r-\delta}^{\circ})$$

in $\mathbb{Z}[\mathbb{X}_N^*]^{\text{sym}}$.

Proof. By Lemma B.1.4 and Lemma B.2.6, we have

$$q^{r^{2}+r} \prod_{i=1}^{r} \left(\boldsymbol{\mu}_{i} + q + q^{-1}\right) = q^{r^{2}+r} \sum_{\delta=0}^{r} \frac{q^{\delta+1} + q^{-\delta}}{q+1} \cdot \chi(\rho_{N;r-\delta})$$

$$= q^{r^{2}+r} \sum_{\delta=0}^{r} \frac{q^{\delta+1} + q^{-\delta}}{q+1} \cdot q^{-(r-\delta)(r+1+\delta)} \sum_{i=0}^{r-\delta} \begin{bmatrix} 2r+1-2i \\ r-\delta-i \end{bmatrix}_{-q} \operatorname{Sat}_{N}^{\circ}(\mathsf{T}_{N;i}^{\circ})$$

$$= \frac{1}{q+1} \sum_{i=0}^{r} \left(\sum_{\delta=0}^{r-i} (q^{2\delta+1} + 1)q^{\delta^{2}} \begin{bmatrix} 2(r-i) + 1 \\ r-i-\delta \end{bmatrix}_{-q} \right) \operatorname{Sat}_{N}^{\circ}(\mathsf{T}_{N;i}^{\circ})$$

$$= \frac{1}{q+1} \sum_{i=0}^{r} \left(\sum_{\delta=-(r-i)-1}^{r-i} q^{\delta^{2}} \begin{bmatrix} 2(r-i) + 1 \\ r-i-\delta \end{bmatrix}_{-q} \right) \operatorname{Sat}_{N}^{\circ}(\mathsf{T}_{N;i}^{\circ}).$$

Thus the lemma is equivalent to the identity

$$\sum_{\delta=-k-1}^{k} q^{\delta^2} {\binom{2k+1}{k-\delta}}_{-q} = (q+1)(q^3+1)\cdots(q^{2k+1}+1)$$

for every integer $k \ge 0$. By Lemma B.2.7, we have

$$\sum_{\delta=-k-1}^{k+1} q^{\delta^2} \begin{bmatrix} 2k+2\\k+1-\delta \end{bmatrix}_{-q} = (q+1)(q^3+1)\cdots(q^{2k+1}+1).$$

Thus, it remains to show

$$\sum_{\delta=-k-1}^{k+1} q^{\delta^2} \begin{bmatrix} 2k+2\\k+1-\delta \end{bmatrix}_{-q} = \sum_{\delta=-k-1}^{k} q^{\delta^2} \begin{bmatrix} 2k+1\\k-\delta \end{bmatrix}_{-q}.$$

However, the difference equals

$$\sum_{\delta=-k-1}^{k+1} q^{a^2} \left(\begin{bmatrix} 2k+2\\k+1-\delta \end{bmatrix}_{-q} - \begin{bmatrix} 2k+1\\k-\delta \end{bmatrix}_{-q} \right) = \sum_{\delta=-k-1}^{k+1} q^{a^2} (-q)^{k+1-a} \begin{bmatrix} 2k+1\\k+1-\delta \end{bmatrix}_{-q}$$
$$= (-q)^{k+1} \sum_{\delta=-k-1}^k (-1)^{\delta} q^{\delta^2+\delta} \begin{bmatrix} 2k+1\\k+1-\delta \end{bmatrix}_{-q}$$

which equals zero as the term of δ and the term of $-\delta - 1$ cancel each other. The lemma follows. **Lemma B.4.2.** We have the identity

$$q^{r^2+r}\prod_{i=1}^r \left(\boldsymbol{\mu}_i - 2\right) = \sum_{\delta=0}^r \mathsf{d}_{\delta,q} \cdot \operatorname{Sat}_N^\circ(\mathsf{T}_{N;r-\delta}^\circ)$$

in $\mathbb{Z}[\mathbb{X}_N^*]^{\text{sym}}$, in which the numbers $d_{\delta,q}$ are introduced in Notation 1.3.2.

Proof. By Lemma B.1.4 and Lemma B.2.6, we have

$$q^{r^{2}+r} \prod_{i=1}^{r} (\boldsymbol{\mu}_{i}-2) = q^{r^{2}+r} \sum_{\delta=0}^{r} (-1)^{\delta} (2\delta+1) \cdot \chi(\rho_{N;r-\delta})$$

$$= q^{r^{2}+r} \sum_{\delta=0}^{r} (-1)^{\delta} (2\delta+1) \cdot q^{-(r-\delta)(r+1+\delta)} \sum_{i=0}^{r-\delta} \begin{bmatrix} 2r+1-2i \\ r-\delta-i \end{bmatrix}_{-q} \operatorname{Sat}_{N}^{\circ}(\mathsf{T}_{N;i}^{\circ})$$

$$= \sum_{i=0}^{r} \left(\sum_{\delta=0}^{r-i} (-1)^{\delta} (2\delta+1) q^{\delta(\delta+1)} \sum_{i=0}^{r-\delta} \begin{bmatrix} 2(r-i)+1 \\ r-i-\delta \end{bmatrix}_{-q} \right) \operatorname{Sat}_{N}^{\circ}(\mathsf{T}_{N;i}^{\circ})$$

$$= \sum_{\delta=0}^{r} \mathsf{d}_{\delta,q} \cdot \operatorname{Sat}_{N}^{\circ}(\mathsf{T}_{N;r-\delta}^{\circ}).$$

The lemma is proved.

Proposition B.4.3. Let *L* be a ring. Consider an *N*-tuple $\boldsymbol{\alpha} = (\alpha_1, \ldots, \alpha_N) \in L^N$ satisfying $\alpha_i \alpha_{N+1-i} = 1$, which determines a homomorphism $\phi_{\boldsymbol{\alpha}}^\circ \colon \mathbb{T}_N^\circ \to L$ as in Notation B.2.3.

(1) We have

$$\phi_{\alpha}^{\circ}(\mathbb{I}_{N}^{\circ}) = q^{r^{2}+r} \prod_{i=1}^{r} \left(\alpha_{i} + \frac{1}{\alpha_{i}} + q + \frac{1}{q} \right).$$

(2) We have

$$\phi_{\alpha}^{\circ}(\mathbf{T}_{N}^{\circ}) = q^{r^{2}+r} \prod_{i=1}^{r} \left(\alpha_{i} + \frac{1}{\alpha_{i}} - 2\right)$$

where

$$\mathtt{T}_N^{\circ}\coloneqq \sum_{\delta=0}^r \mathtt{d}_{r-\delta,q}\cdot \mathtt{T}_{N;\delta}^{\circ}$$

in which the numbers $d_{r-\delta,q}$ are introduced in Notation 1.3.2.

Proof. Part (1) follows from Lemma B.2.5 and Lemma B.4.1. Part (2) follows from Lemma B.4.2. \Box

Lemma B.4.4. We have

$$\mathbf{T}_{N}^{\bullet\circ}\circ\mathbf{T}_{N}^{\circ}=\left((q+1)^{2}\mathbf{T}_{N}^{\bullet}+\mathbf{T}_{N}^{\bullet\circ}\circ\mathbf{T}_{N}^{\circ\bullet}\right)\circ\mathbf{T}_{N}^{\bullet\circ}$$

in $\mathbb{Z}[\mathrm{K}^{\bullet}_{N} \setminus \mathrm{U}(\mathrm{V}_{N})(F^{+})/\mathrm{K}^{\circ}_{N}]$, where T°_{N} is defined in Proposition B.4.3(2), and

$$\mathsf{T}^{ullet}_N\coloneqq \sum_{\delta=0}^{r-1}\mathsf{d}^{ullet}_{r-\delta,q}\cdot\mathsf{T}^{ullet}_{N;\delta}.$$

This lemma is a hard exercise in combinatorics. In fact, our proof below is by brutal force; it would be interesting to find a conceptual proof.

Proof. It suffices to show that for every element $f \in \mathbb{Z}[\operatorname{Lat}_N^\circ]$, we have

(B.2)
$$\left((q+1)^2 \mathsf{T}_N^{\bullet} + \mathsf{T}_N^{\bullet\circ} \circ \mathsf{T}_N^{\circ\bullet} \right) (\mathsf{T}_N^{\bullet\circ}(f)) = \mathsf{T}_N^{\bullet\circ}(\mathsf{T}_N^{\circ}(f))$$

in $\mathbb{Z}[\operatorname{Lat}_{N}^{\bullet}]$. Without lost of generality, we may just consider their values on Λ_{N}^{\bullet} . For every $L \in \operatorname{Lat}_{N}^{\circ}$ and $0 \leq \delta \leq r$, we denote

O $c^{\bullet}_{\delta}(L)$ the number of L[•] ∈ Lat[•]_N satisfying L ⊆ L[•] and (L[•] + Λ[•]_N)/Λ[•]_N ≃ (O_F/𝔅)^{⊕δ}; and O $c^{\bullet}_{\delta}(L)$ the number of L[°] ∈ Lat[°]_N satisfying L[°] ⊆ Λ[•]_N and L/(L ∩ L[°]) ≃ (O_F/𝔅)^{⊕δ}.

We then have

$$(\mathbf{T}_{N;\delta}^{\bullet\circ}(\mathbf{T}_{N}^{\circ\circ}(f)))(\Lambda_{N}^{\bullet}) = \sum_{\mathbf{L}\in\mathrm{Lat}_{N}^{\circ\circ}} c_{\delta}^{\bullet}(\mathbf{L}) \cdot f(\mathbf{L}),$$
$$(\mathbf{T}_{N}^{\bullet\circ}(\mathbf{T}_{N;\delta}^{\circ}(f)))(\Lambda_{N}^{\bullet}) = \sum_{\mathbf{L}\in\mathrm{Lat}_{N}^{\circ\circ}} c_{\delta}^{\circ}(\mathbf{L}) \cdot f(\mathbf{L}).$$

We claim the following identities

(B.3)
$$c^{\bullet}_{\delta}(\mathbf{L}) = \begin{cases} q^{(\delta-\gamma)(\delta-\gamma+2)} \begin{bmatrix} r-\gamma\\\delta-\gamma \end{bmatrix}_{q^2}, & \text{if } (\mathbf{L}+\Lambda^{\bullet}_N)/\Lambda^{\bullet}_N \simeq (O_F/\mathfrak{p})^{\oplus\gamma} \text{ for some } 0 \le \gamma \le \delta; \\ 0, & \text{otherwise;} \end{cases}$$

(B.4)
$$c^{\circ}_{\delta}(\mathbf{L}) = \begin{cases} q^{(\delta-\gamma)^2} \begin{bmatrix} r-\gamma\\\delta-\gamma \end{bmatrix}_{q^2}, & \text{if } (\mathbf{L}+\Lambda^{\bullet}_N)/\Lambda^{\bullet}_N \simeq (O_F/\mathfrak{p})^{\oplus\gamma} \text{ for some } 0 \le \gamma \le \delta; \\ 0, & \text{otherwise.} \end{cases}$$

For (B.3), we must have $(L + \Lambda_N^{\bullet})/\Lambda_N^{\bullet} \subseteq (L^{\bullet} + \Lambda_N^{\bullet})/\Lambda_N^{\bullet} \simeq (O_F/\mathfrak{p})^{\oplus \delta}$. Thus, the otherwise case is confirmed. Suppose $(L + \Lambda_N^{\bullet})/\Lambda_N^{\bullet} \simeq (O_F/\mathfrak{p})^{\oplus \gamma}$ for some $0 \leq \gamma \leq \delta$. Then $(\mathfrak{p}\Lambda_N^{\bullet} + L)/L$ is an isotropic subspace of $\mathfrak{p}^{-1}L/L$ of dimension γ . Moreover, $c_{\delta}^{\bullet}(L)$ is the same as the number of maximal isotropic subspaces of $((\mathfrak{p}\Lambda_N^{\bullet} + L)/L)^{\perp}/((\mathfrak{p}\Lambda_N^{\bullet} + L)/L)$ whose intersection with (the image of) $(\mathfrak{p}^{-1}L \cap \Lambda_N^{\bullet} + L)/L$, which itself is a maximal isotropic subspace, has dimension $r - \delta$. Thus, we obtain (B.3) by Lemma B.4.5 below since $((\mathfrak{p}\Lambda_N^{\bullet} + L)/L)^{\perp}/((\mathfrak{p}\Lambda_N^{\bullet} + L)/L)$ has dimension $2r + 1 - 2\gamma$.

For (B.4), we must have $(L+\Lambda_N^{\bullet})/\Lambda_N^{\bullet} \simeq L/(L\cap\Lambda_N^{\bullet})$ which is a quotient of $L/(L\cap L^{\circ}) \simeq (O_F/\mathfrak{p})^{\oplus \delta}$. Thus, the otherwise case is confirmed. Suppose $(L + \Lambda_N^{\bullet})/\Lambda_N^{\bullet} \simeq (O_F/\mathfrak{p})^{\oplus \gamma}$ for some $0 \leq \gamma \leq \delta$. Then $(L + \Lambda_N^{\bullet})/\Lambda_N^{\bullet}$ is an isotropic subspace of $\mathfrak{p}^{-1}\Lambda_N^{\bullet}/\Lambda_N^{\bullet}$ of dimension γ . Moreover, $c_{\delta}^{\circ}(L)$ is the same as the number of maximal isotropic subspaces of $((L + \Lambda_N^{\bullet})/\Lambda_N^{\bullet})^{\perp}/((L + \Lambda_N^{\bullet})/\Lambda_N^{\bullet})$ whose intersection with (the image of) $(\mathfrak{p}^{-1}\Lambda_N^{\bullet} \cap \mathfrak{p}^{-1}L + \Lambda_N^{\bullet})/\Lambda_N^{\bullet}$, which itself is a maximal isotropic subspace, has dimension $r - \delta$. Thus, we obtain (B.4) by Lemma B.4.5 since $((L + \Lambda_N^{\bullet})/\Lambda_N^{\bullet})^{\perp}/((L + \Lambda_N^{\bullet})/\Lambda_N^{\bullet})^{\perp}/((L + \Lambda_N^{\bullet})/\Lambda_N^{\bullet})$ has dimension $2r - 2\gamma$.

Now we come back to the values of (B.2) on Λ_N^{\bullet} . By a similar proof of Lemma B.2.5, we have

$$\mathbf{T}_{N}^{\bullet\circ}\circ\mathbf{T}_{N}^{\circ\bullet} = \mathbf{T}_{N;r}^{\bullet} + (q+1)\mathbf{T}_{N;r-1}^{\bullet} + (q+1)(q^{3}+1)\mathbf{T}_{N;r-2}^{\bullet} + \dots + \prod_{i=1}^{r}(q^{2i-1}+1)\mathbf{T}_{N;0}^{\bullet}$$

in \mathbb{T}_N^{\bullet} . Then under Notation 1.3.2, we have

(B.5)
$$\left((q+1)^2 \mathsf{T}_N^{\bullet} + \mathsf{T}_N^{\bullet\circ} \circ \mathsf{T}_N^{\circ\bullet} \right) \circ \mathsf{T}_N^{\bullet\circ}$$

= $\mathsf{T}_{N;r}^{\bullet} \circ \mathsf{T}_N^{\bullet\circ} + \sum_{\delta=0}^{r-1} \left((q+1) \mathsf{d}_{r-\delta,q} + (-q)^{r-\delta+1} (q+1) (q^3+1) \cdots (q^{2(r-\delta)-1}+1) \right) \mathsf{T}_{N;\delta}^{\bullet} \circ \mathsf{T}_N^{\bullet\circ}.$

By (B.3), (B.4) and (B.5), the lemma is equivalent to that for every integer $k \ge 0$, we have

$$\sum_{\delta=0}^{k} \mathbf{d}_{k-\delta,q} q^{\delta^{2}} \begin{bmatrix} k \\ \delta \end{bmatrix}_{q^{2}} = q^{k(k+2)} + \sum_{\delta=0}^{k-1} \left((q+1)\mathbf{d}_{k-\delta,q} + (-q)^{k-\delta+1}(q+1)(q^{3}+1)\cdots(q^{2(k-\delta)-1}) \right) q^{\delta(\delta+2)} \begin{bmatrix} k \\ \delta \end{bmatrix}_{q^{2}},$$

or equivalently,

(B.6)

$$\sum_{\delta=0}^{k} \mathsf{d}_{\delta,q} q^{(k-\delta)^{2}} \begin{bmatrix} k \\ \delta \end{bmatrix}_{q^{2}} = q^{k(k+2)} + \sum_{\delta=1}^{k} \left((q+1) \mathsf{d}_{\delta,q} + (-q)^{\delta+1} (q+1) (q^{3}+1) \cdots (q^{2\delta-1}+1) \right) q^{(k-\delta)(k-\delta+2)} \begin{bmatrix} k \\ \delta \end{bmatrix}_{q^{2}}.$$

By Lemma B.2.8, we have

$$(-q)^{\delta+1}(q+1)(q^{3}+1)\cdots(q^{2\delta-1}+1)$$

= $-q\sum_{j=-\delta-1}^{\delta}(-1)^{j}jq^{j^{2}+j} \begin{bmatrix} 2\delta+1\\\delta-j \end{bmatrix}_{-q} + q\sum_{j=-\delta}^{\delta}(-1)^{j}jq^{j^{2}+j} \begin{bmatrix} 2\delta\\\delta-j \end{bmatrix}_{-q}$
= $-qd_{\delta,q} + q\sum_{j=-\delta}^{\delta}(-1)^{j}jq^{j^{2}+j} \begin{bmatrix} 2\delta\\\delta-j \end{bmatrix}_{-q}$.

Thus, (B.6) is equivalent to

$$\sum_{\delta=0}^{k} \mathbf{d}_{\delta,q} q^{(k-\delta)^{2}} \begin{bmatrix} k \\ \delta \end{bmatrix}_{q^{2}} = \sum_{\delta=0}^{k} \left(\mathbf{d}_{\delta,q} + q \sum_{j=-\delta}^{\delta} (-1)^{j} j q^{j^{2}+j} \begin{bmatrix} 2\delta \\ \delta - j \end{bmatrix}_{-q} \right) q^{(k-\delta)(k-\delta+2)} \begin{bmatrix} k \\ \delta \end{bmatrix}_{q^{2}},$$

or equivalently,

(B.7)
$$\sum_{\delta=0}^{k} \mathsf{d}_{\delta,q} q^{(k-\delta)^{2}} (q^{2(k-\delta)} - 1) {k \brack \delta}_{q^{2}} = -\sum_{\delta=0}^{k} \sum_{j=-\delta}^{\delta} (-1)^{j} j q^{j^{2}+j} {2\delta \brack \delta - j}_{-q} q^{(k-\delta+1)^{2}} {k \brack \delta}_{q^{2}}.$$

However, we have

$$\begin{split} &\sum_{\delta=0}^{k} \mathbf{d}_{\delta,q} q^{(k-\delta)^{2}} (q^{2(k-\delta)}-1) \begin{bmatrix} k\\ \delta \end{bmatrix}_{q^{2}} \\ &= \sum_{\delta=0}^{k-1} \mathbf{d}_{\delta,q} q^{(k-\delta)^{2}} (q^{2(k-\delta)}-1) \begin{bmatrix} k\\ \delta \end{bmatrix}_{q^{2}} \\ &= \sum_{\delta=0}^{k-1} \sum_{j=-\delta-1}^{\delta} (-1)^{j} j q^{j^{2}+j} \begin{bmatrix} 2\delta+1\\ \delta-j \end{bmatrix}_{-q} q^{(k-\delta)^{2}} (q^{2(k-\delta)}-1) \begin{bmatrix} k\\ \delta \end{bmatrix}_{q^{2}} \\ &= \sum_{\delta=0}^{k-1} \sum_{j=-\delta-1}^{\delta} (-1)^{j} j q^{j^{2}+j} \begin{bmatrix} 2\delta+1\\ \delta-j \end{bmatrix}_{-q} q^{(k-\delta)^{2}} (q^{2\delta+2}-1) \begin{bmatrix} k\\ \delta+1 \end{bmatrix}_{q^{2}} \\ &= \sum_{\delta=0}^{k-1} \sum_{j=-\delta-1}^{\delta} (-1)^{j} j q^{(k-\delta)^{2}+j^{2}+j} ((-q)^{2\delta+2}-1) \begin{bmatrix} 2\delta+1\\ \delta-j \end{bmatrix}_{-q} \begin{bmatrix} k\\ \delta+1 \end{bmatrix}_{q^{2}} \\ &= \sum_{\delta=0}^{k-1} \sum_{j=-\delta}^{\delta} (-1)^{j} j q^{(k-\delta)^{2}+j^{2}+j} ((-q)^{\delta-j+1}-1) \begin{bmatrix} 2\delta+2\\ \delta-j+1 \end{bmatrix}_{-q} \begin{bmatrix} k\\ \delta+1 \end{bmatrix}_{q^{2}} \\ &= \sum_{\delta=0}^{k} \sum_{j=-\delta}^{\delta-1} (-1)^{j} j q^{(k+1-\delta)^{2}+j^{2}+j} ((-q)^{\delta-j}-1) \begin{bmatrix} 2\delta\\ \delta-j \end{bmatrix}_{-q} \begin{bmatrix} k\\ \delta \end{bmatrix}_{q^{2}} . \end{split}$$

Thus, (B.7) is equivalent to

$$\sum_{\delta=0}^{k} \sum_{j=-\delta}^{\delta} (-1)^{j} j q^{(k+1-\delta)^{2}+j^{2}+j} (-q)^{\delta-j} \begin{bmatrix} 2\delta \\ \delta-j \end{bmatrix}_{-q} \begin{bmatrix} k \\ \delta \end{bmatrix}_{q^{2}} = 0,$$

which is obvious since

$$\sum_{j=-\delta}^{\delta} j q^{j^2} \begin{bmatrix} 2\delta\\\delta-j \end{bmatrix}_{-q} = 0$$

The lemma is finally proved.

Lemma B.4.5. Let V be a (nondegenerate) hermitian space over O_F/\mathfrak{p} of dimension $m \geq 1$ with $r = \lfloor \frac{m}{2} \rfloor$, and $Y_0 \subseteq V$ a maximal isotropic subspace. Then the number of maximal isotropic subspaces $Y \subseteq V$ satisfying $\dim_{O_F/\mathfrak{p}}(Y \cap Y_0) = r - s$ with $0 \leq s \leq r$ is given by

$$\begin{cases} q^{s(s+2)} \begin{bmatrix} r \\ s \end{bmatrix}_{q^2}, & \text{if } m = 2r+1, \\ q^{s^2} \begin{bmatrix} r \\ s \end{bmatrix}_{q^2}, & \text{if } m = 2r. \end{cases}$$

Proof. We will prove the case for m odd and leave the similar case for m even to readers. We fix an integer $0 \le s \le r$. It is easy to see that the number of choices of the intersection $Y \cap Y_0$ (of dimension r-s) is

$$\frac{(q^{2r}-1)(q^{2(r-1)}-1)\cdots(q^{2(r-s+1)}-1)}{(q^{2s}-1)(q^{2(s-1)}-1)\cdots(q^2-1)} = \begin{bmatrix} r\\ s \end{bmatrix}_{q^2}$$

Then we count the number of Y with $Y \cap Y_0$ fixed. We take a basis $\{e_{-r}, \ldots, e_r\}$ of V such that $(e_{-i}, e_j)_V = \delta_{i,j}$ for $0 \le i, j \le r$; Y_0 is spanned by $\{e_{-r}, \ldots, e_{-1}\}$; and $Y \cap Y_0$ is spanned by $\{e_{-r}, \ldots, e_{-s-1}\}$. Let $\{f_1, \ldots, f_s\}$ be an element in Y^s such that $\{e_{-r}, \ldots, e_{-s-1}, f_1, \ldots, f_s\}$ form a basis of Y. Then since Y is isotropic, the coefficients on $\{e_{s+1}, \ldots, e_r\}$ of each f_i have to be zero. In particular, there is unique such element $\{f_1, \ldots, f_s\} \in Y^s$ that is of the form

$$(f_1, \dots, f_s) = (e_1, \dots, e_s) + (e_{-s}, \dots, e_{-1}, e_0) \begin{pmatrix} A \\ v \end{pmatrix}$$

with (uniquely determined) $A \in M_{s,s}(O_F/\mathfrak{p})$ and $v \in M_{1,s}(O_F/\mathfrak{p})$. Moreover, the isotropic condition tion on Y is equivalent to that A + A' = 0 where A' is the q-th Frobenius of A (and no condition on v). It follows that the number for such Y with given $Y \cap Y_0$ (of dimension r - s) is $q^{s(s+2)}$. Thus, the lemma follows.

APPENDIX C. Some representation theory for unitary groups

In this section, we prove several results for representations of unitary groups. Unless specified otherwise, all representations will have coefficients in \mathbb{C} . In Subsection C.1, we recall some general facts about the local base change for unitary groups. In Subsection 6.2, we study the representation appeared in the cohomology of Fermat hypersurfaces, and also compute the local base change of some semistable representations. In Subsection C.3, we collect everything we need from the endoscopic classification for unitary groups in Proposition C.3.1 and derive two corollaries from it.

C.1. Local base change for unitary groups. In this subsection, we fix an unramified quadratic extension F/F^+ of nonarchimedean local fields.

Consider a hermitian space V over F (with respect to F/F^+) of rank N. Put G := U(V). For an irreducible admissible representation π of $G(F^+)$, we denote by $BC(\pi)$ its base change, which is an irreducible admissible representation of $GL_N(F)$. Such local base change is defined by [Rog90] when $N \leq 3$ and by [Mok15, KMSW] for general N.

We review the construction of $BC(\pi)$ in certain special cases. For a parabolic subgroup P of G and an admissible representation σ of $P(F^+)$, we denote by $I_P^G(\sigma)$ the normalized parabolic induction, which is an admissible representation of $G(F^+)$. Fix a minimal parabolic subgroup P_{\min} of G.

We first review Langlands classification of irreducible admissible representations of $G(F^+)$ (see, for example, [Kon03]). For an irreducible admissible representation π of $G(F^+)$, there is a unique

parabolic subgroup P of G containing P_{\min} with Levi quotient M_P , a unique tempered representation τ of $M_P(F^+)$, and a unique strictly positive (unramified) character χ of $P_{\pi}(F^+)$, such that π is isomorphic to the unique irreducible quotient of $I_P^G(\tau\chi)$, which we denote by $J_P^G(\tau\chi)$, known as the Langlands quotient.

We then review the construction of tempered representations from discrete series representations (see, for example, [Jan14]). Let τ be an irreducible admissible tempered representation of $G(F^+)$. Then there is a unique parabolic subgroup P of G containing P_{\min} , and a discrete series representation σ of $M_P(F^+)$ such that τ is a direct summand of $I_P^G(\sigma)$. In fact, $I_P^G(\sigma)$ is a direct sum of finitely many tempered representations of multiplicity one.

Now suppose that $\pi \simeq J_P^G(\tau \chi)$ is a Langlands quotient. Then we may write

$$M_P = G_0 \times \operatorname{Res}_{F/F^+} \operatorname{GL}_{r_1} \times \cdots \times \operatorname{Res}_{F/F^+} \operatorname{GL}_{r_2}$$

with G_0 the unitary factor, under which

$$\chi = \mathbf{1} \boxtimes |\det_{r_1}|_F^{-s_1} \boxtimes \cdots \boxtimes |\det_{r_t}|_F^{-s_t}$$

for unique real numbers $0 < s_1 < \cdots < s_t$, where \det_r denotes the determinant on $\operatorname{GL}_r(F)$. Suppose $\tau = \tau_0 \boxtimes \tau_1 \boxtimes \cdots \boxtimes \tau_t$ under the above decomposition. Consider a standard parabolic subgroup \tilde{P} of GL_N whose Levi is $\operatorname{GL}_{r_t} \times \cdots \times \operatorname{GL}_{r_1} \times \operatorname{GL}_{N_0} \times \operatorname{GL}_{r_1} \times \cdots \times \operatorname{GL}_{r_t}$. Then we have

$$\mathrm{BC}(\pi) \simeq \mathrm{J}_{\tilde{P}}^{\mathrm{GL}_{N}}\left(\tau_{t}^{\vee \mathsf{c}} |\mathrm{det}_{r_{t}}|_{F}^{s_{t}} \boxtimes \cdots \boxtimes \tau_{1}^{\vee \mathsf{c}} |\mathrm{det}_{r_{1}}|_{F}^{s_{1}} \boxtimes \mathrm{BC}(\tau_{0}) \boxtimes \tau_{1} |\mathrm{det}_{r_{1}}|_{F}^{-s_{1}} \boxtimes \cdots \boxtimes \tau_{t} |\mathrm{det}_{r_{t}}|_{F}^{-s_{t}}\right)$$

which is a Langlands quotient of $\operatorname{GL}_N(F)$. Here, τ^{c} stands for $\tau \circ \mathsf{c}$.

Now suppose that τ is an irreducible admissible tempered representation of $G(F^+)$, which is a direct summand of $I_P^G(\sigma)$ for some square-integrable representation σ of $P(F^+)$. Write $\sigma = \sigma_0 \boxtimes \sigma_1 \boxtimes \cdots \boxtimes \sigma_t$, similar to the previous case. Then under the same notation, we have

$$\mathrm{BC}(\tau) \simeq \mathrm{I}_{\tilde{P}}^{\mathrm{GL}_{N}}\left(\sigma_{t}^{\vee \mathsf{c}} \boxtimes \cdots \boxtimes \sigma_{1}^{\vee \mathsf{c}} \boxtimes \mathrm{BC}(\sigma_{0}) \boxtimes \sigma_{1} \boxtimes \cdots \boxtimes \sigma_{t}\right)$$

which is an irreducible admissible representation of $GL_N(F)$.

Finally, if π is an irreducible admissible representation of $G(F^+)$ that is a constituent of an unramified principal series, then BC(π) is a constituent of an unramified principal series of GL_N(F). Thus, it makes sense to talk about the Satake parameter of BC(π), denoted by $\alpha(BC(\pi))$.

In what follows, we will suppress the parabolic subgroup \tilde{P} of GL_N when it is clear. We denote by St_N the Steinberg representation of $GL_N(F)$.

C.2. Tate–Thompson representations. In this subsection, we fix an unramified quadratic extension F/F^+ of nonarchimedean local fields, with residue field extension κ/κ^+ . Let q be the residue cardinality of F^+ and \mathfrak{p} the maximal ideal of O_F .

Let $N \geq 2$ be an integer with $r \coloneqq \lfloor \frac{N}{2} \rfloor$. Consider a hermitian space V_N over F of rank N together with a self-dual lattice Λ_N . Put $U_N \coloneqq U(V_N)$, and let K_N be the stabilizer of Λ_N which is a hyperspecial maximal subgroup of $U_N(F^+)$. Put $\bar{\Lambda}_N \coloneqq \Lambda_N \otimes_{O_{F^+}} \kappa^+$ and $\bar{U}_N \coloneqq U(\bar{\Lambda}_N)$. Then we have the reduction homomorphism $K_N \to \bar{U}_N(\kappa^+)$.

Let $\operatorname{Iso}(\bar{\Lambda}_N) \subseteq \mathbb{P}(\bar{\Lambda}_N)$ be the isotropic locus, that is, it parameterizes hyperplanes H of $\bar{\Lambda}_N$ satisfying $H^{\perp} \subseteq H$. Then $\operatorname{Iso}(\bar{\Lambda}_N)$ is a smooth hypersurface in $\mathbb{P}(\bar{\Lambda}_N)$. In particular, $\operatorname{Iso}(\bar{\Lambda}_N)$ has dimension N-2 and admits a natural action by $\overline{U}_N(\kappa^+)$. For a rational prime ℓ that is invertible in κ , put

$$\mathrm{H}^{\mathrm{prim}}(\mathrm{Iso}(\bar{\Lambda}_N)_{\overline{\kappa}}, \overline{\mathbb{Q}}_{\ell}) \coloneqq \ker\left(\cup c_1(\mathcal{O}_{\mathbb{P}(\bar{\Lambda}_N)}(1)) \colon \mathrm{H}^{N-2}_{\mathrm{\acute{e}t}}(\mathrm{Iso}(\bar{\Lambda}_N)_{\overline{\kappa}}, \overline{\mathbb{Q}}_{\ell}) \to \mathrm{H}^N_{\mathrm{\acute{e}t}}(\mathrm{Iso}(\bar{\Lambda}_N)_{\overline{\kappa}}, \overline{\mathbb{Q}}_{\ell}(1))\right).$$

It is well-known by Tate–Thompson that (see, for example, [HM78]) there is a unique irreducible representation Ω_N of $\overline{U}_N(\kappa^+)$ such that Ω_N is isomorphic to $\iota_\ell^{-1} \mathrm{H}^{\mathrm{prim}}(\mathrm{Iso}(\overline{\Lambda}_N)_{\overline{\kappa}}, \overline{\mathbb{Q}}_\ell)$ as representations of $U_N(\kappa^+)$ for every isomorphism $\iota_\ell \colon \mathbb{C} \xrightarrow{\sim} \overline{\mathbb{Q}}_\ell$. We call Ω_N the Tate-Thompson represen*tation.* We often regard Ω_N as a representation of K_N by inflation according to the context.

To describe Ω_N , we first recall some notation from parabolic induction of finite reductive groups. For every N, we fix a Borel subgroup P_N of \overline{U}_N . For positive integers r_1, \ldots, r_t satisfying $r_1 + r_2$ $\cdots + r_t \leq r$, we obtain a parabolic subgroup $\mathbf{P}_N^{(r_1,\ldots,r_t)}$ of $\bar{\mathbf{U}}_N$ containing \mathbf{P}_N , whose Levi quotient $\mathcal{M}_{N}^{(r_1,\ldots,r_t)}$ is isomorphic to $\overline{\mathcal{U}}_{N-2(r_1+\cdots+r_t)} \times \operatorname{Res}_{\kappa/\kappa^+} \operatorname{GL}_{r_1} \times \cdots \times \operatorname{Res}_{\kappa/\kappa^+} \operatorname{GL}_{r_t}$. For example, we have $\mathbf{P}_{N}^{(1r)} = \mathbf{P}_{N}$. Given a representation σ of $\mathbf{M}_{N}^{(r_{1},...,r_{t})}(\kappa^{+})$, we denote by $\operatorname{Ind}_{\mathbf{P}_{N}^{(r_{1},...,r_{t})}}^{\mathbf{U}_{N}}\sigma$ the parabolic

induction, which is a representation of $\overline{U}_N(\kappa^+)$.

Now we suppose N = 2r even. The irreducible constituents of $\operatorname{Ind}_{P_N}^{U_N} \mathbf{1}$ is parameterized by irreducible representations of the Weyl group $W_N \simeq \{\pm 1\}^r \rtimes \mathfrak{S}_r$. For every irreducible representation ϵ of W_N , we denote by $PS(\epsilon)$ the corresponding irreducible representation of $\overline{U}_N(\kappa^+)$. We now specify a character ϵ_N^{TT} : $W_N \to \{\pm 1\}$ as the extension of the product homomorphism $\{\pm 1\}^r \to \{\pm 1\}$, which is invariant under the \mathfrak{S}_r -action, to W_N that is trivial on $\{+1\}^r \rtimes \mathfrak{S}_r$.

Proposition C.2.1. We have

- (1) When N = 2r is even, the representation Ω_N is isomorphic to $PS(\epsilon_N^{TT})$.
- (2) When N = 2r is even, Ω_N is the unique nontrivial irreducible representation of $\overline{U}_N(\kappa^+)$ satisfying dim $\Omega_N^{\mathbf{P}_N(\kappa^+)} = \dim \Omega_N^{\mathbf{P}_N^{(r)}(\kappa^+)} = 1.$
- (3) The representation Ω_3 is the (unique) cuspidal unipotent representation of $\overline{U}_3(\kappa^+)$.
- (4) When N = 2r + 1 is odd with $r \ge 1$, the representation Ω_N is a multiplicity free constituent of $\operatorname{Ind}_{\operatorname{P}_{N}^{(1r-1)}}^{\overline{\mathrm{U}}_{N}}\Omega_{3}\boxtimes \mathbf{1}^{\boxtimes r-1}.$

Proof. We recall some notion of Deligne-Lusztig characters. Let \mathfrak{S}_N be the group of Npermutations, and \mathfrak{P}_N its conjugacy classes which is canonically identified with the set of partitions of N. For every $\pi \in \mathfrak{P}_N$, we let R_{π} be the Deligne-Lusztig character (of $U_N(\kappa^+)$) [DL76, Corollary 4.3] associated to the trivial representation of the maximal torus corresponding to π . Let R_N be the character of the representation Ω_N . Then by [HM78, Theorem 1], we have

(C.1)
$$R_N = (-1)^{N+1} \sum_{\pi \in \mathfrak{P}_N} \frac{\chi_N(\pi)}{z_{\pi}} R_{\pi}$$

where χ_N is the character function (on \mathfrak{P}_N) of the unique nontrivial subrepresentation of the standard representation of \mathfrak{S}_N ; and $N!/z_{\pi}$ is the cardinality of the conjugacy class π . By [DL76, Theorem 6.8, we have the following orthogonality relation

(C.2)
$$\langle R_{\pi}, R_{\pi'} \rangle = \begin{cases} 0, & \text{if } \pi \neq \pi'; \\ z_{\pi}, & \text{if } \pi = \pi'. \end{cases}$$

We are ready to prove the proposition.

For (1), note that ϵ_N^{TT} is the unique nontrivial character of W_N that is trivial on $\{+1\}^r \rtimes \mathfrak{S}_r$. Thus, (1) follows from (2) by [Cur79, Theorem 4.4.5].

For (2), it suffices to show that dim $\Omega_N^{\mathbf{P}_N(\kappa^+)} = 1$ and $\Omega_N^{\mathbf{P}_N^{(r)}(\kappa^+)} \neq 0$. Let R'_{2r} be the character of $\operatorname{Ind}_{\mathcal{P}_{2r}}^{\overline{U}_{2r}} \mathbf{1}$. Then by [DL76, Proposition 8.2], we have $R'_{2r} = R_{(2^r)}$. By (C.1) and (C.2), we have

$$\langle R_{2r}, R'_{2r} \rangle = \left\langle -\sum_{\pi \in \mathfrak{P}_{2r}} \frac{\chi_{2r}(\pi)}{z_{\pi}} R_{\pi}, R_{(2^r)} \right\rangle = -\chi_{2r}((2^r)) = -(-1) = 1,$$

which implies dim $\Omega_N^{\mathcal{P}_N(\kappa^+)} = 1$. Let $\mathcal{Y}_N \subseteq \overline{\Lambda}_N$ be the maximal isotropic subspace stabilized by $\mathcal{P}_N^{(r)}$. Then $\mathbb{P}(\mathcal{Y}_N)$ is contained in $\mathrm{Iso}(\overline{\Lambda}_N)$, which gives rise to an element in $\mathrm{CH}^{r-1}(\mathrm{Iso}(\overline{\Lambda}_N))$. It is well-known that its cohomology class subtracted by $c_1(\mathcal{O}_{\mathbb{P}(\overline{\Lambda}_N)\overline{\kappa}}(1))$ is a nonzero element in $\mathrm{H}^{\mathrm{prim}}(\mathrm{Iso}(\overline{\Lambda}_N)_{\overline{\kappa}}, \overline{\mathbb{Q}}_\ell)(r-1)$, which is fixed by $\mathcal{P}_N^{(r)}(\kappa^+)$ by construction. Thus, we have $\Omega_N^{\mathcal{P}_N^{(r)}(\kappa^+)} \neq 0$; and (1) and (2) follow.

For (3), we have $R_3 = \frac{1}{3}(R_{(1^3)} - R_{(3)})$ by (C.1). Then as computed in [Pra, Example 6.2], Ω_3 is the unique cuspidal unipotent representation of $\overline{U}_3(\kappa^+)$.

For (4), let R'_{2r+1} be the character of $\operatorname{Ind}_{\mathcal{P}_{2r+1}^{(1^{r-1})}}^{\overline{\mathbb{U}_{2r+1}}}(\Omega_3 \boxtimes \mathbf{1}^{\boxtimes r-1})$. Then by [DL76, Proposition 8.2], we have

$$R'_{2r+1} = \frac{1}{3} \left(R_{(2^{r-1},1^3)} - R_{(2^{r-1},3)} \right).$$

By (C.1) and (C.2), we have

$$\langle R_{2r+1}, R'_{2r+1} \rangle = \left\langle \sum_{\pi \in \mathfrak{P}_{2r+1}} \frac{\chi_{2r+1}(\pi)}{z_{\pi}} R_{\pi}, \frac{1}{3} \left(R_{(2^{r-1},1^3)} - R_{(2^{r-1},3)} \right) \right\rangle$$

= $\frac{1}{3} \left(\chi_{2r+1}((2^{r-1},1^3)) - \chi_{2r+1}((2^{r-1},3)) \right) = \frac{1}{3}(2 - (-1)) = 1.$

Thus, (4) follows.

From now on, we assume that N = 2r is even. Let V'_{2r} be another hermitian space over F together with a lattice Λ'_{2r} satisfying $\Lambda'_{2r} \subseteq (\Lambda'_{2r})^{\vee}$ and $(\Lambda'_{2r})^{\vee}/\Lambda'_{2r} \simeq \kappa$. Put $U'_{2r} \coloneqq U(V'_{2r})$, and let K'_{2r} be the stabilizer of Λ'_{2r} which is a special maximal subgroup of $U'_{2r}(F^+)$. The following proposition exhibits an example of the local Jacquet–Langlands correspondence.

Proposition C.2.2. Define

- S to be the set of isomorphism classes of irreducible admissible representations π of $U_{2r}(F^+)$ such that $\pi|_{K_{2r}}$ contains Ω_{2r} and that the Satake parameter of BC(π) contains {q, q⁻¹} with multiplicity one but does not contain {-1, -1} (Remark 3.1.6);
- \bigcirc S' to be the set of isomorphism classes of irreducible admissible representations π' of $U'_{2r}(F^+)$ such that $\pi'|_{K'_{2r}}$ contains the trivial representation and that the Satake parameter of BC(π) contains {q, q⁻¹} with multiplicity one but does not contain {-1, -1}.

Then there is a unique bijection between S and S' such that π and π' correspond if and only if $BC(\pi) \simeq BC(\pi')$.

Proof. We first note that both $BC(\pi)$ and $BC(\pi')$ are constituents of unramified principal series. We define a correspondence between S and S' via the condition that the two Satake parameters $\alpha(BC(\pi))$ and $\alpha(BC(\pi'))$ coincide, which is clearly a bijection. By Lemma C.2.3(2) and Lemma C.2.4 below, we have $BC(\pi) \simeq BC(\pi')$ if π and π' correspond. The proposition is proved.

Lemma C.2.3. For every irreducible admissible representation π of $U_{2r}(F^+)$ such that $\pi|_{K_{2r}}$ contains Ω_{2r} (hence π is a constituent of an unramified principal series).

- (1) If the Satake parameter of BC(π) does not contain $\{q, q^{-1}\}$ and does not contain $\{-1, -1\}$, then π is unramified.
- (2) If the Satake parameter of $BC(\pi)$ contains $\{q, q^{-1}\}$ with multiplicity one, then there exists a unique multi-subset $\{s_2, \ldots, s_r\} \subseteq \mathbb{C}/(\log q)^{-1}\pi i$ with $\operatorname{Re} s_i \geq 0$ such that if we arrange them so that $0 \leq \operatorname{Re} s_2 \leq \cdots \leq \operatorname{Re} s_r$ holds, then $BC(\pi)$ is isomorphic to the unique irreducible quotient of

$$\mathrm{I}^{\mathrm{GL}_{2r}}\left(\left|\begin{array}{c}|s_{r} \boxtimes \cdots \boxtimes |\\F \boxtimes \end{array}\right| |_{F}^{s_{2}} \boxtimes \mathrm{St}_{2} \boxtimes \left|\begin{array}{c}|-s_{2} \boxtimes \cdots \boxtimes |\\F \boxtimes \end{array}\right| |_{F}^{-s_{r}}\right).$$

Proof. We fix a decomposition

$$\Lambda_{2r} = O_F e_{-r} \oplus \cdots \oplus O_F e_{-1} \oplus O_F e_1 \oplus \cdots \oplus O_F e_r$$

in which $(e_{-i}, e_j) = \delta_{ij}$ for $1 \le i, j \le r$. For $0 \le i \le r$, put

$$V_{2i} \coloneqq Fe_{-i} \oplus \cdots \oplus Fe_{-1} \oplus Fe_1 \oplus \cdots \oplus Fe_i$$

which is a hermitian subspace of V_{2r} . We take the minimal parabolic (Borel) subgroup P_{\min} of $G \coloneqq U_{2r}$ to be the stabilizer of the flag $Fe_{-r} \subseteq \cdots \subseteq Fe_{-r} \oplus \cdots \oplus Fe_{-1}$. We also fix a Levi subgroup of P_{\min} to be $\operatorname{Res}_{F/F^+} \operatorname{GL}(Fe_1) \times \cdots \times \operatorname{Res}_{F/F^+} \operatorname{GL}(Fe_r)$.

Put $K := K_{2r}$, which is a hyperspecial maximal subgroup of $G(F^+)$. Let I be the subgroup of Kof elements whose reduction modulo \mathfrak{p} stabilizes the flag $\kappa e_{-r} \subseteq \cdots \subseteq \kappa e_{-r} \oplus \cdots \oplus \kappa e_{-1}$, which is an Iwahori subgroup of $G(F^+)$. Let J be the subgroup of K of elements whose reduction modulo \mathfrak{p} stabilizes the subspace $\kappa e_{-r} \oplus \cdots \oplus \kappa e_{-1}$, which is a parahoric subgroup of $G(F^+)$. We clearly have $I \subseteq J \subseteq K$. Now we realize the Weyl group $W_{2r} \simeq {\pm 1}^r \rtimes \mathfrak{S}_r$ explicitly as a subgroup of K. For $1 \leq i \leq r$, we let *i*-th -1 in W_{2r} correspond to the element that only switches e_{-i} and e_i , denoted by w_i . For every $\sigma \in \mathfrak{S}_r$, we let $(1^r, \sigma) \in W_{2r}$ correspond to the element that sends $e_{\pm i}$ to $e_{\pm\sigma(i)}$, denoted by $w'_{\sigma} \in J$. Then $\{w_1, w'_{(1,2)}, \ldots, w'_{(r-1,r)}\}$ is a set of distinguished generators of W_{2r} . We recall the Bruhat decompositions

$$K = \prod_{w \in W_{2r}} IwI, \quad K = \coprod_{i=0} Jw_1 \cdots w_i J.$$

For $w \in W$, we let $0 \le i(w) \le r$ be the unique integer such that $w \in Jw_1 \cdots w_{i(w)}J$.

By Proposition C.2.1(2), we have a K-equivariant embedding $\Omega_{2r} \hookrightarrow \mathbb{C}[I \setminus K]$, unique up to scalar, hence obtain a distinguished subspace $\Omega_{2r}^I \subseteq \mathbb{C}[I \setminus K/I]$ of dimension one. We would like to find a generator of Ω_{2r}^I . Now we compute the character of the $\mathbb{C}[I \setminus K/I]$ -module Ω_{2r}^I . By Proposition C.2.1(1), the element $\mathbb{1}_{Iw_1I}$ acts on Ω_{2r}^I by -1; and by Proposition C.2.1(2), which implies that Ω_{2r}^I is contained in $\mathbb{C}[J \setminus K/J]$, the element $\mathbb{1}_{Iw'_{(i,i+1)}I}$ acts on Ω_{2r}^I by $[Iw'_{(i,i+1)}I:I] = q^2$ for every $1 \leq i \leq r-1$. It follows that Ω_{2r}^I is spanned by the following function:

$$f \coloneqq \sum_{w \in W} (-1)^{i(w)} q^{r-i(w)} \cdot \mathbb{1}_{IwI} \in \mathbb{C}[I \setminus K/I].$$

Take an irreducible admissible representation π of $U_{2r}(F^+)$ such that $\pi|_K$ contains Ω_{2r} . Then it is a constituent of an unramified principal series. In other words, there exists a unique multi-subset $\{s_1, \ldots, s_r\} \subseteq \mathbb{C}/(\log q)^{-1}\pi i$ with $\operatorname{Re} s_i \geq 0$ such that π is a constituent of

$$\mathrm{I}_{P_{\min}}^{G}\left(\mid\mid_{F}^{-s_{1}}\boxtimes\cdots\boxtimes\mid\mid_{F}^{-s_{r}}\right).$$

We recall the projection map

$$\mathscr{P}\colon \mathbb{C}[I\backslash K/I] \to \mathrm{I}_{P_{\min}}^{G}\left(\mid \mid_{F}^{-s_{1}} \boxtimes \cdots \boxtimes \mid \mid_{F}^{-s_{r}}\right)^{I}$$

defined at the beginning of [Cas80, Section 2], which is $\mathbb{C}[I \setminus K/I]$ -equivariant. Put $\phi \coloneqq \mathscr{P}f$.

Now we separate the discussion.

Suppose that we are in the situation of (1). We may assume $0 \leq \text{Re } s_1 \leq \cdots \leq \text{Re } s_r$. Then there exist a unique nonnegative integer r_0 and unique positive integers r_1, \ldots, r_t satisfying $r_0 + \cdots + r_t = r$, such that

 $0 = \operatorname{Re} s_1 = \dots = \operatorname{Re} s_{r_0} < \operatorname{Re} s_{r_0+1} = \dots = \operatorname{Re} s_{r_0+r_1} < \dots < \operatorname{Re} s_{r_0+\dots+r_{t-1}+1} = \dots = \operatorname{Re} s_r$ holds. For every $1 \le i \le t$, put

$$\tau_i \coloneqq \mathbf{I}^{\mathrm{GL}_{r_i}} \left(\mid \mid_F^{-s_{r_0}+\dots+r_{i-1}+1} \boxtimes \dots \boxtimes \mid \mid_F^{-s_{r_0}+\dots+r_i} \right) \otimes \left| \det_{r_i} \right|_F^{\operatorname{Re} s_{r_0}+\dots+r_i}$$

which is an irreducible tempered representation of $\operatorname{GL}_{r_i}(F)$. Put $G_0 \coloneqq \operatorname{U}(\operatorname{V}_{2r_0})$ and $P_{0\min} \coloneqq G_0 \cap P_{\min}$. As $||_F^{-s_1} \boxtimes \cdots \boxtimes ||_F^{-s_{r_0}}$ is a discrete series representation of $P_{0\min}(F^+)$, the parabolic induction

$$\tau_0 \coloneqq \mathbf{I}_{P_{0\min}}^{G_0} \left(\mid \mid_F^{-s_1} \boxtimes \cdots \boxtimes \mid \mid_F^{-s_{r_0}} \right)$$

is a finite direct sum of irreducible tempered representations of $G_0(F^+)$. As $\{s_1, \ldots, s_{r_0}\}$ does not contain $(2 \log q)^{-1} \pi i$, τ_0 is actually irreducible by [Gol95, Theorem 1.4 & Theorem 3.4]. In particular, we obtain a Langlands quotient

$$\mathbf{J}_{P}^{G}\left(\tau_{0}\boxtimes\left(\boxtimes_{i=1}^{t}\tau_{i}|\operatorname{det}_{r_{i}}|_{F}^{-\operatorname{Re}s_{r_{0}}+\cdots+r_{i}}\right)\right),$$

where P is the parabolic subgroup of G containing P_0 whose Levi quotient is isomorphic to $G_0 \times \operatorname{Res}_{F/F^+} \operatorname{GL}_{r_1} \times \cdots \times \operatorname{Res}_{F/F^+} \operatorname{GL}_{r_t}$. We claim

(C.3)
$$\phi \neq 0 \in \mathcal{J}_P^G\left(\tau_0 \boxtimes \left(\boxtimes_{i=1}^t \tau_i |\det_{r_i}|_F^{-\operatorname{Re} s_{r_0+\cdots+r_i}}\right)\right).$$

Assuming this claim, then π is isomorphic to $J_P^G\left(\tau_0 \boxtimes \left(\boxtimes_{i=1}^t \tau_i |\det_{r_i}|_F^{-\operatorname{Re} s_{r_0}+\cdots+r_i}\right)\right)$, the unique irreducible quotient of $I_{P_{\min}}^G\left(||_F^{-s_1} \boxtimes \cdots \boxtimes ||_F^{-s_r}\right)$, which is unramified. Thus, (1) follows.

Now we prove (C.3). Let $w \in W_{2r}$ be the element acting trivially on V_{2r_0} and switching $e_{-(r_0+\cdots+r_{i-1}+j)}$ with $e_{r_0+\cdots+r_i+1-j}$ for every $1 \leq j \leq r_i$ and then every $1 \leq i \leq t$. By [Kon03, Corollary 3.2], (C.3) is equivalent to

(C.4)
$$\int_{N(F^+)} \phi(wn) \mathrm{d}n \neq 0$$

in which N is the unipotent radical of P and the integral is absolutely convergent. By [Cas80, Theorem 3.4], we have for $r_0 + 1 \le i \le r$ that

$$T_{w_i}(\phi) = \frac{q^{1-2s_i} - 1}{q(1 - q^{-2s_i})} \cdot \phi,$$

which is nonzero as $\operatorname{Re} s_i > 0$ and $s_i \neq \frac{1}{2}$. Here, the operator T_w is defined at the beginning of [Cas80, Section 3]. From this we obtain (C.4) hence (C.3).

Suppose that we are in the situation of (2). Then we may assume $s_1 = \frac{1}{2}$ and $s_i \neq \frac{1}{2}$ for $2 \leq i \leq r$. Let Q be the parabolic subgroup of G stabilizing the flag $Fe_{-r} \subseteq \cdots \subseteq Fe_{-r} \oplus \cdots \oplus Fe_{-2}$, whose Levi quotient is $U(V_2) \times \operatorname{Res}_{F/F^+} \operatorname{GL}(Fe_2) \times \cdots \times \operatorname{Res}_{F/F^+} \operatorname{GL}(Fe_r)$. Then we have a canonical inclusion

$$I_Q^G\left(\operatorname{Sp}_2\boxtimes|_F^{-s_2}\boxtimes\cdots\boxtimes|_F^{-s_r}\right)\subseteq I_{P_{\min}}^G\left(|_F^{-1/2}\boxtimes|_F^{-s_2}\boxtimes\cdots\boxtimes|_F^{-s_r}\right)$$

where Sp₂ denotes the Steinberg representation of $U(V_2)(F^+)$. As $\mathbb{1}_{Iw_1I}$ acts by -1 on ϕ , we have

$$\phi \in \mathcal{I}_Q^G\left(\operatorname{Sp}_2 \boxtimes | \mid_F^{-s_2} \boxtimes \cdots \boxtimes | \mid_F^{-s_r}\right).$$

In particular, it follows that π is a constituent of $I_Q^G(\operatorname{Sp}_2 \boxtimes | |_F^{-s_2} \boxtimes \cdots \boxtimes | |_F^{-s_r})$.

To proceed, we may assume $0 \leq \operatorname{Re} s_2 \leq \cdots \leq \operatorname{Re} s_r$. Then there exist unique positive integers r_0, \ldots, r_t satisfying $r_0 + \cdots + r_t = r$, such that

 $0 = \operatorname{Re} s_2 = \cdots = \operatorname{Re} s_{r_0} < \operatorname{Re} s_{r_0+1} = \cdots = \operatorname{Re} s_{r_0+r_1} < \cdots < \operatorname{Re} s_{r_0+\dots+r_{t-1}+1} = \cdots = \operatorname{Re} s_r$ holds. For every $1 \le i \le t$, put

$$\tau_i \coloneqq \mathrm{I}^{\mathrm{GL}_{r_i}}\left(\mid |_F^{-s_{r_0}+\dots+r_{i-1}+1} \boxtimes \dots \boxtimes \mid |_F^{-s_{r_0}+\dots+r_i} \right) \otimes |\det_{r_i}|_F^{\operatorname{Re} s_{r_0}+\dots+r_i}$$

which is an irreducible tempered representation of $\operatorname{GL}_{r_i}(F)$. Put $G_0 \coloneqq \operatorname{U}(\operatorname{V}_{2r_0})$ and $Q_0 \coloneqq G_0 \cap Q$. As $\operatorname{Sp}_2 \boxtimes ||_F^{-s_2} \boxtimes \cdots \boxtimes ||_F^{-s_{r_0}}$ is a discrete series representation of $Q_0(F^+)$, the parabolic induction

$$\mathbf{I}_{Q_0}^{G_0}\left(\mathrm{Sp}_2\boxtimes|\mid_F^{-s_2}\boxtimes\cdots\boxtimes|\mid_F^{-s_{r_0}}\right)$$

is a finite direct sum of irreducible tempered representations of $G_0(F^+)$. Let τ_0 be the unique direct summand such that ϕ is contained in the subspace

$$\mathrm{I}_{P}^{G}\left(\tau_{0}\boxtimes\left(\boxtimes_{i=1}^{t}\tau_{i}|\mathrm{det}_{r_{i}}|_{F}^{-\operatorname{Re}s_{r_{0}}+\cdots+r_{i}}\right)\right)\subseteq\mathrm{I}_{Q}^{G}\left(\operatorname{Sp}_{2}\boxtimes||_{F}^{-s_{2}}\boxtimes\cdots\boxtimes||_{F}^{-s_{r}}\right),$$

where P is the parabolic subgroup of G containing P_0 whose Levi quotient is isomorphic to $G_0 \times \operatorname{Res}_{F/F^+} \operatorname{GL}_{r_1} \times \cdots \times \operatorname{Res}_{F/F^+} \operatorname{GL}_{r_t}$. In particular, we obtain a Langlands quotient

$$\mathbf{J}_{P}^{G}\left(\tau_{0}\boxtimes\left(\boxtimes_{i=1}^{t}\tau_{i}|\mathrm{det}_{r_{i}}|_{F}^{-\operatorname{Re}s_{r_{0}}+\cdots+r_{i}}\right)\right).$$

By the same proof of (C.3), we obtain

$$\phi \neq 0 \in \mathcal{J}_P^G\left(\tau_0 \boxtimes \left(\boxtimes_{i=1}^t \tau_i |\det_{r_i}|_F^{-\operatorname{Re} s_{r_0+\cdots+r_i}}\right)\right)$$

Then $BC(\pi)$ is isomorphic to the unique irreducible quotient of

$$\mathbf{I}^{\mathrm{GL}_{2r}}\left(\left(\bigotimes_{i=t}^{1}\tau_{i}^{\vee \mathbf{c}}|\det_{r_{i}}|_{F}^{\operatorname{Re}s_{r_{0}}+\cdots+r_{i}}\right)\boxtimes \operatorname{BC}(\tau_{0})\boxtimes\left(\bigotimes_{i=1}^{t}\tau_{i}|\det_{r_{i}}|_{F}^{-\operatorname{Re}s_{r_{0}}+\cdots+r_{i}}\right)\right)$$

However, $BC(\tau_0)$ is isomorphic to

$$\mathbf{I}^{\mathrm{GL}_{2r_0}}\left(\mid \mid_F^{s_{r_0}} \boxtimes \cdots \boxtimes \mid \mid_F^{s_2} \boxtimes \mathrm{BC}(\mathrm{Sp}_2) \boxtimes \mid \mid_F^{-s_2} \boxtimes \cdots \boxtimes \mid \mid_F^{-s_{r_0}}\right)$$
$$\simeq \mathbf{I}^{\mathrm{GL}_{2r_0}}\left(\mid \mid_F^{s_{r_0}} \boxtimes \cdots \boxtimes \mid \mid_F^{s_2} \boxtimes \mathrm{St}_2 \boxtimes \mid \mid_F^{-s_2} \boxtimes \cdots \boxtimes \mid \mid_F^{-s_{r_0}}\right)$$

which is irreducible. Thus, (2) follows.

The lemma is proved.

Lemma C.2.4. For every irreducible admissible representation π' of $U'_{2r}(F^+)$ such that $(\pi')^{K'_{2r}} \neq \{0\}$, there exists a unique multi-subset $\{s_2, \ldots, s_r\} \subseteq \mathbb{C}/(\log q)^{-1}\pi i$ with $\operatorname{Re} s_i \geq 0$ such that if we arrange them so that $0 \leq \operatorname{Re} s_2 \leq \cdots \leq \operatorname{Re} s_r$ holds, then $\operatorname{BC}(\pi')$ is isomorphic to the unique irreducible quotient of

$$\mathbf{I}^{\mathrm{GL}_{2r}}\left(\left|\begin{array}{c}|_{F}^{s_{r}}\boxtimes\cdots\boxtimes|\\|_{F}^{s_{2}}\boxtimes\operatorname{St}_{2}\boxtimes|\\|_{F}^{-s_{2}}\boxtimes\cdots\boxtimes|\\|_{F}^{-s_{r}}\right).\right.$$

Proof. We fix a decomposition

$$\Lambda'_{2r} = O_F e_{-r} \oplus \dots \oplus O_F e_{-2} \oplus \Lambda'_2 \oplus O_F e_2 \oplus \dots \oplus O_F e_r$$

in which $(e_{-i}, e_j) = \delta_{ij}$ for $2 \le i, j \le r$. For $1 \le i \le r$, put

$$V'_{2i} := Fe_{-i} \oplus \dots \oplus Fe_{-2} \oplus \Lambda'_2 \otimes_{O_F} F \oplus Fe_2 \oplus \dots \oplus Fe_i$$

which is a hermitian subspace of V'_{2r} . We take the minimal parabolic subgroup P_{\min} of $G := U'_{2r}$ to be the stabilizer of the flag $Fe_{-r} \subseteq \cdots \subseteq Fe_{-r} \oplus \cdots \oplus Fe_{-2}$. We also fix a Levi subgroup of P_{\min} to be $U(V'_2) \times \operatorname{Res}_{F/F^+} \operatorname{GL}(Fe_2) \times \cdots \times \operatorname{Res}_{F/F^+} \operatorname{GL}(Fe_r)$.

Take an irreducible admissible representation π' of $G(F^+)$ such that $(\pi')^{K'_{2r}} \neq 0$. Then it is a constituent of an unramified principal series. In other words, there exists a unique multi-subset $\{s_2, \ldots, s_r\} \subseteq \mathbb{C}/(\log q)^{-1}\pi i$ with $\operatorname{Re} s_i \geq 0$ such that π' is a constituent of

$$\mathbb{I}_{P_{\min}}^{G}\left(\mathbf{1}_{2}^{\prime}\boxtimes\mid\mid_{F}^{-s_{2}}\boxtimes\cdots\boxtimes\mid\mid_{F}^{-s_{r}}
ight),$$

where $\mathbf{1}'_2$ denotes the trivial representation of $U(V'_2)(F^+)$.

To proceed, we may assume $0 \leq \operatorname{Re} s_2 \leq \cdots \leq \operatorname{Re} s_r$. Then there exist unique positive integers r_0, \ldots, r_t satisfying $r_0 + \cdots + r_t = r$, such that

$$0 = \operatorname{Re} s_2 = \dots = \operatorname{Re} s_{r_0} < \operatorname{Re} s_{r_0+1} = \dots = \operatorname{Re} s_{r_0+r_1} < \dots < \operatorname{Re} s_{r_0+\dots+r_{t-1}+1} = \dots = \operatorname{Re} s_r$$

holds. For every $1 \le i \le t$, put

$$\tau_i \coloneqq \mathbf{I}^{\mathrm{GL}_{r_i}} \left(\mid |_F^{-s_{r_0}+\dots+r_{i-1}+1} \boxtimes \dots \boxtimes \mid |_F^{-s_{r_0}+\dots+r_i} \right) \otimes |\det_{r_i}|_F^{\mathrm{Re}\,s_{r_0}+\dots+r_i}$$

which is an irreducible tempered representation of $\operatorname{GL}_{r_i}(F)$. Put $G_0 \coloneqq \operatorname{U}(\operatorname{V}'_{2r_0})$ and $P_{0\min} \coloneqq G_0 \cap P_{\min}$. As $\mathbf{1}'_2 \boxtimes |_F^{-s_2} \boxtimes \cdots \boxtimes |_F^{-s_{r_0}}$ is a discrete series representation of $P_{0\min}(F^+)$, the parabolic induction

$$\mathrm{I}_{P_{0\min}}^{G_0}\left(\mathbf{1}_2'\boxtimes|\mid_F^{-s_2}\boxtimes\cdots\boxtimes|\mid_F^{-s_{r_0}}
ight)$$

is a finite direct sum of irreducible tempered representations of $G_0(F^+)$. Let τ_0 be the unique direct summand with nonzero invariants under $K'_{2r} \cap G_0(F^+)$. In particular, we obtain a Langlands quotient

$$\mathbf{J}_{P}^{G}\left(\tau_{0}\boxtimes\left(\boxtimes_{i=1}^{t}\tau_{i}|\mathrm{det}_{r_{i}}|_{F}^{-\operatorname{Re}s_{r_{0}}+\cdots+r_{i}}\right)\right)$$

where P is the parabolic subgroup of G containing P_0 whose Levi quotient is isomorphic to $G_0 \times \operatorname{Res}_{F/F^+} \operatorname{GL}_{r_1} \times \cdots \times \operatorname{Res}_{F/F^+} \operatorname{GL}_{r_t}$. We claim

(C.5)
$$\mathbf{J}_{P}^{G}\left(\tau_{0}\boxtimes\left(\boxtimes_{i=1}^{t}\tau_{i}|\operatorname{det}_{r_{i}}|_{F}^{-\operatorname{Re}s_{r_{0}}+\cdots+r_{i}}\right)\right)^{\mathbf{K}_{2r}^{\prime}}\neq\{0\}$$

Assuming this claim, then $BC(\pi')$ is isomorphic to the unique irreducible quotient of

$$\mathbf{I}^{\mathrm{GL}_{2r}}\left(\left(\bigotimes_{i=t}^{1}\tau_{i}^{\vee \mathsf{c}}|\det_{r_{i}}|_{F}^{\operatorname{Re}s_{r_{0}}+\cdots+r_{i}}\right)\boxtimes \operatorname{BC}(\tau_{0})\boxtimes\left(\bigotimes_{i=1}^{t}\tau_{i}|\det_{r_{i}}|_{F}^{-\operatorname{Re}s_{r_{0}}+\cdots+r_{i}}\right)\right).$$

However, $BC(\tau_0)$ is isomorphic to

$$\mathbf{I}^{\mathrm{GL}_{2r_0}}\left(\left|\begin{array}{c}|\\F\end{array}\right|_{F}^{s_{r_0}}\boxtimes\cdots\boxtimes|\begin{array}{c}|\\F\end{array}\boxtimes\mathrm{BC}(\mathbf{1}_{2}')\boxtimes|\begin{array}{c}|\\F\end{array}\otimes\cdots\boxtimes|\begin{array}{c}|\\F\end{array}\otimes\cdots\boxtimes|\begin{array}{c}|\\F\end{array}\right)$$
$$\simeq\mathbf{I}^{\mathrm{GL}_{2r_0}}\left(\left|\begin{array}{c}|\\F\end{array}\boxtimes\cdots\boxtimes|\begin{array}{c}|\\F\end{array}\boxtimes\mathrm{St}_{2}\boxtimes\mathrm{St}_{2}\boxtimes|\begin{array}{c}|\\F\end{array}\otimes\cdots\boxtimes|\begin{array}{c}|\\F\end{array}\otimes\cdots\boxtimes|\begin{array}{c}|\\F\end{array}\right)$$

which is irreducible. The lemma follows.

Now we prove (C.5). Let $w \in G(F^+)$ be the element acting trivially on V'_{2r_0} and switching $e_{-(r_0+\cdots+r_{i-1}+j)}$ with $e_{r_0+\cdots+r_i+1-j}$ for every $1 \leq j \leq r_i$ and then every $1 \leq i \leq t$. Note that we have a canonical inclusion

$$\mathbf{I}_{P}^{G}\left(\tau_{0}\boxtimes\left(\boxtimes_{i=1}^{t}\tau_{i}|\det_{r_{i}}|_{F}^{-\operatorname{Re}s_{r_{0}}+\cdots+r_{i}}\right)\right)\subseteq\mathbf{I}_{P_{\min}}^{G}\left(\mathbf{1}_{2}^{\prime}\boxtimes|_{F}^{-s_{2}}\boxtimes\cdots\boxtimes|_{F}^{-s_{r}}\right)$$

which is $G(F^+)$ -equivariant. Let ϕ be the unique element in the latter space, realized as functions on $G(F^+)$, that takes value 1 on K'_{2r} , which belongs to the former space by our choice of τ_0 . By [Kon03, Corollary 3.2], (C.5) is equivalent to

(C.6)
$$\int_{N(F^+)} \phi(wn) \mathrm{d}n \neq 0$$

in which N is the unipotent radical of P and the integral is absolutely convergent. By [Cas80, Theorem 3.1], we have for $r_0 + 1 \le i \le r$ that

$$T_{w_i}(\phi) = \frac{1 - q^{-1 - 2s_i}}{1 - q^{-2s_i}} \cdot \phi$$

which is nonzero as $\operatorname{Re} s_i > 0$. Here, the operator T_w is defined at the beginning of [Cas80, Section 3]. From this we obtain (C.6) hence (C.5).

Remark C.2.5. In fact, for $\pi \in S$ and $\pi' \in S'$ in Proposition C.2.2 that correspond to each other, they also correspond under the local theta correspondence with respect to the unramified additive character and the trivial splitting character.

C.3. Results from endoscopic classification. Let F/F^+ be a totally imaginary quadratic extension of a totally real number field in the main text. We state the following proposition, which summarises all we need from Arthur's endoscopic classification for unitary groups in this article. In particular, we will use the notion of local base change for unitary groups defined over F_v^+ for every place v of F^+ , denoted by BC as well, for which we have discussed some special cases when v is inert in F in Subsection C.1.

Proposition C.3.1. Take a relevant representation (Definition 1.1.3) Π of $GL_N(\mathbb{A}_F)$. Let V be a standard definite or indefinite hermitian space over F of rank N and $\pi = \bigotimes_v \pi_v$ an irreducible admissible representation of $U(V)(\mathbb{A}_{F^+})$. We have

- (1) If $BC(\pi_v) \simeq \Pi_v$ for every place v of F^+ , then the discrete automorphic multiplicity of π is 1. In particular, (V, π) is a relevant pair (Definition 3.2.7).
- (2) If (V, π) is a relevant pair such that $BC(\pi) \simeq \Pi^{26}$ then we have $BC(\pi_v) \simeq \Pi_v$ for every place v of F^+ . In particular, the discrete automorphic multiplicity of π is 1 by (1).
- (3) If v is archimedean but not $\underline{\tau}_{\infty}$, then $BC(\pi_v) \simeq \Pi_v$ if and only if π_v is the trivial representation.
- (4) If $v = \underline{\tau}_{\infty}$, then BC(π_v) $\simeq \Pi_v$ if and only if π_v is the trivial representation (resp. is one of the N discrete series representations with the Harish-Chandra parameter $\left\{\frac{1-N}{2}, \frac{3-N}{2}, \dots, \frac{N-3}{2}, \frac{N-1}{2}\right\}\right)$ when V is definite (resp. indefinite).

Proof. Parts (1) and (2) are consequences of [KMSW, Theorem 1.7.1] for generic packets. Parts (3) and (4) follow from (1), (2), and the definition of relevant representations.

The above proposition has the following two immediate corollaries as two examples of the global Jacquet–Langlands correspondence.

Corollary C.3.2. Take a finite place \mathfrak{p} of F^+ inert in F. Let V and V' be a standard definite and a standard indefinite hermitian space over F, respectively, of even rank N = 2r, satisfying $V_v \simeq V'_v$ (for which we fix) for every place v of F^+ other than $\underline{\tau}_{\infty}$ and \mathfrak{p} . Let π be a stable automorphic representation of $U(V)(\mathbb{A}_{F^+})$ (Definition 3.2.3) such that π_{∞} is trivial and $\pi_{\mathfrak{p}}$ belongs to the set \mathcal{S} in Proposition C.2.2 (in particular, $V \otimes_{F^+} F^+_{\mathfrak{p}}$ admits a self-dual lattice). Consider the representation $\pi' \coloneqq \pi'_{\underline{\tau}_{\infty}} \otimes \pi'_{\mathfrak{p}} \otimes \pi^{\underline{\tau}_{\infty},\mathfrak{p}} \text{ of } \mathrm{U}(\mathrm{V}')(\mathbb{A}_{F^+}) \text{ where }$

- O $\pi'_{\underline{\tau}_{\infty}}$ is a discrete series representation of $U(V')(F^+_{\underline{\tau}_{\infty}})$ with the Harish-Chandra parameter
- $\{r \frac{1}{2}, r \frac{3}{2}, \dots, \frac{3}{2} r, \frac{1}{2} r\}; and$ O $\pi'_{\mathfrak{p}} \in \mathcal{S}'$ is the representation of $\mathrm{U}(\mathrm{V}')(F_{\mathfrak{p}}^+)$ corresponding to $\pi_{\mathfrak{p}}$ as in Proposition C.2.2. Then the discrete automorphic multiplicity of π' is 1.

Proof. Put $\Pi := BC(\pi)$. By Proposition C.3.1 and Proposition C.2.2, we have $BC(\pi'_v) \simeq \Pi_v$ for every place v of F^+ . The corollary follows by Proposition C.3.1(1).

Corollary C.3.3. Take a finite place \mathfrak{p} of F^+ inert in F. Let V and V' be a standard definite and a standard indefinite hermitian space over F, respectively, of odd rank N = 2r + 1, satisfying $V_v \simeq V'_v$ (for which we fix) for every place v of F^+ other than $\underline{\tau}_{\infty}$ and \mathfrak{p} . Let π' be a stable automorphic representation of $U(V')(\mathbb{A}_{F^+})$ such that $\pi'_{\mathcal{I}_{\infty}}$ is a discrete series representation of $U(V')(F_{\mathcal{I}_{\infty}}^+)$ (Definition 3.2.3) with the Harish-Chandra parameter $\{r, r-1, \ldots, 1-r, -r\}; \pi'_{\mathcal{I}}$ is trivial for every archimedean place $\underline{\tau} \neq \underline{\tau}_{\infty}$; and $\pi'_{\mathfrak{p}}$ is unramified. Consider the representation $\pi \coloneqq \pi_{\tau_{\infty}} \otimes \pi_{\mathfrak{p}} \otimes (\pi')^{\underline{\tau}_{\infty},\mathfrak{p}} \text{ of } \mathrm{U}(\mathrm{V})(\mathbb{A}_{F^+}) \text{ where}$

- O $\pi_{\underline{\tau}_{\infty}}$ is trivial; and
- O $\pi_{\mathfrak{p}}^{-\infty}$ is unramified satisfying $\mathrm{BC}(\pi_{\mathfrak{p}}) \simeq \mathrm{BC}(\pi'_{\mathfrak{p}})$.

²⁶Note that BC(π) exists by Proposition 3.2.8.

Then the discrete automorphic multiplicity of π is 1.

Proof. Put $\Pi' := BC(\pi')$. By Proposition C.3.1 and Proposition C.2.2, we have $BC(\pi_v) \simeq \Pi'_v$ for every place v of F^+ . The corollary follows by Proposition C.3.1(1).

APPENDIX D. SOME TRACE FORMULAE ARGUMENT

This appendix has two goals. In Subsection D.1, we remove some conditions in a theorem of Caraiani and Scholze [CS17]. In Subsection D.2, we prove a formula computing the dimension of old forms in an L-packet for unitary groups. These two subsections are independent on a logical level; we collect them together in one appendix mainly because the argument we use are similar, namely, trace formulae.

We keep the setup in the main text.

D.1. Vanishing of cohomology off middle degree.

Definition D.1.1. Let $N \geq 1$ be an integer, and Σ^+ a finite set of nonarchimedean places of F^+ containing Σ^+_{ram} . Consider a homomorphism $\phi \colon \mathbb{T}_N^{\Sigma^+} \to \kappa$ with κ a field. We say that ϕ is cohomologically generic if

$$\mathrm{H}^{i}_{\mathrm{\acute{e}t}}(\mathrm{Sh}(\mathrm{V},\mathrm{K}),\kappa)_{\mathbb{T}^{\Sigma^{+\prime}}_{\mathrm{M}}\cap\ker\phi}=0$$

holds for

- O every finite set $\Sigma^{+\prime}$ of nonarchimedean places of F^+ containing Σ^+ ,
- O every integer $i \neq N 1$, and
- O every standard indefinite hermitian space V over F and every object K ∈ $\mathfrak{K}(V)$ of the form $K_{\Sigma^{+\prime}} \times \prod_{v \notin \Sigma^+_{\infty} \cup \Sigma^{+\prime}} U(\Lambda)(O_{F_v^+})$ for a self-dual $\prod_{v \notin \Sigma^+_{\infty} \cup \Sigma^{+\prime}} O_{F_v}$ -lattice Λ in V ⊗_F $\mathbb{A}_F^{\Sigma^+_{\infty} \cup \Sigma^{+\prime}}$.

The following definition is essentially [CS17, Definition 1.9].

Definition D.1.2. Let $\phi: \mathbb{T}_N^{\Sigma^+} \to \kappa$ be a homomorphism with κ a field. For a place w of F^+ not in Σ^+ that splits in F, we say that ϕ is *decomposed generic at* w if $\phi(H_w) \in \kappa[T]$ has distinct (nonzero) roots in which there is no pair with ratio equal to $||w||^{27}$ Here, $H_w \in \mathbb{T}_{N,w}[T]$ is the Hecke polynomial.

Theorem D.1.3. Let $N \ge 1$ be an integer, and Σ^+ a finite set of nonarchimedean places of F^+ containing Σ^+_{ram} . Let V be a standard indefinite hermitian space over F. Let $\phi \colon \mathbb{T}_N^{\Sigma^+} \to \overline{\mathbb{F}}_{\ell}$ be a homomorphism. Suppose $[F^+ : \mathbb{Q}] > 1$. Suppose that there exists a place w of F^+ not in $\Sigma^+ \cup \Sigma^+_{\ell}$ that splits in F, such that ϕ is decomposed generic at w. Then we have

$$\mathrm{H}^{i}_{\mathrm{\acute{e}t}}(\mathrm{Sh}(\mathrm{V},\mathrm{K}),\mathbb{F}_{\ell})_{\ker\phi}=0$$

for every integer $i \neq N-1$, and every object $K \in \mathfrak{K}(V)$ of the form $K_{\Sigma^+} \times \prod_{v \notin \Sigma_{\infty}^+ \cup \Sigma^+} U(\Lambda)(O_{F_v^+})$ for a self-dual $\prod_{v \notin \Sigma_{\infty}^+ \cup \Sigma^+} O_{F_v}$ -lattice Λ in $V \otimes_F \mathbb{A}_F^{\Sigma_{\infty}^+ \cup \Sigma^+}$.

Proof. When F contains an imaginary quadratic field and every place in Σ^+ splits in F (which implies $[F^+ : \mathbb{Q}] > 1$), the statement of the theorem can be deduced from the analogous statement for the unitary similitude group, namely Case 2 of [CS17, Theorem 6.3.1(2)]. In this subsection, we will explain how to remove these restrictions.

In the statement of the theorem, let w_0 be the underlying rational prime of w. We fix an isomorphism $\mathbb{C} \simeq \overline{\mathbb{Q}}_{w_0}$ that induces the place w of F. Put $G \coloneqq \operatorname{Res}_{F^+/\mathbb{Q}} U(V)$. We have the Deligne homomorphism h: $\operatorname{Res}_{\mathbb{C}/\mathbb{R}} \mathbf{G}_m \to G \otimes_{\mathbb{Q}} \mathbb{R}$ as in Section 3.2. Put $K_{w_0,0} \coloneqq \prod_{v|w_0} U(\Lambda)(O_{F_v^+})$, which is a hyperspecial maximal subgroup of $G(\mathbb{Q}_{w_0})$. We fix a character $\varpi \colon F^{\times} \setminus \mathbb{A}_F^{\times} \to \mathbb{C}^{\times}$ that

²⁷In fact, as pointed out in [CS, Remark 1.4], there is no need to assume that the roots are distinct.

is unramified outside Σ^+ such that $\varpi|_{\mathbb{A}_{F^+}^{\times}}$ is the quadratic character η_{F/F^+} associated to F/F^+ . Put $\Sigma \coloneqq \{p \mid \Sigma_p^+ \cap \Sigma^+ \neq \emptyset\}.$

We define a subtorus $T \subseteq \operatorname{Res}_{F/\mathbb{Q}} \mathbf{G}_m$ such that for every commutative \mathbb{Q} -algebra R,

$$\mathbf{T}(R) = \{ a \in F \otimes_{\mathbb{Q}} R \mid \operatorname{Nm}_{F/F^+} a \in R^{\times} \}.$$

We fix a CM type Φ containing τ_{∞} , and a sufficiently small open compact subgroup $K_T \subseteq T(\mathbb{A}^{\infty})$ such that $(K_T)_p$ is maximal for every $p \notin \Sigma$. Then Φ induces a Deligne homomorphism $h_{\Phi} \colon \operatorname{Res}_{\mathbb{C}/\mathbb{R}} \mathbf{G}_m \to T \otimes_{\mathbb{Q}} \mathbb{R}$. We also put $\mathfrak{T} \coloneqq T(\mathbb{A}^{\infty,w_0})/T(\mathbb{Z}_{(w_0)})K_T^{w_0}$ similar to Definition 3.5.5.

Put $G := G \times T$ and $h := h \times h_{\Phi}$. Then we have the Shimura datum (G, h), which is of Hodge type. Its reflex field is the composition $F.F_{\Phi} \subseteq \mathbb{C}$. Therefore, for every sufficiently small open compact subgroup $K \subseteq G(\mathbb{A}^{\infty})$, we have the Shimura variety $\operatorname{Sh}(\tilde{G}, \tilde{h})_{K \times K_T}$, which is smooth projective (as $[F^+ : \mathbb{Q}] > 1$) over $F.F_{\Phi}$ of dimension N-1. When K is of the form $K^{w_0}K_{w_0,0}$, it has a canonical smooth projective model $\mathscr{S}(\tilde{G}, \tilde{h})_{K^{w_0}}$ over $W(\overline{\mathbb{F}}_{w_0})$ which admits a moduli interpretation similar to the one introduced in Section 4.1. Note that $F.F_{\Phi}$ is contained in $W(\overline{\mathbb{F}}_{w_0})_{\mathbb{Q}}$ under the isomorphism $\mathbb{C} \simeq \overline{\mathbb{Q}}_{w_0}$.

The discussion in [CS17], except in Section 5, is valid for all proper Shimura varieties of Hodge type including the above one. Thus, we need to modify the argument in [CS17, Section 5] for our case.

Let μ and $\tilde{\mu}$ be the Hodge cocharacters corresponding to h and h, respectively. We have the natural projection map $B(\tilde{G}, \tilde{\mu}) \to B(G, \mu)$ of Kottwitz sets, which is a bijection. For every $b \in B(G, \mu)$, we have the corresponding Kottwitz groups \tilde{J}_b and J_b , with a canonical isomorphism $\tilde{J}_b \simeq J_b \times T$. For every (sufficiently small) open compact subgroup $K^{w_0} \subseteq G(\mathbb{A}^{\infty,w_0})$ and positive integer m, we have the Igusa variety $\mathscr{I}^b_{Mant,K^{w_0},m}$ for the integral model $\mathscr{S}(\tilde{G},\tilde{h})_{K^{w_0}}$, which is a \mathfrak{T} -scheme over $\overline{\mathbb{F}}_{w_0}$. Define

$$[\mathrm{H}_{\mathfrak{T},c}(\mathscr{I}^{b}_{\mathrm{Mant}},\overline{\mathbb{Q}}_{\ell})] \coloneqq \bigoplus_{i} (-1)^{i} \lim_{\mathrm{K}^{w_{0}},m} \mathrm{H}^{i}_{\mathfrak{T},c}(\mathscr{I}^{b}_{\mathrm{Mant},\mathrm{K}^{w_{0}},m},\overline{\mathbb{Q}}_{\ell}),$$

which is virtual representation of $G(\mathbb{A}^{\infty,w_0}) \times J_b(\mathbb{Q}_{w_0})$. The crucial point is that our G is the honest unitary group, rather than the unitary similitude group. Then [CS17, Theorem 5.2.3] is modified as

$$\operatorname{tr}\left(\phi \mid \mathrm{H}_{\mathfrak{T},c}(\mathscr{I}_{\mathrm{Mant}}^{b}, \overline{\mathbb{Q}}_{\ell})\right) = \sum_{(\mathrm{H},s,\eta)} \iota(\mathrm{G}, \mathrm{H}) \mathrm{ST}_{e}^{\mathrm{H}}(\phi^{\mathrm{H}})$$

where is sum is taken over equivalent classes of elliptic endoscopic triples (H, s, η) of G; and we use the character ϖ for the Langlands–Shelstad transfer. This formula can be proved in the same way as for [Shi10, Theorem 7.2] since our Shimura variety has a similar moduli interpretation as seen in Subsection 4.1, although the Shimura datum (\tilde{G}, \tilde{h}) is not of PEL type in the sense of Kottwitz. We can fix the representatives of the triples (H, s, η) as in [CS17, Page 734] but without the similitude factor. In particular, [CS17, Corollary 5.2.5] is modified as

$$\operatorname{tr}\left(\phi \mid \operatorname{H}_{\mathfrak{T},c}(\mathscr{I}_{\operatorname{Mant}}^{b}, \overline{\mathbb{Q}}_{\ell})\right) = \sum_{\operatorname{G}_{\vec{n}}} \iota(\operatorname{G}, \operatorname{G}_{\vec{n}}) \operatorname{ST}_{e}^{\operatorname{G}_{\vec{n}}}(\phi^{\vec{n}}).$$

The next statement [CS17, Proposition 5.3.1] or rather [Shi11, Corollary 4.7], namely,

$$I_{\text{geom}}^{\mathbb{G}_{\vec{n}}\theta}(f^{\vec{n}}\theta) = \tau(\mathbf{G}_{\vec{n}})^{-1} \mathbf{ST}_{e}^{\mathbf{G}_{\vec{n}}}(\phi^{\vec{n}})$$

holds as long as $f^{\vec{n}}$ and $\phi^{\vec{n}}$ are associated in the sense of [Lab99, 3.2]. Here, $\mathbb{G}_{\vec{n}}$ is the group $\operatorname{Res}_{F/\mathbb{Q}}\operatorname{GL}_{\vec{n}} \rtimes \{1, \theta\}$. Note that, for rational primes in Σ , we do not have explicit local base change transfer. However, we will see shortly that there are enough associated pairs at these primes to make the remaining argument work, following an idea in [Shi].

For the test function $\phi \in C_c^{\infty}(G(\mathbb{A}^{\infty,w_0}) \times J_b(\mathbb{Q}_{w_0}))$ in [CS17, Theorem 5.3.2], if we assume $\phi =$ $\phi_{\Sigma} \otimes \phi^{\Sigma}$ in which ϕ_{Σ} is the characteristic function of some open compact subgroup $K_{\Sigma} \subseteq G(\mathbb{Q}_{\Sigma})$, then for every $G_{\vec{n}}$, $\phi^{\vec{n}}$ is associated to some function $f^{\vec{n}}$ in the sense above. This is shown in the claim in the proof of [Shi, Proposition 1.4]. In particular, for such ϕ , we have

$$\operatorname{tr}\left(\phi \mid \mathcal{H}_{\mathfrak{T},c}(\mathscr{I}^{b}_{\operatorname{Mant}}, \overline{\mathbb{Q}}_{\ell})\right) = \sum_{\mathcal{G}_{\vec{n}}} \iota(\mathcal{G}, \mathcal{G}_{\vec{n}}) I^{\mathbb{G}_{\vec{n}}\theta}_{\operatorname{spec}}(f^{\vec{n}}\theta)$$

in view the above identities and [CS17, (5.3.2)]. The remaining argument toward [CS17, Theorem 5.5.7] is same as it is on the GL-side, for which it suffices to use the above test functions ϕ . In fact, our case is slightly easier as we do not have the similitude factor.

The argument towards Theorem D.1.3 or [CS17, Theorem 6.3.1(2)] only uses [CS17, Theorem 5.5.7]. Therefore, Theorem D.1.3 holds.

Corollary D.1.4. Let the situation be as in Subsection 6.1. Suppose $[F^+:\mathbb{Q}] > 1$. Then for all but finitely many primes λ of E, the composite homomorphism

(D.1)
$$\mathbb{T}_N^{\Sigma^+} \xrightarrow{\phi_{\Pi}} O_E \to O_E / \lambda$$

is cohomologically generic (Definition D.1.1).

Proof. As pointed out in the proof of [CH13, Proposition 3.2.5], we can choose a nonarchimedean place w of F such that Π_w is unramified with distinct Satake parameters. Let $\{\alpha_1, \ldots, \alpha_N\}$ be the Satake parameter of Π_w , which are algebraic integers. Since Π_w is generic, we have $\alpha_i/\alpha_i \notin$ $\{1, \|w\|\}$ for $i \neq j$. Thus, for every sufficiently large rational prime ℓ , we have $\alpha_i / \alpha_i \notin \{1, \|w\|\}$ for $i \neq j$ even in \mathbb{F}_{ℓ} . Let λ be a prime of E above such a rational prime ℓ . Applying the Chebotarev density theorem to any residual Galois representation $\bar{\rho}_{\Pi,\lambda}$ of $\rho_{\Pi,\lambda}$, we conclude that there are infinitely many nonarchimedean places w of F^+ not in $\Sigma^+ \cup \Sigma^+_{\ell}$ that splits in F, such that (D.1) is decomposed generic at w (Definition D.1.2). Thus, (D.1) is cohomologically generic by Theorem D.1.3. The corollary follows.

D.2. Dimension of old forms. Let N = 2r be an even positive integer. We consider

- O a relevant representation Π of $\operatorname{GL}_N(\mathbb{A}_F)$,
- O two disjoint finite sets Σ_{\min}^+ and Σ_{\ln}^+ of nonarchimedean places of F^+ such that Σ_{\min}^+ contains $\Sigma_{\rm ram}^+$; $\Sigma_{\rm min}^+ \cup \Sigma_{\rm lr}^+$ contains Σ_{Π}^+ (Notation 3.1.4); and every place in $\Sigma_{\rm lr}^+$ is inert in F, O a finite set Σ^+ of nonarchimedean places of F^+ containing $\Sigma_{\rm min}^+ \cup \Sigma_{\rm lr}^+$,
- O a standard definite or indefinite hermitian space V over F of rank N such that V_v is not split for $v \in \Sigma_{lr}^+$,
- $O \text{ a self-dual } \prod_{v \notin \Sigma_{\min}^+ \cup \Sigma_{\ln}^+} O_{F_v} \text{-lattice } \Lambda \text{ in } \mathcal{V} \otimes_F \mathbb{A}_F^{\Sigma_{\infty}^+ \cup \Sigma_{\min}^+ \cup \Sigma_{\ln}^+},$
- **O** an object $K \in \mathfrak{K}(V)$ of the form

$$\mathbf{K} = \prod_{v \in \Sigma_{\min}^+ \cup \Sigma_{\mathrm{lr}}^+} \mathbf{K}_v \times \prod_{v \notin \Sigma_{\infty}^+ \cup \Sigma_{\min}^+ \cup \Sigma_{\mathrm{lr}}^+} \mathbf{U}(\Lambda)(O_{F_v^+}),$$

satisfying that K_v is special maximal for $v \in \Sigma_{lr}^+$. We have the homomorphism

$$\phi_{\Pi} \colon \mathbb{T}_{N}^{\Sigma^{+}} \to \overline{\mathbb{Q}}_{\ell}$$

given by Π . Fix an isomorphism $\iota_{\ell} \colon \mathbb{C} \xrightarrow{\sim} \overline{\mathbb{Q}}_{\ell}$.

Definition D.2.1. Let v be a nonarchimedean place of F^+ . We say that an open compact subgroup K_v of $U(V)(F_v^+)$ is transferable if the following two conditions are satisfied.

- (1) For every endoscopic group H of U(V_v), if we let $f_{K_v}^H$ be the endoscopic transfer of $\mathbb{1}_{K_v}$, then there exists a compactly supported smooth function $\phi_{K_v}^H$ on H(F_v) such that $\mathbb{1}_{K_v}$ and $\phi_{K_v}^H$ are associated in the sense of [Lab99, 3.2].
- (2) When H is the quasi-split unitary group of rank N, we can take $\phi_{K_v}^{H}$ to be supported on $GL_N(O_{F_v})$ once we identify $H(F_v)$ with $GL_N(F_v)$.

We call the function $\phi_{K_v}^{H}$ in (2) an *inertial transfer* of K_v if K_v is transferable, and will drop the superscript H in practice.

Lemma D.2.2. Let v be a nonarchimedean place of F^+ .

- (1) If v splits in F, then every open compact subgroup K_v is transferable.
- (2) If v is not in $\Sigma^+_{\infty} \cup \Sigma^+_{\min} \cup \Sigma^+_{lr}$, then the hyperspecial maximal subgroup $U(\Lambda)(O_{F_v^+})$ is transferable which admits $\mathbb{1}_{\mathrm{GL}_N(O_{F_n})}$ as an inertial transfer.
- (3) If v is in $\Sigma_{\min}^+ \cup \Sigma_{lr}^+$, then every sufficiently small open compact subgroup K_v is transferable.

Proof. Part (1) is trivial. Part (2) is the combination of the endoscopic fundamental lemma and the base change fundamental lemma.

For (3), for sufficiently small K_v , condition (1) in Definition D.2.1 is proved in [Mor10, Lemma 8.4.1(1)]; and condition (2) can be achieved by [Lab99, Proposition 3.1.7(2)] (see the proof of [Lab99, Proposition 3.3.2]).

Proposition D.2.3. Suppose that K_v is transferable for $v \in \Sigma_{\min}^+$. For every $v \in \Sigma_{\ln}^+$, let c_v be equal to 1 (resp. 0) if one can (resp. cannot) find purely imaginary complex numbers s_2, \ldots, s_r such that Π_v is isomorphic to the induction

$$\mathbf{I}^{\mathrm{GL}_{2r}}\left(\left|\begin{array}{c}|_{F}^{s_{r}}\boxtimes\cdots\boxtimes|\\F\end{array}\right| \xrightarrow{s_{2}}\boxtimes\mathrm{St}_{2}\boxtimes\left|\begin{array}{c}|_{F}^{-s_{2}}\boxtimes\cdots\boxtimes|\\F\end{array}\right|\right)$$

(see Subsection C.1 for the notation of induced representations). Then we have the identities

$$\dim \overline{\mathbb{Q}}_{\ell}[\operatorname{Sh}(\mathbf{V},\mathbf{K})][\iota_{\ell}\phi_{\Pi}] = \left| \prod_{v \in \Sigma_{\min}^{+}} \operatorname{tr}(\Pi_{v}(\phi_{\mathbf{K}_{v}}) \circ A_{\Pi_{v}}) \prod_{v \in \Sigma_{\operatorname{lr}}^{+}} c_{v} \right|,$$
$$\dim \mathrm{H}_{\operatorname{\acute{e}t}}^{N-1}(\operatorname{Sh}(\mathbf{V},\mathbf{K})_{\overline{F}}, \overline{\mathbb{Q}}_{\ell})[\iota_{\ell}\phi_{\Pi}] = N \left| \prod_{v \in \Sigma_{\min}^{+}} \operatorname{tr}(\Pi_{v}(\phi_{\mathbf{K}_{v}}) \circ A_{\Pi_{v}}) \prod_{v \in \Sigma_{\operatorname{lr}}^{+}} c_{v} \right|,$$

when V is definite and indefinite, respectively, for any inertial transfer ϕ_{K_v} for K_v and any normalized intertwining operator A_{Π_v} for Π_v [Shi11, Section 4.1], for $v \in \Sigma_{\min}^+$.

Proof. We only prove the case where V is indefinite, and leave the case where V is definite (which is slightly easier) to the readers.

By Proposition 3.2.4(1), we know that Π is tempered everywhere. Moreover, every discrete automorphic representation of $U(V)(\mathbb{A}_{F^+})$ whose global base change is isomorphic to Π has to be cuspidal as well. Thus, we have $H^i_{\text{ét}}(Sh(V, K)_{\overline{F}}, \overline{\mathbb{Q}}_{\ell})[\iota_{\ell}\phi_{\Pi}] = 0$ for $i \neq N - 1$.

If there exists $v \in \Sigma_{\mathrm{lr}}^+$ such that $c_v = 0$, then by Lemma C.2.4 and the above fact that Π_v is tempered, we have $\mathrm{H}_{\mathrm{\acute{e}t}}^{N-1}(\mathrm{Sh}(\mathrm{V},\mathrm{K})_{\overline{F}},\overline{\mathbb{Q}}_{\ell})[\iota_{\ell}\phi_{\Pi}] = 0$. Thus, the proposition follows. In what follows, we assume $c_v = 1$ for every $v \in \Sigma_{\mathrm{lr}}^+$.

By Proposition C.3.1 and Lemma C.2.4, we have

$$\dim \mathrm{H}^{N-1}_{\mathrm{\acute{e}t}}(\mathrm{Sh}(\mathrm{V},\mathrm{K})_{\overline{F}},\overline{\mathbb{Q}}_{\ell})[\iota_{\ell}\phi_{\Pi}] = N \prod_{v \in \Sigma^{+}_{\min}} \sum_{\mathrm{BC}(\pi_{v}) \simeq \Pi_{v}} \dim(\pi_{v})^{\mathrm{K}_{v}},$$

where the sum is taken over isomorphism classes of irreducible admissible representations π_v of $U(V)(F_v^+)$ such that $BC(\pi_v) \simeq \Pi_v$ (for $v \in \Sigma_{\min}^+$). Thus, our goal is to show

(D.2)
$$\prod_{v \in \Sigma_{\min}^+} \sum_{\mathrm{BC}(\pi_v) \simeq \Pi_v} \dim(\pi_v)^{\mathrm{K}_v} = \left| \prod_{v \in \Sigma_{\min}^+} \mathrm{tr}(\Pi_v(\phi_{\mathrm{K}_v}) \circ A_{\Pi_v}) \right|.$$

Now for $v \in \Sigma_{lr}^+$, we replace K_v by a smaller subgroup (with the same notation) that is transferable by Lemma D.2.2(3).²⁸ Let $K' \in \mathfrak{K}(V)$ be the correspondent product. So we have

$$\dim \mathrm{H}^{N-1}_{\mathrm{\acute{e}t}}(\mathrm{Sh}(\mathrm{V},\mathrm{K}')_{\overline{F}},\overline{\mathbb{Q}}_{\ell})[\iota_{\ell}\phi_{\Pi}] = N \prod_{v \in \Sigma^{+}_{\mathrm{min}} \cup \Sigma^{+}_{\mathrm{lr}}} \sum_{\mathrm{BC}(\pi_{v}) \simeq \Pi_{v}} \dim(\pi_{v})^{\mathrm{K}_{v}}.$$

On the other hand, by [Shi, (1.8) & (1.9)]²⁹ with j = 0 and ξ the trivial representation, we have

$$\dim \mathrm{H}^{N-1}_{\mathrm{\acute{e}t}}(\mathrm{Sh}(\mathrm{V},\mathrm{K})_{\overline{F}},\overline{\mathbb{Q}}_{\ell})[\iota_{\ell}\phi_{\Pi}] = N \left| \prod_{v \in \Sigma^{+}_{\min} \cup \Sigma^{+}_{\mathrm{Ir}}} \mathrm{tr}(\Pi_{v}(\phi_{\mathrm{K}_{v}}) \circ A_{\Pi_{v}}) \right|,$$

where $\phi_{\mathbf{K}_v}$ is any inertial transfer for \mathbf{K}_v and A_{Π_v} is any normalized intertwining operator for Π_v , for $v \in \Sigma_{\mathrm{lr}}^+$. Since $\sum_{\mathrm{BC}(\pi_v)\simeq\Pi_v} \dim(\pi_v)^{\mathbf{K}_v} \ge 1$ for every $v \in \Sigma_{\mathrm{lr}}^+$, (D.2) will be implied by

(D.3)
$$\prod_{v \in \Sigma_{\mathrm{lr}}^+} \sum_{\mathrm{BC}(\pi_v) \simeq \Pi_v} \dim(\pi_v)^{\mathrm{K}_v} = \left| \prod_{v \in \Sigma_{\mathrm{lr}}^+} \operatorname{tr}(\Pi_v(\phi_{\mathrm{K}_v}) \circ A_{\Pi_v}) \right|.$$

For this, we choose an imaginary quadratic number field $E \subseteq \mathbb{C}$ satisfying

- O E is not included in F;
- O if a rational prime p underlies Σ_{\min}^+ , then p splits in E;
- O if a rational prime p underlies Σ_{lr}^+ , then p is inert in E;
- O the quadratic base change of Π to $\check{F} \coloneqq F.E$, denoted by $\check{\Pi}$, remains cuspidal (hence relevant).

Let $\check{F}^+ \subseteq \check{F}$ be the maximal totally real subfield; let $\check{\Sigma}^+_{\rm lr}$ be the set of places of \check{F}^+ above $\check{\Sigma}^+_{\rm lr}$; and let $\check{\Sigma}^+$ the (finite set) of nonarchimedean places \check{v} of \check{F}^+ not in $\check{\Sigma}^+_{\rm lr}$ such that $\check{\Pi}_{\check{v}}$ is ramified. By our choice of E, \check{F}/\check{F}^+ is everywhere unramified; every place in $\check{\Sigma}^+$ splits in \check{F} ; and every place in $\check{\Sigma}^+_{\rm lr}$ splits into two places of \check{F}^+ both inert in \check{F} . Let \check{V} be the standard definite hermitian space over \check{F} of rank N such that $\check{V}_{\check{v}}$ (for a nonarchimedean place \check{v}) is not split if and only if $\check{v} \in \check{\Sigma}^+_{\rm lr}$. Take an object $\check{K} \in \mathfrak{K}(\check{V})$ of the form $\check{K} = \prod \check{K}_{\check{v}}$ satisfying

- O $\breve{K}_{\breve{v}}$ is hyperspecial maximal if \breve{v} is inert in \breve{F} and not in $\breve{\Sigma}_{lr}^+$;
- O $\breve{K}_{\breve{v}}$ is given by K_v for $\breve{v} \in \breve{\Sigma}_{lr}^+$, where $v \in \Sigma_{lr}^+$ underlies \breve{v} ;
- O $\check{\pi}_{\check{v}}$ has nonzero $\check{K}_{\check{v}}$ invariants if \check{v} splits in \check{F} , where $\check{\pi}_{\check{v}}$ is the descent of $\check{\Pi}_{\check{v}}$ to an irreducible admissible representation of $U(\check{V})(\check{F}_{\check{v}}^+)$.

Then we have

(D.4)
$$\dim \overline{\mathbb{Q}}_{\ell}[\mathrm{Sh}(\breve{\mathrm{V}},\breve{\mathrm{K}})][\iota_{\ell}\phi_{\breve{\mathrm{II}}}] = \prod_{\breve{v}\in\breve{\Sigma}^{+}\cup\breve{\Sigma}_{\mathrm{lr}}^{+}}\sum_{\mathrm{BC}(\breve{\pi}_{\breve{v}})\simeq\breve{\mathrm{II}}_{\breve{v}}}\dim(\breve{\pi}_{\breve{v}})^{\breve{\mathrm{K}}_{\breve{v}}}.$$

 $^{^{28}\}mathrm{We}$ expect that the special maximal subgroup K_v is already transferable; but we do not need to address this issue.

²⁹Strictly speaking, Shin's formulae are stated for unitary similitude groups and assuming F containing an imaginary quadratic subfield. However, we can modify his argument to honest unitary groups and without the constrain on F by our moduli interpretation as we did in the proof of Theorem D.1.3.

On the other hand, by [Shi, (1.8) & (1.9)], we have

(D.5)
$$\dim \overline{\mathbb{Q}}_{\ell}[\operatorname{Sh}(\breve{\mathrm{V}},\breve{\mathrm{K}})][\iota_{\ell}\phi_{\breve{\Pi}}] = \left|\prod_{\breve{v}\in\breve{\Sigma}^{+}\cup\breve{\Sigma}_{\operatorname{lr}}^{+}}\operatorname{tr}(\breve{\Pi}_{\breve{v}}(\phi_{\breve{\mathrm{K}}_{\breve{v}}})\circ A_{\breve{\Pi}_{\breve{v}}})\right|.$$

Here, for $\check{v} \in \check{\Sigma}^+$, we take $\phi_{\check{K}_{\check{v}}}$ to be $\mathbb{1}_{\check{K}_{\check{v}}} \otimes \mathbb{1}_{\check{K}_{\check{v}}}$; and for $\check{v} \in \check{\Sigma}_{\mathrm{lr}}^+$, we take $\phi_{\check{K}_{\check{v}}}$ to be ϕ_{K_v} where v is place of F^+ underlying \check{v} . Then it is easy to see that for $\check{v} \in \check{\Sigma}^+$, we have

(D.6)
$$\left| \operatorname{tr}(\breve{\Pi}_{\breve{v}}(\phi_{\breve{K}_{\breve{v}}}) \circ A_{\breve{\Pi}_{\breve{v}}}) \right| = \sum_{\mathrm{BC}(\breve{\pi}_{\breve{v}}) \simeq \breve{\Pi}_{\breve{v}}} \dim(\breve{\pi}_{\breve{v}})^{\breve{K}_{\breve{v}}} \ge 1$$

(in fact, the sum is taken over a singleton). Combining (D.4), (D.5), and (D.6), we obtain

$$\prod_{\breve{v}\in \check{\Sigma}_{\mathrm{lr}}^+} \sum_{\mathrm{BC}(\breve{\pi}_{\breve{v}})\simeq \breve{\Pi}_{\breve{v}}} \dim(\breve{\pi}_{\breve{v}})^{\breve{K}_{\breve{v}}} = \left|\prod_{\breve{v}\in \check{\Sigma}_{\mathrm{lr}}^+} \mathrm{tr}(\breve{\Pi}_{\breve{v}}(\phi_{\breve{K}_{\breve{v}}})\circ A_{\breve{\Pi}_{\breve{v}}})\right|,$$

which is nothing but

$$\left(\prod_{v\in\Sigma_{\mathrm{lr}}^+}\sum_{\mathrm{BC}(\pi_v)\simeq\Pi_v}\dim(\pi_v)^{\mathrm{K}_v}\right)^2 = \left|\prod_{v\in\Sigma_{\mathrm{lr}}^+}\mathrm{tr}(\Pi_v(\phi_{\mathrm{K}_v})\circ A_{\Pi_v})\right|^2.$$

Thus, (D.3) follows. The proposition is proved.

Appendix E. Deformation of Galois representations

We consider a subfield $F \subseteq \mathbb{C}$ that is a CM number field. We adopt the notation concerning ground fields in Subsection 1.3; and we put $\eta := \eta_{F/F^+}$ for short. The main objective of this appendix is to generalize results in [CHT08] and [Tho12] concerning the relation between the Galois deformation algebra and the Hecke algebra, informally known as R=T theorems. In Subsection E.1, we collect some facts concerning essentially conjugate self-dual representations, which is also frequently used in the main text. In Subsection E.2, we recall the notion and facts of lifting and deformation of Galois representations. In Subsection E.3, we study Fontaine–Laffaille local deformations. In Subsection E.4, we study representations of tame groups, which will be used in the next two subsections. In Subsection E.5, we study minimally ramified local deformations. In Subsection E.6, we study local deformations related to the level raising. In Subsection E.7, we state and prove our R=T theorem for both unitary Shimura sets and unitary Shimura varieties. In Subsection E.8, we study the rigidity property for reduction of automorphic Galois representations, in the sense of Definition E.7.1.

In this appendix, we shall slightly change our notation system from Section 2 to fit the one used in [CHT08]. We fix an odd prime ℓ and an isomorphism $\iota_{\ell} \colon \mathbb{C} \xrightarrow{\sim} \overline{\mathbb{Q}}_{\ell}$. Consider a finite extension E_{λ} of \mathbb{Q}_{ℓ} inside $\overline{\mathbb{Q}}_{\ell}$. Let \mathscr{O} (rather than O_{λ} in Section 2) be the ring of integers of E_{λ} , λ the maximal ideal of \mathscr{O} , and $k := \mathscr{O}/\lambda$ the residue field. Following [CHT08], we denote by $\mathscr{C}_{\mathscr{O}}^{f}$ the category of commutative local Artinian \mathscr{O} -algebras with residue field k, and let $\mathscr{C}_{\mathscr{O}}$ be the category of topological local \mathscr{O} -algebras whose objects are inverse limits of objects of $\mathscr{C}_{\mathscr{O}}^{f}$. For an object Rof $\mathscr{C}_{\mathscr{O}}$, we shall denote by \mathfrak{m}_{R} its maximal ideal. For an \mathscr{O} -valued character, we will use the same notation for its induced R-valued character for every object R of $\mathscr{C}_{\mathscr{O}}$.

Take an integer $N \ge 1$.

E.1. Extension of essentially conjugate self-dual representations. In this subsection, we collect some notion and facts on the extension of essentially conjugate self-dual representations.

Notation E.1.1. We recall the group scheme \mathscr{G}_N from [CHT08, Section 1]. Put

$$\mathscr{G}_N \coloneqq (\mathrm{GL}_N \times \mathrm{GL}_1) \rtimes \{1, \mathfrak{c}\}$$

with $c^2 = 1$ and

$$\mathfrak{c}(g,\mu)\mathfrak{c} = (\mu^{\mathrm{t}}g^{-1},\mu)$$

for $(g, \mu) \in \operatorname{GL}_N \times \operatorname{GL}_1$. In what follows, we will often regard GL_N as a subgroup of \mathscr{G}_N via the embedding $g \mapsto (g, 1, 1)$. Denote by $\nu \colon \mathscr{G}_N \to \operatorname{GL}_1$ the homomorphism such that $\nu|_{\operatorname{GL}_N \times \operatorname{GL}_1}$ is the projection to the factor GL_1 and that $\nu(\mathfrak{c}) = -1$. We have the adjoint action ad of \mathscr{G}_N on M_N , given by

$$ad(g,\mu)(x) = gxg^{-1}, \quad ad(c)(x) = -{}^{t}x$$

for $x \in \mathcal{M}_N$ and $(g, \mu) \in \mathcal{GL}_N \times \mathcal{GL}_1$.

Let $\tilde{\Gamma}$ be a topological group, and $\Gamma \subseteq \tilde{\Gamma}$ a subgroup of index at most two.

Notation E.1.2. Let R (rather than L in Section 2) be a commutative topological \mathbb{Z}_{ℓ} -algebra. For a continuous homomorphism

$$r: \tilde{\Gamma} \to \mathscr{G}_N(R)$$

such that the image of $r|_{\Gamma}$ is contained in $\operatorname{GL}_N(R) \times R^{\times}$, we denote

 $r^{\natural} \colon \Gamma \to \operatorname{GL}_N(R) \times R^{\times} \to \operatorname{GL}_N(R)$

the composition of $r|_{\Gamma}$ with the projection to $\operatorname{GL}_N(R)$.

Lemma E.1.3. Suppose $[\tilde{\Gamma} : \Gamma] = 2$. Let R be a commutative topological \mathbb{Z}_{ℓ} -algebra and $\chi : \tilde{\Gamma} \to R^{\times}$ a continuous character. We have

(1) If $r: \tilde{\Gamma} \to \mathscr{G}_N(R)$ is a continuous homomorphism satisfying $r^{-1}(\operatorname{GL}_N(R) \times R^{\times}) = \Gamma$ and $\nu \circ r = \chi$, then for every $\gamma \in \tilde{\Gamma} \setminus \Gamma$, we have

$$r^{\natural,\gamma} = B \circ \chi r^{\natural,\vee} \circ B^{-1},$$

where A is given from $r(\gamma) = (B, -\chi(\gamma), \mathfrak{c})$.

(2) Let $\rho: \Gamma \to \operatorname{GL}_N(R)$ be a continuous homomorphism, γ an element in $\tilde{\Gamma} \setminus \Gamma$, and $B \in \operatorname{GL}_N(R)$ such that $\rho^{\gamma} = B \circ \chi \rho^{\vee} \circ B^{-1}$ and $B^{\dagger}B^{-1} = \mu_B \chi(\gamma)^{-1} \rho(\gamma^2)$ for some $\mu_B \in \{\pm 1\}$. Then there exists a unique continuous homomorphism

$$r \colon \tilde{\Gamma} \to \mathscr{G}_N(R)$$

satisfying $r|_{\Gamma} = (\rho, \chi|_{\Gamma}, 1)$ and $r(\gamma) = (B, \mu_B \chi(\gamma), \mathfrak{c})$.

(3) Suppose in (2) that R is a field and ρ is absolutely irreducible. If ρ^{γ} and $\chi \rho^{\vee}$ are conjugate, then ρ induces a homomorphism $r \colon \tilde{\Gamma} \to \mathscr{G}_N(R)$ satisfying $r|_{\Gamma} = (\rho, \chi)$, unique up to changing the $\operatorname{GL}_N(R)$ -component of $r(\gamma)$ by a scalar in R^{\times} .

Proof. Part (1) is a special case of [CHT08, Lemma 2.1.1].

For (2), we check that

$$r(\gamma^2) = (B, \mu_B \chi(\gamma), \mathfrak{c}) \cdot (B, \mu_B \chi(\gamma), \mathfrak{c}) = (\mu_B \chi(\gamma) B^{\mathsf{t}} B^{-1}, \chi(\gamma^2), 1) = (\rho(\gamma^2), \chi(\gamma^2), 1).$$

Since Γ is generated by Γ and γ , we obtain a unique continuous homomorphism $r \colon \tilde{\Gamma} \to \mathscr{G}_N(R)$ as in (2).

For (3), by Schur's lemma, the element B is unique up to scalar in R^{\times} , which implies the existence and also the uniqueness of μ_B . Thus, (3) follows immediately.

E.2. Deformation problems. In this subsection, we introduce the notion of deformation problems. Let $\tilde{\Gamma}$ be a topological group, and $\Gamma \subseteq \tilde{\Gamma}$ a subgroup of index at most two.

Notation E.2.1. We consider a pair (\bar{r}, χ) , where

 $O \ \bar{r} \colon \tilde{\Gamma} \to \mathscr{G}_N(k)$ is a homomorphism,

 $O_{\chi}: \tilde{\Gamma} \to \mathscr{O}^{\times}$ a continuous homomorphism, known as the *similitude character*,

subject to the relation $\bar{r}^{-1}(\operatorname{GL}_N(k) \times k^{\times}) = \Gamma$ and $\nu \circ \bar{r} = \chi$.

The following definition slightly generalizes [CHT08, Definition 2.2.1].

Definition E.2.2. A lifting of \bar{r} to an object R of $\mathscr{C}_{\mathscr{O}}$ is a continuous homomorphism $r: \tilde{\Gamma} \to \mathscr{G}_N(R)$ satisfying $r \mod \mathfrak{m}_R = \bar{r}$ and $\nu \circ r = \chi$. We say that two liftings are equivalent if they are conjugate by an element in $1 + \operatorname{M}_N(\mathfrak{m}_R) \subset \operatorname{GL}_N(R) \subset \mathscr{G}_N(R)$. By a deformation of \bar{r} , we mean an equivalence class of liftings of r.³⁰

Now suppose that Γ is topologically finitely generated. Then there exists a universal lifting

$$r^{\text{univ}} \colon \widetilde{\Gamma} \to \mathscr{G}_N(\mathsf{R}^{\text{loc}}_{\overline{r}})$$

of \bar{r} to an object $\mathsf{R}_{\bar{r}}^{\mathrm{loc}}$ of $\mathscr{C}_{\mathscr{O}}$ such that, for every object R of $\mathscr{C}_{\mathscr{O}}$, the set of liftings of \bar{r} to R is in natural bijection with $\mathrm{Hom}_{\mathscr{C}_{\mathscr{O}}}(\mathsf{R}_{\bar{r}}^{\mathrm{loc}}, R)$. Since Γ is topologically finitely generated, it is well-known that $\mathsf{R}_{\bar{r}}^{\mathrm{loc}}$ is Noetherian; and there exists natural isomorphisms

$$\operatorname{Hom}_{k}\left(\mathfrak{m}_{\mathsf{R}^{\operatorname{loc}}_{\bar{r}}}/(\mathfrak{m}^{2}_{\mathsf{R}^{\operatorname{loc}}_{\bar{r}}},\lambda),k\right)\simeq\operatorname{Hom}_{\mathscr{C}_{\mathcal{O}}}\left(\mathsf{R}^{\operatorname{loc}}_{\bar{r}},k[\varepsilon]/(\varepsilon^{2})\right)\simeq\operatorname{Z}^{1}(\tilde{\Gamma},\operatorname{ad}\bar{r}),$$

where $Z^1(\tilde{\Gamma}, \operatorname{ad} \bar{r})$ denotes the group of 1-cocycles of $\tilde{\Gamma}$ with values in the adjoint representation $(\operatorname{ad} \bar{r}, \operatorname{M}_N(k))$. Explicitly, for $\phi \in Z^1(\tilde{\Gamma}, \operatorname{ad} \bar{r})$, the corresponding lifting of \bar{r} to $k[\varepsilon]/(\varepsilon^2)$ is given by

$$r_{\phi}(g) = (1 + \varepsilon \phi(g))\bar{r}(g)$$

for every $g \in \tilde{\Gamma}$. For two cocycles $\phi_1, \phi_2 \in Z^1(\tilde{\Gamma}, \operatorname{ad} \bar{r})$, the corresponding liftings r_{ϕ_1} and r_{ϕ_2} are equivalent if and only if there exists an element $x \in M_N(k)$ such that

$$\phi_1(g) - \phi_2(g) = (1 - \operatorname{ad} \bar{r}(g))(x)$$

for every $g \in \tilde{\Gamma}$. Thus, the equivalence classes of liftings of \bar{r} to $k[\varepsilon]/(\varepsilon^2)$ is in natural bijection with $\mathrm{H}^1(\tilde{\Gamma}, \mathrm{ad}\,\bar{r})$.

Definition E.2.3. A local deformation problem of \bar{r} is a closed formal subscheme \mathscr{D} of Spf $\mathsf{R}_{\bar{r}}^{\mathrm{loc}}$ that is invariant under the conjugate action by $1 + \mathrm{M}_{N}(\mathfrak{m}_{\mathsf{R}^{\mathrm{loc}}})$.

By the moduli interpretation of $\mathsf{R}_{\bar{r}}^{\mathrm{loc}}$, giving a local deformation problem of \bar{r} is equivalent to giving a collection of liftings of \bar{r} to objects in $\mathscr{C}_{\mathscr{O}}$ satisfying certain conditions (see [CHT08, Definition 2.2.2 & Lemma 2.2.3]).

Definition E.2.4. For a local deformation problem \mathscr{D} of \bar{r} , we define the *tangent space of* \mathscr{D} , denoted by $L(\mathscr{D})$, to be the image of the subspace

$$\mathrm{L}^{1}(\mathscr{D}) \coloneqq \mathrm{Hom}_{k}\left(\mathfrak{m}_{\mathsf{R}^{\mathrm{loc}}_{\bar{r}}}/(\mathfrak{m}^{2}_{\mathsf{R}^{\mathrm{loc}}_{\bar{r}}},\mathscr{I},\lambda),k\right) \subseteq \mathrm{Z}^{1}(\tilde{\Gamma},\mathrm{ad}\,\bar{r})$$

under the natural map $Z^1(\tilde{\Gamma}, \operatorname{ad} \bar{r}) \to H^1(\tilde{\Gamma}, \operatorname{ad} \bar{r})$, where $\mathscr{I} \subseteq \mathsf{R}^{\operatorname{loc}}_{\bar{r}}$ is the closed ideal defining \mathscr{D} .

Note that we have the identity

(E.1)
$$\dim_k \mathcal{L}^1(\mathscr{D}) = N^2 + \dim_k \mathcal{L}(\mathscr{D}) - \dim_k \mathcal{H}^0(\tilde{\Gamma}, \operatorname{ad} \bar{r}).$$

³⁰Strictly speaking, a lifting or a deformation of \bar{r} depends on the similitude character χ . But we choose to follow the terminology in [CHT08] by not spelling the characters out, as the relevance on the similitude character is always clear from the context.

Remark E.2.5. Later, when we consider a nonarchimedean place v of F^+ and take $\tilde{\Gamma} = \Gamma_{F_v^+}$, the subgroup Γ we implicitly take is always $\Gamma_{F_v^+} \cap \Gamma_F$.

Now we apply Notation E.2.1 and Definition E.2.2 to the case where $\tilde{\Gamma} = \Gamma_{F^+}$ and $\Gamma = \Gamma_F$.

Definition E.2.6. A global deformation problem is a tuple $(\bar{r}, \chi, S, \{\mathscr{D}_v\}_{v \in S})$, where

- O (\bar{r}, χ) is a pair as in Notation E.2.1;
- O S is a finite set of nonarchimedean places of F^+ containing all ℓ -adic places and those places v such that \bar{r}_v is ramified;
- O \mathscr{D}_v is a local deformation problem of \bar{r}_v (Remark E.2.5) for each $v \in S$.

We take a global deformation problem $\mathscr{S} \coloneqq (\bar{r}, \chi, S, \{\mathscr{D}_v\}_{v \in S})$. For $v \in S$, we denote by \mathscr{I}_v the closed ideal of $\mathsf{R}^{\mathrm{loc}}_{\bar{r}_v}$ defining \mathscr{D}_v . For a subset $T \subseteq S$, put

(E.2)
$$\mathsf{R}^{\mathrm{loc}}_{\mathscr{S},\mathrm{T}} \coloneqq \widehat{\bigotimes}_{v\in\mathrm{T}} \mathsf{R}^{\mathrm{loc}}_{\bar{r}_v} / \mathscr{I}_v,$$

where the completed tensor product is taken over \mathscr{O} . Recall from [CHT08, Definition 2.2.1] that a T-framed lifting of \bar{r} to an object R of $\mathscr{C}_{\mathscr{O}}$ is a tuple $(r; \{\beta_v\}_{v\in T})$, where r is a lifting of \bar{r} to R (Definition E.2.2), and $\beta_v \in 1 + M_N(\mathfrak{m}_R)$ for $v \in T$. Two T-framed liftings $(r; \{\beta_v\}_{v\in T})$ and $(r'; \{\beta'_v\}_{v\in T})$ of \bar{r} to R are said equivalent, if there exists $x \in 1 + M_N(\mathfrak{m}_R)$ such that $r' = x^{-1} \circ r \circ x$ and $\beta'_v = x^{-1}\beta_v$ for every $v \in T$. A T-framed deformation of \bar{r} is an equivalence class of T-framed liftings of \bar{r} . We say that a T-framed lifting $(r; \{\beta_v\}_{v\in T})$ is of type \mathscr{S} if r_v belongs to \mathscr{D}_v for every $v \in S$, and is unramified for every $v \notin S$. Note that being of type \mathscr{S} is a property invariant under the conjugate action by $1 + M_N(\mathfrak{m}_R)$. Thus it makes sense to speak of T-framed deformation of type \mathscr{S} . Let $\operatorname{Def}_{\mathscr{S}}^{\Box_T} : \mathscr{C}_{\mathscr{O}} \to \operatorname{Set}$ be the functor that sends an object R to the set of T-framed deformations of \bar{r} to R of type \mathscr{S} .

Let $\Gamma_{F^+,S}$ be the Galois group of the maximal subextension of \overline{F}/F^+ that is unramified outside S. Recall the cohomology group $\mathrm{H}^i_{\mathscr{S},\mathrm{T}}(\Gamma_{F^+,S},\mathrm{ad}\,\bar{r})$ for $i \geq 0$ introduced after [CHT08, Definition 2.2.7]. By [CHT08, Lemma 2.3.4], these are finite dimensional k-vector spaces, and satisfy $\mathrm{H}^i_{\mathscr{S},\mathrm{T}}(\Gamma_{F^+,S},\mathrm{ad}\,\bar{r}) = 0$ for i > 3.

Proposition E.2.7. Assume that $\bar{r}|_{\Gamma_F}$ is absolutely irreducible. Then for every subset $T \subseteq S$, the functor $\operatorname{Def}_{\mathscr{S}}^{\Box_T}$ is represented by a Noetherian \mathscr{O} -algebra $\mathsf{R}_{\mathscr{S}}^{\Box_T}$ in $\mathscr{C}_{\mathscr{O}}$. Put $\mathsf{R}_{\mathscr{S}}^{\operatorname{univ}} \coloneqq \mathsf{R}_{\mathscr{S}}^{\Box_{\emptyset}}$. We further have

(1) There is a canonical isomorphism

$$\operatorname{Hom}_{k}\left(\mathfrak{m}_{\mathsf{R}_{\mathscr{S}}^{\Box_{\mathrm{T}}}}/(\mathfrak{m}_{\mathsf{R}_{\mathscr{S}}^{\Box_{\mathrm{T}}}}^{2},\lambda,\mathfrak{m}_{\mathsf{R}_{\mathscr{S},\mathrm{T}}^{\mathrm{loc}}}),k\right)\simeq \mathrm{H}_{\mathscr{S},\mathrm{T}}^{1}(\Gamma_{F^{+},\mathrm{S}},\mathrm{ad}\,\bar{r}).$$

where we regard $\mathfrak{m}_{\mathsf{R}^{\mathrm{loc}}_{\mathscr{S},\mathrm{T}}}$ as its image under the tautological homomorphism $\mathsf{R}^{\mathrm{loc}}_{\mathscr{S},\mathrm{T}} \to \mathsf{R}^{\mathrm{D}_{\mathrm{T}}}_{\mathscr{S}}$. Moreover, if $\mathrm{H}^{2}_{\mathscr{S},\mathrm{T}}(\Gamma_{F^{+},\mathrm{S}}, \mathrm{ad}\,\bar{r}) = 0$ and for $v \in \mathrm{S} \setminus \mathrm{T}$, \mathscr{D}_{v} is formally smooth over \mathscr{O} , then $\mathsf{R}^{\mathrm{D}_{\mathrm{T}}}_{\mathscr{S}}$ is a power series ring over $\mathsf{R}^{\mathrm{loc}}_{\mathscr{S},\mathrm{T}}$ in $\dim_{k} \mathrm{H}^{1}_{\mathscr{S},\mathrm{T}}(\Gamma_{F^{+},\mathrm{S}}, \mathrm{ad}\,\bar{r})$ variables.

(2) The choice of a lifting $r_{\mathscr{S}}^{\text{univ}} \colon \Gamma_{F^+} \to \mathscr{G}_N(\mathsf{R}_{\mathscr{S}}^{\text{univ}})$ in the universal deformation determines an extension of the tautological homomorphism $\mathsf{R}_{\mathscr{S}}^{\text{univ}} \to \mathsf{R}_{\mathscr{S}}^{\Box_{\mathrm{T}}}$ to an isomorphism

$$\mathsf{R}^{\mathrm{univ}}_{\mathscr{S}}[[X_{v;i,j}]]_{v\in\mathrm{T};1\leq i,j\leq N}\xrightarrow{\sim}\mathsf{R}^{\Box_{\mathrm{T}}}_{\mathscr{S}}$$

such that, for every $v \in T$, the universal frame at v is given by $\beta_v = 1 + (X_{v;i,j})_{1 \le i,j \le N}$.

Proof. These are exactly [CHT08, Proposition 2.2.9 & Corollary 2.2.13] except that they consider only local deformation problems at split places (that is, they assume that all places in S are split in F). However, the same argument can be applied to the general case without change.
E.3. Fontaine-Laffaille deformations. In this subsection, we study Fontaine-Laffaille deformations at ℓ -adic places. We take a place v of F^+ above ℓ ; and let w be the place of F above v induced by the inclusion $F \subseteq \overline{F}_v^+$. We assume that ℓ is unramified in F, and denote by $\sigma \in \operatorname{Gal}(F_w/\mathbb{Q}_\ell)$ the absolute Frobenius element.

Definition E.3.1. We say that E_{λ} is *F*-inclusive if E_{λ} contains the image of all embeddings of F into $\overline{\mathbb{Q}}_{\ell}$.

We first suppose that E_{λ} is *F*-inclusive, and put $\Sigma_w := \operatorname{Hom}_{\mathbb{Z}_{\ell}}(O_{F_w}, \mathscr{O})$. Following [CHT08], we use a covariant version of the Fontaine–Laffaille theory [FL82]. Let $\mathscr{MF}_{\mathscr{O},w}$ be the category of $O_{F_w} \otimes_{\mathbb{Z}_{\ell}} \mathscr{O}$ -modules *M* of finite length equipped with

- O a decreasing filtration $\{\operatorname{Fil}^{i} M\}_{i \in \mathbb{Z}}$ by $O_{F_{w}} \otimes_{\mathbb{Z}_{\ell}} \mathscr{O}$ -submodules that are $O_{F_{w}}$ -direct summands, satisfying $\operatorname{Fil}^{0} M = M$ and $\operatorname{Fil}^{\ell-1} M = 0$, and
- O a Frobenius structure, that is, $\sigma \otimes 1$ -linear maps Φ^i : Fil^{*i*} $M \to M$ for $i \in \mathbb{Z}$, satisfying the relations $\Phi^i|_{\operatorname{Fil}^{i+1}M} = \ell \Phi^{i+1}$ and $\sum_{i \in \mathbb{Z}} \Phi^i \operatorname{Fil}^i M = M$.

Let $\mathscr{MF}_{k,w}$ be the full subcategory of $\mathscr{MF}_{\mathscr{O},w}$ of objects that are annihilated by λ .

For an object M of $\mathcal{MF}_{\mathcal{O},w}$, there is canonical decomposition

$$M = \bigoplus_{\tau \in \Sigma_w} M_{\tau},$$

where $M_{\tau} := M \otimes_{O_{F_w} \otimes_{\mathbb{Z}_\ell} \mathscr{O}, \tau \otimes 1} \mathscr{O}$. Then we have $\operatorname{Fil}^i M = \bigoplus_{\tau \in \Sigma_w} \operatorname{Fil}^i M_{\tau}$ with $\operatorname{Fil}^i M_{\tau} = M_{\tau} \cap \operatorname{Fil}^i M$, and that Φ^i induces \mathscr{O} -linear maps

$$\Phi^i_{\tau} \colon \operatorname{Fil}^i M_{\tau} \to M_{\tau \circ \sigma^{-1}}.$$

We put

$$\operatorname{gr}^{i}M_{\tau} \coloneqq \operatorname{Fil}^{i}M_{\tau}/\operatorname{Fil}^{i+1}M_{\tau}, \quad \operatorname{gr}^{\bullet}M_{\tau} \coloneqq \bigoplus_{i} \operatorname{gr}^{i}M_{\tau}, \quad \operatorname{gr}^{\bullet}M \coloneqq \bigoplus_{\tau \in \Sigma_{w}} \operatorname{gr}^{\bullet}M_{\tau}.$$

We define the set of τ -Fontaine-Laffaille weights of M to be

$$\operatorname{HT}_{\tau}(M) \coloneqq \{ i \in \mathbb{Z} \mid \operatorname{gr}^{i} M_{\tau} \neq 0 \}.$$

We say that M has regular Fontaine-Laffaille weights if $\operatorname{gr}^{i} M_{\tau}$ is generated over \mathscr{O} by at most one element for every $\tau \in \Sigma_{w}$ and every $i \in \mathbb{Z}$.

For every integer *a* satisfying $0 \le a \le \ell - 2$, let $\mathscr{MF}^{[0,a]}_{\mathscr{O},w}$ be the full subcategory of $\mathscr{MF}_{\mathscr{O},w}$ consisting of objects *M* satisfying Fil^{*a*+1} *M* = 0. In particular, we have $\mathscr{MF}^{[0,\ell-2]}_{\mathscr{O},w} = \mathscr{MF}_{\mathscr{O},w}$ by definition. There is a duality functor $D^{[0,a]}$ on the category $\mathscr{MF}^{[0,a]}_{\mathscr{O},w}$ such that for every object *M* of $\mathscr{MF}^{[0,a]}_{\mathscr{O},w}$, the object $D^{[0,a]}(M)$ is defined as follows:

- O the underlying $O_{F_w} \otimes_{\mathbb{Z}_\ell} \mathscr{O}$ -module of $D^{[0,a]}(M)$ is $\operatorname{Hom}_{O_{F_w}}(M, F_w/O_{F_w})$;
- O Fil^{*i*} D^[0,a](M) = Hom_{O_{Fw}}(M/Fil^{*a*+1-*i*} M, F_w/O_{Fw});
- **O** for $f \in \operatorname{Fil}^i \operatorname{D}^{[0,a]}(M)$ and $m \in \operatorname{Fil}^j M$, we have

$$\Phi^{i}(f)(\Phi^{j}(m)) = \begin{cases} \ell^{a-i-j} f(m)^{\sigma} & \text{if } i+j \leq a; \\ 0 & \text{if } i+j > a. \end{cases}$$

It is easy to see that $D^{[0,a]}(M)$ is a well-defined object of $\mathscr{MF}_{\mathscr{O},w}$ (see [CHT08, Page 34]), and that $D^{[0,a]}(D^{[0,a]}(M)) = M$.

Let $\mathscr{O}[\Gamma_{F_w}]^{f.i.}$ be the category of \mathscr{O} -modules of finite length equipped with a continuous action of Γ_{F_w} . In [CHT08, 2.4.1], the authors defined an exact fully faithful, covariant \mathscr{O} -linear functor

$$\mathbf{G}_w \colon \mathscr{MF}_{\mathscr{O},w} \to \mathscr{O}[\Gamma_{F_w}]^{\mathrm{f.l.}}$$

whose essential image is closed under taking sub-objects and quotient objects.³¹ The length of an object M in $\mathscr{MF}_{\mathcal{O},w}$ as an \mathscr{O} -module equals $[F_w : \mathbb{Q}_\ell]$ times the length of $\mathbf{G}_w(M)$ as an \mathscr{O} -module. If M belongs to $\mathscr{MF}_{\mathcal{O},w}^{[0,a]}$, then we have

$$\mathbf{G}_w(\mathbf{D}^{[0,a]}(M)) = \mathbf{G}_w(M)^{\vee}(-a),$$

where $\mathbf{G}_w(M)^{\vee} \coloneqq \operatorname{Hom}_{\mathscr{O}}(\mathbf{G}_w(M), E_{\lambda}/\mathscr{O})$ with the dual Galois action. For two objects M_1, M_2 of $\mathscr{MF}_{\mathscr{O},w}$, we have a canonical isomorphism

$$\operatorname{Hom}_{\mathscr{MF}_{\mathscr{O},w}}(M_1, M_2) \xrightarrow{\sim} \operatorname{H}^0(F_w, \operatorname{Hom}_{\mathscr{O}}(\mathbf{G}_w(M_1), \mathbf{G}_w(M_2)))$$

and a canonical injective map

$$\operatorname{Ext}^{1}_{\mathscr{MF}_{\mathscr{O},w}}(M_{1},M_{2}) \hookrightarrow \operatorname{Ext}^{1}_{\mathscr{O}[\Gamma_{F_{w}}]^{\mathrm{f.l.}}}(\mathbf{G}_{w}(M_{1}),\mathbf{G}_{w}(M_{2})),$$

where the target is canonically isomorphic to $\mathrm{H}^1(F_w, \mathrm{Hom}_k(\mathbf{G}_w(M_1), \mathbf{G}_w(M_2)))$ if M_1, M_2 are both objects of $\mathscr{MF}_{k,w}$.

Example E.3.2. For an integer *a* satisfying $0 \le a \le \ell - 2$ and an object *R* of $\mathscr{C}^{f}_{\mathscr{O}}$, we have an object $R\{a\}$ of $\mathscr{MF}_{\mathscr{O},w}$ defined as follows: the underlying $O_{F_{w}} \otimes_{\mathbb{Z}_{\ell}} \mathscr{O}$ -module is simply $((F_{w}/O_{F_{w}}) \otimes R)e_{a}$, with the filtration given by

$$\operatorname{Fil}^{i} R\{a\} = \begin{cases} ((F_{w}/O_{F_{w}}) \otimes R)e_{a} & \text{if } i \leq a; \\ 0 & \text{if } i > a. \end{cases}$$

Finally, the Frobenius structure is determined by $\Phi^a(e_a) = e_a$. Then we have

$$\mathbf{G}_w(R\{a\}) \simeq R(-a)$$

as $\mathscr{O}[\Gamma_{F_w}]$ -modules, where (-a) denotes the Tate twist, and $D^{[0,b]}(R\{a\}) \simeq R\{b-a\}$ for every integer b satisfying $a \leq b \leq \ell - 2$.

Construction E.3.3. We construct a functor $-^{\sigma} : \mathscr{MF}_{\mathcal{O},w} \to \mathscr{MF}_{\mathcal{O},w}$ as follows: for an object M of $\mathscr{MF}_{\mathcal{O},w}$, the underlying $O_{F_w} \otimes_{\mathbb{Z}_{\ell}} \mathcal{O}$ -module of M^{σ} is $O_{F_w} \otimes_{O_{F_w},\sigma} M$ with the induced filtration and Frobenius structure. Then we have $M_{\tau}^{\sigma} = M_{\tau \circ \sigma^{-1}}$ for every $\tau \in \Sigma_w$, and that $\mathbf{G}_w(M^{\sigma})$ is isomorphic to $\mathbf{G}_w(M)$ but with the action of Γ_{F_w} twisted by the absolute Frobenius of F_w : if we denote by ρ and ρ_{σ} the actions of Γ_{F_w} on $\mathbf{G}_w(M)$ and $\mathbf{G}_w(M^{\sigma})$, respectively, then they satisfy

$$\rho_{\sigma}(g) = \rho(\tilde{\sigma}^{-1}g\tilde{\sigma})$$

where $\tilde{\sigma} \in \operatorname{Gal}(\overline{F}_v^+/\mathbb{Q}_\ell)$ is a lift of the absolute Frobenius.

We now let $-c: \mathscr{MF}_{\mathcal{O},w} \to \mathscr{MF}_{\mathcal{O},w}$ be the $[F_v^+ : \mathbb{Q}_\ell]$ -th iteration of the functor -c constructed above. Take an object M of $\mathscr{MF}_{\mathcal{O},w}$. Suppose that M is finite free over $O_{F_w} \otimes R$ for some object R of $\mathscr{C}_{\mathcal{O}}^f$. Then giving an isomorphism $M \simeq D^{[0,a]}(M^c)$ is equivalent to giving a perfect pairing

$$\langle , \rangle \colon M^{\mathsf{c}} \times M \to R\{a\}$$

in the category $\mathcal{MF}_{\mathcal{O},w}$, where $R\{a\}$ is the object in Example E.3.2. The latter is equivalent to giving, for each $\tau \in \Sigma_w$, an *R*-bilinear perfect pairing $\langle , \rangle_{\tau} \colon M_{\tau^c} \times M_{\tau} \to (\mathbb{Q}_{\ell}/\mathbb{Z}_{\ell}) \otimes R$ satisfying that

- (1) for every $i, j \in \mathbb{Z}$ and every $x \in \operatorname{Fil}^{i} M_{\tau}$ and $y \in \operatorname{Fil}^{j} M_{\tau}$, $\langle \Phi_{\tau^{c}}^{i} x, \Phi_{\tau}^{j} y \rangle_{\tau}$ equals $\ell^{a-i-j} \langle x, y \rangle_{\tau}$ (resp. 0) if $i + j \leq a$ (resp. i + j > a); and
- (2) for every $i \in \mathbb{Z}$, the annihilator of $\operatorname{Fil}^{i} M_{\tau}$ under \langle , \rangle_{τ} is $\operatorname{Fil}^{a+1-i} M_{\tau^{c}}$; in particular, \langle , \rangle_{τ} induces an *R*-linear isomorphism $\operatorname{gr}^{i} M_{\tau} \simeq \operatorname{Hom}_{R}(\operatorname{gr}^{a-i} M_{\tau^{c}}, (\mathbb{Q}_{\ell}/\mathbb{Z}_{\ell}) \otimes R).$

³¹In fact, if \mathbf{U}_S is the contravariant functor from $\mathscr{MF}_{\mathscr{O},w}$ to $\mathscr{O}[\Gamma_{F_w}]^{\mathrm{f.l.}}$ defined in [FL82], then we have $\mathbf{G}_w(M) \simeq \mathbf{U}_S(\mathrm{D}^{[0,\ell-2]}(M))(2-\ell)$.

Definition E.3.4. Let R be an object of $\mathscr{C}^f_{\mathscr{O}}$, and $\rho: \Gamma_{F_w} \to \operatorname{GL}_n(R)$ a continuous representation. We say ρ is crystalline with Fontaine–Laffaille weights in [0, a] for some $0 \leq a \leq \ell - 2$ if (R^n, ρ) lies in the essential image of the functor $\mathbf{G}_w: \mathscr{MF}^{[0,a]}_{\mathscr{O},w} \to \mathcal{O}[\Gamma_{F_w}]^{\mathrm{f.l.}}$; in this case, we say that ρ has regular Fontaine–Laffaille weights if so does $\mathbf{G}^{-1}_w(\rho)$.

Now we consider a pair (\bar{r}, χ) from Notation E.2.1 with $\tilde{\Gamma} = \Gamma_{F_v^+}$ and $\Gamma = \Gamma_{F_v^+} \cap \Gamma_F = \Gamma_{F_w}$. We do not assume that E_{λ} is *F*-inclusive. We choose an *F*-inclusive finite unramified extension E'_{λ} of E_{λ} , with the ring of integers \mathscr{O}' and the residue field k'.

Assumption E.3.5. There exist an integer a satisfying $0 \le a \le \ell - 2$ and an element $b \in \mathbb{Z}/2\mathbb{Z}$ such that

- (1) $\chi = \eta_v^b \epsilon_{\ell,v}^{-a}$; and
- (2) $\bar{r}^{\ddagger} \otimes_k k'$ is crystalline with regular Fontaine–Laffaille weights in [0, a].

Definition E.3.6. Under Assumption E.3.5, we define $\mathscr{D}^{\mathrm{FL}}$ to be the local deformation problem of \bar{r} that classifies the liftings $r: \Gamma_{F_v^+} \to \mathscr{G}_N(R)$ of \bar{r} to objects R of $\mathscr{C}_{\mathscr{O}}$ such that for every Artinian quotient R' of $R \otimes_{\mathscr{O}} \mathscr{O}'$, $r^{\natural} \otimes_R R'$ is crystalline with Fontaine–Laffaille weights in [0, a] (Definition E.3.4).

Remark E.3.7. It is straightforward to check that Assumption E.3.5 and Definition E.3.6 do not depend on the choice of E'_{λ} .

Lemma E.3.8. Suppose $\ell > N$ and assume Assumption E.3.5. We have

$$\dim_k \mathcal{L}(\mathscr{D}^{\mathrm{FL}}) - \dim_k \mathcal{H}^0(F_v^+, \operatorname{ad} \bar{r}) = [F_v^+ : \mathbb{Q}_\ell] \cdot \frac{N(N-1)}{2}$$

Proof. We may assume $E_{\lambda} = E'_{\lambda}$.

Suppose first that v is split in F. Then we have $F_w = F_v^+$, and that a lifting r in $\mathscr{D}^{\mathrm{FL}}(R)$ of \bar{r} is of the form $r = (\rho, \epsilon_{\ell,v}^{-a}) \colon \Gamma_{F_w} \to \mathrm{GL}_N(R) \times R^{\times}$ such that for every Artinian quotient R' of R, $\rho \otimes_R R'$ lies in the essential image of the functor \mathbf{G}_w . Then the lemma is exactly [CHT08, Corollary 2.4.3].

Suppose now that v is inert in F; and denote by $\Gamma_{w/v}$ the Galois group of the quadratic extension F_w/F_v^+ . Then the restriction map induces an isomorphism

$$\mathrm{H}^{1}(F_{v}^{+}, \mathrm{ad}\,\bar{r}) \xrightarrow{\sim} \mathrm{H}^{1}(F_{w}, \mathrm{ad}\,\bar{r})^{\Gamma_{w/v}}.$$

Put $M := \mathbf{G}_w^{-1}(\bar{r}^{\natural})$. Then the deformations of \bar{r} to $k[\varepsilon]/(\varepsilon^2)$ that lie in the essential image of \mathbf{G}_w are classified by $\operatorname{Ext}^1_{\mathscr{MF}_{k,w}}(M, M)$, which is canonically a $\Gamma_{w/v}$ -stable subspace of $\mathrm{H}^1(F_w, \operatorname{ad} \bar{r})$. Therefore, we have

$$\mathcal{L}(\mathscr{D}^{\mathrm{FL}}) = \mathrm{Ext}^{1}_{\mathscr{MF}_{k,w}}(M,M) \cap \mathrm{H}^{1}(F_{w}, \mathrm{ad}\,\bar{r})^{\Gamma_{w/v}} = \mathrm{Ext}^{1}_{\mathscr{MF}_{k,w}}(M,M)^{\Gamma_{w/v}}.$$

In fact, the induced action of $\Gamma_{w/v}$ on $\operatorname{Ext}^{1}_{\mathscr{MF}_{k,w}}(M, M)$ can be described as follows. Recall the functor $-^{\mathsf{c}}$ in Construction E.3.3. Then $\mathbf{G}_{w}(M^{\mathsf{c}})$ is isomorphic to $\bar{r}^{\sharp,\mathsf{c}}|_{\Gamma_{F_{w}}}$. Since $\bar{r}^{\sharp,\mathsf{c}}$ and $\bar{r}^{\sharp,\vee}(-a)$ are conjugate, we have $M \simeq \mathrm{D}^{[0,a]}(M^{\mathsf{c}})$. We fix such an isomorphism $M \simeq \mathrm{D}^{[0,a]}(M^{\mathsf{c}})$ hence obtain a pairing \langle , \rangle (with R = k) as in Construction E.3.3. Then for an element $[E] \in \operatorname{Ext}^{1}_{\mathscr{MF}_{k,w}}(M, M)$ represented by an extension $0 \to M \to E \to M \to 0$, the image of [E] under the action of the (unique) non-trivial element in $\Gamma_{w/v}$ is obtained by applying the functor $\mathrm{D}^{[0,a]}(-^{\mathsf{c}})$ to $0 \to M \to E \to M \to 0$.

To compute $\operatorname{Ext}^{1}_{\mathscr{MF}_{k,w}}(M, M)^{\Gamma_{w/v}}$, we recall first the following long exact sequence in [CHT08, Lemma 2.4.2]:

(E.3)

$$0 \longrightarrow \operatorname{End}_{\mathscr{MF}_{k,w}}(M) \longrightarrow \operatorname{Fil}^{0} \operatorname{Hom}_{O_{F_{w}} \otimes_{\mathbb{Z}_{\ell}} \mathscr{O}}(M, M) \xrightarrow{\alpha} \operatorname{Hom}_{O_{F_{w}} \otimes_{\mathbb{Z}_{\ell}} \mathscr{O}, \sigma \otimes 1}(\operatorname{gr}^{\bullet} M, M) \xrightarrow{\beta} \operatorname{Ext}^{1}_{\mathscr{MF}_{k,w}}(M, M) \longrightarrow 0,$$

where

- O Fil⁰ Hom_{$O_{F_w \otimes_{\mathbb{Z}_\ell} \mathscr{O}}(M, M)$ denotes the $O_{F_w} \otimes_{\mathbb{Z}_\ell} \mathscr{O}$ -submodule of Hom_{$O_{F_w \otimes_{\mathbb{Z}_\ell} \mathscr{O}}(M, M)$ of endomorphisms that preserve the filtration;}}
- O the map α takes an element $f \in \operatorname{Fil}^0 \operatorname{Hom}_{O_{F_w} \otimes_{\mathbb{Z}_\ell} \mathscr{O}}(M, M)$ to $(f\Phi^i \Phi^i f)_{i \in \mathbb{Z}}$; and
- O the map β is defined as follows: if $\varphi = (\varphi^i)_{i \in \mathbb{Z}}$ is a $\sigma \otimes 1$ -linear map from $\operatorname{gr}^{\bullet} M$ to M, then $\beta(\varphi)$ is given by the extension class of $E = M \oplus M$ with the filtration $\operatorname{Fil}^i E = \operatorname{Fil}^i M \oplus \operatorname{Fil}^i M$ and the Frobenius structure

$$\Phi^i_E \coloneqq \begin{pmatrix} \Phi^i & \varphi^i \\ 0 & \Phi^i \end{pmatrix}.$$

To prove the lemma, we need to derive an analogous long exact sequence similar to (E.3) but with the last term $\operatorname{Ext}^{1}_{\mathscr{MF}_{k,w}}(M, M)^{\Gamma_{w/v}}$. For the first term, note that we have a canonical isomorphism $\operatorname{End}_{\mathscr{MF}_{k,w}}(M) \simeq \operatorname{H}^{0}(F_{w}, \operatorname{ad} \bar{r})$, which contains $\operatorname{H}^{0}(F_{v}^{+}, \operatorname{ad} \bar{r})$ as a submodule. For the second term, let $\operatorname{Fil}^{0}\operatorname{Hom}_{O_{F_{w}}\otimes_{\mathbb{Z}_{\ell}}\mathscr{O}}(M, M)^{+}$ be the submodule of $\operatorname{Fil}^{0}\operatorname{Hom}_{O_{F_{w}}\otimes_{\mathbb{Z}_{\ell}}\mathscr{O}}(M, M)$ consisting of elements $f = (f_{\tau})_{\tau \in \Sigma_{w}}$ such that $-f_{\tau^{c}}$ is the adjoint of f_{τ} under the pairing \langle , \rangle_{τ} for every $\tau \in \Sigma_{w}$. For the third term, let $\operatorname{Hom}_{O_{F_{w}}\otimes_{\mathbb{Z}_{\ell}}\mathscr{O},\sigma\otimes 1}(\operatorname{gr}^{\bullet}M, M)^{+}$ denote by the submodule of $\operatorname{Hom}_{O_{F_{w}}\otimes_{\mathbb{Z}_{\ell}}\mathscr{O},\sigma\otimes 1}(\operatorname{gr}^{\bullet}M, M)$ consisting of $\varphi = (\varphi^{i})_{i\in\mathbb{Z}}$ such that

(E.4)
$$\langle \Phi^i_{\tau^{\mathsf{c}}}(x), \varphi^{a-i}_{\tau}(y) \rangle_{\tau} + \langle \varphi^i_{\tau^{\mathsf{c}}}(x), \Phi^{a-i}_{\tau}(y) \rangle_{\tau} = 0$$

is satisfies for every $x \in \operatorname{gr}^{i} M_{\tau^{c}}$ and $y \in \operatorname{gr}^{a-i} M_{\tau}$.

Then (E.3) induces an exact sequence

$$0 \longrightarrow \mathrm{H}^{0}(F_{v}^{+}, \mathrm{ad}\,\bar{r}) \longrightarrow \mathrm{Fil}^{0} \operatorname{Hom}_{O_{F_{w}} \otimes_{\mathbb{Z}_{\ell}} \mathscr{O}}(M, M)^{+} \xrightarrow{\alpha} \operatorname{Hom}_{O_{F_{w}} \otimes_{\mathbb{Z}_{\ell}} \mathscr{O}, \sigma \otimes 1}(\mathrm{gr}^{\bullet}M, M)^{+} \xrightarrow{\beta} \operatorname{Ext}^{1}_{\mathscr{M}_{F_{k,w}}}(M, M)^{\Gamma_{w/v}} \longrightarrow 0$$

of k-vector spaces. We now compute the dimension of the middle two terms. From the description of $\operatorname{Fil}^0 \operatorname{Hom}_{O_{F_w} \otimes_{\mathbb{Z}_\ell} \mathscr{O}}(M, M)^+$, it is clear that f_{τ^c} is determined by f_{τ} for every $\tau \in \Sigma_w$. On the other hand, for each fixed τ , all the possible choices of f_{τ} form a k-vector space of dimension $\frac{N(N+1)}{2}$. Thus, we have

$$\dim_k \operatorname{Fil}^0 \operatorname{Hom}_{O_{F_w} \otimes_{\mathbb{Z}_\ell} \mathscr{O}}(M, M)^+ = [F_v^+ : \mathbb{Q}_\ell] \cdot \frac{N(N+1)}{2}.$$

For $\operatorname{Hom}_{O_{F_w}\otimes_{\mathbb{Z}_\ell}\mathcal{O},\sigma\otimes 1}(\operatorname{gr}^{\bullet} M, M)^+$, we first note that the map

$$\bigoplus_i \Phi^i_\tau \colon \operatorname{gr}^{\bullet} M_\tau \to M_{\tau \circ \sigma^{-1}}$$

is an isomorphism for every $\tau \in \Sigma_w$. It follows from (E.4) that $\varphi_{\tau^c} \coloneqq \bigoplus_i \varphi_{\tau^c}^i$ is determined by $\varphi_{\tau} \coloneqq \bigoplus_i \varphi_{\tau^c}^i$. On the other hand, for each fixed τ , all the possible choices of $\varphi_{\tau} \colon \operatorname{gr}^{\bullet} M_{\tau} \to M_{\tau \circ \sigma^{-1}}$ form a k-vector space of dimension N^2 . Thus, we have

$$\dim_k \operatorname{Hom}_{O_{F_w} \otimes_{\mathbb{Z}_\ell} \mathscr{O}, \sigma \otimes 1}(\operatorname{gr}^{\bullet} M, M)^+ = [F_v^+ : \mathbb{Q}_\ell] \cdot N^2.$$

The Lemma follows immediately.

Proposition E.3.9. Suppose $\ell > N$ and assume Assumption E.3.5. The local deformation problem \mathscr{D}^{FL} is formally smooth over $\text{Spf } \mathscr{O}$ of pure relative dimension $N^2 + [F_v^+ : \mathbb{Q}_\ell] \cdot \frac{N(N-1)}{2}$.

Proof. By Lemma E.3.8, it suffices to show that \mathscr{D}^{FL} is formally smooth over \mathscr{O} . We may again assume $E_{\lambda} = E'_{\lambda}$. When v is split in F/F^+ , the proposition has been proved in [CHT08, Lemma 2.4.1].

Now we suppose that v is inert in F. Fix a subset $\Sigma_w^+ \subset \Sigma_w$ so that $\Sigma_w = \Sigma_w^+ \coprod \Sigma_w^{+,c}$. Let R be an object of $\mathscr{C}_{\mathscr{O}}^f$ and $I \subset R$ an ideal satisfying $\mathfrak{m}_R I = (0)$. Let r be a lifting of \bar{r} to R/I, and put

 $M := \mathbf{G}_w^{-1}(r^{\natural})$. Note that M is an object of $\mathscr{M}\mathscr{P}_{\mathscr{O},w}^{[0,a]}$, and that for every $\tau \in \Sigma_v$, M_{τ} is free of rank N over R/I and $\operatorname{Fil}^i M_{\tau}$ is a R/I-direct summand of M_{τ} .

Recall the functor $-^{\mathfrak{c}}$ in Construction E.3.3. Then $\mathbf{G}_w(M^{\mathfrak{c}})$ is isomorphic to $\bar{r}^{\sharp,\mathfrak{c}}|_{\Gamma_{F_w}}$. Since $\bar{r}^{\sharp,\mathfrak{c}}$ and $\bar{r}^{\sharp,\vee}(-a)$ are conjugate, we have $M \simeq \mathrm{D}^{[0,a]}(M^{\mathfrak{c}})$. We fix such an isomorphism $M \simeq \mathrm{D}^{[0,a]}(M^{\mathfrak{c}})$ hence obtain a pairing \langle , \rangle (with R = R/I) as in Construction E.3.3. In view of Assumption E.3.5(2), let $m_{\tau,1} < \cdots < m_{\tau,N}$ be the τ -Hodge–Tate weights of M for every $\tau \in \Sigma_w$. Then there exists a basis $e_{\tau,1}, \ldots, e_{\tau,N}$ of M_{τ} over R/I satisfying $\mathrm{Fil}^{m_{\tau,N+1-i}} M_{\tau} = \bigoplus_{j=1}^{i} (R/I) e_{\tau,j}$ for every $1 \leq i \leq N$. By duality, we have $m_{\tau^c,i} + m_{\tau,N+1-i} = a$. Then we may choose the basis $(e_{\tau,i})$ such that $\langle e_{\tau^c,i}, e_{\tau,j} \rangle_{\tau} = \delta_{i,N+1-j}$ for every $\tau \in \Sigma_w^+$ and every $1 \leq i, j \leq N$.

We now define an object $\widehat{M} = \bigoplus_{\tau \in \Sigma_w} \widehat{M_\tau}$ of $\mathscr{MF}_{\ell,w}$ that reduces to M, whose underlying $O_{F_w} \otimes \mathscr{O}$ -module is free over $O_{F_w} \otimes R$, and an isomorphism $\widetilde{M} \simeq D^{[0,a]}(\widetilde{M}^c)$ that reduces to the previous isomorphism $M \simeq D^{[0,a]}(M^c)$, as follows. As an R-module, we take $\widetilde{M}_\tau = R^{\oplus N}$ with the basis $(\tilde{e}_{\tau,i})$ that lifts the basis $(\tilde{e}_{\tau,i})$ of M_τ . We lift $\langle \ , \ \rangle_\tau$ to an R-bilinear perfect pairing $\widetilde{M}_{\tau^c} \times \widetilde{M}_\tau \to R$ such that $\langle \tilde{e}_{\tau^c,i}, \tilde{e}_{\tau,j} \rangle_\tau = \delta_{i,N+1-j}$ still holds for every $\tau \in \Sigma_w^+$ and every $1 \leq i, j \leq N$. For the filtration, we put $\operatorname{Fil}^m \widetilde{M}_\tau := \bigoplus_{j=1}^i R\widetilde{e}_{\tau,j}$ for m satisfying $m_{\tau,N-i} < m \leq m_{\tau,N+1-i}$. Then $\widetilde{M} \otimes_R R/I$ is isomorphic to M as filtered $O_{F_w} \otimes R/I$ -modules; and the condition (2) in Construction E.3.3 holds for \widetilde{M} as well. For the Frobenius structure on \widetilde{M} , we first define maps $\widetilde{\Phi}_\tau^{m_{\tau,i}}$: $\operatorname{Fil}^{m_{\tau,i}} \widetilde{M}_\tau \to \widetilde{M}_{\tau o \sigma^{-1}}$ for $\tau \in \Sigma_w^+$ by the recursive induction on i. For i = N, we take $\widetilde{\Phi}_\tau^{m_{\tau,i}}$ to be an arbitrary lift of $\Phi_\tau^{m_{\tau,i}} : \operatorname{Fil}^{m_{\tau,N}} M_\tau \to M_{\tau o \sigma^{-1}}$ for $\tau \in \Sigma_w^+$. For $i \leq N - 1$, we take $\widetilde{\Phi}_\tau^{m_{\tau,i}}$ to be a lift of $\Phi_\tau^{m_{\tau,i}} : \operatorname{Fil}^{m_{\tau,i}} M_\tau \to M_{\tau o \sigma^{-1}}$ that restricts to $\ell^{m_{\tau,i-1}m_{\tau,i+1}} \widetilde{\Phi}_\tau^{m_{\tau,i+1}}$ on $\operatorname{Fil}^{m_{\tau,i+1}} \widetilde{M}_\tau$. By Nakayama's lemma, we have

$$\widetilde{M}_{\tau \circ \sigma^{-1}} = \sum_{i} \widetilde{\Phi}_{\tau}^{m_{\tau,i}} (\operatorname{Fil}^{m_{\tau,i}} \widetilde{M}_{\tau})$$

for every $\tau \in \Sigma_w^+$. Finally, we define $\widetilde{\Phi}_{\tau^{\mathfrak{c}}}^i \colon \widetilde{M}_{\tau^{\mathfrak{c}}} \to \widetilde{M}_{\tau^{\mathfrak{c}} \circ \sigma^{-1}}$ for $\tau \in \Sigma_w^+$ to be the unique *R*-linear map satisfying the condition (1) in Construction E.3.3 for \widetilde{M} . This finishes the construction of \widetilde{M} and the isomorphism $\widetilde{M} \simeq \mathrm{D}^{[0,a]}(\widetilde{M}^{\mathfrak{c}})$, which together give rise to a lifting \widetilde{r} of \overline{r} to *R* that reduces to *r*. Thus, $\mathscr{D}^{\mathrm{FL}}$ is formally smooth over \mathscr{O} .

The proposition is proved.

E.4. **Representations of the tame group.** In this subsection, we will study conjugate self-dual representations of the tame group, and define the notion of minimally ramified deformations of such representations.

Definition E.4.1. Let $q \ge 1$ be a positive integer coprime to ℓ . We define the *q*-tame group, denoted by T_q , to be the semidirect product topological group $t^{\mathbb{Z}_\ell} \rtimes \phi_q^{\widehat{\mathbb{Z}}}$ where ϕ_q maps t to t^q , that is, $\phi_q t \phi_q^{-1} = t^q$. For every integer $b \ge 1$, We identify T_{q^d} as a subgroup of T_q topologically generated by t and $\phi_{q^b} = \phi_q^b$.

We consider a reductive group G over \mathscr{O} , together with a surjective homomorphism $\nu: G \to H$ over \mathscr{O} , where H is an algebraic group over \mathscr{O} of multiplicative type. Consider a pair $(\bar{\varrho}, \mu)$ in which $\bar{\varrho}: T_q \to G(k)$ and $\mu: T_q \to H(\mathscr{O})$ are continuous homomorphisms satisfying $\nu \circ \bar{\varrho} = \bar{\mu}$ and $\mu(t) = 1$. Similar to the case in Subsection E.2, let $\mathsf{R}^{\mathrm{loc}}_{\varrho}$ be the \mathscr{O} -algebra in $\mathscr{C}_{\mathscr{O}}$ that parameterizes liftings ϱ of $\bar{\varrho}$ satisfying $\nu \circ \varrho = \mu$.³² The following proposition generalizes the tame case of [Sho18, Theorem 2.5].

³²Here, once again we omit the similitude character μ in the ring $\mathsf{R}_{\varrho}^{\mathrm{loc}}$, in order to be consistent with the previous convention.

Proposition E.4.2. The ring $\mathsf{R}^{\mathrm{loc}}_{\bar{\varrho}}$ is a local complete intersection, flat and of pure relative dimension d over \mathscr{O} , where d is the relative dimension of the kernel of ν over \mathscr{O} .

Proof. We follow the same line as in the proof of [Sho18, Theorem 2.5]. Let G_0 and G_1 be the fibers at 1 and $\mu(\phi_q)$ of the homomorphism ν , respectively. Define the subscheme $\mathscr{M}(G,q)$ of $G_0 \times_{\text{Spec }} \mathscr{O}G_1$ such that for every object R of $\mathscr{C}_{\mathscr{O}}$, $\mathscr{M}(G,q)(R)$ consists of pairs $(\mathbf{A}, \mathbf{B}) \in G_0(R) \times G_1(R)$ satisfying

$$(E.5) BAB^{-1} = A^q$$

It suffices to show that $\mathscr{M}(G,q)$ is a local complete intersection, flat and of pure relative dimension d over \mathscr{O} , since $\mathsf{R}^{\mathrm{loc}}_{\bar{\varrho}}$ is the completion of $\mathscr{M}(G,q)$ at the k-point $(\bar{\varrho}(t), \bar{\varrho}(\phi_q))$.

First, we show that every geometric fiber of $\mathscr{M}(G,q) \to \operatorname{Spec} \mathscr{O}$ is of pure dimension d. Consider the natural projection

$$p: \mathscr{M}(G,q) \to G_0$$

to the first factor. Take a geometric point Spec $K \to \text{Spec } \mathcal{O}$. For a point $\mathbf{A}_0 \in G_0(K)$ in the image of p(K), let $Z(\mathbf{A}_0)$ be the centralizer of \mathbf{A}_0 in $G_{0,K}$ as a closed subscheme of $G_{0,K}$, and $C(\mathbf{A}_0)$ the conjugacy class of \mathbf{A}_0 , which is a locally closed subscheme of $G_{0,K}$ isomorphic to $G_{0,K}/Z(\mathbf{A}_0)$. Then $C(\mathbf{A}_0)$ lies in the image of p_K . For every point $\mathbf{A} \in C(\mathbf{A}_0)(K)$, the fiber $p_K^{-1}(\mathbf{A})$ is a torsor under the group $Z(\mathbf{A})$, which is conjugate to $Z(\mathbf{A}_0)$. Thus, $p_K^{-1}(C(\mathbf{A}_0))$ is irreducible of dimension

$$\dim p_K^{-1}(C(\boldsymbol{A}_0)) = \dim C(\boldsymbol{A}_0) + \dim Z(\boldsymbol{A}_0) = \dim G_{0,K} = d.$$

To continue, we choose an embedding $e: G_K \to \operatorname{GL}_{m,K}$ of algebraic groups over K for some integer $m \geq 1$. By (E.5), the image of $e(K) \circ p(K)$ consists only of matrices whose generalized eigenvalues are $(q^{m!} - 1)$ -th roots of unity, hence finitely many conjugacy classes in $\operatorname{GL}_m(K)$. We claim that the image of p(K) consists of finitely many conjugacy classes in $G_0(K)$ as well, which implies that $\mathscr{M}(G,q)_K$ is of pure dimension d. In fact, we have the following commutative diagram

of sets, in which the bottom map is finite since the morphism $G_{0,K}/\!\!/G_{0,K} \to \operatorname{GL}_{m,K}/\!\!/\operatorname{GL}_{m,K}$ is; and the left map is also finite due to the finiteness of unipotent conjugacy classes [Lus76]; it follows that the upper map is finite as well.

The above discussion shows that the morphism $\mathscr{M}(G,q) \to \operatorname{Spec} \mathscr{O}$ is of pure relative dimension d. Now we take a closed point (\bar{A}, \bar{B}) of $\mathscr{M}(G,q)$, which induces a homomorphism

$$\mathscr{O}_{\mathscr{M}(G,q),(\bar{A},\bar{B})} \to \mathscr{O}_{G_0,\bar{A}} \otimes_{\mathscr{O}} \mathscr{O}_{G_1,\bar{B}}$$

of corresponding complete local rings. As both G_0 and G_1 are smooth over \mathscr{O} of pure relative dimension d, both $\mathscr{O}_{G_0,\bar{A}}$ and $\mathscr{O}_{G_1,\bar{B}}$ are power series rings over \mathscr{O} in d variables. The relation (E.5), or equivalently, the relation $\mathbf{A} = \mathbf{B}^{-1}\mathbf{A}^q\mathbf{B}$, is defined by d equations in $\mathscr{O}_{G_0,\bar{A}} \otimes_{\mathscr{O}} \mathscr{O}_{G_1,\bar{B}}$. In other words, $\mathscr{M}(G,q)$ is a local complete intersection hence Cohen-Macaulay. Therefore, $\mathscr{M}(G,q)$ is flat over \mathscr{O} . The proposition is proved.

Take an integer $n \geq 1$. Now we apply the above discussion to the homomorphism $\nu : \mathscr{G}_n \to \mathrm{GL}_1$ in Notation E.1.1. Consider a pair $(\bar{\varrho}, \mu)$ from Notation E.2.1 with $\tilde{\Gamma} = \mathrm{T}_q$ and $\Gamma = \mathrm{T}_{q^2}$, such that $\mu(t) = 1$. In particular, $\bar{\varrho} : \mathrm{T}_q \to \mathscr{G}_n(k)$ is a homomorphism and $\mu : \mathrm{T}_q \to \mathscr{O}^{\times}$ is a (continuous) similitude character. Write

(E.6)
$$\bar{\varrho}(t) = \bar{\boldsymbol{A}} = (\bar{A}, 1, 1), \quad \bar{\varrho}(\phi_q) = \bar{\boldsymbol{B}} = (\bar{B}, -\mu(\phi_q), \mathfrak{c})$$

for $\bar{A}, \bar{B} \in \operatorname{GL}_n(k)$. For a lifting ρ of $\bar{\rho}$ to an object R of $\mathscr{C}_{\mathcal{O}}$, we write $\rho(t) = \mathbf{A} = (A, 1, 1)$ and $\rho(\phi_q) = \mathbf{B} = (B, -\mu(\phi_q), \mathfrak{c})$. Then the pair (A, B) reduce to (\bar{A}, \bar{B}) , and satisfy the relation (E.7) $B^{\mathsf{t}}A^{-1}B^{-1} = A^q$.

Corollary E.4.3. The ring $\mathsf{R}^{\mathrm{loc}}_{\bar{\varrho}}$ is a local complete intersection, flat and of pure relative dimension n^2 over \mathscr{O} .

Proof. This follows immediately from Proposition E.4.2 since the kernel of $\nu: \mathscr{G}_n \to \operatorname{GL}_1$ is of dimension n^2 .

From now till the end of this subsection, we assume $l \ge n$. Denote by \mathcal{N}_n (resp. \mathcal{U}_n) the closed subscheme of \mathcal{M}_n (resp. GL_n) defined by the equation $X^n = 0$ (resp. $(A - 1)^n = 0$). For every object R of $\mathscr{C}_{\mathcal{O}}$, we have the *truncated exponential map* exp: $\mathcal{N}_n(R) \to \mathcal{U}_n(R)$ defined by the formula

$$\exp X = 1 + X + \dots + \frac{X^{n-1}}{(n-1)!},$$

which is an bijection. Its inverse is given by the truncated logarithm map log: $\mathcal{U}_n(R) \to \mathcal{N}_n(R)$ defined by the formula

$$\log A = \sum_{i=1}^{n-1} (-1)^{i-1} \frac{(A-1)^i}{i}.$$

Let \mathfrak{P}_n be the set of partitions of n. By the classification of nilpotent orbits in GL_n , for $K = k, E_\lambda$, we have canonical surjective maps $\pi \colon \mathcal{N}_n(K) \to \mathfrak{P}_n$ such that the fibers of π are exactly the orbits in $\mathcal{N}_n(K)$ under the conjugate action of $\mathrm{GL}_n(K)$.

By the continuity of $\bar{\varrho}$, we know that \bar{A} in (E.6) is unipotent, which implies $\bar{A} \in \mathcal{U}_n(k)$. Put $\bar{X} \coloneqq \log \bar{A} \in \mathcal{N}_n(k)$. Following [Boo19, Definition 3.9], we define the functor $\operatorname{Nil}_{\bar{X}} : \mathscr{C}_{\mathcal{O}} \to \operatorname{Set}$ that sends an object R of $\mathscr{C}_{\mathcal{O}}$ to the set of elements $X \in \mathcal{N}_n(R)$ that reduce to \bar{X} and are of the form CX_0C^{-1} , where X_0 is an element in $\mathcal{N}_n(\mathcal{O})$ satisfying $\pi(X_0) = \pi(\bar{X})$ and $C \in \operatorname{GL}_n(R)$, where we regard X_0 as an element in $\mathcal{N}_n(E_{\lambda})$ in the notation $\pi(X_0)$.

Definition E.4.4. We say that a lifting ρ of $\overline{\rho}$ to an object R of $\mathscr{C}_{\mathcal{O}}$ is minimally ramified if there exists an element $X \in \operatorname{Nil}_{\overline{X}}(R)$ such that $\rho^{\natural}(t) = \exp X$.

Let $\mathscr{D}_{\bar{\varrho}}^{\min}$ be the local deformation problem of $\bar{\varrho}$ (Definition E.2.3) that classifies minimally ramified liftings of $\bar{\varrho}$.

Proposition E.4.5. The local deformation problem $\mathscr{D}_{\bar{\varrho}}^{\min}$ is formally smooth over Spf \mathscr{O} of pure relative dimension n^2 .

Proof. We follow the approach of [Boo19, Proposition 5.6], where a similar result for symplectic or orthogonal representations was proved.

Consider the morphism $\alpha \colon \mathscr{D}_{\bar{\varrho}}^{\min} \to \operatorname{Nil}_{\bar{X}}$ that sends a lifting ϱ to $\log A$ if $\varrho^{\natural}(t) = A$. In the definition of $\operatorname{Nil}_{\bar{X}}$, we may fix the nilpotent element $X_0 \in \mathcal{N}_n(\mathscr{O})$. Moreover, up to conjugation in $\operatorname{GL}_n(\mathscr{O})$, we may assume

$$X_0 = \begin{pmatrix} J_{n_1} & & \\ & \ddots & \\ & & J_{n_r} \end{pmatrix},$$

where $n = n_1 + \cdots + n_r$, and J_{n_i} is the Jordan block of size n_i -by- n_i as in Subsection 1.3. Let $Z_n(X_0)$ be the centralizer of X_0 in $\operatorname{GL}_{n,\mathscr{O}}$, which is a closed subscheme of $\operatorname{GL}_{n,\mathscr{O}}$. By [Boo19, Remark 4.18], $Z_n(X_0)$ is smooth over \mathscr{O} . By [Boo19, Lemma 3.11], $\operatorname{Nil}_{\bar{X}}$ is represented by a formal power series ring over \mathscr{O} in $n^2 - \dim_{\mathscr{O}} Z_n(X_0)$ variables, where $\dim_{\mathscr{O}} Z_n(X_0)$ denotes the relative dimension

of $Z_n(X_0)$ over \mathscr{O} . Thus, it suffices to show that α is represented by a formal scheme formally smooth of pure relative dimension $\dim_{\mathscr{O}} Z_n(X_0)$ over $\operatorname{Nil}_{\bar{X}}$.

Take a lifting ρ of $\bar{\rho}$ to an object R of $\mathscr{C}_{\mathscr{O}}$. Then $\rho(\phi_q)$ has the form $(B, -\mu(\phi_q), \mathfrak{c})$ with $B \in \operatorname{GL}_n(R)$ that reduces to \bar{B} and satisfies

$$(E.8) B^{t}XB^{-1} = -qX$$

by (E.7). For each given $X \in \operatorname{Nil}_{\bar{X}}$, if there exists $B \in \operatorname{GL}_n(R)$ that reduces to \bar{B} and satisfies (E.8), then the set of all elements B form a torsor under the group

$$Z_n(X)(R) \coloneqq \{g \in 1 + \mathcal{M}_n(\mathfrak{m}_R) \mid gXg^{-1} = X\},\$$

which is isomorphic to the group of R-valued points of the formal completion of the group scheme $Z_n(X_0)$ along the unit section. Thus, to finish the proof, it suffices to show that the equation (E.8) admits at least one solution for B that reduces to \overline{B} .

Assume first $X = X_0$ in $\mathcal{N}_n(R)$. Then

$$B_0 \coloneqq \begin{pmatrix} A_{n_1} & & \\ & \ddots & \\ & & A_{n_r} \end{pmatrix}, \text{ where } A_{n_i} \coloneqq \begin{pmatrix} & & (-q)^{n_i - 1} \\ & \ddots & \\ & -q & \\ 1 & & \end{pmatrix}$$

is a solution to (E.8). In the general case, we write $X = CX_0C^{-1}$ for some $C \in GL_n(R)$. Then $B := CB_0 {}^{\mathrm{t}}C$ satisfies the equation (E.8). Up to multiplying C by an element in $Z_n(X_0)(R)$ from the right, we can make $B \in GL_n(R)$ to reduce to \overline{B} . This finishes the proof of the proposition. \Box

Recall from Definition E.2.4 that $L(\mathscr{D}_{\bar{\varrho}}^{\min}) \subseteq H^1(T_q, \operatorname{ad} \bar{\varrho})$ is tangent space of the local deformation problem $\mathscr{D}_{\bar{\varrho}}^{\min}$.

Corollary E.4.6. We have $\dim_k L(\mathscr{D}_{\bar{\rho}}^{\min}) = \dim_k H^0(T_q, \operatorname{ad} \bar{\varrho}).$

Proof. Suppose $\mathscr{D}_{\bar{\varrho}}^{\min} = \operatorname{Spf} \mathsf{R}_{\bar{\varrho}}^{\min}$. By (E.1), we have

$$\dim_k \mathfrak{m}_{\mathsf{R}^{\min}_{\bar{\varrho}}}/(\lambda,\mathfrak{m}^{2}_{\mathsf{R}^{\min}_{\bar{\varrho}}}) = \dim_k \mathcal{L}(\mathscr{D}^{\min}_{\bar{\varrho}}) + n^2 - \dim_k \mathcal{H}^0(\mathcal{T}_q, \mathrm{ad}\,\bar{\varrho}).$$

From this, the corollary follows immediately from Proposition E.4.5.

To end this subsection, we record the following lemma concerning decomposition of representations of the q-tame group, in which part (1) will be used later and part (2) is only for complement.

Lemma E.4.7. Let $(\bar{\rho}, \bar{M})$ be an unramified representation of $T_q = t^{\mathbb{Z}_\ell} \rtimes \phi_q^{\widehat{\mathbb{Z}}}$ over k of dimension N. Suppose that \bar{M} admits a decomposition

$$\overline{M} = \overline{M}_1 \oplus \dots \oplus \overline{M}_s$$

stable under the action of $\bar{\rho}(\phi_q)$ such that the characteristic polynomials of $\bar{\rho}(\phi_q)$ on M_i are mutually coprime for $1 \leq i \leq s$. Let (ρ, M) be a lifting of $(\bar{\rho}, \bar{M})$ to an object R of $\mathscr{C}_{\mathcal{O}}$. Then we have

(1) There is a unique decomposition

$$M = M_1 \oplus \cdots \oplus M_s$$

of free R-modules, such that M_i is stable under the action of $\rho(\phi_q)$ and it is a lifting of M_i as a ϕ_q -module.

(2) Write $\rho(t) = (\rho(t)_{i,j})$ with $\rho(t)_{i,j} \in \operatorname{Hom}_R(M_j, M_i)$. Suppose that q is not an eigenvalue for the canonical action of ϕ_q on $\operatorname{Hom}_k(\bar{M}_j, \bar{M}_i)$ for all $i \neq j$. Then we have $\rho(t)_{i,j} = 0$ for all $i \neq j$; in other words, the decomposition in (1) is stable under the whole group T_q .

Proof. For (1), let $P(T) \in R[T]$ be the characteristic polynomial of $\rho(\phi_q)$ on M. By Hensel's lemma, P(T) admits a unique decomposition

$$P(T) = \prod_{i=1}^{s} P_i(T)$$

such that $P_i(T) \mod \mathfrak{m}_R$ is the characteristic polynomial of $\bar{\rho}(\phi_q)$ on $\overline{M_i}$ for $1 \leq i \leq s$. We put $Q_i(T) = \prod_{j \neq i} P_j(T)$ for $1 \leq i \leq s$. We put $M_i \coloneqq Q_i(\phi_q)M$ and $N_i \coloneqq P_i(\phi_q)M$. Then both M_i and N_i are both stable under ϕ_q ; M_i is annihilated by $P_i(\phi_q)$; N_i is annihilated by $Q_i(\phi_q)$; hence $M_i \cap N_i = \{0\}$. Using Nakayama's lemma, it is easy to see that $(P_i(T), Q_i(T)) = R[T]$. Thus, there exist polynomials $F_i, G_i \in R[T]$ such that $F_i P_i + G_i Q_i = 1$ in R[T]. We then obtain

$$F_i(\phi_q)N_i + G_i(\phi_q)M_i = M,$$

hence $M = M_i \oplus N_i$. To complete the proof of (1), it suffices to show that $N_i = \bigoplus_{j \neq i} M_j$. By definition, it is clear that $M_j \subseteq N_i$ for every $j \neq i$, hence $\bigoplus_{j \neq i} M_j \subseteq N_i$. The inverse inclusion follows from the fact that the ideal of R[T] generated by Q_j for $j \neq i$ is same as the ideal generated by P_i .

For (2), we choose a basis of M over R adapted to the decomposition of M in (1). We identify $\rho(t)$ and $\rho(\phi_q)$ with their matrices under this basis. We have $\rho(\phi_q)_{i,j} = 0$ for $i \neq j$ since each M_i is stable under $\rho(\phi_q)$. Let $J \subset R$ be the ideal generated by the coefficients of $\rho(t)_{i,j}$ for $i \neq j$. We have to show that J = 0. By Nakayama's lemma, it suffices to show that $J = \mathfrak{m}_R J$. As

$$\rho(t)^{q} = (1 + (\rho(t) - 1))^{q} = 1 + q(\rho(t) - 1) + \sum_{a \ge 2} \binom{q}{a} (\rho(t) - 1)^{a},$$

and $\rho(t) \equiv 1 \mod \mathfrak{m}_R$, we have

 $(\rho(t)^q)_{i,j} \equiv q\rho(t)_{i,j} \mod \mathfrak{m}_R J$

for $i \neq j$. The relation $\phi_q t = t^q \phi_q$ implies that

$$\rho(\phi_q)_{i,i}\rho(t)_{i,j} = (\rho(t)^q)_{i,j}\rho(\phi_q)_{j,j} \equiv q\rho(t)_{i,j}\rho(\phi_q)_{j,j} \mod \mathfrak{m}_R J.$$

It follows that

$$\rho(t)_{i,j} P_i(q\rho(\phi_q)_{j,j}) \equiv P_i(\rho(\phi_q)_{i,i})\rho(t)_{i,j} \equiv 0 \mod \mathfrak{m}_R J$$

for $i \neq j$. By assumption, if $\bar{\alpha}$ is an eigenvalue of $\bar{\rho}(\phi_q)_{i,i}$, then $q^{-1}\bar{\alpha}$ is not an eigenvalue of $\bar{\rho}(\phi_q)_{j,j}$. It follows that $P_i(q\rho(\phi_q)_{j,j})$ is invertible, hence $\rho(t)_{i,j} \equiv 0 \mod \mathfrak{m}_R J$.

The lemma is proved.

E.5. Minimally ramified deformations. In this subsection, we define and study the minimally ramified deformations at places coprime to ℓ . Thus, we take a nonarchimedean place v of F^+ that is not above ℓ .

When v is split in F, the problem has been studied in [CHT08, 2.4.4]. So we assume that v is nonsplit in F. Let w be the unique prime of F above v. Let $I_{F_v^+} \subseteq \Gamma_{F_v^+}$ be the inertia subgroup, and $P_{F_v^+}$ the maximal closed subgroup of $I_{F_v^+}$ of pro-order coprime to ℓ . Put $T_v := \Gamma_{F_v^+}/P_{F_v^+}$. Similarly, we have Γ_{F_w} , I_{F_w} , P_{F_w} , and T_w . Finally, put $T_w^+ := \Gamma_{F_v^+}/P_{F_w}$.

Remark E.5.1. The group T_v is a ||v||-tame group (Definition E.4.1). When w is unramified over v, we have $P_{F_w} = P_{F_v^+}$, $T_w^+ = T_v$, and that the subgroup T_w of T_v is a $||v||^2$ -tame group. When w is ramified over v, we have $P_{F_v^+}/P_{F_w} \simeq \mathbb{Z}/2\mathbb{Z}$, that the natural map $T_w \to T_v$ is an isomorphism, and a canonically split short exact sequence

$$1 \to \mathrm{P}_{F_v^+}/\mathrm{P}_{F_w} \to \mathrm{T}_w^+ \to \mathrm{T}_v \to 1.$$

We recall some facts about extensions of representations of P_{F_w} from [CHT08]. For an irreducible representation τ of P_{F_w} with coefficients in k, we put

$$\Gamma_{\tau} \coloneqq \{ \sigma \in \Gamma_{F_w} \mid \tau^{\sigma} \simeq \tau \},\$$

where we recall from Subsection 1.3 that τ^{σ} denotes the representation given by $\tau^{\sigma}(g) = \tau(\sigma g \sigma^{-1})$ for $g \in P_{F_w}$. Let T_{τ} be the image of Γ_{τ} in $T_w = \Gamma_{F_w}/P_{F_w}$. As P_{F_w} is normal in $\Gamma_{F_v^+}$, we may similar define

$$\Gamma_{\tau}^{+} \coloneqq \{ \sigma \in \Gamma_{F_{v}^{+}} \mid \tau^{\sigma} \simeq \tau \}$$

and denote by T_{τ}^+ its image in T_{w}^+ .

Lemma E.5.2. We have the following properties for τ :

- (1) the dimension of τ is coprime to ℓ ; and τ has a unique deformation to a representation $\tilde{\tau}$ of P_{F_w} over \mathscr{O} ;
- (2) $\tilde{\tau}$ in (1) admits a unique extension to a representation of $\Gamma_{\tau} \cap I_{F_w}$ over \mathscr{O} whose determinant has order coprime to ℓ ;
- (3) there exists an extension of $\tilde{\tau}$ in (2) to a representation of Γ_{τ} over \mathscr{O} .

Proof. This is [CHT08, Lemma 2.4.11].

Now we consider a pair (\bar{r}, χ) from Notation E.2.1 with $\tilde{\Gamma} = \Gamma_{F_v^+}$ and $\Gamma = \Gamma_{F_v^+} \cap \Gamma_F = \Gamma_{F_w}$. Our first goal is to define the notion of *minimally ramified liftings* of \bar{r} (Definition E.5.8).

Recall from Notation E.1.2 that we have the induced homomorphism $\bar{r}^{\natural} \colon \Gamma_{F_w} \to \mathrm{GL}_N(k)$. For an irreducible representation τ of P_{F_w} with coefficients in k, we put

$$M_{\tau}(\bar{r}) \coloneqq \operatorname{Hom}_{k[\operatorname{P}_{F_w}]}(\tau, \bar{r}^{\natural}).$$

Then $\tau \otimes_k M_{\tau}(\bar{r})$ is canonically the τ -isotypic component of \bar{r}^{\natural} . As τ extends to a representation of Γ_{τ} , the k-vector space $M_{\tau}(\bar{r})$ is equipped with a natural action by T_{τ} ; and $\tau \otimes_k M_{\tau}(\bar{r})$ is equipped with a natural action by T_{τ} .

We denote by $\mathfrak{T} = \mathfrak{T}(\bar{r})$ the set of isomorphism classes of irreducible representations τ of P_{F_w} such that $M_{\tau}(\bar{r}) \neq 0$. Then Γ_{F_w} acts on \mathfrak{T} by conjugation, whose orbits we denote by $\mathfrak{T}/\Gamma_{F_w}$. For $\tau \in \mathfrak{T}$, we write $[\tau]$ for its orbit in $\mathfrak{T}/\Gamma_{F_w}$.

Definition E.5.3. We say that E_{λ} is \bar{r} -inclusive if every $\tau \in \mathfrak{T}(\bar{r})$ is absolutely irreducible.

We first suppose that E_{λ} is \bar{r} -inclusive. Let \tilde{E}_{λ} be an unramified quadratic extension of E_{λ} , with the ring of integers $\tilde{\mathscr{O}}$ and the residue field \tilde{k} .

Choose an element $\gamma \in \Gamma_{F_v^+} \setminus \Gamma_{F_w}$. By Lemma E.1.3, the homomorphism \bar{r} is determined by an element $\bar{\Psi} \in \operatorname{GL}_N(k)$ satisfying

$$\bar{r}^{\natural,\gamma} = \bar{\Psi} \circ \chi \bar{r}^{\natural,\vee} \circ \bar{\Psi}^{-1}, \quad \bar{\Psi}^{\,\mathrm{t}} \bar{\Psi}^{-1} = -\chi(\gamma)^{-1} \bar{r}^{\natural}(\gamma^2).$$

In what follows, we will adopt the following simplified notation: for a representation τ of a subgroup of $\Gamma_{F_v^+}$, we write τ^* for $\chi\tau^{\vee}$. Due to the existence of $\overline{\Psi}$, we know that if $\tau \in \mathfrak{T}$, then $\tau^{\gamma,*} \in \mathfrak{T}$ as well. As $\gamma^2 \in \Gamma_{F_w}$, the assignment $\tau \mapsto \tau^{\gamma,*}$ induces an involution on the set $\mathfrak{T}/\Gamma_{F_w}$, which does not depend on the choice of γ .

Construction E.5.4. We now would like to construct a Γ_{F_w} -stable partition $\mathfrak{T} = \mathfrak{T}_1 \sqcup \mathfrak{T}_2 \sqcup \mathfrak{T}_3$. For each subset \mathfrak{T}_i , we will specify, for every $\tau \in \mathfrak{T}_i$, an extension of $\tilde{\tau}$ in Lemma E.5.2(2) to a representation of Γ_{τ} with coefficients in \mathscr{O} ($\tilde{\mathscr{O}}$ if i = 3) in a compatible way, specified below.

We start from the following observation. Suppose $[\tau] = [\tau^{\gamma,*}]$ in $\mathfrak{T}/\Gamma_{F_w}$. Then there exists an element $h \in \Gamma_{F_w}$, unique up to left multiplication by an element in Γ_{τ} , such that $\tau^{\gamma,*} \simeq \tau^{h^{-1}}$, or equivalently, $\tau^{h\gamma} \simeq \tau^*$. Then we have $(h\gamma)^2 \in \Gamma_{\tau}$ but $h\gamma \notin \Gamma_{\tau}$. Denote by $\tilde{\Gamma}_{\tau}$ the subgroup of $\Gamma_{F_{\tau}^+}$.

generated by Γ_{τ} and $h\gamma$, which contains Γ_{τ} as a subgroup of index two. Let \tilde{T}_{τ} be the image of $\tilde{\Gamma}_{\tau}$ in T_{w}^{+} , which contains T_{τ} as a subgroup of index two.

- (1) We define \mathfrak{T}_1 to be the subset of \mathfrak{T} consisting of τ such that $[\tau] \neq [\tau^{\gamma,*}]$. We choose a subset $\mathfrak{T}_1^{\heartsuit} \subseteq \mathfrak{T}_1$ such that $\{\tau, \tau^{\gamma,*} \mid \tau \in \mathfrak{T}_1^{\heartsuit}\}$ is a set of representatives for the Γ_{τ} -action on \mathfrak{T}_1 . For each element $\tau \in \mathfrak{T}_1^{\heartsuit}$, we choose an extension of $\tilde{\tau}$ in Lemma E.5.2(2) to a representation of Γ_{τ} with coefficients in \mathscr{O} , which we still denote by $\tilde{\tau}$. For a general element $\tau \in \mathfrak{T}_1$, there are two cases. If $\tau \simeq \tau_1^h$ for (unique) $\tau_1 \in \mathfrak{T}_1^{\heartsuit}$ and some $h \in \Gamma_{F_w}$, then we choose $\tilde{\tau}$ to be $\tilde{\tau}_1^h$, as the extension to $\Gamma_{\tau} = h^{-1}\Gamma_{\tau_1}h$. If $\tau \simeq (\tau_1^h)^{\gamma,*}$ for (unique) $\tau_1 \in \mathfrak{T}_1^{\heartsuit}$ and some $h \in \Gamma_{F_w}$, then we choose $\tilde{\tau}$ to be $(\tilde{\tau}_1^h)^{\gamma,*}$, as the extension to $\Gamma_{\tau} = \gamma^{-1}h^{-1}\Gamma_{\tau_1}h\gamma$.
- (2) We define \mathfrak{T}_2 to be the subset of \mathfrak{T} consisting of τ such that $[\tau] = [\tau^{\gamma,*}]$, and that the images of Γ_{τ} and $\tilde{\Gamma}_{\tau}$ in $\Gamma_{F_v^+}/I_{F_v^+}$ are different. We choose a subset $\mathfrak{T}_2^{\heartsuit} \subseteq \mathfrak{T}_2$ of representatives for the Γ_{τ} -action on \mathfrak{T}_2 . For each element $\tau \in \mathfrak{T}_2^{\heartsuit}$, we choose an extension $\tilde{\tau}$ from Lemma E.5.6(1) below to a representation of Γ_{τ} with coefficients in \mathscr{O} . For $\tau \in \mathfrak{T}_2$ in general, we have $\tau \simeq \tau_2^h$ for (unique) $\tau_2 \in \mathfrak{T}_2^{\heartsuit}$ and some $h \in \Gamma_{F_w}$; and we choose $\tilde{\tau}$ to be $\tilde{\tau}_2^h$, as the extension to $\Gamma_{\tau} = h^{-1}\Gamma_{\tau_2}h$.
- (3) We define \mathfrak{T}_3 to be the subset of \mathfrak{T} consisting of τ such that $[\tau] = [\tau^{\gamma,*}]$, and that the images of Γ_{τ} and $\tilde{\Gamma}_{\tau}$ in $\Gamma_{F_v^+}/I_{F_v^+}$ are the same. We choose a subset $\mathfrak{T}_3^{\heartsuit} \subseteq \mathfrak{T}_3$ of representatives for the Γ_{τ} -action on \mathfrak{T}_3 . For each element $\tau \in \mathfrak{T}_3^{\heartsuit}$, we choose an extension $\tilde{\tau}$ from Lemma E.5.6(2) below to a representation of Γ_{τ} with coefficients in $\tilde{\mathscr{O}}$. For $\tau \in \mathfrak{T}_3$ in general, we have $\tau \simeq \tau_3^h$ for (unique) $\tau_3 \in \mathfrak{T}_3^{\heartsuit}$ and some $h \in \Gamma_{F_w}$; and we choose $\tilde{\tau}$ to be $\tilde{\tau}_3^h$, as the extension to $\Gamma_{\tau} = h^{-1}\Gamma_{\tau_3}h$.

In addition, we put $\mathfrak{T}^{\heartsuit} \coloneqq \mathfrak{T}_1^{\heartsuit} \sqcup \mathfrak{T}_2^{\heartsuit} \sqcup \mathfrak{T}_3^{\heartsuit}$.

Remark E.5.5. The partition $\mathfrak{T} = \mathfrak{T}_1 \sqcup \mathfrak{T}_2 \sqcup \mathfrak{T}_3$ does not depend on the choice of γ . Moreover, if \mathfrak{T}_3 is nonempty, then w is ramified over v.

Lemma E.5.6. Let $\tau \in \mathfrak{T}$ be an element of dimension d.

- (1) If $\tau \in \mathfrak{T}_2$, then the representation $\tilde{\tau}$ in Lemma E.5.2(2) extends to a representation of Γ_{τ} with coefficients in \mathscr{O} such that $\tilde{\tau}^{\gamma'} \simeq \tilde{\tau}^*$ still holds for every $\gamma' \in \tilde{\Gamma}_{\tau} \setminus \Gamma_{\tau}$.
- (2) If $\tau \in \mathfrak{T}_3$, then the representation $\tilde{\tau}$ in Lemma E.5.2(2) extends to a representation of Γ_{τ} with coefficients in $\tilde{\mathcal{O}}$ such that $\tilde{\tau}^{\gamma'} \simeq \tilde{\tau}^*$ still holds for every $\gamma' \in \tilde{\Gamma}_{\tau} \setminus \Gamma_{\tau}$.

Proof. We fix a splitting $\Gamma_{F_v^+} \simeq \mathcal{P}_{F_v^+} \rtimes \mathcal{T}_v$ and an isomorphism $\mathcal{T}_v \simeq \mathcal{T}_q = t^{\mathbb{Z}_\ell} \rtimes \phi_q^{\mathbb{Z}}$ with the q-tame group (Definition E.4.1) where q = ||v||. Then we have the induced splitting $\Gamma_\tau \simeq \mathcal{P}_{F_w} \rtimes \mathcal{T}_\tau$, where $\mathcal{T}_\tau = t_\tau^{\mathbb{Z}_\ell} \rtimes \phi_\tau^{\mathbb{Z}}$ is a subgroup of \mathcal{T}_q , with $t_\tau = t^{\ell^a}$ and $\phi_\tau = \phi_q^b$ for unique integers $a \ge 0$ and b > 0. To extend $\tilde{\tau}$ in Lemma E.5.2(2) to a representation of Γ_τ , it suffices to specify $\tilde{\tau}(\phi_\tau)$.

For (1), there are two cases.

First, we suppose that w is unramified over v. Then b is even; and \tilde{T}_{τ} is the image of $\tilde{\Gamma}_{\tau}$ in T_v . Then \tilde{T}_{τ} is generated by T_{τ} and an element $\gamma' \in T_v$ of the form $(\tilde{t}_{\tau}, \phi_q^{b/2})$ such that $\gamma'^2 = (\tilde{t}_{\tau}^{q^{b/2}+1}, \phi_q^b)$ lies in Γ_{τ} . As $[\tau] = [\tau^{\gamma,*}]$, we have $\tau^{\gamma'} \simeq \tau^*$. We choose a basis of τ hence regard τ as a homomorphism $\tau \colon P_{F_w} \to \operatorname{GL}_d(k)$. By Lemma E.5.2(1,2), we have a continuous homomorphism $\tilde{\tau} \colon \Gamma_{\tau} \cap I_{F_w} \to \operatorname{GL}_d(\mathcal{O})$ such that $\tilde{\tau}|_{P_{F_w}}$ is a lifting of τ , unique up to conjugation in $1 + M_d(\lambda)$. In particular, there is an element $A \in \operatorname{GL}_d(\mathcal{O})$, unique up to scalar in \mathcal{O}^{\times} , such that

$$\begin{split} \tilde{\tau}^{\gamma'}(g) &= B\tilde{\tau}^*(g)B^{-1} \text{ for every } g \in \Gamma_{\tau} \cap I_{F_w}. \text{ Since } \phi_{\tau} = \tilde{t}_{\tau}^{-q^{b/2}-1}\gamma'^2, \text{ we have} \\ \tilde{\tau}(\phi_{\tau}g\phi_{\tau}^{-1}) &= \tilde{\tau}(\tilde{t}_{\tau}^{-q^{b/2}-1}\gamma'^2g\gamma'^{-2}\tilde{t}_{\tau}^{q^{b/2}+1}) = \tilde{\tau}(\tilde{t}_{\tau}^{-q^{b/2}-1})\tilde{\tau}(\gamma'^2g\gamma'^{-2})\tilde{\tau}(\tilde{t}_{\tau}^{q^{b/2}+1}) \\ &= \tilde{\tau}(\tilde{t}_{\tau}^{-q^{b/2}-1})B\tilde{\tau}^*(\gamma'^{-1}g\gamma')B^{-1}\tilde{\tau}(\tilde{t}_{\tau}^{q^{b/2}+1}) \\ &= \tilde{\tau}(\tilde{t}_{\tau}^{-q^{b/2}-1})(B^{\mathsf{t}}B^{-1})\tilde{\tau}(g)({}^{\mathsf{t}}BB^{-1})\tilde{\tau}(\tilde{t}_{\tau}^{q^{b/2}+1}) \end{split}$$

for every $g \in \Gamma_{\tau} \cap I_{F_w}$. We put $\tilde{\tau}(\phi_{\tau}) \coloneqq -\chi(\phi_q^{b/2})\tilde{\tau}(\tilde{t}_{\tau}^{-q^{b/2}-1})(B^{t}B^{-1})$. Then we obtain the desired extension as in (1).

Second, we suppose that w is ramified over v. By the definition of \mathfrak{T}_2 , the image of $\tilde{\Gamma}_{\tau}$ in $\Gamma_{F_v^+}/I_{F_v^+}$ contains $\phi_{\tau}^{\mathbb{Z}}$ as a subgroup of index two. Thus, there exists an element $\gamma' \in \tilde{\Gamma}_{\tau} \setminus \Gamma_{\tau}$ such that $\gamma^2 = h\phi_{\tau}$ for some $h \in \Gamma_{\tau} \cap I_{F_w}$. The remaining argument is same to the above case.

For (2), by the definition of \mathfrak{T}_3 , the image of $\tilde{\Gamma}_{\tau}$ in $\Gamma_{F_v^+}/I_{F_v^+}$ coincides with $\phi_{\tau}^{\mathbb{Z}}$. In particular, we can find an element $\gamma' \in \tilde{\Gamma}_{\tau} \setminus \Gamma_{\tau}$ contained in $I_{F_v^+} \setminus I_{F_w}$. By Lemma E.5.2(1,2), we have a continuous homomorphism $\tilde{\tau} \colon \Gamma_{\tau} \cap I_{F_w} \to \operatorname{GL}_d(\mathcal{O})$ such that $\tilde{\tau}|_{P_{F_w}}$ is a lifting of τ , unique up to conjugation in $1 + \operatorname{M}_d(\lambda)$. As we have $\tau^{\gamma'} \simeq \tau^*$ and $\tau^{\phi_{\tau}} \simeq \tau$, there are elements $A, B \in \operatorname{GL}_d(\mathcal{O})$ such that

(E.9)
$$\tilde{\tau}^{\gamma'}(g) = A\tilde{\tau}^*(g)A^{-1}$$

(E.10)
$$\tilde{\tau}^{\phi_{\tau}}(g) = B\tilde{\tau}(g)B^{-1},$$

for every $g \in \Gamma_{\tau} \cap I_{F_w}$. It follows from (E.9) that the desired element $\tilde{\tau}(\phi_{\tau}) \in \operatorname{GL}_d(\mathcal{O})$ has to satisfy the equation

(E.11)
$$\chi(\phi_{\tau})A^{t}\tilde{\tau}(\phi_{\tau})^{-1}A^{-1} = \tilde{\tau}(\gamma'\phi_{\tau}\gamma'^{-1}) = \tilde{\tau}(\gamma'\phi_{\tau}\gamma'^{-1}\phi_{\tau}^{-1})\tilde{\tau}(\phi_{\tau}),$$

where we note that $\gamma' \phi_{\tau} \gamma'^{-1} \phi_{\tau}^{-1} \in \Gamma_{\tau} \cap I_{F_w}$. However, by (E.10), we have

$$\tilde{\tau}^{\gamma'\phi_{\tau}\gamma'^{-1}}(g) = (\tilde{\tau}(\gamma'\phi_{\tau}\gamma'^{-1}\phi_{\tau}^{-1})B)\tilde{\tau}(g)(\tilde{\tau}(\gamma'\phi_{\tau}\gamma'^{-1}\phi_{\tau}^{-1})B)^{-1}$$

for every $g \in \Gamma_{\tau} \cap I_{F_w}$. On the other hand, by (E.9) and (E.10), we have

$$\tilde{\tau}^{\gamma'^{-1}\phi_{\tau}\gamma'}(g) = (A^{\mathsf{t}}B^{-1}A^{-1})\tilde{\tau}(g)(A^{\mathsf{t}}B^{-1}A^{-1})^{-1}$$

for every $g \in \Gamma_{\tau} \cap I_{F_w}$. Since τ is absolutely irreducible, it follows that there exists $\beta \in \mathscr{O}^{\times}$ such that

$$A^{\mathsf{t}}B^{-1}A^{-1} = \beta \cdot \tilde{\tau}(\gamma'\phi_{\tau}\gamma'^{-1}\phi_{\tau}^{-1})B.$$

Take an element $\alpha \in \tilde{\mathscr{O}}^{\times}$ such that $\alpha^2 = \beta \chi(\phi_{\tau})$. Then it is clear that $\tilde{\tau}(\phi_{\tau}) = \alpha B \in \operatorname{GL}_d(\tilde{\mathscr{O}})$ is a solution to (E.11).

The lemma is proved.

Using Construction E.5.4, we now discuss the structure of liftings of \bar{r} . We replace E_{λ} by E_{λ} so that every representations $\tilde{\tau}$ in Construction E.5.4 have coefficients in \mathscr{O} . Let $r \colon \Gamma_{F_v^+} \to \mathscr{G}_N(R)$ be a lifting of \bar{r} to an object R of $\mathscr{C}_{\mathscr{O}}$. By Lemma E.1.3, to give such a lifting r is equivalent to giving an element $\Psi \in \operatorname{GL}_N(R)$ that reduces to $\bar{\Psi}$ and satisfies

$$r^{\natural,\gamma} = \Psi \circ \chi r^{\natural,\vee} \circ \Psi^{-1}, \quad \Psi^{\mathsf{t}} \Psi^{-1} = -\chi(\gamma)^{-1} r^{\natural}(\gamma^2).$$

For every $\tau \in \mathfrak{T}$, put

$$M_{\tau}(r) \coloneqq \operatorname{Hom}_{R[P_{F_w}]}(\tilde{\tau} \otimes_{\mathscr{O}} R, r^{\natural}),$$

which is a finite free *R*-module equipped with the induced continuous action by T_{τ} . Denote by $m_{\tau} \geq 1$ the rank of $M_{\tau}(r)$. Let $\tau' \in \mathfrak{T}$ be the unique element such that $\tau^{\gamma} \simeq \tau'^*$. Choose an

isomorphism $\iota_{\tau} \colon \tau^{\gamma} \xrightarrow{\sim} \tau'^*$, which, by construction E.5.4, lifts to an isomorphism $\iota_{\tilde{\tau}} \colon \tilde{\tau}^{\gamma} \xrightarrow{\sim} \tilde{\tau}'^*$ of representations of $\Gamma_{\tau'}$. Then we have isomorphisms

$$M_{\tau}(r)^{\gamma} \xrightarrow{\sim} \operatorname{Hom}_{R[\operatorname{P}_{F_w}]}(\tilde{\tau}^{\gamma} \otimes_{\mathscr{O}} R, r^{\natural, \gamma}) \xrightarrow{\sim} \operatorname{Hom}_{R[\operatorname{P}_{F_w}]}(\tilde{\tau}'^* \otimes_{\mathscr{O}} R, r^{\natural, *}) \xrightarrow{\sim} \operatorname{Hom}_{R[\operatorname{P}_{F_w}]}(r^{\natural}, \tilde{\tau}' \otimes_{\mathscr{O}} R),$$

where the second isomorphism is induced by $\iota_{\tilde{\tau}}$ and Ψ . As τ' is absolutely irreducible, we obtain a perfect *R*-bilinear pairing

$$M_{\tau}(r)^{\gamma} \times M_{\tau'}(r) \to \operatorname{End}_{R[\operatorname{P}_{F_w}]}(\tilde{\tau}' \otimes_{\mathscr{O}} R) = R,$$

which induces an isomorphism

$$\theta_{\tilde{\tau},r} \colon M_{\tau}(r)^{\gamma} \xrightarrow{\sim} M_{\tau'}(r)^{\vee} \coloneqq \operatorname{Hom}_{R}(M_{\tau'}(r), R)$$

of $R[T_{\tau'}]$ -modules. In particular, we have

$$r^{\natural} \simeq \left(\bigoplus_{\tau \in \mathfrak{T}_{1}^{\heartsuit}} \left(\operatorname{Ind}_{\Gamma_{\tau}}^{\Gamma_{F_{w}}}(\tilde{\tau} \otimes_{\mathscr{O}} M_{\tau}(r)) \oplus \operatorname{Ind}_{\Gamma_{\tau}\gamma}^{\Gamma_{F_{w}}}(\tilde{\tau}^{\gamma,*} \otimes_{\mathscr{O}} M_{\tau}(r)^{\gamma,\vee}) \right) \right) \oplus \left(\bigoplus_{\tau \in \mathfrak{T}_{2}^{\heartsuit} \sqcup \mathfrak{T}_{3}^{\heartsuit}} \operatorname{Ind}_{\Gamma_{\tau}}^{\Gamma_{F_{w}}}(\tilde{\tau} \otimes_{\mathscr{O}} M_{\tau}(r)) \right)$$

as representations of Γ_{F_w} .

Now for every τ , we fix an isomorphism $b_{\tau} \colon M_{\tau}(\bar{r}) \xrightarrow{\sim} k^{\oplus m_{\tau}}$ of k-vector spaces, and let $\bar{\varrho}_{\tau} \colon T_{\tau} \to k^{\oplus m_{\tau}}$ $\operatorname{GL}_{m_{\tau}}(k)$ be the induced homomorphism. There are two cases.

(a) Suppose $\tau \in \mathfrak{T}_1$. Then $M_{\tau'}(r)$ is determined by $M_{\tau}(r)$. If we choose an isomorphism $M_{\tau}(r) \simeq R^{\oplus m_{\tau}}$ of R-modules that reduces to b_{τ} , then we obtain a continuous homomorphism

$$\varrho_\tau\colon \mathrm{T}_\tau\to \mathrm{GL}_{m_\tau}(R)$$

that reduces to $\bar{\rho}_{\tau}$.

(b) Suppose $\tau \in \mathfrak{T}_2 \sqcup \mathfrak{T}_3$. Let h be element from Construction E.5.4. Then $\theta_{\tilde{\tau},r}$ induces an isomorphism $M_{\tau}(r)^{h\gamma} \xrightarrow{\sim} M_{\tau}(\tau)^{\vee}$ of $R[T_{\tau}]$ -modules. Applying Lemma E.1.3 to $r = \bar{r}$, we obtain a homomorphism

$$\bar{p}_{\tau} \colon \tilde{\mathrm{T}}_{\tau} \to \mathscr{G}_{m_{\tau}}(k)$$

satisfying $\bar{\varrho}^{-1}(\operatorname{GL}_{m_{\tau}}(k) \times k^{\times}) = \operatorname{T}_{\tau}$ and $\nu \circ \bar{\varrho}_{\tau} = \eta_{v}^{\mu_{\tau}}$ for some $\mu_{\tau} \in \mathbb{Z}/2\mathbb{Z}$ determined by $\tilde{\tau}^{33}$. In general, if we choose an isomorphism $M_{\tau}(r) \simeq R^{\oplus m_{\tau}}$ of *R*-modules that reduces b_{τ} , then we obtain a continuous homomorphism

$$\varrho_{\tau}\colon \mathbf{T}_{\tau}\to \mathscr{G}_{m_{\tau}}(R)$$

that reduces to $\bar{\varrho}_{\tau}$ and satisfies $\nu \circ \rho_{\tau} = \eta_{\nu}^{\mu_{\tau}}$.

The following proposition is the counterpart of [CHT08, Corollary 2.4.13] when v is nonsplit in F.

Proposition E.5.7. We keep the choices of $\gamma \in \Gamma_{F_w^+} \setminus \Gamma_{F_w}$, those in Construction E.5.4, ι_{τ} , and b_{τ} . We also recall that E_{λ} now is an unramified quadratic extension of a \bar{r} -inclusive extension of \mathbb{Q}_{ℓ} . For every object R of $\mathscr{C}_{\mathscr{O}}$, the assignment

$$r\mapsto (\varrho_\tau)_{\tau\in\mathfrak{T}^\heartsuit}$$

establishes a bijection between deformations of \bar{r} to R and equivalence classes of tuples $(\rho_{\tau})_{\tau\in\mathfrak{I}^{\otimes}}$ where

- (a) for $\tau \in \mathfrak{T}_1^{\heartsuit}$, $\varrho_\tau \colon \mathrm{T}_\tau \to \mathrm{GL}_{m_\tau}(R)$ is a continuous homomorphism that reduces to $\bar{\varrho}_\tau$; (b) for $\tau \in \mathfrak{T}_2^{\heartsuit} \sqcup \mathfrak{T}_3^{\heartsuit}$, $\varrho_\tau \colon \tilde{\mathrm{T}}_\tau \to \mathscr{G}_{m_\tau}(R)$ is a continuous homomorphism that reduces to $\bar{\varrho}_\tau$ and satisfies $\nu \circ \rho_{\tau} = \eta_{v}^{\mu_{\tau}}$.

³³In fact, when $\tau \in \mathfrak{T}_2$, one can always modify $\tilde{\tau}$ to make $\mu_{\tau} = 0$; but when $\tau \in \mathfrak{T}_3$, μ_{τ} is determined by τ .

Here, two tuples $(\varrho_{\tau})_{\tau \in \mathfrak{T}^{\heartsuit}}$ and $(\varrho'_{\tau})_{\tau \in \mathfrak{T}^{\heartsuit}}$ are said to be equivalent if ϱ_{τ} and ϱ'_{τ} are conjugate by elements in $1 + M_{m_{\tau}}(\mathfrak{m}_R)$ for every $\tau \in \mathfrak{T}^{\heartsuit}$.

Proof. We now attach to every tuple $(\varrho_{\tau})_{\tau \in \mathfrak{T}^{\heartsuit}}$ as in the statement a lifting r explicitly. Denote by M_{τ} the $R[\mathbf{T}_{\tau}]$ -module corresponding to ϱ_{τ} . Consider

$$M \coloneqq \left(\bigoplus_{\tau \in \mathfrak{T}_1^{\heartsuit}} \left(\operatorname{Ind}_{\Gamma_{\tau}}^{\Gamma_{F_w}} (\tilde{\tau} \otimes_{\mathscr{O}} M_{\tau}) \oplus \operatorname{Ind}_{\Gamma_{\tau}^{\gamma}}^{\Gamma_{F_w}} (\tilde{\tau}^{\gamma, *} \otimes_{\mathscr{O}} M_{\tau}^{\gamma, \vee}) \right) \right) \bigoplus \left(\bigoplus_{\tau \in \mathfrak{T}_2^{\heartsuit} \sqcup \mathfrak{T}_3^{\heartsuit}} \operatorname{Ind}_{\Gamma_{\tau}}^{\Gamma_{F_w}} (\tilde{\tau} \otimes_{\mathscr{O}} M_{\tau}) \right)$$

which is a free *R*-module of rank *N*, equipped with a continuous action by Γ_{F_w} . Moreover, we have $M \otimes_R R/\mathfrak{m}_R \simeq \overline{r}^{\natural}$ as representations of Γ_{F_w} by (E.12). Thus, we may fix an isomorphism $M \simeq R^{\otimes N}$ such that the induced continuous homomorphism $\rho = \rho_M \colon \Gamma_{F_w} \to \operatorname{GL}_N(R)$ reduces to \overline{r}^{\natural} . Thus, by Lemma E.1.3, to construct the desired lifting *r* from ρ , it amounts to finding an element $\Psi \in \operatorname{GL}_N(R)$ satisfying

(E.13)
$$\rho^{\gamma} = \Psi \circ \chi \rho^{\vee} \circ \Psi^{-1}, \quad \Psi^{\mathsf{t}} \Psi^{-1} = -\chi(\gamma)^{-1} \rho(\gamma^2).$$

We will construct Ψ as a direct sum of Ψ_{τ} for $\tau \in \mathfrak{T}^{\heartsuit}$.

For $\tau \in \mathfrak{T}_1^{\heartsuit}$, we note that $\tilde{\tau}(\gamma^{-2}) \otimes \varrho_{\tau}(\gamma^{-2})$ induces an isomorphism

$$\operatorname{Ind}^{\gamma^{-2}} \colon \operatorname{Ind}_{\Gamma_{\tau\gamma^{2}}}^{\Gamma_{F_{w}}}(\tilde{\tau}^{\gamma^{2}} \otimes_{\mathscr{O}} M_{\tau}^{\gamma^{2}}) \simeq \left(\operatorname{Ind}_{\Gamma_{\tau\gamma}}^{\Gamma_{F_{w}}}(\tilde{\tau}^{\gamma,*} \otimes_{\mathscr{O}} M_{\tau}^{\gamma,\vee})\right)^{\gamma,*} \xrightarrow{\sim} \operatorname{Ind}_{\Gamma_{\tau}}^{\Gamma_{F_{w}}}(\tilde{\tau} \otimes_{\mathscr{O}} M_{\tau})$$

Thus, we obtain an isomorphism

$$\left(\operatorname{Ind}_{\Gamma_{\tau}}^{\Gamma_{F_{w}}}(\tilde{\tau}\otimes_{\mathscr{O}}M_{\tau})\oplus\operatorname{Ind}_{\Gamma_{\tau}\gamma}^{\Gamma_{F_{w}}}(\tilde{\tau}^{\gamma,*}\otimes_{\mathscr{O}}M_{\tau}^{\gamma,\vee})\right)^{\gamma,*} \xrightarrow{\sim} \operatorname{Ind}_{\Gamma_{\tau}}^{\Gamma_{F_{w}}}(\tilde{\tau}\otimes_{\mathscr{O}}M_{\tau})\oplus\operatorname{Ind}_{\Gamma_{\tau}\gamma}^{\Gamma_{F_{w}}}(\tilde{\tau}^{\gamma,*}\otimes_{\mathscr{O}}M_{\tau}^{\gamma,\vee})$$

as the composition of the canonical isomorphism

$$\left(\operatorname{Ind}_{\Gamma_{\tau}}^{\Gamma_{F_{w}}}(\tilde{\tau}\otimes_{\mathscr{O}}M_{\tau})\oplus\operatorname{Ind}_{\Gamma_{\tau}\gamma}^{\Gamma_{F_{w}}}(\tilde{\tau}^{\gamma,*}\otimes_{\mathscr{O}}M_{\tau}^{\gamma,\vee})\right)^{\gamma,*} \xrightarrow{\sim} \operatorname{Ind}_{\Gamma_{\tau}\gamma}^{\Gamma_{F_{w}}}(\tilde{\tau}^{\gamma,*}\otimes_{\mathscr{O}}M_{\tau}^{\gamma,\vee}) \oplus \left(\operatorname{Ind}_{\Gamma_{\tau}\gamma}^{\Gamma_{F_{w}}}(\tilde{\tau}^{\gamma,*}\otimes_{\mathscr{O}}M_{\tau}^{\gamma,\vee})\right)^{\gamma,*}$$

and the isomorphism

$$\operatorname{Ind}_{\Gamma_{\tau\gamma}}^{\Gamma_{F_w}}(\tilde{\tau}^{\gamma,*} \otimes_{\mathscr{O}} M_{\tau}^{\gamma,\vee}) \oplus \left(\operatorname{Ind}_{\Gamma_{\tau\gamma}}^{\Gamma_{F_w}}(\tilde{\tau}^{\gamma,*} \otimes_{\mathscr{O}} M_{\tau}^{\gamma,\vee})\right)^{\gamma,*} \xrightarrow{\sim} \operatorname{Ind}_{\Gamma_{\tau}}^{\Gamma_{F_w}}(\tilde{\tau} \otimes_{\mathscr{O}} M_{\tau}) \oplus \operatorname{Ind}_{\Gamma_{\tau\gamma}}^{\Gamma_{F_w}}(\tilde{\tau}^{\gamma,*} \otimes_{\mathscr{O}} M_{\tau}^{\gamma,\vee})$$

given by the matrix $\begin{pmatrix} 0 & -\chi(\gamma) \operatorname{Ind}^{\gamma^{-2}} \\ 1 & 0 \end{pmatrix}$. We now let Ψ_{τ} be the matrix representing the isomorphism

$$\left(\operatorname{Ind}_{\Gamma_{\tau}}^{\Gamma_{F_{w}}}(\tilde{\tau}\otimes_{\mathscr{O}}M_{\tau}) \oplus \operatorname{Ind}_{\Gamma_{\tau}\gamma}^{\Gamma_{F_{w}}}(\tilde{\tau}^{\gamma,*}\otimes_{\mathscr{O}}M_{\tau}^{\gamma,\vee})\right)^{*} \xrightarrow{\sim} \left(\operatorname{Ind}_{\Gamma_{\tau}}^{\Gamma_{F_{w}}}(\tilde{\tau}\otimes_{\mathscr{O}}M_{\tau}) \oplus \operatorname{Ind}_{\Gamma_{\tau}\gamma}^{\Gamma_{F_{w}}}(\tilde{\tau}^{\gamma,*}\otimes_{\mathscr{O}}M_{\tau}^{\gamma,\vee})\right)^{\gamma}$$

induced from (E.14) by duality.

For $\tau \in \mathfrak{T}_2^{\heartsuit} \sqcup \mathfrak{T}_3^{\heartsuit'}$, let *h* be the element in Construction E.5.4. Put $\gamma' \coloneqq h\gamma$, which is an element in $\tilde{\Gamma}_{\tau} \setminus \Gamma_{\tau}$. The homomorphism $\varrho_{\tau} :: \tilde{T}_{\tau} \to \mathscr{G}_{m_{\tau}}(R)$ induces an isomorphism $M_{\tau}^{\gamma'} \xrightarrow{\sim} M_{\tau}^{\vee}$ by Lemma E.1.3(1), which induces an isomorphism $M_{\tau}^{\gamma,\vee} \xrightarrow{\sim} M_{\tau}^{h^{-1}}$. On the other hand, by Lemma E.5.6, we have an isomorphism $\tilde{\tau}^{\gamma,*} \simeq \tilde{\tau}^{h^{-1}}$. Thus, we obtain an isomorphism

(E.15)
$$\left(\operatorname{Ind}_{\Gamma_{\tau}}^{\Gamma_{F_{w}}}(\tilde{\tau}\otimes_{\mathscr{O}}M_{\tau})\right)^{\gamma,*} \xrightarrow{\sim} \operatorname{Ind}_{\Gamma_{\tau}}^{\Gamma_{F_{w}}}(\tilde{\tau}\otimes_{\mathscr{O}}M_{\tau})$$

as the composition of the canonical isomorphism

$$\left(\operatorname{Ind}_{\Gamma_{\tau}}^{\Gamma_{F_{w}}}(\tilde{\tau}\otimes_{\mathscr{O}}M_{\tau})\right)^{\gamma,*} \xrightarrow{\sim} \operatorname{Ind}_{\Gamma_{\tau}^{\gamma}}^{\Gamma_{F_{w}}}(\tilde{\tau}^{\gamma,*}\otimes_{\mathscr{O}}M_{\tau}^{\gamma,\vee}),$$

the isomorphism

$$\operatorname{Ind}_{\Gamma_{\tau^{\gamma}}}^{\Gamma_{F_{w}}}(\tilde{\tau}^{\gamma,*}\otimes_{\mathscr{O}}M_{\tau}^{\gamma,\vee})\xrightarrow{\sim}\operatorname{Ind}_{\Gamma_{\tau^{h^{-1}}}}^{\Gamma_{F_{w}}}(\tilde{\tau}^{h^{-1}}\otimes_{\mathscr{O}}M_{\tau}^{h^{-1}})$$

specified above, and the isomorphism

$$\operatorname{Ind}_{\Gamma_{\tau^{h^{-1}}}}^{\Gamma_{F_w}}(\tilde{\tau}^{h^{-1}}\otimes_{\mathscr{O}} M_{\tau}^{h^{-1}}) \xrightarrow{\sim} \operatorname{Ind}_{\Gamma_{\tau}}^{\Gamma_{F_w}}(\tilde{\tau}\otimes_{\mathscr{O}} M_{\tau})$$

given by the action of h^{-1} . We now let Ψ_{τ} be the matrix representing the isomorphism

$$\left(\operatorname{Ind}_{\Gamma_{\tau}}^{\Gamma_{F_{w}}}(\tilde{\tau}\otimes_{\mathscr{O}}M_{\tau})\right)^{*}\xrightarrow{\sim}\left(\operatorname{Ind}_{\Gamma_{\tau}}^{\Gamma_{F_{w}}}(\tilde{\tau}\otimes_{\mathscr{O}}M_{\tau})\right)^{*}$$

induced from (E.15) by duality.

Finally, we put $\Psi := \bigoplus_{\tau \in \mathfrak{T}^{\heartsuit}} \Psi_{\tau}$. Then (E.13) follows by construction. In other words, we have assigned a lifting r from the tuple $(\varrho_{\tau})_{\tau \in \mathfrak{T}^{\heartsuit}}$. It is straightforward to check that such assignment is inverse to the assignment in the proposition. The proposition follows.

From now till the end of this subsection, we assume $l \ge N$. Using Proposition E.5.7, we can define minimally ramified liftings of \bar{r} . We now do *not* assume that E_{λ} is \bar{r} -inclusive. We choose an unramified quadratic extension E'_{λ} of an \bar{r} -inclusive unramified extension of E_{λ} , with the ring of integers \mathscr{O}' and the residue field k'. We also keep the choices of $\gamma \in \Gamma_{F_v^+} \setminus \Gamma_{F_w}$, those in Construction E.5.4, ι_{τ} , and b_{τ} , as in Proposition E.5.7 (with respect to E'_{λ}).

Definition E.5.8. We say that a lifting r of \bar{r} to some object R of $\mathscr{C}_{\mathscr{O}}$ is minimally ramified if in the tuple $(\varrho_{\tau})_{\tau \in \mathfrak{T}^{\heartsuit}}$ corresponding to the composition $\Gamma_{F_v^+} \xrightarrow{r} \mathscr{G}_N(R) \to \mathscr{G}_N(R \otimes_{\mathscr{O}} \mathscr{O}')$, every homomorphism ϱ_{τ} is a minimally ramified lifting of $\bar{\varrho}_{\tau}$ in the following sense.

- (1) For $\tau \in \mathfrak{T}_1^{\heartsuit}$, minimally ramified liftings of $\bar{\varrho}_{\tau}$ is defined in the sense of [CHT08, Definition 2.4.14].
- (2) For $\tau \in \mathfrak{T}_2^{\heartsuit}$, note that \tilde{T}_{τ} is isomorphic to the q_{τ} -tame group for some power q_{τ} of ||v||under which the subgroup T_{τ} is the q_{τ}^2 -tame group. Thus, we may define minimally ramified liftings of $\bar{\varrho}_{\tau}$ using Definition E.4.4 (with respect to the similitude character $\eta_v^{\mu_{\tau}}$, which is trivial on T_{τ});
- (3) For $\tau \in \mathfrak{T}_3^{\heartsuit}$, note that $\tilde{\mathrm{T}}_{\tau} \simeq \mathrm{T}_{\tau} \times \mathbb{Z}/2\mathbb{Z}$. Then, by Lemma E.1.3, we may regard the homomorphism ϱ_{τ} as a continuous homomorphism $\varrho_{\tau} \colon \mathrm{T}_{\tau} \to G(R)$, where G is a symplectic (resp. orthogonal) group of rank m_{τ} if μ_{τ} is 0 (resp. 1). Thus, we may define minimally ramified liftings of $\bar{\varrho}_{\tau}$ using [Boo19, Definition 5.4].

Remark E.5.9. It is straightforward to check that Definition E.5.8 do not depend on the choice of E'_{λ} , $\gamma \in \Gamma_{F_v} \setminus \Gamma_{F_w}$, those in Construction E.5.4, ι_{τ} , and b_{τ} .

Now we allow v to be a nonarchimedean place of F^+ that is not above ℓ , but not necessarily nonsplit in F. Again, we consider a pair (\bar{r}, χ) from Notation E.2.1 with $\tilde{\Gamma} = \Gamma_{F^+_{\pi}}$ and $\Gamma = \Gamma_{F^+_{\pi}} \cap \Gamma_F$.

Definition E.5.10. We define \mathscr{D}^{\min} to be the local deformation problem of \bar{r} that classifies minimally ramified liftings in the sense of Definition E.5.8 (resp. [CHT08, Definition 2.4.14]) when v is nonsplit (resp. split) in F.

Proposition E.5.11. We have

- (1) The ring $\mathsf{R}^{\mathrm{loc}}_{\bar{r}}$ is a reduced local complete intersection, flat and of pure relative dimension N^2 over \mathscr{O} .
- (2) Every irreducible component of Spf $\mathsf{R}^{\mathsf{loc}}_{\bar{r}}$ is a local deformation problem (Definition E.2.3).
- (3) If $\ell \geq N$, then \mathscr{D}^{\min} is an irreducible component of $\operatorname{Spf} \mathsf{R}^{\operatorname{loc}}_{\bar{r}}$ and is formally smooth over $\operatorname{Spf} \mathscr{O}$ of pure relative dimension N^2 .

Proof. We may assume $E_{\lambda} = E'_{\lambda}$.

For (1), when v splits in F, this is [Sho18, Theorem 2.5]. Thus, we may assume that v is nonsplit in F. By Proposition E.5.7, $\mathsf{R}_{\bar{r}}^{\mathsf{loc}}$ is a power series ring over

$$\widehat{\bigotimes}_{\tau\in\mathfrak{T}^{\heartsuit}}\mathsf{R}^{\mathrm{loc}}_{\bar{\varrho}\tau}.$$

We now claim that for every $\tau \in \mathfrak{T}^{\heartsuit}$, $\mathsf{R}^{\mathsf{loc}}_{\bar{\varrho}_{\tau}}$ a local complete intersection, flat and equidimensional. Indeed, for $\tau \in \mathfrak{T}^{\heartsuit}_{1}$, this is [Sho18, Theorem 2.5]; for $\tau \in \mathfrak{T}^{\heartsuit}_{2}$, this is Corollary E.4.3; for $\tau \in \mathfrak{T}^{\heartsuit}_{3}$, this is Proposition E.4.2 for G a symplectic or orthogonal group with the trivial similitude character. On the other hand, by [BG19, Theorem 3.3.2] or [BP, Theorem 1], we know that $\mathsf{R}^{\mathsf{loc}}_{\bar{r}}[1/\ell]$ is reduced and of pure dimension dim $\mathscr{G}_N = N^2$. Thus, $\mathsf{R}^{\mathsf{loc}}_{\bar{r}}$ is a local complete intersection, flat and of pure relative dimension N^2 over \mathscr{O} . Since $\mathsf{R}^{\mathsf{loc}}_{\bar{r}}$ is generically reduced and Cohen–Macaulay, it is reduced. (1) is proved.

For (2), take an irreducible component \mathscr{D} of Spf $\mathsf{R}^{\mathrm{loc}}_{\bar{r}}$, and let \mathscr{L}_N be the formal completion of $\mathrm{GL}_{N,\mathscr{O}}$ along the unit section. Then the conjugation action induces a homomorphism $\mathscr{L}_N \times_{\mathrm{Spf}\,\mathscr{O}} \mathscr{D} \to \mathrm{Spf}\,\mathsf{R}^{\mathrm{loc}}_{\bar{r}}$ whose image contains \mathscr{D} . Since \mathscr{L}_N is irreducible, the image is irreducible hence has to be \mathscr{D} . In other words, \mathscr{D} is a local deformation problem.

For (3), since \mathscr{D}^{\min} is Zariski closed in Spf $\mathbb{R}^{\text{loc}}_{\bar{r}}$ from its definition, it suffices to show that \mathscr{D}^{\min} is formally smooth over Spf \mathscr{O} of pure relative dimension N^2 . When v splits in F, this is [CHT08, Corollary 2.4.21]. Thus, we may assume that v is nonsplit in F. For $\tau \in \mathfrak{T}^{\heartsuit}$, let $\mathscr{D}_{\bar{\varrho}_{\tau}}^{\min}$ be the local deformation problem of $\bar{\varrho}_{\tau}$ classifying minimally ramified liftings of $\bar{\varrho}_{\tau}$ in various cases in Definition E.5.8. By Proposition E.5.7 and Definition E.5.8, \mathscr{D}^{\min} is formally smooth over

$$\prod_{\tau \in \mathfrak{T}^{\heartsuit}} \mathscr{D}_{\bar{\varrho}_{\tau}}^{\min}.$$

We claim that for every $\tau \in \mathfrak{T}^{\heartsuit}$, $\mathscr{D}_{\bar{\varrho}_{\tau}}^{\min}$ is formally smooth over Spf \mathscr{O} . Indeed, for $\tau \in \mathfrak{T}_{1}^{\heartsuit}$, this is [CHT08, Lemma 2.4.19]; for $\tau \in \mathfrak{T}_{2}^{\heartsuit}$, this is Proposition E.4.5; for $\tau \in \mathfrak{T}_{3}^{\heartsuit}$, this is a part of [Boo19, Theorem 1.1]. Thus, \mathscr{D}^{\min} is formally smooth over Spf \mathscr{O} .

It remains to compute the dimension. By (E.1), it suffices to show that

(E.16)
$$\dim_k \mathcal{L}(\mathscr{D}^{\min}) = \dim_k \mathcal{H}^0(F_v^+, \operatorname{ad} \bar{r}).$$

For every $\tau \in \mathfrak{T}^{\heartsuit}$, let $L(\mathscr{D}_{\bar{\varrho}_{\tau}}^{\min})$ be the tangent space of the deformation problem $\mathscr{D}_{\bar{\varrho}_{\tau}}^{\min}$, which is a subspace of $H^{1}(T_{\tau}, \operatorname{ad} \bar{\varrho})$ (resp. $H^{1}(\tilde{T}_{\tau}, \operatorname{ad} \bar{\varrho})$) if $\tau \in \mathfrak{T}_{1}^{\heartsuit}$ (resp. $\tau \in \mathfrak{T}_{2}^{\heartsuit} \sqcup \mathfrak{T}_{3}^{\heartsuit}$). By Proposition E.5.7, we have

(E.17)
$$\dim_k \mathcal{L}(\mathscr{D}^{\min}) = \sum_{\tau \in \mathfrak{T}^{\heartsuit}} \dim_k \mathcal{L}(\mathscr{D}_{\bar{\varrho}_{\tau}}^{\min})$$

We claim that

(E.18)
$$\dim_k \mathcal{L}(\mathscr{D}_{\bar{\varrho}_{\tau}}^{\min}) = \begin{cases} \dim_k \mathcal{H}^0(\mathcal{T}_{\tau}, \operatorname{ad} \bar{\varrho}_{\tau}) & \text{if } \tau \in \mathfrak{T}_1^{\heartsuit}; \\ \dim_k \mathcal{H}^0(\tilde{\mathcal{T}}_{\tau}, \operatorname{ad} \bar{\varrho}_{\tau}) & \text{if } \tau \in \mathfrak{T}_2^{\heartsuit} \sqcup \mathfrak{T}_3^{\heartsuit} \end{cases}$$

Indeed, for $\tau \in \mathfrak{T}_1^{\heartsuit}$, this is [CHT08, Corollary 2.4.20]; for $\tau \in \mathfrak{T}_2^{\heartsuit}$, this is Corollary E.4.6; for $\tau \in \mathfrak{T}_3^{\heartsuit}$, this is a part of [Boo19, Theorem 1.1] as $\dim_k \mathrm{H}^0(\tilde{\mathrm{T}}_{\tau}, \mathrm{ad}\,\bar{\varrho}_{\tau}) = \dim_k \mathrm{H}^0(\mathrm{T}_{\tau}, \mathrm{ad}^0\,\bar{\varrho}_{\tau})$. From (E.12) for \bar{r} , we have

$$\mathrm{H}^{0}(F_{v}^{+}, \mathrm{ad}\,\bar{r}) \simeq \left(\bigoplus_{\tau \in \mathfrak{T}_{1}^{\heartsuit}} \mathrm{H}^{0}(\mathrm{T}_{\tau}, \mathrm{ad}\,\bar{\varrho}_{\tau})\right) \bigoplus \left(\bigoplus_{\tau \in \mathfrak{T}_{2}^{\heartsuit} \sqcup \mathfrak{T}_{3}^{\heartsuit}} \mathrm{H}^{0}(\tilde{\mathrm{T}}_{\tau}, \mathrm{ad}\,\bar{\varrho}_{\tau})\right).$$

Together with (E.17) and (E.18), we obtain (E.16).

The proposition is proved.

Now we discuss one example of minimally ramified liftings, which is used in Proposition 8.1.5. Consider an elliptic curve A over F_v^+ . For every rational prime ℓ , we fix an isomorphism $\mathrm{H}^1_{\mathrm{\acute{e}t}}(A_{\alpha\overline{F}},\mathbb{Z}_\ell) \simeq \mathbb{Z}_\ell^{\oplus 2}$, hence obtain a continuous homomorphism $\rho_{A,\ell} \colon \Gamma_{F_v^+} \to \mathrm{GL}_2(\mathbb{Z}_\ell)$. Suppose $N \geq 2$. We obtain a continuous homomorphism

$$r_{A,\ell} \colon \Gamma_{F^+_n} \to \mathscr{G}_N(\mathbb{Z}_\ell) = (\mathrm{GL}_N(\mathbb{Z}_\ell) \times \mathbb{Z}_\ell^{\times}) \rtimes \{1, \mathfrak{c}\}$$

by the formula $r_{A,\ell}(\gamma) = (\operatorname{Sym}^{N-1} \rho_{A,\ell}(\gamma), \eta_v^{N-1} \epsilon_{\ell,v}^{1-N}(\gamma), \mathfrak{c}(\gamma))$, where $\mathfrak{c}(\gamma) = \mathfrak{c}$ if and only if $\gamma \in \Gamma_{F_v}^+ \setminus \Gamma_F$. Denote by $\bar{r}_{A,\ell}$ the composition of $r_{A,\ell}$ and the projection $\mathscr{G}_N(\mathbb{Z}_\ell) \to \mathscr{G}_N(\mathbb{F}_\ell)$.

Proposition E.5.12. For all but finitely many rational primes $\ell \geq N$, every lifting of $\bar{r}_{A,\ell}$ to an object R of $\mathscr{C}_{\mathbb{Z}_{\ell}}$ (with respect to the similation character $\eta_v^N \epsilon_{\ell_v}^{1-N}$) is minimally ramified in the sense of Definition E.5.8.

Proof. For simplicity, we only prove the proposition for v nonsplit in F. The split case is similar and easier, which we leave to readers. Thus, let w be the unique place of F above v. Fix a finite extension F'_w of F_w in \overline{F}_v^+ so that $A' \coloneqq A \otimes_{F_v^+} F'_w$ has either good or split multiplicative reduction. We further request that F'_w/F_w is totally ramified if A' has good reduction. Let T'_w be the image of the $\operatorname{Gal}(\overline{F}_v^+/F'_w)$ of Γ_{F_w} in $T_w = \Gamma_{F_w}/P_{F_w}$. We fix an isomorphism $T_w \simeq T_q = t^{\mathbb{Z}_\ell} \rtimes \phi_q^{\mathbb{Z}}$ with the q-tame group, where q = ||w||. We now assume $\ell > [F'_w : F_w]$. Then T'_w is generated by t and ϕ_q^b for a unique integer b > 0. We then also assume $\ell \ge q^{b \cdot N!}(>N)$. Let $\mathfrak{T} = \mathfrak{T}(\overline{r}_{A,\ell})$ be the set of isomorphism classes of absolutely irreducible representations of P_{F_w} appearing in $\overline{r}_{A,\ell}^{\natural}$ as before.

We first consider the case where A' has split multiplicative reduction. Let u be the valuation of the *j*-invariant j(A) in F'_w , which is a negative integer. Assume further that ℓ is coprime to u. Then $\rho_{A',\ell}(t)$ is conjugate to $\begin{pmatrix} 1 & 1 \\ 0 & 1 \end{pmatrix}$ in $\operatorname{GL}_2(\mathbb{Z}_\ell)$, which implies that $\operatorname{Sym}^{N-1}\rho_{A',\ell}(t)$ is conjugate to $1 + J_N$ in $\operatorname{GL}_N(\mathbb{Z}_\ell)$. It follows that \mathfrak{T} is a singleton, say $\{\tau\}$; and every lifting ρ_{τ} of $\bar{\rho}_{\tau}$ is minimally ramified. Thus, every lifting r of $\bar{r}_{A,\ell}$ is minimally ramified.

We then consider the case where A' has good reduction. Then $T'_w = T_w$ hence b = 1. Let $\alpha, \beta \in \overline{\mathbb{Q}}_{\ell}$ be the two eigenvalues of $\rho_{A',\ell}(\phi_q)$. Then α, β are Weil $q^{-1/2}$ -numbers in $\overline{\mathbb{Q}}$, which depend only on A', not on ℓ . We further assume that ℓ satisfies that $\alpha, \beta \in \overline{\mathbb{Z}}_{\ell}^{\times}$, and that the image of the set $\{(\alpha/\beta)^{N-1}, (\alpha/\beta)^{N-3}, \ldots, (\alpha/\beta)^{3-N}, (\alpha/\beta)^{1-N}\}$ in $\overline{\mathbb{F}}_{\ell}^{\times}$ does not contain q. It follows that for every $\tau \in \mathfrak{T}$, every lifting ρ_{τ} of $\bar{\rho}_{\tau}$ is actually unramified, hence minimally ramified. Thus, every lifting r of $\bar{r}_{A,\ell}$ is minimally ramified.

Since in both cases, we only exclude finitely many rational primes ℓ , the proposition follows. \Box

More generally, we would like to propose the following conjecture. As in the initial setup of Subsection 6.1, let Π be a relevant representation of $\operatorname{GL}_N(\mathbb{A}_F)$ (Definition 1.1.3), and $E \subseteq \mathbb{C}$ a strong coefficient field of Π (Definition 3.2.5). Then for every prime λ of E, we have a continuous homomorphism $\rho_{\Pi,\lambda} \colon \Gamma_F \to \operatorname{GL}_N(E_{\lambda})$.

E.6. Level-raising deformations. In this subsection, we discuss level-raising deformations. Assume $\ell \geq N \geq 2$. We take a nonarchimedean place v of F^+ that is inert in F and not above ℓ . Let w be the unique place of F above v. Recall that we have $T_v = \Gamma_{F_v^+}/P_{F_v^+}$ and $T_w = \Gamma_{F_w}/P_{F_w}$. Then T_v is isomorphic to the q-tame group and the subgroup T_w is the q^2 -tame group (Definition E.4.1), where q = ||v||.

We consider a pair (\bar{r}, χ) from Notation E.2.1 with $\tilde{\Gamma} = \Gamma_{F_v^+}$ and $\Gamma = \Gamma_{F_v^+} \cap \Gamma_F = \Gamma_{F_w}$, such that \bar{r} is unramified and $\chi = \eta_v^{\mu} \epsilon_{\ell,v}^{1-N}$ for some $\mu \in \mathbb{Z}/2\mathbb{Z}$. Then by Lemma E.5.2(1), every lifting r of \bar{r} to an object R of $\mathscr{C}_{\mathcal{O}}$ factors through T_v . In particular, we may apply the discussion in Subsection E.4 to the pair (\bar{r}, χ) .

Now assume $\ell \nmid (q^2 - 1)$ and that the generalized eigenvalues of $\overline{r}^{\natural}(\phi_w)$ in $\overline{\mathbb{F}}_{\ell}$ contain the pair $\{q^{-N}, q^{-N+2}\}$ exactly once. By Lemma E.4.7(1), for every lifting r of \overline{r} to an object R of $\mathscr{C}_{\mathscr{O}}$, we

have a canonical decomposition

(E.19)
$$R^{\oplus N} = M_0 \oplus M$$

of free R-modules such that if we write $P_0(T)$ for the characteristic polynomial for $r^{\natural}(\phi_w)$, then $P_0(T) \equiv (T - q^{-N})(T - q^{-N+2}) \operatorname{mod} \mathfrak{m}_R.$

Definition E.6.1. Let (\bar{r}, χ) be as above. We define \mathscr{D}^{mix} to be the local deformation problem of \bar{r} (Definition E.2.3) that classifies liftings r to an object R of $\mathscr{C}_{\mathscr{O}}$ such that

- O the decomposition (E.19) is stable under the action of $r^{\natural}(\mathbf{I}_{F_{w}})$;
- O the action of $r^{\natural}(\mathbf{I}_{F_w})$ on M_1 is trivial;
- **O** for every $t \in I_{F_w}$, the characteristic polynomial of $r^{\natural}(t)$ on M_0 is $(T-1)^2$.

We define

- (1) \mathscr{D}^{unr} to be the local deformation problem contained in \mathscr{D}^{mix} so that the action of $r^{\natural}(\mathbf{I}_{F_w})$ on M_0 is also trivial;
- (2) \mathscr{D}^{ram} to be the local deformation problem contained in \mathscr{D}^{mix} so that $P_0(T) = (T q^{-N})(T q^{-N})$ q^{-N+2}) in R[T].

It is clear that \mathscr{D}^{unr} coincides with \mathscr{D}^{\min} from Definition E.5.10.

Proposition E.6.2. Suppose $\ell \nmid (q^2 - 1)$ and that the generalized eigenvalues of $\overline{r}^{\natural}(\phi_w)$ in $\overline{\mathbb{F}}_{\ell}$ contain the pair $\{q^{-N}, q^{-N+2}\}$ exactly once. Then the formal scheme \mathscr{D}^{mix} is formally smooth over Spf $\mathscr{O}[[x_0, x_1]]/(x_0 x_1)$ of pure relative dimension $N^2 - 1$ such that the irreducible components defined by $x_0 = 0$ and $x_1 = 0$ are \mathscr{D}^{unr} and \mathscr{D}^{ram} , respectively. In particular, \mathscr{D}^{ram} is formally smooth over Spf \mathscr{O} of pure relative dimension N^2 .

Proof. We fix an isomorphism $T_v \simeq T_q = t^{\mathbb{Z}_\ell} \rtimes \phi_q^{\widehat{\mathbb{Z}}}$ so that $\phi_w = \phi_q^2$. We write $k^{\oplus N} = \overline{M}_0 \oplus \overline{M}_1$ so that $\bar{r}^{\natural}(\phi_q^2)$ has eigenvalues q^{-N} and q^{-N+2} on \overline{M}_0 . Without lost of generality, we may assume that \overline{M}_0 is spanned by the first two factors and \overline{M}_0 is spanned by the last N-2 factors. Thus, we obtain two unramified homomorphisms $\bar{r}_0: T_q \to \mathscr{G}_2(k)$ and $\bar{r}_1: T_q \to \mathscr{G}_{N-2}(k)$. Let \mathscr{D}_0 be the local deformation problem of \bar{r}_0 classifying liftings r_0 of \bar{r}_0 so that the characteristic polynomial of $r_0^{\natural}(t)$ is $(T-1)^2$. Let \mathscr{D}_1 be the local deformation problem of \bar{r}_1 classifying unramified liftings.

Suppose $N \geq 3$. We say that lifting r of \bar{r} to an object R of $\mathscr{C}_{\mathscr{O}}$ is standard if

$$r^{\natural}(t) = \begin{pmatrix} A_0 & 0\\ 0 & 1_{N-2} \end{pmatrix}, \quad r(\phi_q) = \left(\begin{pmatrix} B_0 & 0\\ 0 & B_1 \end{pmatrix}, (-1)^{\mu+1} q^{1-N}, \mathfrak{c} \right)$$

for some $A_0, B_0 \in \operatorname{GL}_2(R)$ and $B_1 \in \operatorname{GL}_{N-2}(R)$. Let $\mathscr{D}_{0,1}^{\min} \subseteq \mathscr{D}^{\min}$ be the locus of standard liftings. Then we have a natural isomorphism

$$\mathscr{D}_{0,1}^{\mathrm{mix}} \simeq \mathscr{D}_0 \times_{\mathrm{Spf}\,\mathscr{O}} \mathscr{D}_1$$

of formal schemes over Spf \mathcal{O} .

For $n \geq 1$, denote by \mathscr{L}_n the formal completion of $\operatorname{GL}_{n,\mathscr{O}}$ along the unit section. Then \mathscr{L}_N acts on \mathscr{D}^{mix} by conjugation. We claim that $\mathscr{D}_{0,1}^{\text{mix}}$ generates \mathscr{D}^{mix} under the action of \mathscr{L}_N . For this, it suffices to show that for every lifting r of \bar{r} to an object R of $\mathscr{C}_{\mathscr{O}}$, the maps

$$B\colon M_0 \to R^{\oplus N} \to M_1, \quad B\colon M_1 \to R^{\oplus N} \to M_0$$

induced by B from Lemma E.1.3(1) for $\gamma = \phi_q$, are both zero. Since the two maps intertwine the actions r^{\natural} and $r^{\natural,\vee} \otimes \epsilon_{\ell}^{1-N}$ of T_{q^2} , it suffices to show that the generalized eigenvalues of $r_0^{\natural,\vee} \otimes \epsilon_{\ell}^{1-N}(\phi_q^2)$ and the generalized eigenvalues of $r_1^{\natural}(\phi_a^2)$ are disjoint. However, this follows from the condition that the generalized eigenvalues of $\overline{r}^{\natural}(\phi_w)$ in $\overline{\mathbb{F}}_{\ell}$ contain the pair $\{q^{-N}, q^{-N+2}\}$ exactly once.

The above claim induces a canonical isomorphism

$$\mathscr{D}_{0,1}^{\min} \times_{\operatorname{Spf} \mathscr{O}} (\mathscr{L}_2 \times_{\operatorname{Spf} \mathscr{O}} \mathscr{L}_{N-2} \backslash \mathscr{L}_N) \xrightarrow{\sim} \mathscr{D}^{\min}.$$

By Proposition E.4.5, \mathscr{D}_1 is formally smooth over Spf \mathscr{O} of pure relative dimensions $(N-2)^2$. Since $\mathscr{L}_2 \times_{\operatorname{Spf} \mathscr{O}} \mathscr{L}_{N-2} \setminus \mathscr{L}_N$ is formally smooth over $\operatorname{Spf} \mathscr{O}$ of pure relative dimension $N^2 - (N-2)^2 - 4$, it suffices to prove the proposition for N = 2.

Now we assume N = 2. After changing a basis, we may assume

$$\bar{r}(\phi_q)(\bar{B},(-1)^{\mu+1}q^{1-N},\mathfrak{c}), \quad \bar{B} = \begin{pmatrix} 0 & (-1)^{\mu+1} \\ q & 0 \end{pmatrix}.$$

Then we have

$$\bar{r}^{\natural}(\phi_q^2) = (-1)^{\mu+1} q^{1-N} \bar{B}^{\, t} \bar{B}^{-1} = \begin{pmatrix} q^{-N} & 0\\ 0 & q^{-N+2} \end{pmatrix}.$$

For every object R of $\mathscr{C}_{\mathscr{O}}$, the set $\mathscr{D}^{\min}(R)$ is bijective to the set of pairs (B, X) where $B \in GL_2(R)$ and $X \in M_2(R)$ satisfying $B \equiv \overline{B} \mod \mathfrak{m}_R$, $X \equiv 0 \mod \mathfrak{m}_R$, that the characteristic polynomial of X is T^2 , and the relation

(E.20)
$$B^{t}XB^{-1} = -qX.$$

Indeed, the bijection is given by $r(\phi_q) = (B, (-1)^{\mu+1}q^{1-N}, \mathfrak{c})$ and $r^{\natural}(t) = 1_2 + X$. We let $\mathscr{D}_0^{\text{mix}}$ be the subscheme of \mathscr{D}^{mix} defined by the condition that $r^{\natural}(\phi_q^2) = (-1)^{\mu+1}q^{1-N}B^{\dagger}B^{-1}$ is a diagonal matrix. Take a lifting $r \in \mathscr{D}_0^{\min}(R)$ corresponding to the pair (B, X); we must have

$$B = \begin{pmatrix} 0 & (-1)^{\mu+1}(1+x) \\ q(1+y) & 0 \end{pmatrix}, \quad r^{\natural}(\phi_q^2) = \begin{pmatrix} q^{-N}(1+x)(1+y)^{-1} & 0 \\ 0 & q^{-N+2}(1+y)(1+x)^{-1} \end{pmatrix}$$

for some $x, y \in \mathfrak{m}_R$. Then by (E.20), $X = \begin{pmatrix} 0 & 0 \\ x_0 & 0 \end{pmatrix}$ for some $x_0 \in \mathfrak{m}_R$ satisfying $(x - y)x_0 = 0$. Put $x_1 \coloneqq x - y$. Then we obtain an isomorphism

$$\mathscr{D}_0^{\min} \simeq \operatorname{Spf} \mathscr{O}[[x_0, x_1, y]]/(x_0 x_1)$$

such that

 $O x_0 = 0$ if and only if r is unramified;

O $x_1 = 0$ if and only if $P_0(T) = (T - q^{-N})(T - q^{-N+2})$, where P_0 is the characteristic polynomial of $r^{\natural}(\phi_w) = r^{\natural}(\phi_a^2)$.

Finally, not that \mathscr{L}_2 acts on \mathscr{D}^{mix} by conjugation, which induces a canonical isomorphism

$$\mathscr{D}_{0}^{\mathrm{mix}} \times_{\mathrm{Spf}\,\mathscr{O}} (\mathscr{L}_{1} \times_{\mathrm{Spf}\,\mathscr{O}} \mathscr{L}_{1} \backslash \mathscr{L}_{2}) \xrightarrow{\sim} \mathscr{D}^{\mathrm{mix}}$$

The proposition (for N = 2) follows as $\mathscr{L}_1 \times_{\operatorname{Spf} \mathscr{O}} \mathscr{L}_1 \setminus \mathscr{L}_2$ is formally smooth over $\operatorname{Spf} \mathscr{O}$ of pure relative dimension 2. The entire proposition is now proved.

E.7. An almost minimal R=T theorem. In this subsection, we prove a version of the R=T theorem for a global Galois representation. Assume $\ell > N > 2$ and that ℓ is unramified in F.

We consider a pair (\bar{r}, χ) from Notation E.2.1 with $\tilde{\Gamma} = \Gamma_{F^+}$ and $\Gamma = \Gamma_F$, in which $\chi = \eta^{\mu} \epsilon_{\ell}^{1-N}$ for some $\mu \in \mathbb{Z}/2\mathbb{Z}$. We take two finite sets Σ_{\min}^+ and Σ_{\ln}^+ of nonarchimedean places of F^+ such that

- O Σ_{\min}^+ , Σ_{lr}^+ , and Σ_{ℓ}^+ are mutually disjoint;
- O Σ_{\min}^+ contains Σ_{\max}^+ ; O every place $v \in F^+$ is inert in F and satisfies $\ell \nmid (||v||^2 1)$.

Definition E.7.1. We say that \bar{r} is rigid for $(\Sigma_{\min}^+, \Sigma_{lr}^+)$ if the following are satisfied:

(1) For v in Σ_{\min}^+ , every lifting of \bar{r}_v is minimally ramified.

- (2) For v in Σ_{lr}^+ , the generalized eigenvalues of $\overline{r}_v^{\natural}(\phi_w)$ in $\overline{\mathbb{F}}_{\ell}$ contain the pair $\{\|v\|^{-N}, \|v\|^{-N+2}\}$ exactly once, where w is the unique place of F above v.
- (3) For v in Σ_{ℓ}^+ , \bar{r}_v^{\sharp} is crystalline with regular Fontaine–Laffaille weights in [0, N-1] (Definition E.3.4).
- (4) For a nonarchimedean place v of F^+ not in $\Sigma_{\min}^+ \cup \Sigma_{lr}^+ \cup \Sigma_{\ell}^+$, the homomorphism \overline{r}_v is unramified.

Suppose now that \bar{r} is rigid for $(\Sigma_{\min}^+, \Sigma_{lr}^+)$. Consider a global deformation problem (Definition E.2.6)

$$\mathscr{S} \coloneqq (\bar{r}, \eta^{\mu} \epsilon_{\ell}^{1-N}, \Sigma_{\min}^{+} \cup \Sigma_{\mathrm{lr}}^{+} \cup \Sigma_{\ell}^{+}, \{\mathscr{D}_{v}\}_{v \in \Sigma_{\min}^{+} \cup \Sigma_{\mathrm{lr}}^{+} \cup \Sigma_{\ell}^{+}})$$

where

O for $v \in \Sigma_{\min}^+$, \mathscr{D}_v is the local deformation problem classifying all liftings of \bar{r}_v ;

O for $v \in \Sigma_{\mathrm{lr}}^+$, \mathscr{D}_v is the local deformation problem $\mathscr{D}^{\mathrm{ram}}$ of \bar{r}_v from Definition E.6.1;

O for $v \in \Sigma_{\ell}^+$, \mathscr{D}_v is the local deformation problem \mathscr{D}^{FL} of \bar{r}_v from Definition E.3.6.

Then we have the global universal deformation ring $\mathsf{R}^{\mathrm{univ}}_{\mathscr{C}}$ from Proposition E.2.7.

Remark E.7.2. It is possible that \bar{r} is rigid for two pairs $(\Sigma_{\min}^+, \Sigma_{lr}^+)$ and $(\Sigma_{\min}^{+\prime}, \Sigma_{lr}^{+\prime})$. Then $\mathsf{R}^{\operatorname{univ}}_{\mathscr{S}'}$ and $\mathsf{R}^{\operatorname{univ}}_{\mathscr{S}'}$ are different in general, where \mathscr{S}' denotes the corresponding global deformation problem for $(\Sigma_{\min}^{+\prime}, \Sigma_{lr}^{+\prime})$.

Now we state an R=T theorem. Let V be a standard definite or indefinite hermitian space (Definition 3.2.1) over F of rank N, such that V_v is not split for $v \in \Sigma_{lr}^+$. We fix a self-dual $\prod_{v \notin \Sigma_{w}^+ \cup \Sigma_{lr}^+} O_{F_v}$ -lattice Λ in $V \otimes_F \mathbb{A}_F^{\Sigma_{w}^+ \cup \Sigma_{lr}^+}$, and an element $K \in \mathfrak{K}(V)$ (Definition 3.1.11) of the form

$$\mathbf{K} = \prod_{v \in \Sigma_{\min}^+ \cup \Sigma_{\mathrm{lr}}^+} \mathbf{K}_v \times \prod_{v \notin \Sigma_{\infty}^+ \cup \Sigma_{\min}^+ \cup \Sigma_{\mathrm{lr}}^+} \mathbf{U}(\Lambda)(O_{F_v^+})$$

in which K_v is special maximal for $v \in \Sigma_{lr}^+$.

Let Σ^+ be a finite set of nonarchimedean places of F^+ containing $\Sigma^+_{\min} \cup \Sigma^+_{lr}$; so we have the abstract unitary Hecke algebra $\mathbb{T}_N^{\Sigma^+}$ (Definition 3.1.9). Let $\phi \colon \mathbb{T}_N^{\Sigma^+} \to k$ be a homomorphism such that

- O for every nonarchimedean place v of F^+ not in $\Sigma^+ \cup \Sigma^+_{\ell}$ that induces one place w of F, we have $\phi|_{\mathbb{T}_{N,v}} = \phi_{\alpha}$ (Construction 3.1.8) where $\alpha = (\alpha_1, \ldots, \alpha_N)$ is the unitary abstract Hecke parameter at v (Definition 3.1.3) satisfying that $\{\alpha_1 \|v\|^{N-1}, \ldots, \alpha_N \|v\|^{N-1}\}$ are the generalized eigenvalues of $\bar{r}_v^{\natural}(\phi_w^{-1})$ in $\overline{\mathbb{F}}_{\ell}$;
- O for every nonarchimedean place v of F^+ not in $\Sigma^+ \cup \Sigma^+_{\ell}$ that splits into two places w_1 and w_2 of F, we have $\phi|_{\mathbb{T}_{N,v}} = \phi_{\alpha}$ (Construction 3.1.8) where $\boldsymbol{\alpha} = ((\alpha_{1,1}, \ldots, \alpha_{1,N}); (\alpha_{2,1}, \ldots, \alpha_{2,N}))$ is the unitary abstract Hecke parameter at v (Definition 3.1.3) satisfying that for $i = 1, 2, \{\alpha_{i,1}\sqrt{\|v\|}^{N-1}, \ldots, \alpha_{i,N}\sqrt{\|v\|}^{N-1}\}^{34}$ are the generalized eigenvalues of $\bar{r}^{\ddagger}_{v}(\phi^{-1}_{w_{i}})$ in $\overline{\mathbb{F}}_{\ell}$.

We write $\mathfrak m$ for the kernel of ϕ

Theorem E.7.3. Suppose $\Sigma_{lr}^+ = \emptyset$ if N is odd. Under the above setup, we further assume **(D1):** ℓ is unramified in F, and $\ell \ge 2(N+1)$; **(D2):** $\bar{r}^{\natural}|_{\text{Gal}(\overline{F}/F(\zeta_{\ell}))}$ is absolutely irreducible;

(D3): \bar{r} is rigid for $(\Sigma_{\min}^+, \Sigma_{\ln}^+)$ (Definition E.7.1);

(D4): ϕ is cohomologically generic (Definition D.1.1) when V is indefinite.

³⁴Recall that we have fixed an isomorphism $\iota_{\ell} \colon \mathbb{C} \xrightarrow{\sim} \overline{\mathbb{Q}}_{\ell}$ at the beginning of this section. Thus, for every positive integer q coprime to ℓ , \sqrt{q} is a well-defined element in $\overline{\mathbb{Z}}_{\ell}$ hence in $\overline{\mathbb{F}}_{\ell}$.

Let T be the image of $\mathbb{T}_N^{\Sigma^+}$ in $\operatorname{End}_{\mathscr{O}}(\operatorname{H}^{N-1}_{\operatorname{\acute{e}t}}(\operatorname{Sh}(V, \operatorname{K})_{\overline{F}}, \mathscr{O}))$ (resp. $\operatorname{End}_{\mathscr{O}}(\mathscr{O}[\operatorname{Sh}(V, \operatorname{K})]))$ when V is indefinite (resp. definite). If $\mathsf{T}_{\mathfrak{m}} \neq 0$, then we have the following:

- (1) There is a canonical isomorphism $\mathsf{R}^{\mathrm{univ}}_{\mathscr{S}} \xrightarrow{\sim} \mathsf{T}_{\mathfrak{m}}$ of local complete intersection \mathscr{O} -algebras.
- (2) When V is indefinite (resp. definite), $H^{N-1}_{\acute{e}t}(Sh(V, K)_{\overline{F}}, \mathscr{O})_{\mathfrak{m}}$ (resp. $\mathscr{O}[Sh(V, K)]_{\mathfrak{m}}$) is a finite free $\mathsf{T}_{\mathfrak{m}}$ -module.
- (3) We have $\mu \equiv N \mod 2$.

The rest of this subsection is devoted to the proof of the theorem. We will use the Taylor– Wiles patching argument following [CHT08] and [Tho12]. Put $S := \Sigma_{\min}^+ \cup \Sigma_{lr}^+ \cup \Sigma_{\ell}^+$. To prove the theorem, we may replace E_{λ} by a finite unramified extension. Thus, we may assume that kcontains all eigenvalues of matrices in $r^{\natural}(\Gamma_F)$.

Remark E.7.4. By (D1), we know that F is not contained in $F^+(\zeta_{\ell})$. Thus, by [Tho12, Theorem A.9], (D1) and (D2) imply that $\bar{r}(\operatorname{Gal}(\overline{F}/F^+(\zeta_{\ell})))$ is adequate in the sense of [Tho12, Definition 2.3].

Recall that a prime v of F^+ is called a *Taylor–Wiles prime* for the global deformation problem \mathscr{S} if

- $O v \notin S$; v splits in F; and $||v|| \equiv 1 \mod \ell$;
- O \bar{r}_v is unramified;
- $\bigcirc \bar{r}_v^{\natural}(\phi_w)$ is not a scalar and admits an eigenvalue $\bar{\alpha}_v \in k$, called *special eigenvalue*, such that $\bar{r}_v^{\natural}(\phi_w)$ acts semisimply on the generalized eigenspace for $\bar{\alpha}_v$, where w is the place of F above v induced by the inclusion $F \subseteq \overline{F}_v^+$.

A Taylor-Wiles system is a tuple $(Q, \{\bar{\alpha}_v\}_{v \in Q})$ where Q is a finite set of Taylor-Wiles primes, and $\bar{\alpha}_v$ is a special eigenvalue for every $v \in Q$. For such a system, we write $r_v^{\natural} = r_v^{\bullet} \oplus r_v^{\circ}$ for every $v \in Q$, where r_v^{\bullet} (resp. r_v°) is the generalized eigenspace for $\bar{\alpha}_v$ (resp. for generalized eigenvalues other than α_v). Then we have another global deformation problem (see [Tho12, Definition 4.1])

$$\mathscr{S}(\mathbf{Q}) \coloneqq (\bar{r}, \eta^{\mu} \epsilon_{\ell}^{1-N}, \mathbf{S} \cup \mathbf{Q}, \{\mathscr{D}_{v}\}_{v \in \mathbf{S} \cup \mathbf{Q}})$$

where \mathscr{D}_v is the same as in \mathscr{S} for $v \in S$; and for $v \in Q$, \mathscr{D}_v is the local deformation problem of \bar{r}_v that classifies liftings r_v so that r_v^{\natural} is of the form $r_v^{\bullet} \oplus r_v^{\circ}$ in which r_v^{\bullet} is a lifting of \bar{r}_v^{\bullet} on which I_{F_w} acts by scalars, and r_v° is an unramified lifting of \bar{r}_v° .

We now discuss the existence of Taylor–Wiles systems. For each $v \in S$, we have the tangent space $L(\mathscr{D}_v) \subseteq H^1(F_v^+, \operatorname{ad} \bar{r})$ from Definition E.2.4. Let $L(\mathscr{D}_v)^{\perp} \subseteq H^1(F_v^+, \operatorname{ad} \bar{r}(1))$ be the annihilator of $L(\mathscr{D}_v)$ under the local Tate duality induced by the perfect pairing $\operatorname{ad} \bar{r} \times \operatorname{ad} \bar{r}(1) \to k(1)$ sending (x, y) to $\operatorname{tr}(xy)$. Recall that $\Gamma_{F^+,S}$ is the Galois group of the maximal subextension of \overline{F}/F^+ that is unramified outside S. For every subset $T \subseteq S$, we define $H^1_{\mathscr{L}^{\perp},T}(\Gamma_{F^+,S}, \operatorname{ad} \bar{r}(1))$ to be the kernel of the natural map

$$\mathrm{H}^{1}(\Gamma_{F^{+},\mathrm{S}},\mathrm{ad}\,\bar{r}(1)) \to \bigoplus_{v\in\mathrm{S}\backslash\mathrm{T}}\mathrm{H}^{1}(F_{v}^{+},\mathrm{ad}\,\bar{r}(1))/\mathrm{L}(\mathscr{D}_{v})^{\perp}$$

Recall the \mathscr{O} -algebras $\mathsf{R}^{\mathsf{loc}}_{\mathscr{S},\mathsf{T}}$ (E.2) and $\mathsf{R}^{\Box_{\mathsf{T}}}_{\mathscr{S}(\mathsf{Q})}$ from Proposition E.2.7. Moreover, $\mathsf{R}^{\Box_{\mathsf{T}}}_{\mathscr{S}(\mathsf{Q})}$ is naturally an algebra over $\mathsf{R}^{\mathsf{loc}}_{\mathscr{S},\mathsf{T}}$.

Lemma E.7.5. Let the situation be as in Theorem E.7.3. Let T be a subset of S. For every integer $b \ge \dim_k \operatorname{H}^1_{\mathscr{L},T}(\Gamma_{F^+,S}, \operatorname{ad} \bar{r}(1))$ and every integer $n \ge 1$, there is a Taylor–Wiles system $(Q_n, \{\bar{\alpha}_v\}_{v \in Q_n})$ satisfying

(1) $|\mathbf{Q}_n| = b;$ (2) $||v|| \equiv 1 \mod \ell^n;$ (3) $\mathsf{R}_{\mathscr{I}(\mathbf{Q}_n)}^{\Box_{\mathrm{T}}}$ can be topologically generated over $\mathsf{R}_{\mathscr{I},\mathrm{T}}^{\mathrm{loc}}$ by

$$g_{b,\mathrm{T}} \coloneqq b - \sum_{v \in \mathrm{T} \cap \Sigma_{\ell}^+} [F_v^+ : \mathbb{Q}_{\ell}] \frac{N(N-1)}{2} - N[F^+ : \mathbb{Q}] \frac{1 + (-1)^{\mu+1-N}}{2}$$

elements.

Proof. By (E.1), Proposition E.3.9(3), Proposition E.5.11, and Proposition E.6.2, we have for every $v \in \mathscr{S}$ that

$$\dim_k \mathcal{L}(\mathscr{D}_v) - \dim_k \mathcal{H}^0(F_v^+, \operatorname{ad} \bar{r}(1)) = \begin{cases} [F_v^+ : \mathbb{Q}_\ell] \frac{N(N-1)}{2} & \text{if } v \mid \ell \\ 0 & \text{if } v \nmid \ell \end{cases}$$

Then the lemma follows from $[Tho 12, Proposition 4.4]^{35}$ in view of Remark E.7.4.

Now we take a Taylor–Wiles system $(Q_n, \{\bar{\alpha}_v\}_{v \in Q_n})$ as in the above lemma. For each $v \in Q_n$, we

- O put $d_v \coloneqq \dim_k \bar{r}_v^{\bullet}$;
- O let $P_{d_v} \subseteq GL_N$ be the standard upper-triangular parabolic subgroup corresponding to the partition $(N d_v, d_v)$;
- O let κ_v be the residue field of F_v^+ , and Δ_v the maximal quotient of κ_v^{\times} of ℓ -power order;
- O fix an isomorphism $K_v \simeq GL_N(O_{F_v^+})$ and denote by $K_{v,0} \subseteq K_v$ the parahoric subgroup corresponding to P_{d_v} ;
- O let $K_{v,1}$ be the kernel of the canonical map

$$\mathrm{K}_{v,0} \to \mathrm{P}_{d_v}(\kappa_v) \to \mathrm{GL}_{d_v}(\kappa_v) \xrightarrow{\mathrm{det}} \kappa_v^{\times} \to \Delta_v.$$

We then

- O put $\Delta_{\mathbf{Q}_n} := \prod_{v \in \mathbf{Q}_n} \Delta_v$; and let $\mathfrak{a}_{\mathbf{Q}_n}$ be the augmentation ideal of $\mathscr{O}[\Delta_{\mathbf{Q}_n}]$;
- $O \text{ write } \mathfrak{m}_{\mathbf{Q}_n} \text{ for the kernel of the composite homomorphism } \mathbb{T}_N^{\Sigma^+ \cup \mathbf{Q}_n} \to \mathbb{T}_N^{\Sigma^+} \xrightarrow{\phi} k;$

O or i = 0, 1, put

$$\mathbf{K}_{i}(\mathbf{Q}_{n}) = \prod_{v \notin \mathbf{Q}_{n}} \mathbf{K}_{v} \times \prod_{v \in \mathbf{Q}_{n}} \mathbf{K}_{v,i},$$

which are subgroups of K.

In particular, $K_1(Q_n)$ is a normal subgroup of $K_0(Q_n)$; and we have a canonical isomorphism (E.21) $K_0(Q_n)/K_1(Q_n) \xrightarrow{\sim} \Delta_{Q_n}$.

Now we introduce some patching module from automorphic input. For every open compact subgroup $k \in \{K, K_0(Q_n), K_1(Q_n)\}$, we put

$$H_{\mathbf{k}} \coloneqq \begin{cases} \mathscr{O}[\mathrm{Sh}(\mathbf{V},\mathbf{k})] & \text{if } \mathbf{V} \text{ is definite;} \\ \mathbf{H}_{N-1}(\mathrm{Sh}(\mathbf{V},\mathbf{k})(\mathbb{C}),\mathscr{O}) & \text{if } \mathbf{V} \text{ is indefinite.} \end{cases}$$

Here, H_{N-1} stands for the singular homology for complex manifolds. By (E.21), $H_{K_1(Q_n)}$ is canonically a module over $\mathscr{O}[\Delta_{Q_n}]$.

Lemma E.7.6. Let the situation be as in Theorem E.7.3. The $\mathscr{O}[\Delta_{Q_n}]$ -module $H_{K_1(Q_n),\mathfrak{m}_{Q_n}}$ is a finite and free. Moreover, the canonical map

$$H_{\mathrm{K}_{1}(\mathrm{Q}_{n}),\mathfrak{m}_{\mathrm{Q}_{n}}}/\mathfrak{a}_{\mathrm{Q}_{n}} \to H_{\mathrm{K}_{0}(\mathrm{Q}_{n}),\mathfrak{m}_{\mathrm{Q}_{n}}}$$

is an isomorphism.

³⁵Strictly speaking, the set S in [Tho12, Proposition 4.4] consists of only places split in F. But the same argument works in our case as well.

Proof. Note that by Definition 3.1.11, every open compact subgroup $k \in \{K, K_0(Q_n), K_1(Q_n)\}$ is neat; in particular, $t^{-1}U(V)(F^+)t \cap k$ has no torsion elements for every $t \in U(V)(\mathbb{A}_{F^+}^{\infty})$.

When V is definite, the lemma (even without localization at \mathfrak{m}_{Q_n}) follows by the same argument for [Tho12, Lemma 6.4].

Suppose now that V is indefinite. For an abelian group A, we let $C_{\bullet}(Sh(V, k)(\mathbb{C}), A)$ be the complex of singular chains for the complex manifold $Sh(V, k)(\mathbb{C})$ with coefficients in A. By Artin's comparison theorem between the singular cohomology and the étale cohomology, the dual complex $Hom_A(C_{\bullet}(Sh(V, k)(\mathbb{C}), A), A)$ calculates $H^{\bullet}_{\acute{e}t}(Sh(V, k)_{\mathbb{C}}, A)$. Now take $A = \mathscr{O}/\lambda^m$ for integers $m \geq 1$. By (D4), we know that

$$\mathrm{H}_{i}(\mathrm{Sh}(\mathrm{V},\mathrm{k})(\mathbb{C}),\mathscr{O}/\lambda^{m})_{\mathfrak{m}_{\mathrm{Q}_{n}}}=\mathrm{H}_{i}(\mathrm{C}_{\bullet}(\mathrm{Sh}(\mathrm{V},\mathrm{k})(\mathbb{C}),\mathscr{O}/\lambda^{m}))_{\mathfrak{m}_{\mathrm{Q}_{n}}}=0$$

for every $i \neq N-1$. On the other hand, by [KT17, Lemma 6.9], for every $m \geq 1$, $C_{\bullet}(Sh(V, K_1(Q_n))(\mathbb{C}), \mathscr{O}/\lambda^m)$ is a perfect complex of free $\mathscr{O}/\lambda^m[\Delta_{Q_n}]$ -modules; and there is a canonical isomorphism

$$C_{\bullet}(Sh(V, K_1(Q_n))(\mathbb{C}), \mathscr{O}/\lambda^m) \otimes_{\mathscr{O}[\Delta_{Q_n}]} \mathscr{O}[\Delta_{Q_n}] / \mathfrak{a}_{Q_n} \simeq C_{\bullet}(Sh(V, K_0(Q_n))(\mathbb{C}), \mathscr{O}/\lambda^m).$$

Then the lemma follows easily by taking the homology group and passing to the limit for $m \to \infty$.

Proof of Theorem E.7.3. When V is indefinite, by (D4) and Artin's comparison theorem between the singular cohomology and the étale cohomology, we have a canonical isomorphism $H_{\mathrm{K},\mathfrak{m}} \simeq \mathrm{Hom}_{\mathscr{O}}(\mathrm{H}^{N-1}_{\mathrm{\acute{e}t}}(\mathrm{Sh}(\mathrm{V},\mathrm{K})_{\overline{F}},\mathscr{O})_{\mathfrak{m}},\mathscr{O})$, under which $\mathrm{T}_{\mathfrak{m}}$ is identified with the image of $\mathbb{T}_{N}^{\Sigma^{+}}$ in $\mathrm{End}_{\mathscr{O}}(H_{\mathrm{K},\mathfrak{m}})$. Thus, in both cases, $H_{\mathrm{K},\mathfrak{m}}$ is a finite free \mathscr{O} -module.

First, we need to construct a canonical homomorphism $\mathsf{R}^{\mathrm{univ}}_{\mathscr{S}} \to \mathsf{T}_{\mathfrak{m}}$. It is well-known that $\mathsf{T}_{\mathfrak{m}}[1/\ell]$ is a reduced finite E_{λ} -algebra. As $H_{\mathrm{K},\mathfrak{m}}$ is a finite free \mathscr{O} -module, $\mathsf{T}_{\mathfrak{m}}$ is a reduced finite flat \mathscr{O} -algebra. Every point $x \in \operatorname{Spec} \mathsf{T}_{\mathfrak{m}}[1/\ell]$ corresponds to a relevant representation Π_x of $\operatorname{GL}_N(\mathbb{A}_F)$ (Definition 1.1.3) such that

- O the associated Galois representation ρ_{Π_x,ι_ℓ} from Proposition 3.2.4(2) is residually isomorphic to $\bar{r}^{\natural} \otimes_k \overline{\mathbb{F}}_{\ell}$ (hence residually absolutely irreducible by (D2));
- O there exists a cuspidal automorphic representation of $U(V)(\mathbb{A}_{F^+})$ satisfying $BC(\pi) \simeq \Pi$ and $\pi^{K} \neq \{0\}$.

In fact, ρ_{Π_x, ι_ℓ} comes from a continuous homomorphism

$$\rho_x \colon \Gamma_F \to \mathrm{GL}_N(\mathsf{T}_x),$$

which is a lifting of \bar{r}^{\ddagger} . By a theorem of Carayol [Car94, Théorème 2], the product homomorphism

$$\prod_{x \in \operatorname{Spec} \mathsf{T}_{\mathfrak{m}}[1/\ell]} \rho_x \colon \Gamma_F \to \operatorname{GL}_N(\prod_x \mathsf{T}_x)$$

is conjugate to some continuous homomorphism $\rho_{\mathfrak{m}}: \Gamma_F \to \operatorname{GL}_N(\mathsf{T}_{\mathfrak{m}})$ that is a lifting of \bar{r}^{\ddagger} . Moreover, by Proposition 3.2.4(2), we know that $\rho_{\mathfrak{m}}$ is (1 - N)-polarizable (Definition 2.4.7). Thus, by Lemma E.1.3, we obtain a continuous homomorphism

$$r_{\mathfrak{m}} \colon \Gamma_{F^+} \to \mathscr{G}_N(\mathsf{T}_{\mathfrak{m}})$$

satisfying $r_{\mathfrak{m}}^{\natural} = \rho_{\mathfrak{m}}$, which is a lifting of \bar{r} . We claim that $r_{\mathfrak{m}}$ satisfies the global deformation problem \mathscr{S} . Indeed, since $\Pi_{x,w}$ is unramified for nonarchimedean places w of F not above $\Sigma_{\min}^+ \cup \Sigma_{\mathrm{lr}}^+$, we know that $\bar{r}_{\mathfrak{m},v}$ belongs to $\mathscr{D}_v^{\mathrm{FL}}$ for $v \in \Sigma_{\ell}^+$ by [CH13, Theorem 3.2.3(b,c)]; and $\bar{r}_{\mathfrak{m},v}$ is unramified for $v \notin S$ by Proposition 3.2.4(2). By Lemma C.2.4 and Proposition 3.2.4(2), $\bar{r}_{\mathfrak{m},v}$ belongs to $\mathscr{D}_v^{\mathrm{ram}}$

for $v \in \Sigma_{lr}^+$.³⁶ Therefore, by the universal property of $\mathsf{R}^{\text{univ}}_{\mathscr{S}}$, we obtain a canonical homomorphism

(E.22)
$$\varphi \colon \mathsf{R}^{\mathrm{univ}}_{\mathscr{S}} \to \mathsf{T}_{\mathfrak{m}}$$

of \mathcal{O} -algebras. Moreover, it is clear that our homomorphism $r_{\mathfrak{m}}$ satisfies [CHT08, Proposition 3.4.4](2,3) as well, which implies that φ is surjective. Thus, it remains to show that φ is injective.

We follow the strategy for [Tho12, Theorem 6.8]. We take an integer $n \ge 1$, and a Taylor–Wiles system $(Q_n, \{\bar{\alpha}_v\}_{v \in Q_n})$ from Lemma E.7.5. For each $v \in Q_n$, we

- O let $\operatorname{Art}_v \colon F_v^{\times} \to \Gamma_{F_v^+}^{\operatorname{ab}}$ be the local Artin map;
- O let $\varpi_v \in F_v^+$ be the uniformizer such that $\operatorname{Art}_v(\varpi_v)$ coincides with the image of ϕ_v^{-1} in $\Gamma_{F_v^+}^{\operatorname{ab}}$;
- O let $pr_{\overline{\omega}_{n}}$ be the commuting projection defined in [Tho12, Propositions 5.9 & 5.12];
- O for every $\alpha \in O_{F_v^+}^{\times}$, let $\mathbb{V}_v^{\alpha} \in \mathbb{Z}[\mathbb{K}_{v,1} \setminus \mathbb{K}_{v,0}/\mathbb{K}_{v,1}]$ be the characteristic function of the double coset

$$\mathbf{K}_{v,1} \begin{pmatrix} \mathbf{1}_{N-1} & \mathbf{0} \\ \mathbf{0} & \alpha \end{pmatrix} \mathbf{K}_{v,1}$$

For i = 0, 1, we put

$$M_{i,\mathbf{Q}_n} \coloneqq \left(\prod_{v \in \mathbf{Q}_n} \mathrm{pr}_{\varpi_v}\right) H_{\mathrm{K}_i(\mathbf{Q}_n),\mathfrak{m}_{\mathbf{Q}_n}},$$

and let $\mathsf{T}_{i,\mathbf{Q}_n}$ be the image of $\mathbb{T}_N^{\Sigma^+\cup\mathbf{Q}_n}$ in $\mathrm{End}_{\mathscr{O}}(M_{i,\mathbf{Q}_n})$. We also put

$$M \coloneqq H_{\mathrm{K},\mathfrak{m}}$$

Then the canonical map $M \to H_{K,\mathfrak{m}_{Q_n}}$ is an isomorphism, hence we obtain canonical surjective homomorphisms

$$\mathsf{T}_{1,\mathbf{Q}_n} \twoheadrightarrow \mathsf{T}_{0,\mathbf{Q}_n} \twoheadrightarrow \mathsf{T}_{\mathfrak{m}}$$

of \mathcal{O} -algebras. Similar to $\mathsf{T}_{\mathfrak{m}}$, we obtain a continuous homomorphism

$$r_{i,\mathbf{Q}_n}\colon \Gamma_{F^+} \to \mathscr{G}_N(\mathsf{T}_{i,\mathbf{Q}_n}),$$

which is a lifting of \bar{r} , for i = 0, 1. We have the following two claims:

- (1) For every $v \in Q_n$, there is a continuous character $\mathbf{v}_v \colon O_{F_*}^{\times} \to \mathsf{T}_{1,Q_n}^{\times}$ such that
 - (a) for every $\alpha \in O_{F^+}^{\times}$, the actions of \mathbb{V}_v^{α} and $\mathfrak{v}_v(\alpha)$ on M_{1,\mathbb{Q}_n} coincide;
 - (b) $r_{1,Q_n,v}^{\natural}$ has a (unique) decomposition $r_{1,Q_n,v}^{\bullet} \oplus r_{1,Q_n,v}^{\circ}$ such that $r_{1,Q_n,v}^{\bullet}$ is a lifting of \bar{r}_v^{\bullet} on which $I_{F_v^+}$ acts via the character $\mathbf{v}_v \circ \operatorname{Art}_v^{-1}$, and $r_{1,Q_n,v}^{\circ}$ is an unramified lifting of \bar{r}_v° .
- (2) The composite map

$$M = H_{\mathcal{K},\mathfrak{m}_{\mathcal{Q}_n}} \to H_{\mathcal{K}_0(\mathcal{Q}_n),\mathfrak{m}_{\mathcal{Q}_n}} \xrightarrow{\prod_{v \in \mathcal{Q}_n} \operatorname{pr}_{\varpi_v}} M_{0,\mathcal{Q}_n}$$

is an isomorphism. In particular, the canonical homomorphism $\mathsf{T}_{0,\mathbf{Q}_n} \to \mathsf{T}_{\mathfrak{m}}$ is an isomorphism; and r_{0,\mathbf{Q}_n} and $r_{\mathfrak{m}}$ are equivalent liftings of \bar{r} .

³⁶This is not correct if N is odd, which is the only reason that we suppose $\Sigma_{lr}^+ = \emptyset$ if N is odd in the statement of the theorem.

Indeed, these claims follow easily from [Tho12, Propositions 5.9 & 5.12].

It follows from (1) that r_{1,Q_n} satisfies the global deformation problem $\mathscr{S}(Q_n)$, which induces a canonical surjective homomorphism

$$\varphi_n \colon \mathsf{R}^{\mathrm{univ}}_{\mathscr{S}(\mathbf{Q}_n)} \twoheadrightarrow \mathsf{T}_{1,\mathbf{Q}_n}$$

of \mathscr{O} -algebras. Now we claim that φ_n is naturally a homomorphism of $\mathscr{O}[\Delta_{\mathbf{Q}_n}]$ -algebras. Indeed, take a universal lifting $r_{\mathscr{S}(\mathbf{Q}_n)}^{\mathrm{univ}}$ for \bar{r} over $\mathsf{R}_{\mathscr{S}(\mathbf{Q}_n)}^{\mathrm{univ}}$. Then for each $v \in \mathbf{Q}_n$, there is a unique character $\mathbf{v}_v^{\mathrm{univ}} \colon \Delta_v \to (\mathsf{R}_{\mathscr{S}(\mathbf{Q}_n)}^{\mathrm{univ}})^{\times}$ such that $\mathbf{I}_{F_v^+}$ acts on $r_{\mathscr{S}(\mathbf{Q}_n),v}^{\mathrm{univ},\bullet}$ via the character

$$I_{F_v^+} \xrightarrow{\operatorname{Art}_v^{-1}} O_{F_v^+}^{\times} \to \kappa_v^{\times} \to \Delta_v \xrightarrow{\mathfrak{v}_v^{\operatorname{univ}}} (\mathsf{R}_{\mathscr{S}(\mathbf{Q}_n)}^{\operatorname{univ}})^{\times}.$$

Then $\mathsf{R}^{\mathrm{univ}}_{\mathscr{S}(\mathbf{Q}_n)}$ becomes an $\mathscr{O}[\Delta_{\mathbf{Q}_n}]$ -algebra via the character $\prod_{v \in \mathbf{Q}_n} \mathbf{v}^{\mathrm{univ}}_v \colon \Delta_{\mathbf{Q}_n} \to (\mathsf{R}^{\mathrm{univ}}_{\mathscr{S}(\mathbf{Q}_n)})^{\times}$. Moreover, φ_n is a homomorphism of $\mathscr{O}[\Delta_{\mathbf{Q}_n}]$ -algebras. By (2) and Lemma E.7.6, we obtain a canonical commutative diagram

$$\begin{array}{c} \mathsf{R}^{\mathrm{univ}}_{\mathscr{S}(\mathbf{Q}_n)} / \mathfrak{a}_{\mathbf{Q}_n} & \xrightarrow{\sim} & \mathsf{R}^{\mathrm{univ}}_{\mathscr{S}} \\ \varphi_n / \mathfrak{a}_{\mathbf{Q}_n} & & & \downarrow^{\varphi} \\ \mathsf{T}_{1,\mathbf{Q}_n} / \mathfrak{a}_{\mathbf{Q}_n} & \xrightarrow{\sim} & \mathsf{T}_{0,\mathbf{Q}_n} & \xrightarrow{\sim} & \mathsf{T}_{\mathfrak{m}} \end{array}$$

of \mathcal{O} -algebras where all horizontal arrows are isomorphisms.

Now we fix a subset $T \subseteq S$ of cardinality t. Choose universal liftings

$$r^{\mathrm{univ}}_{\mathscr{S}} \colon \Gamma_{F^+} \to \mathscr{G}_N(\mathsf{R}^{\mathrm{univ}}_{\mathscr{S}}), \quad r^{\mathrm{univ}}_{\mathscr{S}(\mathbf{Q}_n)} \colon \Gamma_{F^+} \to \mathscr{G}_N(\mathsf{R}^{\mathrm{univ}}_{\mathscr{S}(\mathbf{Q}_n)})$$

for \bar{r} over $\mathsf{R}^{\text{univ}}_{\mathscr{S}}$ and $\mathsf{R}^{\text{univ}}_{\mathscr{S}(\mathbf{Q}_n)}$, respectively, such that $r^{\text{univ}}_{\mathscr{S}} = r^{\text{univ}}_{\mathscr{S}(\mathbf{Q}_n)} \mod \mathfrak{a}_{\mathbf{Q}_n}$. By Proposition E.2.7(2), we obtain isomorphisms

$$\mathsf{R}^{\mathrm{univ}}_{\mathscr{S}}[[X_{v;i,j}]]_{v\in\mathrm{T};1\leq i,j\leq N}\xrightarrow{\sim}\mathsf{R}^{\Box_{\mathrm{T}}}_{\mathscr{S}},\quad\mathsf{R}^{\mathrm{univ}}_{\mathscr{S}(\mathrm{Q}_n)}[[X_{v;i,j}]]_{v\in\mathrm{T};1\leq i,j\leq N}\xrightarrow{\sim}\mathsf{R}^{\Box_{\mathrm{T}}}_{\mathscr{S}(\mathrm{Q}_n)}$$

of \mathcal{O} -algebras. In particular, we have a surjective homomorphism $\mathsf{R}^{\Box_{\mathrm{T}}}_{\mathscr{S}} \to \mathsf{R}^{\mathrm{univ}}_{\mathscr{S}}$, which makes $\mathsf{R}^{\mathrm{univ}}_{\mathscr{S}}$ an algebra over $\mathsf{R}^{\mathrm{loc}}_{\mathscr{S},\mathrm{T}}$.

We put

$$S_{\infty} \coloneqq \mathscr{O}[[X_{v;i,j}]]_{v \in \mathrm{T}; 1 \le i,j \le N}[[Y_1, \dots, Y_b]]$$

and let $\mathfrak{a}_{\infty} \subseteq S_{\infty}$ be the augmentation ideal. Put

$$R_{\infty} \coloneqq \mathsf{R}^{\mathrm{loc}}_{\mathscr{S},\mathrm{T}}[[Z_1,\ldots,Z_{g_{b,\mathrm{T}}}]]$$

where $g_{b,T}$ is the number appeared in Lemma E.7.5. Applying the usual patching lemma (see the proof of [BLGG11, Theorem 3.6.1], or [Tho12, Lemma 6.10]), we have the following:

O There exists a homomorphism $S_{\infty} \to R_{\infty}$ of \mathscr{O} -algebras so that we have an isomorphism $R_{\infty}/\mathfrak{a}_{\infty}R_{\infty} \simeq \mathsf{R}^{\mathrm{univ}}_{\mathscr{S},\mathrm{T}}$ -algebras.

O There exist an R_{∞} -module M_{∞} and an isomorphism $M_{\infty}/\mathfrak{a}_{\infty}M_{\infty} \simeq M$ of $\mathsf{R}^{\mathrm{univ}}_{\mathscr{S}}$ -modules.

O As an S_{∞} -module, M_{∞} is finite and free.

In particular, we have

$$\operatorname{depth}_{R_{\infty}}(M_{\infty}) \ge \dim S_{\infty} = 1 + |\mathbf{T}|N^2 + b.$$

On the other hand, by Proposition E.3.9, Proposition E.5.11(3), and Proposition E.6.2, we know that $\mathsf{R}^{\mathsf{loc}}_{\mathscr{S},\mathsf{T}}$ is a formal power series ring over \mathscr{O} in

$$|\mathbf{T}|N^2 + \sum_{v \in \mathbf{T} \cap \Sigma_{\ell}^+} [F_v^+ : \mathbb{Q}_{\ell}] \frac{N(N-1)}{2}$$

variables. It follows that R_{∞} is a regular local ring of dimension

$$1 + |\mathbf{T}|N^2 + \sum_{v \in \mathbf{T} \cap \Sigma_{\ell}^+} [F_v^+ : \mathbb{Q}_{\ell}] \frac{N(N-1)}{2} + g_{b,\mathrm{T}} = 1 + |\mathbf{T}|N^2 + b - N[F^+ : \mathbb{Q}] \frac{1 + (-1)^{\mu+1-N}}{2}$$

As dim $R_{\infty} \geq \text{depth}_{R_{\infty}}(M_{\infty})$, we obtain Theorem E.7.3(3). By the Auslander–Buchsbaum theorem, M_{∞} is a finite free R_{∞} -module. Thus, M is a finite free $\mathbb{R}_{\mathscr{S}}^{\text{univ}}$ -module. In particular, the surjective homomorphism φ (E.22) is injective hence an isomorphism. Theorem E.7.3(1,2) are proved.

E.8. Rigidity of automorphic Galois representations. In this subsection, we study the rigidity property for reduction of automorphic Galois representations. Let us take the initial setup of Subsection 6.1, hence let Π be a relevant representation of $\operatorname{GL}_N(\mathbb{A}_F)$ (Definition 1.1.3) for $N \geq 2$, and $E \subseteq \mathbb{C}$ a strong coefficient field of Π (Definition 3.2.5). Then for every prime λ of E, we have a continuous homomorphism $\rho_{\Pi,\lambda} \colon \Gamma_F \to \operatorname{GL}_N(E_{\lambda})$.

Conjecture E.8.1. Let Π and E be as above. Fix a finite set Σ^+ of nonarchimedean place of F^+ containing Σ^+_{Π} (Notation 3.1.4). Then for all but finitely many primes λ of E, we have

- (1) $\rho_{\Pi,\lambda}$ is residually absolutely irreducible (so we have the residual homomorphism $\bar{\rho}_{\Pi,\lambda}$ and may put $\bar{r}_{\Pi,\lambda} \coloneqq \bar{\rho}_{\Pi,\lambda,+}$ from Remark 6.1.6), with the similitude character $\chi_{\lambda} \coloneqq \eta_{F/F^+}^N \epsilon_{\ell}^{1-N}$;
- (2) $\bar{\rho}_{\Pi,\lambda}|_{\mathrm{Gal}(\overline{F}/F(\zeta_{\ell}))}$ is absolutely irreducible, where ℓ is the underlying rational prime of λ ;
- (3) $\bar{r}_{\Pi,\lambda}$ is rigid for $(\Sigma_{\Pi}^+, \emptyset)$ (Definition E.7.1).

Remark E.8.2. When N = 2, Conjecture E.8.1 is not hard to verify. In fact, if the coefficient field of Π is \mathbb{Q} , then it follows from Proposition E.5.12 and Serre's theorem on the image of residual Galois representations of elliptic curves [Ser72].

Concerning Conjecture E.8.1(1), we have the following lemma.

Lemma E.8.3. Let Π and E be as above. Suppose that there exists a nonarchimedean place w of F such that Π_w is supercuspidal. Then there exists a finite set Λ_1 of primes of E depending on Π_w only, such that for every $\lambda \notin \Lambda_1$, $\rho_{\Pi,\lambda}$ is residually absolutely irreducible.

Proof. Let W_{F_w} be the Weil group of F_w . Since Π_w is supercuspidal, we have the induced continuous representation $\rho_{\Pi_w} \colon W_{F_w} \to \operatorname{GL}_N(\mathbb{C})$ via the local Langlands correspondence, which is irreducible. Fix an arithmetic Frobenius element ϕ_w in W_{F_w} . We have a natural quotient map $W_{F_w} \to \mathbb{Z}$ sending ϕ_w to 1. For every integer $b \ge 1$, let $W^b_{F_w}$ be the inverse image of $b\mathbb{Z}$. Then there exist an absolutely irreducible representation τ of I_{F_w} and a character χ of $W^b_{F_w}$, such that the underlying representation of ρ_{Π_w} is isomorphic to $\operatorname{Ind}_{W^b_{F_w}}^{W_{F_w}} \tau \otimes \chi$, where b is the smallest positive integer satisfying $\tau^{\phi_w^b} \simeq \tau$. We may choose a finite extension E' of E inside \mathbb{C} , and a finite set Λ' of primes of E', such that both τ and χ are defined over $O_{E',(\Lambda')}$. In particular, the image of ρ_{Π_w} is contained in $\operatorname{GL}_N(O_{E',(\Lambda')})$, up to conjugation. We claim that there exists a finite set Λ'_1 of primes of E' containing Λ' , such that the composite map

$$\rho_{\Pi_w} \colon W_{F_w} \to \operatorname{GL}_N(O_{E',(\Lambda')}) \to \operatorname{GL}_N(O_{E'}/\lambda')$$

is absolutely irreducible for $\lambda' \notin \Lambda'_1$. Now let Λ_1 be the set of primes of E underlying Λ'_1 . Then the lemma follows by Proposition 3.2.4(2).

It remains to show the claim. In fact, let Λ'_1 be the smallest set of primes of E' containing Λ' such that every $\lambda' \notin \Lambda'_1$ satisfies

- O the underlying rational prime does not divide $b|I_{F_w}/\ker\rho_{\Pi_w}|$;
- O $\bar{\tau}_{\lambda'} \coloneqq \tau \otimes_{O_{E'}(\Lambda')} O_{E'}/\lambda'$ remains absolutely irreducible;

O *b* remains the smallest positive integer that satisfies $\bar{\tau}_{\lambda'}^{\phi_w^b} \simeq \bar{\tau}_{\lambda'}$.

Then Λ'_1 is a finite set, meeting the requirement in the claim. The lemma is proved.

Concerning the entire Conjecture E.8.1, we have the following theorem.

Theorem E.8.4. Let Π and E be as above. Suppose that there exist two nonarchimedean places w_c and w_s of F such that Π_{w_c} is supercuspidal and Π_{w_s} is a twist of the Steinberg representation, respectively, in which w_s is not above Σ^+_{ram} . Then Conjecture E.8.1 holds for Π and E.

The rest of this subsection is devoted to the proof of the theorem. Let v_c and v_s be the primes of F^+ underlying w_c and w_s , respectively, both of which are in Σ_{Π}^+ . Without lost of generality, we may assume that w_c and w_s are induced by the embedding $F \hookrightarrow \overline{F}_{v_c}^+$ and $F \hookrightarrow \overline{F}_{v_s}^+$, respectively. Let p_s be the underlying rational prime of w_s , which is then unramified in F.

Lemma E.8.5. Under the situation of Theorem E.8.4, there exists a finite set Λ_2 of primes of E containing Λ_1 such that for every λ not in Λ_2 , the restriction $\bar{\rho}_{\Pi,\lambda}|_{\text{Gal}(\overline{F}/F(\zeta_{\ell}))}$ remains absolutely irreducible, where ℓ is the underlying rational prime of λ .

Proof. By twisting Π with a Dirichlet character, we may assume that Π_{w_s} is just the Steinberg representation.

We first use the integral model of certain Shimura variety to compute the reduction of the monodromy operator at w_s . Choose a finite extension $F \subseteq \check{F} \subseteq \mathbb{C}$ satisfying

- O \check{F} is a CM field, with $\check{F}^+ \coloneqq \check{F}^{c=1}$;
- $\bigcirc \breve{F}/F$ is Galois and soluble;
- O \breve{F} contains an imaginary quadratic field;
- $O[\breve{F}^+:\mathbb{Q}]$ is even;
- O p_s is unramified in \breve{F} ;
- O the extension \breve{F}/\breve{F}^+ is unramified at all nonarchimedean places;
- O every place of \breve{F}^+ above p splits in \breve{F} if p underlies Σ_{Π}^+ ;
- O w_c splits completely in \check{F} ;
- O for every prime w of F not above v_c , the local base change of Π_w to $\check{F}_{\check{w}}$ has nonzero Iwahori fixed vectors for every prime \check{w} of \check{F} above w.

Let \check{w}_s be the prime of \check{F} induced by the embedding $\check{F} \hookrightarrow \overline{F}_{v_s}^+$ which is above w_s , and \check{v}_s the prime of \check{F}^+ underlying \check{w}_s . Let $\check{\Pi}$ be the base change of Π to \check{F} by [AC89], which is again a relevant representation of $\operatorname{GL}_N(\mathbb{A}_{\check{F}})$. Moreover, we have $\rho_{\check{\Pi},\lambda} = \rho_{\Pi,\lambda}|_{\operatorname{Gal}(\overline{F}/\check{F})}$ for every prime λ of E, and that $\rho_{\check{\Pi},\lambda}$ is absolutely irreducible for every λ not in Λ_1 from Lemma E.8.3 (with $w = w_c$).

We fix following data

- O a central division algebra B over \breve{F} of dimension N^2 as in [HT01, Section I.7], whose invariants at \breve{w}_s and \breve{w}_s^c are 1/N and -1/N, respectively, and that splits at all other places of \breve{F} ;
- O an element $\beta \in B^{\times}$ as in [HT01, Lemma I.7.1] with τ the default archimedean place, which gives rise to a reductive group $G \coloneqq G_{\beta}$ over \mathbb{Q} ;
- O a cuspidal automorphic representation π of $G(\mathbb{A})$ (as in [HT01, Theorem VI.2.1]) satisfying - for every rational prime p not underlying Σ_{Π}^+ , π_p is unramified;
 - $\operatorname{BC}(\pi) = (\psi, \Pi_B)$, where Π_B is the Jacquet–Langlands transfer of Π to $(B^{\operatorname{op}})^{\times}$, and ψ is a Dirichlet character that is unramified at p_s ;
 - $-\pi^{\infty}$ is defined over E (by possibly replacing E by a finite extension in \mathbb{C});
- O a decomposable open compact subgroup U of $G(\mathbb{A}^\infty)$ satisfying
 - -U is neat;

- $-\pi^{U} \neq \{0\};$
- $-U_p$ is hyperspecial maximal for p not underlying Σ_{Π}^+ ;
- U_{p_s} is of the form $\mathbb{Z}_{p_s}^{\times} \times U_{\check{v}_s} \times U_{p_s}^{\check{v}_s}$ in which $U_{\check{v}_s}$ is the maximal open compact subgroup of $B_{\check{w}_s}^{\times}$.

Then we obtain a Shimura variety X_U over \check{F} as in [HT01, Section III.1], which is projective and smooth of relative dimension N-1. Moreover, $X_U \otimes_{\check{F}} \check{F}_{\check{w}_s}$ admits a natural projective integral model X_U over $O_{\check{F}_{\check{w}_s}}$, which is strictly semistable, whose strata on the special fiber are strata of the reduced subscheme of Drinfeld's formal upper half space of relative dimension N-1 (see, for example, [Th014, Theorem 6.2]). For every prime λ of E, we have the associated weight spectral sequence $E_{s,\lambda}^{p,q}$ converging to $H_{\acute{e}t}^{p+q}(X_U \otimes_{\check{F}} \bar{F}, O_\lambda)$, together with a monodromy map $\mu_{\lambda} \colon E_{1,\lambda}^{\bullet,\bullet} \to E_{1,\lambda}^{\bullet+2,\bullet-2}$. Now since strata of the reduced subscheme of Drinfeld's formal upper half space satisfy [It005, Assumption 2.1] by the proof of [It005, Theorem 1.1], we have a canonical finite bicomplex $E_1^{p,q}$ of finite free O_E -modules with only horizontal differentials together with a monodromy map $\mu \colon E_{1,O_k}^{\bullet,\bullet} \to E_{1,O_k}^{\bullet+2,\bullet-2}$, such that for every prime λ of E, $E_{1,O_E}^{p,q} \otimes_{O_E} O_\lambda$ is canonically isomorphic to $E_{1,O_k}^{p,q}$ as bicomplexes in $Mod(O_\lambda)$, under which $\mu \otimes_{O_E} O_\lambda$ coincides with μ_{λ} . Let $E_2^{p,q}$ be the cohomology of $E_1^{p,q}$ (under horizontal differentials). Let Λ_{mot} be the support of the finite O_E -module $\bigoplus_{p,q} (E_2^{p,q})_{tor}$, which is a finite set of primes of E. Then for every prime $\lambda \notin \Lambda_{mot}$, the spectral sequence $E_{s,\lambda}^{p,q}$ degenerates at the second page by the incompatibility of weights, similarly to the proof of [Tho14, Proposition 6.5]. Moreover, if we put

$$\mathrm{H}^{n}_{\mathrm{mot}}(X_{U}, O_{E}) \coloneqq \bigoplus_{p+q=n} \mathrm{E}_{2}^{p,q}$$

with the induced monodromy map $\mu_{\text{mot}} \colon \operatorname{H}^n_{\text{mot}}(X_U, O_E) \to \operatorname{H}^n_{\text{mot}}(X_U, O_E)$, then for every $\lambda \notin \Lambda_{\text{mot}}$, we have a canonical isomorphism $\operatorname{H}^n_{\text{mot}}(X_U, O_E) \otimes_{O_E} O_\lambda \simeq \operatorname{H}^n_{\text{\acute{e}t}}(X_U \otimes_{\breve{F}} \overline{F}, O_\lambda)$ such that there exists a topological generator t_λ in the \mathbb{Z}_{ℓ} -quotient of $\operatorname{I}_{\breve{F}_{\breve{w}_s}}$ whose action on $\operatorname{H}^n_{\text{\acute{e}t}}(X_U \otimes_{\breve{F}} \overline{F}, O_\lambda)$ is given by $1 + \mu_{\text{mot}}$.

Put $\operatorname{H}^{n}_{\operatorname{mot}}(X_{U}, \mathbb{C}) \coloneqq \operatorname{H}^{n}_{\operatorname{mot}}(X_{U}, O_{E}) \otimes_{O_{E}} \mathbb{C}$. Note that the Hecke algebra $\mathbb{Z}[U^{p} \setminus G(\mathbb{A}^{\infty,p})/U^{p}]$ acts on $\operatorname{E}^{p,q}_{1}$ hence on $\operatorname{H}^{n}_{\operatorname{mot}}(X_{U}, \mathbb{C})$. Let $\operatorname{H}^{n}_{\operatorname{mot}}(X_{U}, \mathbb{C})[\pi]$ be the $\pi^{\infty,p}$ -isotypic part of $\operatorname{H}^{n}_{\operatorname{mot}}(X_{U}, \mathbb{C})$, and put $\operatorname{H}^{n}_{\operatorname{mot}}(X_{U}, O_{E})[\pi] \coloneqq \operatorname{H}^{n}_{\operatorname{mot}}(X_{U}, \mathbb{C})[\pi] \cap \operatorname{H}^{n}_{\operatorname{mot}}(X_{U}, O_{E})$. For every $\lambda \notin \Lambda_{\operatorname{mot}}$, we put $\operatorname{H}^{n}_{\operatorname{mot}}(X_{U}, O_{\lambda})[\pi] \coloneqq \operatorname{H}^{n}_{\operatorname{mot}}(X_{U}, O_{E})[\pi] \otimes_{O_{E}} O_{\lambda}$, which is a subspace of $\operatorname{H}^{n}_{\operatorname{\acute{e}t}}(X_{U} \otimes_{\breve{F}} \overline{F}, O_{\lambda})$ stable under the action of $\operatorname{Gal}(\overline{F}/\breve{F})$, and is nonzero if and only if n = N - 1. Now we characterize $\operatorname{H}^{n}_{\operatorname{mot}}(X_{U}, O_{\lambda})[\pi]$ as an $O_{\lambda}[\operatorname{Gal}(\overline{F}/\breve{F})]$ -module for $\lambda \notin \Lambda_{1} \cup \Lambda_{\operatorname{mot}}$. For $\lambda \notin \Lambda_{1} \cup \Lambda_{\operatorname{mot}}, \rho_{\breve{\Pi},\lambda}$ is residually absolutely irreducible, for which we may fix a $\operatorname{Gal}(\overline{F}/\breve{F})$ -stable lattice $\operatorname{R}_{\lambda}$, which is a free O_{λ} -module of rank N. Now we let $\operatorname{R}^{\psi}_{\lambda}$ be the $O_{\lambda}[\operatorname{Gal}(\overline{F}/\breve{F})]$ -module on which $\operatorname{Gal}(\overline{F}/\breve{F})$ acts by $\rho_{\breve{\Pi},\lambda}$ twisted by the character $\operatorname{res}(\psi)|_{\operatorname{Gal}(\overline{F}/\breve{F})}^{-1}$ as in [HT01, Corollary VII.1.10], which is again residually absolutely irreducible. By [HT01, Corollary VII.1.10], we obtain an isomorphism

(E.23)
$$\mathrm{H}^{n}_{\mathrm{mot}}(X_{U}, O_{\lambda})[\pi] \otimes_{O_{\lambda}} E_{\lambda} \simeq (\mathrm{R}^{\psi}_{\lambda} \otimes_{O_{\lambda}} E_{\lambda})^{\oplus n}$$

of $E_{\lambda}[\operatorname{Gal}(\overline{F}/\check{F})]$ -modules, for some positive integer m independent of λ . Since $\check{\Pi}_{\check{w}_s}$ is the Steinberg representation and $\operatorname{res}(\psi)|_{\operatorname{Gal}(\overline{F}/\check{F})}^{-1}$ is unramified at p_s , by Proposition 3.2.4(2), the restriction endomorphism $1 + \mu_{\text{mot}}$ of $\operatorname{H}^n_{\text{mot}}(X_U, O_E)[\pi]$ is conjugate to $(1 + J_N)^{\oplus m}$. Thus, there exists a finite set Λ_2 of primes of E containing $\Lambda_1 \cup \Lambda_{\text{mot}}$ and all those λ whose underlying rational prime ramifies in \check{F} , such that for every $\lambda \notin \Lambda_2$, the reduction of $1 + \mu_{\text{mot}}$ as, an endomorphism of $\operatorname{H}^n_{\text{mot}}(X_U, O_E)[\pi] \otimes_{O_E} O_E/\lambda$, is again conjugate to $(1 + J_N)^{\oplus m}$. By (E.23), $\bar{\mathrm{R}}^{\psi}_{\lambda}$ is an $O_{\lambda}/\lambda[\operatorname{Gal}(\overline{F}/\check{F})]$ -submodule of $\operatorname{H}^n_{\text{mot}}(X_U, O_E)[\pi] \otimes_{O_E} O_E/\lambda$. Thus, the image of $\operatorname{Gal}(\overline{F}/\check{F}(\zeta_\ell))$ in End $(\bar{\mathrm{R}}^{\psi}_{\lambda})$, hence $\bar{\rho}_{\check{\mathrm{II}},\lambda}(\operatorname{Gal}(\overline{F}/\check{F}(\zeta_\ell)))$, contain a unipotent matrix that is conjugate to $1 + J_N$. In particular, $\bar{\rho}_{\check{\Pi},\lambda}|_{\operatorname{Gal}(\overline{F}/\check{F}(\zeta_{\ell}))}$ is absolutely indecomposable. On the other hand, since $\operatorname{Gal}(\overline{F}/\check{F}(\zeta_{\ell}))$ is a normal subgroup of $\operatorname{Gal}(\overline{F}/\check{F})$ of finite index, $\bar{\rho}_{\check{\Pi},\lambda}|_{\operatorname{Gal}(\overline{F}/\check{F}(\zeta_{\ell}))}$ is absolutely semisimple, hence has to be absolutely irreducible. Thus, $\bar{\rho}_{\Pi,\lambda}|_{\operatorname{Gal}(\overline{F}/F(\zeta_{\ell}))}$ remains absolutely irreducible for $\lambda \notin \Lambda_2$. \Box

Proof of Theorem E.8.4. Let Λ_2 be the set in Lemma E.8.5. It suffices to study (3) in Theorem E.8.4. We take a prime λ of E not in Λ_2 , whose underlying rational prime ℓ does not underlie Σ^+ , and satisfies $\ell \geq 2(N+1)$. In particular, we have

- (a) ℓ is unramified in F;
- (b) Π_w is unramified for every place w of F above ℓ ;
- (c) $\bar{\rho}_{\Pi,\lambda}|_{\operatorname{Gal}(\overline{F}/F(\zeta_{\ell}))}$ is absolutely irreducible, which implies that $\bar{r}_{\Pi,\lambda}(\operatorname{Gal}(\overline{F}/F^+(\zeta_{\ell})))$ is adequate by Remark E.7.4;
- (d) Proposition E.3.9 holds for the local deformation problem $\mathscr{D}^{\mathrm{FL}}$ of $\bar{r}_{\Pi,\lambda,v}$ for every $v \in \Sigma_{\ell}^+$;
- (e) Proposition E.5.11 holds for $\bar{r}_{\Pi,\lambda,v}$ for every $v \in \Sigma^+$.

For a collection $\mathscr{D}_{\Sigma^+} = \{\mathscr{D}_v \mid v \in \Sigma^+\}$ in which \mathscr{D}_v is an irreducible component of Spf $\mathsf{R}^{\mathrm{loc}}_{\bar{r}_{\Pi,\lambda,v}}$ for $v \in \Sigma^+$, we define a global deformation problem (Definition E.2.6)

$$\mathscr{S}(\mathscr{D}_{\Sigma^+}) \coloneqq (\bar{r}_{\Pi,\lambda}, \eta^{\mu}_{F/F^+} \epsilon^{1-N}_{\ell}, \Sigma^+ \cup \Sigma^+_{\ell}, \{\mathscr{D}_v\}_{v \in \Sigma^+ \cup \Sigma^+_{\ell}})$$

where for $v \in \Sigma^+$, \mathscr{D}_v is the prescribed irreducible component (which is a local deformation problem by Proposition E.5.11(2)) in \mathscr{D}_{Σ^+} ; and for $v \in \Sigma_{\ell}^+$, \mathscr{D}_v is the local deformation problem $\mathscr{D}^{\mathrm{FL}}$ of $\bar{r}_{\Pi,\lambda,v}$ from Definition E.3.6. Now by (a–e), and the same proof of [Tho12, Theorem 10.1] (which assumes that $\Sigma^+ \cup \Sigma_{\ell}^+$ consists only of places split in F), we know that the global universal deformation ring $\mathsf{R}^{\mathrm{univ}}_{\mathscr{P}(\mathscr{D}_{\Sigma^+})}$ is a finite \mathscr{O} -module. Moreover, we have $\mu \equiv N \mod 2$. By (d,e), and the same proof of [Gee11, Lemma 5.1.3] (which assumes that $\Sigma^+ \cup \Sigma_{\ell}^+$ consists only of places split in F), we know that the Krull dimension of $\mathsf{R}^{\mathrm{univ}}_{\mathscr{P}(\mathscr{D}_{\Sigma^+})}$ is at least one. Thus, $\mathsf{R}^{\mathrm{univ}}_{\mathscr{P}(\mathscr{D}_{\Sigma^+})}[1/\ell]$ is nonzero. By choosing a $\overline{\mathbb{Q}}_{\ell}$ -point of Spec $\mathsf{R}^{\mathrm{univ}}_{\mathscr{P}(\mathscr{D}_{\Sigma^+})}[1/\ell]$, we obtain a relevant representation $\Pi(\mathscr{D}_{\Sigma^+})$ of $\mathrm{GL}_N(\mathbb{A}_F)$ satisfying

- O $\Pi(\mathscr{D}_{\Sigma^+})$ is unramified outside Σ^+ ;
- O for every place w of F above Σ^+ , there is an open compact subgroup U_w of $\operatorname{GL}_N(F_w)$ depending only on Π_w , such that $\Pi(\mathscr{D}_{\Sigma^+})^{U_w}_w \neq \{0\}$;
- O $\rho_{\Pi(\mathscr{D}_{\Sigma^+}),\iota_{\ell}}$ and $\rho_{\Pi,\lambda} \otimes_{E_{\lambda}} \overline{\mathbb{Q}}_{\ell}$ are residually isomorphic.

In fact, the second property is a consequence of Proposition E.5.7. Note that there are only finitely relevant representations of $\operatorname{GL}_N(\mathbb{A}_F)$ up to isomorphism, satisfying the first two properties. By the strong multiplicity one property for GL_N [PS79], we know that for ℓ large enough, Π is the only relevant representation of $\operatorname{GL}_N(\mathbb{A}_F)$ up to isomorphism, satisfying all the three properties.

Now we claim that for two different collections \mathscr{D}_{Σ^+} and \mathscr{D}'_{Σ^+} , the relevant representations $\Pi(\mathscr{D}_{\Sigma^+})$ and $\Pi(\mathscr{D}'_{\Sigma^+})$ are not isomorphic. Assuming this claim, then for ℓ large enough, we have only one collection, which is $\{\mathscr{D}_v^{\min} \mid v \in \Sigma^+\}$, that is, $\bar{r}_{\Pi,\lambda}$ is rigid for $(\Sigma_{\Pi}^+, \emptyset)$. The theorem is proved.

For the claim itself, we take a place $v \in \Sigma^+$. Then the local components of $\Pi(\mathscr{D}_{\Sigma^+})$ above v give rise to a continuous homomorphism $r \colon \Gamma_{F_v^+} \to \mathscr{G}_N(\overline{\mathbb{Q}}_\ell)$, which corresponds to a $\overline{\mathbb{Q}}_\ell$ -point x_r in Spec $\mathsf{R}^{\mathrm{loc}}_{\overline{r}_{\Pi,\lambda,v}}[1/\ell]$. Now the dimension of the tangent space of Spec $\mathsf{R}^{\mathrm{loc}}_{\overline{r}_{\Pi,\lambda,v}}[1/\ell]$ at x_r is equal to

$$N^{2} + \dim_{\overline{\mathbb{Q}}_{\ell}} \mathrm{H}^{1}(F_{v}^{+}, \operatorname{ad} r) - \dim_{\overline{\mathbb{Q}}_{\ell}} \mathrm{H}^{0}(F_{v}^{+}, \operatorname{ad} r) = N^{2} + \dim_{\overline{\mathbb{Q}}_{\ell}} \mathrm{H}^{2}(F_{v}^{+}, \operatorname{ad} r)$$
$$= N^{2} + \dim_{\overline{\mathbb{Q}}_{\ell}} \mathrm{H}^{0}(F_{v}^{+}, (\operatorname{ad} r)(1)) \leq N^{2} + \dim_{\overline{\mathbb{Q}}_{\ell}} \mathrm{H}^{0}(F_{w}, (\operatorname{ad} r^{\natural})(1)),$$

where w is the place of F induced by the embedding $F \hookrightarrow \overline{F}_v^+$. However, since $\Pi(\mathscr{D}_{\Sigma^+})_w$ is generic, we have $\dim_{\overline{\mathbb{Q}}_\ell} \operatorname{H}^0(F_w, (\operatorname{ad} r^{\natural})(1)) = 0$ by [BLGGT14, Lemma 1.3.2(1)]. Thus, by Proposition E.5.11(1), Spec $\mathsf{R}^{\mathrm{loc}}_{\bar{r}_{\Pi,\lambda,v}}[1/\ell]$ is smooth at x_r , which implies that x_r can not lie on two irreducible components. The claim then follows.

Remark E.8.6. In fact, using the same proof, one can obtain Theorem E.8.4 for Π satisfying a weaker condition, namely, by asking Π to be regular algebraic rather than Definition 1.1.3(3).

References

- [AC89] J. Arthur and L. Clozel, Simple algebras, base change, and the advanced theory of the trace formula, Annals of Mathematics Studies, vol. 120, Princeton University Press, Princeton, NJ, 1989. MR1007299 ↑138, 206
- [BLGG11] T. Barnet-Lamb, T. Gee, and D. Geraghty, *The Sato-Tate conjecture for Hilbert modular forms*, J. Amer. Math. Soc. **24** (2011), no. 2, 411–469, DOI 10.1090/S0894-0347-2010-00689-3. MR2748398 ↑204
- [BLGGT14] T. Barnet-Lamb, T. Gee, D. Geraghty, and R. Taylor, Potential automorphy and change of weight, Ann. of Math. (2) 179 (2014), no. 2, 501–609, DOI 10.4007/annals.2014.179.2.3. MR3152941 ↑119, 208
 - [BG19] R. Bellovin and T. Gee, G-valued local deformation rings and global lifts, Algebra Number Theory 13 (2019), no. 2, 333–378, DOI 10.2140/ant.2019.13.333. MR3927049 ↑195
 - [BBM82] P. Berthelot, L. Breen, and W. Messing, Théorie de Dieudonné cristalline. II, Lecture Notes in Mathematics, vol. 930, Springer-Verlag, Berlin, 1982 (French). MR667344 ↑33
 - [BD05] M. Bertolini and H. Darmon, Iwasawa's main conjecture for elliptic curves over anticyclotomic Z_pextensions, Ann. of Math. (2) 162 (2005), no. 1, 1–64, DOI 10.4007/annals.2005.162.1. MR2178960 ↑5
 - [BPLZZ] R. Beuzart-Plessis, Y. Liu, W. Zhang, and X. Zhu, Isolation of cuspidal spectrum, with application to the Gan-Gross-Prasad conjecture. arXiv:1912.07169. ↑6, 137
 - [BK90] S. Bloch and K. Kato, *L*-functions and Tamagawa numbers of motives, The Grothendieck Festschrift, Vol. I, Progr. Math., vol. 86, Birkhäuser Boston, Boston, MA, 1990, pp. 333–400. MR1086888 (92g:11063) ↑16
 - [BR06] C. Bonnafé and R. Rouquier, On the irreducibility of Deligne-Lusztig varieties, C. R. Math. Acad. Sci. Paris 343 (2006), no. 1, 37–39, DOI 10.1016/j.crma.2006.04.014 (English, with English and French summaries). MR2241956 ↑144, 146
 - [Boo19] J. Booher, Minimally ramified deformations when $\ell \neq p$, Compos. Math. **155** (2019), no. 1, 1–37, DOI 10.1112/S0010437X18007546. MR3875451 \uparrow 186, 194, 195
 - [BP] J. Booher and S. Patrikis, G-valued Galois deformation rings when $\ell \neq p$. arXiv:1708.07434. \uparrow 195
 - [Car12] A. Caraiani, Local-global compatibility and the action of monodromy on nearby cycles, Duke Math. J. 161 (2012), no. 12, 2311–2413, DOI 10.1215/00127094-1723706. MR2972460 ↑26
 - [Car14] _____, Monodromy and local-global compatibility for l = p, Algebra Number Theory 8 (2014), no. 7, 1597–1646, DOI 10.2140/ant.2014.8.1597. MR3272276 \uparrow 26
 - [CS17] A. Caraiani and P. Scholze, On the generic part of the cohomology of compact unitary Shimura varieties, Ann. of Math. (2) 186 (2017), no. 3, 649–766, DOI 10.4007/annals.2017.186.3.1. MR3702677 ↑3, 171, 172, 173
 - [CS] ____, On the generic part of the cohomology of non-compact unitary Shimura varieties. arXiv:1909.01898. ↑171
 - [Car94] H. Carayol, Formes modulaires et représentations galoisiennes à valeurs dans un anneau local complet, p-adic monodromy and the Birch and Swinnerton-Dyer conjecture (Boston, MA, 1991), Contemp. Math., vol. 165, Amer. Math. Soc., Providence, RI, 1994, pp. 213–237, DOI 10.1090/conm/165/01601, (to appear in print) (French). MR1279611 ↑202
 - [Cas80] W. Casselman, The unramified principal series of p-adic groups. I. The spherical function, Compositio Math. 40 (1980), no. 3, 387–406. MR571057 ↑166, 167, 169
 - [CH13] G. Chenevier and M. Harris, Construction of automorphic Galois representations, II, Camb. J. Math. 1 (2013), no. 1, 53–73, DOI 10.4310/CJM.2013.v1.n1.a2. MR3272052 ↑26, 27, 173, 202
 - [Clo90] L. Clozel, Motifs et formes automorphes: applications du principe de fonctorialité, Automorphic forms, Shimura varieties, and L-functions, Vol. I (Ann Arbor, MI, 1988), Perspect. Math., vol. 10, Academic Press, Boston, MA, 1990, pp. 77–159 (French). MR1044819 ↑25

- [CHT08] L. Clozel, M. Harris, and R. Taylor, Automorphy for some l-adic lifts of automorphic mod l Galois representations, Publ. Math. Inst. Hautes Études Sci. 108 (2008), 1–181, DOI 10.1007/s10240-008-0016-1. With Appendix A, summarizing unpublished work of Russ Mann, and Appendix B by Marie-France Vignéras. MR2470687 ↑7, 18, 176, 177, 178, 179, 180, 182, 183, 188, 189, 192, 194, 195, 200, 203
 - [CT14] L. Clozel and J. A. Thorne, Level-raising and symmetric power functoriality, I, Compos. Math. 150 (2014), no. 5, 729–748, DOI 10.1112/S0010437X13007653. MR3209793 ↑139
 - [CT15] _____, Level raising and symmetric power functoriality, II, Ann. of Math. (2) **181** (2015), no. 1, 303–359, DOI 10.4007/annals.2015.181.1.5. MR3272927 ↑139
 - [CT17] _____, Level-raising and symmetric power functoriality, III, Duke Math. J. 166 (2017), no. 2, 325–402, DOI 10.1215/00127094-3714971. MR3600753 ↑139
- [Cur79] C. W. Curtis, Representations of finite groups of Lie type, Bull. Amer. Math. Soc. (N.S.) 1 (1979), no. 5, 721–757, DOI 10.1090/S0273-0979-1979-14648-2. MR537625 ↑164
- [dJ93] A. J. de Jong, The moduli spaces of principally polarized abelian varieties with Γ₀(p)-level structure, J. Algebraic Geom. 2 (1993), no. 4, 667–688. MR1227472 ↑33
- [DL76] P. Deligne and G. Lusztig, Representations of reductive groups over finite fields, Ann. of Math. (2) 103 (1976), no. 1, 103–161, DOI 10.2307/1971021. MR0393266 ↑164, 165
- [FL82] J.-M. Fontaine and G. Laffaille, Construction de représentations p-adiques, Ann. Sci. École Norm. Sup. (4) 15 (1982), no. 4, 547–608 (1983) (French). MR707328 ↑180, 181
- [FK88] E. Freitag and R. Kiehl, *Étale cohomology and the Weil conjecture*, Ergebnisse der Mathematik und ihrer Grenzgebiete (3) [Results in Mathematics and Related Areas (3)], vol. 13, Springer-Verlag, Berlin, 1988. Translated from the German by Betty S. Waterhouse and William C. Waterhouse; With an historical introduction by J. A. Dieudonné. MR926276 ↑87
- [GGP12] W. T. Gan, B. H. Gross, and D. Prasad, Symplectic local root numbers, central critical L values, and restriction problems in the representation theory of classical groups, Astérisque 346 (2012), 1–109 (English, with English and French summaries). Sur les conjectures de Gross et Prasad. I. MR3202556 ↑3, 6
- [Gee11] T. Gee, Automorphic lifts of prescribed types, Math. Ann. **350** (2011), no. 1, 107–144, DOI 10.1007/s00208-010-0545-z. MR2785764 ↑208
- [GJ76] S. Gelbart and H. Jacquet, A relation between automorphic forms on GL(2) and GL(3), Proc. Nat. Acad. Sci. U.S.A. 73 (1976), no. 10, 3348–3350. MR0412156 ↑139
- [Gol95] D. Goldberg, *R*-groups and elliptic representations for unitary groups, Proc. Amer. Math. Soc. **123** (1995), no. 4, 1267–1276, DOI 10.2307/2160730. MR1224616 ↑167
- [Gro91] B. H. Gross, Kolyvagin's work on modular elliptic curves, L-functions and arithmetic (Durham, 1989), London Math. Soc. Lecture Note Ser., vol. 153, Cambridge Univ. Press, Cambridge, 1991, pp. 235–256, DOI 10.1017/CBO9780511526053.009, (to appear in print). MR1110395 ↑15
- [Gur99] R. M. Guralnick, Small representations are completely reducible, J. Algebra 220 (1999), no. 2, 531–541, DOI 10.1006/jabr.1999.7963. MR1717357 ↑15
- [HT01] M. Harris and R. Taylor, The geometry and cohomology of some simple Shimura varieties, Annals of Mathematics Studies, vol. 151, Princeton University Press, Princeton, NJ, 2001. With an appendix by Vladimir G. Berkovich. MR1876802 ↑206, 207
- [HM78] R. Hotta and K. Matsui, On a lemma of Tate-Thompson, Hiroshima Math. J. 8 (1978), no. 2, 255–268. MR0486178 ↑71, 144, 164
- [Ito05] T. Ito, Weight-monodromy conjecture for p-adically uniformized varieties, Invent. Math. 159 (2005), no. 3, 607–656, DOI 10.1007/s00222-004-0395-y. MR2125735 ↑207
- [Jan14] C. Jantzen, Tempered representations for classical p-adic groups, Manuscripta Math. 145 (2014), no. 3-4, 319–387, DOI 10.1007/s00229-014-0679-5. MR3268853 ↑163
- [KMSW] T. Kaletha, A. Minguez, S. W. Shin, and P.-J. White, Endoscopic Classification of Representations: Inner Forms of Unitary Groups. arXiv:1409.3731. ↑162, 170
- [Kat81] N. Katz, Serre-Tate local moduli, Algebraic surfaces (Orsay, 1976), Lecture Notes in Math., vol. 868, Springer, Berlin-New York, 1981, pp. 138–202. MR638600 ↑30
- [KT17] C. B. Khare and J. A. Thorne, Potential automorphy and the Leopoldt conjecture, Amer. J. Math. 139 (2017), no. 5, 1205–1273, DOI 10.1353/ajm.2017.0030. MR3702498 ↑202
- [Kim03] H. H. Kim, Functoriality for the exterior square of GL₄ and the symmetric fourth of GL₂, J. Amer. Math. Soc. 16 (2003), no. 1, 139–183, DOI 10.1090/S0894-0347-02-00410-1. With appendix 1 by Dinakar Ramakrishnan and appendix 2 by Kim and Peter Sarnak. MR1937203 ↑139

- [KS02] H. H. Kim and F. Shahidi, Functorial products for GL₂×GL₃ and the symmetric cube for GL₂, Ann. of Math. (2) **155** (2002), no. 3, 837–893, DOI 10.2307/3062134. With an appendix by Colin J. Bushnell and Guy Henniart. MR1923967 ↑139
- [Kis09] M. Kisin, Lectures on deformations of Galois representations (2009). http://www.math.harvard. edu/~kisin/notes/notes.pdf, preprint. ↑120, 121
- [KSZ] M. Kisin, S. W. Shin, and Y. Zhu, The stable trace formula for certain Shimura varieties of abelian type. in preparation. ↑27
- [Kol90] V. A. Kolyvagin, Euler systems, The Grothendieck Festschrift, Vol. II, Progr. Math., vol. 87, Birkhäuser Boston, Boston, MA, 1990, pp. 435–483. MR1106906 (92g:11109) ↑5
- [Kon03] T. Konno, A note on the Langlands classification and irreducibility of induced representations of p-adic groups, Kyushu J. Math. 57 (2003), no. 2, 383–409, DOI 10.2206/kyushujm.57.383. MR2050093 ↑162, 167, 169
- [KR14] S. Kudla and M. Rapoport, Special cycles on unitary Shimura varieties II: Global theory, J. Reine Angew. Math. 697 (2014), 91–157, DOI 10.1515/crelle-2012-0121. MR3281653 ↑45
- [Lab99] J.-P. Labesse, Cohomologie, stabilisation et changement de base, Astérisque 257 (1999), vi+161 (French, with English and French summaries). Appendix A by Laurent Clozel and Labesse, and Appendix B by Lawrence Breen. MR1695940 ↑172, 174
 - [Lab] _____, Changement de base CM et séries discrètes. https://www.imj-prg.fr/fa/bpFiles/ Labesse2.pdf, preprint. ↑27
- [Lan13] K.-W. Lan, Arithmetic compactifications of PEL-type Shimura varieties, London Mathematical Society Monographs Series, vol. 36, Princeton University Press, Princeton, NJ, 2013. MR3186092 ↑25
- [LS18] K.-W. Lan and B. Stroh, Nearby cycles of automorphic étale sheaves, Compos. Math. 154 (2018), no. 1, 80–119, DOI 10.1112/S0010437X1700745X. MR3719245 ↑10, 38, 59, 99
- [Liu02] Q. Liu, Algebraic geometry and arithmetic curves, Oxford Graduate Texts in Mathematics, vol. 6, Oxford University Press, Oxford, 2002. Translated from the French by Reinie Erné; Oxford Science Publications. MR1917232 ↑89
- [Liu07] T. Liu, Torsion p-adic Galois representations and a conjecture of Fontaine, Ann. Sci. École Norm. Sup. (4) 40 (2007), no. 4, 633–674, DOI 10.1016/j.ansens.2007.05.002 (English, with English and French summaries). MR2191528 ↑17
- [Liu16] Y. Liu, Hirzebruch-Zagier cycles and twisted triple product Selmer groups, Invent. Math. 205 (2016), no. 3, 693–780, DOI 10.1007/s00222-016-0645-9. MR3539925 ↑3, 5, 11, 12, 14, 15, 16, 17, 18, 19
- [Liu19] _____, Bounding cubic-triple product Selmer groups of elliptic curves, J. Eur. Math. Soc. (JEMS) **21** (2019), no. 5, 1411–1508, DOI 10.4171/JEMS/865. MR3941496 ↑5, 11, 13, 106, 107, 116, 125, 126
 - [Liu] _____, Fourier-Jacobi cycles and arithmetic relative trace formula (with an appendix by Chao Li and Yihang Zhu). https://users.math.yale.edu/~yl2269/FJcycle.pdf, preprint. \27, 121
 - [LT] Y. Liu and Y. Tian, Supersingular locus of Hilbert modular varieties, arithmetic level raising, and Selmer groups. https://users.math.yale.edu/~yl2269/supersingular_locus.pdf, preprint. ↑5
- [Lus76] G. Lusztig, On the finiteness of the number of unipotent classes, Invent. Math. 34 (1976), no. 3, 201–213, DOI 10.1007/BF01403067. MR419635 ↑185
- [Mes72] W. Messing, The crystals associated to Barsotti-Tate groups: with applications to abelian schemes, Lecture Notes in Mathematics, Vol. 264, Springer-Verlag, Berlin-New York, 1972. MR0347836 ↑30
- [Mok15] C. P. Mok, Endoscopic classification of representations of quasi-split unitary groups, Mem. Amer. Math. Soc. 235 (2015), no. 1108, vi+248, DOI 10.1090/memo/1108. MR3338302 ↑162
- [Mor10] S. Morel, On the cohomology of certain noncompact Shimura varieties, Annals of Mathematics Studies, vol. 173, Princeton University Press, Princeton, NJ, 2010. With an appendix by Robert Kottwitz. MR2567740 ↑174
- [MP96] A. Moy and G. Prasad, Jacquet functors and unrefined minimal K-types, Comment. Math. Helv. 71 (1996), no. 1, 98–121, DOI 10.1007/BF02566411. MR1371680 ↑72
- [Nek07] J. Nekovář, On the parity of ranks of Selmer groups. III, Doc. Math. 12 (2007), 243–274. MR2350290 ↑17, 114
- [Niz93] W. Nizioł, Cohomology of crystalline representations, Duke Math. J. 71 (1993), no. 3, 747–791, DOI 10.1215/S0012-7094-93-07128-1. MR1240602 ↑14
- [PS79] I. I. Piatetski-Shapiro, Multiplicity one theorems, Automorphic forms, representations and L-functions (Proc. Sympos. Pure Math., Oregon State Univ., Corvallis, Ore., 1977), Proc. Sympos. Pure Math., XXXIII, Amer. Math. Soc., Providence, R.I., 1979, pp. 209–212. MR546599 ↑26, 208
 - [Pra] D. Prasad, Notes on representations of finite groups of Lie type. arXiv:1404.0861. ↑165

- [RSZ] M. Rapoport, B. Smithling, and W. Zhang, Arithmetic diagonal cycles on unitary Shimura varieties. arXiv:1710.06962. ↑37
- [Rog90] J. D. Rogawski, Automorphic representations of unitary groups in three variables, Annals of Mathematics Studies, vol. 123, Princeton University Press, Princeton, NJ, 1990. MR1081540 ↑72, 162
- [Rog92] _____, Analytic expression for the number of points mod p, The zeta functions of Picard modular surfaces, Univ. Montréal, Montreal, QC, 1992, pp. 65–109. MR1155227 ↑27
- [Sai03] T. Saito, Weight spectral sequences and independence of l, J. Inst. Math. Jussieu 2 (2003), no. 4, 583–634, DOI 10.1017/S1474748003000173. MR2006800 ↑86
- [Sch18] P. Scholze, On the p-adic cohomology of the Lubin-Tate tower, Ann. Sci. Éc. Norm. Supér. (4) 51 (2018), no. 4, 811–863, DOI 10.24033/asens.2367 (English, with English and French summaries). With an appendix by Michael Rapoport. MR3861564 ↑118, 120
- [Ser72] J.-P. Serre, Propriétés galoisiennes des points d'ordre fini des courbes elliptiques, Invent. Math. 15 (1972), no. 4, 259–331 (French). MR0387283 (52 #8126) ↑21, 133, 205
- [Ser94] _____, Sur la semi-simplicité des produits tensoriels de représentations de groupes, Invent. Math. 116 (1994), no. 1-3, 513–530, DOI 10.1007/BF01231571 (French). MR1253203 ↑15, 138
- [Shi10] S. W. Shin, A stable trace formula for Igusa varieties, J. Inst. Math. Jussieu 9 (2010), no. 4, 847–895, DOI 10.1017/S1474748010000046. MR2684263 ↑172
- [Shi11] _____, Galois representations arising from some compact Shimura varieties, Ann. of Math. (2) 173 (2011), no. 3, 1645–1741, DOI 10.4007/annals.2011.173.3.9. MR2800722 ↑118, 172, 174
- [Shi] _____, On the cohomological base change for unitary similitude groups. https://math.berkeley. edu/~swshin/BCofU.pdf, an appendix to Galois representations associated to holomorphic limits of discrete series by W. Goldring. ↑27, 172, 173, 175, 176
- [Sho18] J. Shotton, The Breuil-Mézard conjecture when $l \neq p$, Duke Math. J. **167** (2018), no. 4, 603–678, DOI 10.1215/00127094-2017-0039. MR3769675 \uparrow 184, 185, 195
- [TY07] R. Taylor and T. Yoshida, Compatibility of local and global Langlands correspondences, J. Amer. Math. Soc. 20 (2007), no. 2, 467–493, DOI 10.1090/S0894-0347-06-00542-X. MR2276777 ↑114, 132
- [Tho12] J. Thorne, On the automorphy of l-adic Galois representations with small residual image, J. Inst. Math. Jussieu 11 (2012), no. 4, 855–920, DOI 10.1017/S1474748012000023. With an appendix by Robert Guralnick, Florian Herzig, Richard Taylor and Thorne. MR2979825 ↑7, 176, 200, 201, 202, 203, 204, 208
- [Tho14] J. A. Thorne, Raising the level for GL_n , Forum Math. Sigma 2 (2014), e16, 35, DOI 10.1017/fm-s.2014.14. MR3264255 \uparrow 207
 - [XZ] L. Xiao and X. Zhu, Cycles on Shimura varieties via geometric Satake. arXiv:1707.05700. ↑7, 47, 149, 153
 - [Yao] Z. Yao, The Breuil-Mézard conjecture for function fields. arXiv:1808.09433. ¹¹⁹
- [Zha14] W. Zhang, Fourier transform and the global Gan-Gross-Prasad conjecture for unitary groups, Ann. of Math. (2) 180 (2014), no. 3, 971–1049, DOI 10.4007/annals.2014.180.3.4. MR3245011 ↑6

DEPARTMENT OF MATHEMATICS, YALE UNIVERSITY, NEW HAVEN CT 06511, UNITED STATES *E-mail address*: yifeng.liu@yale.edu

UNIVERSITÉ DE STRASBOURG, STRASBOURG, FRANCE E-mail address: tian@math.unistra.fr

BEIJING INTERNATIONAL CENTER FOR MATHEMATICAL RESEARCH, PEKING UNIVERSITY, BEIJING 100871, CHINA

E-mail address: lxiao@bicmr.pku.edu.cn

DEPARTMENT OF MATHEMATICS, MASSACHUSETTS INSTITUTE OF TECHNOLOGY, CAMBRIDGE MA 02139, UNITED STATES

E-mail address: weizhang@mit.edu

DIVISION OF PHYSICS, MATHEMATICS AND ASTRONOMY, CALIFORNIA INSTITUTE OF TECHNOLOGY, PASADE-NA CA 91125, UNITED STATES

E-mail address: xzhu@caltech.edu